## **NEW JERSEY TRANSIT CORPORATION**

# NOTICE TO CONTRACTORS

Gladstone Line Pole Foundations Project

#### INVITATION FOR BID NO. 13-071X

Notice is hereby given that this is an electronic bid submission. Electronic bids will be received via the Internet until **2:00 P.M. on Thursday, July 10, 2014**, by NJ TRANSIT for the Gladstone Line Pole Foundations Replacement Project.

Bids for the project will be downloaded from the Bid Express website on the scheduled Bid Due Date (subject to change by Addenda) at 2:00:00 P.M., and publicly opened and read immediately thereafter at NJ TRANSIT's offices located at One Penn Plaza East, 6<sup>th</sup> floor Bid Room, Newark, NJ 07105. The Bidder must upload its bid prior to the hour named so that it is included in the project download. Late bids cannot be accepted. This is the only way to submit a bid for this project; paper bids will not be accepted.

The work under this Invitation for Bids generally consists of, but is not limited to, The general construction necessary to install one hundred sixty three (163) new concrete foundations and steel poles between New Providence Station and Gladstone Rail Yard on NJ TRANSIT's Gladstone Branch. In addition, the fabrication and installation/erection of new steel catenary poles will be required to support the existing overhead catenary system (OCS) and ancillary lines. The project involves civil work, site work, and concrete foundations as outlined in the Contract Documents. All work is to be conducted by means of high-rail equipment. All work must be in strict conformance with the drawings and specifications provided in the Bid Documents.

In order to obtain all pertinent Bid Documents, interested firms must register with BID EXPRESS at https://www.bidx.com. Contractors that are currently registered with BID EXPRESS to bid NJDOT or any other public construction project do not require further registration, only an additional digital ID. To subscribe, follow instructions on the website. Fees apply to downloading documents and plans and bidding access. The fee schedule is available on the Bid Express website. All fees are directly payable to Bid Express. In order to submit a bid, NJ TRANSIT recommends that Contractors apply for a digital ID at least seven business days prior to the Bid Due Date.

All firms must be prequalified by NJ TRANSIT prior to submitting a bid. Contractors must be prequalified for an amount of work that is equal to or greater than their bid amount. Contractors must be prequalified for "Heavy Construction-Rail "HR" (Operating Railroad Environment) with a classification of "GC"-General Contractor or a subcategory classification of "ET"-Catenary /Electrification Contractor, for an amount equal to or exceeding their bid amount. NJTRANSIT suggests a minimum rating of "O" \$10,000,001 to \$15,000,000. Prequalification questionnaires are available for download from the Bid Express website <a href="www.bidx.com">www.bidx.com</a>. Completed prequalification forms **MUST** be submitted to NJTRANSIT's Bid Desk at least **fourteen (14)** days prior to the scheduled bid opening date <a href="but NJ TRANSIT encourages Prequalification Questionnaire submissions as early as possible">but NJ TRANSIT encourages Prequalification Questionnaire submissions as early as possible</a>. Late prequalification submissions **MAY NOT** be considered.

A pre-bid conference has been scheduled for Wednesday, June18, 2014 at 10:00A.M., at New Jersey Transit's Headquarters located at (intersection of Raymond Boulevard and Penn Plaza East), Sixth Floor Bid Room, Newark, New Jersey. There is no site tour scheduled for this project. Bidders are advised that attendance at both the pre-bid conference and site visit is strongly recommended.

This project is subject to the requirements of the Public Works Contractor Registration Act (N.J.S.A. 34:56-48 et seq). Contractors and subcontractors are also required to comply with the State of New Jersey, Division of Revenue, Business Registration Certificate requirements (N.J.S.A. 52:32-44.).

Bidders are required to comply with the requirements of N.J.S.A. 10:5-31et seq. and N.J.A.C. 17:27, regarding Equal Employment Opportunity Laws and Regulations. **Disadvantaged Business Enterprises, in accordance with the Department of Transportation (DOT) Regulations 49 CFR, Part 26, shall have the maximum opportunity to participate in the performance of this contract.** The Race Conscious DBE goal for this project has been set at **twenty percent (20%).** 

All bids, proposals, contracts, laws and related documentation will be subject to the financial assistance contract between the project sponsor and the U.S. Department of Transportation, under the Federal Transit Act of 1964, as amended, based upon availability of funds.

In accordance with N.J.S.A. 27:25-11(b)(3) and N.J.A.C. 16:72-2.2 bidders are notified that a bid bond in the amount of 10% of the bid price is required with their bids.

# New Jersey Transit Corporation Instructions To Bidders For Construction

# **Table of Contents**

A.	ABB	REVIATIONS AND DEFINITIONS	1
	A.1	ABBREVIATIONS	1
	A.2	DEFINITIONS	1
B.	COM	MUNICATIONS	1
	B.1	OBTAINING BID DOCUMENTS	1
	B.2	INQUIRIES AND REQUESTS FOR CLARIFICATION	1
	B.3	ACKNOWLEDGMENT OF RECEIPT OF ADDENDA	2
	B.4	IMPROPER COMMUNICATIONS	2
	B.5	PRE-BID CONFERENCE	2
C.	REG	ISTRATIONS, PREQUALIFICATION AND DBE CERTIFICATION	2
	C.1	CORPORATE REGISTRATION	2
	C.2	PREQUALIFICATION OF BIDDERS	3
	C.3.	DBE CERTIFICATION	4
	C.4	DEPARTMENT OF LABOR REGISTRATION	4
	C.5	OWNERSHIP DISCLOSURE	5
	C.6	DISCLOSURE OF POLITICAL CONTRIBUTIONS (N.J.S.A. 19:44A-20.27)	5
	C.7	BUSINESS REGISTRATION NOTICE	5
D.	ETH	ICAL REQUIREMENTS FOR BIDDERS	6
	D.1	CODE OF ETHICS FOR BIDDERS	6
	D.2	CERTIFICATION REGARDING DEBARMENT, SUSPENSION, INELIGIBILITY AND VOLUNTARY EXCLUSION - LOWER TIER COVERED TRANSACTION	6
	D.3	STATE DEBARMENT AND SUSPENSION CERTIFICATION	8
	D.4	LIMITATIONS ON LOBBYING	8
	D.5	DISCLOSURE OF INVESTMENT ACTIVITIES IN IRAN	9

# New Jersey Transit Corporation Instructions To Bidders For Construction

# **Table of Contents**

E.	BID	PREPARATION AND SUBMISSION PROCEDURES9
	E.1	EXAMINATION OF CONTRACT DOCUMENTS 9
	E.2	BID FORM QUANTITIES 9
	E.3	PREPARATION OF THE BID
	E.4	DBE FORMS AND CERTIFICATION
	E.5	BID GUARANTEE
	E.6	BID VALIDITY
	E.7	WITHDRAWAL OF BIDS
	E.8	SUBMISSION OF MULTIPLE BIDS
	E.9	SUBMISSION OF BIDS
F.	BID	OPENING
	F.1	PUBLIC OPENING OF BIDS
	F.2	UNOPENED BIDS
	F.3	CONDITIONALLY ACCEPTED
G.	IRR	EGULAR BIDS13
	G.1	IRREGULARITIES AND NON-CONFORMANCES
	G.2	RIGHT TO REJECT OR WAIVE
н.	AWA	ARD AND EXECUTION OF CONTRACT14
	H.1	CALCULATION OF BIDS
	H.2	SINGLE BIDS
	H.3	DETERMINATION OF LOW BID
	H.4	AWARD PROCESS
	H.5	CANCELLATION OF AWARD
	H.6	CONTRACT BONDS

# New Jersey Transit Corporation Instructions To Bidders For Construction

# **Table of Contents**

	H.7	EXECUTION OF CONTRACT	16
	H.8	FAILURE TO EXECUTE CONTRACT	16
	H.9	RELIEF OF BIDDERS	16
	H.10	RELEASE OF BID GUARANTEES	16
I.	PRO	TEST PROCEDURE	17
	1.1	PURPOSE	17
	12	PROCEDURE	17

#### A. ABBREVIATIONS AND DEFINITIONS

#### A.1 ABBREVIATIONS

See Article 1.1.1 of the General Provisions for Construction.

#### A.2 DEFINITIONS

See Article 1.1.2 of the General Provisions for Construction.

#### B. COMMUNICATIONS

## B.1 OBTAINING BID DOCUMENTS

This Project is being bid by use of an electronic bidding process. Electronic bidding information is available on NJ TRANSIT's electronic bidding website: <a href="www.bidx.com">www.bidx.com</a>. Registration and a subscription fee are required to access the Bid Documents. The Bidder shall download the bidding software. The Bidder shall not alter or in any way change the software.

The Bidder shall download all bid documents from NJ TRANSIT's electronic bidding website. NJ TRANSIT assumes no responsibility for errors or omissions in the downloaded documents except as specifically provided for in the Contract Documents. The Bidder shall address questions or problems with downloading or using the electronic files, not the requirements of the Contract, to:

NJ TRANSIT Bid Express Administrator

E-Mail: <u>e-bidding@njtransit.com</u>

or

Bid Express Customer Support

Tel: (352) 381-4888 Fax: (352) 381-4444

E-Mail: <a href="mailto:customer.support@bidx.com">customer.support@bidx.com</a>

#### B.2 INQUIRIES AND REQUESTS FOR CLARIFICATION

All inquiries and requests for clarifications regarding the Contract Documents shall be submitted by e-mail to the Contract Specialist identified in the Special Provisions. Such requests shall state the Bid number and name of Project. Any response that NJ TRANSIT may choose to make will be by a written addendum to the Bid. NJ TRANSIT will not be bound by any informal explanation, clarification, or interpretation, oral or written, by whomsoever made, that is not incorporated into an addendum to the Bid. All such Addenda will be posted on the electronic bidding website prior to the opening of Bids. It is the obligation of the Bidder to check the website for addenda. Certain Addenda will contain

1

Amendments. The Bidder shall ensure that the Schedule of Bid Items to be bid contains all applicable Amendments. NJ TRANSIT has the right to reject bids that do not contain all applicable Amendments to the Schedule of Bids Items to be bid. No response may be made by NJ TRANSIT to inquiries received less than ten (10) days prior to the scheduled Bid opening.

#### B.3 ACKNOWLEDGMENT OF RECEIPT OF ADDENDA

Bidders are required to acknowledge receipt of all Addenda on the Bid Form in the space provided. Failure to acknowledge receipt of all Addenda may cause the rejection of the Bid as non-responsive.

#### B.4 IMPROPER COMMUNICATIONS

In accordance with N.J.S.A. 52:34-10.1, Bidders are advised that communications with NJ TRANSIT that in any way relate to this Project shall only be conducted with the authorized representative of the Contracting Officer in NJ TRANSIT's Procurement Department. All other contacts, with the exception of contacting NJ TRANSIT's Business Diversity Office for matters related to SBE requirements, are strictly prohibited and are considered improper.

Bidders are advised that violation of this prohibition, as required under N.J.S.A. 52:34-10.1, may result in the removal of the Bidder from consideration for award of this Contract.

#### B.5 PRE-BID CONFERENCE

A pre-bid conference may be held with prospective Bidders to review Contract Documents and generally discuss the Project. The time and place will be specified in the Advertisement. Bidders are strongly urged to attend.

#### C. REGISTRATIONS, PREQUALIFICATION AND DBE CERTIFICATION

#### C.1 CORPORATE REGISTRATION

In accordance with N.J.S.A. 14A: 13-3, all Out-of-State companies contracting to do business with NJ TRANSIT must have a certificate of authority to do business in the State of New Jersey. If you are not authorized to transact business in the State of New Jersey, contact the Department of Treasury's Commercial Recording/Corporate Filing Unit at (609) 292-9292, or go to the State of New Jersey's Business Gateway Registry Services (NJBGS) website at <a href="https://www.state.nj.us/njbgs">www.state.nj.us/njbgs</a>. The relevant documents, contained in the New Jersey Complete Business Registration Package (NJ-REG), instructions and customer service assistance are available from the above locations. No contract will be entered into by NJ TRANSIT with a firm unless the firm has procured a certificate of authority from the Secretary of State to do business in New Jersey. In addition, all Out-Of State companies

must have a registered agent in the State of New Jersey upon whom process may be served.

#### C.2 PREQUALIFICATION OF BIDDERS

In accordance with N.J.A.C. 16:72-2.4, prospective Contractors, prior to bidding on capital improvement projects, must be prequalified as to the character and amount of work for which they are permitted to submit Bids. Prequalification shall be assigned Contractors based on all factors related to Contractors responsibility as set forth in N.J.A.C. 16:72-1.5, and any pertinent information relating to the qualification of Contractors. A Prequalification Classification and Rating shall be assigned Contractors based on information submitted by them in response to the Prequalification Questionnaire available through NJ TRANSIT's electronic bidding website. The Prequalification Classification and Rating shall be valid for thirty-six (36) months from the date of the Notice of Classification letter.

Please note that the Prequalification classification process will only be conducted for those firms subscribing to NJ TRANSIT's electronic bidding service, have downloaded Bid Documents from NJ TRANSIT's electronic bidding website <a href="www.bidx.com">www.bidx.com</a>, are intending to bid as a prime contractor or as a member of a joint venture, and have submitted a completed Prequalification Questionnaire.

Prospective bidders are advised that a key consideration in the evaluation, classification and rating of the Contractor is past experience with work similar both in nature and in magnitude to the Pregualification classification and rating requested by the Contractor.

Failure, either in the Prequalification Questionnaire or in any subsequent information provided by the prospective bidder, to demonstrate responsible experience with work of the nature and magnitude of the classification and rating requested may result in NJ TRANSIT assigning a different classification, lower rating, or both, or a rejection of the Prequalification request pending the submission of evidence of prior responsible experience.

Fully completed Prequalification Questionnaires must be received by NJ TRANSIT no later than fourteen (14) calendar days prior to the due date for receipt of bids. Failure of a prospective Bidder to submit its fully completed Prequalification Questionnaire on a timely basis may result in a denial of Prequalification for this Bid.

Firms seeking to submit bids as Joint Ventures are advised that both firms must submit Prequalification Questionnaires independently, in addition to a Statement of Joint Venture form executed by both firms. A rating and classification shall be assigned to each firm and

a determination shall be made as to the Joint Venture's eligibility to submit a bid in response to this Bid only.

Two or more firms which maintain an ongoing Joint Venture relationship may request Prequalification as a single entity, provided they can demonstrate a past history of responsible performance as a single entity. In such event, the Classification and Rating assigned to the entity shall be valid for thirty six (36) months.

#### C.3. DBE CERTIFICATION

Disadvantaged Business Enterprises, in accordance with the Department of Transportation (DOT) Regulations 49 CFR, Part 26, shall have the maximum opportunity to participate in the performance of this contract. The DBE subcontracting goals are identified in the Special Provisions for this Project.

#### C.4 DEPARTMENT OF LABOR REGISTRATION

Pursuant to The Public Works Contractor Registration Act, N.J.S.A. 34:11-56.48 et seq., Bidders are advised that Contractors shall not bid on any contract for public work as defined in N.J.S.A. 34:11-56.48 unless the Contractor is registered pursuant to the act. Contractors shall not list a Subcontractor in a bid for the Contract unless the Subcontractor is registered at the time the bid is made. Contractors or Subcontractors, including a Subcontractor not listed in the bid, shall not engage in the performance of any public work subject to the Contract, unless the Contractor or Subcontractor is registered pursuant to the act.

The Bidder shall possess a valid Public Works Contractor Registration Certificate at time of bid submission. Bidder is requested to provide its Public Works Registration Certificate number with bid submission. If requested, proof of valid registration shall be submitted to NJ TRANSIT, in care of the Contract Specialist, within seven (7) business days of NJ TRANSIT's request.

Each Contractor shall, after the bid is made and prior to the awarding of the contract, submit to NJ TRANSIT the certificates of registration for all Subcontractors listed in the bid. Applications for registration from either the Contractor or any Subcontractor shall not be accepted as a substitute for a certificate of registration.

Registration forms, copies of the Act and other relevant information may be obtained by contacting:

New Jersey Department of Labor and Workforce Development Division of Wage & Hour Compliance

P.O. Box 389

Trenton, New Jersey 08625-0389

Telephone: (609) 292-9464

Fax: (609) 633-8591

Companies not listed in the bid that quote successfully as Subcontractors at any tier will be required to submit evidence of registration with the Department of Labor as part of the post-award Subcontractor approval process.

#### C.5 OWNERSHIP DISCLOSURE

Pursuant to N.J.S.A. 52:25-24.2, in the event the Bidder is a corporation, partnership or sole proprietorship, the Bidder must complete a current, Ownership Disclosure Form prior to the receipt of the bid or accompanying the bid. Failure to submit the form will preclude the award of a contract.

#### C.6 DISCLOSURE OF POLITICAL CONTRIBUTIONS (N.J.S.A. 19:44A-20.27)

The Contractor is advised of its responsibility to file an annual disclosure statement on political contributions with the New Jersey Election Law Enforcement Commission (ELEC), pursuant to N.J.S.A. 19:44A-20.27, as amended, if the contractor receives contracts in excess of \$50,000 from a public entity in a calendar year. It is the contractor's responsibility to determine if filing is necessary. Failure to so file can result in the imposition of financial penalties by ELEC. Additional information about this requirement is available from ELEC at 888-313-3532 or at <a href="https://www.elec.state.nj.us">www.elec.state.nj.us</a>.

#### C.7 BUSINESS REGISTRATION NOTICE

In accordance with N.J.S.A. 52:32-44, all New Jersey and out of State business organizations must obtain a Business Registration Certificate (BRC) from the Department of the Treasury, Division of Revenue, prior to conducting business with NJ TRANSIT. Bidder is requested to provide its Business Registration Certificate number and Taxpayer Identification (FEIN) with bid submission. The Bidder shall possess a valid Business Registration Certificate prior to the time of contract award. The business registration form (Form NJ-REG) can be found online at http://www.state.nj.us/treasury/revenue/gettingregistered.htm#busentity.

No contract with a Subcontractor shall be entered into by any Contractor unless the Subcontractor first provides proof of valid business registration. In conjunction with the

Subcontractor approval process detailed in General Provision 1.9.2, the Contractor shall provide a copy of the Business Registration Certificate for each Subcontractor.

#### D. ETHICAL REQUIREMENTS FOR BIDDERS

#### D.1 CODE OF ETHICS FOR BIDDERS

Bidders are informed that it is NJ TRANSIT policy that Bidders who do or may do business with NJ TRANSIT must avoid all situations where proprietary or financial interest, or the opportunity for financial gain, could lead a NJ TRANSIT officer or employee to secure favored treatment for any organization or individual.

Bidders must also avoid circumstances and conduct which may not constitute actual wrongdoing, or conflict of interest, but might nevertheless appear questionable to the general public, thus compromising the integrity of NJ TRANSIT. All Bidders must comply with NJ TRANSIT's Code of Ethics as set forth in Article 10.4 of the General Provisions for Construction.

D.2 CERTIFICATION REGARDING DEBARMENT, SUSPENSION, INELIGIBILITY AND VOLUNTARY EXCLUSION – LOWER TIER COVERED TRANSACTION.

By signing and submitting this bid ("proposal"), the prospective lower tier participant, defined as the Contractor and its Subcontractors, is providing the certification set out below. The certification in this clause is a material representation of fact upon which reliance was placed when this transaction was entered into. If it is later determined that the prospective lower tier participant knowingly rendered an erroneous certification, in addition to other remedies available to the Federal Government, NJ TRANSIT may pursue available remedies, including suspension and/or debarment.

The prospective lower tier participant shall provide immediate written notice to NJ TRANSIT if at any time the prospective lower tier participant learns that its certification was erroneous when submitted or has become erroneous by reason of changed circumstances.

Certain terms used in this clause have the meanings set out in 2 CFR Part 1200 and 2CFR Part 180.

The prospective lower tier participant agrees by submitting this proposal that, should the proposed covered transaction be entered into, it shall not knowingly enter into any lower tier covered transaction with a person who is debarred, suspended, declared ineligible, or

voluntarily excluded from participation in this covered transaction, unless authorized in writing by NJ TRANSIT.

The prospective lower tier participant further agrees by submitting this proposal that it will include the clause titled "Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion – Lower Tier Covered Transaction", without modification, in all lower tier covered transactions (valued at \$25,000 or more) and in all solicitations for lower tier covered transactions.

A participant in a covered transaction may rely upon a certification of a prospective participant in a lower tier covered transaction that it is not debarred, suspended, ineligible, or voluntarily excluded from the covered transaction, unless it knows that the certification is erroneous. Each participant shall check the U.S. Government System for Award Management (SAM) database.

Nothing contained in the foregoing shall be construed to require establishment of a system of records in order to render in good faith the certification required by this clause. The knowledge and information of a participant is not required to exceed that which is normally possessed by a prudent person in the ordinary course of business dealings.

Except for transactions authorized under the fourth paragraph above, if a participant in a covered transaction knowingly enters into a lower tier covered transaction with a person who is suspended, debarred, ineligible, or voluntarily excluded from participation in this transaction, in addition to all remedies available to the Federal Government, NJ TRANSIT may pursue available remedies including suspension and/or debarment.

The prospective lower tier participant certifies by submission of this bid or proposal that neither it nor its principals (as defined at 2 CFR 180.995) is presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participation in this transaction by any Federal department or agency. A participant may decide the method and frequency by which it determines the eligibility of its principals.

When the prospective lower tier participant is unable to certify to the statements in this certification, such prospective participant shall attach an explanation to this proposal.

The prospective lower tier participant shall also be currently registered and active with no exclusion on the U.S. Government System for Award Management (SAM) database.

#### D.3 STATE DEBARMENT AND SUSPENSION CERTIFICATION

The Bidder shall certify that neither it or its principals are included on the State of New Jersey, Department of the Treasury, Consolidated Debarment Report or on the State of New Jersey, Department of Labor and Workforce Development, Division of Wage and Hour Compliance, Prevailing Wage Debarment List.

.

If the Bidder is included on such report, he may not be eligible for Award of Contract.

#### D.4 LIMITATIONS ON LOBBYING

(APPLICABLE FOR FEDERALLY FUNDED PROJECTS VALUED OVER \$100,000)

By signing and submitting this bid or proposal, the prospective lower tier participant, defined as a Contractor or Subcontractor, is providing the signed certification set out below. This certification is a material representation of fact upon which reliance was placed when this transaction was made or entered into. Submission of this certification is a prerequisite for making or entering into this transaction imposed by Section 1352, Title 31, U.S. Code. Any person who fails to file or amend a declaration required to be filed or amended under Section 1352, Title 31, U.S. Code shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.

Bidder certifies, to the best of his or her knowledge and belief, that:

- (1) No Federal appropriated funds have been paid or will be paid, by or on behalf of the undersigned, to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with the awarding of any Federal Contract, the making of any Federal grant, the making of any Federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment, or modification of any Federal Contract, grant, loan, or cooperative agreement.
- (2) If any funds other than Federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with this Federal Contract, grant, loan, or cooperative agreement, the undersigned shall complete and submit "Disclosure of Lobbying Activities," in accordance with its instructions.

- (3) At the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the information contained in any declaration previously filed in accordance with D.4.1 or D.4.2 above, Bidders shall file an updated certification or declaration, as appropriate, in accordance with 31 USC 1352.
- (4) The undersigned shall require that the language of this certification be included in the award documents for all subawards at all tiers (including subcontracts, subgrants, and contracts under grants, loans, and cooperative agreements) and that all subrecipients shall certify and disclose accordingly.

#### D.5 DISCLOSURE OF INVESTMENT ACTIVITIES IN IRAN

Pursuant to N.J.S.A. 52: 32-55 et seq., a Bidder that, at the time of bid opening, is identified on a list created pursuant to such law by the New Jersey Department of the Treasury as a person or entity engaging in investment activities in Iran as described in such law, shall be ineligible to, and shall not, bid on or enter into a contract with NJ TRANSIT. As required by such law, the Bidder must complete the certification with its Bid to attest under penalty of perjury, that neither the person or entity nor any of its parents, subsidiaries or affiliates is identified on the Department of Treasury's Chapter 25 list as a person or entity engaging in investment activities in Iran. Failure to complete the certification will render the bid non-responsive.

#### E. BID PREPARATION AND SUBMISSION PROCEDURES

#### E.1 EXAMINATION OF CONTRACT DOCUMENTS

The Bidder shall examine carefully the Contract Documents, project site and conditions affecting the procurement. By submitting a Bid, the Bidder acknowledges that he has carefully examined the Contract Documents and project site, and has satisfied himself as to the conditions affecting the procurement. NJ TRANSIT assumes no responsibility for any conclusions or interpretations made by the Bidder on the basis of the information made available by NJ TRANSIT.

#### E.2 BID FORM QUANTITIES

E.2.1 The quantities stipulated are approximate and are given only as a basis for the comparison of bids. NJ TRANSIT does not expressly, or by implication, represent that the actual amount of work will correspond to the estimated quantities stipulated in the Bid Form.

NJ TRANSIT reserves the right to increase or decrease the amount of any class or portion of the Work or omit portions of the Work.

E.2.2 The Bid Form may include Pay I tems that have been designated as Allowances. Such items may or may not be shown on the Contract Drawings. NJ TRANSIT, in its sole discretion, may or may not incorporate Allowances into the Project.

#### E.3 PREPARATION OF THE BID

- E.3.1 The Bidder shall ensure that all Addenda, including Amendments to the Bid Form, are applied to the Bid and properly acknowledged. The Bidder shall insert the price for each Pay Item in the appropriate box provided under the column designated "Unit Price". The only entries permitted in the Bid Form will be the lump sum prices or unit prices for items that shall be bid. The software will perform all extensions of the unit prices and calculate the total bid amounts.
- E.3.2 Prices shall be provided for all items called for on the Bid Form. All bid prices must be quoted in numeric form. When the Bidder intends to bid zero dollars (\$0.00) for a Pay Item, the Bidder shall insert a "0" in the appropriate box under the "Unit Price" column. Where no figure is provided by the Bidder in the "Unit Price" column for one or more Contract Items, the Bid will be considered to be nonconforming and shall be rejected.
- E.3.4 Alternate bids for any item will not be considered unless specifically requested in the Bid.
- E.3.4 The Bidder shall check the bid before submission using the software. The Bidder shall select "tools" and then select "check bid" and ensure there are no errors prior to submitting the electronic bid. For bids submitted by Joint Ventures select "tools" from the software menu and mark the electronic bid as "Joint Bid." The Bidder may print a completed Proposal Form for their records after completing the bid.

#### E.4 DBE FORMS AND CERTIFICATION

E.4.1 The apparent low bidder and the second low bidder must submit DBE Forms A, A-1, A-2, B, C and D, if necessary, to NJ TRANSIT, in care of the Contract Specialist, within seven (7) business days of the date for receipt of bids. In addition, the apparent low bidder and second low bidder are required to submit DBE Certificates, as issued by the certifying agency, for each firm listed on DBE Form A to NJ TRANSIT, in care of the Contract Specialist, within seven (7) business days of the date for receipt of bids. Bidders are requested to review Forms carefully and complete all Forms in their entirety.

#### E.5 BID GUARANTEE

- E.5.1 Bids shall be accompanied by Bid Security in the form of a Bid Bond, Cashier's Check, Certified Check or irrevocable Letter of Credit. Cash is not considered an acceptable form of security.
- E.5.2 Whenever the Bid Security accompanying the Bid is a Bid Bond in electronic form, the Bidder shall complete the electronic bond form. The Bidder shall ensure that the Bid Bond is properly completed and furnished by a corporation or corporations authorized to issue surety bonds in the State of New Jersey and listed in the current U.S. Treasury Department Circular 570 as of the date for the receipt of Bids and made payable to the New Jersey Transit Corporation.

Whenever the Bid Security accompanying the Bid is a Bid Bond in non-electronic form, the Bidder shall ensure that the Bid Bond is properly completed and furnished by a corporation or corporations authorized to issue surety bonds in the State of New Jersey and listed in the current U.S. Treasury Department Circular 570 as of the date for the receipt of Bids and made payable to the New Jersey Transit Corporation. The Bid Bond shall be delivered to NJ TRANSIT's Bid Desk located at One Penn Plaza East, 6<sup>th</sup> Floor, Newark, New Jersey prior to the time and date set for the opening of bids. Failure to deliver such Bid Bond to NJ TRANSIT's Bid Desk prior to the time set for the opening of bids shall result in rejection of the bid as non-responsive.

Whenever the Bid Security is a Letter of Credit, it shall be issued by an "insured bank" within the meaning of the Act creating the Federal Deposit Insurance Corporation (12 U.S.C. 1811). When the Bid Security accompanying the Bid is a Cashier's Check or a Certified Check, it shall also be drawn on an account with an "insured bank" within the meaning of the Act creating the Federal Deposit Insurance Corporation (12 U.S.C. 1811). The Letter of Credit, Cashier's Check or Certified Check shall be delivered to NJ TRANSIT's Bid Desk located at One Penn Plaza East, 6<sup>th</sup> Floor, Newark, New Jersey prior to the time and date set for the opening of bids. Failure to deliver such Cashier's Check, Certified Check or irrevocable Letter of Credit to NJ TRANSIT's Bid Desk prior to the time set for the opening of bids shall result in rejection of the bid as non-responsive.

Cashier's Checks, Certified Checks or irrevocable Letters of Credit based on accounts with or guaranteed by persons, corporations or institutions other than those identified in the preceding paragraph shall not be considered acceptable Bid Security and the Bid shall be declared non-responsive.

E.5.4 The amount of the Bid Security shall equal ten percent (10%) of the total amount bid. A Bid shall be rejected as non-responsive if it is not accompanied by satisfactory Bid Security.

#### E.6 BID VALIDITY

NJ TRANSIT reserves the right to issue a Notice of Award to the successful Bidder, as determined by NJ TRANSIT, at any time for a period of ninety (90) days following the date of bid opening. If necessary, requests for extensions of the date for issuance of the notice of award may be made by NJ TRANSIT by written request to bidders. The bid shall remain valid through the date NJ TRANSIT fully executes the contract.

#### E.7 WITHDRAWAL OF BIDS

Bids may be withdrawn at any time prior to the time specified for the opening of Bids by using the "Tools" feature of the software, selecting "View Submitted Bids" and then selecting "Withdraw Bid." The withdrawal of a Bid does not prejudice the right of the Bidder to file a new Bid. Withdrawals received after the time specified for the opening of Bids will not be considered nor may any Bid be withdrawn after that time.

#### E.8 SUBMISSION OF MULTIPLE BIDS

Submission of more than one Bid from an individual, firm, partnership, corporation or combination thereof under the same or different names shall be cause for disqualification of the Bids submitted by such entities. Reasonable grounds for believing that any individual, firm, partnership, corporation, or combination thereof, is interested as a principal in more than one Bid for the procurement contemplated may cause the rejection of all Bids submitted by such individual, firm, partnership, corporation, or combination thereof.

#### E.9 SUBMISSION OF BIDS

- E.9.1 Once the Bidder has completed its bid, the Bidder shall submit the electronically signed bid via the Internet. The Bidder shall ensure delivery of its bid with all required components and attachments.
- E.9.2 The Bidder is solely responsible for any and all errors and for timely submission of the bid, all components thereof, and all attachments thereto, through the electronic bidding system; NJ TRANSIT assumes no responsibility for any claim arising from the failure of any Bidder or of the electronic delivery system to cause any bid, bid component, or attachment to not be delivered to NJ TRANSIT on or before the time set for bid opening.

#### F. BID OPENING

#### F.1 PUBLIC OPENING OF BIDS

Bids will be publicly opened and read aloud at the advertised time and place set for such Bid opening, unless NJ TRANSIT extends the time for opening of Bids. NJ TRANSIT reserves the right at its sole discretion to extend the time for opening of Bids at any time prior to the opening of the Bids. Their content will be made public for the information of Bidders and other interested parties.

## F.2 UNOPENED BIDS

NJ TRANSIT may reject a bid and leave the bid unopened, if the Bidder has not been prequalified in accordance with Paragraph C.2, above, or for other material violations of NJ TRANSIT's bidding requirements.

#### F.3 CONDITIONALLY ACCEPTED

All Bids opened by NJ TRANSIT shall be considered conditionally accepted pending NJ TRANSIT's detailed review and examination of the Bids.

#### G. IRREGULAR BIDS

## G.1 IRREGULARITIES AND NON-CONFORMANCES

- A. Bids will be considered irregular and shall be rejected for the following reasons:
  - 1. If the Bid materially fails to conform to the requirements of the Invitation for Bids.
  - 2. If the Bid is received from a firm which is not adequately prequalified or is otherwise determined to be not responsible.
  - 3. If the Bid is not received by the specified date and time.
  - 4. If the Bidder fails to furnish Bid Security in accordance with Paragraph E.5.
  - 5. If the Bidder takes material exception to any of the Bid requirements.
  - 6. If the Bid is not digitally signed by the authorized representative of the Bidder.
  - 7. If the Bidder fails to deliver before receipt of the bid or with the bid, the Ownership Disclosure Form as required by N.J.S.A. 52:25-24.2.
  - 8. If the Bidder fails to complete the Disclosure of Investment Activities in Iran certification.
- B. Bids will be considered irregular and may be rejected for the following reasons:
  - 1. If the Bidder fails to comply with all material EEO/DBE requirements.
  - 2. Subject to Paragraph G.2, if the Bid is not properly completed.
  - 3. If all addenda are not properly acknowledged.
  - 4. If the Bid contains prices that are materially unbalanced.

5. For other reasons, if NJ TRANSIT deems it advisable to do so in the public interest.

#### G.2 RIGHT TO REJECT OR WAIVE

NJ TRANSIT, in its sole discretion, reserves the right to reject any and all Bids and to waive informalities and minor irregularities in Bids received, notwithstanding other provisions of the Contract Documents.

#### H. AWARD AND EXECUTION OF CONTRACT

#### H.1 CALCULATION OF BIDS

- H.1.1 After the Bids are opened and read, they will be compared on the basis of the correctly determined summation of the correctly determined products of all the quantities for Contract Items shown in the Bid multiplied by the unit prices bid. The results of such comparisons will be made available to the public. Award will be made on the basis of the correct total Contract price.
- H.1.2 In the event an error is encountered with the software and a discrepancy exists between the unit price bid for any Contract Item and the extension shown for that item under the column of the Bid designated "Extension," the unit price shall govern.

#### H.2 SINGLE BIDS

If only one Bid is received in response to the Invitation for Bids, a detailed cost proposal may be requested of the single Bidder. A cost/price analysis and evaluation and/or audit may be performed of the cost proposal in order to determine if the price is fair and reasonable.

#### H.3 DETERMINATION OF LOW BID

The award of the firm-fixed-price Contract, if it is awarded, will be to that responsible Bidder whose bid, conforming to the Invitation for Bids, is lowest in price.

#### H.4 AWARD PROCESS

Contract Award is a process initiated by NJ TRANSIT when it issues the Notice of Award, includes the period when the Contractor executes and returns the Contract, and is completed when NJ TRANSIT executes the Contract.

#### H.5 CANCELLATION OF AWARD

NJ TRANSIT reserves the right to cancel the Award of any Contract before execution of the Contract by NJ TRANSIT, even if it has been executed by the Contractor, if NJ TRANSIT

deems such cancellation to be in its best interests. In no event will NJ TRANSIT have any liability for the cancellation of such award. The Contractor assumes sole risk and responsibility for expenses prior to execution of the Contract and shall not commence work until receipt of the written Notice to Proceed.

#### H.6 CONTRACT BONDS

The Bidder to whom the Contract is awarded shall furnish the following bonds in a form substantially similar to that provided by NJ TRANSIT.

- H.6.1 A Performance Bond equal to 100 percent of the Contract price to secure fulfillment of the Contractor's obligations specified in the Contract.
- H.6.2 A Payment Bond equal to 100 percent of the Contract price to protect firms or persons supplying labor or materials to the Contractor for the performance of work provided for in the Contract.
- H.6.3 Performance and Payment Bonds must be executed by surety companies licensed to do business in the State of New Jersey and listed in the U.S. Treasury Department Circular 570 as of the date for receipt of Bids.
- H.6.4 Provisions of the Performance and Payment Bonds shall not limit any liability of the Contractor to NJ TRANSIT.
- H.6.5 The Performance and Payment Bonds shall continue in full force and effect until receipt by NJ TRANSIT of Contractor affidavits of payments, debts, claims and until final Acceptance of the Work.
- H.6.6 All alterations, extensions of time, extra and additional work, and other changes authorized by the Contract Documents may be made without securing the consent of the Surety or Sureties on the Performance Bond and Payment Bond.
- H.6.7 The Performance and Payment Bonds will be paid for at the lump sum price bid therefore or at the actual cost of the Bonds to the Contractor, whichever price may be lower. Payment for the Bonds will be made only upon delivery to the Engineer of a receipted bill for the Bonds.
- H.6.8 If any Surety upon any Bond furnished in connection with this Contract becomes unacceptable to NJ TRANSIT, the Contractor shall promptly furnish such additional security

as may be required from time to time to protect the interests of NJ TRANSIT and of persons supplying labor or materials in the prosecution of the Work contemplated by the Contract.

#### H.7 EXECUTION OF CONTRACT

Within ten (10) business days of receipt of a Notice of Award, the Bidder to whom the Contract is awarded shall deliver to NJ TRANSIT the Bonds, specified certificates and policies of insurance, and other specified documents required herein by NJ TRANSIT and shall properly execute two (2) copies of the Contract by signing the Contract Execution Form, as shown in Appendix C of the General Provisions. NJ TRANSIT will execute both copies of the Contract and will return one (1) copy to the Contractor.

H.7.1 This contract is subject to the provisions of N.J.S.A. 10:2-1 through 10:2-4 and N.J.S.A. 10:5-31 et seq. (P.L. 1975, c.127), and in accordance with the rules and regulations promulgated pursuant thereto, the Bidder agrees to comply with the following:

At the time the signed contract is returned to NJ TRANSIT, the said Bidder (contractor) shall submit to NJ TRANSIT and the New Jersey Department of Treasury, Division of Public Contracts Equal Employment Opportunity Compliance an Initial Project Workforce Report (Form AA-201) consisting of forms provided by NJ TRANSIT and completed by the Bidder (contractor) in accordance with Subchapter 7 of the regulations promulgated by the Treasurer pursuant to P.L. 1975, c.127. Bidders shall comply with NJ TRANSIT's General Provisions for Construction, Article 11.2 entitled "Equal Opportunity."

#### H.8 FAILURE TO EXECUTE CONTRACT

Failure of a Bidder to whom the Contract is awarded to execute the Contract or to file acceptable bonds or certificates and policies of insurance, as provided herein, shall be just cause for the annulment of the award and the forfeiture of such Bidder's Bid Security.

#### H.9 RELIEF OF BIDDERS

No Bidder shall be relieved of his Bid, nor shall any change be made in his Bid, because of a mistake, except as provided under applicable New Jersey Statutes and Administrative Codes.

### H.10 RELEASE OF BID GUARANTEES

Within thirty (30) working days of Bid opening, NJ TRANSIT will release or return, as appropriate, the Bid Security of all but the two lowest responsible Bidders. The Bid Security of the two lowest Bidders shall be released or returned upon execution and delivery of the

Contract and Performance and Payment Bonds by the Bidder to whom the Contract has been awarded as required by H.6.

#### I. PROTEST PROCEDURE

#### I.1 PURPOSE

This section describes the policies and procedures governing the receipt and resolution of Bidder protests in connection with an Invitation for Bid (IFB).

## I.2 PROCEDURE

#### I.2.1 Parties

Only an interested party may file a protest.

## 1.2.2 Types of Protests/Time Limits

- a. Protests based upon alleged restrictive specifications or alleged improprieties in NJ TRANSIT's procurement process must be filed no later than five (5) days prior to the bid opening date.
- b. Protests based upon alleged improprieties of a Bid shall be filed no later than five (5) days after the Protestor knows or should have known of the facts giving rise thereto.
- c. Protests based upon the award of a contract shall be filed no later than five (5) days after the notification to the unsuccessful firms of NJ TRANSIT's intent to award, or no later than five (5) days after an unsuccessful firm becomes aware of NJ TRANSIT's intent to award a contract, whichever comes first.
- d. All protests must be filed in writing. Oral protests will not be accepted.

#### I.2.3 Where To File

Protests must be filed directly with NJ TRANSIT's Contracting Officer at the address below and identifying the Invitation for Bid No.

Mr. James Schworn, Chief, Procurement and Support Services
NJ TRANSIT
One Penn Plaza East, Sixth Floor
Newark, NJ 07105
IFB NO.:

#### 1.2.4 The Protest

- a. The protest must contain the following information:
  - (i) The name, address and telephone number of the protestor.

- (ii) Identity of the IFB (by number and description).
- (iii) A statement of the specific grounds for protest and any supporting documentation. Additional materials in support of the protest will only be considered if filed within the time limits set in Paragraph 1.2.2.
- (iv) An indication of the ruling or relief desired from NJ TRANSIT.
- b. If the protest is filed before contract award, the potential contractor will be advised by NJ TRANSIT of the pending protest.
- c. If deemed appropriate by NJ TRANSIT, an informal conference on the merits of the protest may be conducted with all interested parties allowed to attend.

# I.2.5 Confidentiality of Protest

Material submitted by a protestor will not be withheld from any interested party, except to the extent that the withholding of information is permitted or required by law or regulation. If the protestor considers that the protest contains proprietary material which should be withheld, a statement advising of this fact must be affixed to the front page of the protest documents and the alleged proprietary information must be so identified wherever it appears.

## I.2.6 Response to the Protest

NJ TRANSIT's Contracting Officer, or designee, will respond to the protest within a reasonable time after receipt of the protest by NJ TRANSIT. NJ TRANSIT's response shall address only the issues raised originally by the protestor.

## I.2.7 Rebuttal to NJ TRANSIT Response

The protestor may submit a written rebuttal to NJ TRANSIT's response, addressed to the Contracting Officer, but must do so within five (5) days after receipt of the original NJ TRANSIT response. New issues in the rebuttal will not be addressed by NJ TRANSIT. After receipt of the protestor's rebuttal, the Contracting Officer will review the protest and notify the protestor of his final decision.

#### I.2.8 Request for Additional Information

Failure of the protestor to comply expeditiously with a request for information as specified by NJ TRANSIT's Contracting Officer, or designee, may result in determination of the protest without consideration of the additional information. If any parties to the protest request information from another party, the request shall be made to NJ TRANSIT's Contracting Officer, or designee, and shall be complied with by the other party within five (5) days if NJ TRANSIT so directs.

#### 1.2.9 Request for Reconsideration

If data becomes available that were not previously known, or there has been an error of law, a protestor may submit a request for reconsideration of the protest. NJ TRANSIT's Contracting Officer will again review the protest considering all currently available information. The Contracting Officer's determination will be made within a reasonable period of time, and his decision will be considered final.

#### I.2.10 Procurement Process Status

Upon timely receipt of a protest, NJ TRANSIT will delay the opening of bids until after resolution of the protest for protests filed prior to the bid opening, or withhold award until after resolution of the protest for protests filed after bid opening. However, NJ TRANSIT may open bids or award a contract whenever NJ TRANSIT, at its sole discretion, determines that:

- a. The items or work to be procured are urgently required; or
- b. Delivery or performance will be unduly delayed by failure to make the award promptly or;

Failure to make prompt award will otherwise cause undue harm to NJ TRANSIT or the Federal Government.

#### 1.2.11 Federal Transit Administration (FTA) Involvement

Where procurements are funded by the FTA, the protestor may protest to the FTA only where the protest alleges that NJ TRANSIT failed to have or to adhere to its protest procedures, failed to review a complaint or protest, or violated a Federal law or regulation. Any protest to the FTA must be filed in accordance with FTA Circular 4220.1F.

## 1.2.12 Definitions Applicable to Instructions to Bidders, Section I, entitled "Protest Procedure"

- a. "Days" means working days.
- b. "File or Submit" means date of receipt by NJ TRANSIT's Contracting Officer.
- c. "Federal Law or Regulation" means any valid requirement imposed by Federal statue or regulation governing contracts awarded pursuant to a grant agreement between NJ TRANSIT and the FTA. This includes the requirements as stated in FTA Circular 4220.1F.
- d. "Contracting Officer" means the Chief of Procurement & Support Services or his designee as indicated in the solicitation document.
- e. "Interested Party" means all bidders/offers. It may also include a Subcontractor or supplier provided they have a substantial economic interest in a portion of the IFB.
- f. "Potential Contractor" means the bidder that is in line for award of the contract in the event that the protest is denied.

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Joseph D. Bertoni, Acting Board Chairman Veronique Hakim, Executive Director



July 31, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 13

To whom it may concern:

The following constitutes Addendum No. 13 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

# ITEM#I: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Question 1, Set 9, dated July 28, 2014 is from Pete Dunn of Railroad Construction Company, Incorporated (RCCI), Paterson, New Jersey.

RCCI Question 1) We are in receipt of Addendum No 11 which has attempted to answer our inquiry about the *grade of steel* to be used for the anchor bolts.

- Should it be F1554 **Grade 36** (Low carbon 36 ksi yield steel), or
- Should it be F1554 Grade 55 (High strength, low alloy, 55 ksi yield steel), or
- Should it be F1554 **Grade 105** (Alloy, heat treated, high strength, 105 ksi yield steel)?

# NJ TRANSIT RESPONSE: Anchor bolts shall conform to ASTM F1554 Grade 55 (High strength, low alloy, 55 ksi yield steel).

Question 1, Set 10, dated July 30, 2014 is from Pete Dunn of Railroad Construction Company, Incorporated (RCCI), Paterson, New Jersey. RCCI Question 1)

- Specification 16600, 4.01 MEASUREMENT, B. says "Measurement for each Cathodic Protection Location shall be made, and is part of this section.
- Item 0440 is for Cathodic Protection 36 EA.
- Drawing ST-920, DETAIL 1, shows 4 Magnesium Anodes protecting 1 Catenary Pole.

Our question is: when it says "Measurement for each Cathodic Protection *Location*", does that mean that 4 anodes constitute a Location, so that each catenary pole is 1 (one) Location and eligible for payment of 1 EA. Cathodic Protection? Or is it saying that each magnesium anode bag is a Location and since there are 4 bags = 4 (four) locations and eligible for payment of 4 EA. Cathodic Protection? <a href="NJ TRANSIT RESPONSE: Each catenary pole/foundation location that has a complete and operating cathodic protection system installed is considered one location.">Considered one location.</a>

The cathodic protection system includes anodes, cables, connectors, test stations and any other equipment required for complete and operating cathodic protection. Each magnesium anode bag is an item in the cathodic protection system.

# Refer to detail 1 and note 6 of ST-920.

# Refer also to Addendum 6 ,RESPONSES TO PROSPECTIVE BIDDERS QUESTIONS DMRSCI Set 3 Question 1dated 6.25.14.

Question 1, Set 2 dated July 30, 2014 is from John Leslie of JH REID General Contractor (JHRGC), South Plainfield, New Jersey

JHRGC Question 1) Has there been any discussions relative to any further postonements on this project? **NJ TRANSIT RESPONSE: No.** 

This concludes Addendum No. 13. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

Managing Contract Specialist

Robert Delitto

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

CC: R. Duran/K. Donovan/D. Maine/IFB File

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Joseph D. Bertoni, Acting Board Chairman Veronique Hakim, Executive Director



July 29, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 12 To whom it may concern:

The following constitutes Addendum No. 12 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

# ITEM#I: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Questions 1-2, Set 6, dated July 28, 2014 are from Mike Dunn of Daidone Electric, Inc. (DEI), Newark NJ:

DEI Question 1) Will there be line outages provided with the track outages? If so, will the train power, signal power and any other misc. lines in the area be taken out with an outage during the weekday/weekend track outages provided? NJ TRANSIT RESPONSE: When the Track outages are in place catenary and signal feeders impacted by construction will be denergized – refer to the Special Provisions of the contract.

DEI Question 2) What are the line heights from track or grade to the existing lines to remain during construction of the new foundations? NJ TRANSIT RESPONSE: Contractor can determine these heights based on the heights of respective steel support arms shown on the Erection Diagrams (ED), ST-022 to ST-514. Heights on ED are from top of rail unless shown otherwise.

Question 1, Set 7, dated July 28, 2014 is from Mike Dunn of Daidone Electric, Inc. (DEI), Newark NJ: DEI Question 1) If rock is encountered during the excavation of the cathodic protection holes how will this be paid? Will this be covered under the additional rock (Line 170)? NJ TRANSIT RESPONSE: Rock encountered during excavation for cathodic protection will not be considered separately for payment. Per note 7 of drawing ST-920, the installation depth of 25 ft. is typical, and may be less at rock locations. However, Contractor must ensure that the anode will produce the desired output.

This concludes Addendum No. 12. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

Managing Contract Specialist

Robert Delite

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

CC: R. Duran/K. Donovan/D. Maine/IFB File

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Veronique Hakim, Executive Director



July 25, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 11

To whom it may concern:

The following constitutes Addendum No. 11 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

# ITEM#I: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Question 1, Set 7, dated July 23, 2014 is from Pete Dunn of Railroad Construction Company, Incorporated (RCCI), Paterson, New Jersey.

RCCI Question 1)

Add 10, page 1 of 3, the table of OUTAGES still says

- Week-Ends –
- 6:00 a.m. Saturday thru
- 9:00 p.m. Sunday.

I think it was changed by previous addendum to

- 6:00 a.m. Saturday to 9:00 p.m. Saturday and
- 6:00 a.m. Sunday to 9:00 p.m. Sunday

#### NJTRANSIT RESPONSE: Yes see corrected SP 7 see below:

#### **Special Provisions SP 7- USE OF PREMISES (4.2)**

Revise SP 7, Paragraph 4 to read:

The window of time available for all construction and any required outages(s) or obstruction(s) will be at the following time periods:

• OUTAGES	2015	2016
<ul><li>Week days</li><li>10:00 a.m. thru 3:00 p.m.</li></ul>	July 6 August 28	July 5 – August 26
Week Ends –  6:00 a.m. Saturday to 9:00 p.m. Saturday and 6:00 a.m. Sunday to 9:00 p.m. Sunday	Jan 24 – November 15	January 23 - November 18

Question 1, Set 8, dated July 24, 2014 is from Pete Dunn of Railroad Construction Company, Incorporated (RCCI), Paterson, New Jersey.

RCCI Question 1) Drawing GP-004, upper left corner, <u>STRUCTURAL STEEL</u> notes, note 2 says "All structural bolts...ASTM A325, Type 1, SC..."

Drawing ST-92, upper right corner, TYPICAL ANCHOR BOLT DETAIL shows the anchor bolts as "ASTM F1554"

ASTM F1554 covers the following Grades of steel for the anchor bolts:

F1554 Grade 36 Low carbon, 6 ksi yield steel anchor bolts

F1554 Grade 55 High strength, low alloy, 55 ksi yield steel anchor bolts

F1554 Grade 105 Alloy, heat treated, high strength, 105 ksi yield steel anchor bolts

Our question is, what Grade of steel is required for the Anchor Bolts

NJ TRANSIT RESPONSE: Structural steel bolts shall be in conformance with ASTM A325 unless otherwise noted. Anchor bolts shall conform to ASTM F1554.

This concludes Addendum No. 11. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

**Managing Contract Specialist** 

Robert Delitte

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

CC: R. Duran/K. Donovan/D. Maine/IFB File

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Veronique Hakim, Executive Director



July 23, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 10

To whom it may concern:

The following constitutes Addendum No. 10 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

#### ITEM#I-The following Information applies to this Procurement:

- 1 <u>Bidders Proposal Forms have been revised</u> <u>E-Bid Procurement Cost Estimate Gladstone 14-7-22-AD 10 REVISED;</u>
- 2 Add Specifications Guidelines for Temporary Shoring EP-2 -NJ Transit Rail Operations;
- 3 <u>Drawings Nos. CA-001 thru CA-138 have been revised.</u> Refer to Access Roads- Conformed Submittal Layout April2007.;
- 4 Special Provisions SP 7- USE OF PREMISES (4.2)

Revise SP 7, Paragraph 4 to read:

The window of time available for all construction and any required outages(s) or obstruction(s) will be at the following time periods:

• OUTAGES	2015	2016
<ul><li>Week days</li><li>10:00 a.m. thru 3:00 p.m.</li></ul>	July 6 August 28	July 5 – August 26
<ul><li>Week Ends –</li><li>6:00 a.m. Saturday thru</li><li>9:00 p.m. Sunday</li></ul>	Jan 24 – November 15	January 23 - November 18

# ITEM#II: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Question 2, Set 5, dated July 15, 2014 is from Michael Dyshuk of Daidone Electric, Inc.(DEI), Newark NJ: DEI Question 2) Are there any vibratory restrictions while performing work on rail. NJ TRANSIT RESPONSE: Yes, Refer to the Guidelines for Temporary Shoring - EP-2 -NJ Transit Rail Operations is included as part of this addendum

Question 1, Set 3, dated July 16, 2014 is from Pete Dunn of Railroad Construction Company, Incorporated (RCCI), Paterson, New Jersey.

RCCI Question 1) Drawings CA-001 thru CA-138 contain an "ACCESS ROAD LEGEND."

The legend describes 3 conditions of the Access Road.

However there is no Access Road depicted on the drawings at all.

Would it be possible to include access road information on the drawings? NJ TRANSIT RESPONSE: Yes, access road information should be showing on the plans. Drawings will be updated accordingly with access road information. However for bid purposes, Drawings CA-001 to CA-138 from the Priority I

# Conformed set is being made available to be used as reference for bid purposes (Access Roads-Conformed Submittal Layout April2007) These drawings show the access road information.

Question 1, Set 4, dated July 16, 2014 is from Pete Dunn of Railroad Construction Company, Incorporated (RCCI), Paterson, New Jersey.

RCCI Question 1) Addendum No 4 at DEI Question 5) the response informs us that "Yes, sidings will be available."

May we interpret that as meaning that sidings at each of the Stations will be available? For example,

- at the Gladstone yard, may the Contractor use one or more of the storage tracks? <u>NJ TRANSIT</u> RESPONSE NO
- may the Contractor use the siding at the Far Hills Station? NJ TRANSIT RESPONSE NO
- may the Contractor use the siding at the Bernardsville Station? <u>NJ TRANSIT RESPONSE No in Station YES West of station Ferrante's crossing</u>
- may the Contractor use the siding between Sta. 1646+42 to 1631+80? NJ TRANSIT RESPONSE Lyons siding YES
- may the Contractor use the siding at the Stirling Station? <u>NJ TRANSIT RESPONSE NO</u>
- may the Contractor use the siding at the Berkeley Heights Station? NJ TRANSIT RESPONSE YES
- may the Contractor use the siding at the Murray Hill Station? NJ TRANSIT RESPONSE NO

At the siding locations which the Contractor may use, would NJT operate the switches during Contractor working times? **NJ TRANSIT RESPONSE**: **Response to follow via addendum.** 

Question 1, Set 5, dated July 17, 2014 is from Pete Dunn of Railroad Construction Company, Incorporated(RCCI), Paterson, New Jersey.

RCCI Question 1) Addendum # 7 Response to Question 1 confirms that "All load bearing members shall be A572 Gr 50."

But Structural Steel Specification 05500-3,

Part 2.01A states that "General requirements for Rolled Steel Plates. Shapes, and Bars shall conform to ASTM A36."

Part 2.01B states that "Carbon Steel Plates, Shapes, and Bars shall conform to ASTM A36,"

Does this mean to say that all columns, beams, and channels shall conform to A572 Gr. 50; and plates, bars, angles shall conform to ASTM A36? **NJ TRANSIT RESPONSE**: **YES.** 

Questions 1-2, Set 6, are dated July 17, 2014 are from Pete Dunn of Railroad Construction Company, Incorporated (RCCI), Paterson, New Jersey.

RCCI Question 1). Spec 02010-4 3.03, A, 4. "In some instances, material excavated during construction may be used along the right of way. Reuse of excavation spoils must be pre-approved by the Construction Manager." Could you please elaborate on what instances would prevent surplus soil from foundations being used along the right of way; or used near or at the point of excavation. NJ TRANSIT RESPONSE: The Bidder shall consider all soil to be disturbed as non-hazardous contaminated soil (NJDEP solid waste classification ID-27). All ID-27 contaminated soil shall be disposed of in accordance with all applicable federal, state and local standards and regulations. Therefore, only excavation soils deemed suitable with respect to geotechnical properties, may be re-used subject to NJ Transit approval.

RCCI Question 2). Bid Item 14 is "Drilled Foundation for Guy 3'-0" without Rock Socket", Quantity 2 EA. However Drawing ST-904, GUY FOUNDATION SCHEDULE shows 3 guy foundations without Rock Sockets. Similarly Item 15 "Drilled Foundation for Guy 3'-0" with Rock Socket", Quantity 2 EA; However Drawing ST-904, GUY FOUNDATION SCHEDULE shows 1 guy foundation with Rock Socket. Should the bid item quantities match the

guy foundation schedule quantities? NJ TRANSIT RESPONSE: <u>Bid item quantities now match the Guy</u> Foundation Schedule Quantities shown on sheet ST-904. Bidders proposal form has been revised.

This concludes Addendum No. 10. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

**Managing Contract Specialist** 

Robert Delitto

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

Revised Bidders Proposal FORM E-Bid Procurement Cost Estimate - Gladstone - 14-7-22-AD 10 REVISED

Revised Drawings: Access Roads- Conformed Submittal Layout April2007

New Technical Provisions: Guidelines for Temporary Shoring - EP-2 -NJ Transit Rail Operations

CC: R. Duran/K. Donovan/D. Maine/IFB File

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Veronique Hakim, Executive Director



July 17, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 9

To whom it may concern:

The following constitutes Addendum No. 9 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

# ITEM#I- The Bid Due Date has been revised. Electronic bids will be received via the Internet until 2:00:00 P.M., Thursday, August 7, 2014, by NJ TRANSIT.

# ITEM#II: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Question 1, Set 2, dated July 15, 2014 is from Luz Carrillo of Railroad Construction Company, Incorporated(RCCI), Paterson, New Jersey.

RCCI Question 1) We respectfully request an immediate two (2) week postponement of the date of receipt of bids, to Thursday, August 7, 2014. We submitted the Pre-qualification information before the deadline and as of this date, we have not received confirmation that the Joint Venture will be prequalified to submit a bid. Also, we need at least seven (7) business days to apply for a digital ID through Bid Express. NJ TRANSIT: REFER TO ITEM I ABOVE. In order to obtain all pertinent Bid Documents, interested Joint Venture firms must register with BID EXPRESS at https://www.bidx.com. Joint Venture firms that are currently registered with BID EXPRESS to bid NJDOT or any other public construction project do not require further registration, only an additional digital ID. To subscribe, follow instructions on the website. Fees apply to downloading documents and plans and bidding access. The fee schedule is available on the Bid Express website. All fees are directly payable to Bid Express. In order to submit a Joint Venture bid, NJ TRANSIT recommends that proposed Joint Venture Firms, whether approved or not, apply for a digital ID at least seven business days from the date of this Addendum.

Questions 1-4, Set 5, dated July 15, 2014 are from Michael Dyshuk of Daidone Electric, Inc.(DEI), Newark NJ: DEI Question 1) Line 170, Item C03-005-004.0 is for an additional 200 LF of additional rock drilling. If more than 200 LF of additional rock is encountered how will it be paid? NJ TRANSIT RESPONSE: Refer to General Provisions, Article 3 "CONTRACT CHANGES" along with Article 8; EXCAVATION AND DIFFERING SITE CONDITIONS and the Construction Specifications applicable to the foundations. The drawings, together with the supporting documents indicate the appropriate action in cases of different site conditions.

DEI Question 2) Are there any vibratory restrictions while performing work on rail. **NJ TRANSIT: Response to follow via addendum.** 

DEI Question 3) Project specifications calls out for 90 minutes for concrete pour time from time that concrete is batched (water is introduced). Is retarder an acceptable additive for concrete on this project and will an extension be allowed from the 90 minutes with the use of retarder per ACI specifications? <a href="Multiple-NJ TRANSIT: Yes. Curing retarding additives are acceptable for use assuming they are applied per the product manufacturer's specifications, and are applied per the applicable guidelines. The use of the selected product must be submitted for approval prior to use, and must be included in the concrete mix design. The 90 minute limit can be extended if the retarding additive is applied per ACI, and the workability of the concrete is

maintained at the slump levels specified. The Construction Manager, at his discretion, may reject any mixed concrete that is found to be unworkable or unsuitable for use.

DEI Question 4) Technical Specifications spec section 03300 Cast-In-Place Concrete 3.05.C.1 calls for concrete to be placed/deposited in horizontal layers not deeper than 24 inches. Can you please clarify what the intent is for this spec? NJ TRANSIT: The intent of this statement is to ensure the concrete is placed in a uniform and continuous manner such that "cold joints" are avoided, and the concrete is properly consolidated. The use of construction joints (cold joints) is not permitted.

Question 1, Set 3, dated July 16, 2014 is from Pete Dunn of Railroad Construction Company, Incorporated(RCCI), Paterson, New Jersey.

RCCI Question 1) Drawings CA-001 thru CA-138 contain an "ACCESS ROAD LEGEND."

The legend describes 3 conditions of the Access Road.

However there is no Access Rod depicted on the drawings at all.

Would it be possible to include access road information on the drawings? **NJ TRANSIT: Response to follow via addendum.** 

Question 1, Set 4, dated July 16, 2014 is from Pete Dunn of Railroad Construction Company, Incorporated(RCCI), Paterson, New Jersey.

RCCI Question 1) Addendum No 4 at DEI Question 5) the response informs us that "Yes, sidings will be available."

May we interpret that as meaning that sidings at each of the Stations will be available? For example.

- at the Gladstone yard, may the Contractor use one or more of the storage tracks?
- may the Contractor use the siding at the Far Hills Station?
- may the Contractor use the siding at the Bernardsville Station?
- may the Contractor use the siding between Sta. 1646+42 to 1631+80?
- may the Contractor use the siding at the Stirling Station?
- may the Contractor use the siding at the Berkeley Heights Station?
- may the Contractor use the siding at the Murray Hill Station?

#### NJ TRANSIT: Response to follow via addendum.

At the siding locations which the Contractor may use, would NJT operate the switches during Contractor working times? **NJ TRANSIT: Response to follow via addendum.** 

Question 1, Set 5, dated July 17, 2014 is from Pete Dunn of Railroad Construction Company, Incorporated(RCCI), Paterson, New Jersey.

RCCI Question 1) Addendum # 7 Response to Question 1 confirms that "All load bearing members shall be A572 Gr 50."

But Structural Steel Specification 05500-3,

Part 2.01A states that "General requirements for Rolled Steel Plates. Shapes, and Bars shall conform to ASTM A36."

Part 2.01B states that "Carbon Steel Plates. Shapes, and Bars shall conform to ASTM A36."

Does this mean to say that all columns, beams, and channels shall conform to A572 Gr. 50; and plates, bars, angles shall conform to ASTM A36? **NJ TRANSIT: Response to follow via addendum.** 

This concludes Addendum No.9. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,



Robert Delitto

Managing Contract Specialist
Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

CC: R. Duran/K. Donovan/D. Maine/IFB File

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Veronique Hakim, Executive Director



July 14, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 8

To whom it may concern:

The following constitutes Addendum No. 8 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

# ITEM#I: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Question 1, Set 2 dated July 9, 2014 is from Roy Friedlander, Case Foundation Company (CFC), Berwyn ,PA:

CFC Question 1) We have the following question regarding the Gladstone project: Special Provisions item SP.7, 4.2.11 Track Outages indicates that "the window of time for construction on weekends will be from 6:00AM Saturday thru 9:00PM on Sunday." This infers we will be able to work continuously for 39 hours on the weekends. On drawing GP-003, notes # 1 and 2 under "Local Ordinances" seem to differ from the above provision. Note #1 infers we will not have access from 9:00PM on Saturday to 6:00AM on Sunday. Note #2 further restricts our work due to noise restrictions. Please clarify when we will have track time and when we may work. Thanks. NJ TRANSIT Response: Noise restrictions will be in place. Window of time for weekend construction shall be 6:00 a.m. till 9:00 p.m. Saturday and Sunday. SP's will be revised and included in the conformed set.

This concludes Addendum No.8. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

Managing Contract Specialist

Robert Delite

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

CC: R. Duran/K. Donovan/D. Maine/IFB File

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Veronique Hakim, Executive Director



July 8, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 7

To whom it may concern:

The following constitutes Addendum No. 7 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

# ITEM#I: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Questions 1-10, Set 1 dated July 1, 2014 are from Justin Lukens, Project Engineer, Case Foundation Company (CFC), Berwyn ,PA:

CFC Question 1) Dwg GP-004 calls for all structural steel to be A572 Gr 50. Is that correct. **NJ TRANSIT RESPONSE: That is correct. All load bearing members shall be A572 Gr 50.** 

CFC Question 2) Dwg GP-004 calls for all structural to be Charpy impact tested. Is that correct. **NJ TRANSIT RESPONSE**: <u>That is correct.</u>

CFC Question 3) Specification 05500 section 3.02 sub section d. call for all shop fillet welds to be magnetic particle inspected. Is that correct or only Partial and complete penetration. **NJ TRANSIT RESPONSE:** <u>That is correct.</u>

CFC Question 4) What is the finish to be on the anchor bolt embedment plates. (Galv or Plain Finish). **NJ TRANSIT RESPONSE**: <u>Anchor embedment plate is to be galvanized</u>..

CFC Question 5) What is the material grade of the CB-1 cantilever bracket sag rod and turnbuckle size. NJ TRANSIT RESPONSE: <u>Material grade for the sag rod is A449.</u> Turnbuckle size is 1" with 18" of take-up.

CFC Question 6) Are the CB-1 cantilever bracket all to be the same length from column center line. It is not clear how to determine the overall length. NJ TRANSIT RESPONSE: <u>From end of beam to the first sliding connector is 2'-6"</u>. Some of the cantilever brackets on the <u>erection diagrams were not adjusted once the sliding connectors were correctly placed</u>.

CFC Question 7) Is a splice connection allowed on W14 columns over 60 feet high. NJ TRANSIT RESPONSE: Yes, columns can be spliced for ease of shipping on column sections exceeding 60' in length...

CFC Question 8.) Are pole steps required on the W14 columns. **NJ TRANSIT RESPONSE: No.**.

CFC Question 9) Confirm that all connection hardware is to be included. **NJ TRANSIT RESPONSE:** <u>Include all connection hardware to erect structures complete in place in accordance with the drawings and specifications.</u>

CFC Question 10) Is the structural steel fabricator required to provide AISC certified. NJ TRANSIT RESPONSE: AISC certification is required, however, this requirement can be waived if the fabricator can demonstrate that they have completed five (5) successful projects of similar scope and application.

Questions 1-2, Set 4, dated July 2, 2014 are from Mike Dunn, Vice President, Daidone Electric (DEI), Newark, New Jersey

DEI Question 1) ST-920 Note 5: Calls for the contractor to perform soil resistivity tests prior to foundation installation. Is the intent to determine which locations have resistivity below 2000 ohms to determine which locations require cathodic NJT RESPONSE: Soil resistivity shall be measured at all pole locations to properly determine which locations require cathodic protection.

DEI Question 2) ST-022 to ST-513 calls for Assy E-9: Ground Wire Assy and references drawing CA-410. Are we to provide this ground assembly? If so, please provide detail CA-410 as it is not included in the bid set. NJT RESPONSE: Ground wire assembly E-9 is NJ Transit responsibility. Please refer to sketch 1, in Specification Section 01010-3, Summary of Work.

This concludes Addendum No.7. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

Managing Contract Specialist

Robert Delitto

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

CC: R Duran/K Donovan/D Maine/IFB File

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Veronique Hakim, Executive Director



July 2, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 6

To whom it may concern:

The following constitutes Addendum No. 6 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

# ITEM#I- The Bid Due Date has been revised. Electronic bids will be received via the Internet until 2:00:00 P.M., Thursday, July 24, 2014, by NJ TRANSIT.

ITEM#II: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Question 1, Set 3, dated June 24, 2014 is from Mike Dunn, Vice President, Daidone Electric (DEI), Newark, New Jersey

DEI Question 1) Thanks Bob. How long do you think it will take to get a response on our prequalification application? NJT RESPONSE: Response turnaround time is contingent on the accuracy of DEI's request. If the response is not as we hope the next step is for us to JV with Aldridge in which case the prequalification would be due on Thursday. If this is the case we would ask for a postponement in the bid due date so we can also extend the 14 day period for the prequalification package. I think what I submitted is reasonable and hope we don't have to go to the next step. NJT RESPONSE: Refer to Item I above..

Questions 1-2, Set 2, dated June 24, 2014 are from Paul Kaczkin, DMR Construction Services, Incorporated (DMRCSI), Waldwick, New Jersey DMRCSI Question 1) Bid form dated generated 05/28/14/ Revised 06/18/14 / line 0420 indicates there are 171 Steel Columns, Plan sheets as well as invitation to bid indicate there are 163 with additional foundations for Guy Wire. Can you please clarify what the correct amount of units there are. NJT RESPONSE: There are 171 columns and cross beams, of which 163 are columns and 8 cross beams..

DMRCSI Question 2). All the plans sheet have inch numerals with arrows pointing perpendicular each way on track (Example CA 084 shows three # 12 " and three # 9 ") please advise as to what these numbers represent. NJT RESPONSE: The arrows indicate direction of stagger. The numbers indicate the stagger value. Where 2 sets of values are shown, it signifies different stagger values of the contact and messenger wires. Refer to the erection diagrams. For example on CA 084, looking towards Gladstone, both contact wire and messenger wire are staggered 12 to the right at G1820+01. At

G1821+71, the contact wire is staggered 9 inches to the left, and the messenger wire is staggered 9 inches from the contact wire to the right. The legend in the plan set will be updated in the conformed set of drawings.

Questions 1-2, Set 1 dated June 24, 2014 are from Alex Grebel of JH REID General Contractor (JHRGC), South Plainfield, New Jersey

JHRGC Question 1) There are a couple of other questions we have and need clarified in order to submit the bid: The instructions to bidders mention filling out a Statement of Joint Venture. I have looked through the contract documents and online and cannot find this form. Is it a NJ Transit form or are you just looking for a statement that we provide in our own wording? NJT RESPONSE: Attached hereto and included as part of this addendum is a copy of the STATEMENT OF JOINT VENTURE FOR CONTRACTORS. JV Statements are due within 14 calendar days of the bid opening.

JHRGC Question 2) Bid X - please clarify if the JV will require its own Digital ID to bid on Bid X. If so, that means we need a distinct account for Bid X with the JV's name on it? NJT RESPONSE: Whether submitting a bid as a Joint Venture or a Single entity, it is the bidders responsibility to obtain the proper digital ID. Some clarity on how the electronic bidding works is needed so we can obtain a Digital ID in time for the bid. My understanding is it takes up to 7 days to obtain a Digital ID from Bid X, so we require this information by the middle of next week or we will request a short extension of time. NJT RESPONSE: Refer to Item I above.

Questions 1-2, Set 3 dated June 25, 2014 are from Paul Kaczkin, DMR Construction Services, Incorporated (DMRCSI), Waldwick, New Jersey

DMRCSI Question 1) Specification Section 16600 and Bid Form Line Items 0440
Specification section 16600 requires the contractor to test soil resistivity using the Wenner 4-pin method at 5, 15 and 30 feet spacing. The project's bid form only requires cathodic protection at 36 poles. Are we required to perform the Wenner 4-pin testing at 36 poles only, or are we required to test all poles? Furthermore, line item 0440 has the quantity of 36 locations. If the locations are predetermined, which implies the resistivity testing of the soil has already been measured, are we required to perform the Wenner testing at the 36 locations? NJT RESPONSE: Soil resistivity shall be measured at all pole locations to properly determine which locations require cathodic protection. However, we have estimated for bid purposes that no more than 36 locations will require cathodic protection. Total cathodic protection locations have not been predetermined..

#### **DMRCSI Question 2) Bid Form Line Items 0450**

The bid form's line item 0450 pricing is for 25 "additional ground rods." At this present moment the bid form only contains a line item for additional ground rods and no line item for the required ground rods. Please provide a line item for the ground rods that are not additional. NJT RESPONSE: Ground rods are measured and paid for per each concrete foundation installed and is not considered a separate bid item. Additional ground rods are paid for separately and the quantity is subject to field resistivity tests. However, we have estimated for bid purposes that no more than 25 will be required.

Questions 1-2, set 1 dated June 26, 2014 are from Bill Goodworth, Lead Estimator, Vanalt Electrical Construction, Inc. (VECI), Yeadon, Pa.

VECI Question 1) Further to the response given in Addendum No. 5, Item #1, we request additional clarification on the CB-1 cantilever brackets. One, the 2'-6" dimension from the first sliding connector to end of bracket does not work in all cases, such as the structure shown on the SED Drawing ST-469. NJ TRANSIT RESPONSE: The 2'-6" dimension should work in all cases, unless otherwise noted. Cantilever bracket shown on the plans at some locations may not have been adjusted to reflect this 2'-6" requirement once the sliding connectors were accurately placed. Therefore at some locations the cantilever bracket is shown longer than it should actually be.

VECI Question 2) Please clarify the determination of CB-1 length. Two, is the 4'-6" vertical dimension shown on Drawing ST-861 an absolute, or is a minimum angle of the 1" sag rod in relation to the W6x25 beam to be maintained, thereby necessitating a taller dimension in certain cases where the cantilever bracket arm is longer? NJ TRANSIT RESPONSE: The 4'-6" vertical dimension is to be maintained, unless otherwise noted elsewhere. Minimum angle for the sag rod to W6x25 beam is not indicated on the plans..

Questions 1-4, Set 1 are from RCCI June 27,2014 from Pete Dunn of Railroad Construction Company, Incorporated(RCCI), Paterson, New Jersey.

RCCI Question 1) We had just received Addendum No 2 late last week containing the 405 page Geotechnical Final Report. We have been studying the voluminous report and comparing the data and borings to the topography of the railroad. There is a huge amount of information to evaluate. We are endeavoring to determine access for specialized foundation equipment for the locations where it will be required. Toward that objective, may we request a NJ Transit sponsored site tour, either on your hi-rail equipment or perhaps on a revenue train. The tour on the tracks will permit site inspection, notations on mileposts and corroboration of access assumptions for the length of the ROW. We'd like to record the site tour by video recording, unless you or your design engineers have made a video already. If so, may we have access to it? NJ TRANSIT RESPONSE: There are no plans for a hi-rail trip. NJ TRANSIT is not aware of any existing videos.

RCCI Question 2) If you do not have an available video, we request a hi-rail inspection where speed and detail can be controlled. Please advise if the are other opportunities for detailed site inspection.

NJ TRANSIT RESPONSE: There are no plans for a hi-rail detailed site inspection..

RCCI Question 3) Lacking the video recording(s) we would request permission and an escort to make inspection of specific areas of importance or concern.

NJ TRANSIT RESPONSE: There are no plans for site inspections of specific areas.

RCCI Question 4) Please consider a three week postponement of the bid date to permit complete evaluation of the recently received geotechnical data. A postponement would further allow a mutually beneficial opportunity for bidders to inspect first hand the site conditions. **NJ TRANSIT RESPONSE:** Refer to Item I above.

Question 1, set 2 dated June 30, 2014 is from Bill Goodworth, Lead Estimator, Vanalt Electrical Construction, Inc. (VECI), Yeadon, Pa.

VECI Question 1) Please clarify the contractor's scope of work regarding the Down Guy Assemblies (Drawing ST-882). The Bid Item C05-012-001.13 description states "Down Guy Assembly G-3 (material only)". However, specification sections 05550 & 16126 both reference installation of down guy assemblies also. If the contractor is to supply material only, does the requirement for the tension measuring instrument in 05550 1.03 D apply? NJ TRANSIT RESPONSE: Down guy assembly materials shall be provided by the Contractor and down guy assembly shall be installed and secured in place by the Contractor with provisions for future adjustments once the catenary wires are installed by NJ Transit. The Contractor shall hand over tension measuring instrument to NJ Transit as required...

Questions 1-10, Set 1 dated July 1, 2014 are from Justin Lukens, Project Engineer, Case Foundation Company (CFC), Berwyn ,PA:

CFC Question 1) Dwg GP-004 calls for all structural steel to be A572 Gr 50. Is that correct. **NJ TRANSIT RESPONSE**: Response to Follow via addendum..

CFC Question 2) Dwg GP-004 calls for all structural to be Charpy impact tested. Is that correct. **NJ TRANSIT RESPONSE: Response to Follow via addendum.**.

CFC Question 3) Specification 05500 section 3.02 sub section d. call for all shop fillet welds to be magnetic particle inspected. Is that correct or only Partial and complete penetration. **NJ TRANSIT RESPONSE: Response to Follow via addendum.** 

CFC Question 4) What is the finish to be on the anchor bolt embedment plates. (Galv or Plain Finish). NJ TRANSIT RESPONSE: Response to Follow via addendum..

CFC Question 5) What is the material grade of the CB-1 cantilever bracket sag rod and turnbuckle size. **NJ TRANSIT RESPONSE:** Response to Follow via addendum..

CFC Question 6) Are the CB-1 cantilever bracket all to be the same length from column center line. It is not clear how to determine the overall length. NJ TRANSIT RESPONSE: Response to Follow via addendum..

CFC Question 7) Is a splice connection allowed on W14 columns over 60 feet high. NJ TRANSIT RESPONSE: Response to Follow via addendum..

CFC Question 8.) Are pole steps required on the W14 columns. **NJ TRANSIT RESPONSE:** Response to Follow via addendum..

CFC Question 9) Confirm that all connection hardware is to be included. **NJ TRANSIT RESPONSE:** Response to Follow via addendum...

CFC Question 10) Is the structural steel fabricator required to provide AISC certified. NJ TRANSIT RESPONSE: Response to Follow via addendum..

This concludes Addendum No.6. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

Managing Contract Specialist

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

Attachment (3 pages)-NJ TRANSIT STATEMENT OF JOINT VENTURES FOR CONTRACTORS

CC: R. Duran/K. Donovan/D. Maine/IFB File

#### STATEMENT OF JOINT VENTURE

WE, THE UNDERSIGNED, BEING DULY SWORN ACCORDING TO LAW, UPON OUR RESPECTIVE OATHS DEPOSE AND SAY THAT:

- 1. THE CONTRACTOR, UNDER WHOSE NAME WE HAVE AFFIXED OUR RESPECTIVE SIGNATURES, HAS DULY AUTHORIZED AND EMPOWERED US TO EXECUTE THIS <u>STATEMENT OF JOINT VENTURE</u> IN THE NAME OF AND ON BEHALF OF SUCH CONTRACTOR FOR THE PURPOSES HEREIN FURTHER SET FORTH.
  - 2. THE FOLLOWING NAMED CONTRACTORS:

(a)	;	;	;
	( ) Indîvidual	( ) Partnership	( ) Corporation
(b)	;	;	;
	( ) Individual	( ) Partnership	( ) Corporation
(c)	;	;	;
	( ) Individual	( ) Partnership	( ) Corporation

EACH OF WHOM IS PRE-QUALIFIED TO BID FOR WORK OF NEW JERSEY TRANSIT CORPORATION (NJ TRANSIT) HAVE ENTERED INTO A <u>JOINT VENTURE</u> FOR THE SPECIAL PURPOSE OF CARRYING ON THE WORK AND IMPROVEMENT HEREINAFTER DESCRIBED.

- 3. UNDER THE PROVISIONS OF SUCH <u>JOINT VENTURE</u> THE ASSETS OF EACH OF THE CONTRACTORS NAMED IN PARAGRAPH 2 HEREOF, AND IN CASE ANY CONTRACTOR SO NAMED ABOVE IS A PARTNERSHIP THE ASSETS OF THE INDIVIDUAL MEMBERS OF SUCH PARTNERSHIP, WILL BE AVAILABLE FOR THE PERFORMANCE OF SUCH <u>JOINT VENTURE</u> AND LIABLE THEREFOR AND FOR ALL OBLIGATIONS INCURRED IN CONNECTION THEREWITH.
- 4. THE ASSETS AND LIABILITIES OF THE NAMED CONTRACTORS FOR WHOM WE RESPECTIVELY EXECUTE THIS <u>STATEMENT OF JOINT VENTURE</u> ARE NOW SUBSTANTIALLY THE SAME AS SET FORTH IN THE LAST CONTRACTOR'S PREQUALIFICATION STATEMENT FILED WITH NJ TRANSIT FOR PURPOSES OF PRE-QUALIFICATION.
- 5. THIS <u>STATEMENT OF JOINT VENTURE</u> IS EXECUTED SO THAT THE NAMED CONTRACTORS MAY, UNDER SUCH <u>JOINT VENTURE</u>, BID UPON THE WORK AND IMPROVEMENT HEREIN MENTIONED AND THEY MAY, IF THE SUCCESSFUL BIDDER THEREFOR, BE AWARDED THE CONTRACT FOR SUCH WORK AND IMPROVEMENT. ANY BID, BID DOCUMENT, BOND AND CONTRACT RELATING TO THE WORK AND IMPROVEMENT HEREINAFTER SPECIFIED SHALL BE EXECUTED BY ANY PERSON AUTHORIZED TO BIND ANY MEMBER OF THIS <u>JOINT VENTURE</u>, AND WHEN SO EXECUTED SHALL BIND THIS <u>JOINT VENTURE</u> AND EACH AND EVERY CONTRACTOR NAMED HEREIN, SEVERALLY AND JOINTLY. SIMULTANEOUS WITH THE EXECUTION OF THE CONTRACT THE JOINT VENTURERS SHALL DESIGNATE AND APPOINT A PROJECT SUPERVISOR TO ACT AS THEIR TRUE AND LAWFUL AGENT WITH FULL POWER AND AUTHORITY TO DO AND PERFORM ANY AND ALL ACTS OR THINGS NECESSARY TO CARRY OUT THE CONSTRUCTION WORK SET FORTH IN SAID CONTRACT.

### STATEMENT OF JOINT VENTURE (Continued)

6. IMPROVEM EXECUTE EACH OF	ENTS THIS	AS STA	<u>JOINT</u> ATEMEN	VENT	<u>URERS</u> , JOINT	WE BI	ND TH <u>RE</u> IN	E CONT FIRM	AGREEM	FOR	WHOM	WE RES	SPECTI	VELY
7. ENTERED						/EMENT	FOR	WHICH	THIS	JOIN	IT VEI	NTURE	HAS	BEEN
• • • • • • • • • • • • • • • • • • • •	,							• • • • • •						,
				• • • • •		,								
,	, , , ,	• • • •									,.,			
, , , , , , ,	,					* * * * * *					,			
SUBSCRIBEI THIS				DA	Y OF	ВҮ		• • • • • • • •			(Name	of Con		
SUBSCRIBEE					Y 0F	ВҮ					(Name	of Con		•
SUBSCRIBED					Y 0F	ВҮ					(Name	of Con  type or		•

## TO BE EXECUTED BY EACH JOINT VENTURER

AUTHORIZATION AND DESIGNATION OF RESPECTIVE AFFIANTS TO THE STATEMENT OF JOINT VENTURE HERETO ATTACHED TO ACT FOR AND ON BEHALF OF THE CONTRACTORS NAMED IN PARAGRAPH 2 THEREOF:
(a) HEREBY CERTIFIES THAT
(Name of Representative)  HAS BEEN AND IS HEREBY EMPOWERED
TO SIGN THE STATEMENT OF JOINT VENTURE ATTACHED HERETO AS THE AUTHORIZED REPRESENTATIVE OF(Name of Contractor)
FOR THE SPECIAL PURPOSE THEREIN EXPRESSED.
ATTEST
(SEAL NECESSARY IF CORPORATION)
• • • • •
(b) HEREBY CERTIFIES THAT
(Name of Representative)  HAS BEEN AND IS HEREBY EMPOWERED
TO SIGN THE STATEMENT OF JOINT VENTURE ATTACHED HERETO AS THE AUTHORIZED REPRESENTATIVE OF(Name of Contractor)
FOR THE SPECIAL PURPOSE THEREIN EXPRESSED.
ATTEST
(SEAL NECESSARY IF CORPORATION)
(c) HEREBY CERTIFIES THAT (Name of Contractor)
(Name of Representative)  HAS BEEN AND IS HEREBY EMPOWERED
TO SIGN THE STATEMENT OF JOINT VENTURE ATTACHED HERETO AS THE AUTHORIZED REPRESENTATIVE OF(Name of Contractor)
FOR THE SPECIAL PURPOSE THEREIN EXPRESSED.
ATTEST
(SEAL NECESSARY IF CORPORATION)

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Veronique Hakim, Executive Director



June 24, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 5
To whom it may concern:

The following constitutes Addendum No. 5 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

# ITEM#I: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Question 1, Set 2, dated June 20, 2014 is from Mike Dunn, Vice President, Daidone Electric (DEI), Newark, New Jersey

DEI Question 1) The length of the Cantilevered Brackets CB-1 shown on ST-861 is not given. Referring to the applicable SED shown, a typical reference dimension of 13'-0 is shown from the center of the poles to the center of the track, however, there is no dimension given from the center of the track to the end of the CB1 Beam that extends beyond the track. Please provide a standard length for the CB-1 bracket. NJT RESPONSE: The end of cantilever bracket to the first sliding connector from the end is 2'-6". This dimension was previously shown on drawing ST-861, but was omitted in the conformed set. The dimensions from sliding connector to center of track and from center of track to pole are shown on the erection diagrams. Post award drawing ST-861 will be updated with the conformed set of drawings.

This concludes Addendum No.5. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

Managing Contract Specialist

Robert Delitte

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

CC: R. Duran/K. Donovan/D. Maine/IFB File

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Veronique Hakim, Executive Director



June 20, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 4

To whom it may concern:

The following constitutes Addendum No. 4 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

# ITEM#I: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Questions 1-7, Set 1, dated June 18, 2014 are from Mike Dunn, Vice President, Daidone Electric (DEI), Newark, New Jersey

DEI Question 1) Please confirm that the procurement and installation of all Support and Registration Assemblies as shown on drawings ST-22 through ST-514 will by others and are Not In Contract per Technical Specification Sketch 01010-3. **NJ TRANSIT RESPONSE: That is correct.** 

DEI Question 2) Please confirm that the procurement and installation of the following Feeding and Switching devices and mounts as shown on drawings ST-22 through ST-514 will by others and are Not In Contract: If these are to be included in our bid please provide details. Signal Power Assembly (SPA)

- a. Signal Power XFMR (SGL)
- b. Feeder Support (E1A)
- c. Sig Power Support (E3A)

#### NJ TRANSIT RESPONSE: That is correct.

DEI Question 3) Please confirm that the Signal Power XFMR Mount is to be included in the bid (SPS-1). **NJ TRANSIT RESPONSE: That is correct.** 

DEI Question 4) Please advise if the Whistleboard Signs at 1878+76 and 2171+41 are to be included in the contract. **NJ TRANSIT RESPONSE: No..** 

DEI Question 5) Are the Pocket Tracks or sidings available for the contractors use to store equipment on rail between track outages? **NJ TRANSIT RESPONSE**: **Yes sidings will be available**.

DEI Question 6) How is the pay item of a given pole height determined? For example, the pole at 1191+81.5 is 42.521 LF Tall (Top of Pole to top of Rail height + Pole of Rail to Top of Foundation height). Is this pole considered a 40 foot pole or a 45 foot pole? What is the rule for establishing the pay item. NJ TRANSIT RESPONSE: The pole at 1191+81.5 is considered a 45 ft. pole. Pole

height is shown on pole detail drawing ST-860, as pole length, from top of foundation to top of column, and is paid for according to Specification Section 05500, 4.02.

DEI Question 7) The height of one of the poles for 3 of the portal structures is not detailed. Please provide:

1246+76 1518+64 1529+01

**NJ TRANSIT RESPONSES:** 

1246+76 = 40 ft pole height

1518+64 = 40 ft pole height

1529+01 = 34 ft pole height.

Questions 1-2, Set 1, dated June 18, 2014 are from John Mason, PKF-MARK III, Inc. (PKF), Newtown, PA:

PKF Question # 1- Drawing ST-920 indicates that the installation of the Magnesium Anode unit are to be installed in a 10" dia x 25' deep shaft. If rock is encountered will the transit authority require that we stop at the rock or continue to the 25' depth? NJ TRANSIT RESPONSE: Per note 7 of drawing ST-920, the installation depth of 25 ft. is typical, and may be less at rock locations. However, Contractor must ensure that the anode will produce the desired output.

PKF Question # 2 –Drawing ST-920 Indicates that the ground conductor be installed at a min. depth of 54". If rock is encounter will the authority require the 54' depth or can the ground conductor be installed on top of the rock. NJ TRANSIT RESPONSE: The 54" depth is minimum design requirement to ensure wire remains protected from nicks, cuts, bruises or breaks. However, if rock is encountered, Contractor may seek approval of the Engineer to install cable on top of rock, and if given, shall install and protect cable accordingly. Minimum code requirements for buried cable installations must be conformed to.

This concludes Addendum No.4. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

Managing Contract Specialist

Robert Delitto

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

CC: R. Duran/K. Donovan/D. Maine/IFB File

Chris Christie, Governor Kim Guadagno, Lieutenant Governor Veronique Hakim, Executive Director



June 19, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 3

To whom it may concern:

The following constitutes Addendum No. 3 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

#### ITEMI- The following General Information Qualifies for this Procurement:

- NJ TRANSIT has revised the qualifications to bid this project. **NO FIRMS ARE CURRENTLY PREQUALIFIED TO BID THIS PROJECT.** These requirements are project specific and all prequalification requests will be kept confidential between NJ TRANSIT and the requestors. Firms currently pre-qualified must submit references sheets (sample attached) in accordance with item 2. Reference sheets can also be downloaded from the Request for Prequalification documents located on bidx. Firms whose pre-qualifications have expired must submit both the Specialty references along with the standard types of work references. All requests are due within 14 calendar days of the bid due date. The undersigned will notify applicants the status of each request.
- 2 <u>DELETE SPECIAL PROVISIONS (SP 2)- PREQUALIFICATION OF BIDDERS and Replace with the following:</u>

#### SP.2 PREQUALIFICATION OF BIDDERS

Supplement Article C.2 of NJ TRANSIT's Instructions to Bidders for Construction by adding the following:

All firms must be prequalified by NJ TRANSIT prior to submitting a bid. Contractors must be prequalified for Heavy Construction Rail "HR" (Operating Railroad Environment) with a classification of "GC" General Contractor or a subcategory classification of "ET"-Catenary / Electrification Contractor: SP-Specialty: Completed projects using Hi-Rail Drilling Equipment methods with five (5) years of experience and a minimum of three (3) projects with a minimum rating of "o" \$10,000,001 to \$15,000,000.

The General Contractor shall submit related project experience to NJ TRANSIT that they have a minimum of five (5) years of experience and minimum of three (3) projects which the contractor has used Hi-Rail Drilling Equipment methods. The Contractor shall provide the scope of the On-Track Drilling Equipment methods used and duration of the construction for each project.

In addition, the General Contractor must submit his On-Track Driller subcontractor for approval. The On-Track Driller subcontractor, shall submit related project experience to NJ TRANSIT that they have a minimum of <u>five (5) years of experience and a minimum of three (3) projects</u> where Hi-Rail Equipment Drill Rigs were used to perform self-performed work similar to the work shown in the Contract documents.

In accordance with Item 17 of the Prequalification, A bidder to be prequalified to perform construction work in an "Operating Railroad Environment" shall have identified in their Bidder's Prequalification Questionnaire specific completed projects which demonstrate construction experience working on an FRA regulated high speed commuter or passenger railroad, in both electrified and diesel fuel environment, working on or within four feet of the active railroad tracks.

Completed prequalification questionnaire and supporting documentation are due 14 calendar days prior to the bid opening date and shall be submitted to:

NJ TRANSIT
Mr. Robert Delitto
Managing Contract Specialist
One Penn Plaza East-6<sup>th</sup> Floor
Newark, NJ 07105
rdelitto@njtransit.com

# ITEM#II: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Questions 1-7, Set 1, dated June 18, 2014 are from Mike Dunn, Vice President, Daidone Electric (DEI), Newark, New Jersey

DEI Question 1) Please confirm that the procurement and installation of all Support and Registration Assemblies as shown on drawings ST-22 through ST-514 will by others and are Not In Contract per Technical Specification Sketch 01010-3. <a href="https://www.numerical.new.numerical.

DEI Question 2) Please confirm that the procurement and installation of the following Feeding and Switching devices and mounts as shown on drawings ST-22 through ST-514 will by others and are Not In Contract: If these are to be included in our bid please provide details. 

NJ TRANSIT RESPONSE:

Response to follow via addendum.

- a. Signal Power Assembly (SPA)
- b. Signal Power XFMR (SGL)
- c. Feeder Support (E1A)
- d. Sig Power Support (E3A)

DEI Question 3) Please confirm that the Signal Power XFMR Mount is to be included in the bid (SPS-1). **NJ TRANSIT RESPONSE:** Response to follow via addendum.

DEI Question 4) Please advise if the Whistleboard Signs at 1878+76 and 2171+41 are to be included in the contract. **NJ TRANSIT RESPONSE: Response to follow via addendum.** 

DEI Question 5) Are the Pocket Tracks or sidings available for the contractors use to store equipment on rail between track outages? **NJ TRANSIT RESPONSE**: **Response to follow via addendum.** 

DEI Question 6) How is the pay item of a given pole height determined? For example, the pole at 1191+81.5 is 42.521 LF Tall (Top of Pole to top of Rail height + Pole of Rail to Top of Foundation

height). Is this pole considered a 40 foot pole or a 45 foot pole? What is the rule for establishing the pay item. **NJ TRANSIT RESPONSE:** Response to follow via addendum.

DEI Question 7) The height of one of the poles for 3 of the portal structures is not detailed. Please provide. **NJ TRANSIT RESPONSE**: **Response to follow via addendum.** 

- e. 1246+76
- f. 1518+64
- q. 1529+01

Questions 1-2, Set 1, dated June 18, 2014 are from John Mason, PKF-MARK III, Inc. (PKF), Newtown, PA:

PKF Question # 1- Drawing ST-920 indicates that the installation of the Magnesium Anode unit are to be installed in a 10" dia x 25' deep shaft. If rock is encountered will the transit authority require that we stop at the rock or continue to the 25' depth? **NJ TRANSIT RESPONSE: Response to follow via addendum.** 

PKF Question # 2 –Drawing ST-920 Indicates that the ground conductor be installed at a min. depth of 54". If rock is encounter will the authority require the 54' depth or can the ground conductor be installed on top of the rock. **NJ TRANSIT RESPONSE: Response to follow via addendum.** 

Question 1, Set 1, dated June 19, 2014 is from Roy Friedlander ,Case Foundation Company (CFC), Berwyn, PA:

CFC Question #1) Addendum #1 sets the time for completion of the work at 730 calendar days after notice to proceed. Based upon the information communicated to everyone yesterday at the "pre-bid", it appears it is N.J Transit's intention to give a notice to proceed in the fall of 2014, which means that the project must be completed in the fall of 2016. Special Provisions Item SP7, Article 4.2.11 shows available work windows in 2017. Please confirm that these work windows will not be available. NJ TRANSIT RESPONSE: The 2017 work window shown in the chart under SP7 (article 4.2.11) are not available to the contractor. NJT anticipates a NTP by the end of this year.

This concludes Addendum No.3. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

Managing Contract Specialist

Robert Delitto

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

Attachments (RELATED PROJECT EXPERIENCE)

CC: R. Duran/K. Donovan/D. Maine/IFB File

## **NJ TRANSIT**

#### II. RELATED PROJECT EXPERIENCE

Fill in complete information for at least five (5) projects finished recently for which the organization performed the type of work for which prequalification is requested.

Project Name:	Contract Price: \$
	Date Completed:
Type of Work:	
Client:	
Contact Person:	
Telephone No.: ( )	
Fax No.: ( )	
Project Name:	Contract Price: \$
	Date Completed:
Type of Work:	
Client:	
Contact Person:	
Telephone No.: ( )	
Fax No.: ( )	
Project Name:	Contract Price: \$
	Date Completed:
Type of Work:	
Client:	
Contact Person:	
Telephone No.: ( )	
Fax No.: ( )	
Project Name:	Contract Price: \$
	Date Completed:
Type of Work:	
Client:	
Contact Person:	
Telephone No.: ( )	
Fax No.: ( )	
Project Name:	Contract Price: \$
	Date Completed:
Type of Work:	
Client:	
Contact Person:	
Telephone No.: ( )	
Fax No : ( )	



June 18, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 2

To whom it may concern:

The following constitutes Addendum No. 2 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

### ITEM#I: The following General Information Qualifies for this Procurement:

- A) Special Provisions No. SP 3- The DBE Point of Contact has been revised to
  - a. Jacquelin Rush-Gilbert

Senior Business Development Specialist

Office of Business Development-Contract Compliance

Office of Civil Rights & diversity Programs

(973) 491-8061

<u>irush-gilbert@nitransit.com</u>

B) A pre-bid conference was held on 10:00AM, Wednesday, June 18, 2014 at NJTRANSIT Headquarters. The meeting minutes were as follows:

Discussion included Item I-Bid Forms, Item II-Instructions to Bidders, Item III-Special Provisions, Item IV-General Provisions, Item V-Technical Provisions and Drawings, Item VI-A DBE race neutral goal has been set for this project. There was no site tour.

Critical dates for this Procurement include:

Request for Prequalification are due fourteen (14) calendar days prior to the bid opening date;

All RFIs are due by 4:00PM, Monday, June 30, 2014;

Bid Opening is 2:00PM, Thursday, July 10, 2014;

First and second low bids must submit required Race Conscious DBE FORMS( See Special Provisions) within 7 calendar days after the bid due date;

- 1) All attendees were notified that no verbal communications qualified for this Procurement and that all communications must be submitted through the Contracting Officer Representative-Robert Delitto.
  - 2) Attached hereto is a copy of the pre-bid meeting attendance sheet.

# ITEM#II: Incorporate the following question/request for information along with NJ Transit's responses (underlined):

Question 1, Set 1, dated June 11, 2014 is from Luz Carrillo, Railroad Construction Company, Inc (RCCI), Paterson, New Jersey

RCCI Question 1) After reviewing the bid documents, we noticed that the soil borings were missing. Please provide a copy through BidEx. NJ TRANSIT RESPONSE: <u>Available geotech data has been uploaded to BidX as part of this addendum.</u>

Questions 1-2, Set 1, dated June 13, 2014 are from Paul Kaczkin, DMR Construction Services, Incorporated (DMRCSI), Waldwick, New Jersey

DMRCSI Question#1) The plan sections mention that there are geotechnical profiles for this project but do not see them anywhere in the bid documents will they be forthcoming shortly? NJ TRANSIT RESPONSE: Available geotech data has been uploaded to BidX as part of this addendum.

DMRCSI Question#2) The documents also mention that the locations of the ID 27, should we anticipate that all the soil encountered at all the 163 holes sites should be considered contaminated and to be removed from the site? NJ TRANSIT RESPONSE: Any soil removed from the site shall be considered.

This concludes Addendum No.2. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

Managing Contract Specialist

Robert Delitto

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

Attachments

(1 Page) Pre-bid Conference Sign-In Sheet

Geo Tech Report (Volume I)

Geo Tech Report (Volume II)

CC: R. Duran/K. Donovan/D. Maine/J. Rush-Gilbert/IFB File

# **NJ TRANSIT**

IFB No. 13-071X

Gladstone Line Pole Foundations Replacement Project 10:00AM: Wednesday, June 18, 2014 BID ROOM – 6<sup>TH</sup> FLOOR Page 1 of \_\_\_\_\_\_

#### PRE-CONSTRUCTION MEETING ATTENDANCE SHEET

NAME	CO	MPANY, ADDRESS & TELEPHONE
- Kohortou	NTTRANSIT	9734917542
RAFAEL BURAN		973 - 491- 7580
MARIE PATZI		MING 610 650 7730
BRYAN SHOBER	GANNETT FLEMING	610-650-7730
Jackel Kush-Gilbe	eret OBD	X 8061
DAN MAINE	NJTERNSIA	201 800 1363
Kevin MHchell	HNTB	856-924-4512)
KORRY DONOUN	NJT	973-491-7332
DEREY LABATCH	N.R.C	732-489-0598
REGIS DOVIETTE	CHECK RID 201	-664-5487
Jeffrey Glisson	PKF-Mark ITT, Inc	267-759-3075
Anthon Arry	George Horns	132-988-4004 x 338
PAT REDMOND	CORNELL FABRY ATOON	
JUAN ROMONO	FERREIRA CONTINUCTION	
Tony Lauro	Vanal+ Elactrical Construct	
ROY FRIEDLANDED	CASE FOUNDATION	
JUSTIN LUKENS	LASE FOUNDATION (D.	610 - 353 - 0600
MikeDUNN	Daidone Electric, Inc.	973-690-5216
Reter Blunckburgo	Mortherest Structural Steel	914-699-9900
ADAM BLUM	NESS	414-644-9900
NITHIN THOMAS	J. H. Reid General (mho	cm. 732-757-4050
DUANN HAMMA	J. H. RETE GENERAL CONT	
Angela andena	underground Otilities (10	
Robert Lubarsky	Underground Utilities C	
KENNETH WIGG	MORETRENCH AMERICAN	973-677-2100
PAUL KACZILIH	DWR COHST. SERIERS	201-652 2411



## Medina Consultants

# GEOTECHNICAL REPORT (FINAL)

For

# Gladstone Line Catenary Wood Pole Replacement Project

(VOLUME I OF II)

Gladstone – New Providence New Jersey

April 5, 2006

Prepared for:

New Jersey Transit And Gannett Fleming

Prepared by: PU

Reviewed by: SS/SP

handu J. Bhoraniya, P. J.

NJ License No. 24GE03835500

Project Manager Date: 04/05/2006

Medina Project No. 05032

Prepared by:
Medina Consultants

One Edgeview Drive Hackettstown, NJ 07840 Ph: (908) 850-3366 Fax: (908) 850-3326

FOR THIS CONTRACT ARE NOT REPRINTED IERE DUE TO SIZE

Chris Christie, Governor Kim Guadagno, Lieutenant Governor James S. Simpson, Board Chairman Veronique Hakim, Executive Director



June 5, 2014

RE: INVITATION FOR BID NO.13-071X Gladstone Line Pole Foundations Replacement Project. Addendum No. 1

To whom it may concern:

The following constitutes Addendum No. 1 and must be acknowledged by each Bidder acknowledging receipt of addendum with your e-bid submission. Failure to do so may render a bid as non-responsive.

#### ITEM#I: The following General Information Qualifies for this Procurement:

- 1) Revise Instructions to Bidders Section E6-BID VALIDITY. Bid Validity has been revised to 120 calendar days.
- 2) Delete Note #3 on drawings CA-001 through CA-138, note #3 (REMOVE AND DISPOSE OF EXISTING WOOD POLES AND RELATED COMPONENTS AS PER SPECIFICATIONS.) does not apply. NJT force account will perform that work. The specs indicate that NJT will perform this work.
- 3) Replace Special Provisions 6 with the following:

#### SP 6: TIME OF COMPLETION - DELAY - LIQUIDATED DAMAGES (2.1)

Add the following to Article 2.1.1:

<u>Daily track outages per hour</u> - The Contractor shall pay NJ TRANSIT the amount of two thousand three hundred dollars (\$2,300.00) as liquidated damages for **each hour** and fraction thereof that the Contractor fails to vacate the track to permit resumption of rail service beyond the time specified in SP-7.

Add the following to Article 2.1.2:

The time of completion for the contract work shall be seven hundred thirty (730) calendar days after the effective date of the Notice to Proceed.

This concludes Addendum No.1. Bidders must acknowledge receipt of this addendum on the acknowledgment of receipt of addendum with your e-bid submission.

Sincerely,

Robert Delitto

Managing Contract Specialist

Contracts/Division of Procurement E mail: RDELITTO@NJTRANSIT.COMFax(973)491-7597

CC: R. Duran/K. Donovan/D. Maine/J. Hannah/IFB File

# **NEW JERSEY TRANSIT CORPORATION**

## **SPECIAL PROVISIONS**

#### **FOR**

# GLADSTONE LINE CATENARY POLE FOUNDATIONS REPLACEMENT

IFB#13-071X

### **TABLE OF CONTENTS**

# **GLADSTONE LINE POLE FOUNDATIONS PROJECT**

### **SPECIAL PROVISIONS**

SP.	1	COMMUNICATIONS WITH NJ TRANSIT (B4)	1
SP.	2	PREQUALIFICATION OF BIDDERS (C.2)	1
SP.	3	DBE CERTIFICATION (C.3)	1
SP.	4	DBE FORMS AND CERTIFICATION (E.4.1)	2
SP.	5	DESCRIPTION OF WORK (1.18)	2
SP.	6	TIME OF COMPLETION - DELAY - LIQUIDATED DAMAGES (2.1)	3
SP.	7	USE OF PREMISES (4.2)	3
SP.	8	MAINTENANCE OF MARINE TRAFFIC (4.10)	5
SP.	9	MAINTENANCE OF RAILROAD TRAFFIC (4.11)	5
SP.	10	ARTS IN TRANSIT (4.13)	6
SP.	11	CONSTRUCTION PROGRESS SCHEDULE (6.2)	6
SP.	12	CONSTRUCTION LAYOUT (6.12)	7
SP.	13	MATERIALS - WORKMANSHIP - LABOR (7.2)	8
SP.	14	NEW JERSEY PREVAILING WAGE ACT (11.1)	8
SP. 1	15	MEASUREMENT AND PAYMENT (12.1)	8
SP. 1	16	EMPLOYEE PROTECTIONS-CONSTRUCTION ACTIVITIES (LABOR PROVISIONS) (14.1)	9

#### **ATTACHMENTS**

- ATTACHMENT A THE STATE OF NEW JERSEY PREVAILING WAGE DETERMINATION FOR SOMERSET COUNTY, NEW JERSEY (69 PAGES)
- **ATTACHMENT B** DBE REQUIREMENTS FOR RACE CONSCIOUS FEDERAL PROCUREMENT ACTIVITIES (29 pages)
- ATTACHMENT C THE FEDERAL GENERAL WAGE DETERMINATION FOR SOMERSET COUNTY, NEW JERSEY 8 PAGES)

# NEW JERSEY TRANSIT CORPORATION SPECIAL PROVISIONS FOR GLADSTONE LINE POLE FOUNDATIONS REPLACEMENT

### SP.1 COMMUNICATIONS WITH NJ TRANSIT (B4)

Modify Article B.4 of NJ TRANSIT's Instructions to Bidders for Construction by adding the following:

All questions must be directed to Robert Delitto, Managing Contract Specialist, NJ TRANSIT Procurement Department at 973-491-7542 or <a href="mailto:red-itto@nitransit.com">red-itto@nitransit.com</a>.

### SP.2 PREQUALIFICATION OF BIDDERS (C.2)

Supplement Article C.2 of NJ TRANSIT's Instructions to Bidders for Construction by adding the following:

Contractors must be prequalified for "Heavy Construction-Rail "HR" (Operating Railroad Environment) with a classification of "GC"- General Contractor or a subcategory classification of "ET"-Catenary /Electrification Contractor. NJTRANSIT suggests a minimum rating of "O" \$10,000,001 to \$15,000.000.00.

The following is hereby added to the end of the fifth paragraph:

The paper original of the completed Prequalification Questionnaire and photocopies of all supporting documents shall be submitted 14 calendar days prior to the bid opening date and shall be submitted to:

Mr. Robert Delitto
Managing Contract Specialist
NJ TRANSIT
One Penn Plaza East-6<sup>TH</sup> Floor
Newark, NJ 07105
RE: IFB 13-071X
Pregualification Documents

#### SP3 DBE CERTIFICATION (C.3)

Supplement Article C.3 "DBE CERTIFICATION" of the Instructions to Bidders by adding the following:

As an aid in meeting its commitment to its Disadvantaged Business Enterprise (DBE) Program, NJ TRANSIT has assigned a Twenty percent (20%) Race Conscious DBE goal on the gross sum amount of the bid or Contract value for DBE subcontracting participation.

NJ TRANSIT's DBE Program shall be accorded the same priority as compliance with all other legal obligations required by NJ TRANSIT and the USDOT. Bidders shall comply

with and refer to New Jersey Transit Corporation, DBE Requirements for Race-Conscious Federal Procurement Activities included in the Contract Documents in the award and administration of NJ TRANSIT contracts.

All NJ Unified Certification Program (NJUCP) certified DBE firms, including suppliers, are eligible to participate in this contract. Questions regarding a firms DBE certification should be directed to Ms. Jacqueline Hannah, Senior Contract Compliance Specialist, of the Contract Compliance Unit of the Office of Business Development (OBD) at 973-491-7539 or by email at <a href="mailto:ihannah@njtransit.coom">ihannah@njtransit.coom</a> or in writing at:

Ms. Jacqueline Hannah
Principal Business Development Specialist
NJ TRANSIT
Office of Business Development, Contract Compliance Unit
One Penn Plaza East
Newark, NJ 07105-2246

#### SP.4 DBE FORMS AND CERTIFICATION (E.4.1)

Supplement Article E.4.1 of the Instructions to Bidders by adding the following:

Bidders are encouraged to submit these mandatory attached required DBE forms (A, A1, A2, B, D and DBE certificate) forms on the bid opening date to expedite the review of the Bidders proposed DBE Program and to prevent delay in the contract award. These applicable mandatory required DBE forms (ATTACHMENT B-DBE REQUIREMENTS FOR RACE CONSCIOUS FEDERAL PROCUREMENT ACTIVITIES) shall be completed in their entirety with no blank fields. Carrying out these requirements is a matter of Bidder responsibility and failure to do so may render the bidder non-responsible which may affect the award of the contract.

#### SP.5 DESCRIPTION OF WORK (1.18)

Add the following new article 1.18:

1.18 Description of Work

The work under this contract consists of the construction necessary to install reinforced concrete foundations within 4 feet of existing wood poles that support the existing overhead catenary system (OCS) and ancillary lines. The project, in general, involves civil work, site work, and concrete foundations as outlined in the Contract Documents.

The project site is on the Gladstone Branch of NJ TRANSIT's Morris & Essex (M&E) Rail Line. Construction will be comprised of the installation of one hundred sixty three (163) concrete foundations and steel poles between New Providence Station and Gladstone Rail Yard..

### SP.6 TIME OF COMPLETION - DELAY - LIQUIDATED DAMAGES (2.1)

Add the following to Article 2.1.1:

<u>Daily track outages per hour</u> - The Contractor shall pay NJ TRANSIT the amount of two thousand three hundred dollars (\$2,300.00) as liquidated damages for **each hour** and fraction thereof that the Contractor fails to vacate the track to permit resumption of rail service beyond the time specified in SP-7.

Add the following to Article 2.1.2:

The time of completion for this project shall be no later than December 17, 2017

#### SP.7 USE OF PREMISES (4.2)

Add the following to Article 4.2.10:

The first Thirty (30) days from Notice to Proceed shall be used for shop drawing / preparation submittals, review and approvals. Access to the site / actual start of construction will commence after this Thirty (30) day period. Any access to the site prior to the Thirty (30) day period will be subject to NJ TRANSIT review and approval.

In the event of any emergency involving any Railroad facility, the Contractor shall immediately telephone the NJ TRANSIT CONSTRUCTION MANAGER and NJ TRANSIT Rail Operations Trouble Desk at 1-800-742-2921.

Add the following new article 4.2.11:

#### 4.2.11 Track Outages

Track outages are critical resources and the contractor is advised that his success in obtaining future track outages is directly related to his demonstrated performance in returning the track to service at the times previously agreed upon.

The window of time available for all construction and any required outages(s) or obstruction(s) will be at the following time periods:

•	OUTAGES	2015	2016	2017
•	WEEK DAYS 10:00 A.M. THRU 3:00 P.M.	July 6 August 28	July 5 – August 26	July 5 – August 25
•	WEEK ENDS – 6:00 A.M. SATURDAY THRU 9:00 P.M. SUNDAY	Jan 24 – December 20	January 23 - December 18	Jan 21 – December 17

Track Outages will not be available on Easter Day, Memorial Day weekend, Labor Day weekend and the Steeple Chase weekend (end of October).

(Note: NJ Transit Rail requires 1 hour at the beginning and end of each work period to de-energize/energize the electric traction and signal power facilities in the applicable zones)

Add the following new article 4.2.12:

#### 4.2.12 Non-Availability of Track Outages

- A. The Contractor acknowledges that even when all planning and approvals for track outages have been accomplished as specified, railroad operations may dictate that track outages be denied to the Contractor. NJ TRANSIT is providing the following procedure to reimburse Contractor for costs for delays caused by such non-availability of programmed track outages. Payment will not be made for delays that are less than 30 minutes per shift.
- B. Monetary reimbursement will be granted only when all of the following have occurred:
  - 1. A Work Plan for the specific work was reviewed and approved.
  - 2. The specific work activity was scheduled on the lasted bi-weekly update of the approved construction progress schedule.
  - 3. Confirmation was received that the specific track outage was previously reviewed by NJ TRANSIT.
  - 4. The specific track outage was delayed for more than 30 minutes or canceled without prior notice.
  - 5. "Prior notice cancellation" shall mean notification to Contractor no later than 16 hours before the start of the scheduled shift.
  - 6. Contractor has productively used all prior granted track outages. 
    "Productively used" means Contractor diligently performed he work scheduled for the subject track outages as planned in the approved Work Plan.
  - 7. The reason for the cancellation/delay of the track outage was solely for the benefit of the railroad and not attributable, in whole or in part, to any acts, errors, omissions on the part of Contractor. NJ TRANSIT reserves the right to cancel track outages without notice or compensation because of the Contractor's failure to abide by the rules of railroad operations, the terms of the Contract or the instruction of NJ TRANSIT, the flagman or the Construction Manager.
  - 8. Contractor was not able to reassign that day's resources on other parts of the work during this period.
- C. Payments for direct costs actually incurred due to such non-availability of track outages will be limited to:

- 1. Hourly labor specifically assigned to the work actively requiring the track outage.
- 2. Major Equipment and Plant Rentals required specifically for the outage. Small tools, on-site equipment, and vehicles will be considered part of the overhead for the purpose of this provision.
- 3. Ten percent (10%) overhead calculated on 1 and 2 above.
- D. No profit will be allowed under this Provision.
- E. Payment will be made from the allowance established under the bid time TRACK TIME ALLOWANCES listed in the Proposal, as part of the monthly progress payments when all back-up information has been submitted.
- F. Payment for canceled or delayed track outages will be made for labor and equipment operating in work sites scheduled to be active during that specific track outage. Payment will not be made for idle labor or equipment that is not, and was not scheduled to be, operating in the active work site.

#### SP.8 MAINTENANCE OF MARINE TRAFFIC (4.10)

Delete Article 4.10 of the General Provisions for Construction in its entirety.

### SP.9 MAINTENANCE OF RAILROAD TRAFFIC (4.11)

Supplement Sub-Article 4.11.1 of the General Provisions for Construction as follows:

The Contractor fully acknowledges that the Contract Work is within the confines of an active railroad right-of-way vital to the operations of NJ TRANSIT and that the Contractor's access and use of the premises may be restricted by NJ TRANSIT without advance notice to insure continued operations, despite having received prior permission/approval to enter on and/or occupy said premises. Furthermore, the Contractor acknowledges that the Contract Work must be done to accommodate NJ TRANSIT's operations at all times and that all of the Contractor's activities in all areas must be performed in such a manner so as to minimize or eliminate all disruptions, interferences or hindrances whatsoever to NJ TRANSIT's operations.

When work is performed within 18 feet of centerline of the nearest track or within 18 feet of any Overhead Energized circuit, or equipment has the potential to come within 18 feet of the these items, the track is considered "fouled." All work that "fouls track" shall be performed during the times as listed in SP.7 unless specifically approved in writing by NJ Transit.

The Contractor will be allowed to work simultaneously on both side of the Railroad for work that NJ TRANSIT has determined requires only flag protection. Temporary protective fencing must be installed a minimum of 8'-6" from centerline of track if Contractor proposes to work beyond outages provided in SP.7.

The Contractor shall notify the Construction Manager in writing two weeks prior to any utility shutdown that may affect NJ TRANSIT's operations and obtain the Construction Manager's express written approval before proceeding with a utility shutdown. NJ TRANSIT will be responsible to mark-out all its underground facilities but the Contractor must use NJ One-Call as required.

The Contractor clearly recognizes that the above cited restrictions, limitations and stipulations may adversely impact his productivity and efficiency during the construction and the same has been taken into account when preparing his bid prices.

Article 4.11.2 of the General Provisions for Construction is supplemented by adding the following:

All high-rail equipment proposed for use on-track must pass inspection prior to having access to the rails in both phases. The inspection of the vehicles will be arranged through the Construction Manager and conducted by NJ TRANSIT at its Maintenance of Way equipment maintenance facility located in Wood-Ridge, N.J. It shall be the Contractor's responsibility to ensure that all vehicles pass inspection. Once inspected, vehicles will not need to be re-inspected for the duration of the phase unless maintenance work is performed on any of the vehicle track(s), suspension, steering or propulsion equipment, at which point the vehicle shall be re-inspected by NJ TRANSIT at it Wood-Ridge facility. Each individual vehicle must be inspected, even if it is of a type similar to one already approved and in use of similar design. The vehicle inspection conducted by NJ TRANSIT shall consider the following:

- Vehicle Inspection
- Tire and wheel condition
- · Rail wheel condition and wear
- Steering wheel locks for high rail if required
- High-rail safety locking pins
- High-rail mounting bolts
- Brakes
- · Headlights, tail lights, brake lights
- Horn
- Fire extinguisher
- Strobe light
- Back-up alarm

#### SP.10 ARTS IN TRANSIT (4.13)

Delete Article 4.13 of the General Provisions for Construction in its entirety.

### SP.11 CONSTRUCTION PROGRESS SCHEDULE (6.2)

Revise the 4th sentence of GP 6.2.6 by changing "monthly" to bi-weekly"

Add the following as new sub-article 6.2.16 of the General Provisions for Construction:

The Detailed Baseline Project Schedule shall identify all long-lead items essential to the major components of the project, indicating the projected duration of the submittal, approval and fabrication for each item. Upon review of the schedule, if in the view of NJ TRANSIT there are essential items that are not reflected in the schedule as long-lead items, the Contractor should be advised of such and include those items in a resubmittal of the Detailed Project Schedule.

Once the Detailed Baseline Project Schedule has been approved, the original duration indicated for the submittal, approval and fabrication of all long-lead items shall not change without an explanation supporting the change in the narrative report, and the approval of the changes by NJ TRANSIT.

#### SP.12 CONSTRUCTION LAYOUT (6.12)

Add the following as a new Article 6.12 Construction Layout of the General Provisions for Construction:

The Contractor shall provide all work required in connection with the layout for construction of the Project using the control points and data furnished by the NJ TRANSIT Construction Manager. The Contractor shall employ a NJ Licensed Land Surveyor to perform Construction Layout.

The Contractor shall furnish all necessary qualified personnel and adequate equipment to preserve such controls throughout the duration of the contract and shall lay out all of the lines and grades necessary for the complete construction of the project.

The Contractor shall provide the NJ TRANSIT Construction Manager with copies of all calculations and staking data at least two working days before the use of said stakeout information. The survey notes and data shall include, but not be limited to, field notes that were used to establish and set construction stakes, field notes for the alignment, horizontal, and vertical project control and field notes to document the easement and monument location.

The Contractor shall make all necessary computations to establish the exact position of all the work from the control points that are shown on the plans or furnished by the NJ TRANSIT Construction Manager. All the work shall be referenced to baselines which the Contractor shall establish from control points, re-establish when necessary, and maintain throughout the life of the contract so as not to delay the NJ TRANSIT Construction Manager from making necessary preliminary, interim, and final measurements and from checking the Contractor's layout if the NJ TRANSIT Construction Manager so desires.

The NJ TRANSIT Construction Manager will notify the Contractor, in writing, not less than five days in advance of when the baselines are required to be established by the Contractor.

The Contractor shall be responsible for the preservation of all control points furnished by the NJ TRANSIT Construction Manager for its use in staking out the work. If such control points are damaged, lost, displaced, or removed, they shall be reset at no cost to NJ TRANSIT.

The Contractor shall be responsible for maintaining the points it has established. Any error or apparent discrepancies found in the Plans or Specifications shall be called to the NJ TRANSIT Construction Manager's attention in writing for interpretation before proceeding with the work.

Payment for construction layout will not be made on a separate basis but will be covered in all bid items provided.

## SP.13 MATERIALS - WORKMANSHIP - LABOR (7.2)

Supplement Article 7.2.5 of the General Provisions for Construction with the following:

The Contractor assumes all the responsibility and liability associated with ensuring that all workers in its employment and that of its subcontractors have the capacity to fully understand all verbal and written training, instructions and commands given to them by NJ TRANSIT safety, security and supervisory personnel. The Contractor shall immediately remove and replace all workers who fail to strictly completely and immediately remove and replace all workers who fail to strictly completely and immediately comply with all NJ TRANSIT training, instructions and commands.

## SP.14 NEW JERSEY PREVAILING WAGE ACT (11.1)

Supplement Sub-article 11.1.1 of the General Provisions for Construction by adding the following:

ATTACHMENT A THE STATE OF NEW JERSEY PREVAILING WAGE DETERMINATION FOR SOMERSET COUNTY, NEW JERSEY (69 PAGES)

## SP.15 MEASUREMENT AND PAYMENT (12.1)

Supplement Sub-article 12.1.4 of the General Provisions for Construction by adding the following:

Unless otherwise mentioned in the Special Provisions the costs for work to be paid by an Allowance shall be measured and paid in accordance

with the order of priority of use set forth in Article 3.2.5 and the overhead and profit markups set forth in table 3.2.10 of the General Provisions.

## SP.16 EMPLOYEE PROTECTIONS - CONSTRUCTION ACTIVITIES (14.1)

Supplement Article 14.1.1 of the General Provisions for Construction by adding the following:

ATTACHMENT C: THE FEDERAL GENERAL WAGE DETERMINATION FOR SOMERSET COUNTY, NEW JERSEY (8 pages).

**END OF SECTION** 

ATTACHMENT A THE STATE OF NEW JERSEY PREVAILING WAGE DETERMINATION FOR SOMERSET COUNTY, NEW JERSEY (69 PAGES)



#### STATE OF NEW JERSEY

Department of Labor and Workforce Development
Division of Wage and Hour Compliance - Public Contracts Section
PO Box 389
Trenton, NJ 08625-0389

# PREVAILING WAGE RATE DETERMINATION

The New Jersey Prevailing Wage Act (N.J.S.A. 34:11-56.25 et seq.) requires that the Department of Labor and Workforce Development establish and enforce a prevailing wage level for workers engaged in public works in order to safeguard their efficiency and general well being and to protect them as well as their employers from the effects of serious and unfair competition.

Prevailing wage rates are wage and fringe benefit rates based on the collective bargaining agreements established for a particular craft or trade in the locality in which the public work is performed. In New Jersey, these rates vary by county and by the type of work performed.

Applicable prevailing wage rates are those wages and fringe benefits in effect on the date the contract is awarded. All pre-determined rate increases listed at the time the contract is awarded must also be paid, beginning on the dates specified. Rates that have expired will remain in effect until new rates are posted.

# **Prevailing Wage Rate**

The prevailing wage rate for each craft will list the effective date of the rate and the following information:

W = Wage Rate per Hour

B = Fringe Benefit Rate per Hour\*

T = Total Rate per Hour

\* Fringe benefits are an integral part of the prevailing wage rate. Employers not providing such benefits must pay the fringe benefit amount directly to the employee each payday. Employers providing benefits worth less than the fringe benefit amount must pay the balance directly to the employee each payday.

Unless otherwise stated in the Prevailing Wage Rate Determination, the fringe benefit rate for overtime hours remains at the straight time rate.

When the Overtime Notes in the Prevailing Wage Rate Determination state that the overtime rates are "inclusive of benefits," the benefit rate is increased by the same factor as the wage rate (i.e. multiplied by 1.5 for time and one-half, multiplied by 2 for double time, etc.).

### Apprentice Rate Schedule

An "apprentice" is an individual who is registered with the United States Department of Labor - Office of Apprenticeship and enrolled in a certified apprenticeship program during the period in which they are working on the public works project.

The apprentice <u>wage</u> rate is a percentage of the journeyman wage rate, unless otherwise indicated. The apprentice <u>benefit</u> rate is the full journeyman benefit rate, unless otherwise indicated.

If there is no apprentice rate schedule listed, the individual must be paid at least the journeyman rate even if that individual is in a certified apprentice program for that trade.

If there is no ratio of apprentices to journeymen listed for a particular craft, then the ratio shall be one (1) apprentice to every four (4) journeymen.

5/13/2014

#### Comments/Notes

For each craft listed there will be comments/notes that cover the definition of the regular workday, shift differentials, overtime, recognized holidays, and any other relevant information.

# **Public Works Contractor Registration**

The Public Works Contractor Registration Act (N.J.S.A. 34:11-56.48, et seq.) requires that all contractors, subcontractors, or lower tier subcontractors who are working on or who bid on public works projects register with the Department of Labor and Workforce Development. Applications are available at <a href="https://www.nj.gov/labor">www.nj.gov/labor</a> (click on Wage & Hour and then go to Registration & Permits).

### Pursuant to N.J.S.A. 34:11-56.51:

No contractor shall bid on any contract for public work as defined in section 2 of P.L. 1963, c. 150 (C.34:11-56.26) unless the contractor is registered pursuant to this act. No contractor shall list a subcontractor in a bid proposal for the contract unless the subcontractor is registered pursuant to P.L. 1999, c.238 (C.34:11-56.48 et seq.) at the time the bid is made. No contractor or subcontractor, including a subcontractor not listed in the bid proposal, shall engage in the performance of any public work subject to the contract, unless the contractor or subcontractor is registered pursuant to that act.

## **Snow Plowing**

Snow plowing contracts are <u>not</u> subject to the New Jersey Prevailing Wage Act or the Public Works Contractor Registration Act.

County - SOMERSET

Craft: Air Conditioning & Refrigeration - Service and Repair

PREVAILING WAGE RATE

	03/01/14
Journeyman (Mechanic)	W34.93
	B19.39
	T54.32

Expiration Date: 02/28/2015

Craft: Air Conditioning & Refrigeration - Service and Repair

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERI	OD AND RA	TES					
As Shown	Mo. 1-3	Mo. 4-12	2nd Year	3rd Year	4th Year	5th Year	Wage = %	of Jnymn	Wage
Wage and Bene	50%	55%	60%	65%	75%	85%	Bene = %	of Jnymn	Bene

Ratio of Apprentices to Journeymen - 1:4

APPRENTICE RATE SCHEDULE FOR THOSE APPRENTICES ENTERING PROGRAM AFTER 3-1-13:

**INTERVAL** 

PERIOD AND RATES

As Shown Wage and Benefit 1st Year 40%

2nd Year 3rd Year 50% 60%

r 4th Year 70%

ear 5th Year

Wage =% of Jnymn Wage

Bene. =% of Jnymn Wage

Craft: Air Conditioning & Refrigeration - Service and Repair

COMMENTS/NOTES

# THESE RATES MAY BE USED FOR THE FOLLOWING:

- Service/Repair/Maintenance Work to EXISTING facilities.
- Replacement or Installation of air conditioning and refrigeration equipment when the combined tonnage does not exceed 15 tons for refrigeration, or 25 tons for air conditioning.
- Replacement or Installation of "packaged" or "unitary" rooftop-type units when the combined tonnage of the units does not exceed 75 tons.

NOTE: These rates may NOT be used for any work in new construction (including work on new additions).

The regular workday shall consist of 8 hours, starting between 6:00 AM and 10:00 AM, Monday through Friday.

### SHIFT DIFFERENTIALS:

- The second and third shifts shall be paid an additional 15% of the hourly rate.
- All shifts must run for a minimum of 5 consecutive days.

## **OVERTIME:**

Hours in excess of 8 per day, hours before or after the regular workday that are not shift work, and all hours on Saturdays shall be paid at time and one-half the hourly rate. All hours on Sundays and holidays shall be paid at double the hourly rate.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Veterans' Day, Thanksgiving Day, Christmas Day.

County - SOMERSET

Craft: Boilermaker

PREVAILING WAGE RATE

	01/01/14
Foreman	W44.39
	B38.63
	T83.02
General Foreman	W46.39
	B39.61
	T86.00
Journeyman	W40.39
	B37.03
	T77.42

Expiration Date: 12/31/2014

Craft: Boilermaker

APPRENTICE RATE SCHEDULE

INTERVAL		PĘR	HOD AND R	ATES			<del></del>	 	
1000 Hours	65%	70%	75%	80%	85%	90%	95%		
Benefit =	30.94	31.81	32.69	33.55	34.43	35.30	36,16		

# Ratio of Apprentices to Journeymen - \*

\* 1 apprentice will be allowed for the first 5 journeymen, 1 apprentice for the next 10 journeymen and 1 apprentice for each succeeding 20 journeymen up to a maximum of 5 apprentices per contractor on any one job.

Craft: Boilermaker

COMMENTS/NOTES

HIGH WORK: All apprentices working on the erection, repair, or dismantling of smoke stacks, standpipes, or water towers shall be paid the Journeyman rate.

The regular workday shall consist of 8 hours, between 8:00 AM and 4:30 PM.

## SHIFT DIFFERENTIALS:

- The second shift shall work 7½ hours and receive 8 hours pay, at a rate equal to the regular hourly rate plus 10%.
- The third shift shall work 7 hours and receive 8 hours pay, at a rate equal to the regular hourly rate plus 20%.
- For "Municipal Water Works" projects only, the following shall apply: Two, four day, 10 hour shifts may be worked at straight time Monday through Thursday. The day shift shall work four days, at 10 hours, for 10 hours pay. The second shift shall work four days, at nine and a half hours, for 10 hours pay, plus 10% the hourly rate for new work and .25 cents on repair work. Friday may be used as a make-up day at straight time, due to weather conditions, hoilday or any other circumstances beyond the employer's control.

# **OVERTIME:**

- Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the hourly rate. All hours on Sundays and holidays (except Labor Day) shall be paid at double the hourly rate. All hours on Labor Day shall be paid at four times the hourly rate.
- If any other craft employed by the same contractor, or a subcontractor thereof, receives double time in lieu of time and one-half, then the Boilermaker shall receive double time in lieu of time and one-half.
- For "Municipal Water Works" projects only, the following shall apply: Four 10 hour days may be worked Monday through

County - SOMERSET

Thursday at straight time. Friday may be used as a make-up day for a day lost to inclement weather, holiday or other conditions beyond the control of the employer. Overtime shall be paid for any hours that exceed 10 hours per day or 40 hours per week.

RECOGNIZED HOLIDAYS: New Year's Day, Washington's Birthday, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day. Sunday holidays observed the following Monday.

County - SOMERSET

# Craft: Boilermaker - Minor Repairs

## PREVAILING WAGE RATE

	01/01/14
Foreman	W30.29
	B16.17
	T46.46
General Foreman	W30.79
	B16.17
	T46.96
Mechanic	W28.79
	B16.17
	T44.96

Expiration Date: 12/31/2014

Craft: Boilermaker - Minor Repairs

COMMENTS/NOTES

NOTE: These rates apply to MINOR REPAIR WORK ONLY (repair work in the field for which the contract amount does not exceed \$20,000.00).

# OVERTIME:

Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the hourly rate. All hours on Sundays and holidays (except Labor Day) shall be paid at double the hourly rate. All hours on Labor Day shall be paid at four times the hourly rate.

RECOGNIZED HOLIDAYS: New Year's Day, Washington's Birthday, Good Friday, Memorial Day, July 4th, Labor Day, Presidential Election Day, Thanksgiving Day, day after Thanksgiving, Christmas Day. Saturday holidays observed the following Monday.

County - SOMERSET

Craft: Bricklayer, Stone Mason

# PREVAILING WAGE RATE

	05/01/14
Deputy Foreman	W41.25
	B28.26
	T69.51
Foreman	W44.25
	B28.26
	T72.51
Journeyman	W38.25
	B28.26
	T66.51

Expiration Date: 10/31/2014

Craft: Bricklayer, Stone Mason

### APPRENTICE RATE SCHEDULE

INTERVAL		PE	RIOD AND R	ATES	·		<del></del>		 
6 Months	40%	50%	55%	60%	65%	70%	75%	80%	
Benefits	3.72	4.65	5.12	5.58	18.72	19.93	21.18	22.41	

Ratio of Apprentices to Journeymen - 1:5

Craft: Bricklayer, Stone Mason

COMMENTS/NOTES

The regular workday shall consist of 8 hours, between 6:00 AM and 4:30 PM.

### SHIFT DIFFERENTIALS:

- When a 2 shift schedule (including a day shift) is established, the first, or day shift, shall be established on an 8 hour basis. The second shift shall be established on an 8 hour basis, and receive the regular rate plus 15%, inclusive of benefits.
- When a three shift schedule is established, the first shift shall be established on an 8 hour basis, the second shift on a 7.5 hour basis, and the third shift on a 7 hour basis. The first shift shall receive the regular hourly rate, the second shift shall receive the regular rate plus 15%, inclusive of benefits, and the third shift shall receive the regular rate plus 20%, inclusive of benefits.
- When there is no day shift, and a second or third shift is established, it shall be established on an 8 hour basis. The second shift shall receive the regular rate plus 15%, inclusive of benefits, and the third shift shall receive the regular rate plus 20%, inclusive of benefits.

### OVERTIME:

- The first 2 hours in excess of 8 per day, or before or after the regular workday that are not shift work, Monday through Friday, shall be paid at time and one-half the regular rate, inclusive of benefits. Any additional overtime shall be paid at time and one-half the regular rate, inclusive of benefits. The first 10 hours on Saturday shall be paid at time and one-half the regular rate, inclusive of benefits. Any additional overtime shall be paid at double the regular rate, inclusive of benefits. All hours on Sundays and holidays shall be paid at double the regular rate, inclusive of benefits.
- Saturday may be used as a make-up day for hours lost to inclement weather.
- When Bricklayers/Stone Masons work on Saturday with Laborers, and no other crafts are working on the project for the day, benefits may be paid at straight time. If other crafts are present, the applicable overtime rate for benefits shall be paid.

County - SOMERSET

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day. Sunday holidays will be observed the following Monday.

#### County - SOMERSET

Craft: Carpenter

#### PREVAILING WAGE RATE

	05/01/14	11/01/14	05/01/15	11/01/15
Foreman	W49.57	W0.00	W0.00	W0.00
	B27.75 T77.32	B0.00 T77.92	B0.00 T79.17	B0.00 T80.42
Journeyman	W43.10	W0.00	W0.00	W0.00
	B24.13 T67.23	B0.00 T68.23	B0.00 T69.48	B0.00 T70.73

Expiration Date: 04/30/2016

Craft: Carpenter

### APPRENTICE RATE SCHEDULE

INTERVAL		PERIC	DD AND RAT	ES						
6 Months	40%	45%	50%	55%	60%	65%	70%	75%	85%	95%
Benefits	56% of	Appren	tice	Wage	Rate		for all	intervals		

Ratio of Apprentices to Journeymen - 1:4

Craft: Carpenter

**COMMENTS/NOTES** 

## FOREMAN REQUIREMENTS:

- When there are 2 or more Carpenters on a job, 1 shall be designated as a Foreman.
- When there are 21 or more Carpenters on a job, 2 shall be designated as Foremen.

The regular workday shall consist of 8 hours, starting between 7:00 AM and 9:00 AM.

## SHIFT DIFFERENTIALS:

- When a 2 shift schedule (including a day shift) is established, the day shift shall be established on an 8 hour basis. The second shift shall be established on an 8 hour basis, and receive the regular rate plus 15%, inclusive of benefits.
- When a three shift schedule is established, the first shift shall be established on an 8 hour basis, the second shift on a 7.5 hour basis, and the third shift on a 7 hour basis. The first shift shall receive the regular hourly rate, the second shift shall receive the regular rate plus 15% and the third shift shall receive the regular rate plus 20%, inclusive of benefits.
- When there is no day shift, and a second or third shift is established, it shall be established on an 8 hour basis. The second shift shall receive the regular rate plus 15% and the third shift shall receive the regular rate plus 20%, inclusive of benefits.

### OVERTIME:

- All hours in excess of 8 per day, or before or after an established shift that are not shift work, and all hours on Saturdays shall be paid at time and one-half the hourly rate, inclusive of benefits. All hours on Sundays and holidays shall be paid at double the hourly rate, inclusive of benefits.
- Four 10-hour days may be worked, Monday to Thursday, at straight time. Friday may be used as a make-up day for a day lost due to inclement weather. If Friday is not a make-up day, all hours on Friday shall be paid at time and one-half the hourly rate, inclusive of benefits.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election

County - SOMERSET

Day, Veterans' Day, Thanksgiving Day, Christmas Day. Sunday holidays observed the following Monday.

County - SOMERSET

Craft: Carpenter - Resilient Flooring

PREVAILING WAGE RATE

	05/01/14
Foreman	W49.57
	B27.75
	T77.32
Journeyman	W43.10
	B24.13
	T67.23

Expiration Date: 10/31/2014

Craft: Carpenter - Resilient Flooring

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIO	DD AND RAT	ES			<u></u>		<u>.</u> .	
6 Months	40%	45%	50%	55%	60%	65%	70%	75%	85%	95%
Benefit	56%	of	Appren	tice	Wage	Rate		for all	intervals	

Ratio of Apprentices to Journeymen - \*

\* 1 apprentice shall be allowed to every 2 journeymen or major fraction thereof. No more than 3 apprentices on any one job or project.

Craft: Carpenter - Resilient Flooring

COMMENTS/NOTES

# FOREMAN REQUIREMENTS:

- On any job where there are 4 or more Carpenters of Resilient Flooring, 1 must be designated a Foreman.

The regular workday consists of 8 hours, starting between 6:00 AM and 9:00 AM.

# SHIFT DIFFERENTIALS:

- When a 2 shift schedule (including a day shift) is established, the day shift, shall be established on an 8 hour basis. The second shift shall be established on an 8 hour basis, and receive the regular wage rate plus 15%.
- When a three shift schedule is established, the first shift shall be established on an 8 hour basis, the second shift on a 7.5 hour basis, and the third shift on a 7 hour basis. The first shift shall receive the regular wage rate plus 15% and the third shift shall receive the regular wage rate plus 20%.
- When there is no day shift, and a second or third shift is established, it shall be established on an 8 hour basis. The second shift shall receive the regular wage rate plus 15% and the third shift shall receive the regular wage rate plus 20%.

#### **OVERTIME:**

- Hours in excess of 8 per day or 40 per week, or before or after the regular workday, Monday through Friday, shall be paid at time and one-half the wage rate. Saturday may be used as a make-up day, at straight time, up to 8 hours, for hours lost to reasons beyond the control of the employer, up to a total of 40 hours per week; hours in excess of 8 on Saturday shall then be paid at time and one-half the wage rate. If Saturday is not a make-up day, all hours on Saturday shall be paid at time and one-half the wage rate. All hours on Sundays and holidays shall be paid at double the wage rate.
- Four 10-hour days may be worked, Monday to Thursday, at straight time. Friday may be used as a make-up day for hours lost to reasons beyond the control of the employer. If Friday is not a make-up day, all hours on Friday shall be paid at time and one-half the wage rate.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election

5/13/2014

**County - SOMERSET** 

Day, Veterans' Day, Thanksgiving Day, Christmas Day. Sunday holidays will be observed the following Monday.

5/13/2014 Page 12 of 69

**County - SOMERSET** 

Craft: Cement Mason

PREVAILING WAGE RATE

See " Bricklayer, Stone Mason" Rates

**Expiration Date:** 

Cruft: Cement Mason

**COMMENTS/NOTES** 

\*\*\*See Bricklayer, Stone Mason" Rates

## County - SOMERSET

Craft: Diver

PREVAILING WAGE RATE

	04/01/14
Diver	W53.21
	B40.45
	T93.66
Tender	W41.25
	B40.45
	T81.70

Expiration Date: 04/30/2014

Craft: Diver

**COMMENTS/NOTES** 

NOTE: All dive crews must consist of a Tender, a Diver, and a standby Diver (standby Diver is the same rate as a Diver).

DEPTH & PENETRATION RATES: Divers shall be paid the following depth and penetration rates, in addition to the regular hourly rate, when applicable:

AIR DIVES:

MIXED GAS DIVES:

0-59 feet: No additional wage 60-74 feet: + \$0.25 per foot 5-125 feet: + \$0.78 per foot 0-74 feet: No additional wage 75-125 feet: + \$0.78 per foot 126-200 feet: + \$1.60 per foot

126-200 feet: + \$1.60 per foot

# PENETRATION DIVES:

126-200 feet: + \$1.00 per foot 201-275 feet: + \$1.25 per foot 276-350 feet: + \$1.50 per foot 351-425 feet: + \$2.00 per foot

# OVERTIME:

Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the hourly rate. All hours on Sundays and holidays shall be paid at double the hourly rate.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Veterans' Day, Presidential Election Day, Thanksgiving Day, Christmas Day. Veterans' Day may be switched with the day after Thanksgiving.

#### County - SOMERSET

Craft: Dockbuilder

PREVAILING WAGE RATE

	04/01/14
Foreman	W47.69
	B40.45
	T88.14
Foreman	W47.69
(Concrete Form Work)	B32.61
	T80.30
Journeyman	W41.47
	B40.45
	T81.92
Journeyman	W41.47
(Concrete Form Work)	B32.61
	T74.08

Expiration Date: 04/30/2014

Craft: Dockbuilder

### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES								
Yearty	16.59	5.59 20.73 26.95 33.18								
Benefits	27.57	for all	intervals		Concrete	Form	only -	Benefit=	22,05 ali	intervals

# Ratio of Apprentices to Journeymen - \*

Craft: Dockbuilder

COMMENTS/NOTES

### **CREOSOTE HANDLING:**

When handling creosote products on land piledriving, floating marine construction, and construction of wharves, the worker shall receive an additional \$0.25 per hour.

#### HAZARDOUS WASTE WORK:

- Hazardous waste removal work on a state or federally designated hazardous waste site where Level A, B, or C personal protection is required: an additional 20% of the hourly rate, per hour.
- Hazardous waste removal work in Level D, or where personal protection is not required; an additional \$1.00 per hour.

CERTIFIED WELDER: When required on the job by the project owner, a Certified Welder shall receive an additional \$1.00 per hour.

## **FOREMAN REQUIREMENTS:**

The first Dockbuilder on the job shall be designated a Foreman.

#### **OVERTIME:**

Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the

5/13/2014 Page 15 of 69

<sup>\*</sup> When there are 4 or fewer Dockbuilders on a job, no more than 1 may be an apprentice. When there are 5 or more Dockbuilders, there may be 1 apprentice for every 5 Dockbuilders.

**County - SOMERSET** 

hourly rate. All hours on Sundays and holidays shall be paid at double the hourly rate.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Veterans' Day, Presidential Election Day, Thanksgiving Day, Christmas Day. Veterans' Day may be switched with the day after Thanksgiving.

**County - SOMERSET** 

Craft: Drywall Finisher

PREVAILING WAGE RATE

	05/08/14
Foreman	W41.80
	B21.60
	T63.40
eneral Foreman	W43.70
	B21.60
	T65.30
Journeyman	W38.00
	B21.60
	T59.60

Expiration Date: 04/30/2015

Craft: Drywall Finisher

APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES									
4 Months	30%	40%	50%	60%	70%	75%	80%	85%	90%		
Benefits	Intervals	1 to 3 =	8.85	Intervals	4 to 6 =	11.28	Intervals	7 to 9 =	14.45		

Ratio of Apprentices to Journeymen - 1:4

Craft: Drywall Finisher

**COMMENTS/NOTES** 

The regular workday shall consist of 8 hours between 7:00 AM and 5:30 PM.

### SHIFT DIFFERENTIALS:

- The second shift shall receive an additional 10% of the hourly rate, per hour, and the third shift shall receive an additional 15% of the hourly rate, per hour.
- When 3 shifts are worked, the second shift shall receive 8 hours pay for 7.5 hours of work, and the third shift shall receive 8 hours pay for 7 hours of work.
- Shift work must run for a minimum of 5 consecutive workdays.

#### OVERTIME:

- Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the regular rate, inclusive of benefits. All hours on Sundays and holidays shall be paid at double the regular rate, inclusive of benefits.
- Saturday or Sunday may be used to make up a day lost to inclement weather, at straight time.

RECOGNIZED HOLIDAYS: New Year's Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day. Saturday holiday observed the preceding Friday. Sunday holiday observed the following Monday.

## County - SOMERSET

Craft: Electrician

PREVAILING WAGE RATE

	06/03/13	06/02/14
	W55.17	W56.29
	B31.15	B32.17
	T86.32	T88.46
Foreman	W56.17	W57.31
	B31.71	B32.76
	T87.88	T90.07
Journeyman	W50.15	W51.17
	B28.31	B29.25
	T78.46	T80.42

Expiration Date: 05/31/2015

Craft: Electrician

APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES									
Yearly	40%	49%	58%	68%	80%		of Jour	neyman	Wage	Rate	
Benefit	40%	49%	58%	68%	80%		of Jour	neyman	Benefit	Rate	

Ratio of Apprentices to Journeymen - 2:3

Craft: Electrician

COMMENTS/NOTES

# THESE RATES ALSO APPLY TO THE FOLLOWING:

- -All burglar and fire alarm work.
- -All fiber optic work.
- -Teledata work in new construction.
- -Teledata work involving 16 Voice/Data Lines or more.

The regular workday shall be 8 hours, between 8:00 AM and 4:30 PM.

#### FOREMAN REQUIREMENTS:

- On any job where there are 1 to 10 Journeymen electricians, 1 shall be designated a Foreman.

# SHIFT DIFFERENTIALS:

- Shift work must run for a minimum of 5 consecutive workdays.
- 2nd Shift (4:30 PM to 12:30 AM) shall receive 8 hours pay for 7.5 hours work + an additional 10% of the regular rate, per hour, inclusive of benefits.
- 3rd Shift (12:30 AM to 8:00 AM) shall receive 8 hours pay for 7 hours work + an additional 15% of the regular rate, per hour, inclusive of benefits.

## **OVERTIME:**

Hours in excess of 8 per day, or outside of the regular workday, Monday through Friday, and all hours on Saturdays, shall be paid at time and one-half the regular rate, inclusive of benefits. All hours on Sundays and holidays shall be paid at double the regular rate, inclusive of benefits.

5/13/2014 Page 18 of 69

County - SOMERSET

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day.

#### **County - SOMERSET**

Craft: Electrician - Teledata (15 Voice/Data Lines & Less)

PREVAILING WAGE RATE

	04/29/13
Master	W48.93
Technician/General	B27.63
Foreman	T76.56
Senior Technician/Lead	W44.79
Foreman	B25.29
(21-30 Workers on Job)	T70.08
Technician A/Foreman	W42.91
(11-20 Workers on Job)	B24.23
	T67.14
Technician B/Working	W41.02
Foreman	B23.16
(4-10 Workers on Job)	T64.18
Technician C/Journeyman	W37.64
(1-3 Workers on Job)	B21.25
	T58.89

Expiration Date: 10/31/2014

Craft: Electrician - Teledata (15 Voice/Data Lines & Less)

# APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES									
6 months	18.12	19.32	21.34	23.75	26.57	28.99	31.80	34.62			
Benefits	7.52	8.01	8.85	9.85	11.02	12.02	13.19	14.35			

Ratio of Apprentices to Journeymen - 2:3

Craft: Electrician - Teledata (15 Voice/Data Lines & Less)

**COMMENTS/NOTES** 

APPRENTICE RATE SCHEDULE AS OF 4-29-13:

INTERVAL

PERIOD AND RATES

6 Months 18.76 20.01 22.10 24.60 27.52 30.02 32.94 35.86 Benefits 7.78 8.30 9.17 10.20 11.41 12.45 13.66 14.87

### NOTES:

- These rates are for service, maintenance, moves, and/or changes affecting
- 15 Voice/Data (teledata) lines or less. These rates may NOT be used for any teledata work in new construction (including additions) or any fiber optic work.
- The number of Teledata workers on the jobsite is the determining factor for which Foreman category applies .

The regular workday shall be 8 hours, between 8:00 AM and 4:30 PM.

## SHIFT DIFFERENTIALS:

5/13/2014 Page 20 of 69

## **County - SOMERSET**

- Shift work must run for a minimum of 5 consecutive workdays.
- 2nd Shift (4:30 PM to 12:30 AM) shall receive 8 hours pay for 7.5 hours work + an additional 10% of the regular rate, per hour, inclusive of benefits.
- 3rd Shift (12:30 AM to 8:00 AM) shall receive 8 hours pay for 7 hours work + an additional 15% of the regular rate, per hour, inclusive of benefits.

## OVERTIME:

Hours in excess of 8 per day, or outside of the regular workday, Monday through Friday, and all hours on Saturdays, shall be paid at time and one-half the regular rate, inclusive of benefits. All hours on Sundays and holidays shall be paid at double the regular rate, inclusive of benefits.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day.

# County - SOMERSET

Craft: Electrician - Teledata (16 Instruments & More)	PREVAILING WAGE RATE
See "Electrician" Rates	
Expiration Date:	
Craft: Electrician - Teledata (16 Instruments & More)	COMMENTS/NOTES

\*\*\*See ELECTRICIAN Rates\*\*\*

**County - SOMERSET** 

# Craft: Electrician- Outside Commercial

# PREVAILING WAGE RATE

	06/03/13	06/02/14
Cable Splicer	W55.34	W56.56
	B30.97	B31.92
	T86.31	T88.48
Certified Welder	W52.83	W53.99
	B29.56	B30.46
	T82.39	T84.45
Equipment Operator	W50.31	W51.42
	B28.15	B29.01
	T78.46	T80.43
Foreman (1-10	W56.35	W57.59
Journeyman workers on	B31.53	B32.49
job)	T87.88	T90.08
Foreman (11-20	W57.86	W59.13
Journeyman workers on	B32.38	B33.37
job)	T90.24	T92.50
General Foreman (21-30	W59.37	W60.67
	B33.22	B34.23
job)	T92.59	T94.90
General Foreman (31-60	W64.40	W65.82
	B36.04	B37.13
job)	T100.44	T102.95
Groundman	W30.19	W30.85
	B16.90	B17.41
	T47.09	T48.26
Journeyman	W50.31	W51.42
Lineman/Technician	B28.15	B29.01
	T78.46	T80.43
Sub-Foreman	W56.35	W57.59
	B31.53	B32.49
General Foreman (21-30 dourneyman workers on ob)  General Foreman (31-60 dourneyman workers on ob)  Groundman  Gourneyman ineman/Technician	T87.88	T90.08

Expiration Date: 06/01/2015

# Craft: Electrician- Outside Commercial

# APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES									
1000 Hours	60%	65%	70%	75%	80%	85%	90%				
Benefits	55.95% of	Journey	man	wage	+\$.01						

**County - SOMERSET** 

Craft: Electrician- Outside Commercial

COMMENTS/NOTES

APPRENTICE RATE SCHEDULE AS OF 6-2-14:

INTERVAL

PERIOD AND RATES

1000 Hours 60% 65% 70% 75% 80% 85% 90%

**Benefits** 

56.41% of Journeyman wage + \$.01

The regular worday shall be 8 hours, between 8:00 AM and 4:30 PM.

# FOREMAN REQUIREMENTS:

On jobs where there are 4 Journeymen, one shall be a Foreman. The following number of Foreman, General Foreman, Assistant General Foreman and Sub-Foreman shall be required with respect to number of Journeymen on site: 4-10 Journeymen ( 1 Foreman)

11-20 Journeymen (1 Foreman and 1 Sub-Foreman)

21-30 Journeymen (1 Foreman and 2 Sub-Foremen)

# SHIFT DIFFERENTIALS:

Shift work must run for a minimum of 5 consecutive workdays.

2nd Shift (4:30 PM to 12:30 AM): 8 hrs. pay for 7.5 hrs. work + an additional 10% of the regular rate, inclusive of

3rd Shift (12:30 AM to 8:00 AM): 8 hrs. pay for 7 hrs. work + an additional 15% of the regular rate per hour, inclusive benefits.

#### OVERTIME:

Hours in excess of 8 per day, or outside of the regular workday, Monday through Friday, and all hours on Saturdays, shall be paid at time and one-half the regular rate, inclusive of benefits. All hours on Sundays and holidays shall be paid at double the regular rate, inclusive of benefits.

### RECOGNIZED HOLIDAYS:

New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day and Christmas Day.

<sup>\*</sup> FOR UTILITY WORK PLEASE SEE STATEWIDE RATES

County - SOMERSET

Craft: Electrician-Utility Work (North)

PREVAILING WAGE RATE

Rates are located in the "Statewide" rate package

Expiration Date:

Craft: Electrician-Utility Work (North)

# APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES									
6 Months	60%	65%	70%	75%	80%	85%	90%				
Benefits	62.5% of	Appren	tice	Wage	Rate	for all	intervals				

Craft: Electrician-Utility Work (North)

**COMMENTS/NOTES** 

Electrician-Utility Work (North) rates are located in the "Statewide" rate package.

**County - SOMERSET** 

Craft: Electrician-Utility Work (South)

PREVAILING WAGE RATE

Rates are located in the "Statewide" rate package

**Expiration Date:** 

Craft: Electrician-Utility Work (South)

# APPRENTICE RATE SCHEDULE

INTERVAL		PERI	OD AND RAT	ES			<del></del>	 <del></del>	_
6 Months	25.69	27.83	29.97	32.12	34.26	36.40	38.54		
Benefits	20.89	22.06	23.22	24.40	25.56	26.73	27.91	-	

Craft: Electrician-Utility Work (South)

**COMMENTS/NOTES** 

Electrician-Utility Work (South) rates are located in the "Statewide" rate package.

County - SOMERSET

Craft: Elevator Constructor

PREVAILING WAGE RATE

	03/20/14
Journeyman	W58.23
	B29.76
	T87.99

Expiration Date: 03/16/2015

Craft: Elevator Constructor

# APPRENTICE RATE SCHEDULE

INTERVAL		PERIO	DD AND RAT	ES		 <u></u>		
Yearly	26.06	32.03	37.85	43.67				
Benefits	25.76	26.16	26.96	27.76			_	

Ratio of Apprentices to Journeymen - 1:1

Craft: Elevator Constructor

COMMENTS/NOTES

The regular workday shall consist of either 7 or 8 hours to be established at the beginning of the project, between 7:00 AM and 4:30 PM.

#### OVERTIME:

For all hours worked before or after the regular workday, Monday through Friday, and all hours on Saturday and Sunday, shall be paid at double the hourly rate. Holiday pay is one days wages (8 hours) plus double the hourly rate for all hours worked.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Good Friday, Memorial Day, July 4th, Labor Day, Columbus Day, Veterans' Day, Thanksgiving Day and the day after, Christmas Day. Saturday holidays shall be observed on the previous Friday and Sunday holidays shall be observed on the following Monday.

**County - SOMERSET** 

Craft: Elevator Modernization & Service

PREVAILING WAGE RATE

03/20/14
W46.00
B29.61
T75,61

Expiration Date: 03/16/2015

Craft: Elevator Modernization & Service

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIO	OD AND RAT	ES
Yearly	26.06	25.30	29.90	34.50
Benefits	25.68	26.08	26.86	27.65

Ratio of Apprentices to Journeymen - 1:1

Craft: Elevator Modernization & Service

### **COMMENTS/NOTES**

MODERNIZATION (addition, replacement, refurbishing, relocation, or changes in design or appearance, of elevator equipment in existing buildings):

- The regular workday consists of 8 hours, between 7:00 AM and 4:30 PM.
- Overtime:

Hours in excess of 8 per day, or before or after the regular workday, Monday through Friday, and all hours on Saturday and Sunday shall be paid at time and one-half the hourly rate. Holiday pay is one days wages (8 hours) plus time and one-half the hourly rate for all hours worked.

SERVICE (repair or replacement of parts for the purpose of maintaining elevator equipment in good operating condition):

- The regular workday consists of 8 hours, between 6:00 AM and 6:00 PM.
- Overtime:

Hours in excess of 8 per day, or before or after the regular workday, Monday through Friday, and all hours on Saturday shall be paid at time and one-half the hourly rate. All hours on Sunday and holidays shall be paid at double the hourly rate.

RECOGNIZED HOLIDAYS (Modernization and Service): New Year's Day, Presidents' Day, Good Friday, Memorial Day, July 4th, Labor Day, Columbus Day, Veterans' Day, Thanksgiving Day and the day after, Christmas Day. Saturday holidays shall be observed on the previous Friday and Sunday holidays shall be observed on the following Monday.

## **County - SOMERSET**

Craft: Glazier

PREVAILING WAGE RATE

	05/01/13
Foreman	W44.94
	B21.21
	T66.15
General Foreman	W46.94
	B21.45
73	T68.39
Journeyman	W40.94
	B20.73
	T61.67

Expiration Date: 04/30/2014

Craft: Glazier

APPRENTICE RATE SCHEDULE

INTERVAL		PERI	OD AND RAT	TES .	<u>.</u>				
4 Months	30%	40%	50%	60%	70%	75%	80%	85%	90%
Benefits	Intervals	1 to 3 =	6.34	Intervals	4 to 6 =	9.04	Intervals	7 to 9 =	11,33

Ratio of Apprentices to Journeymen - 1:4

Craft: Glazier

**COMMENTS/NOTES** 

Hazard/Height Pay: +\$1.00 per hour

# FOREMAN REQUIREMENTS:

- When there are 4 or more Glaziers on a job, 1 must be designated a Foreman.
- When there are 15 or more Glaziers on a job, 1 must be designated a General Foreman.

The regular workday shall consist of 8 hours, between 7:00 AM and 5:30 PM, Monday to Friday.

## SHIFT DIFFERENTIALS:

- The second shift shall receive an additional 10% of the hourly rate, per hour, and the third shift shall receive an additional 15% of the hourly rate, per hour.
- When 3 shifts are worked, the second shift shall receive 8 hours pay for 7.5 hours of work, and the third shift shall receive 8 hours pay for 7 hours of work.

### **OVERTIME:**

Hours in excess of 8 per day, or before or after the regular workday Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the regular rate. All hours on Sundays and holidays shall be paid at double the regular rate.

RECOGNIZED HOLIDAYS: New Year's Day, Memorial Day, July 4th, Labor Day, General Election Day, Veterans' Day, Thanksgiving Day, Christmas Day. Saturday holiday observed the preceding Friday. Sunday holiday observed the following Monday.

5/13/2014

County - SOMERSET

Craft: Heat & Frost Insulator

# PREVAILING WAGE RATE

	07/01/13	07/01/14	07/01/15	07/01/16
Foreman	W41.32	W0.00	W0.00	W0.00
	B30.33	B0.00	B0.00	B0.00
	T71.65	T72.80	T73.90	T75.00
Journeyman	W40.32	W0.00	W0.00	W0.00
	B30.33	B0.00	B0.00	B0.00
	T70.65	T71.80	T72.90	T74.00

Expiration Date: 06/30/2017

Craft: Heat & Frost Insulator

# APPRENTICE RATE SCHEDULE

INTERVAL		PEI	RIOD AND R	ATES				
Yearly	45%	55%	65%	75%	80%			
Benefit	27.08	for	all	intervals				

# Ratio of Apprentices to Journeymen - \*

\* Ratio = 1:4 on a "company-wide" basis (i.e. the total number of apprentices and journeymen employed by the company). There is no limit to the number of apprentices allowed on any one job, provided there is at least 1 journeyman on the job.

Craft: Heat & Frost Insulator

COMMENTS/NOTES

# FOREMAN REQUIREMENTS:

- If there is only 1 Insulator on the job, he must be designated a Foreman.
- If there are 2 to 10 Insulators on the job, 1 must be designated a Foreman.
- If there are 11 or more insulators on the job, 1 must be designated a General Foreman and receive the following additional pay (% above Journeyman wage rate):
  - 11 20 Insulators on site: 10%; 21 30 Insulators on site: 15%;
  - 31 40 Insulators on site: 20%; 41 50 Insulators on site: 25%

The regular workday shall be 8 hours between 7:00 AM and 3:30 PM.

## SHIFT DIFFERENTIALS:

- Shift work must run for a minimum of two (2) consecutive days and a minimum of two (2) shifts per day must be worked Additionally, no less than two (2) employees may work on any one (1) shift. If these requirements are not met then shift work would not apply and the applicable overtime rate shall be paid.
- 1st Shift (8:00 AM- 4:00 PM).
- 2nd Shift (4:00 PM 12:00 AM): additional 15% of the regular rate, inclusive of benefits.
- 3rd Shift (12:00 AM 8:00 AM): additional 20% of the regular rate, inclusive of benefits.

# OVERTIME:

- Hours in excess of 8 per day, or before or after the regular workday Monday through Friday, that are not shift work, and all hours on Saturdays, shall be paid at time and one-half the regular rate, inclusive of benefits. All hours on Sundays and holidays (except Labor Day) shall be paid at double the regular rate, inclusive of benefits. All hours on Labor Day shall be paid at triple the regular rate, inclusive of benefits.

RECOGNIZED HOLIDAYS: New Year's Day, Memorial Day, July 4th, Labor Day, Veterans' Day, Presidential Election Day,

5/13/2014

County - SOMERSET

Thanksgiving Day, Christmas Day. Saturday holidays observed the preceding Friday, Sunday holidays observed the following Monday.

County - SOMERSET

Craft: Heat & Frost Insulator - Asbestos Worker

PREVAILING WAGE RATE

=	07/01/13	07/01/14	07/01/15	07/01/16
Foreman	W41.32	W0.00	W0.00	W0.00
	B30.33	B0.00	B0.00	B0.00
	T71.65	T72.80	T73.90	T75.00
Journeyman	W40.32	W0.00	W0.00	W0.00
	B30.33	B0.00	B0.00	B0.00
	T70.65	T71.80	T72.90	T74.00

Expiration Date: 06/30/2017

Craft: Heat & Frost Insulator - Asbestos Worker

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIO	OD AND RAT	ES	<del></del>	<del></del> -	<del></del> -	 <u> </u>
	SEE	Heat &	Frost	Insulator				

Craft: Heat & Frost Insulator - Asbestos Worker

#### **COMMENTS/NOTES**

NOTE: These rates apply only to the REMOVAL of insulation materials/asbestos from mechanical systems, including containment erection and demolition, and placing material in appropriate containers.

# FOREMAN REQUIREMENTS:

- If there is only 1 Asbestos Worker on the job, he must be designated an Abatement Foreman.
- If there are 2 to 10 Asbestos Workers on the job, 1 must be designated an Abatement Foreman.
- If there are 11 or more Asbestos Workers on the job, 1 must be designated a General Foreman and receive the following additional pay (% above Abatement Mechanic wage rate):
  - 11 20 Insulators on site: 10%; 21 30 Insulators on site: 15%;
  - 31 40 Insulators on site: 20%; 41 50 Insulators on site: 25%

# **MECHANIC-TO-APPRENTICE RATIO:**

- Maximum of 5 Apprentices for each Abatement Mechanic on the job.

## OVERTIME:

- Hours in excess of 8 per day, and all hours on Saturdays, shall be paid at time and one-half the regular rate, inclusive of benefits. All hours on Sundays and holidays (except Labor Day) shall be paid at double the regular rate, inclusive of benefits.

RECOGNIZED HOLIDAYS: New Year's Day, Memorial Day, July 4th, Labor Day, Veterans' Day, Presidential Election Day, Thanksgiving Day, Christmas Day. Saturday holidays observed the preceding Friday, Sunday holidays observed the following Monday.

#### County - SOMERSET

Craft: Ironworker

PREVAILING WAGE RATE

	07/19/13	07/01/14
Rod Foreman	W40.24	W0.00
	B40.52	B0.00
	T80.76	T82.26
Rod Journeyman	W37.24	W0.00
	B40.52	B0,00
	T77.76	179.26
Structural Foreman	W43.04	W0.00
	B40.52	B0.00
	T83.56	T85.31
Structural Journeyman	W40.04	W0.00
	B40.52	B0.00
<u> </u>	T80.56	T82.31

Expiration Date: 06/30/2015

Craft: Ironworker

APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES										
6 Months	50%	60%		Yearly	70%	80%	90%					

Ratio of Apprentices to Journeymen - 1:4

Craft: Ironworker

COMMENTS/NOTES

HAZARDOUS WASTE WORK: On hazardous waste removal work on a state or federally designated hazardous waste site where the Ironworker is required to wear Level A,B, or C personal protection: + \$3.00 per hour

The regular workday consists of 8 hours between 6:00 AM and 4:30 PM.

# FOREMAN REQUIREMENTS:

When there are 2 or more Ironworkers on a job, 1 shall be designated a Foreman.

### SHIFT DIFFERENTIALS:

- When a 2 shift schedule is established, the first, or day shift, shall be established on an 8 hour basis. The second shift shall be established on an 8 hour basis, and receive the regular rate plus 15%.
- When a three shift schedule is established, the first shift shall be established on an 8 hour basis, the second shift on a 7.5 hour basis, and the third shift on a 7 hour basis. The first shift shall receive the regular hourly rate, the second shift shall receive the regular rate plus 15%, and the third shift shall receive the regular rate plus 20%.
- When there is no day shift, and a second or third shift is established, it shall be established on an 8 hour basis.
- When an irregular shift is established for 5 consecutive days, the rate shall be paid at the regular rate and benefit rate, with no wage premium included. When an irregular shift is established for less than 5 days, the rate shall be paid at the regular rate plus 15%.

County - SOMERSET

#### OVERTIME:

- All hours in excess of 8 per day, or before or after an established shift that are not shift work, and all hours on Saturday, shall be paid at time and one-half the regular rate, inclusive of benefits. All hours on Sunday and holidays shall be paid at double the hourly rate, inclusive of benefits. Saturday may be used as a make-up day for a day lost to inclement weather. If Saturday is not a make-up day, all hours on Saturday shall be paid at time and one-half the hourly rate, inclusive of benefits.
- Four 10-hour days may be worked, Monday to Thursday, at straight time. Friday may be used as a make-up day for a day lost to inclement weather. If Friday is not a make-up day, all hours on Friday shall be paid at time and one-half the hourly rate, inclusive of benefits.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans Day, Thanksgiving Day, Christmas Day.

#### **County - SOMERSET**

Craft: Laborer - Ashestos & Hazardous Waste Removal

PREVAILING WAGE RATE

	04/24/14	12/01/14
Foreman	W32.42	W0.00
	B22.87	B0.00
	T55.29	T55.79
Journeyman (Handler)	W28.87	W0.00
	B22.87	B0.00
	T51.74	T52.24

Expiration Date: 11/30/2015

Craft: Laborer - Asbestos & Hazardous Waste Removal

# APPRENTICE RATE SCHEDULE

INTERVAL	PERIOD AND RATES									
6 Months	60%	70%	80%	90%						
Benefits	21.22	for	all	intervals						

Ratio of Apprentices to Journeymen - \*

Craft: Laborer - Asbestos & Hazardous Waste Removal

## COMMENTS/NOTES

NOTE: These rates apply to work in connection with Asbestos, Radiation, Hazardous Waste, Lead, Chemical, Biological, Mold Remediation and Abatement.

The regular workday shall be 8 hours between 6:00 AM and 6:00 PM.

### OVERTIME:

- Hours in excess of 8 per day, Monday through Saturday, and all hours on Sunday and holidays shall be paid at time and one-half the regular rate.
- When the owner (Public Body) mandates that work is to be performed on Sunday, those hours may be worked at straight time, up to 8 hours per day, up to 40 hours per week.
- Benefits on ALL overtime hours shall be paid at straight time.

RECOGNIZED HOLIDAYS: New Year's Day, President's Day, Good Friday, Easter, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day.

<sup>\*</sup> Ratio of apprentices to journeymen shall not be more than one apprentice for the first journeyman and no more than one (1) apprentice for each additional three (3) journeymen.

**County - SOMERSET** 

Craft: Laborer - Building

PREVAILING WAGE RATE

	05/13/14
Class A Journeyman	W30.65
	B23.37
	T54.02
Class B Journeyman	W30.15
	B23.37
	T53.52
Class C Journeyman	W25.63
	B23.37
	T49.00
Foreman	W34.48
	B23.37
	T57.85
General Foreman	W38.31
	B23.37
	T61.68

Expiration Date: 04/30/2015

Craft: Laborer - Building

# APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES									
6 Months	60%	70%	80%	90%			T				
Benefit	20.92	20.92	20.92	20.92							

Ratio of Apprentices to Journeymen - \*

\* Ratio of apprentices to journeymen shall not be more than one apprentice for the first journeyman and no more than one (1) apprentice for each additional three (3) journeymen.

Craft: Laborer - Building

# COMMENTS/NOTES

CLASS A: Specialist laborer including mason tender or concrete pour crew; scaffold builder (scaffolds up to 14 feet in height); operator of forklifts, Bobcats (or equivalent machinery), jack hammers, tampers, motorized tampers and compactors, vibrators, street cleaning machines, hydro demolition equipment, riding motor buggies, conveyors, burners; and nozzlemen on gunite work.

CLASS B: Basic laborer - includes all laborer work not listed in Class A or Class C.

CLASS C: Janitorial-type light clean-up work associated with the TURNOVER of a project, or part of a project, to the owner. All other clean-up work is Class B.

The regular workday shall be 8 hours between 6:00 AM and 6:00 PM.

# SHIFT DIFFERENTIALS:

- Shift work must run for a minimum of 5 consecutive workdays.
- When a 2-shift schedule is worked, including a day shift, both shifts shall be establised on the basis of 8 hours pay for 8 hours worked. The second shift shall receive the regular rate plus an additional 10%.

#### **County - SOMERSET**

- When a 3-shift schedule is worked, the day shift shall be established on the basis of 8 hours pay for 8 hours worked, the second shift shall be established on the basis of 8 hours pay for 7.5 hours worked, and the third shift shall be established on the basis of 8 hours pay for 7 hours worked. The day shift shall receive the regular rate, the second shift shall receive the regular rate plus an additional 10%, and the third shift shall receive the regular rate plus an additional 15%.
- When a second or third shift is worked with no day shift, the second or third shift shall be established on the basis of 8 hours pay for 8 hours worked. The second shift shall receive the regular rate plus an additional 10%, and the third shift shall receive the regular rate plus an additional 15%.

### OVERTIME:

- Hours in excess of 8 per day, or outside the regular workday that are not shift work, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the regular rate. All hours on Sundays and holidays shall be paid at double the regular rate.
- Four 10-hour days may be worked Monday to Thursday, at straight time, with Friday used a make-up day for a day lost to inclement weather. If Friday is not a make-up day, all hours on Friday shall be paid at time and one-half the regular rate.
- Benefits on ALL overtime hours shall be paid at time and one-half.

RECOGNIZED HOLIDAYS: New Year's Day, Washington's Birthday, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day, Sunday holidays observed the following Monday.

**County - SOMERSET** 

PREVAILING WAGE RATE

Rates are located in the "Statewide" rate package

**Expiration Date:** 

Craft: Laborer - Heavy & General

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIO	DD AND RAT	ES		-		 
1000 Hours	60%	70%	80%	90%			_	
Benefit	16.28	for	all	intervals			_	

Ratio of Apprentices to Journeymen - \*

Craft: Laborer - Heavy & General

**COMMENTS/NOTES** 

Heavy & General Laborer rates are located in the "Statewide" rate package.

<sup>\*</sup> No more than 1 apprentice for the first journeyman and no more than 1 apprentice for each additional 3 journeymen.

#### County - SOMERSET

Craft: Millwright

#### PREVAILING WAGE RATE

	05/01/14	11/01/14	05/01/15	11/01/15
Foreman	W50.46	W0.00	W0.00	W0.00
	B28.25	B0.00	B0.00	B0.00
	T78.71	T79.34	T80.59	T81.84
Journeyman	W43.88	W0.00	W0.00	W0.00
	B24.57	B0.00	B0.00	B0.00
	T68.45	T69.45	T70.70	T71.95

Expiration Date: 04/30/2016

Craft: Millwright

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES								
6 Months	40%	45%	50%	55%	60%	65%	70%	75%	85%	95%
Benefits	56% of	Appren	tice	Wage	Rate					-

Ratio of Apprentices to Journeymen - 1:4

Craft: Millwright

COMMENTS/NOTES

#### FOREMAN REQUIREMENTS:

- When there are 2 or more Millwrights on a job, 1 shall be designated as a Foreman.
- When there are 21 or more Millwrights on a job, 2 shall be designated as Foremen.

The regular workday shall consist of 8 hours, starting between 7:00 AM and 9:00 AM.

#### SHIFT DIFFERENTIALS:

- When a 2 shift schedule (including a day shift) is established, the day shift shall be established on an 8 hour basis. The second shift shall be established on an 8 hour basis, and receive the regular rate plus 15%, inclusive of benefits.
- When a three shift schedule is established, the first shift shall be established on an 8 hour basis, the second shift on a 7.5 hour basis, and the third shift on a 7 hour basis. The first shift shall receive the regular hourly rate, the second shift shall receive the regular rate plus 15% and the third shift shall receive the regular rate plus 20%, inclusive of benefits.
- When there is no day shift, and a second or third shift is established, it shall be established on an 8 hour basis. The second shift shall receive the regular rate plus 15% and the third shift shall receive the regular rate plus 20%, inclusive of benefits.

#### OVERTIME:

- All hours in excess of 8 per day, or before or after an established shift that are not shift work, and all hours on Saturdays shall be paid at time and one-half the hourly rate, inclusive of benefits. All hours on Sundays and holidays shall be paid at double the hourly rate, inclusive of benefits.
- Four 10-hour days may be worked, Monday to Thursday, at straight time. Friday may be used as a make-up day for a day lost due to inclement weather. If Friday is not a make-up day, all hours on Friday shall be paid at time and one-half the hourly rate, inclusive of benefits.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election

5/13/2014 Page 39 of 69

**County - SOMERSET** 

Day, Veterans' Day, Thanksgiving Day, Christmas Day. Sunday holidays will be observed the following Monday.

5/13/2014 Page 40 of 69

**County - SOMERSET** 

Craft: Operating Engineer

PREVAILING WAGE RATE

Rates are located in the "Statewide" rate package

**Expiration Date:** 

Craft: Operating Engineer

APPRENTICE RATE SCHEDULE

INTERVAL		PERIC	DD AND RAT	ES			
Yearly	60%	70%	80%	90%			
			×				

Ratio of Apprentices to Journeymen - \*

Craft: Operating Engineer

**COMMENTS/NOTES** 

Operating Engineer rates are located in the "Statewide" rate package.

<sup>\* 1</sup> apprentice for each piece of heavy equipment. At least 10 pieces of heavy equipment or a minimum of 5 Operating Engineers must be on site.

County - SOMERSET

Craft: Operating Engineer - Field Engineer

PREVAILING WAGE RATE

Rates are located in the "Statewide" rate package

Expiration Date:

Craft: Operating Engineer - Field Engineer

			- "Bille	er								
	INTERVAL				A	PPRENTICE	E RATE SCIII	Fhu s				
	Yearly	<del> </del>	PERI	OD AND RA	TES			CDOFF				
H		70%	750/	of D	T	T						
L	Yearly				Chainman	Wage		T	T			1
Ra	itio of A			80%	90%	Transit/	<del> </del>	<del> </del>	L 1	~		
* N	atio of Apprentices	io Journeym	len = *			Transit/	Instrument	man	Wage			
Cra	lo more than 1 Figure	eld Engineer	Apprentice	Der Sum	_				- Sc		- 1	
	ift: Operating En	gineer - Field	l Engineer	rer ourvey (	Crew.							

Craft: Operating Engineer - Field Engineer

COMMENTS/NOTES

Operating Engineer - Field Engineer rates are located in the "Statewide" rate package.

5/13/2014 Page 42 of 69

County - SOMERSET

Craft: Painter - Bridges

PREVAILING WAGE RATE

05/01/13
W55.68
B22.62
T78.30
W58.18
B22.62
T80.80
W50.68
B22.62
T73.30

Expiration Date: 04/30/2014

Craft: Painter - Bridges

APPRENTICE RATE SCHEDULE

INTERVAL		PERI	OD AND RA	ATES		<del></del>				
6 Months	50%	55%			60%	65%		75%	85%	
Benefits	Intervals	1 to 2 =	8.13	intervals	3 to 4 =	9.63	Intervals	5 to 6 =	11.13	

Ratio of Apprentices to Journeymen - 1:4

Craft: Painter - Bridges

COMMENTS/NOTES

These rates apply to: All bridges that span waterways, roadways, railways and canyons. All tunnels, overpasses, viaducts and all appurtenances.

### FOREMEN REQUIREMENTS:

- When there are 4 or more Painters on a job, 1 shall be designated a Foreman.
- When there are 15 or more Painters on a job, 1 shall be designated a General Foreman.

The regular workday shall consist of 8 hours between 7:00 AM and 5:30 PM.

### SHIFT DIFFERENTIALS:

- The second shift shall receive an additional 10% of the hourly rate, per hour, and the third shift shall receive an additional 15% of the hourly rate, per hour.

#### OVERTIME:

- Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays and Sundays shall be paid at time and one-half the regular rate. All hours on holidays shall be paid at double the regular rate.
- Saturday or Sunday may be used to make up a day lost to inclement weather, at straight time.
- Four 10-hour days may be worked, at straight time, Monday through Friday.

RECOGNIZED HOLIDAYS: New Year's Day, President's Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day. Saturday holiday observed the preceding Friday. Sunday holiday observed the following Monday.

### County - SOMERSET

Craft: Painter - Line Striping

#### PREVAILING WAGE RATE

	07/09/13
Foreman (Charge Person)	W34.33
	B13.05
	T47.38
Helper (1st Year-2nd 6	W25.99
Mos.)	B12.96
	T38.95
Helper (2nd Year)	W27.55
	B12.99
	T40.54
Helper (3rd Year)	W29.62
	B13.00
	T42.62
Journeyman	W33.83
	B13.05
<u> </u>	T46.88
Probationary Helper (1st-6	W22.86
Mos.)	B12.93
	T35.79

Expiration Date: 06/30/2014

Craft: Painter - Line Striping

COMMENTS/NOTES

#### **OVERTIME:**

Hours in excess of 8 per day, Monday through Saturday, and all hours on Sundays and holidays shall be paid at time and one-half the hourly rate.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Veterans Day, Thanksgiving Day and Christmas Day. Veterans Day may be substituted for the day after Thanksgiving.

County - SOMERSET

### Craft: Painter - New Construction

### PREVAILING WAGE RATE

	05/03/13
Foreman	W40.94
	B20.78
	T61.72
General Foreman	W42.80
	B21.01
	T63.81
Journeyman	W37.22
	B20.34
	T57.56

Expiration Date: 04/30/2014

Craft: Painter - New Construction

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIO	OD AND RAT	TES			<del></del>			
4 Months	30%	40%	50%	60%	70%	75%	80%	85%	90%	
Benefits	Intervals	1 to 3 =	8.00	Intervals	4 to 6 =	10.00	Intervals	7 to 9 =	11.00	

Ratio of Apprentices to Journeymen - 1:4

Craft: Painter - New Construction

#### COMMENTS/NOTES

Spraying, sandblasting, lead abatement, work on tanks or stacks, work performed above 3 stories or 30 feet in height, or using swing scaffolds requires an additional 10% of the wage rate.

### FOREMEN REQUIREMENTS:

- When there are 4 or more Painters on a job, 1 shall be designated a Foreman.
- When there are 15 or more Painters on a job, 1 shall be designated a General Foreman.

The regular workday shall consist of 8 hours between 7:00 AM and 5:30 PM.

#### SHIFT DIFFERENTIALS:

- The second shift shall receive an additional 10% of the hourly rate, per hour, and the third shift shall receive an additional 15% of the hourly rate, per hour.

### **OVERTIME:**

- Hours in excess of 8 per day, or before or after the regular workday, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the regular rate. All hours on Sundays and holidays shall be paid at double the regular rate.
- Saturday or Sunday may be used to make up a day lost to inclement weather, at straight time.
- Four 10-hour days may be worked, at straight time, Monday through Friday.

RECOGNIZED HOLIDAYS: New Year's Day, President's Day, Memorial Day, July 4th, Labor Day, General Election Day, Veterans' Day, Thanksgiving Day, Christmas Day.

5/13/2014

**County - SOMERSET** 

Craft: Painter - Repainting

#### PREVAILING WAGE RATE

	05/03/13
Foreman	W31.41
	B17.48
	T48.89
General Foreman	W34.27
	B25.50
	T59.77
Journeyman	W28.56
	B17.15
	T45.71

Expiration Date: 04/30/2014

Craft: Painter - Repainting

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES								
	SEE	PAINTER	NEW	CONSTR	TION					

Ratio of Apprentices to Journeymen - 1:4

Craft: Painter - Repainting

#### **COMMENTS/NOTES**

NOTE: These rates may only be used on jobs where no major alterations occur, and where not more than 3 other trades are present on the job, but may NOT, under any circumstances, be used for work on bridges, stacks, elevated tank, or generating stations.

Spraying, sandblasting, lead abatement, work on tanks or stacks, work performed above 3 stories or 30 feet in height, or using swing scaffolds requires an additional 10% of the wage rate.

#### FOREMEN REQUIREMENTS:

- When there are 4 or more Painters on a job, 1 shall be designated a Foreman.
- When there are 15 or more Painters on a job, 1 shall be designated a General Foreman.

### OVERTIME:

- Hours in excess of 8 per day and 40 per week shall be paid at time and one-half the regular rate.
- Four 10-hour days may be worked, at straight time, Monday through Sunday.

RECOGNIZED HOLIDAYS: New Year's Day, President's Day, Memorial Day, July 4th, Labor Day, General Election Day, Veterans' Day, Thanksgiving Day, Christmas Day.

#### **County - SOMERSET**

Craft: Painter- Containment

#### PREVAILING WAGE RATE

05/01/13
W31.37
B20.92
T52.29

Expiration Date: 04/30/2014

Craft: Painter- Containment

COMMENTS/NOTES

NOTE: These rates shall require no painting, but used in a supporting capacity only, such as wrapping, boxing, fencing, etc. on new tanks and structural steel only.

The regular workday shall consist of 8 hours between 7:00 AM and 5:30 PM.

#### SHIFT DIFFERENTIALS:

- The second shift shall receive an additional 10% of the hourly rate, per hour, and the third shift shall receive an additional 15% of the hourly rate, per hour.

#### OVERTIME:

- Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays and Sundays shall be paid at time and one-half the regular rate. All hours on holidays shall be paid at double the regular rate.

RECOGNIZED HOLIDAYS: New Year's Day President's Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day. Saturday holiday observed the preceding Friday. Sunday holiday observed the following Monday.

County - SOMERSET

Craft: Painter-Structural Steel and Tanks (New Construction)

PREVAILING WAGE RATE

	05/01/13
Foreman	W44.21
	B20.92
	T65.13
General Foreman	W46.71
	B20.92
	T67.63
Journeyman	W39.21
	B20.92
	T60.13

Expiration Date: 04/30/2014

Craft: Painter- Structural Steel and Tanks (New Construction)

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES								
	SEE	PAINTER	BRIDGES	·						
				_						

Craft: Painter-Structural Steel and Tanks (New Construction)

#### COMMENTS/NOTES

These rates apply to: All work in nuclear plants, on towers, on steeples, on dams, on hangers and open steel whether new or repaint. All new work in refineries, tank farms, water/sewerage treatment facilities and on piplines, and tanks, including all elevated and water tanks, tank interiors and repaint of ground tanks over sixty (60) feet in height.

### FOREMEN REQUIREMENTS:

- When there are 4 or more Painters on a job, 1 shall be designated a Foreman.
- When there are 15 or more Painters on a job, 1 shall be designated a General Foreman.

The regular workday shall consist of 8 hours between 7:00 AM and 5:30 PM.

#### SHIFT DIFFERENTIALS:

- The second shift shall receive an additional 10% of the hourly rate, per hour, and the third shift shall receive an additional 15% of the hourly rate, per hour.

#### OVERTIME:

- Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays and Sundays shall be paid at time and one-half the regular rate. All hours on holidays shall be paid at double the regular rate.
- Saturday or Sunday may be used to make up a day lost to inclement weather, at straight time.
- Four 10-hour days may be worked, at straight time, Monday through Friday.

RECOGNIZED HOLIDAYS: New Year's Day, President's Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day. Saturday holiday observed the preceding Friday. Sunday holiday observed the following Monday.

5/13/2014 Page 48 of 69

#### County - SOMERSET

Craft: Painter-Structural Steel and Tanks (Repaint)

PREVAILING WAGE RATE

	05/01/13
Foreman	W35.31
	B17.65
	T52.96
General Foreman	W37.81
	B17.65
	T55.46
Journeyman	W30.31
	B17.65
	T47.96

Expiration Date: 04/30/2014

Craft: Painter- Structural Steel and Tanks (Repaint)

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES									
	SEE	PAINTER	BRIDGES								
							_	·			

Craft: Painter- Structural Steel and Tanks (Repaint)

#### COMMENTS/NOTES

These rates apply to: All repaint work in refineries, tank farms, water/sewerage treatment facilities and on pipelines and repainting of all other tanks.

### **FOREMEN REQUIREMENTS:**

- When there are 4 or more Painters on a job, 1 shall be designated a Foreman.
- When there are 15 or more Painters on a job, 1 shall be designated a General Foreman.

The regular workday shall consist of 8 hours between 7:00 AM and 5:30 PM.

#### SHIFT DIFFERENTIALS:

- The second shift shall receive an additional 10% of the hourly rate, per hour, and the third shift shall receive an additional 15% of the hourly rate, per hour.

#### OVERTIME:

- Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays and Sundays shall be paid at time and one-half the regular rate. All hours on holidays shall be paid at double the regular rate.
- Saturday or Sunday may be used to make up a day lost to inclement weather, at straight time.
- Four 10-hour days may be worked, at straight time, Monday through Friday.

RECOGNIZED HOLIDAYS: New Year's Day, President's Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day.

**County - SOMERSET** 

Craft: Painter- Tender

PREVAILING WAGE RATE

05/01/13
W18.50
B9.05
T27,55

Expiration Date: 04/30/2014

Craft: Painter-Tender

COMMENTS/NOTES

These rates apply to: The handling of all materials, manning of safety boats, handling of traffic controls, loading and unloading of trucks, cleaning of abrasive materials and other clean-up work.

NOTE: THIS WORK CLASSIFICATION SHALL ONLY APPLY IN SUPPORT OF THE FOLLOWING WORK CLASSIFICATIONS), Painter-Structural Steel and Tanks (New Construction), Painter-Structural Steel and Tanks (Repaint) and Painter-Bridges.

The regular workday shall consist of 8 hours between 7:00 AM and 5:30 PM.

#### SHIFT DIFFERENTIALS:

- The second shift shall receive an additional 10% of the hourly rate, per hour, and the third shift shall receive an additional 15% of the hourly rate, per hour.

#### **OVERTIME:**

- Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays and Sundays shall be paid at time and one-half the regular rate. All hours on holidays shall be paid at double the regular rate.
- Saturday or Sunday may be used to make up a day lost to inclement weather, at straight time.
- Four 10-hour days may be worked, at straight time, Monday through Friday.

RECOGNIZED HOLIDAYS: New Year's Day, President's Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day.

County - SOMERSET

Craft: Paperhanger - New Construction

PREVAILING WAGE RATE

	05/03/13
Foreman	W41.93
	B20.90
	T62.83
Journeyman	W38.12
	B20.44
	T58.56

Expiration Date: 04/30/2014

Craft: Paperhanger - New Construction

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES								
4 Months	30%	40%	50%	60%	70%	75%	80%	85%	90%	
Benefits	Intervals	1 to 3 =	8.00	Intervals	4 to 6 =	10,00	intervals	7 to 9 =	11.00	

Ratio of Apprentices to Journeymen - 1:4

Craft: Paperhanger - New Construction

**COMMENTS/NOTES** 

#### **FOREMEN REQUIREMENTS:**

- When there are 4 or more Paperhangers on a job, 1 shall be designated a Foreman.

The regular workday shall consist of 8 hours between 7:00 AM and 5:30 PM.

#### SHIFT DIFFERENTIALS:

- The second shift shall receive an additional 10% of the hourly rate, per hour, and the third shift shall receive an additional 15% of the hourly rate, per hour.

#### **OVERTIME:**

- Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the regular rate. All hours on Sundays and holidays shall be paid at double the regular rate.
- Saturday or Sunday may be used to make up a day lost to inclement weather, at straight time.
- Four 10-hour days may be worked, at straight time, Monday through Friday.

RECOGNIZED HOLIDAYS: New Year's Day, President's Day, Memorial Day, July 4th, Labor Day, General Election Day, Veterans' Day, Thanksgiving Day, Christmas Day.

**County - SOMERSET** 

Craft: Paperhanger - Renovation

PREVAILING WAGE RATE

	05/03/13
Foreman	W32.20
	B17.58
	T49.78
Journeyman	W29.28
	B17.23
	T46.51

Expiration Date: 04/30/2014

Craft: Paperhanger - Renovation

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES								
	SEE	PAPER-	HANGER	NEW	CONSTR	TION				
										-

Ratio of Apprentices to Journeymen - 1:4

Craft: Paperhanger - Renovation

COMMENTS/NOTES

NOTE: These rates may only be used on jobs where no major alterations occur, and where not more than 3 other trades are present on the job, but may NOT, under any circumstances, be used for work on bridges, stacks, elevated tanks, or generating stations.

#### FOREMEN REQUIREMENTS:

- When there are 4 or more Paperhangers on a job, 1 shall be designated a Foreman.

#### **OVERTIME:**

- Hours in excess of 8 per day and 40 per week shall be paid at time and one-half the regular rate.
- Four 10-hour days may be worked, at straight time, Monday through Sunday.

County - SOMERSET

Craft: Pipelitter

PREVAILING WAGE RATE

	05/03/13
Foreman	W49.36
	B35.63
	T84.99
Journeyman	W45.61
	B32.93
	T78.54

Expiration Date: 04/30/2014

Craft: Pipefitter

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIC	DD AND RAT	ES		-	<del> </del>	 	
Yearly	35%	45%	55%	70%	80%			-	
Benefit	72% of	Appren	tice	Wage	Rate + .10	cents	-		

Ratio of Apprentices to Journeymen - 1:5

Craft: Pipefitter

COMMENTS/NOTES

#### **FOREMAN REQUIREMENTS:**

- When there are 2 or more Journeyman Pipefitters on a job, 1 shall be designated a Foreman.
- There shall be a Foreman for every 8 Journeyman Pipefitters on a job.

The regular workday shall be 8 hours between 7:00 AM and 3:30 PM.

#### SHIFT DIFFERENTIAL:

- 2nd Shift (3:30 PM-11:30 PM) shall work 7.5 hours and receive 8 hours pay at the regular rate, plus 25% per hour.
- 3rd Shift (11:30 PM-7:00 AM) shall work 7 hours and receive 8 hours pay at the regular rate, plus 30% per hour.

#### OVERTIME:

- All hours worked in excess of 8 per day, Monday through Friday, and all hours worked on Saturday, shall be paid at time and one-half, inclusive of benefits. All hours on Sunday and holidays shall be paid at double time, inclusive of benefits.
- By mutual agreement, employees may work four 10-hour days, Monday to Thursday, at straight time rate. Friday may be used as a make-up day for a day lost to inclement weather, and may be paid at straight time. If Friday is not a make-up day, the first 8 hours shall be paid at time and one-half, inclusive of benefits; hours in excess of 8 shall be paid at double time, inclusive of benefits.

### SHIFT DIFFERENTIAL (Maintenance Work Only):

- 2nd Shift (3:30 PM-11:30 PM) shall work 7.5 hours and receive 8 hours pay at the regular rate, plus 10% per hour.
- 3rd Shift (11:30 PM-7:00 AM) shall work 7 hours and receive 8 hours pay at the regular rate, plus 15% per hour.

#### OVERTIME (Maintenance Work Only):

- All hours in excess of 8 per day, Monday through Saturday, shall be paid at time and one-half, inclusive of benefits. All hours on Sundays and holidays shall be paid at double time, inclusive of benefits.

5/13/2014 Page 53 of 69

### **County - SOMERSET**

NOTE: Maintenance work is work to repair, restore, or improve the efficiency of existing facilities. This does NOT apply to ANY new construction.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day. Sunday holidays are observed the following Monday.

5/13/2014 Page 54 of 69

### County - SOMERSET

Craft: Plasterer	PREVAILING WAGE RATE
See "Cement Mason" R	ates
Expiration Date:	
Craft: Plasterer	COMMENTS/NOTES
***See CEMENT MASON Rates***	

5/13/2014 Page 55 of 69

#### County - SOMERSET

Craft: Plumber

PREVAILING WAGE RATE

	12000	
i\	05/01/14	05/01/15
Foreman	W53.31	W0.00
	B30.99	B0.00
	T84.30	T86.18
General Foreman	W56.76	W0.00
	B30.99	B0.00
	T87.75	T89.55
Journeyman	W49.36	W0.00
	B30.99	B0.00
	T80.35	T82.30

Expiration Date: 04/30/2016

Craft: Plumber

APPRENTICE RATE SCHEDULE

INTERVAL		PERI	OD AND RAT	ES	7.5		<u>_</u>	<del></del>	
Yearly	30%	45%	55%	65%	75%				
Benefits	11.89	17.51	19.04	20.06	22.13	_	 _		

### Ratio of Apprentices to Journeymen - \*

\* Employers may employ 1 apprentice on any job where 1 or 2 journeymen are employed. Thereafter, 1 apprentice may be employed for every 4 journeymen.

Craft: Plumber

COMMENTS/NOTES

### FOREMAN REQUIREMENTS:

- On any job having 2 or more Plumbers, 1 must be designated a Foreman.
- On any job having 9 or more Plumbers, 2 shall be designated as Foremen.

The regular workday shall consist of 8 hours between 7:00 AM and 4:30 PM.

### SHIFT DIFFERENTIALS:

- Shift work must continue for a minimum of 5 consecutive workdays.
- When two shifts are worked, the second shift shall work 7.5 hours and receive 8 hours pay, at a rate equal to the hourly rate plus 10%, inclusive of benefits.
- When a third shift is worked, the third shift shall work 7 hours and receive 8 hours pay, at a rate equal to the hourly rate plus 15%, inclusive of benefits.

#### **OVERTIME:**

- All hours in excess of 8 per day, or before of after the regular workday that are not shift work, Monday through Friday, and all hours on Saturday, shall be paid at time and one-half the regular rate, inclusive of benefits. All hours on Sunday and holidays, shall be paid at double the hourly rate, inclusive of benefits.
- Four 10-hour days may be worked, Monday to Thursday, at straight time. Friday may be used as a make-up day for a day lost due to inclement weather. If Friday is not a make-up day, all hours on Friday shall be paid at time and one-half, inclusive of benefits.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election

5/13/2014

**County - SOMERSET** 

Day, Veterans' Day, Thanksgiving Day, Christmas Day. Sunday holidays will be observed the following Monday.

#### County - SOMERSET

Craft: Roofer

PREVAILING WAGE RATE

	08/06/12
Foreman	W34.77
	B21.52
	T56.29
Journeyman	W33.77
	B21.52
	T55.29

Expiration Date: 05/31/2015

Craft: Roofer

APPRENTICE RATE SCHEDULE

INTERVAL		PEI	RIOD AND R	ATES			<u></u>		
6 Months	40%	50%	60%	70%	80%	90%			
Benefits	1.10	1.10	19.00	19.00	19.00	19.00			

### Ratio of Apprentices to Journeymen - \*

- \* [A] For roofing jobs that are of the 1 or single ply nature: 1:2 or fraction thereof
- [B] For roofing jobs on new built up roofs: 1:3 or fraction thereof
- [C] For roofing jobs that are of a tear-off nature: 1:2 or fraction thereof
- [D] For re-roofing jobs (not requiring complete removal of existing systems; installation done over existing roof): 1:3 or fraction thereof

Craft: Roofer

Crait: Rooier

COMMENTS/NOTES

Pitch: +.50 per hour

Mop Man: +.30 per hour

The regular workday consists of 8 hours between 8:00 AM and 4:30 PM.

#### OVERTIME:

Hours in excess of 8 per day, or before or after the regular workday, Monday through Friday, and all hours on Saturdays, and holidays shall be paid at time and one-half the regular rate.

RECOGNIZED HOLIDAYS: New Year's Day, Good Friday, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day.

County - SOMERSET

Craft: Sheet Metal Sign Installation

### PREVAILING WAGE RATE

	04/03/14
Foreman	W32.03
	B26.77
	T58.80
Journeyman	W30.78
	B26.77
	T57.55

Expiration Date: 09/30/2014

Craft: Sheet Metal Sign Installation

### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES									
1000 hours	35%	40%	45%	50%	55%	60%	65%	70%	75%	80%	
Benefits	10.77	12.31	13.85	15.39	16.93	18.47	20.01	21.55	23.09	24.62	

Ratio of Apprentices to Journeymen - 1:3

Craft: Sheet Metal Sign Installation

COMMENTS/NOTES

### FOREMAN REQUIREMENT:

When there are 6 or more Sheet Metal Sign Installers on a job, 1 shall be designated a Foreman.

The regular workday consists of 8 hours, between 7:00 AM and 3:30 PM.

### OVERTIME:

Hours before or after the regular workday, Monday though Friday, and all hours worked on Saturday shall be paid at time and one-half the hourly rate. All hours on Sunday and holidays shall be paid at double the hourly rate.

Four(4) 10 hour days may be worked, Monday through Friday, at straight time, for projects lasting at least one week in duration. The fifth day may be used as a make-up day at straight time for a day lost due to inclement weather. However, if the fifth day is not a make-up day, all hours worked will be paid at time and one-half the hourly rate.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Good Friday, Memorial Day, July 4th, Labor Day, Veterans' Day, Thanksgiving Day and the day after, Christmas Day. Saturday holidays observed the following Monday.

County - SOMERSET

Craft: Sheet Metal Worker

#### PREVAILING WAGE RATE

	06/05/13	06/01/14
Foreman	W46.87	W0.00
	B33.85	B0.00
	T80.72	T82.47
General Foreman	W47.87	W0.00
	B33.85	B0.00
	T81.72	T83.47
Journeyman	W44.37	W0.00
	B33.85	B0.00
	T78.22	T79.97

Expiration Date: 05/31/2015

Craft: Sheet Metal Worker

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIC	DD AND RAT	ES				·		
Yearly	35%	45%	55%	65%	of	Journey	man	Wage	Rate	
Benefit	35%	45%	55%	65%	of	Journey	man	Benefit	Rate	

Ratio of Apprentices to Journeymen - 1:4

Craft: Sheet Metal Worker

**COMMENTS/NOTES** 

### FOREMAN REQUIREMENTS:

- When there are 2 or more Sheet Metal Workers on a project, 1 must be designated a Foreman.
- When there are 17 or more Sheet Metal Workers on a project, 1 must be designated a General Foreman.
- When there is only 1 Sheet Metal Worker (1 Journeyman) on a project, he/she shall receive \$1.00 more than the regular Journeyman's rate.

The regular workday is 8 hours between 7:00 AM and 4:30 PM.

#### SHIFT DIFFERENTIAL:

- 2nd Shift (3:30 PM 12:00 AM) : +17% of regular hourly rate
- Shift work must run for a minimum of 5 consecutive workdays.

### OVERTIME:

- Hours in excess of 8 per day, or before or after the regular workday, that are not shift work, and the first 10 hours on Saturdays shall be paid at time and one-half of the regular rate, inclusive of benefits. Hours in excess of 10 per day on Saturday, and all hours on Sundays and holidays shall be at double the regular rate, inclusive of benefits.
- Four 10-hour days may be worked, Monday through Friday, at straight time, with hours in excess of 10 per day, and hours in excess of 40 per week paid at the overtime rates listed above.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Good Friday, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day. Sunday holidays will be observed the following Monday.

#### County - SOMERSET

Craft: Sprinkler Fitter

### PREVAILING WAGE RATE

	07/03/13	07/01/14	07/01/15
Foreman	W59.63	W0.00	W0.00
	B23.12	B0.00	80.00
<u> </u>	T82.75	T84.65	T86.65
General Foreman	W62.31	W0.00	W0.00
	B23.12	B0.00	B0.00
ja	T85.43	T87.55	T89.79
Journeyman	W55.63	W0.00	W0.00
	B23.12	B0.00	B0.00
	T78.75	T80.65	T82.65

Expiration Date: 06/30/2016

Craft: Sprinkler Fitter

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES										
1000 hours	9.50	11.25	27.82	30.60	33,38	36.16	38.94	41.72	44.50	47,29		
Benefits	10.12	10.12	19.37	19.37	19.37	19.37	Intervals	7 to 10	Jourymn	Ben.		

#### Ratio of Apprentices to Journeymen - 1:3

### APPRENTICE RATE SCHEDULE FOR THOSE APPRENTICES REGISTERED AS OF 7-1-13:

INTERVAL PERIOD AND RATES

1000 hours 13.90 16.69 22.25 25.03 30.60 33.38 38.94 41.72 47.29 50.07

Benefits 10.12 10.12 19.37 19.37 19.37 Intervals 7 to 10 receive Journeyman Ben.

Craft: Sprinkler Fitter COMMENTS/NOTES

The regular workday consists of 8 consecutive hours between 6:00 AM and 4:30 PM.

### FOREMAN REQUIREMENTS:

- The first Sprinkler Fitter on the job must be designated a Foreman.
- On any job having 12 or more Sprinkler Fitters, one must be designated a General Foreman.

### SHIFT DIFFERENTIALS:

- Shift work must run for a minimum of 2 consecutive workdays.
- 2nd and 3rd shift shall receive an additional 15% of the regular rate, per hour.
- Any "off hours" shift starting at 8:00 PM or later shall receive an additional 25% of the regular rate, per hour.

#### **OVERTIME**

The first 2 hours in excess of 8 per day, after the regular workday that are not shift work, Monday through Friday, shall be paid at time and one-half the regular rate. Hours worked in excess of 10 per day, Monday through Friday, and all hours on Saturday, Sunday and holidays, shall be paid double the regular rate.

Four 10 hour days may be worked, Monday through Thursday, at striaght-time.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Good Friday, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day.

5/13/2014

### **County - SOMERSET**

Craft: Tile Finisher-Marble

#### PREVAILING WAGE RATE

	01/01/14	07/01/14	01/01/15	07/01/15	01/01/16	07/01/16
Finisher	W44.79	W45.68	W46.58	W47.47	W48.37	W49.44
	B28.17	B28.17	B28.17	B28.17	B28.17	B28.17
	T72.96	T73.85	T74.75	T75.64	T76.54	T77.61

Expiration Date: 12/31/2016

Craft: Tile Finisher-Marble

#### APPRENTICE RATE SCHEDULE

INTERVAL,		PERIC	DD AND RAT	ES		<del></del>			<u>,</u>	
750 Hours	50%	55%	65%	70%	75%	85%	90%	95%		
					,					<del>-</del> .:

Ratio of Apprentices to Journeymen - 1:4

Craft: Tile Finisher-Marble

COMMENTS/NOTES

#### OVERTIME:

Hours in excess of 7 per day, Monday through Friday, and the first 7 hours on Saturdays shall be paid at time and one half the regular rate, inclusive of benefits. Hours in excess of 7 on Saturdays and all hours on Sundays and holidays shall be paid at double the regular rate, inclusive of benefits.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Good Friday, Memorial Day, July 4th, Labor Day, Columbus Day, Veterans' Day, Thanksgiving Day and the day after, Christmas Day, Sunday holidays observed the following Monday.

#### **County - SOMERSET**

Craft: Tile Setter - Ceramic

#### PREVAILING WAGE RATE

	12/07/13	06/07/14	12/06/14	06/06/15	12/05/15	06/04/16	12/03/16
Finisher	W40.46	W41.28	W42.10	W42.92	W43.74	W44.56	W45.38
	B27.07	B27.07	B27.07	B27.07	B27.07	B27.07	B27.07
_	T67.53	T68.35	T69.17	T69.99	<b>T</b> 70.81	T71.63	T72.45
Setter	W51.78	W52.91	W54.04	W55.17	W56.30	W57.43	W58.56
	B29.96	B29.96	B29.96	B29.96	B29.96	B29.96	B29.96
	T81.74	T82.87	T84.00	T85.13	T86.26	T87.39	T88.52

Expiration Date: 12/02/2016

Craft: Tile Setter - Ceramic

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES								
750 Hours	50%	55%	60%	65%	70%	75%	85%	95%	100%	
										· -

Ratio of Apprentices to Journeymen - 1:4

Craft: Tile Setter - Ceramic

COMMENTS/NOTES

#### **OVERTIME:**

Hours in excess of 7 per day, and the first 10 hours on Saturdays shall be paid at time and one-half the hourly rate. All hours on Sundays after 10 hours shall be paid double the hourly rate. All hours on Sundays and holidays shall be paid at double the hourly rate.

RECOGNIZED HOLIDAYS; New Year's Day, Memorial Day, July 4th, Labor Day, Thanksgiving Day, Christmas Day.

5/13/2014

#### **County - SOMERSET**

Craft: Tile Setter - Marble

#### PREVAILING WAGE RATE

	01/01/14	07/01/14	01/01/15	07/01/15	01/01/16	07/01/16
Tile Setter	W56.15	W57.19	W58.22	W59.46	W60.71	W62.16
	B28.39	B28.39	B28.39	B28.39	B28.39	B28.39
	T84.54	T85.58	T86.61	T87.85	T89.10	T90.55

Expiration Date: 12/31/2016

Craft: Tile Setter - Marble

#### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES								
750 Hours	50%	55%	65%	70%	75%	85%	90%	95%		

Ratio of Apprentices to Journeymen - 1:4

Craft: Tile Setter - Marble

COMMENTS/NOTES

#### OVERTIME:

Hours in excess of 7 per day, Monday through Friday, and the first 7 hours on Saturdays shall be paid at time and one-half the regular rate, inclusive of benefits. Hours in excess of 7 on Saturdays, and all hours on Sundays and holidays shall be paid at double the regular rate, inclusive of benefits.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Good Friday, Memorial Day, July 4th, Labor Day, Columbus Day, Veterans' Day, Thanksgiving Day and the day after, Christmas Day. Sunday holidays observed the following Monday.

**County - SOMERSET** 

Craft: Tile Setter - Mosaic & Terrazzo

### PREVAILING WAGE RATE

	01/01/14	07/01/14	01/01/15	07/01/15	01/01/16	07/01/16
Grinder or Assistant	W46.69	W47.65	W48.61	W49.76	W50.91	W52.06
	B31.79	B31.79	B31.79	B31.79	B31.79	B31.79
	T78.48	T79.44	T80.40	T81.55	T82.70	T83.85
Mechanic	W48.30	W49.26	W50.22	W51.37	W52.52	W53.67
	B31.80	B31.80	B31.80	B31.80	B31.80	B31.80
	T80.10	T81.06	T82.02	T83.17	T84.32	T85.47

Expiration Date: 12/31/2016

Craft: Tile Setter - Mosaic & Terrazzo

### APPRENTICE RATE SCHEDULE

INTERVAL		PERIOD AND RATES								
750 Hours	50%	55%	65%	70%	75%	85%	90%	95%		

Ratio of Apprentices to Journeymen - 1:5

Cruft: Tile Setter - Mosaic & Terrazzo

**COMMENTS/NOTES** 

The regular workday consists of 7 hours, between 8:00 AM and 3:30 PM.

### OVERTIME:

 Hours in excess of 7 per day, or before or after the regular workday, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the hourly rate. All hours on Sundays and holidays shall be paid at double the hourly rate.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Good Friday, Monday after Easter, Memorial Day, July 4th, Labor Day, Columbus Day, Veterans' Day, Thanksgiving Day and the day after, Christmas Day. Sunday holidays observed the following Monday.

### **County - SOMERSET**

Craft: Truck Driver

### PREVAILING WAGE RATE

	10/11/13
Bucket, Utility, Pick-up,	W35.80
Fuel Delivery trucks	B26,48
<u> </u>	T62.28
Dump truck, Asphalt	W35.80
Distributor, Tack Spreader	B26.48
	T62.28
Euclid-type vehicles (large	W35.95
off-road equipment)	B26.48
	T62.43
Helper on Asphalt	W35.80
Distributor truck	B26.48
<u> </u>	T62.28
Slurry Seal,	W35.80
Seeding/Fertilizing/Mulchi	B26.48
ng truck	T62.28
Straight 3-axle truck	W35.85
	B26.48
	T62.33
Tractor-Trailer truck (all	W35.95
types)	B26.48
	T62.43
Vacuum or Vac-All truck	W35.80
(entire unit)	B26.48
	T62.28
Winch Trailer Driver	W36.05
	B26.48
	T62.53

Expiration Date: 04/30/2014

Craft: Truck Driver

**COMMENTS/NOTES** 

Foreman: + \$.75 cents per hour. Overtime rate shall be increased accordingly.

### HAZARDOUS WASTE REMOVAL WORK:

- On a hazardous waste site requiring Level A, B, or C personal protection for any worker: + \$3.00 per hour.
- On a hazardous waste site not designated Level A, B, or C: + \$1.00 per hour.

The regular workday consists of 8 hours starting at either 6:00 AM or 8:00 AM.

### SHIFT DIFFERENTIAL:

Any shift starting at a time other than 6:00 AM or 8:00 AM shall receive an additional \$2.50 per hour.

County - SOMERSET

#### **BLENDED RATE:**

- When a truck driver is performing work on site and also serving as a material delivery driver, the driver shall be paid a "blended rate" which shall be 80% of the above-listed wage rates, plus the full benefit rate. This rate shall be used when the driver "round robins" for a minimum of 6 hours during the work day.

#### OVERTIME

- Hours in excess of 8 per day, or before or after the regular workday that are not shift work, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the hourly rate. All hours on Sundays and holidays shall be paid at double the hourly rate.
- Benefits on overtime shall be \$30.38. As of 11-1-13, benefits on overtime shall be \$31.28.
- Four 10-hour days may be worked, Monday through Thursday, at straight time, with Friday used as a make-up day for a day lost to inclement weather. If Friday is not a make-up day, all hours on Friday shall be paid at time and one-half the hourly rate.

RECOGNIZED HOLIDAYS: New Year's Day, Presidents' Day, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day. Sunday holidays will be observed the following Monday.

County - SOMERSET

Craft: Truck Driver-Material Delivery Driver

PREVAILING WAGE RATE

	09/04/12
Driver	W19.43
	B15.98
	T35.41
New Hires: 1st Year	W18.43
	B15.98
	T34.41

Expiration Date: 04/30/2012

Craft: Truck Driver-Material Delivery Driver

**COMMENTS/NOTES** 

NOTE: These rates may only be used for the delivery of materials to the jobsite.

OVERTIME: Hours in excess of 8 per day, Monday through Friday, and all hours on Saturdays shall be paid at time and one-half the hourly rate. All hours on Sundays and holidays shall be paid at double the hourly rate.

RECOGNIZED HOLIDAYS: New Year's Day, Washington's Birthday, Memorial Day, July 4th, Labor Day, Presidential Election Day, Veterans' Day, Thanksgiving Day, Christmas Day, Sunday holidays will be observed the following Monday.

County - SOMERSET

Craft:	Welder
Cran:	Welder

PREVAILING WAGE RATE

Welder

**Expiration Date:** 

Craft: Welder

**COMMENTS/NOTES** 

Welders rate is the same as the craft to which the welding is incidental.

ATTACHMENT B DBE REQUIREMENTS FOR RACE CONSCIOUS FEDERAL PROCUREMENT ACTIVITIES (29 pages)

# NEW JERSEY TRANSIT CORPORATION DBE REQUIREMENTS FOR

### **RACE-CONSCIOUS**

### FEDERAL PROCUREMENT ACTIVITIES

### **TABLE OF CONTENTS**

TITLE	OF ARTICLE P	AGE
1.1	POLICY	2
1.2	ASSURANCE	2
1.3	DBE GOAL FOR THIS PROJECT	3
1.4	GUIDANCE TO BIDDER/PROPOSERS/PRIMES	3
1.5	TRANSIT VEHICLE MANUFACTURERS (TVM)	3-4
1.6	RESPONSIBLE BID/PROPOSAL CRITERIA	. 4
2.0	GUIDANCE ON A GOOD FAITH	4
2.2	A GOOD FAITH EFFORT	5
2.3	REQUIRED FORMS	5-6
2.4	SUPPLEMENTAL REQUIRED FORMS (IF APPLICABLE)	
	7	
2.5	INSTRUCTIONS FOR COMPLETING REQUIRED FORMS	7-9
2.6	INSTRUCTIONS FOR COMPLETING SUPPLEMENTAL REQUIRED FORMS	9-10
3.0	GUIDANCE ON COUNTING DBE PARTICIPATION	11
3.2	COUNTING DBE PARTICIPATION	11-12
3.3	DBE PRIME CONTRACTOR GUIDANCE	12
3.4	DBE TRUCKING FIRM GUIDANCE	12
3.5	LEASING TRUCKS	12-13
3.6	COUNTING MATERIALS AND SUPPLIES	13
4.0	TERMINATION OF DBE(s)	13
4.2	ADDITION OF DBE(s)	14
4.3	REPLACEMENT OR REMOVAL OF DBE(s)	14-15
4.4	WITHDRAWN DBE(s)	15
4.5	DECERTIFIED DBE(s)	15-16
5.0	AWARD OBLIGATIONS	16
5.2	POST AWARD DELIVERABLES	16-17
5.3	PROMPT PAYMENT OF INVOICES TO SUBCONTRACTORS	17-18
5.4	SUBCONTRACT PAYMENT DISPUTE RESOLUTION	. 18
5.5	PROMPT PAYMENT OF SUBCONTRACTOR RETAINAGE	18-19
5.6	REMEDIES AND PENALITES	. 19
5.7	LIQUIDATED DAMAGES	19-20
APPEND	DIX 1 (GLOSSARY)	21-22

# NEW JERSEY TRANSIT CORPORATION DBE REQUIREMENTS FOR RACE-CONSCIOUS FEDERAL PROCUREMENT ACTIVITIES

The following pages provide Bidders/Proposers/Primes on federal contracts with New Jersey Transit (NJT), information about NJT's Disadvantaged Business Enterprise (DBE) Program, administered by the Office of Business Development (OBD). Prospective Bidders/Proposers/Primes will have an opportunity to ask questions regarding the directives contained in the DBE specifications at the pre-bid/pre-proposal conference(s). Further clarification of the DBE specifications, along with assistance in completing the forms, can be obtained by calling (973) 491-7593.

A list of certified DBE firms may be found in the NJ Unified Certification Program (NJUCP) Directory at <a href="https://www.njucp.net">www.njucp.net</a>.

Note: Use of this list does not relieve the Bidder/Proposer/Prime contractor/consultant of responsibility to seek

DBE participation from other sources. The list is updated daily and must be checked periodically, as firms are certified and decertified daily.

These DBE specifications are a part of the Contract and shall be binding upon the successful Bidder/Proposer and Prime in the pre and post-award stages of NJT professional services, construction, and goods and services contracts. These specifications shall be binding upon sub-recipients and imposed on their contractors.

### 1.1 POLICY

As defined in the U.S. Department of Transportation (USDOT) Regulation 49 CFR Part 26, it is the policy of NJT that Disadvantaged Business Enterprises shall have the opportunity to compete for and participate in the performance of contracts financed in whole or in part with federal funds. Each subcontract a Prime signs with a subcontractor/subconsultant must include the following assurance referenced in **article 1.2.** 

### 1.2 ASSURANCE

- 1.2.1 The Prime contractor/consultant, or subcontractor/subconsultant shall not discriminate on the basis of race, color, national origin, or sex in the performance of this contract. The Prime contractor/consultant shall carry out applicable requirements of 49 CFR Part 26 in the award and administration of DOT-assisted contracts. Failure by the contractor/consultant to carry out these requirements is a material breach of this contract, which may result in the termination of this contract or such other remedy, as the recipient deems appropriate (49 CFR 26.13).
- 1.2.2 This language is included to comply with relevant Federal law and to ensure that all persons who enter into any direct or indirect form of contractual agreement with NJT are aware of their responsibilities and

the commitment of NJT to see that NJT's DBE Policy is carried out in all instances.

### 1.3 DBE GOAL FOR THIS PROJECT

As an aid in meeting the commitment of its DBE Program, NJT is setting a *race conscious* goal of awarding percent of the gross sum bid/proposal to certified NJUCP DBE firms. Should the actual contract amount increase or decrease, through approved change order(s), the assigned goal may remain. The OBD will determine if the change orders and/or contract phases will result in an adjustment to the DBE participation goal.

### 1.4 GUIDANCE TO BIDDER/PROPOSERS/PRIMES

- 1.4.1 Failure by a Bidder/Proposer/Prime to comply with any of the requirements contained herein shall result in breach of contract and it shall be subject to the appropriate penalties, remedies, or liquidated damage(s). Refer to articles 5.6-5.7
- 1.4.2 All required forms, including the supplemental section (see articles 2.3-2.4) must be submitted in accordance with the requirements. Firms must be certified under the NJUCP at the time of contract award in order to obtain DBE credit toward the goal.
- 1.4.3 Price alone is not an acceptable basis for rejecting a DBE subcontractor/subconsultant's bid.
- 1.4.4 The Bidder/Proposer/Prime shall, at a minimum, seek DBEs in the same geographic area in which it generally seeks subcontractors/subconsultants. However, the Bidder/Proposer/Prime may be required to expand its search under specific circumstances as determined by OBD. Refer to article 2.0
- 1.4.5 Agreements between a Bidder/Proposer/Prime and a DBE in which the DBE promises not to provide subcontracting quotations to other Bidder/Proposers are prohibited.
- 1.4.6 The desire of a Bidder/Proposer/Prime to self-perform the work of a contract with its own organization is not an acceptable basis to not meet the goal or demonstrate a good faith effort to do so.
- 1.4.7 The Bidder/Proposer/Prime is responsible for verifying that the DBE is certified under the appropriate NAICS code for the scope of work identified. DBE credit shall be given only for work performed in the NAICS code(s) under which the DBE is certified.
- 1.4.8 A DBE firm listed on the First-Tier DBE Utilization Form (Form A) shall constitute a binding representation to NJT, by the Bidder/Proposer/Prime, that the DBE firm is qualified, available, and certified under the appropriate and required NAICS code to perform the scope of work identified. Refer to article 2.5a

### 1.5 TRANSIT VEHICLE MANUFACTURERS (TVM)

- 1.5.1 As a transit vehicle manufacturer, you must establish and submit for FTA's approval an annual overall DBE percentage goal. A TVM must certify that it submitted the annual DBE goal required by 49 CFR 26.49 and FTA has approved it or not disapproved it.
- (a) As a condition of being authorized to bid or propose on FTA assisted transit vehicle procurements, the Bidder/Proposer must complete and submit the TVM Certification form with the bid/proposal certifying that it has complied with the requirements of 49 CFR 26.49.

NJT may, with FTA approval, establish project-specific goals for DBE participation in the procurement of transit 1.5.2 vehicles in lieu of complying with the procedures of this section.

### 1.6 RESPONSIBLE BID/PROPOSAL CRITERIA

- 1.6.1 As a matter of responsibility, the two lowest Bidders or two highest ranked Proposers must submit the required forms, including the supplemental section (if applicable), with the bid/proposal or within seven (7) days after the bid opening or proposal due date. NJT may grant a formal written request to extend this 7day requirement at its sole discretion on a case-by-case basis.
- 1.6.2 Failure to satisfactorily complete or submit all required forms when due may result in determination by NJT that the Bidder/Proposer is non-responsible and may cause rejection of the bid or proposal.
- If the two lowest Bidders/highest ranked Proposers submit the DBE forms, but fail to meet the DBE goal, the 1.6.3 OBD will consider the efforts made to determine if a Bidder/Proposer/Prime has in fact, demonstrated a good faith effort. See article 2.0
- 1.6.4 If it is determined that efforts were made to include DBE participation on the contract, however these efforts did not result in meeting the goal, NJT may request that additional efforts be made within 10 business days of the request. If at this time the Bidder/Proposer fails to demonstrate a good faith effort to achieve the goal, NJT shall consider awarding the contract to the next lowest bidder or highest ranked proposer who offers a reasonable price and meets the DBE goal or demonstrates a good faith effort and other bid requirements or requirements of 49 CFR Part 26.

### 2. GUIDANCE ON A GOOD FAITH EFFORT

- To demonstrate a good faith effort to meet the DBE goal, a Bidder/Proposer/Prime shall provide written 2.1 documentation in addition to Form D (article 2.3e), of the steps it has taken, prior to the bid opening/proposal due date, or during the life of the contract to obtain DBE participation. The Bidder/Proposer/Prime can meet this requirement in either of two ways:
  - (1) The Bidder/ Proposer/Prime can meet the goal.
  - (2) The Bidder/Proposer/Prime shall exhaust the available options referenced in article 2.2 in making a continuous good faith effort to meet the assigned contract goal for the life of the contract.
- The efforts employed by the bidder should be those that one could reasonably expect a bidder to take if the (a) bidder were actively and aggressively trying to obtain DBE participation sufficient to meet the contract goal.
- In determining a good faith effort, the OBD will consider the quality, quantity, and intensity of the different (b) kinds of efforts that the Bidder/Proposer/Prime has made. Mere pro forma efforts will not be considered as demonstration of good faith effort to meet the DBE contract requirements.
- The Bidder/Proposer/Prime shall use good business judgment and consider a number of factors in (c) negotiating with subcontractors/subconsultants, including DBE subcontractors/ subconsultants, and should take a firm's price and capabilities as well as contract goals into consideration. The fact that there may be some additional costs involved in finding and using DBEs is not in itself sufficient reason for a Bidder's/Proposer's failure to meet the contract DBE goal, as long as such costs are reasonable as determined by NJT. Primes are not, however, required to accept higher quotes from DBEs if the

price difference is excessive or unreasonable.

- (d) The Bidder/Proposer/Prime's ability or desire to perform the work of a contract with its own organization (self-performance) does not relieve the Bidder/Proposer/Prime of the responsibility to meet the goal or demonstrate a good faith effort.
- (e) The Bidder/Proposer/Prime shall not reject DBEs as being unqualified without sound reasons based on a thorough investigation of their capabilities. The Bidder/Proposer/Prime's standing within its industry, membership in specific groups, organizations, or associations and political or social affiliations (for example, union vs. non-union employee status) are not legitimate causes for the rejection or non-solicitation of bids in the Bidder/Proposer/Prime's efforts to meet the assigned project goal.
- (f) The OBD will support the Bidder/Proposer/Prime in indentifying ways to meet the assigned contract goal.

### 2.2 A GOOD FAITH EFFORT

The following is a list of actions that NJT will consider as evidence of a Bidder/Proposer/Prime's good faith effort to obtain DBE participation. While exhausting the available options in this list may count as a good faith effort, this list is not intended to be a mandatory checklist, nor is this list intended to be exclusive or exhaustive of all the efforts a Bidder/Proposer/Prime might make to achieve the assigned DBE goal. NJT may require a Bidder/Proposer/Prime to take action above and beyond those listed below to meet the assigned DBE goal.

- (a) The Bidder/Proposer/Prime shall solicit through all reasonable and available means (e.g. attendance at pre-bid meetings, advertising and/or written notices) the interest of all certified DBEs who have the capacity to perform the work of the contract.
  - (1) The Bidder/Proposer/Prime must solicit this interest within sufficient time to allow the DBE to respond to the solicitation.
  - (2) The Bidder/Proposer/Prime must take appropriate steps to follow up on initial solicitations in order to determine with certainty if the DBE firms are interested.
- (b) The Bidder/Proposer/Prime shall select portions of the work to be performed by DBEs in order to increase the likelihood that the DBE goals will be achieved. This includes, where appropriate, breaking out contract work items into economically feasible units to facilitate DBE participation, even when the Prime might otherwise prefer to perform these work items with its own forces.
- (c) The Bidder/Proposer/Prime shall provide interested DBEs with adequate information about the plans, specifications, and requirements of the contract in a timely manner to assist them in responding to a solicitation.
- (d) The Bidder/Proposer/Prime shall negotiate with a DBE(s) with the intent to enter into a contract. It is the Bidder/Proposer's responsibility to make a portion of the work available to DBE subcontractors/subconsultants and suppliers and to select those portions of the work or material needs consistent with the available DBE Primes and suppliers, so as to facilitate DBE participation.
  - (1) Evidence of such negotiation includes: the names, addresses, and telephone numbers of DBEs that were considered; a description of the information provided regarding the plans and specifications for the work selected for subcontracting; and evidence as to why additional agreements could not be reached for DBEs to perform the work.
- (e) The Bidder/Proposer/Prime shall make efforts to assist interested DBEs in obtaining bonding, lines of credit, or DBE Requirements for Federal Procurement Activities

- insurance as required by NJT or the Prime contractor.
- (f) The Bidder/Proposer/Prime shall make efforts to assist interested DBEs in obtaining necessary equipment, supplies, materials, or related assistance or services.
- (g) The Bidder/Proposer/Prime shall effectively use the services of available minority/women community organizations; minority/women Prime contractors groups; local, State and Federal minority/women business assistance offices; and other organizations as allowed on a case-by-case basis to provide assistance in the recruitment and placement of DBEs.

### 2.3 REQUIRED FORMS

- (a) Form A First Tier DBE Utilization: Lists all First Tier DBE firms scheduled to participate on this contract.
- (b) Form A1 Bidder/Proposer Solicitation and Contractor Information: Lists all DBE and Non-DBE sub contractor/subconsultants, including suppliers, solicited for, and participating on this contract.
- (c) <u>Form A2- Non-DBE Sub Utilization</u>: Lists all DBE and Non-DBE subcontractors/ subconsultants, including suppliers, participating on this contract.
- (d) Form B Intent to Perform as a DBE Sub: Identifies the work the 1<sup>st</sup> Tier DBE intends to perform including scope of work, subcontract dollar value, etc.
- (e) <u>DBE Good Faith Effort Form (if applicable):</u> Identifies any DBE subcontractor invited to quote, but declined to do so for any reason.
- (f) <u>Trucking Commitment Agreement (if applicable):</u> Identifies all trucking firms (DBE and Non-DBE) participating on this contract, at any tier.
- (g) NJ UCP DBE Certification & NAICS Code Verification: Confirms the DBE status and NAICS code(s) of each First Tier DBE subcontractor/subconsultant.
- (h) \*Form E Contractor's Monthly DBE Payment Report & Payment Certification Voucher (Post-Award):

  Records monthly payments issued to each DBE subcontractor/subconsultant/supplier and monthly payments issued by NJ TRANSIT to the Prime. Certifies that DBE subs have been paid for previous month's invoices.
- (i) Form E2 DBE's Monthly Payment Report (Post-Award): Records monthly invoices submitted by the DBE, payments owed to the DBE on past due invoices and payments received from the prime by each DBE subcontractor/subconsultant.
- (j) <u>Form E1- DBE Prime's Monthly Payment Report (For DBE Prime Only):</u> Records monthly payments <u>issued</u> to each DBE Prime by NJ TRANSIT to.

\*This form is due from the Prime in each month following the notice to proceed issued by NJ TRANSIT. Refer to article 5.2.4

### 2.4 SUPPLEMENTAL REQUIRED FORMS (IF APPLICABLE)

(a) Form AA – Second Tier DBE Utilization: Lists all Second Tier DBE firms scheduled to participate on the DBE sub-Prime's contract.

- (b) <u>Form AA1 Second Tier Bidder/Proposer Solicitation and Contractor Information</u>: Lists all Second Tier DBE firms participating on this contract as indicated on Form AA and Form AA2.
- (c) <u>Form AA2- Second Tier Non-DBE Sub Utilization:</u> Lists all DBE and Non-DBE firms including suppliers participating on the DBE sub-Prime's contract.
- (d) Form BB Intent to Perform as a Second Tier DBE Sub: Identifies the work the 2<sup>nd</sup> Tier DBE intends to perform including scope of work, subcontract dollar value, etc.
- (e) NJ UCP DBE Certification & NAICS Code Verification: Confirms the DBE status and NAICS code(s) of each Second Tier DBE subcontractor/subconsultant.

### 2.5 INSTRUCTIONS FOR COMPLETING REQUIRED FORMS (see glossary for definition of terms)

### (a) Form A - First Tier DBE Utilization:

Form A is a formal agreement between the Bidder/Proposer and the DBE(s). Replacement/removal of DBE subcontractors/subconsultants/supplier identified on Form A is prohibited after the bid or proposal is submitted to NJT. **Refer to article 4.3**. A DBE Bidder/Proposer, which lists itself on Form A, is committed to performing the work indicated with its own personnel.

DBEs performing as second tier sub(s) to a **non-DBE sub Prime** should be listed with the name of the non-DBE sub Prime's firm name in parenthesis next to the DBE sub's name. {Ex: DBE Electric Co. (Prime Contractor, Inc.)}

- (1) A first Tier DBE is required to perform at least 51% of its subcontract value with its own forces. Bidders/Proposers/Primes will not receive any credit for DBEs performing less than 51% and therefore must not be listed on this form.
- (2) For DBE suppliers, identify all manufacturers, regular dealers, and brokers. If a DBE supplier is a *manufacturer*, indicate the full value of its subcontract. If a DBE supplier is a *regular dealer*, show its total contract value multiplied by 60% (Ex. \$100K x 60%= \$60K). If a DBE supplier is *neither a manufacturer nor a dealer*, indicate the <u>fee/commission only</u>, *not the cost of materials or supplies*. See article 3.0 for direction on determining credit toward the goal.
- (3) A detailed scope of work must be provided; <u>one-word descriptions are not acceptable</u>. (Ex. Haul and dispose of approximately 192,000 tons of contaminated soil to a clean earth facility at \$34.00 a ton).

### (b) Form A1 - Bidder/Proposer/Prime Solicitation and Contractor Information:

The Bidder/Proposer must complete and submit **page one** (1). The DBE and non-DBE subcontractors/subconsultants, including suppliers, solicited for, participating on, or expressed interest in this contract must complete **page two** (2).

### (c) Form A2- Non-DBE Sub Utilization:

Bidders/Proposers are required to report and submit all dollars committed to non-DBE subcontractors/subconsultants/suppliers. The non-DBE portion of work is <u>not</u> counted toward the assigned DBE goal. **See article 3.0** 

A detailed scope of work must be provided; one-word descriptions are not acceptable.

(Ex. Hauf and dispose of approximately 192,000 tons of contaminated soil to a clean earth facility at aton). \$34.00

### (d) Form B - Intent to Perform as a First Tier DBE Sub:

Each DBE subcontractor/subconsultant/supplier listed on Form A must complete and sign Form B. Note: <u>The Bidder/Proposer/Prime is prohibited from completing any portion of the form and from directing DBE(s) to sign a blank form.</u>

- (1) The Bidder/Proposer/Prime must provide interested DBEs with a copy of appropriate plans, specifications, and requirements of the contract in a timely manner to allow the DBE to prepare an appropriate price quote and submit on time.
- (2) First Tier DBEs must perform at least 51% of the total dollar value of its subcontract, with its own forces. The firm must indicate the percentage of the total portion of work to be subcontracted to DBE and non-DBE firms. The <u>non-DBE</u> percentage of work is <u>not counted</u> toward the assigned goal.
- (3) The OBD encourages DBE-to-DBE subcontracting in order to preserve DBE participation credit. See article 3.0
- (4) The DBE must provide a detailed scope of work; <u>one-word descriptions are not acceptable</u>. Descriptions should include: *type of services provided, total number of units, price per unit, total cost, etc.*

### (e) <u>DBE Good Faith Effort</u>: (If Applicable)

Form D applies to any Bidder/Proposer/Prime who failed to meet the assigned DBE goal. This form will assist the Bidder/Proposer/Prime in demonstrating a good faith effort.

If the DBE(s) solicited declines to sign this form, the completed form should be submitted with the Bidder/Proposer's signature only and the OBD will verify the information provided with the firm. Refer to articles 2.0-2.2 for guidance.

### (f) Trucking Commitment Agreement: (If Applicable)

DBEs must provide information for all DBE and non-DBE trucking firms it will lease from or subcontract to. Subcontracting to a non-DBE trucker means that the non-DBE will perform a portion of the DBE firm's subcontract. **Refer to article 3.4** 

The following documents must be attached for all trucks owned: copy of title(s)/finance agreement(s), registration card(s), insurance card(s), apportioned cab card(s) and/or hazardous material license(s) if applicable. A copy of the title or finance agreement is the only acceptable proof of ownership.

The following documents must be attached for all trucks leased: copy of lease agreement(s) established between both firms, title(s), registration card(s), insurance card(s), lease agreement(s), apportioned cab card(s) and/or hazardous material license(s) if applicable.

### (g) NJ UCP DBE Certification and NAICS Code Verification:

All DBEs listed on Form A must be certified at the time of contract award. It is the Bidder/ Proposer's responsibility to ensure that DBEs are certified and that their NAICS code(s) match the scope of work to be performed on this contract. Credit will not be given for any work to be performed without the appropriate NAICS code. Status can be verified through <a href="https://www.njucp.net">www.njucp.net</a> and <a href="https://wwww.njucp.net">www.njucp.net</a> and <a href="h

### (h) Form E - Contractor's Monthly DBE Payment Report & Payment Certification Voucher:

Beginning the month following the contract's notice to proceed, the Prime must report monthly payment activity for each DBE subcontractor/subconsultant/supplier; certifies each DBE sub has been paid any amounts due from previous or current progress payments paid to the Prime. (article 5.2.4)

All invoices 30 days past due from NJT must be listed in the appropriate field.

This report is due even if there is no payment activity. This form must be completed and submitted to the OBD by the 7<sup>th</sup> of each month to the attention of the OBD's Manager of Contract Compliance.

### (i) Form E1- DBE Prime's Monthly Payment Report (For DBE Prime Only)

Beginning the subsequent month following the contract's execution date, the DBE Prime must report its monthly payments received by NJT. Refer to article 5.2.5.

All invoices 30 days past due from NJT must be listed in the appropriate field.

This report is due even if there is no payment activity. This form must be completed and submitted to the OBD by the 7<sup>th</sup> of each month to the attention of the OBD's Manager of Contract Compliance.

### (j) Form E2 – DBE's Monthly Payment Report:

The Prime must provide a copy of the Form E2 to each DBE subcontractor/subconsultant/ supplier(s). Beginning the subsequent month following the DBE's execution date, the DBE firm must report its monthly payment activity.

This report is due even if there is no payment activity. This form must be completed and submitted by the DBE only to the OBD by the 7<sup>th</sup> of each month to the attention of the OBD's Manager of Contract Compliance.

All invoices 30 days past due must be listed in the appropriate field. Identify concerns or issues in the comments section to be addressed by the OBD. (Refer to article 5.2.6)

### 2.6 INSTRUCTIONS FOR COMPLETING SUPPLEMENTAL REQUIRED FORMS:

(a) Form AA – Second Tier DBE Utilization: The Second Tier DBE must perform 100% of its subcontract with its own forces. A formal request to waive this requirement may be granted, solely at the discretion of the OBD; however, approval is required.

> DBE Requirements for Federal Procurement Activities 9 of 20

A detailed scope of work must be provided; one-word descriptions are not acceptable.

(Ex. Haul and dispose of approximately 192,000 tons of contaminated soil to a clean earth facility at \$34.00 a ton).

### (b) Form AA1 - Second Tier Bidder/Proposer Solicitation and Contractor Information:

The DBE sub-Prime must submit and complete **page one** (1). Second Tier DBE(s) solicited for and participating on this contract must complete **page two** (2).

### (c) Form AA2- Second Tier Non-DBE Subcontractor Utilization:

DBE sub-Primes are required to report and submit all dollars committed to non-DBEs. The non-DBE portion of work is not counted toward DBE participation credit. **Refer to article 3.0** 

A detailed scope of work must be provided; one-word descriptions are not acceptable.

(Ex. Haul and dispose of approximately 192,000 tons of contaminated soil to a clean earth facility at \$34.00 a ton).

### (d) Form BB - Intent to Perform as a Second Tier DBE Subcontractor:

Each DBE firm listed on Form AA, must complete, and sign. Only Second Tier DBE(s) must complete and sign this form.

The Second Tier DBE must provide a detailed scope of work; one-word descriptions are not acceptable. Descriptions should include: type of services provided, total number of units, price per unit, total cost, etc

### (e) NJ UCP DBE Certification & NAICS Code Verification:

All DBEs listed on Form AA must be certified at the time of contract award. It is the Bidder/Proposer/Prime's responsibility to ensure that DBEs are certified and that their NAICS code(s) match the scope of work to be performed on this contract. Credit will <u>not</u> be given for any work to be performed without the appropriate NAICS code. Status can be verified through <u>www.njcup.net</u> and <u>www.census.gov/eos/www/naics/</u>.

### 3.0 GUIDANCE ON COUNTING DBE PARTICIPATION

- 3.1 If a firm is not currently certified as a DBE in accordance with 49 CFR part 26 at the time of the execution of the contract, the firm's participation will <u>not count</u> toward the DBE goal.
- 3.1.1 A DBE performing less than 51% of its subcontract will not count toward the assigned goal and should not be listed on any forms.
- 3.1.2 When a DBE subcontracts part of the work of its contract to another firm, the value of the subcontracted work may be counted toward DBE goals only if the DBEs subcontractor/subconsultant is a DBE.
- (a) Work that a DBE subcontracts to a <u>non-DBE firm does not count</u> toward the DBE contract goal.
- (b) When a DBE performs as a participant in a **joint venture with a Non-DBE**, count the portion of the total dollar value of the contract equal to the distinct, clearly defined portion of the work of the contract that the DBE performs with its own forces toward DBE goals.
- (c) A DBE performs a *commercially useful function* when it is responsible for execution of the work of the contract and is carrying out its responsibilities by actually performing, managing and supervising the work involved.

A DBE does not perform a *commercially useful function* if its role is limited to that of an extra participant in a transaction, contract, or project through which funds are passed in order to obtain the appearance of DBE participation.

### 3.2 COUNTING DBE PARTICIPATION

- 3.2.1 When a DBE participates in a contract, only the value of the work actually performed by the DBE is counted toward DBE goals.
- (a) The entire amount of that portion of a contract that is performed by the DBE's own forces is counted. This includes the cost of supplies and materials obtained by the DBE for the work of the contract, as well as supplies purchased or equipment leased by the DBE (except supplies and equipment the DBE subcontractor/subconsultant purchases or leases from the Prime contractor or its affiliate).
- 3.2.2 The entire amount of fees or commissions charged by a DBE firm for providing a bona fide service, such as professional, technical, consultant, or managerial services, or for providing bonds or insurance specifically required for the performance of a contract, is counted toward DBE goals, provided the fee is reasonable and not excessive as compared with fees customarily allowed for similar services by a DBE.

### 3.3 DBE PRIME CONTRACTOR GUIDANCE

- 3.3.1 If a <u>DBE Prime</u>, expenditures are counted toward DBE goals only if the DBE is performing a commercially useful function on that contract.
- 3.3.2 A DBE Prime must perform or be responsible at least 30% of the total cost of its contract with its own workforce.
- 3.3.3 If a DBE Prime does not perform or exercise responsibility for at least 30% of the total cost of its contract with its own workforce or subcontracts a greater portion of the work of a contract than would be expected on the basis of normal industry practice for the type of work involved, NJT will consider that it is not performing a commercially useful function and the DBE Prime shall be in breach of the contract and subject to the appropriate remedies and penalties. Refer to Articles 5.6-5.7

### 3.4 DBE TRUCKING FIRMS GUIDANCE

- 3.4.1 A DBE trucking firm is performing a commercially useful function if:
- (a) The DBE is responsible for the management and supervision of the entire trucking operation for which it is responsible on a particular contract, and there is not a contrived arrangement for the purpose of meeting DBE goals.
- (b) The DBE itself <u>owns and operates at least one</u> fully licensed, insured, and operational truck to be used on the contract.
- 3.4.2 The DBE receives credit for the total value of the transportation services it provides on the contract using trucks it owns, insures, and operates using drivers it employs.

### 3.5 LEASING TRUCKS

3.5.1 Leased trucks must display the name and identification number of the DBE.

- 3.5.2 The DBE may lease trucks from another DBE firm, including an owner-operator that is certified as a DBE.
  The DBE who leases trucks from another DBE receives credit for the total value of the transportation services the lessee DBE provides on the contract.
- 3.5.3 The DBE may also lease trucks from a non-DBE firm, including an owner-operator.
- (a) The DBE who leases trucks from a non-DBE <u>is entitled to credit only for the fee or commission it</u> receives as a result of the lease arrangement.
- 3.5.4 For the purposes of this section (Leasing), a lease must indicate that the DBE has exclusive use of and control over the truck.
- (a) This does not preclude the leased truck from working for others during the term of the lease with the consent of the DBE, so long as the lease gives the DBE absolute priority for use of the leased truck.

### 3.6 COUNTING MATERIALS AND SUPPLIES

- 3.6.1 Expenditures with DBEs for materials or supplies are counted toward DBE goals as provided in the following:
- (a) If the materials or supplies are obtained from a DBE manufacturer, 100% of the cost of the materials or supplies are counted toward DBE goals.
  - (1) For purposes of this paragraph 3.6.1(a), a manufacturer is a firm that operates or maintains a factory or establishment that produces, on the premises, the material, supplies, articles, or equipment required under the contract and of the general character described by the specifications.
- (b) If the materials or supplies are purchased from a **DBE regular dealer**, sixty percent (60%) of the cost of the materials or supplies is counted toward **DBE** goals.
  - (1) For purposes of this paragraph 3.6.1(b), a <u>regular dealer</u> is a firm that owns, operates, maintains a store, warehouse, or other establishment in which the materials, supplies, articles or equipment of the general character described by the specifications and required under the contract are bought, kept in stock, and regularly sold or leased to the public in the usual course of business.
  - (2) The firm must be an established, regular business that engages, as its principal business and under its own name, in the purchase and sale or lease of the products in question.
  - (3) A person may be a <u>regular dealer</u> in such bulk items as petroleum products, steel, cement, gravel, stone, or asphalt without owning, operating, or maintaining a place of business as provided above if the person both owns and operates distribution equipment for the products. Any supplementing of regular dealers' own distribution equipment shall be by a long-term lease agreement and not on an ad hoc or contract-by-contract basis.
  - (4) Packagers, brokers, manufacturers' representatives, or other persons who arrange or expedite transactions are not regular dealers within the meaning of this paragraph 3.6.1(b).
- (c) With respect to materials or supplies purchased from a DBE, which is neither a manufacturer nor a regular dealer, only the entire amount of fees or commissions charged for assistance in the procurement of the materials and supplies, or fees or transportation charges for the delivery of materials or supplies required on a job site, is counted toward DBE goals, provided the fees are determined to be reasonable and not excessive as compared with fees customarily allowed for similar service. However, any portion of the cost of the materials

### and supplies themselves do not count toward DBE goals.

### 4.0 TERMINATION OF DBE(s)

4.1 The Bidder/Proposer/Prime shall not terminate for convenience, or any other reason, and then perform the work of the terminated subcontractor/subconsultant with its own forces 9self-perform) or those of an affiliate without NJT's prior written consent. Refer to Article 4.3.

Submission of all REQUIRED FORMS is mandatory for the following Articles 4.2 through 4.5

### 4.2 ADDITION OF DBE(s)

- 4.2.1 Should the Bidder/Proposer/Prime wish to add a DBE not listed on Form A, a written request for the addition of a DBE(s) must be submitted by the Bidder/Proposer/Prime.
- 4.2.2 The Bidder/Proposer/Prime must receive written approval of the OBD <u>prior to</u> the addition of the DBE subcontractor/subconsultant in order for the addition to be credited toward the goal.

### 4.3 REPLACEMENT OR REMOVAL OF DBE(s)

- 4.3.1 When a Prime is considering replacing or removing a DBE due to performance issues, the OBD must be contacted as soon as possible.
- 4.3.2 Request for DBE replacement or removal may be made under the following conditions:
  - 1) The DBE materially fails to successfully perform the contract tasks.
  - 2) Under unusual situations referenced in article 4.3.8.
- 4.3.3 A written request for replacement or removal of a DBE(s) listed on Form A, must be submitted by the Bidder/Proposer/Prime to the OBD with complete justification for the request. The process to follow such requests is as follows:
- (a) Written communications (over a period) from the Prime and/or NJT's PM/CM team to the DBE, notifying the DBE of its poor performance must be provided to the OBD.
- (b) The OBD will arrange a meeting with the DBE, the Prime, and a representative from Procurement and project management to discuss the specifics of the performance issue.
- (c) The DBE must provide a written plan identifying the efforts it will make to correct the deficiencies.
- (d) The Prime must provide the DBE with a minimum of 30 calendar days from acceptance of its plan to improve its performance. Throughout the 30-day window, the Prime and/or NJT PM/CM team must provide written communication to the DBE of any additional/continued performance issues, with a copy to the OBD.
- 4.3.4 The Bidder/Proposer/Prime must receive written approval of the OBD <u>prior to</u> replacement or removal of the DBE subcontractor/subconsultant can be made, <u>regardless of the reason for the replacement or removal</u>.
- 4.3.5 If the OBD issues written approval for the removal of a DBE(s), NJT will require a Bidder/Proposer/Prime to continue to demonstrate a good faith effort to replace the removed DBE to the extent needed to meet the contract goal established by NJT for the procurement.
  - These good faith efforts shall be directed at finding another DBE to perform at least the same amount of work under the contract as the removed DBE.

- 4.3.6 Failure to obtain approval from the OBD **prior to** replacing or removing a DBE shall result in the Bidder/Proposer/Prime being found in breach of the contract and subject to the appropriate remedies, audits and penalties. **Articles 5.6-5.7**
- 4.3.7 If NJT finds that the Bidder/Proposer/Prime upon submission of its bid/proposal committed itself to the goal in good faith, the Bidder/Proposer/Prime may, in "unusual situations", be permitted to substitute a DBE subcontractor(s)/subconsultant(s).
- 4.3.8 The term "unusual situations", includes, but is not limited to, the following circumstances:
- (a) Failure to qualify as a DBE, or maintain DBE certification status.
- (b) Death or physical disability of a key individual.
- (c) Dissolution, if a corporation or partnership.
- (d) Bankruptcy of the subcontractor/subconsultant, subject to applicable bankruptcy law, and only in instances where the bankruptcy affects the subcontractor/subconsultant's ability to perform.
- (e) Inability to obtain, or loss of, a license necessary for the performance of the particular category of work.
- (f) Failure or inability to comply with a requirement of law applicable to Primes or, subcontractors/subconsultants.

### 4.4 WITHDRAWN DBE(s)

- 4.4.1 When a DBE is unable to complete a subcontract (withdraws), for any reason, NJT will require a Bidder/Proposer/Prime to make a good faith effort to replace a withdrawn DBE at least to the extent needed to ensure that the Prime contractor is able to meet the contract goal established by NJT for the procurement. These good faith efforts shall be directed at finding another DBE to perform at least the same amount of work under the contract as the removed DBE.
- 4.4.2 The Bidder/Proposer/Prime is required to make a good faith effort to seek other DBE subcontractors/subconsultants in substitution of the original DBE. The good faith efforts described in article 2 are required in finding another DBE to perform at least the same amount of work under the contract as the DBE that was terminated, to the extent needed to meet the contract goal established for the procurement.

### 4.5 DECERTIFIED DBE(S)

- 4.5.1 If the Prime has reason to doubt that a proposed DBE is still eligible due to change in ownership, management, or size, the Prime shall, within 10 days of learning this information, notify NJT of that fact in writing.
- 4.5.2 If the subcontract has not been executed before the DBE's ineligibility occurs, the Prime will not receive credit toward the contract goal for the ineligible DBE. The Prime may continue to use the ineligible DBE, however, DBE participation credit will not be granted.
- (a) To the extent necessary to meet the assigned contract goal, the Prime will make a good faith effort to replace the ineligible DBE within 10 days after notification by the OBD. The OBD will support the Prime in its efforts to replace with an eligible DBE firm in order to meet the contract goal.
- 4.5.3 If the contract has been executed before the firm has been notified of its ineligibility, the Prime may continue to use the firm on the contract and receive credit toward its DBE goal for the duration of that particular phase of or option in the contract. This may not apply to future phases or options, which will be handled on a case-by-case

- basis at the sole discretion of the OBD.
- 4.5.4 Exception: If the DBEs ineligibility is caused solely by its having exceeded the size standard during the performance of the contract the Prime may continue to count its participation on that contract only toward the contract goals. This may not apply to future phases or options and will be subject to determination by the OBD.

### **5.0 AWARD OBLIGATIONS**

- 5.1 The Prime must designate a DBE Liaison Officer. The liaison officer will be responsible to NJT regarding DBE subcontract matters.
- 5.1.1 If at any point during the contract's life, the Prime's DBE participation falls below the assigned goal, the Prime must identify additional work or new work items for which it will subcontract to DBEs to the extent necessary to meet the assigned goal. Any new scope of work issued to the Prime shall still be subject to the assigned goal.
- 5.1.2 The OBD will support the Prime in identifying current/future opportunities in the contract to meet the assigned contract goal.
- 5.1.3 Should the Prime seek a change that addresses the DBE's performance, or affects the work scope and/or compensation, the OBD must be notified, prior to implementation, for its review and approval of the changes as soon as possible. No change will be allowed without <u>prior review and approval</u> by the OBD. Failure to notify the office and obtain approval prior to a change shall result in breach of the contract and may be subject to the appropriate remedies, audits, and penalties.
- 5.1.4 Whenever NJT issues project change orders the goal may still apply; the OBD will determine if increased DBE participation will be required.
- 5.1.5 To ensure that all obligations under subcontracts awarded to DBEs are met NJT shall review the Prime's DBE involvement efforts during the performance of the contract.

### **5.2 POST AWARD DELIVERABLES**

- 5.2.1 After the execution of a contract with NJT, **signed copies** of subcontractor/subconsultant agreements between the Prime and DBE subcontractors must be submitted to the OBD <u>no later than 10 business days after the Prime's contract execution date</u>. The agreement between the Prime and DBE subcontractor shall remain firm for the duration of the contract.
- 5.2.2 The Prime shall provide a list of the anticipated job start date for all DBE subcontractors/subconsultants no later than two days after the initial pre-construction meeting.
- 5.2.3 Certification of DBE(s) Payments submit monthly with the Form E to the Manager of the OBD and with its monthly invoice submittal to NJT project manager of this project. Refer to article 5.3.2
- (a) The Prime will certify, <u>prior to the issuance of each progress payment by NJT</u>, that all DBE subs have been paid any amounts due on past due invoices from previous or current progress payments.
- 5.2.4 Form E (Contractor's Monthly DBE Payment Report & DBE Payment Certification Voucher) submit monthly to the Manager of the OBD. Refer to articles 2.5h and 5.3.2.
  - Failure to submit this report on a monthly basis may result in breach of the contract and be subject to the appropriate remedies, penalties or liquidated damages as indicated in articles 5.6-5.7.
- 5.2.5 Form E1 (DBE Prime's Monthly Payment Report) (For DBE Prime Only) submit monthly to the Manager of DBE Requirements for Federal Procurement Activities

the OBD.

### 5.2.6 Form E2 (DBE's Monthly Payment Report) - Refer to article 2.5j

- (a) Forms E/E1 and E2 will be reviewed monthly to determine compliance with the assigned DBE goal, the subcontractor prompt payment regulation, and the DBE Program.
- (b) Attainment of goals will be monitored and based upon actual payments <u>received</u> by the DBE.
  Failure to submit Form E/E1 may result in suspension of payments or such other remedies as provided in <u>article</u>
  5.6. If at any time, NJT has reason to believe that any person or firm has willfully and knowingly provided incorrect information or made false statements, it shall refer the matter to the USDOT for inquiry.

### 5.3 PROMPT PAYMENT OF INVOICES TO SUBCONTRACTORS

- 5.3.1 The Prime must pay each subcontractor under this contract for satisfactory performance of its work no later than ten (10) days from the receipt of each payment the Prime receives from NJT for the subcontractor's work. Failure to comply with this requirement shall result in breach of the contract and shall be subject to the appropriate remedies as determined by OBD.
- 5.3.2 In accordance with 49 CFR 26.29, the Prime shall certify, prior to the issuance of a progress payment by NJT that all subcontractors have been paid any amounts due on past due invoices (greater than 30 days) from previous or current progress payments. The Prime must submit the Certification of DBE(s) Payments with its monthly invoice submittals to NJT project manager, and with its Form E to the OBD.
- 5.3.3 The Prime will not be reimbursed for work performed by subcontractors/subconsultants unless and until the Prime ensures that the subs are promptly paid for the work performed. Alternatively, the Prime shall certify that a valid basis exists under the terms of the subcontractor's/subconsultant's or supplier's contract to withhold payment from the subcontractor/subconsultant and therefore payment is withheld.
- 5.3.4 If the Prime withholds payment from the subcontractor/subconsultant, the Prime shall provide to the subcontractor/subconsultant or supplier written notice thereof. The notice shall detail the reason for withholding payment and state the amount of the payment withheld. If a performance/payment bond has been provided under this contract, the Prime shall send a copy of the notice to the surety providing the bond for the Prime. A copy of the notice shall also be submitted to NJT with the certification that payments are being withheld.
- 5.3.5 If withholding payment is due to the Prime's failure to promptly pay the DBE in accordance with the prompt payment of invoices and/or retainage clauses, the OBD may request proof of payment to DBE(s) for delinquent invoices and/or retainage in order to issue release of payment to Prime.
- 5.3.6 Failure to comply with the above shall result in breach of the contract and may be subject to the appropriate penalties. See article 5.6

### 5.4 SUBCONTRACTOR PAYMENT DISPUTE RESOLUTION

5.4.1 The Prime is required to notify the OBD of its intention to withhold payment from a DBE as soon as possible and in advance of taking action. Should the Prime provide notice and proceed to withhold payment from any subcontractor/subconsultant or supplier due to a performance issue or unapproved work performed, an OBD representative shall make an effort to resolve the dispute.

DBE Requirements for Federal Procurement Activities 16 of 20

- (a) OBD's efforts shall be limited to meeting with the Prime and the subcontractor/subconsultant, and reviewing the relevant facts with both parties.
- (b) OBD will not act as a decider of fact nor will OBD direct a settlement to the dispute.
- (c) Any OBD effort is solely intended to assist the parties in understanding their respective positions and to encourage a reasonable resolution of the dispute. The Prime is required to send written notification of the above to the OBD immediately.
- 5.4.2 Should payments be withheld that are not related to the previous items mentioned, and/or a determination can be made that the withholding of payments violates the prompt payment clause, NJT may execute the appropriate remedies in accordance with article 5.6.

### 5.5 PROMPT PAYMENT OF SUBCONTRACTOR RETAINAGE (FOR CONSTRUCTION CONTRACTS ONLY)

- 5.5.1 The Prime must include a contract clause in the subcontractor agreement obligating the Prime to pay <u>all</u>

  retainage owed to the subcontractor/subconsultant for satisfactory completion of the accepted scope of work no later than 15 days after the DBE subcontractor's/subconsultant's work is satisfactorily completed.
- 5.5.2 Only subcontractors/subconsultants whose work has been 100% completed, including all punch list work or remaining work, and who have supplied closeout documents shall be eligible for release of retainage. Any delay or postponement of payment from the above referenced time frame may occur only for good cause <u>following written</u> <u>approval of NJT</u>.
- 5.5.3 NJT may agree to release an equivalent amount of Prime retainage provided that:
- (a) There is no offsetting claims from NJT (including, but not limited to, liquidated damages), other subcontractors/subconsultants, material men, or workers;
- (b) None of the other reasons to withhold payments specified under the Prime contract exists.
- 5.5.4 Prior to release of the Prime's retainage, the Prime shall provide to NJT executed copies of the following subcontractor closeout documents, (shown in Appendix A of the contract) as appropriate:
- (a) Consent of Surety to Final Payment to the Subcontractor
- (b) Certificate of Amounts Due and Owing to Subcontractor Employees
- (c) Subcontractor Release of Claims
- (d) Subcontractor Release of Liens and a Certificate of Final Acceptance of Subcontractor Work
- 5.5.5 Notwithstanding NJT's release or partial release of retainage, nothing in this clause shall be deemed to constitute NJT's partial or final acceptance of the work, or any portion thereof, unless either a Certificate of Partial Acceptance or a Certificate of Final Acceptance has been executed by NJT, in the form(s).

### 5.6 Audit and Penalties

During the performance of any contract and for a period of up to three (3) years following completion of the contract work, NJ TRANSIT may conduct reviews for compliance with the requirements of the DBE Program. Such reviews may include, but not be limited to, the evaluation of monthly reports, desk audits and site visitations.

5.6.1 Where a Prime is found to be in breach of the requirements of the DBE Program during the performance of the DBE Requirements for Federal Procurement Activities

contract, and does not promptly take corrective action, the following sanctions may be instituted (singularly, in any combination, and in addition to any other contractual remedies or otherwise provided by law):

- (a) The Prime may be ordered to stop work without penalty to NJT.
- (b) The contract may be terminated for breach.
- (c) Suspension or debarment proceedings may be commenced in accordance with New Jersey law.
- (d) The relevant performance bond may be enforced.
- (e) NJT may withhold payment of specific invoices.

### 5.7 LIQUIDATED DAMAGES

- 5.7.1 Liquidated damages (LD) may be assessed when the Prime fails to meet the established DBE goal on the contract.
- 5.7.2 If the DBE goal is not met, and the Prime has not demonstrated a good faith effort to do so, NJT may elect to subtract from the Prime's payment, as liquidated damages and not a penalty, the following:

  The amount equal to the difference (in dollars) between the total contract value multiplied by the assigned DBE goal percentage, (originally established or as subsequently modified) and the actual DBE participation percentage (total dollars paid to DBEs divided by total dollars paid to the Prime).
- 5.7.3 This may be withheld from a series of payments or from the Prime's final payment, depending on the size of the liquidated damage.
- 5.7.4 If the Prime's final payment is not sufficient to satisfy the LD in full, the balance shall be due and owing from the Prime and subject to repayment terms as determined by NJT. NJT shall waive liquidated damages where good cause is shown for the deficiency in DBE participation upon determination by the OBD.

### **APPENDIX I**

### GLOSSARY

A Good Faith Effort-the efforts employed by the bidder, which should be those that one could reasonably expect a bidder to take if the bidder were actively and aggressively trying to obtain DBE participation sufficient to meet the contract goal.

<u>Certification</u> - means the process by which a business is determined to be a bona fide DBE. Any business applying for DBE certification must complete the appropriate NJ Unified Certification Program Application. Certification Applications are available at the OBD.

### Disadvantaged Business Enterprise or DBE - means a small business concern:

Which is at least 51 percent owned by one or more socially and economically disadvantaged individuals, or in the case of any publicly owned business, at least 51 percent of the stock of which is owned by one or more socially and economically disadvantaged individuals; and where one or more of the socially and economically disadvantaged owners controls management and daily business operations. A DBE shall not include a small business concern where that concern or a group of concerns controlled by the same socially and economically disadvantaged individual or individuals has annual average gross receipts in excess of \$22,410,000 over the previous three fiscal years or is not otherwise eligible as a small business as defined by the Small Business Administration in 13 CFR Part 121.

<u>DBE Goal</u> - means numerically expressed objectives for DBE participation on federal contracts Prime contractors are required to make a good faith effort to achieve to the extent necessary to meet the assigned DBE goal.

<u>DBE Sub-Prime</u> - means any 1<sup>st</sup> Tier DBE subcontractor/subconsultant listed on the Form A that will subcontract any portion of its subcontract/scope of work to a DBE(s) and/or non-DBE(s) firm(s).

<u>DBE Ineligibility</u> – means a firm's DBE status changes or ceases due to change in ownership, management, or size, etc. <u>DBE Prime</u> – means the successful Bidder is a DBE firm and has a direct contract with NJT.

<u>DBE Trucking Firm</u> – owns and operates at least one fully licensed, insured, and operational truck used on the contract. Is responsible for the management and supervision of the entire trucking operation for which it is responsible on a particular contract.

First Tier DBE - refers to any DBE listed on the Bidder/Proposer/Prime's Form A and having a direct contract with the Prime.

<u>Joint Venture</u>—means an association of a DBE firm and one or more other firms to carry out a single, for-profit business enterprise, for which parties combine their property, capital, efforts, skills and knowledge, and in which the DBE is responsible for a distinct, clearly defined portion of the work of the contract and whose share in the capital contribution, control, management, risks, and profits of the joint venture are commensurate with its ownership interest.

<u>Prime</u> - means any contractor or consultant, including a DBE contractor/consultant, who enters into a direct contractual relationship with NJT.

Race-conscious Measure or Program - is one that is focused specifically on assisting only DBEs, including women-owned DBEs.

Race-neutral Measure or Program- is one that is focused specifically on assisting all small businesses equally, including DBEs. Such activities include bonding, insurance, and technical assistance. For the purposes of this part, race-neutral includes gender-neutrality.

Reasonable Bid Price - means a price that shall be considered reasonable if the Bidder/Proposer/Prime would have been awarded the contract had the firm submitted the only bid.

Regular Dealer - means a firm that owns, operates, or maintains an establishment in which the materials or supplies required for the performance of a contract are bought, kept in stock and regularly sold to the public in the usual course of business.

The firm must engage in, as its principal business, and in its own name, the purchase and sale of products in question. Bulk items such as steel, cement and petroleum products need not be stocked, if the dealer owns or operates distribution equipment.

Note: Brokers and packagers are not regarded as regular dealers.

Second Tier DBE - refers to any DBE listed on the DBE Sub-Prime's Form AA.

<u>Subcontractor/Subconsultant</u> - means any contractor/consultant, including suppliers, who enters into a contract issued by a Prime contractor.

Transit Vehicle Manufacturer (TVM) - is a manufacturer of vehicles used by NJT for the primary program purpose of public mass transportation (e.g., buses, railcars, vans). The term does not apply to firms, which rehabilitate old vehicles, or to manufacturers of locomotives or ferryboats. The term refers to distributors of or dealers in transit vehicles with respect to the requirements of 49 CFR Section 26.49.

<u>U.S. DOT</u> – means the U.S. Department of Transportation, including the Office of the Secretary, the Federal Highway Administration (FHWA), the Federal Transit Administration (FTA), and the Federal Railroad Administration (FRA).

Withdrawn DBE - a DBE withdraws, or drops out of its contract, or fails to complete its work on the contract for any reason.

# MANDATORY FORM FOR BIDDER/PROPOSER/PRIME: COMPLETE ENTIRELY

### FORM A (Fed)

## First Tier DBE UTILIZATION - FORM A

Project Name:	NJT Contract No:		
Assigned DBE Goal %:NJT Procurement Specialist:	Contract Value (\$):		
ier DBE must perform at least 51% of its subcontract value	First Tier DBE must perform at least 51% of its subcontract value if subcontracting to a Second -Tier DBE or Non-DBE. Do not count Non-DBE portion toward the goal.	ot count Non-DBE portior	toward the
Name, Address and Telephone # of DBE Subcontractor/Subconsultant	Provide <u>Detailed</u> Scope of Work to be Performed (Identify all suppliers)	Dollar Value of Subcontract/Sub- consultant Work (\$) Awarded	Percentage o Subcontract Work (%)
			0,
For DBE suppliers, show original subcontract value multiplied by 60% (\$2,000*60%=\$1200). For DBE portion of work, subtract Non-DBE portion of work from original subcontract value.	TOTALS	v	

The undersigned will enter into a formal agreement with the DBE(s) listed in this schedule conditioned upon execution of a contract with NJ TRANSIT for the above referenced project. The undersigned understands that removal/replacement of the DBE(s) listed is NOT PERMISSIBLE for any reason (pre or post-award), without submitting a written request to the Office of Business Development. Failure to obtain written approval shall result in the breach of contract and subject to corrective action to be determined by NJ TRANSIT.

Company Name:	Authorized Signature:
Company Address:	Print Name:
	Title:
ederal Tax ID #:	Prime Contractor's DBE Liaison Officer:
Company Tel #:	Date Signed:

To Add Subs Use Additional Forms

## MANDATORY FORM: COMPLETE ENTIRELY

# BIDDER SOLICITATION & CONTRACTOR INFORMATION - FORM A1

Project Title:		Date:	
Prime Contractor/Consultant:		Telephone #:	
Complete the information	Complete the information below for Bidder/Proposer/Prime(s) working on this project. Use Page 2 for all subcontractors/subconsultants	on this project <u>.</u> Use Page 2 for all subcontrac	ors/subconsultants
	Bidder/Proposer/Prime	Bidder/Proposer/Prime	Bidder/Proposer/Prime
Company's Full Name			
Address			
City			
Zip			
County			
Phone			
Fax			
E-mail			
Owner			
Date Established			
Date Certified			
Ethnicity	;		
Gender			
Certification Status: DBE or Non-DBE			
Federal Tax ID # / SSN #			
Annual Gross Receipts: A - Less than \$500K			
B - \$500K to \$1M C - \$1M to \$2M			
D - \$2M to \$5M F - \$5M and Over			
indicate the letter that applies			
Primary NAICS Code:			

Page 1

NJT Subrecip - Fed Form A1 rev Jun 2010, rev 0314

To Add Subs Use Additional Forms

## MANDATORY FORM: COMPLETE ENTIRELY

# BIDDER SOLICITATION & CONTRACTOR INFORMATION - FORM A1

Date:	Telephone #:	
Project Title:	Prime Contractor/Consultant:	

# COMPLETE THE INFORMATION BELOW FOR "ALL" FIRMS INCLUDING SUPPLIERS SOLICITED; INCLUDING THOSE THAT $\overline{WILL}$ WORK ON THIS PROJECT.

		THE HOLD ON THIS PROJECT.	TILL HOLD ON THIS PROJECT.
	Subcontractor/Subconsultant	Subcontractor/Subconsultant	Subcontractor/Subconsultant
Company's Full Name			
Address			
City			
Zip		3	
County			
Phone			
Fax			
E-mail			
Owner			
Date Established			
Date Certified			
Ethnicity			
Gender			
Certification Status: DBE or Non-DBE			
Federal Tax ID # / SSN #			
Annual Gross Receipts: A - Less than \$500K B - \$500K to \$1M			
D - \$2M to \$5M E - \$5M and over			
indicate the letter that applies			
Primary NAICS Code:			

NON-DBE SUBCONTRACTOR UTILIZATION - FORM A2 Directions: To be completed by any Bidder/Proposer/Prime for "all" subs including suppliers participating on this contract.

Bidder/Proposer Prime Name:		Project Title:		
Date:		Prime Contract Value:		
Name, Address and Telephone # of all Subcontractor/Subconsultants	# EES	Provide Detailed Scope of Work to be Performed	Subcontractor/Sub- consultant Work (\$)	Percentage of Subcontract
				(%)
			vs .	%
			n	00
			vo vo	% )
Mist provide a detailed score of make and			S	%
most provide a detailed scope of work; one-word descriptions are not acceptable.	riplions are not acceptable.	TOTALS	и	%

### INTENT TO PERFORM AS A 1<sup>ST</sup> TIER DBE - FORM B

The Bidder/Proposer/Prime is prohibited from completing any portion of this form and directing the DBE to sign a blank form.

<u>DIRECTIONS</u>: DBE(s) listed on the Form A must complete all information on this form.

Print Name Telep	phone #:
Signature of 1 <sup>st</sup> Tier DBE Date Title	
The undersigned DBE will enter into a formal agreement for conditioned upon execution of a contract with NJ TRANSIT. As certification, compliance and monitoring process set forth by NJ 51% of my subcontract with my own workforce for the referenced	a DBE subcontractor, I will cooperate with the TRANSIT. I attest that I will perform at least
DBE Contract Start Date: DBE Contract Completion	
The Prime Contractor <i>projected</i> the following commencement and cor	mpletion date for such work as follows:
The undersigned based the above scope of work and subcontract value contractor named above. Circle one. (Yes or No)	e on detailed project specs received from the Bidder
Total Quantity/Units (if applicable): Per Unit	Cost (if applicable): \$
Dollar Value of DBE Subcontract: \$	
The undersigned will perform the following described work on description of the type of work you will perform on your subcontra Bidder (optional)).	<u>ct</u> . Attach a copy of quote approved and signed by
Intend to subcontract any portion of its scope of work to a Non-DBE(  If yes, must complete and submit Form AA2.	(s)? Circle one. (Yes or No) At what percent?%
Intend to subcontract any portion of its scope of work to a DBE(s)? If yes, DBE Sub-Primes <u>must complete and submit</u> Form AA.	Circle one. (Yes or No)  At what percent?%
Does the undersigned DBE (Ansemble Intend to perform subcontract work in connection with the above-meter or No)	wer Accordingly): entioned project as a Joint Venture? Circle one. (Ye
Project/Contract Name:	n B/R/1 Contract Number:
Project/Control N	IFB/RFP Contract Number:

Failure to adhere to these instructions or the falsification of any information on this form shall result in breach of contract and subject to the appropriate penalties to be determined by NJ TRANSIT.

### Mandatory Form Submit Monthly

PRIME CONTRACTOR Monthly DBE Payment Report - Form E

Fed Form E (Page 1 of 2)

Name of Project:	Name of Project: Prime Original Contract Value:			NJT Cont	JE Fayillelli	NJT Contract #:	act #:			
all	Change Orders (Overall Inc/Dec.):					Heport for t	Report for the Month of: Notice to Proceed Date:			
으	Total Contract Amount to Date:					NJT Project	NJT Project Mar Name:			
<u>\$</u>	Total Payments Received from NJT Date:	ate:				Assigned D	Assigned DBE Goal %:			
	Work Task Performed	Original Confract Amount \$	Change Order Amount \$ (+/-)	Amount of Invoice Received this Month (\$)	Date of Invoice Received in this Month	Payment(s) Made to DBE in this Month in \$\$ (Itemize)	Date(s) Payment Made This Month	Total DBE Payments made to Date in (\$)	% Overall Work Finished	iii E &
	Total(s)→	₩.	<b>6</b> 3	<del>€9</del>	Total(s)	vs.	Total(s)	S		
1										

Itemize payments/invoices and dates if paid/received more than one payment/invoice between the 1st and 31st of THIS Month. Prime Contractor Information:

Project Director Name:	Project Director Signature:	Federal TIN #:
		Jate:
Prime Firm Name:	Address:	Telephone #:

Prime's Past Due Invoice Information: List any invoice more than 40 days past due from the date submitted to NJT at the time you complete this form.

Invoice #	Invoice Date	Invoice Amount (\$)	Number of Days Past Due	Comments:
This form is	This form is not to be altered in any way.		For assistance completing this form,	completing this form, call 973-491-7539, 8058, 8768, 8575, 8069, 8941 Fed Form E rev Sept 2010

# Form E - Prime Contractor's DBE Payment Certification

- Have all DBE subcontractors with executed subcontracts been paid amounts due from previous progress payments?
- If yes, skip the next section and go to number 3.
- If no, please complete fields in box below: (Use additional paper, if needed)

Specific Reason for Withholding	
Invoice Date	
Invoice Number	
Total of Invoice Amount (\$)	
Amount Withheld From Invoice (\$)	
DBE SubContractor Name	

- Have you notified the DBE subcontractor(s) that you are withholding payment and the reason(s) why? તાં
- If yes, provide a copy of written notification to the DBE subcontractor with this form, indicating the date of notification. 00
- If no, lack of prior written notification to the DBE(s) that you are withholding payment violates the prompt payment clause guidelines. Please contact the DBE immediately, and provide a copy of written notification to the subcontractor with this form.
- By signing this form, I certify that all of the above represent true and accurate information. က

Note: CFO or equivalent Sr. Manager must complete and sign off on this form.

DATE

PROJECT DIRECTOR (SIGNATURE)

Additional Reasons/Comments for Withholding Payment:

PROJECT DIRECTOR NAME (PRINT)

DO NOT WRITE BELOW. DEPARTMENTAL USE ONLY.

Approved

Denied

THIS FORM IS DUE ON THE 7TH OF EACH MONTH Please forward to: Office of Civil Rights and Diversity Programs **Business Development** NJ TRANSIT

One Penn Plaza East, 6th Fl

Newark, New Jersey 07105-2246

This form is not to be altered in any way.

Fed Form E rev Sept 2010 For assistance completing this form, call 973-491-7539, 8058, 8768, 8575, 8069, 8941

# DBE PRIME CONTRACTOR'S MONTHLY PAYMENTS FROM NJ TRANSIT INFORMATION ON CONTRACT

DATE CONTRACT EXECUTED:	ä	55	CONTRACT NUMBER: _	REPORT FO	REPORT FOR THE MONTH OF:	YEAR	
ORIGINAL CONTRACT AMOUNT:	UNT:			FED TAX ID #:			
CHANGE ORDERS (OVERALL INC/DEC.):	L INC/DEC.)			PURCHASE ORDER #*	BDEB #:		
TOTAL CONTRACT AMOUNT TO DATE:	TO DATE:			NAME OF PROJECT:	OJECT:		
Original Contract Value	Change Order Amount +/-	Date of Change Order	New Contract Value	Payments Received This Month	Total Payments Received to Date	% Work Completed To Date	Final Pmt V/N
TOTALS:							
PRIME INVOICE 30 DAYS PAST DUE FROM NJT:	DAYS PAS	T DUE FRO	M NJT:		TRANSIT I'N	LISN	
Invoice Date Reference #		No. Days P	No. Days Past Due Amount	Project Manager (Name): Telephone #:	ger (Name): ַ		
PRIME CONTRACTOR INFORMATION	ŞI			FORM	FORM IS DUE ON THE 7 <sup>TH</sup> OF EACH MONTH. PLEASE FORWARD TO:	CH MONTH.	
Compliance Officer (Name):					NJ TRANSIT		
Date: Telephone #:				ō	Manager, Contract Compliance Office of Business Diversity – 6 <sup>TH</sup> FL	ance - 6 <sup>TH</sup> FL	
					One Penn Plaza East		

Fed - Form E-1 rev12/9/08

THIS FORM IS NOT TO BE ALTERED OR CHANGED IN ANY WAY.

Newark, New Jersey 07105-2246

## DBE SUBCONTRACTOR Monthly Payment Report - Form E2

Name of DBE Firm:	Report for the Month of:
DBE's FEIN#:	Contract Number:
DBE Address:	Contract Name:
DBE Telephone #:	DBE Contract Start Date:
Prime Contractor's Information:	
Name of Prime:	Telephone #:
DBE PAYMENT INFO: Itemize payments/invoices and dates if received/submitted more than one payment/invoice between the 1st and 31st of THIS Month.	han one payment/invoice between the 1 <sup>st</sup> and 31 <sup>st</sup> of <u>THIS</u> Month.

	Final Payment? Y or N			
	Total % Work To			
	Total Payments Received by DBE To Date (S)			w
	Date Payment(s) Received in this Month			TOTALS
	Total Payments Received by DBE In this Month * (S)			မ
	Date of Invoice(s) Submitted this Month			TOTALS
	Dollar Amount of Each Invoice Submitted in this Month			S
-  -	Invoice #(s) Submitted in this			TOTALS→
	Change Order Amount (+/-)			s
	Original Subcontract Amount S			ശ
	Work Task Performed			TOTALS →

. Did your final payment include retainage? Yes or No (circle one) Is retainage held on your subcontract? Yes or No (circle one) If yes, how much? \$\_

		_
s than 40 days past due from date <u>submitted</u> to prime at the time you complete this form.	Comments: use additional paper if necessary	
more than 40 days past du	Number of Days Past Due	
Past Due Invoice(s) Information: List any invoice more I	Invoice Amount (\$)	
voice(s) Inform	Invoice Date	
Past Due In	Invoice #	

Note: CFO or equivalent Sr. Manager must complete and sign off on this form.

Date:
Title:
Signature:
Name:

THIS FORM IS DUE ON THE 7<sup>TH</sup> OF EACH MONTH <u>IMMEDIATELY FOLLOWING DBE'S SUBCONTRACT START DATE,</u> EVEN IF PAYMENT NOT RECEIVED.

Please mail this form to: NJ TRANSIT, Office of Business Development, One Penn Plaza East, 6<sup>th</sup> FI, Newark, New Jersey 07105-2246

Do not alter this form in any way.

If you need assistance completing this form please call 973-491-7539, 8058, 8768, 8069, or 8941.

Rev Fed Form E2 - Sept 2010

ATTACHMENT C THE FEDERAL GENERAL WAGE DETERMINATION FOR SOMERSET COUNTY, NEW JERSEY 8 PAGES)

General Decision Number: NJ140063 05/09/2014 NJ63

Superseded General Decision Number: NJ20130063

State: New Jersey

Construction Type: Heavy

County: Somerset County in New Jersey.

HEAVY CONSTRUCTION PROJECTS

Modification Number

Publication Date

01/03/2014

05/09/2014

BRNJ0004-001 05/01/2013

		Rate

Fringes

CEMENT MASON....\$ 38.25

27.39

CARP0006-014 11/01/2012

Rates Fringes

CARPENTER.....\$ 41.49

56%

The first sixty feet at the regular rate, 10% per hour additional for each additional fifty feet thereafter.

### CARP0715-007 05/01/2014

Ra	t	e	S
----	---	---	---

Fringes

Millwright.....\$ 43.88

Work of erection and dismantling of elevators and towers, such as concrete conveyors and temporary material elevators, scaffolding or other structures to be used as scaffolding inside or outside of buildings: the first sixty feet at the regular rate, 10% per hour additional for each additional fifty feet thereafter.

CARP1456-013 11/01/2012

1	Rates	Fringes
CARPENTER (Dock Builder)		
Concrete form work\$		31.40
All other work\$	40.87	31.40

Work on land pile driving, floating marine construction and the construction of wharves, while handling and working with creosote and creosote-impregnated products: \$.25 per hour additional.

Work on hazardous/toxic/contaminated waste removal, on a

hazardous/toxic/contaminated waste site, where the worker comes into contact with hazardous/toxic/contaminated waste material, and when A, B or C personal protective equipment is required and used for respiratory, skin or eye protection: 20% per hour additional.

### ELEC0102-025 06/03/2013

SOMERSET COUNTY (does not include the township of Franklin east of a line following Cedar Grove Lane from the Raritan River, in a southwesterly direction, to the Millstone Branch of the Pennsylvania Railroad; then, west along the railroad to the Delaware and Raritan Canal; then, south along the canal to the Middlesex County line; does not include the Township of Montgomery west and south of a line following U.S. Hwy. 206 (formerly State Hwy. 31) north from the Mercer County line to Harlingen Rd.; then, west along Harlingen Rd. and the Dutchtown-Zion road to the Hillsborough township line)

E	Rates	Fringes
ELECTRICIAN		
Cable Splicer	55.17	55.5%
Electrician\$	50.15	55.5%

Work forty ft. or more above the ground or protective rigging (does not apply to pole work, or to use of a manlift or high reach-type lift): 10% per hour additional.

Work with, or the removal of, asbestos materials: 112% times the journeyman rate.

### ELEC0269-023 10/01/2012

SOMERSET COUNTY (Township of Montgomery (west and south of a line following U.S. Hwy. 206 (formerly State Hwy. 31) north from the Mercer County line to Harlingen Rd.; then, west along Harlingen Rd. and the Dutchtown-Zion road to the Hillsborough township line))

	Rates	Fringes
Electrician	\$ 47.34	61.48% + .25
ELEC0456-009 06/03/2013		

SOMERSET COUNTY (Township of Franklin (east of a line following Cedar Grove Lane from the Raritan River, in a southwesterly direction, to the Millstone Branch of the Pennsylvania Railroad; then, west along the railroad to the Delaware and Raritan Canal; then, south along the canal to the Middlesex County line)):

	Rates	Fringes
Electricians:		
Cable splicers	50.48	66%
Electrician\$	46.33	67.25%

Work on line voltage of 440 volts and over: 10% per hour additional.

Work from trusses, scaffolds and ladders 40 ft. or more from the ground or floor; or under air pressure; or over conveyors or moving equipment or machinery: 10% per hour additional.

ENGI0825-021 07/01/2013

	Rates	Fringes
Power equipment operators: GROUP 1\$	42.00	
GROUP 2\$	43.07	28.50
GROUP 3\$	39.57	28.50 28.50
GROUP 4s	37.94	28.50
GROUP 5\$	36.23	28.50

### Hazardous waste removal work:

Work on a state or federally designated hazardous waste site, where the worker is in direct contact with hazardous material, and when personal protective equipment is required for respiratory, skin and eye protection: 20% per hour additional.

### PAID HOLIDAYS:

New Year's Day, Washington's Birthday observed, Memorial Day, Independence Day, Labor Day, Presidential Election Day, Veteran's Day, Thanksgiving Day and Christmas Day; provided 1) that the worker works three of the preceding five work days before the holiday; or, the work day before the holiday and the work day after the holiday; and, 2) that the worker works the work day before and the work day after the holiday.

### DEFINITION OF GROUPS:

### GROUP 1:

Backhoe, Including Backhoe Track; Boom; Concrete Paving Machine; Crane (all types, including overhead and straddle traveling type); Drill (down-the-hole drill, rotary drill, self-propelled hydraulic drill, self-powered drill); Elevating Grader; Excavator; Front End Loader (5 cu. yd. and over); Piledriver (length of boom, including length of leads, shall determine premium rate applicable)

### GROUP 2:

Backhoe Loader Combo; Concrete Pumper; Grader/Blade (Finish); Hoist; Hydraulic Crane, 10 Tons and under; Front End Loader (2 cu. yd. but less than 5 cu. yd.); Scraper; Side Boom

### GROUP 3:

Asphalt Spreader; Bulldozer; Compressor(2 or 3) (in Battery) (within 100 ft.); Crusher; Forklift; Front End Loader (1 cu. yd. and over but less than 2 cu. yd.); Lull; Mechanic; Paver, Asphalt; Roller, Blacktop; Tractor;

GROUP 4:

Broom; Compressor (Single); Farm Tractor; Front End Loader (under 1 cu. yd.); Roller, Grade; Pump

GROUP 5:

Oiler

780v0011 040 0011

IRON0011-010 08/01/2012

SOMERSET COUNTY (Northern Half of County)

:	Rates	Fringes
Ironworkers:		
Reinforcing\$ Structural, Ornamental,		39.40
Riggers	43.54	39.40

IRON0068-015 07/01/2013

Somerset County (Southern Half of County)

	Rates	Fringes		
IRONWORKER				
Reinforcing\$ Structural, Ornamental,	42.40	20.03		
Rigger\$	44.39	20.03		

Rates

LABO0172-009 09/01/2012

	•
Laborers:	
Common or General Laborer;	
Landscape Laborer, Power	
Tool Operator\$ 33.20	22.40
Pipelayer\$ 33.90	22.40

### Hazardous waste removal work:

Work on a state or federally designated hazardous waste site, where the worker is required to wear Level A, B or C personal protection: \$3.00 per hour additional.

Work on a state or federally designated hazardous waste site, where the worker is not required to wear Level A, B, or C personal protection: \$1.00 per hour additional.

### PAID HOLIDAYS:

New Year's Day, President's Day, Memorial Day, Independence Day, Labor Day, Presidential Election Day, Veteran's Day, Thanksgiving Day and Christmas Day; provided that the worker works three days for the same employer within a period of ten working days consisting of five working days before and five working days after the day upon which the holiday falls or is observed.

LABO0222-013 07/01/2012

Rates

Fringes

Fringes

LABORER MASON TENDER:		
Cement/Concrete	\$ 29.35	23.07
PAIN0711-023 05/01/2009		
	Rates	Fringes
Painters: Work on bridges (Major Bridges Designed for Commercial Navigation)	\$ 46 50	10.10
PAIN0711-024 11/01/2008		19.13
11110711 024 1170172008		
	Rates	Fringes
Painters:  New Construction  Brush and roller  Spray  Repaint work, on projects on which no major alterations occur.  Brush and roller  Spray  Steel  PLUM0009-027 07/01/2013  SOMERSET COUNTY (Township of Medical County of Medical County)	\$ 37.92 \$ 35.81 \$ 26.67 \$ 29.34 \$ 27.74	
	Rates	Fringes
PLUMBER/PIPEFITTERService and Repair	\$ 43.43	33.20 18.53
PLUM0024-020 05/01/2013		
SOMERSET COUNTY (does not inclu Borough of Rocky Hill)	ade the towns	hip of Montgomery;
	Rates	Fringes
PLUMBER		29.74
PLUM0475-019 05/01/2013		
SOMERSET COUNTY (does not inclu Borough of Rocky Hill)	de the Townsi	nip of Montgomery;
	Rates	Fringes
PIPEFITTER		23.04
TEAM0469-008 05/01/2011		
	Rates	Fringes

Truck drivers:

Dump Truck; Flatbed Truck...\$ 34.75 24.385 Off the Road Truck; Pickup Truck.....\$ 34.60 24.385

### Hazardous waste removal work:

Work on a state or federally designated hazardous waste site, where the worker is in direct contact with hazardous material, and when personal protective equipment is required for respiratory, skin and eye protection: \$3.00 per hour additional.

Work on a state or federally designated hazardous waste site, in a zone requiring Level A personal protection for any workers other than the truck driver: \$3.00 per hour additional.

Work on a state or federally designated hazardous waste site where the worker is not working in a zone requiring Level A, B or C personal protection: \$1.00 per hour additional.

### PAID HOLIDAYS:

New Year's Day, President's Day, Decoration Day, Independence Day, Labor Day, Presidential Election Day, Veteran's Day, Thanksgiving Day and Christmas Day.

### VACATION PAY CREDIT:

Workers working or receiving pay for 80 days within a year receive one week paid vacation (48 hours); 125 days receive two weeks paid vacation (96 hours); 145 days receive 15 days paid vacation (120 hours); 15 years seniority and 145 days receive 4 weeks paid vacation (160 hours).

WELDERS - Receive rate prescribed for craft performing operation to which welding is incidental.

Unlisted classifications needed for work not included within the scope of the classifications listed may be added after award only as provided in the labor standards contract clauses (29CFR 5.5 (a) (1) (ii)).

The body of each wage determination lists the classification and wage rates that have been found to be prevailing for the cited type(s) of construction in the area covered by the wage determination. The classifications are listed in alphabetical order of "identifiers" that indicate whether the particular rate is union or non-union.

Union Identifiers

An identifier enclosed in dotted lines beginning with

file:///C:/Users/cprcrxd/AppData/Local/Temp/Low/VOWMYPXO.htm

characters other than "SU" denotes that the union classification and rate have found to be prevailing for that classification. Example: PLUM0198-005 07/01/2011. The first four letters, PLUM, indicate the international union and the four-digit number, 0198, that follows indicates the local union number or district council number where applicable, i.e., Plumbers Local 0198. The next number, 005 in the example, is an internal number used in processing the wage determination. The date, 07/01/2011, following these characters is the effective date of the most current negotiated rate/collective bargaining agreement which would be July 1, 2011 in the above example.

Union prevailing wage rates will be updated to reflect any changes in the collective bargaining agreements governing the rates.

0000/9999: weighted union wage rates will be published annually each  ${\it January}$ .

### Non-Union Identifiers

Classifications listed under an "SU" identifier were derived from survey data by computing average rates and are not union rates; however, the data used in computing these rates may include both union and non-union data. Example: SULA2004-007 5/13/2010. SU indicates the rates are not union majority rates, LA indicates the State of Louisiana; 2004 is the year of the survey; and 007 is an internal number used in producing the wage determination. A 1993 or later date, 5/13/2010, indicates the classifications and rates under that identifier were issued as a General Wage Determination on that date.

Survey wage rates will remain in effect and will not change until a new survey is conducted.

### WAGE DETERMINATION APPEALS PROCESS

- 1.) Has there been an initial decision in the matter? This can be:
- \* an existing published wage determination
- \* a survey underlying a wage determination
- \* a Wage and Hour Division letter setting forth a position on a wage determination matter
- \* a conformance (additional classification and rate) ruling

On survey related matters, initial contact, including requests for summaries of surveys, should be with the Wage and Hour Regional Office for the area in which the survey was conducted because those Regional Offices have responsibility for the Davis-Bacon survey program. If the response from this initial contact is not satisfactory, then the process described in 2.) and 3.) should be followed.

With regard to any other matter not yet ripe for the formal process described here, initial contact should be with the Branch of Construction Wage Determinations. Write to:

Branch of Construction Wage Determinations Wage and Hour Division U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

2.) If the answer to the question in 1.) is yes, then an interested party (those affected by the action) can request review and reconsideration from the Wage and Hour Administrator (See 29 CFR Part 1.8 and 29 CFR Part 7). Write to:

Wage and Hour Administrator U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

The request should be accompanied by a full statement of the interested party's position and by any information (wage payment data, project description, area practice material, etc.) that the requestor considers relevant to the issue.

3.) If the decision of the Administrator is not favorable, an interested party may appeal directly to the Administrative Review Board (formerly the Wage Appeals Board). Write to:

Administrative Review Board U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

4.) All decisions by the Administrative Review Board are final.

END OF GENERAL DECISION

## **GENERAL PROVISIONS FOR CONSTRUCTION**

# **Table of Contents**

	EKAL	REQUIREMENTS	
	1.1	ABBREVIATIONS AND DEFINITIONS	1
	1.2	ORDER OF PRECEDENCE	
	1.3	REFERENCED STANDARDS	
	1.4	INTENTION	8
	1.5	INTERPRETATION	
	1.6	PERMITS, LAWS, AND REGULATIONS	
	1.7	FEDERAL, STATE, AND LOCAL TAXES	
	1.8	INDEPENDENT CONTRACTOR	
	1.9	ASSIGNING AND SUBCONTRACTING CONTRACT	
	1.10	CONTRACTING OFFICER'S AUTHORITY	13
	1.11	PROJECT MANAGER'S AUTHORITY	14
	1.12	CONSTRUCTION MANAGER'S AUTHORITY	14
	1.13	INSPECTOR'S AUTHORITY	14
	1.14	NOTICE AND COMMUNICATION	14
	1.15	DISPUTES	15
	1.16	GOVERNING LAW; CONSENT TO JURISDICTION AND WAIVER OF TRIAL BY JURY	
	1.17	AUDIT AND INSPECTION OF RECORDS	16
2.CON	TRAC	TIME, SUSPENSION AND TERMINATION	16
	2.1	TIME OF COMPLETION - DELAY - LIQUIDATED DAMAGES	16
	2.2	NO DAMAGES FOR DELAY	
	2.3	SUSPENSION OF WORK	
	2.4	TERMINATION FOR CONVENIENCE	
	2.5	TERMINATION FOR CAUSE	
		CHANGES	26
	3.1	CHANGE ORDERS	
	3.2	CHANGE ORDER PRICING	
	3.3	TIME AND MATERIAL (T&M) CHANGE ORDER RECORDS	
	3.4	CONTRACTOR INITIATED CHANGE ORDERS	
	3.5	AUDIT OF CHANGE ORDERS	
	3.6	SUPPLEMENTAL CONSTRUCTION COSTS	
4.PRO	TECTION	ON AND CONTROL OF PREMISES	38
	4.1	RESPONSIBILITY FOR WORK	20
	1 2		
	4.2	USE OF PREMISES	39
	4.3	MAINTENANCE AND CLEANING OF PREMISES	39 41
	4.3 4.4	MAINTENANCE AND CLEANING OF PREMISESFIRE PREVENTION ON PREMISES	39 41 42
	4.3 4.4 4.5	MAINTENANCE AND CLEANING OF PREMISES	39 41 42
	4.3 4.4 4.5 4.6	MAINTENANCE AND CLEANING OF PREMISES	39 41 42 42
	4.3 4.4 4.5 4.6 4.7	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES	
	4.3 4.4 4.5 4.6 4.7 4.8	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS	39 41 42 44 44
	4.3 4.4 4.5 4.6 4.7 4.8 4.9	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS  MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC	
	4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS  MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC  MAINTENANCE OF MARINE TRAFFIC	
	4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS  MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC  MAINTENANCE OF MARINE TRAFFIC  MAINTENANCE OF RAILROAD TRAFFIC	
	4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11 4.12	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS  MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC  MAINTENANCE OF MARINE TRAFFIC  MAINTENANCE OF RAILROAD TRAFFIC  WORK FURNISHED BY OTHERS	
	4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11 4.12 4.13	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS  MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC  MAINTENANCE OF MARINE TRAFFIC  MAINTENANCE OF RAILROAD TRAFFIC  WORK FURNISHED BY OTHERS  ARTS IN TRANSIT	
	4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11 4.12 4.13	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS  MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC  MAINTENANCE OF MARINE TRAFFIC  MAINTENANCE OF RAILROAD TRAFFIC  WORK FURNISHED BY OTHERS  ARTS IN TRANSIT	
5.MOB	4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11 4.12 4.13 BILIZAT 5.1	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS  MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC  MAINTENANCE OF MARINE TRAFFIC  MAINTENANCE OF RAILROAD TRAFFIC  WORK FURNISHED BY OTHERS  ARTS IN TRANSIT  ION AND TEMPORARY FACILITIES  MOBILIZATION	
5.MOB	4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11 4.12 4.13 BILIZAT 5.1 5.2	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS  MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC  MAINTENANCE OF MARINE TRAFFIC  MAINTENANCE OF RAILROAD TRAFFIC  WORK FURNISHED BY OTHERS  ARTS IN TRANSIT  ION AND TEMPORARY FACILITIES  MOBILIZATION  FIELD OFFICE AND SANITARY FACILITIES	
5.MOB	4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11 4.12 4.13 BILIZAT 5.1	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS  MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC  MAINTENANCE OF MARINE TRAFFIC  MAINTENANCE OF RAILROAD TRAFFIC  WORK FURNISHED BY OTHERS  ARTS IN TRANSIT  ION AND TEMPORARY FACILITIES  MOBILIZATION  FIELD OFFICE AND SANITARY FACILITIES  CONSTRUCTION SIGN	
5. <b>M</b> OB	4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11 4.12 4.13 BILIZAT 5.1 5.2	MAINTENANCE AND CLEANING OF PREMISES  FIRE PREVENTION ON PREMISES  PROTECTION AGAINST DAMAGE  PROTECTION OF PRIVATE PROPERTY  PROTECTION OF PUBLIC UTILITIES  PROTECTION OF EXISTING MONUMENTS  MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC  MAINTENANCE OF MARINE TRAFFIC  MAINTENANCE OF RAILROAD TRAFFIC  WORK FURNISHED BY OTHERS  ARTS IN TRANSIT  ION AND TEMPORARY FACILITIES  MOBILIZATION  FIELD OFFICE AND SANITARY FACILITIES	

## **GENERAL PROVISIONS FOR CONSTRUCTION**

## **Table of Contents**

5.6	TEMPORARY HEAT	
5.7	TEMPORARY PARTITIONS, ENCLOSURES, GLAZING BREAKAGE AND CLEANING	61
5.8	TEMPORARY, INTERMEDIATE AND HIDDEN WORK	
5.9	DEMOBILIZATION	62
6.PROJECT	ADMINISTRATION AND DOCUMENT CONTROL	62
6.1	PROJECT MEETINGS	
6.2	CONSTRUCTION PROJECT SCHEDULE	
6.3	CONTRACT DRAWINGS AND SPECIFICATIONS	
6.4	GENERAL REQUIREMENTS FOR SUBMITTALS	
6.5	SHOP AND WORKING DRAWING SUBMITTALS	
6.6	SAMPLES SUBMITTALS	
6.7	PRODUCT AND SUBMITTALS	
6.8	VALUE ENGINEERING SUBMITTALS	
6.9	PROGRESS PHOTOGRAPH SUBMITTALS	
6.10 6.11	REPORTS, RECORDS AND DATA SUBMITTALSAS-BUILT DRAWINGS AND QUANTITIES SUBMITTAL	
	ASSURANCE AND QUALITY CONTROL	81
7.1	QUALITY ASSURANCE	
7.2	MATERIAL - WORKMANSHIP - LABOR	
7.3	INSPECTION OF WORK	
7.4	PLANT INSPECTIONINSTALLED EQUIPMENT TESTING AND TRAINING	
7.5 7.6	LABORATORY TESTING AND INSPECTION	
7.0 7.7	CERTIFICATION OF COMPLIANCE	
7.7	NON-CONFORMING WORK AND MATERIALS	
7.9	WARRANTY AGAINST DEFECTIVE WORK	
8.EXCAVAT	TION AND DIFFERING SITE CONDITIONS	91
8.1	UNCLASSIFIED EXCAVATION	91
8.2	MEASUREMENT OF PAY LIMITS FOR EXCAVATION	
8.3	SOIL BORINGS	92
8.4	DIFFERING, LATENT OR UNUSUAL SITE CONDITIONS	
8.5	ARCHEOLOGICAL FINDINGS	
9.INDEMNI	FICATION AND LIABILITY	93
9.1	INDEMNIFICATION; RISKS ASSUMED BY THE CONTRACTOR	
9.2	INSURANCE	
9.2.5		
9.3 9.4	LIMITATIONS OF LIABILITY NO THIRD PARTY BENEFICIARIES	
9. <del>4</del> 9.5	PERSONAL LIABILITY OF PUBLIC OFFICIALS	
9.6	INTELLECTUAL PROPERTY	
9.7	ENVIRONMENTAL COMPLIANCE AND LIABILITY	
	_ REQUIREMENTS	100
10.1	COVENANT AGAINST CONTINGENT FEES	
10.1	OFFICIALS NOT TO BENEFIT	
10.2	GRATUITIES	
10.4	NJ TRANSIT CODE OF ETHICS	
11. SOCIAL	AND ECONOMIC REQUIREMENTS	103
11.1	NEW JERSEY PREVAILING WAGE ACT	103
11.2	EQUAL OPPORTUNITY	

## **GENERAL PROVISIONS FOR CONSTRUCTION**

## **Table of Contents**

11.3	UTILIZATION OF DISADVANTAGED BUSINESS ENTERPRISES	111
12.MEASUR	EMENT AND PAYMENT	111
12.1	SCOPE OF PAYMENT	111
12.2	QUANTITIES: MEASUREMENT AND PAYMENT	
12.3	PARTIAL PAYMENTS	
12.4	MATERIALS PAYMENTS	114
12.5	RETAINAGE	115
12.6	SUBCONTRACTOR PAYMENTS AND RETAINAGE	
12.7	PAYMENTS WITHHELD	
12.8	FINAL PAYMENT	
12.9	SETTING OFF TAX ARREARS AGAINST SUMS OWED	119
13.ACCEPT	ANCE AND COMPLETION	119
13.1	PARTIAL ACCEPTANCE	
13.2	SUBSTANTIAL COMPLETION	
13.3	COMPLETION AND FINAL ACCEPTANCE OF THE WORK	121
14.FEDERAL	PROVISIONS	122
14.1	EMPLOYEE PROTECTIONS-CONSTRUCTION ACTIVITIES	
14.2	CLEAN WATER AND CLEAN AIR ACTS	
14.3	BUY AMERICA	
14.4	FLY AMERICA REQUIREMENTS	131
14.5	CARGO PREFERENCE - USE OF U.S. FLAG VESSELS	
14.6	ENERGY CONSERVATION	132
14.7	CONTRACT WORK HOURS AND SAFETY STANDARDS	
14.8	TITLE VI COMPLIANCE	_
14.9 14.10	PATENT RIGHTSRCRA REQUIREMENTS: BUILDING INSULATION PRODUCTS AND FLY ASH IN CONCRETE	
14.10	FEDERAL EEO PROVISIONS FOR CONSTRUCTION CONTRACTS	
14.11	LIMITATIONS ON LOBBYING	
14.13	SEISMIC SAFETY	
14.14	EXCLUSIONARY OR DISCRIMINATORY SPECIFICATIONS	
14.15	NO FEDERAL GOVERNMENT OBLIGATIONS TO THIRD PARTIES	
14.16	FALSE OR FRAUDULENT STATEMENTS AND CLAIMS	
14.17	FEDERAL AID, LAWS AND REGULATIONS	
	APPENDIX A – SUBCONTRACTOR EVALUATION DATA FORM	
	APPENDIX B – CLOSEOUT FORMS	
	APPENDIX C – CONTRACT EXECUTION FORM	
	APPENDIX D – GOVERNING RULES FOR WORKING WITHIN THE RAILROAD RIGHT OF WAY	1
	APPENDIX E – AGREEMENT BETWEEN CONTRACTOR AND ARTIST	

#### **GENERAL PROVISIONS FOR CONSTRUCTION**

#### 1. GENERAL REQUIREMENTS

#### 1.1 ABBREVIATIONS AND DEFINITIONS

#### 1.1.1 ABBREVIATIONS:

AAR Association of American Railroads

AASHTO American Association of State Highway and Transportation Officials

AMTRAK National Railroad Passenger Corporation

ANSI American National Standards Institute (Formerly USASI, United States of American

Standards Institute, and ASA, American Standard Association)

AREMA American Railway Engineering and Maintenance Association (Formerly AREA, American

Railway Engineering Association)

<u>ASTM</u> American Society for Testing and Materials

 CONRAIL
 Consolidated Rail Corporation

 CFR
 Code of Federal Regulations

CPM Critical Path Method

<u>DCA</u> New Jersey Department of Community Affairs

FTA Federal Transit Administration

MSDS Material Safety Data Sheet(s)

NFPA National Fire Protection Association

New Jersey Administrative Code

New Jersey Department of Transportation

New Jersey Uniform Fire Code

NJ TRANSIT

New Jersey Transit Corporation

New Jersey Uniform Construction Code

NPC Notice of Proposed Change

OSHA Occupational Safety and Health Administration, and Occupational Safety and Health Act

of 1970, and amendments thereto; United States Department of Labor

RFI Request for Information

<u>USDOT</u> United States Department of Transportation

#### 1.1.2 DEFINITIONS:

Acceptance of the Work: The act of the Contracting Officer, or the Contracting Officer's authorized representative, by which NJ TRANSIT accepts the work performed as partial or complete performance of the requirements of the Contract on the part of the Contractor.

<u>Addenda</u>: Written interpretations, clarifications, and revisions to any of the Contract Documents issued by NJ TRANSIT before the bid opening.

<u>Advertisement</u>: The public announcement, as required by law, inviting bids for work to be performed, materials to be furnished, or both.

<u>Affiliate:</u> Any entity that (1) directly, indirectly, or constructively controls another entity, (2) is directly, indirectly, or constructively controlled by another entity, or (3) is subject to the control of a common entity. An entity controls another entity if it owns, directly or individually, more than 50% of the ownership in that entity.

Allowance: An item included in the Bidder's Proposal for the exclusive use and benefit of NJ TRANSIT.

<u>Approved Equal</u>: A product which, in the opinion of the Engineer, complies with the technical specification requirements in form, fit, functionality, maintainability, useful life and quality of performance.

<u>Approval</u>: The written endorsement, sanction, or authorization by NJ TRANSIT of a proposal, plan, procedure, action, document, report, specification, design, or any part thereof, undertaken, promulgated, or developed by the Contractor in accordance with the indicated requirements of the Contract.

<u>Architect</u>: The term Architect shall be used interchangeably with Engineer and shall have the same meaning as Engineer.

Award: The approval by NJ TRANSIT of the bid and written notice of same to the tenderer of said bid.

<u>Bid</u>: The proposal or offer of the Bidder for the Work, when made out and submitted on the prescribed Bid Forms, properly signed, dated, and guaranteed, and which includes the schedule of Bid Items.

<u>Bid Security</u>: The bid bond, cashier's check, certified check or irrevocable letter of credit accompanying the bid submitted by the Bidder guaranteeing that the Bidder will enter into a Contract with NJ TRANSIT for the performance of the Work indicated and will provide acceptable bonds and insurance if the Contract is awarded to it.

<u>Bidder</u>: An individual, firm, partnership, corporation, joint venture or combination thereof submitting a proposal for the Work contemplated as a single business entity and acting directly or through a duly authorized representative.

<u>Bidder's Proposal</u>: The prescribed Bid Form which contains the schedule of Bid Items and the Bidder's declaration to perform the work for the prices bid.

<u>Business Organization</u>: An individual, partnership, association, joint stock company, trust, corporation, or other legal business entity or successor thereof.

<u>Business Registration</u>: A business registration certificate issued by the Department of the Treasury or such other form or verification that a Contractor or Subcontractor is registered with the Department of Treasury.

Calendar Day: Each and every day shown on the calendar.

<u>Change Order</u>: A written order issued by the Contracting Officer of NJ TRANSIT to the Contractor delineating changes in the Contract Documents in conjunction with one or more Notice(s) of Proposed Change and establishing, if appropriate, an equitable adjustment to the Contract Price or Contract Time for the work affected by the change(s).

<u>Construction Equipment</u>: All machinery and equipment, together with the necessary supplies for upkeep and maintenance, and also tools and apparatus necessary for the proper and acceptable completion of the Work.

<u>Construction Manager</u>: NJ TRANSIT's Assistant Executive Director (AED) of Capital Planning and Programs acting, directly or through the AED's authorized representatives at the Construction Site, within

the scope of duties assigned to him or her.

Construction Site: The geographical area of the property at which the contract work is performed, as

authorized and identified by NJ TRANSIT.

Consultant: A firm or individual contracting with and providing professional services to NJ TRANSIT.

<u>Contract</u>: The written agreement executed by the Contracting Officer of NJ TRANSIT and the Contractor which covers the performance of the Work, the furnishing of labor, materials, tools and equipment and the basis of payment, and which incorporates the various Contract Documents. The Contract represents the entire and integrated agreement between the parties and supersedes all prior negotiations, representations or agreements

either written or oral.

<u>Contract Documents</u>: The Contract Documents include: The General Provisions for Construction, Special Provisions, Technical Provisions, Contract Drawings, Contract Forms, Documents referenced by these

specifications, the Notice(s) to Proceed, Directive Letters, Change Orders, and Addenda.

<u>Contract Drawings</u>: The official plans, sections, elevations and details in the Contract Documents or amendments thereto and supplemental drawings approved by NJ TRANSIT which show the locations,

character, dimensions and details of the Work to be performed.

<u>Contract Item (Pay Item)</u>: A specifically described product or unit of work for which a price is listed in the Contract or the Bidder's Proposal.

<u>Contractor Initiated Change Order Request (CICOR)</u>: A request submitted to NJ TRANSIT by the Contractor specifying a proposed addition, deduction, or change to the Contract Documents.

<u>Contract Limit</u>: The lines shown on the drawings beyond which no construction work shall be performed unless otherwise noted on the drawings or in the specifications.

<u>Contract Time</u>: The number of calendar days or specified date set forth in the Contract for substantial completion of the Work, including amendments authorized by Change Order thereto; also referred to as Time of Completion. Contract Time begins on the date of the Notice to Proceed which is day one (1) of the Contract Time.

<u>Contracting Officer</u>: NJ TRANSIT's Chief of Procurement & Support Services acting, directly or through the Chief of Procurement & Support Services authorized representative(s), within the scope of duties assigned to him or her.

<u>Contractor</u>: The individual, firm, partnerships, corporation, joint venture, or any combination thereof, who, as an independent contractor, has entered into a Contract with NJ TRANSIT, as party or parties of the second part and who is referred to throughout the Contract Documents by singular number and non-specific gender.

<u>Days</u>: Days as used in the Specifications shall be understood to mean calendar days unless otherwise designated.

<u>Directive Letter</u>: A letter issued by the Contracting Officer, or the Contracting Officer's duly authorized designee, directing the Contractor to proceed with added, deleted or changed work.

**Drawings**: See Contract Drawings.

<u>Engineer</u>: The authorized representative of NJ TRANSIT providing design and engineering services for the Project. This may be either internal NJ TRANSIT staff or an outside consultant.

<u>Execution of the Contract</u>: The signing of the Contract by the authorized representative of NJ TRANSIT and the authorized representative of the Contractor.

Executive Director: The chief executive officer of NJ TRANSIT appointed in accordance with N.J.S.A. 27:25-1 et seq. and NJ TRANSIT's by-laws, or his/her designee.

<u>Extra Work</u>: An item of work not provided for in the Contract as awarded but found essential to the acceptable completion of the Contract within its intended scope.

<u>Failure</u>: Inability of a component, equipment, or system to function or perform in accordance with the indicated requirements.

<u>Federal Transit Administration</u>: Formerly known as the Urban Mass Transportation Administration, an agency within the United States Department of Transportation.

<u>Final Inspection</u>: The inspection conducted by the Project Manager to determine if the Work, or any substantial portion thereof, declared by the Contractor to be completed, has been satisfactorily completed in accordance with the requirements of the Contract and Contract Documents, and properly conditioned for final acceptance by the Contracting Officer.

<u>Form, Fit, and Function</u>: The technical documentation describing the physical and functional characteristics of an item as an entity, but not including any characteristics of the elements making up the item.

<u>General Provisions</u>: The general conditions of the Contract set forth in the Contract Documents as the General Provisions for Construction.

Government: The Government of the United States of America.

Holidays: Specific days on which NJ TRANSIT is not open for business.

<u>Inspector</u>: The Construction Manager's authorized representative assigned to observe Contract performance and materials furnished by the Contractor.

<u>Installed Equipment</u>: Equipment incorporated into the Work under this Contract.

<u>Installer</u>: The Contractor or entity (person or firm) engaged by the Contractor or its subcontractor at any tier for the performance of a particular unit of work at the project site, including installation, erection, application and similar required operations.

<u>Invitation for Bids</u>: The set of documents issued by NJ TRANSIT's Procurement Department for the purpose of soliciting bids.

<u>Joint Venture</u>: A legal association of Contractors formed for the purpose of bidding and executing a Contract as a single business entity.

<u>Logo</u>: An abbreviation for logotype, trademark or symbol.

Materials: Substances specified for use in the Work and its appurtenances.

<u>NJ TRANSIT</u>: A public instrumentality of the State of New Jersey established by the New Jersey Public Transportation Act, <u>N.J.S.A.</u> 27:25-1 <u>et seq.</u> (L. 1979, c.150).

<u>Notice of Proposed Change (NPC)</u>: A notice issued to the Contractor by NJ TRANSIT specifying a proposed addition, deduction or change to the Contract Documents. A Notice of Proposed Change is not an order to incorporate revisions into the Work.

<u>Notice(s) to Proceed</u>: The written directive from the Contracting Officer to the Contractor authorizing the Contractor to begin the prosecution of the Work in the initial Contract or for added, deleted or changed Work.

Owner: The New Jersey Transit Corporation (NJ TRANSIT).

<u>Performance and Payment Bonds</u>: The approved form of security furnished by the Contractor and it surety guaranteeing complete performance of the Contract in conformity with the Contract Documents and the payment of legal obligations pertaining to the construction of the Contract.

<u>Project</u>: The specific Work required to be performed by the Contractor as described in the Contract Documents.

<u>Project Data</u>: Illustrations, standard schedules, performance charts, instructions, brochures, diagrams, performance and test data, test procedures, existing drawings, operational manuals, maintenance manuals, spare parts lists and data, descriptive literature, catalogs, catalog cuts, and other information furnished by the Contractor to illustrate a material, product or equipment and to support its test, installation, operation or maintenance.

<u>Project Manager</u>: NJ TRANSIT's Assistant Executive Director (AED) of Capital Planning and Programs or Deputy General Manager (DGM) of Infrastructure Design acting, directly or through the their authorized representative(s), within the scope of the particular duties assigned to him or her.

Proposal: See Bid.

<u>Purchase Order</u>: The written document generated by the Procurement Department and issued to the Contractor for billing purposes.

Quotation: The Contractor's written response to the Notice of Proposed Change.

<u>Reliability</u>: The probability that equipment or a system will perform its intended functions without failure and within design parameters under specified operating conditions for which designed and for a specific period of time.

Remaining Work: Any and all Work remaining to be performed after Substantial Completion, including but not limited to, punch list work, which in the opinion of the Engineer is necessary for full conformance to the Contract.

<u>Samples</u>: Physical examples which illustrate materials, equipment or workmanship and establish standards by which the acceptability of the Work will be judged.

<u>Shop Drawings</u>: Original drawings, submitted to the Construction Manager by or through the Contractor, subcontractor or any lower tier subcontractor pursuant to the Work, including, but not limited to: stress sheets, working drawings, diagrams, illustrations, schedules, performance charts, brochures, erection plans, falsework plans, framework plans, cofferdam plans, bending diagrams for reinforcing steel, reinforced concrete formwork drawings, or other supplementary plans or similar data which are prepared by the Contractor or a subcontractor, manufacturer, supplier or distributor, and which the Contractor is required to submit to the Construction Manager for review and approval by the Engineer.

<u>Special Provisions</u>: Special Provisions are supplementary specified clauses setting forth conditions or requirements peculiar to the Work taking precedence over the General Provisions.

<u>Specifications</u>: The directions, provisions, and requirements contained or referred to in the Contract Documents, together with all duly authorized written agreements and directives made or to be made pertaining to the manner of performing the Work, or to the quantities and qualities of materials to be furnished or the quantities and qualities of work to be performed under the Contract.

State: State of New Jersey.

<u>Subcontractor</u>: Any individual, partnership, firm or corporation who undertakes for the Contractor, with the prior approval of the Project Manager, the partial or total manufacture or installation, or both, of one or more items of work under the terms of the Contract, or who performs other services for the Contractor as required to fulfill the terms of this Contract, by virtue of an agreement with the Contractor.

<u>Substantial Completion</u>: The point at which the Project Manager determines that the performance of work or portion thereof under the Contract, except Remaining Work, has been completed: provided that the Project Manager has determined, in the Project Manager's sole discretion, that (1) the project is safe and convenient for use and occupancy by the public and NJ TRANSIT employees and visitors; and (2) the project and facilities resulting therefrom may be used for the purposes for which they were intended.

<u>Substitution</u>: A product which, in the opinion of the Engineer, does not comply in form, fit, functionality, maintainability, useful life and quality of performance, with the technical specification requirements yet is proposed by the Contractor for incorporation into the Work in lieu of a specified product.

<u>Superintendent</u>: The Contractor's designated representative at the Construction Site responsible for the supervision and coordination of the Work.

<u>Supplier</u>: Any individual, partnership, firm or corporation which provides materials or equipment but not labor or services to the Contractor in partial fulfillment of the Scope of the Work of the Contract and who is responsible to the Contractor by virtue of an agreement.

<u>Surety</u>: The corporate body authorized to issue surety bonds in New Jersey which is bound with and for the Contractor for the guarantee of its proposal and the satisfactory performance of the Work by the Contractor, and the prompt payment in full for materials, labor, equipment, rentals, utility services, and debts and obligations, as provided in the bonds.

<u>Total Bid</u>: The total monetary amount of the bidder's proposal in dollars for performance of the Work of the entire Contract.

<u>Technical Data</u>: Written, typed or printed material prepared by the Contractor, subcontractors, vendors and suppliers or others, whether or not copyrighted, and submitted by the Contractor in response to the requirements set forth in these specifications. Technical data include, but are not limited to, product data, shop drawings, pictorial representations, reports, schedules, studies, and assessments.

<u>Technical Provisions</u>: The Technical Provisions are specific requirements setting forth the materials and methods required to accomplish the Work.

<u>U.S. Department of Transportation</u>: The Secretary of the U.S. Department of Transportation, and other persons who may at the time be acting in the capacity of the Secretary, or an authorized representative or

other persons otherwise authorized to perform the functions to be performed by USDOT, including representatives of the Federal Transit Administration (FTA).

<u>Work</u>: Work shall mean the furnishing of labor, materials, equipment, and other incidentals necessary or convenient to the successful completion of the Project and the carrying out of the duties and obligations imposed by the Contract, including alterations, amendments or extensions.

### 1.2 ORDER OF PRECEDENCE

The sequence of precedence pertaining to the Contract Documents is as follows:

Contract Change Orders and Directive Letters

Addenda, if any

Special Provisions, if any

**General Provisions** 

**Technical Provisions** 

Details

Figured Dimensions

Scaled Dimensions

**Drawings** 

Referenced Standards

### 1.3 REFERENCED STANDARDS

- 1.3.1 All materials, equipment and workmanship, specified by the number, symbol or title of a referenced standard shall comply with the latest edition or revision thereof and all amendments and supplements thereto in effect on the date of the opening bids, except where a particular issue is indicated in the reference. Where products or workmanship are specified by an association, trade or federal standard, Contractor shall comply with the requirements of that standard, except where exceeded by the requirements of the Contract plans and specifications or regulatory authorities. Where referenced standards are in conflict, the more stringent shall apply. In case of conflicting requirements between Referenced Standards and the Contract Documents, the Contract Documents shall govern.
- 1.3.2 Where the New Jersey Uniform Construction Code, its adopted subcodes and their referenced standards, and the other regulations described in Article 1.6 are silent regarding the construction requirements of the work specified herein, installation shall be in accordance with the most current versions of the following:
  - National Fire Protection Association (NFPA) <u>Standard 130 Standard for Fixed Guideway</u>
     Transit Systems;
  - 2.) American Railway Engineering and Maintenance Association (AREMA) Manual of Railway Engineering;
  - 3.) Industrial Risk Insurers (IRI) Standards;
  - 4.) NJ TRANSIT Standards as identified in the Contract Documents;

- 5.) NJ TRANSIT Graphic Standards Manual; and
- 6.) NJDOT Standard Specification for Road Bridge Construction (except Section 100).

#### 1.4 INTENTION

- 1.4.1 The Contract Documents are intended to provide for and comprise everything necessary to the proper and complete finishing of the work in every part notwithstanding that each and every item necessary may not be shown on drawings or mentioned in the specifications. The Contractor shall abide by and comply with the true intent and meaning of all the Contract Documents taken as a whole, and shall not avail itself of any apparent error or omission, should any exist.
- 1.4.2 The Contractor's execution of this Contract constitutes its certification that it satisfied itself, through personal inspection, as to the correctness of information given which may affect the quantity, size and quality of materials required for a satisfactorily completed Contract, whether or not such information is indicated on the Drawings or within the Specifications. The Contractor confirms that it checked and verified conditions outside of the Contract Limit Lines to determine whether or not any conflict exists between elevations or other data shown on the drawings and existing elevations or other data outside of the Contract Limit Lines.
- 1.4.3 Work that may be called for in the Specifications and not shown on the Drawings or shown on the Drawings and not called for in the Specifications, shall be executed and furnished by the Contractor as if described in both. Should work or materials be required which are necessary for the proper carrying out of the intent thereof, the Contractor shall understand same to be implied and required and it shall perform such work and furnish such materials as fully as if they were particularly delineated or described.
- 1.4.4 The Contractor shall not at any time after the execution of this Contract set up any claims whatsoever based upon insufficient data, patent ambiguities, inconsistencies or confusion in the Contract Documents or incorrectly assumed conditions, nor shall it claim any misunderstanding with regard to the nature, conditions or character of the work to be done under the Contract, except as provided for under Article 8.4- DIFFERING, LATENT OR UNUSUAL SITE CONDITIONS.
- 1.4.5 Should any error or discrepancy appear or should any doubt exist or any dispute arise as to the true intent and meaning of the Contract Documents, or should any portion of same be obscure or capable of more than one method of construction, the Contractor shall immediately apply in writing to the Construction Manager for the correction or explanation thereof and, in case of dispute, the Contracting Officer's decision shall be final.
- 1.4.6 Each and every provision of law and clause required by law to be inserted in this Contract shall be deemed to be inserted herein and the Contract shall be read and enforced as though it were included herein, and if through mistake or otherwise such provision is not inserted, or is not correctly inserted, then

upon the application of either party the Contract shall forthwith be physically amended to make such insertion or correction.

- 1.4.7 NJ TRANSIT may act directly through its own employees or may act indirectly through retained independent third party Contractors and consultants and their employees. The Contractor shall be advised by the Contracting Officer of those persons authorized to act on NJ TRANSIT's behalf.
- 1.4.8 A waiver on the part of NJ TRANSIT of any breach of any part of the Contract shall not be held to be a waiver of any other or subsequent breach.

#### 1.5 INTERPRETATION

- 1.5.1 The headings of the articles herein are for convenience of reference only and shall have no bearing on their interpretation. Whenever in these Contract Documents the following terms are used, the intent and meaning shall be interpreted as defined herein. All of the terms used herein are treated throughout the Contract as if each were the singular number and non-specific gender.
- 1.5.2 Wherever in the Contract Documents the words "directed", "required", "ordered", "designated", "prescribed", or words of like import are used, it shall be understood that the "direction", "requirement", "order", "designation", or "prescription" of NJ TRANSIT is intended and similarly the words "approved", "acceptable", "satisfactory", or words of like import, shall mean "approved by", or "acceptable to", or "satisfactory to" NJ TRANSIT unless otherwise expressly stated. Where "as shown", "as indicated", "as detailed", or words of similar import are used, it shall be understood that the reference is made to the Specifications and Drawings accompanying the Contract unless stated otherwise.
- 1.5.3 References to all Articles or Sections include all Sub-articles or subsections under the Article referenced, and references to all Sub-articles include all sub-Sub-articles.

### 1.6 PERMITS, LAWS, AND REGULATIONS

- 1.6.1 NJ TRANSIT in entering into this Contract does not waive its sovereign immunity except as provided in the New Jersey Contractual Liability Act, N.J.S.A. 59:13-1 et seq. The terms and conditions of the Contract are not intended to, and shall not be deemed to, expand the waiver of sovereign immunity as set forth in that Act.
- The work under this Contract is subject to Federal, State, and local laws, ordinances, codes, and regulations, including those of railroad companies and utility companies, for work on the Project. Except as otherwise provided in the Contract Documents, the Contractor shall obtain and pay for necessary permits and make necessary arrangements with the authorities having jurisdiction. Where the Contractor's compliance with Federal, State or local laws, ordinances, codes or regulations may or will

conflict with the Contract, the Contractor shall notify the Construction Manager in writing for appropriate action.

- 1.6.3 NJ TRANSIT will supply the Contractor with applications for construction permits, as required by the New Jersey Department of Community Affairs (DCA). The Contractor shall return within seven (7) calendar days permit applications signed and sealed to NJ TRANSIT. NJ TRANSIT will submit the Contractor's applications along with all fees required, to the DCA. The fees for all DCA applications shall be the responsibility of NJ TRANSIT.
  - (a) Except as otherwise provided in the Contract Documents, all work shall be performed in accordance with the New Jersey Uniform Construction Code (NJUCC), <u>N.J.A.C.</u> 5:23-1 <u>et seq.</u>, its adopted subcodes, and their referenced standards. Specific requirements of the NJUCC and its adopted subcodes shall supersede any conflicting requirements in other documents referenced herein.
  - (b) The Contractor shall be responsible for assisting NJ TRANSIT in the preparation and submission of any new or revised plans and specifications required due to DCA requests for clarification, changes in scope of work, etc., necessary for obtaining amended or additional permits or approvals for work for which NJ TRANSIT has already secured construction permits.
  - (c) The Contractor shall be responsible for requesting utility location markups and obtaining permits and approvals from utility authorities and other authorities having jurisdiction.
- 1.6.4 It shall be the responsibility of the Contractor to keep itself fully informed concerning all requirements of law, including but not limited to, all Federal, State, and local laws, ordinances, codes, and regulations which in any manner affect the Project, and of all such orders and decrees of bodies or tribunals having jurisdiction or authority over the same. The Contractor shall at all times observe, and shall cause its subcontractors, agents, and employees to observe, such requirements of law, and shall defend, indemnify, save, and hold harmless the State of New Jersey and NJ TRANSIT and all of their officers, agents, and employees against claims and liabilities arising from or based upon the violation of such requirement of law whether by the Contractor or its agents, subcontractors or employees.
- 1.6.5 Plumbing, drainage, and sewage disposal work shall conform with applicable Federal, State and all relevant utility regulator environmental laws and regulations.
- 1.6.6 Electrical and applicable mechanical materials and systems shall bear the label of the Underwriters' Laboratories (UL) and shall be listed in the publication issued by the UL. Other materials or systems bearing labels of other testing laboratories may be accepted upon written approval of the Engineer. Rules of the National Fire Protection Association shall be followed explicitly unless deviations are agreed to in writing by the Engineer or otherwise modified by these specifications.

- 1.6.7 Work shall be conducted in accordance with the State of New Jersey Department of Labor and Industry Construction Safety Code, as promulgated by the Commissioner of Labor and Industry under the authority of the Construction Safety Act, N.J.S.A. 34:5-166 to 34:5-181, and applicable provisions of the Occupational Safety and Health Administration (OSHA), 29 CFR 1910, Rules and Regulations. Where the Construction Safety Code refers to the designation of General Contractor for enforcing compliance with the Code, such designation shall be intended to refer to the Contractor.
- 1.6.8 Construction work shall be performed in accordance with the requirements of the New Jersey Uniform Fire Code (NJUFC), N.J.A.C. 5:71-1 et seq.
- 1.6.9 The Contractor shall immediately notify the Construction Manager should any review or inspection by any regulatory agency or official result in the issuance of a citation or notice of violation of any permit, regulation, statute or other governmental rule.

### 1.7 FEDERAL, STATE, AND LOCAL TAXES

Except as may be otherwise provided in the Contract, each Contract Item shall include all applicable taxes and duties. N.J.S.A. 54:32B-9 provides that any sale or service to NJ TRANSIT is not subject to the sales and use taxes imposed under the Sales and Use Tax Act. N.J.S.A. 54:32B-8.22 provides that sales made to Contractors or repairmen of materials, supplies or services for exclusive use in erecting structures, or building on, or otherwise improving, altering or repairing real property of NJ TRANSIT are exempt from the tax on retail sales imposed by the Sales and Use Tax Act. The exemption provided under N.J.S.A. 54:32B-8.22 is conditioned on the person seeking such exemption qualifying therefore pursuant to the rules and regulations and upon the forms prescribed by the New Jersey Division of Taxation. The required form, "Contractor's Exemption Purchase Certificate" (Form No. ST-13), can be obtained by writing or calling the New Jersey Division of Taxation, Tax Information Service (TIS), West State and Willow Streets, Trenton, New Jersey 08625.

NJ TRANSIT is exempt from Federal Excise Taxes. The State of New Jersey's Federal Excise Tax exemption number is 22-75-0050K.

The Contractor and any Subcontractor providing goods or performing services under the Contract, and each of their Affiliates, shall, during the term of the Contract, collect and remit to the Director of the Division of Taxation in the Department of the Treasury the use tax due pursuant to the "Sales and Use Tax Act", P.L 1966, c. 30 (N.J.S.A. 54:32B-1 et seq.) on all their sales of tangible personal property delivered into the State.

#### 1.8 INDEPENDENT CONTRACTOR

The relationship of the Contractor to NJ TRANSIT is that of an independent Contractor, and said Contractor, in accordance with its status as an independent Contractor, covenants and

agrees that it will conduct itself consistent with such status, that it will neither conduct itself as nor claim to be an officer or employee of NJ TRANSIT or the State by reason thereof. The Contractor will not, by reason thereof, make any claim, demand or application to or for any right or privilege applicable to an officer or employee of NJ TRANSIT or the State, including, but not limited to, worker's compensation coverage, unemployment insurance benefits, social security coverage or retirement membership or credit.

### 1.9 ASSIGNING AND SUBCONTRACTING CONTRACT

1.9.1 The Contractor shall not transfer, assign or otherwise dispose of the Contract or Contract funds, due or to become due, or claims of any nature it has against NJ TRANSIT to any other party except upon the express written approval of NJ TRANSIT, which approval NJ TRANSIT shall not unreasonably withhold. Application for subcontracting any part or parts of the work shall be made by the Contractor and shall be addressed to NJ TRANSIT through the Construction Manager. The Contractor shall perform with its own organization and with the assistance of workmen under its immediate superintendence, work amounting to not less than 20 percent of the Contract Price, exclusive of Bid Items for Insurance, Performance/Payment Bonds, Mobilization and Allowances.

1.9.2 At the Preconstruction Meeting, the Contractor shall submit to the Construction Manager a list of, and Subcontractor Evaluation Data Forms (Appendix A) for, all subcontractors to be used on the Project within the first two months. The list shall identify the subcontractor's name, nature of work and value of work to be performed, and date work is to start. Thereafter, the Contractor shall provide the Construction Manager with at least 15 days notice before engaging additional or alternative subcontractors. Each proposed subcontractor shall submit a completed Subcontractor Evaluation Data Form. NJ TRANSIT reserves the right to reject any subcontractor with unsatisfactory qualifications, experience or record of performance. No Contract shall be entered into by the Contractor with a subcontractor before its name has been approved in writing by the Construction Manager.

All Subcontractors are required to comply with the Public Works Contractor Registration Act N.J.S.A. 34:11-56.48 et seq. as amended. Subcontractors shall not engage in the performance of any public work as defined in N.J.S.A. 34:11-56.26 unless the Subcontractor is registered pursuant to the Act. As part of the post award Subcontractor approval process detailed above, certificates of registration shall be submitted to NJ TRANSIT's Construction Manager for each Subcontractor.

In accordance with N.J.S.A. 52:32-44 all Subcontractors must obtain a Business Registration Certificate prior to performing work or providing goods and/or services on the contract. A Subcontractor shall provide a copy of its business registration to the Contractor who shall forward it to NJ TRANSIT. No contract with a Subcontractor shall be entered into by the Contractor unless the Subcontractor first provides proof of valid business registration.

The Contractor shall maintain a list of the names of all Subcontractors and their current addresses, updated as necessary during the course of the Contract performance. In conjunction with the Subcontractor approval process detailed above, the Contractor shall submit a complete and accurate list of the Subcontractors to NJ TRANSIT before for final payment is made.

- 1.9.3 The Contractor agrees that it is as fully responsible to NJ TRANSIT for the acts and omissions of its subcontractors and of persons either directly or indirectly employed by them as it is for the acts and omissions of persons directly employed by the Contractor. The Contractor hereby gives its assurance that, when minimum wage rates are specified, they shall apply to labor performed on all work subcontracted, assigned or otherwise disposed of in any way.
- 1.9.4 The Contractor agrees to bind every subcontractor to and have every subcontractor agree to be bound by the terms of the Contract Documents, as far as applicable to its work. Each Subcontractor Evaluation Data Form shall contain the subcontractor's certification that it has reviewed and is familiar with the Contract Documents in their entirety and that the subcontractor shall comply with all Contract requirements.
- 1.9.5 It is understood, however, that any consent of NJ TRANSIT for the subcontracting of any of the work of the Contract in no way relieves the Contractor from its full obligations under the Contract. Approval by NJ TRANSIT of a subcontractor, including a material supplier, does not relieve the Contractor or its subcontractor of the responsibility of complying with the Contract Documents. Further, the approval of a subcontractor does not imply approval of any material, installed equipment, substitution or additional approved equal.
- 1.9.6 When the Contractor proposes to subcontract a portion of an item which involves a breakdown of the unit of measurement of that item, it shall submit a breakdown of cost showing the value of the portion of the item to be subcontracted in relation to the value of the whole item, which shall be subject to the approval of NJ TRANSIT.
- 1.9.7 Nothing contained in the Contract Documents shall create any contractual relation between any subcontractor and NJ TRANSIT. The consent to subcontract any part of the work shall not be construed to be an approval of the said subcontract or of any of its terms, but shall operate only as an approval of the Contractor's request for the making of a subcontract between the Contractor and its chosen subcontractor.

#### 1.10 CONTRACTING OFFICER'S AUTHORITY

The Contracting Officer is the final interpreter of the terms and conditions of the Contract and the judge of its performance. The Contracting Officer has the sole authority to execute the Contract; order additions, deductions and changes to the Work; render final decisions on disputes; suspend or terminate the Work, or portions thereof; and accept or reject Contractor performance.

#### 1.11 PROJECT MANAGER'S AUTHORITY

In connection with the Work to be performed under the Contract, the Project Manager shall be the technical representative of the Contracting Officer. The Project Manager shall review and approve the Contractor's invoices, after certification by the Construction Manager, and the Project Manager shall authorize payments. When duly authorized in writing by the Contracting Officer, the Project Manager shall have the authority to give approval of and order changes in the Work. The limits of this authorization shall be specified in writing by the Contracting Officer and provided to the Contractor. The Project Manager shall monitor the performance of the Contract, shall determine Substantial Completion of the Work and recommend to the Contracting Officer its Final Acceptance.

#### 1.12 CONSTRUCTION MANAGER'S AUTHORITY

The Construction Manager shall apply the Contract Documents, and shall judge the quantity, quality, fitness and acceptability of all parts of the work. In addition, the Construction Manager shall determine whether specific items of construction work, methods or materials are properly specified in the Contract Documents. In the event of a dispute, the Contractor shall proceed diligently with the performance of the work in accordance with the Construction Manager's determination, pending the decision of the Contracting Officer. The Construction Manager shall certify Contractor's invoices for work performed and materials delivered to the site, and shall be given access to the work for inspection at all times. The Construction Manager shall not have authority to give approval of nor order changes in work which alter the terms or conditions of the Contract, nor which involve additional cost or Contract Time unless duly authorized in writing by the Contracting Officer. The Construction Manager may, however, make recommendations to the Contracting Officer for such changes, whether or not costs are to be revised and the Contracting Officer may act, at the Contracting Officer's discretion, on the basis of the Construction Manager's recommendations. The Construction Manager has the authority to reject unsuitable material or suspend work that is being improperly performed.

#### 1.13 INSPECTOR'S AUTHORITY

Inspectors are authorized to inspect work done and materials furnished. Such inspection may extend to all or any part of the work and to the preparation, fabrication or manufacture of the materials to be used. Inspectors are not authorized to alter or waive the provisions of the Contract. Inspectors are not authorized to issue instructions contrary to the Contract Documents, or to act as foremen for the Contractor; however, they shall have the authority to reject work or materials until a question at issue can be referred to and decided by the Construction Manager.

## 1.14 NOTICE AND COMMUNICATION

1.14.1 Written notice shall be deemed to have been duly served and received by NJ TRANSIT and the Contractor if: (A) Delivered in person to the intended individual, to a member of the firm, an officer of the corporation or their authorized representative on the Work, or (B) Sent by certified mail, or other mail or courier service, with delivery receipt, to the last business address known to the individual who gives the notice, or (C) Sent

by telefacsimile or electronic mail, followed by a hard copy to the last business address known to the individual who gives the notice.

- 1.14.2 Communications to the Contractor shall be transmitted through and coordinated by the Contractor's authorized representative(s), as designated at the Pre-Construction Conference.
- 1.14.3 Communications to NJ TRANSIT shall be transmitted through and coordinated by the Construction Manager. Correspondence shall be addressed to the appropriate NJ TRANSIT authorized representative as set forth in the Contract Documents or as otherwise designated by NJ TRANSIT at the Pre-Construction Conference. Upon written notice to the Contractor, NJ TRANSIT may change its authorized designee(s) at any time.
- 1.14.4 All communications to NJ TRANSIT shall be clearly marked with NJ TRANSIT's Contract Number and Contract Title

#### 1.15 DISPUTES

Disputes arising in the performance of this Contract, which are not resolved by agreement of the parties shall be decided in writing by the authorized representative of the Contracting Officer. This decision shall be final and conclusive unless within ten (10) calendar days from the date of receipt of its copy, the Contractor mails or otherwise furnishes a written appeal to the Contracting Officer. In connection with any such appeal, the Contractor shall be afforded an opportunity to be heard and to offer evidence in support of its position. Pending final decision of a dispute hereunder, the Contractor shall proceed diligently with performance of the Contract in accordance with the decision of the authorized representative of the Contracting Officer. The decision of the Contracting Officer shall be binding upon the Contractor and the Contractor shall abide by the decision. The "New Jersey Contractual Liability Act", N.J.S.A. 59:13-1 et seq., shall govern any action which may be brought by the Contractor as a result of NJ TRANSIT's decision.

- 1.16 GOVERNING LAW; CONSENT TO JURISDICTION AND WAIVER OF TRIAL BY JURY
- 1.16.1 This Contract shall be construed in accordance with and shall be governed by the Constitution and laws of the State of New Jersey.
- 1.16.2 Except as otherwise provided in this Contract, disputes and claims arising under this Contract which are not disposed of by mutual agreement shall be governed by the laws of the State of New Jersey as they may from time to time be in effect. The Contractor, by entering into this Contract, consents and submits to the exclusive jurisdiction of the Courts of the State of New Jersey over any action at law, suit in equity or other proceeding that may arise out of this Contract, and the Contractor agrees, during the period of performance and of Warranty, to maintain within the State of New Jersey an agent to accept service of legal process on its behalf. Notwithstanding the language of N.J.S.A. 59:13-4, the Contractor expressly waives trial by jury

on any and all disputes and claims arising out of this Contract whether by or against the Contractor, NJ TRANSIT or any other person or entity.

### 1.17 AUDIT AND INSPECTION OF RECORDS

The Contractor shall retain all Contract records and permit the authorized representatives of NJ TRANSIT, the State of New Jersey, the USDOT and the Comptroller General of the United States to inspect and audit all data and records of the Contractor relating to products, transactions or services provided under its performance and the performance of its subcontractors under this Contract from the date of the Contract and for five (5) years after final payment under this Contract has been made.

The Contractor further agrees to include in all its subcontracts hereunder a provision to the effect that the subcontractor agrees that it will keep all Contract records and that NJ TRANSIT, the Department of Transportation and the Comptroller General of the United States or any of their duly authorized representatives shall, until the expiration of five (5) years after final payment under the subcontract, have access to and the right to examine any directly pertinent books, documents, papers, and records of such subcontractor, involving transactions related to the subcontractor. The term "subcontractor" as used in this clause excludes (1) purchase orders not exceeding \$10,000 and (2) subcontracts or purchase orders for public utility services at rates established for uniform applicability to the general public.

Pursuant to N.J.S.A. 52:15C-14(d), the Contractor shall maintain all documentation related to products, transactions or services under this contract for a period of five (5) years from the date of final payment. Such records shall be made available to the New Jersey Office of State Comptroller upon request.

The periods of access and examination described above, for records which relate to (1) appeals under the "Disputes" clause of the Contract, (2) litigation or the settlement of claims arising out of the performance of this Contract, or (3) costs and expenses of this Contract as to which exception has been taken by the Comptroller General, USDOT, NJ TRANSIT or the State of New Jersey or any of their duly authorized representatives, shall continue until such appeals, litigation, claims, or exceptions have been disposed of.

## 2. CONTRACT TIME, SUSPENSION AND TERMINATION

- 2.1 TIME OF COMPLETION DELAY LIQUIDATED DAMAGES
- 2.1.1 The Contractor and NJ TRANSIT recognize that delay in completion of the project will result in damage to the State of New Jersey in terms of the effect of the delay in the use of the Project upon the public convenience and economic development of the State of New Jersey, and will also result in additional cost to NJ TRANSIT for engineering, inspection, and administration of the Contract. Because some of this damage is difficult or impossible to estimate, the parties agree that if the Contractor fails to substantially complete the project or any interim milestones within the time stated in the Contract, or within such further time as may have been granted in accordance with the provisions of the Contract, or fails to complete Remaining Work, other than Punch List Work (as described in Article 13.2- SUBSTANTIAL COMPLETION) within the period fixed by NJ TRANSIT, the Contractor shall pay NJ TRANSIT liquidated damages, in accordance with

the amount set forth in the Special Provisions for each day that it is in default on time to complete the work. The days in default shall be the number of calendar days in default when the time for completion of the Project is specified on the basis of calendar days or a fixed date; and shall be the number of working days in default when the time for completion is specified on the basis of working days. NJ TRANSIT shall recover said damages by deducting the amount thereof from monies due or that may become due the Contractor, and if said monies be insufficient to cover said damages, then the Contractor or its Surety shall pay the amount due. This clause is expressly limited in purpose and effect to damages to NJ TRANSIT and the State of New Jersey as a consequence of the Contractor's failure to complete the project on time. It is not intended, and shall not be construed, to apply to any other claim for damages or disputes arising from the Contractor's performance on the Project, nor shall it limit the Contractor's duty to indemnify NJ TRANSIT for claims made by third parties.

2.1.2 The work embraced in this Contract including work authorized under Article 3.1 shall commence as soon as possible but not later than ten (10) days after receipt of a Notice to Proceed unless otherwise specified by NJ TRANSIT.

The Contractor agrees that said work shall be prosecuted regularly, diligently, and without interruption at such rate of progress as will insure substantial completion thereof within the time specified. It is expressly understood and agreed, by and between the Contractor and NJ TRANSIT, that the time for the completion of the work herein is a reasonable time for the completion of the same. It is further agreed that float, or slack time, is not for the exclusive use or benefit of either the Contractor or NJ TRANSIT, as further described in Article 6.2- Construction Project Schedule.

2.1.3 It is further agreed that time is of the essence of each and every portion of this Contract and of the Specifications wherein a definite and certain length of time is fixed for the performance of any act whatsoever, and where under the Contract additional time is allowed for the completion of any work, the new time limit fixed by such extension shall be of the essence of this Contract. The Contractor shall not be charged with liquidated damages when the delay in the completion of the work arises from both: a.) causes beyond the reasonable control and without any fault or negligence of the Contractor, its officers, employees, agents, servants, subcontractors, and suppliers, including but not restricted to, acts of God, or of the public enemy, acts of another Contractor in the performance of a Contract with NJ TRANSIT, fires, floods, epidemics, quarantine restrictions, labor disputes not determined by final judicial or administrative adjudication to have been caused or provoked by the illegal acts of the Contractor or one of its subcontractors or agents, freight embargoes, and unusually severe weather and, b.) such causes arise after the award of the Contract and neither were nor could have been anticipated by the Contractor by reasonable investigation before such award. The basis to define unusually severe weather will be the data showing high and low temperatures, precipitation and wind conditions in the geographic area of the Work for the previous twenty (20) years, as compiled by the recording station of the U.S. National Weather Service located nearest to the Work.

- Even though a cause of delay meets both conditions a.) and b.) above, an extension shall be granted only to the extent that (i) the completion of the affected work is actually and necessarily delayed, (ii) the effect of such cause could not be anticipated and avoided or mitigated by the exercise of all reasonable precautions, efforts and measures (including planning, scheduling and rescheduling) whether before or after the occurrence of the cause of delay, and (iii) the critical path of the project schedule is actually extended due to the delay beyond the required Contract completion date. Any reference in this Article to the Contractor shall be deemed to include materialmen, suppliers, and subcontractors, whether or not in privity of Contract with the Contractor, all of whom shall be considered as agents of the Contractor for the purposes of this Article. A delay resulting from a cause meeting all conditions in Sub-article 2.1.3 and this Sub-article 2.1.4 shall be deemed an Excusable Delay.
- 2.1.5 The period of any extension of time shall be only that which is necessary to make up the time actually lost as determined by NJ TRANSIT. NJ TRANSIT may defer all or part of its decision on an extension and any extension may be rescinded or shortened if it subsequently is found that the delay can or could have been overcome or reduced by the exercise of reasonable precautions, efforts and measures.
- 2.1.6 In case the Contractor shall be actually and necessarily delayed by reason of the failure of NJ TRANSIT to deliver to the Contractor access to the Project Site or any materials or facilities to be furnished by NJ TRANSIT which are actually needed for use in the work, or by any act or omission on the part of NJ TRANSIT, and such delay is recognized by NJ TRANSIT in writing, such delay shall also be deemed to be an Excusable Delay. The time for completion of the Project shall be extended by NJ TRANSIT by the amount of time of such delay as determined by NJ TRANSIT, but no allowance by way of damages of any kind or nature will be made for such failure.
- As a condition precedent to the granting of an extension of time, the Contractor shall give written notice to NJ TRANSIT within five (5) days after the time when the Contractor knows or should know of any cause which might under reasonably foreseeable circumstances result in delay for which it may claim an extension of time (including those causes for which NJ TRANSIT itself is responsible or of which NJ TRANSIT has knowledge), specifically stating in such notice that an extension is or may be claimed, identifying such cause and describing, as fully as practicable at that time, the nature and expected duration of the delay, including justification, and its effect on the completion of that part of the work identified in the notice. Since the possible necessity for an extension of time may materially alter the scheduling plans, and other actions of NJ TRANSIT and since, with sufficient notice, NJ TRANSIT may, if it should so elect, attempt to mitigate the effect of the delay for which an extension of time might be claimed, and since mere oral notice may cause dispute as to the existence or substance thereof, the giving of written notice as above required shall be of the essence of the Contractor's obligations hereunder. Failure to give the written notice within the aforementioned five (5) day period shall deprive the Contractor of any right to an extension of time, except to the extent that NJ TRANSIT may determine otherwise in its sole discretion.

It shall in all cases be presumed that no extension, or further extension of time, is due unless the Contractor shall affirmatively demonstrate the extent thereof to the reasonable satisfaction of NJ TRANSIT. To this end, the Contractor shall maintain adequate records supporting any claim for an extension of time, and in the absence of such records, this presumption shall be deemed conclusive.

- 2.1.8 In regard to an injunction or interference of public authority which may delay or impact the Project, the Contractor shall give NJ TRANSIT a copy of the injunction or other orders and of the papers upon which the same shall have been granted.
- 2.1.9 Within a reasonable time after receipt of a written notice requesting an extension of time NJ TRANSIT will advise the Contractor if such notice is adequate, or if further information is required. Failure of NJ TRANSIT to furnish the Contractor with the foregoing advisement shall not, however, be deemed to waive NJ TRANSIT's right to deny an extension of time. Within a reasonable time after NJ TRANSIT, at its sole discretion, has determined that the Contractor has provided sufficient information for NJ TRANSIT to decide on a request for an extension of time, NJ TRANSIT shall issue a determination on that request. In the event that NJ TRANSIT was unable to decide on a request for an extension, NJ TRANSIT may issue at the completion of the Project, in conjunction with issuing its Final Certificate of Payment, a final determination on the Contractor's request or requests for an extension of time.
- 2.1.10 Only the actual delay or impact to the Contractor necessarily resulting from the causes above-mentioned, as determined by NJ TRANSIT, shall be considered for an extension of time. In case the Contractor shall be delayed or impacted at any time or for any period by two or more of the causes above-mentioned, the Contractor shall not be entitled to a separate extension for each one of the causes but only one period of extension shall be granted for the period of concurrent delay. In case the Contractor shall be actually and necessarily delayed by one or more of the causes above-mentioned in the performance of any portion of the Project, the extension of time to be granted to the Contractor shall be only for such portion of the Project. The Contractor shall not be entitled by reason of such delay to an extension of time for the completion of the remainder of the Project. If the Contractor shall be so delayed as to a portion of the Project, it shall nevertheless proceed continuously and diligently with the prosecution of the remainder of the Project. No demand by the Contractor that NJ TRANSIT determine any matter of extension of time for the completion of the Project or any part thereof will be of any effect whatsoever unless the same be made in writing and duly served upon NJ TRANSIT prior to the issuance of the Final Certificate of Payment as provided for in Article 12.8, FINAL PAYMENT.
- 2.1.11 Delay to or impacts upon the Contractor's performance arising out of any request of the Contractor to change the order of furnishing working drawings as provided elsewhere in this Contract or arising out of any changes made or requested by the Contractor in any matters shown or indicated on the Contract Drawings

will not be cause for an extension of time, and all additional costs to the Contractor, incidental to such request or change, shall be borne by the Contractor.

- 2.1.12 The permitting of the Contractor to go on and finish the Project or any part thereof after the time fixed for completion or after the date to which the time for completion may have been extended or the making of payments to the Contractor after any such period shall not operate as a waiver on the part of NJ TRANSIT of any rights under this Contract, including but not limited to declaring the Contractor in default.
- 2.1.13 The determination of NJ TRANSIT as to any matter of extension of time for completion of the Project or any part thereof shall be binding and conclusive upon the Contractor.

#### 2.2 NO DAMAGES FOR DELAY

- 2.2.1 The Contractor expressly waives the right to make any claim against NJ TRANSIT for damages or additional compensation for any delay to or impact upon the performance of this Contract occasioned by any act or omission to act by NJ TRANSIT or its employees or any third parties for any reason whether or not enumerated in Article 2.1, and agrees that any such claim shall be fully compensated for by an extension of time to complete performance of the work as provided in Article 2.1.
- 2.2.2 Except as provided in Subarticle 2.2.4, he Contractor alone hereby specifically assumes the risk of all delays, obstruction, or interference of any kind, duration or cause whatsoever, whether or not within the contemplation of the parties or foreseeable.
- 2.2.3 The Contractor shall have no right to rescind or terminate this Contract, and Contractor shall have no cause of action under any theory of unjust enrichment, quasi-contract, quantum meruit or additional risk by reason of any delay, obstruction, or interference of any kind or duration whatsoever.
- 2.2.4 Nothing in these General Provisions is intended to limit Contractor's remedy for delayed performance caused by NJ TRANSIT's negligence, bad faith, active interference or other tortious conduct, to the extent such limitation is determined by a court of law to be void and unenforceable under N.J.S.A. 2A:58B-3.

## 2.3 SUSPENSION OF WORK

- 2.3.1 If the Contracting Officer deems it advisable, the Contracting Officer may notify the Contractor in writing to suspend work on one or more occasions on all or any part of the Project, for a period not to exceed 90 days in the aggregate. After notification(s) is delivered to the Contractor, the Contractor shall do no work where so suspended until it has received written notice from the Contracting Officer to resume work.
- 2.3.2 When work is suspended as provided above, payments for the completed parts of the work will be made as provided and a suitable extension of time for completing the suspended work will be granted where appropriate. Should any single suspension be for a period greater than 45 days and the Contractor incurs

unavoidable extended field overhead costs as a result of the suspension, the Contractor shall notify the Contracting Officer within 30 days of the completion of the suspension period and provide a detailed accounting of such extended field overhead costs. No profit markup will be allowed on extended field overhead costs. The Contracting Officer's determination as to the amount of compensation to be paid under this clause shall be final and conclusive. Under no circumstances shall any other compensation or allowance be made on account of such suspension. No payment will be made for work done by the Contractor on suspended work.

2.3.3 Within the period of ninety (90) days (or the lesser period specified) after a notice of suspension is delivered to the Contractor, or within any extension of that period to which the parties shall have agreed, NJ TRANSIT shall either: a.) cancel the notice of suspension, or b.) terminate the work covered by such suspension as provided in Article 2.4, TERMINATION FOR CONVENIENCE, or c.) negotiate reasonable compensation with the Contractor for a further period of suspension.

#### 2.4 TERMINATION FOR CONVENIENCE

- 2.4.1 NJ TRANSIT may terminate performance of work under this Contract in whole or, from time to time, in part if the Contracting Officer determines that a termination is in NJ TRANSIT's interest. The Contracting Officer shall terminate the work by delivering to the Contractor a Notice of Termination specifying the extent of termination and the effective date.
- 2.4.2 After receipt of a Notice of Termination, and except as directed by the Contracting Officer, the Contractor shall immediately proceed with the following obligations, regardless of any delay in determining or adjusting any amounts due under this clause:
  - (a) Stop work as specified in the notice.
  - (b) Place no further subcontracts or orders (referred to as subcontracts in this clause) for materials, services, or facilities, except as necessary to complete the continued portion of the Contract.
  - (c) Terminate all subcontracts to the extent they relate to the work terminated.
  - (d) Assign to NJ TRANSIT, as and if directed by the Contracting Officer, all right, title, and interest of the Contractor under the subcontracts terminated, in which case NJ TRANSIT shall have the right to settle or to pay any termination settlement proposal arising out of those terminations.
  - (e) With approval or ratification to the extent required by the Contracting Officer, settle all outstanding liabilities and termination settlement proposals arising from the termination of subcontracts; the approval or ratification will be final for purposes of this clause.
  - (f) As directed by the Contracting Officer, transfer title and deliver to NJ TRANSIT: (i) the fabricated or unfabricated parts, work in process, completed work, supplies, and other material produced or acquired for the work terminated, and (ii) the completed or partially completed plans, drawings, information, and other property that, if the Contract had been completed, would be required to be furnished to NJ TRANSIT.

- (g) Complete performance of the work not terminated.
- (h) Take any action that may be necessary, or that the Contracting Officer may direct, for the protection and preservation of the property related to this Contract that is in the possession of the Contractor and in which NJ TRANSIT has or may acquire an interest.
- (i) Use its best efforts to sell, as directed or authorized by the Contracting Officer, any property of the types referred to in Subparagraph (f) above; provided, however, that the Contractor (i) is not required to extend credit to any purchaser and (ii) may acquire the property under the conditions prescribed by, and at prices approved by the Contracting Officer. The proceeds of any transfer or disposition will be applied to reduce any payments to be made by NJ TRANSIT under this contract, credited to the price or cost of the work, or paid in any other manner directed by the Contracting Officer.
- 2.4.3 After expiration of the plant clearance period as defined in Subpart 45.6 of the Federal Acquisition Regulation, the Contractor may submit to the Contracting Officer a list, certified as to quantity and quality, of termination inventory not previously disposed of, excluding items authorized for disposition by the Contracting Officer. The Contractor may request NJ TRANSIT to remove those items or enter into an agreement for their storage. Within fifteen (15) days, NJ TRANSIT will accept title to those items and remove them or enter into a storage agreement. The Contracting Officer may verify the list upon removal of the items, or if stored, within forty-five (45) days from submission of the list, and shall correct the list, as necessary, before final settlement.
- 2.4.4 After termination, the Contractor shall submit a final termination settlement proposal to the Contracting Officer in the form and with the certification prescribed by the Contracting Officer. The Contractor shall submit the proposal promptly, but no later than one (1) year from the effective date of termination, unless extended in writing by the Contracting Officer upon written request of the Contractor within this one (1)-year period. However, if the Contracting Officer determines that the facts justify it, a termination settlement proposal may be received and acted on after one (1) year or any extension. If the Contractor fails to submit the proposal within the time allowed, the Contracting Officer may determine, on the basis of information available, the amount, if any, due the Contractor because of the termination and shall pay the amount determined.
- Subject to Sub-article 2.4.4 above, the Contractor and the Contracting Officer may agree upon the whole or any part of the amount to be paid because of the termination. However, the agreed amount, whether under this Sub-article or Sub-article 2.4.6 below, exclusive of costs shown in Sub-article 2.4.6(b) below, may not exceed the total contract price as reduced by: (i) the amount of payments previously made and, (ii) the contract price of work not terminated. The Contract shall be amended, and the Contractor paid the agreed amount. Sub-article 2.4.6 below shall not limit, restrict, or affect the amount that may be agreed upon to be paid under this Paragraph.

- 2.4.6 If the Contractor and Contracting Officer fail to agree on the whole amount to be paid the Contractor because of the termination of work, the Contracting Officer shall pay the Contractor the amounts determined as follows, but without duplication of any amounts agreed upon under Sub-article 2.4.5 above:
  - (a) For contract work performed before the effective date of termination, the total (without duplication of any items) of:
    - (i) The cost of this work;
    - (ii) The cost of settling and paying termination settlement proposals under terminated subcontracts that are properly chargeable to the terminated portion of the Contract if not included in subdivision (i) above; and
    - (iii) A sum, as profit on (i) above, determined by the Contracting Officer to be fair and reasonable; however, if it appears that the Contractor would have sustained a loss on the entire contract had it been completed, the Contracting Officer shall allow no profit under this subdivision (iii) and shall reduce the settlement to reflect the indicated rate of loss.
  - (b) The reasonable costs of settlement of the work terminated, including:
    - (i) Accounting, legal, clerical, and other expenses reasonably necessary for the preparation of termination settlement proposals and supporting data;
    - (ii) The termination and settlement of subcontracts (excluding the amounts of such settlements); and
    - (iii) Storage, transportation, and other costs incurred, reasonably necessary for the preservation, protection, or disposition of the termination inventory.
- 2.4.7 Except for normal spoilage, and except to the extent that NJ TRANSIT expressly assumed the risk of loss, the Contracting Officer shall exclude from the amounts payable to the Contractor under Sub-article 2.4.6 above, the fair value, as determined by the Contracting Officer, of property that is destroyed, lost, stolen, or damaged so as to become undeliverable to NJ TRANSIT or to a buyer.
- 2.4.8 The cost principles and procedures of Part 31 of the Federal Acquisition Regulation, in effect on the date of this contract, shall govern all costs claimed, agreed to, or determined under this clause.
- 2.4.9 The Contractor shall have the right of appeal, under the Disputes clause, from any determination made by the Contracting Officer under Sub-article 2.4.4, 2.4.6, or 2.4.11, except that if the Contractor failed to submit the termination settlement proposal within the time provided in Sub-article 2.4.4 or 2.4.11, and failed to request a time extension, there is no right of appeal. If the Contracting Officer has made a determination of the amount due under Sub-article 2.4.4, 2.4.6, or 2.4.11, NJ TRANSIT shall pay the Contractor (i) the amount determined by the Contracting Officer if there is no right of appeal or if no timely appeal has been taken, or (ii) the amount finally determined on an appeal.

- 2.4.10 In arriving at the amount due the Contractor under this clause, there shall be deducted:
  - (a) All unliquidated advance or other payments to the Contractor under the terminated portion of this contract;
  - (b) Any claim which NJ TRANSIT has against the Contractor under this contract; and
  - (c) The agreed price for, or the proceeds of sale of, materials, supplies, or other things acquired by the Contractor or sold under the provisions of this clause and not recovered by or credited to NJ TRANSIT.
- 2.4.11 If the termination is partial, the Contractor may file a proposal with the Contracting Officer for an equitable adjustment of the price(s) of the continued portion of the contract. The Contracting Officer shall make any equitable adjustment agreed upon. Any proposal by the Contractor for an equitable adjustment under this clause shall be requested within ninety (90) days from the effective date of termination unless extended in writing by the Contracting Officer.
- 2.4.12 NJ TRANSIT may, under the terms and conditions it prescribes, make partial payments and payments against costs incurred by the Contractor for the terminated portion of the contract, if the Contracting Officer believes the total of these payments will not exceed the amount to which the Contractor will be entitled. If the total payments exceed the amount finally determined to be due, the Contractor shall repay the excess to NJ TRANSIT upon demand, together with interest computed at the rate established by the Secretary of the Treasury under 50 USC App. 1215(b)(2). Interest shall be computed for the period from the date the excess payment is received by the Contractor to the date the excess is repaid. Interest shall not be charged on any excess payment due to a reduction in the Contractor's termination settlement proposal because of retention or other disposition of termination inventory until ten (10) days after the date of the retention or disposition, or a later date determined by the Contracting Officer because of the circumstances.
- 2.4.13 Unless otherwise provided in this contract, or by statute, the Contractor shall maintain all records and documents relating to the terminated portion of this Contract for three (3) years after final settlement. This includes all books and other evidence bearing on the Contractor's costs and expenses under this contract. The Contractor shall make these records and documents available to NJ TRANSIT, at the Contractor's office, at all reasonable times, without any direct charge. If approved by the Contracting Officer, photographs, microphotographs, or other authentic reproductions may be maintained instead of original records and documents.

#### 2.5 TERMINATION FOR CAUSE

2.5.1 In the event that any of the provisions of this Contract are violated by the Contractor, or any of its Subcontractors, the Contracting Officer may serve written notice upon the Contractor and Surety of NJ TRANSIT's intention to terminate the Contract for cause. The Notice of Intent to Terminate for Cause shall identify the causes for the proposed termination and demand the elimination of such causes.

- 2.5.2 If the Contractor or Surety, within a period of ten (10) days after such notice or within such additional time as may be granted by the Contracting Officer, does not proceed in accordance therewith to make satisfactory arrangements to eliminate the causes of the proposed termination, then the Contracting Officer may terminate the Contract for cause.
- 2.5.3 The Notice of Termination for Cause will terminate the Contractor's right to proceed with all items of work except as specified in the termination notice. The latter will include all work necessary to ensure the safety of the public, to properly secure existing work already constructed or partially constructed, and to secure the Job Site. The work specified in the notice shall be performed in accordance with the Contract Documents and may include items of work not in the original Contract. Unless otherwise specified in the notice, all insurance policies provided by the Contractor naming NJ TRANSIT and any other parties as additional insureds shall remain in full force and effect until issuance by NJ TRANSIT of a Final Certificate of Payment.
- 2.5.4 Payment for completed or partially completed items of Work shall be made in accordance with the Contract Documents. Payment for new items, if any, will be paid either at agreed prices or paid for by Time and Material methods described in Sub-article 3.2.7. No other costs or compensation will be allowed the Contractor.
- 2.5.5 When all work specified in the termination notice is completed to the satisfaction of NJ TRANSIT, the Contract shall terminate upon issuance by NJ TRANSIT of a Final Certificate of Payment.
- 2.5.6 Upon issuance of a Notice of Termination for Cause, the Surety shall have ten (10) days to advise NJ TRANSIT in writing that it intends to take over and complete the Project in accordance with the Contract terms and conditions, without any further conditions. If so notified, the Surety shall have 30 days from such notice to commence the work. Upon the Surety's failure to comply with either of the above, NJ TRANSIT may take over the work and prosecute the same to completion by contract with another contractor, or use whatever methods it deems necessary to complete the work, including completion of the Work by its own forces for the account and at the expense of the Contractor. NJ TRANSIT may take possession of and utilize in completing the work such materials, appliances and plants as may be on the site of the work and necessary therefor.
- 2.5.7 Whether the Contract Work is completed by NJ TRANSIT, either directly or through other contractors, or the Surety, the Contractor and its Surety shall be liable to NJ TRANSIT for excess costs incurred by NJ TRANSIT and other such damages arising out of the Termination for Cause including liquidated damages caused by the delay to the date of completion of the Project Work.
- 2.5.8 All such costs and damages incurred by NJ TRANSIT will be deducted from any monies due or that may become due the Contractor and Surety. If such costs and damages exceed the sum which is available,

then the Contractor and the Surety shall be liable and shall pay NJ TRANSIT within 30 days of the issuance of an invoice for the amount of such excess.

2.5.9 In terminating the Contract for cause NJ TRANSIT does not waive its right to sue the Contractor and/or Surety for any costs incurred or damages suffered by NJ TRANSIT as a result of the Contractor's default and termination.

2.5.10 If, after a Notice of Termination for Cause has been issued, it is determined for any reason that the provisions of the Contract were not violated by the Contractor, or any of its Subcontractors, or if the termination of the Contract for Cause pursuant to the provisions of this Article is found by a court to be legally improper, then the termination of the Contract for cause will be treated as if it had been a termination for convenience and such termination shall be compensated for in accordance with the provisions of Article 2.4.

#### 3. CONTRACT CHANGES

#### 3.1 CHANGE ORDERS

3.1.1 The Contracting Officer, at the Contracting Officer's sole discretion, may at any time during the progress of the work authorize additions, deductions, or changes to the Work as set forth below, and the Contract shall not be terminated or the surety released thereby. When changes in the work must be performed immediately, the Contracting Officer may issue a written directive to the Contractor detailing the changed work and the basis for determining compensation, and the Contractor will proceed immediately with the Work as directed, pending the execution of a formal Change Order.

If any such change causes an increase or decrease in the cost of the performance of any part of the Work or requires a change in the Contract Time, then a Change Order shall be issued incorporating the change. All Change Orders shall be priced in accordance with Article 3.2. The Change Order shall be a written order to the Contractor and shall describe the change with cost changes and changes to Contract Time. The Change Order shall be signed by the Contractor and returned to NJ TRANSIT. Upon receipt, the Change Order shall be countersigned by the Contracting Officer and shall then become a part of the Contract Documents.

In the event the Contractor and NJ TRANSIT cannot, for whatever reason, reach an agreement on cost changes or changes to Contract Time, the Contracting Officer shall issue a unilateral Change Order incorporating the change and the Contractor shall nonetheless proceed with the Work as directed therein. The unilateral Change Order shall then become a part of the Contract Documents.

3.1.2 The Construction Manager shall have the authority to order, in writing, minor changes in the work not involving an adjustment to the price of any items of work or an extension of time and not inconsistent with the intent of the Contract. Such changes shall be binding on NJ TRANSIT and the Contractor, and shall not

be the basis of increased compensation to the Contractor. Such work shall be executed under the conditions of the original Contract.

- 3.1.3 All additions, deductions or changes to the work as directed by Change Orders shall be executed under the conditions of the original Contract. The Change Order shall recite the additional time granted by NJ TRANSIT to perform the Work, if any. Except as specified in Sub-article 3.1.2 above, or in an emergency endangering life or property, no change shall be made unless pursuant to a written directive of the Contracting Officer or Change Order, and no claim for an addition to the Contract Price or time shall be valid unless so ordered.
- 3.1.4 Should the Contractor dispute the Construction Manager's interpretation of work specified in the Contract Documents and claim that work is Extra Work that will involve additional costs or Contract Time, the Contractor shall proceed with the work in accordance with the Construction Manager's interpretation. In such event, the Contractor shall follow the procedures and maintain the detailed cost records set forth in Article 3.3-T&M CHANGE ORDER RECORDS pending the resolution of the dispute. In all other cases, should the Contractor perform Extra Work without first obtaining a written directive or Change Order from the Contracting Officer's authorized representative, such action shall be construed by NJ TRANSIT as voluntary performance and as a waiver of any and all claims to extra payment and time therefor.
- 3.1.5 The time needed to perform Extra Work shall not be the basis of claims by the Contractor for extra costs of any nature whatsoever.

#### 3.2 CHANGE ORDER PRICING

3.2.1 Proposed additions, deductions and changes shall be defined in a Notice of Proposed Change (NPC) and issued to the Contractor. The Contractor's proposal for all proposed additions, deductions and changes to the work involving cost or Contract Time shall be submitted by the Contractor to the Construction Manager with copies to the Contracting Officer and Project Manager within 15 days, or such other time as the Construction Manager may direct, after the issuance of the Notice of Proposed Change. The Contractor's cost proposal shall be structured in accordance with the format(s) set forth below and shall comply with the pricing specifications set forth in this Article 3.2. NJ TRANSIT shall review the Contractor's proposal and, if necessary, meet with the Contractor to negotiate the proposal. Should the Contracting Officer require additional information, the Contractor will provide the requested information. The Contractor's costs for preparing, submitting, and negotiating proposals will not be paid separately and shall not be included in the proposals, but shall be considered paid for in the Contract Price.

Table 3.2.1: Change Order Cost Proposal Format				
I	II	III		
LABOR <sup>1</sup>	Cost Calculation	Summary of Costs		
Name Each Trade Classification	(Manhours) x (Labor Costs) =	А		

Table 3.2.1: Change Order Cost Proposal Format				
MATERIAL <sup>2</sup>				
Name Each Major Type of Material	(Quantity) x (Unit Costs) =	В		
EQUIPMENT <sup>3</sup>				
Name Each Major Piece of Equipment	(Time) x (Unit Costs) =	С		
SUBTOTAL	A + B + C =	D		
OVERHEAD <sup>4</sup>	(D) x (Overhead %) =	E		
PROFIT <sup>5</sup>	(D) x (Profit %) =	F		
TOTAL	D+E+F=	G		
Footnotes: <sup>1</sup> Labor Costs as specified in Sub-article 3.2.9.1				

<sup>2</sup>Material Costs as specified in Sub-article 3.2.9.2

<sup>5</sup>Profit Markups as specified in Article 3.2.9.6

- 3.2.2 Requests for extension of time for proposed Change Order work shall be included in the Contractor's proposal. Extensions of time will not be granted unless requested in accordance with the provisions of Subarticle 3.2.1.
- 3.2.3 Full documentation supporting all estimated and actual costs shall be furnished to the Construction Manager or Contracting Officer if such is requested. Documentation may consist of records such as actual payroll records and receipted bills for rentals and materials. All Change Orders shall be subject to audit by the Contracting Officer or the Contracting Officer's authorized representative.
- 3.2.4 All proposed and incurred change order costs shall as a minimum be allowable, allocable and reasonable in accordance with the Contract cost principles and procedures in Part 31 of the Federal Acquisition Regulations in effect on the date of the Contract. The Contracting Officer's determination on the allowability, allocability and reasonableness of incurred costs shall be final and conclusive.
- 3.2.5 The value of any change in the Contract shall be determined in accordance with the following pricing bases, listed in the order of priority of use: a) Unit Price, b) Lump Sum and c) Time and Material (T&M). Unit Prices shall govern if contained in the Bidder's Proposal for the applicable work. If no Unit Prices apply, then a Lump Sum pricing approach shall be used. If a Lump Sum cannot be determined, or agreement cannot be reached, or the Contracting Officer determines that work must be performed immediately, then the Contracting Officer will direct the Contractor to proceed on a T & M basis. Whenever the terms "labor", "materials", "equipment", "overhead" and "profit" are used herein with regard to change order cost and price proposals, they are used as these cost and price elements are defined in this Article 3.2.

<sup>&</sup>lt;sup>3</sup>Equipment Costs as specified in Sub-article 3.2.9.3

<sup>&</sup>lt;sup>4</sup>Overhead Markups as specified in Article 3.2.9.4

- 3.2.6 Unit Price Basis: Whenever unit prices govern, the Contractor's cost proposal shall identify the additional estimated quantities required for the work. The unit price included in the Contract, or subsequently agreed upon, shall be used to solely determine the increased or decreased cost of the work. The unit price shall be deemed to include all costs for labor, material, overhead and profit and the increase or decrease in the cost of the work shall be on a dollar for dollar basis.
- 3.2.7 Lump Sum Basis: When unit prices do not apply, the Contractor shall submit a detailed breakdown of labor, materials, and equipment. The Contractor shall add to this overhead and profit markups as specified in Subarticle 3.2.10. Cost proposals for labor and material shall be provided on the stationary of the parties that will be performing the work (subcontractors) and supplying material (suppliers).
- 3.2.8 Time and Material: The Contractor shall submit the same detailed breakdown of costs as set forth in Subarticle 3.2.7 for Lump Sum change orders. In addition, the Contractor shall submit a Guaranteed Maximum Price (GMP) which may be accepted or rejected by the Contracting Officer.
- 3.2.8.1 Time and Material with GMP: If the Guaranteed Maximum Price is accepted, the payment for such work shall not exceed the actual cost for labor, materials, and equipment. To this may be added overhead and profit mark-ups both as specified in Sub-article 3.2.10. However, in no event shall payment exceed the Guaranteed Maximum Price established by agreement between the Contractor and NJ TRANSIT.
- 3.2.8.2 Time and Material with Upset Price: If the Guaranteed Maximum Price submitted by the Contractor is rejected, NJ TRANSIT may direct the Contractor to proceed on a time and material basis with an Upset Price established by NJ TRANSIT. The Upset Price shall be the limit of authorization for performance of the Extra Work by the Contractor. At such time as the Contractor has expended eighty percent (80%) of the authorized limit, NJ TRANSIT may establish a new limit by revising the Upset Price. However, if NJ TRANSIT chooses not to establish a revised Upset Price, the Contractor shall cease the time and material work when the original Upset Price has been reached. The payment for such work shall include the actual cost for labor, materials and equipment. To this may be added overhead and profit mark-ups as specified in Sub-article 3.2.10.
- 3.2.8.3 Emergent Time and Material: Should the Contracting Officer determine that changed work must be performed immediately, and NJ TRANSIT determines that the Contractor has insufficient time to submit a detailed cost proposal in advance of performing the work, NJ TRANSIT may direct the Contractor to proceed on an emergent time and material basis with an Upset Price established by NJ TRANSIT. The terms of performance and payment shall be as set forth in Sub-article 3.2.8.2, except that profit markups shall be as specified in Sub-article 3.2.10.
- 3.2.9 NJ TRANSIT will consider for payment only the labor, material and equipment cost elements as specified herein in conjunction with any cost proposal submitted by the Contractor. These cost elements, individually

or together, shall serve as the cost basis upon which applicable markups for profit and overhead shall be applied, all as specified in Sub-article 3.2.10. These costs elements, together with the applicable markups for profit and overhead, shall constitute full compensation for all direct and indirect costs and shall be deemed to include all items of expense not specifically designated.

### 3.2.9.1 Labor Costs

- a) For necessary labor and foremen in direct charge of the specific operations, whether the employer is the Contractor, subcontractor or another, the Contractor shall receive the rate of wage (or scale) actually paid as shown in its certified payrolls for each and every hour that said labor and foremen are actually engaged in such work.
- b) The Contractor shall also receive the actual costs paid to, or in behalf of, workers by reason of health and welfare benefits, pension fund benefits or other benefits, when such amounts are required by collective bargaining agreements or other employment contracts, or the Contract prevailing wage determination, generally applicable to the classes of labor employed on the work.
- c) The Contractor shall receive the actual cost paid to applicable State and Federal agencies and insurance carriers for Worker's Compensation Insurance, Federal Insurance Compensation Act (FICA, Social Security), Unemployment Insurance and Contractor's General Liability and Worker's Disability.
- 3.2.9.2 <u>Material Costs</u>: Only materials furnished by the Contractor and necessarily used in the performance of the work will be paid for. The cost of such materials will be the cost to the purchaser, whether Contractor, subcontractor or other forces from the supplier thereof, together with transportation charges actually paid by the Contractor, except as the following are applicable.
  - (a) If a cash or trade discount by the actual supplier is offered or available to the purchaser, it shall be credited to NJ TRANSIT notwithstanding the fact that such discount may not have been taken.
  - (b) If materials are procured by the purchaser by any method which is not a direct purchase from and a direct billing by the actual supplier to such purchaser, the cost of such materials shall be deemed to be the price paid to the actual supplier as determined by NJ TRANSIT plus the actual costs, if any, incurred in the handling of such materials.
  - (c) If the materials are obtained from a supply or source owned wholly or in part by the purchaser, the cost of such materials shall not exceed the price paid by the purchaser for similar materials furnished from said source on Contract Items or the current wholesale price for such materials delivered to the job site, whichever price is lower.
  - (d) If the cost of such materials is, in the opinion of NJ TRANSIT, excessive, then the cost of such material shall be deemed to be the lowest current wholesale price at which such materials are available in the quantities concerned delivered to the job site, less any discounts as provided in Subparagraph (a) above.
  - (e) If the Contractor does not furnish satisfactory evidence of the cost of such materials from the actual supplier thereof, the cost shall then be determined in accordance with Paragraph (d) above.

NJ TRANSIT reserves the right to furnish such materials as it deems advisable, and the Contractor shall have no claims for costs and markups on such materials.

### 3.2.9.3 Equipment and Plant Rental Costs:

- (a) Contractor Owned Equipment and Plant The hourly rates for Contractor owned equipment and plant will be based on "Rental Rate Blue Book for Construction Equipment" or the "Rental Rate Blue Book for Older Construction Equipment" (both referred to hereafter as the "Blue Book"), published by Nielsen/DATAQUEST, Inc. of Palo Alto, California. The Blue Book shall be used in the following manner:
  - (1) The hourly rate will be determined by dividing the "monthly" rate set out in the Blue Book by 176. The "weekly," "hourly," and "daily" rates listed in the Blue Book will not be used.
  - (2) The number of hours to be paid for shall be the number of hours that the equipment or plant is actually used on a specific activity.
  - (3) The "current revisions" to the Blue Book will be used in establishing rates. The "current revision" applicable to specific Change Order work will be the "current revision" as of the first day of work performed on that Change Order work and that rate will apply throughout the period the Change Order work is being performed.
  - (4) Area adjustments will not be made. Equipment life adjustments will be made in accordance with the rate adjustment tables.
  - (5) Overtime shall be charged at the same rate indicated in (1), above.
  - (6) The "estimated operating costs per hour" shall be used for each hour that the equipment or plant is in operation on the Change Order work. No such costs shall apply to idle time regardless of the cause of the idleness.
  - (7) Idle time for equipment will not be paid for, except where the equipment has been held on the Project site on a standby basis at the request of NJ TRANSIT and, but for this request, would have left the Project site. Such payment will be made at one half (1/2) the rate established in (1), above.
  - (8) The rates as established above shall be deemed to include the cost of fuel, oil, lubrication, supplies, small tools, necessary attachments, repairs, overhaul, and maintenance of any kind, depreciation, storage, overhead, profits, insurance, costs of moving equipment or plant on to and away from the site, and incidentals (including labor and equipment).
  - (9) Operator costs shall be paid only as provided in Sub-article 3.2.9.1, "Labor," of this Article.

Equipment shall be in good operating condition and suitable for the work, in the opinion of the Construction Manager. Equipment used by the Contractor shall be specifically described and be of suitable size and capacity required for the work to be performed. In the event the Contractor elects to use equipment of a higher rental value than that suitable for the work, payment will be made at the rate applicable to the suitable equipment. The equipment actually used and the

suitable equipment paid for will be recorded as a part of the record for Change Order work. If there is a differential in the rate of pay of the operator of oversize or higher rate equipment, the rate paid for the operator will likewise be that for the suitable equipment.

In the event that a rate is not established in the Blue Book for a particular piece of equipment or plant, NJ TRANSIT shall establish a rate for that piece of equipment or plant that is consistent with its cost and use.

The provisions of this Subparagraph (a), "Contractor Owned Equipment and Plant" shall apply to the equipment and plant owned directly by the Contractor or by entities which are divisions, affiliates, subsidiaries or in any other way related to the Contractor or its parent company.

- (b) Rented Equipment and Plant In the event that the Contractor does not own a specific type of equipment and must obtain it by rental, it shall be paid the actual rental for the equipment for the time that the equipment is actually used to accomplish the work, plus the cost of moving the equipment on, to, and away from the job. The Contractor shall provide a copy of the paid receipt for the rental expense incurred.
- 3.2.9.4 Overhead Costs: Overhead shall be defined to include any and all Contractor Field Office and Home Office overhead and operating expenses whatsoever. Overhead includes, as a minimum, the following categories of expense, regardless of whether or not the Contractor's accounting system allocates such expenses on a direct or indirect basis:
  - (a) Salary and expenses of all Field Office employees, including project managers, supervising officers, supervising employees, superintendents, technical, scheduling or engineering employees, draft persons and clerical or stenographic employees;
  - (b) Charges for minor equipment, small tools, and other miscellaneous supplies and expenses, including computers and telephones, personal protection equipment, shovels, picks, axes, saws, bars, sledges, lanterns, jacks, cables, pails, wrenches, etc.;
  - (c) Charges for trailer rentals, utility and other temporary facility rental and maintenance charges, monthly utility charges, and all other costs to operate and maintain Contractor's Field Office unless otherwise provided as a direct charge elsewhere in the contract.
  - (d) Salary and expenses of Home Office employees, including executive officers, managers, professional and administrative staff, and clerical and support staff;
  - (e) Charges and expenses for drafting, Computer Assisted Design, scheduling, billing, financing, etc.
  - (f) All other costs to operate and maintain the Contractor's Home Office.
  - (g) Bond and insurance costs described in Sub-article 3.2.9.5.

The Contractor agrees that its overhead costs will be fully and fairly compensated by the fixed, non-negotiable overhead percentage markups set forth in Sub-article 3.2.10.

- 3.2.9.5 <u>Bond and Insurance</u>: Compensation for bond premiums and other insurance premiums not listed in Subarticle 3.2.9.1 shall be considered paid for under the overhead percentages added to the sum of the actual cost for labor, material and equipment and will not be considered or paid separately by NJ TRANSIT.
- 3.2.9.6 Profit: The Contractor's profit shall be negotiated as a percentage markup based on the type of work, the value of the change, the pricing basis and the amount of risk to the Contractor associated with the work to be performed. The Contractor agrees that the profit percentage markups are subject to negotiation on each change. However under no circumstances shall negotiated markups exceed the maximum allowable markup set forth in Sub-article 3.2.10.
- 3.2.10 Overhead and profit markups on each change shall be calculated in accordance with this Sub-article 3.2.10. Where work is performed by Subcontractors at any tier, the Contractor shall reach an agreement with such Subcontractors as to the distribution of payments, including overhead and profit markups made by NJ TRANSIT for such work. No additional payment therefor will be made by NJ TRANSIT by reason of the performance of the work of any Subcontractor.
- 3.2.10.1 When work is to be added or deleted on a Unit Price basis, the Unit Price shall govern and is deemed to include all markups for overhead and profit. No additional markups for overhead and profit will be allowed. When a complete Bid Item is deleted, it shall be treated as a Unit Price Bid Item (regardless of whether it is a Lump Sum or Unit Price item) and the total bid price for that item shall be deducted from the Contract Price.
- 3.2.10.2 When work is to be added on a lump sum or time and material basis, markups for profit and overhead shall be as specified in Table 3.2.10.
- 3.2.10.3 When work is to be deleted on a lump sum or time and material basis, markups specified in Table 3.2.10, for overhead costs that will not be incurred and profit that would have been realized if the work had not been deleted, shall be included in the deductive cost proposal submitted by the Contractor. If the Contractor's deductive cost proposal does not include an amount for overhead and profit, the Contracting Officer will add the markups specified in Table 3.2.10 to the cost proposal. When work is to be deleted, the Contractor may include documented cancellation and restocking charges and subtract those charges from the cost basis of the deductive cost proposal.
- 3.2.10.4 When work is to be both added and deleted on a lump sum or time and material basis, the cost basis shall be determined first by calculating both the added and deleted labor, material and equipment costs.

  Overhead and profit markups specified in Table 3.2.10 shall be applied to:

- a) Net increase in cost basis, in which case Paragraph 3.2.10.2 shall govern;
- b) Net decrease in cost basis, in which case Paragraph 3.2.10.3 shall govern.

Should there be a net change in cost basis of zero, there will be no change in the Contract Price.

- 3.2.10.5 When there is a change only to the material being supplied and no additional labor cost will be incurred by the Contractor or subcontractors at any tier, markups for overhead and profit shall be as specified in Table 3.2.10.
- 3.2.10.6 When a change is authorized for standby time, markups for overhead and profit shall be as specified in Table 3.2.10. Any claim for standby time will be rejected unless documented by time sheets signed by the Inspector.
- 3.2.10.7 When a change is authorized for overtime and the work to be performed is an established item of work, markups for overhead and profit shall be as specified in Table 3.2.10 and shall be applied only to the premium portion of labor costs.

Table 3.2.10: Maximum Overhead and Profit Markups for Added Work					
Work Performed By	Change Order Pricing Basis	Cost Basis	Overhead Markup (as a % of Cost Basis)	Maximum Profit Markup (as a % of Cost Basis)	
Contractor	Lump Sum or T&M with GMP	Contractor labor, material and equipment costs	10%	10%	
Subcontractor (at any tier)	Lump Sum or T&M with GMP	Subcontractor labor, material and equipment costs	15% (to be shared between Contractor and Subcontractors)	15% (to be shared between Contractor and Subcontractors)	
Contractor	T&M with NJT Upset Price	Contractor labor, material and equipment costs	10%	5%	
Subcontractor (at any tier)	T&M with NJT Upset Price	Subcontractor labor, material and equipment	15% (to be shared between Contractor and Subcontractors)	7.5% ( to be shared between Contractor and Subcontractors)	
Contractor	Emergent T&M	Contractor labor, material and equipment costs	10%	7.5%	
Subcontractor (at any tier)	Emergent T&M	Subcontractor labor, material and equipment costs	15% (to be shared between Contractor and Subcontractors)	10% (to be shared between Contractors and Subcontractors)	
No Additional Labor	Lump Sum All T&M's	Only Material Costs	2.5%	2.5%	
Standby Time	T&M Only	Labor Costs Only	5%	5%	
Overtime	T&M Only	Premium Labor Costs Only	5%	5%	

# 3.3 TIME AND MATERIAL (T&M) CHANGE ORDER RECORDS

3.3.1 The Contractor shall maintain its records in such a manner as to provide a clear distinction between the direct costs of T & M work and the costs of other Work.

From the above records, the Contractor shall furnish the Construction Manager completed daily work reports for each day's work to be paid for on a T & M basis. The daily T & M work reports shall be detailed as follows:

- (a) Name, classification, date, daily hours, total hours, rate, and extension for each laborer and foreman.
- (b) Designation, dates, daily hours, total hours, rental rate, and extension for each unit of machinery and equipment.
- (c) Quantities of materials, prices, and extensions.
- (d) Transportation of materials.
- (e) Cost of property damage, liability, and worker's compensation insurance premiums, unemployment insurance contributions, bonds, and social security tax.
- 3.3.2 Material charges shall be substantiated by valid copies of vendor's invoices. Such invoices shall be submitted with the daily work reports, or if not available, they shall be submitted with subsequent daily T & M work reports. Should said vendor's invoices not be submitted within 60 days after the date of delivery of the material, NJ TRANSIT reserves the right to establish the cost of such materials at the lowest current wholesale prices at which said materials are available, in the quantities concerned delivered to the location of work less any discounts provided in Sub-article 3.2.9.2(a), above.
- 3.3.3 Said daily time and material work reports shall be signed by the Contractor or its authorized representative.
- 3.3.4 The Construction Manager will compare NJ TRANSIT's records with the completed daily time and material work reports furnished by the Contractor and make any necessary adjustments. Except when daily time and material work reports are submitted for the purpose of recording the cost of disputed items of work, as required under Sub-article 3.1.4, when daily time and material work reports are agreed upon and signed by both parties, said reports shall become the basis of payment for the work performed, but shall not preclude subsequent adjustment based on a later audit by NJ TRANSIT. In the case of disputed items of work, the signature of NJ TRANSIT's Construction Manager represents only the Construction Manager's concurrence with the costs allocated by the Contractor to the disputed work and shall not preclude the Contracting Officer from disputing such work.

### 3.4 CONTRACTOR INITIATED CHANGE ORDERS

- 3.4.1 The Contractor may request a change order for an increase in the cost of the performance of any part of the Work or a change in the Contract Time only when such costs or time impacts are attributable to the following:
  - a.) Latent errors and omissions in the contract documents;

b.) Additional costs or an extension of Contract time for which a change order is expressly permitted under any Article in this Contract.

The Contractor must give immediate notice to the Construction Manager when it becomes aware of the condition causing the initiation of a request for change.

- 3.4.2 Contractor Initiated Change Order Requests (CICOR's) will not be considered unless the Contractor has strictly complied with the notice requirements of the appropriate Articles of this Contract. The Contractor further understands and agrees that neither the procedure established under this Article nor the review of CICOR's by NJ TRANSIT pursuant hereto shall in any way affect the requirements of the filing of a Notice of Claim or the filing of a suit pursuant to the provisions of N.J.S.A. 59:13-1 et seq.
- 3.4.3 Within 15 days of notification by the Contractor of a condition causing the initiation of a request for change, the Contractor must submit the CICOR with sufficient detail to enable NJ TRANSIT to ascertain the basis and amount of said request. As a minimum, the following information must accompany each request submitted pursuant to the provisions of this Sub-article:
  - (a) A detailed factual statement of the CICOR providing all necessary dates, locations and items of work affected by the CICOR;
  - (b) the date on which facts arose which gave rise to the CICOR;
  - (c) the name, function, and activity of each NJ TRANSIT individual, official or employee involved in or knowledgeable about such CICOR;
  - (d) the specific provisions of the Contract which support or mitigate against the CICOR and a statement of the reasons why such provisions support or mitigate against the CICOR;
  - (e) if the CICOR relates to a decision of NJ TRANSIT or the Construction Manager which the Contract leaves to NJ TRANSIT's or the Construction Manager's discretion or as to which the Contract provides that NJ TRANSIT's or the Construction Manager's decision is final, the Contractor shall set out in detail all facts supporting its contention that the decision of NJ TRANSIT or the Construction Manager was fraudulent or capricious or arbitrary or is not supported by substantial evidence;
  - (f) the identification of documents and the substance of oral communications relating to such CICOR;
  - (g) a statement as to whether the additional compensation or extension of time sought is based on the operation of the provisions of the Contract or an alleged breach of contract;
  - (h) if an extension of time is sought, the specific days for which it is sought and the CPM schedule data providing a logical basis for such an extension;
  - (i) if additional compensation is sought, the exact amount sought and a breakdown of that amount in accordance with the pricing specifications set forth in Article 3.2.

It will be the responsibility of the Contractor to furnish within a reasonable time such further information and details as may be required by NJ TRANSIT to determine the facts or contentions involved in the CICOR's, including but not limited to those items identified in Article 3.5.

### 3.5 AUDIT OF CHANGE ORDERS

- 3.5.1 The cost records of the Contractor and its Subcontractors pertaining to change orders shall be open to inspection or audit by representatives of NJ TRANSIT during the life of the Contract and for a period of not less than three years after the date of acceptance thereof, and the Contractor and its Subcontractors shall retain such records for that period. This audit provision shall apply whether or not such change orders are part of a suit pending in the courts of this State pursuant to the New Jersey Contractual Liability Act. The audit may be performed by employees of NJ TRANSIT or by an auditor under contract with NJ TRANSIT. The audit may begin on ten (10) days notice to the Contractor or its Subcontractor. The Contractor or Subcontractor shall provide adequate facilities, acceptable to NJ TRANSIT, for such audit during normal business hours. The Contractor or its subcontractor shall make a good faith effort to cooperate with the auditors.
- 3.5.2 If an audit is to be commenced more than 60 days after the acceptance date of the Contract, the Contractor will be given a reasonable notice of the time when such audit is to begin.
- 3.5.3 As a minimum, the Contractor shall maintain and the auditors shall have available to them the following documents:
  - (a) daily time sheets and foreman's daily reports.
  - (b) union agreements.
  - (c) insurance, welfare and benefits records.
  - (d) payroll registers.
  - (e) earnings records.
  - (f) payroll tax forms.
  - (g) material invoices and/or requisitions.
  - (h) material cost distribution worksheet.
  - (i) equipment records (list of company equipment, rates, etc.)
  - (j) vendors', rental agencies', and subcontractors' invoices.
  - (k) subcontractors' payment certificates.
  - (I) canceled checks (payroll and vendors).
  - (m) job cost report.
  - (n) job payroll ledger.
  - (o) general ledger.
  - (p) cash disbursements journal.
  - (q) financial statements for all years reflecting the operations on this Project.
  - (r) income tax returns for all years reflecting the operations on this Project.

- (s) depreciation records on all company equipment whether such records are maintained by the company involved, or its accountant, or others.
- (t) if a source other than depreciation records is used to develop costs for the Contractor's internal purposes in establishing the actual cost of owning and operating equipment, all such other source documents.
- (u) all documents which reflect the Contractor's actual profit and overhead during the years this Project was being performed and for each of the five years prior to the commencement of this Project.
- (v) all documents related to the preparation of the Contractor's bid including the final calculations on which the bid was based.
- (w) all documents which relate to each and every change order together with all documents which support the amount of claimed costs.
- (x) worksheets used to prepare the CICOR or cost proposal tracing the cost elements of the change order (including, but not limited to, labor, benefits and insurance, materials, equipment and subcontractors) to the primary records which establish the time periods, individuals, hours, rates, materials and equipment involved in the change order.

### 3.6 SUPPLEMENTAL CONSTRUCTION COSTS

Whenever the Bid Item "Supplemental Construction Costs" appears in the Bidder's Proposal, NJ TRANSIT has provided an allowance for additional or supplemental construction work that it has not yet defined. This allowance is provided for the sole convenience of NJ TRANSIT and can only be used for work authorized by NJ TRANSIT.

All additional or supplemental work authorized under this provision will be incorporated into the Contract by Change Order pursuant to Article 3.1. The Change Order will describe the additional or supplemental work with any associated cost changes and will reduce the Supplemental Construction Cost allowance in the amount specified in the Change Order. Residual amounts remaining in the Supplemental Construction Cost Allowance Bid Item at Final Completion will be deleted from the Contract Amount by NJ TRANSIT.

### 4. PROTECTION AND CONTROL OF PREMISES

- 4.1 RESPONSIBILITY FOR WORK
- 4.1.1 The Contractor shall be responsible for damages arising from its work on the Project, to any part of the Project work, both temporary and permanent, to adjoining property and to NJ TRANSIT property both within and outside the project limits. The Contractor shall, at its own expense, protect finished work susceptible to damage and keep the same protected until the Project is completed and accepted by NJ TRANSIT.
- 4.1.2 All Contractor and Sub-Contractor personnel are required to carry, and display when requested, a form of photo identification acceptable to NJ TRANSIT.

- 4.1.3 The Contractor shall make, use, and provide proper, necessary, and sufficient precautions, safeguards, and protection against the occurrence of accident, injury, damage or hurt to person or property during the progress of the work. The Contractor alone shall be responsible for the safety, efficiency, and adequacy of its plant, appliances, and methods, and for damage or injury which may result from its failure to act in a safe, careful, efficient, and workmanlike manner. Any action or direction by NJ TRANSIT or its representatives relating to the adequacy or implementation of the Contractor's precautions, safeguards, and protection shall in no manner relieve the Contractor of any of its obligations or responsibilities hereunder.
- In case of an emergency which threatens persons or property, the Contractor shall act, without previous instructions from the Construction Manager, in a diligent and proper manner to remedy the situation. The Contractor shall notify the Construction Manager immediately. During non-standard work hours (See Sub-Article 4.2.9 for Standard Work Hours) the Rail or Bus Control Center, as appropriate, shall be notified. Claims for compensation by the Contractor for Extra Work arising from emergencies not caused by the Contractor shall be documented and promptly submitted for review and approval. Where the Contractor has notified the Construction Manager of such emergency but has not taken any action, it shall act as instructed or authorized by the Construction Manager.

## 4.2 USE OF PREMISES

- 4.2.1 Prior to the use of NJ TRANSIT premises, the Contractor shall obtain the approval of the Construction Manager for the Contractor's staging area(s), access and egress to the premises, parking area(s) for Contractor vehicles and equipment, elevator use, and any other use of NJ TRANSIT property, facilities, or on site utilities. The Contractor shall notify the Construction Manager no later than 72 hours in advance of any utility shutdowns that affect NJ TRANSIT facilities. All cut overs of existing mechanical and electrical services shall be done at a time convenient to NJ TRANSIT and any other private or public agency having jurisdiction, so as not to interfere with facility operations.
- 4.2.2 The Contractor shall comply with the rules and regulations of NJ TRANSIT. The Contractor shall confine its apparatus, the storage of materials and the operations of workmen to limits indicated by law, ordinances, permits, contract limit lines as established, or directions of the Construction Manager and shall not unreasonably encumber the premises with its materials. The Contractor shall maintain a reasonably clean job site free of debris and litter.
- 4.2.3 The Contractor shall be responsible for hoisting and distributing material and equipment throughout the Project for its work, and the work of its subcontractors. The Contractor shall handle materials in a controlled manner with as few handlings as possible. The Contractor shall not drop or throw materials from heights. The Contractor shall not load or permit any part of a structure to be so loaded as might endanger its safety or integrity.

The Contractor agrees to NJ TRANSIT's use and occupancy of a portion or unit of the Project after the portion or unit has been declared Substantially Complete by NJ TRANSIT.

- 4.2.4 The Contractor shall request of and obtain from NJ TRANSIT specific instructions, rules and regulations regarding the required conduct of the Contractor during the construction so that the security and safety of personnel and property, including both NJ TRANSIT's and the general public's, will not be endangered. NJ TRANSIT will not allow an increase in the Contract amount due to the Contractor's failure to determine the conditions under which it must perform its contractual obligations. The Contractor shall enforce the Construction Manager's instructions regarding but not limited to signs, advertisements, fires, smoking, alcohol, safety and cleanliness on the site.
- 4.2.5 Accessibility to the work area shall be determined by the Contractor and approved by the Construction Manager, unless otherwise indicated in the Contract Documents. It is the Contractor's responsibility to make arrangements for use of public and/or private properties required to execute and complete the work under this Contract.
- 4.2.6 Space that the Contractor may require for plant, equipment, storage or other purposes, in addition to that available therefor at the site of the Project, shall be procured by the Contractor and the cost thereof is considered to be included in the prices bid for the various items scheduled in the Bid. In event of default, NJ TRANSIT has the right to take over and occupy such space, or cause it to be occupied, for the purpose of completing the Project, at the Contractor's expense. If leased, the lease shall contain a provision that in event of default by the Contractor the lease may be assigned to NJ TRANSIT or its nominee. The Contractor agrees, in event of said default, that it will make such assignment. At the time of execution, a copy of all lease agreements shall be submitted to the Construction Manager.
- 4.2.7 The Contractor shall provide watchmen service, when necessary or when directed by the Construction Manager throughout the period of construction, to adequately protect the work, stored materials and temporary structures located on the premises, and to prevent unauthorized persons from entering upon the construction site.
- 4.2.8 The Contractor shall adequately insure, secure and protect its own tools, equipment, materials and supplies.
- 4.2.9 Regular working hours are from 8:00 a.m. to 4:30 p.m., Monday through Friday. The Contractor shall obtain the written approval of the Construction Manager for performance of work other than during regular working hours or on weekends or Holidays. Standard NJ TRANSIT Holidays are as follows: New Year's Day, Martin Luther King Day, President's Day, Good Friday, Memorial Day (Monday observance), Independence Day, Labor Day, Thanksgiving Day and the day after, and Christmas Day. The Contractor shall advise the Construction Manager no less than seven (7) days in advance of work to be performed during such times. This shall not preclude taking prudent and necessary actions in an emergency situation.

- 4.3 MAINTENANCE AND CLEANING OF PREMISES
- 4.3.1 The Contractor shall maintain and clean the premises as necessary to ensure a safe, orderly and clutterfree working environment. The Contractor shall comply with the following cleaning requirements:
- 4.3.2 The Contractor shall retain all stored items in an orderly arrangement to allow maximum access, not impede drainage or traffic, eliminate fire hazards and provide proper protection of materials. Weekly, and more often if necessary, the Contractor shall inspect all material storage conditions on the site and restack, tidy, or otherwise service material storage conditions to maintain an orderly arrangement. Scrap, debris, waste materials, and other items not required for construction of the Work shall be regularly disposed of in accordance with the requirements set forth below. The Contractor shall wet down dry materials to minimize dust and prevent blowing dust. The Contractor shall maintain the site in a neat and orderly condition at all times.
- 4.3.3 The Contractor shall provide adequate storage for all items awaiting removal from the job site, observing all requirements for fire protection, health and protection of the environment. Combustible waste, scrap, rubbish, etc., shall be stored in properly sized metal containers (with metal covers where practical) pending removal from the premises. Pest control services shall be provided as necessary to control vermin, rodents and other pests. Daily, and more often as necessary, the Contractor shall inspect the site and move all scrap debris and waste material to the place designated for their storage. At least once a week and more often if necessary, the Contractor shall completely remove and legally dispose of all scrap, debris and waste material from the job site. Placement of waste containers and carting schedules shall be submitted to the Construction Manager for the Construction Manager's review and approval. If the Contractor fails to remove debris from the site within three days after it has been given written notice to do so by the Construction Manager, the Construction Manager will have the debris removed by others and the cost backcharged to the Contractor.
- 4.3.4 Weekly, and more often if necessary, the Contractor shall sweep all interior spaces clean. "Clean", for the purpose of this Subparagraph, shall be interpreted as meaning free from dust and other material capable of being removed by use of reasonable effort and a hand-held broom.
- 4.3.5 Preparatory to the installation of any succeeding materials, the Contractor shall clean all structures, or pertinent portions thereof, to the degree of cleanliness recommended by the manufacturer of the succeeding material, using all equipment and materials required to achieve the required cleanliness.
- 4.3.6 The Contractor shall schedule cleaning operations so that dust and other contaminants resulting from any cleaning process will not fall on wet, newly painted surfaces.

4.3.7 The Contractor shall schedule final cleaning, as approved by the Construction Manager, to enable NJ TRANSIT to accept a completely clean project. Prior to completion of Work, the Contractor shall remove from the job site all tools, surplus materials, equipment, scrap, debris and waste. The Contractor shall remove all traces of soil, waste material, mortar and paint droppings, grease and other foreign matter from all interior and exterior surfaces. All floor slabs shall receive a final steam cleaning.

## 4.4 FIRE PREVENTION ON PREMISES

- 4.4.1 Each Contractor shall perform its work on or about the premises in a careful manner with full consideration to fire prevention as required by the New Jersey Uniform Fire Code (NJUFC), N.J.A.C. 5:71-1 et seq., and its referenced standards. Fire resistant materials shall be used for temporary enclosures. Storage of flammable materials on the site shall be subject to limitations specified in the NJUFC and the approval of the Construction Manager, and shall be the Contractor's responsibility. Accessability to fire hydrants shall be maintained at all times. On site open burning of rubbish, garbage, trade waste, leaves or plant life is strictly prohibited by New Jersey law.
- 4.4.2 Chemical extinguishers approved by the Construction Manager shall be provided by the Contractor during the progress of the work where specified by the NJUFC or required by Fire Officials from the DCA Bureau of Fire Safety or the local fire department. In addition, the Contractor shall be responsible for furnishing and maintaining his own extinguisher equipment in storage sheds, warehouses, Contractor's offices, and workmen's temporary buildings.
- 4.4.3 The Contractor shall maintain an active program of fire prevention to keep workers fire conscious during the entire Contract duration. It shall designate one member of its organization to execute and coordinate the fire control measures of its own organization, that of all subcontractors under its jurisdiction and that of all other personnel at the site. It shall report to the Construction Manager any lack of cooperation or refusal to participate on the part of any worker or subcontractor with regard to the fire prevention program. Failure of any worker or subcontractor to cooperate with the Contractor in carrying out the above program shall be grounds for barring that individual or firm from the Project.
- 4.4.4 Temporary heating systems provided under Article 5.6 shall conform to the requirements of the NJUFC where the building is fully or partially occupied.
- 4.4.5 Where required under the NJUFC, the Contractor shall be responsible for obtaining required permits from the DCA Bureau of Fire Safety for flammable or combustible gas or liquid storage, fumigation/fogging, blasting, welding, burning, cutting and torch-applied roofing or paint removal.

## 4.5 PROTECTION AGAINST DAMAGE

4.5.1 The Contractor shall protect existing property, structures, curbs, walks, drives, trees, shrubs, lawns, and landscape work on the site or affected by its activities from damage and shall provide such guards and covering

as necessary. Damaged items shall be repaired or replaced at the Contractor's expense to the satisfaction of NJ TRANSIT. No extension of time will be allowed for repair or replacement of damaged items. Should the Contractor not repair or replace such damaged items, NJ TRANSIT will take corrective measures and deduct the cost from the Contract Price.

- 4.5.2 It shall be the responsibility of the Contractor at all times to protect construction excavations, trenches up to 10 feet from structures, and the structures from water damage, including damage by rainwater, ground water, backing up of drains, downspouts, or sewers. The Contractor shall construct and maintain necessary drainage and do pumping required to keep the Project free from water, and shall perform pumping necessary for the full and proper execution of the construction work and protection of the Project including equipment installed therein.
- 4.5.3 Beyond a point 10 feet from facilities, it shall be the responsibility of the Contractor to protect the trenches by shoring or other methods and perform pumping required to dispose of the surface and subsurface water to permit the satisfactory performance of the work. Each Contractor shall provide its own pumping equipment of adequate capacity and shall be responsible for fuel, cost of operators, and supervision.
- 4.5.4 The Contractor shall protect equipment, such as electric switch gear and HVAC equipment, that is subject to damage by moisture during the period from installation of equipment to completion of the Project acceptance, and shall provide temporary waterproof enclosures and ceilings over such equipment. The interior of the enclosure shall be kept dry by whatever measures are necessary. Special openings shall be provided in the enclosures and ceilings in order to service the equipment during the protection period. The Contractor shall procure and maintain, during the protection period, insurance covering the subject equipment in the full amount of the value of the equipment. See Article 9.2- INSURANCE for submission of proof of carriage of insurance.
- 4.5.5 The Contractor shall remove snow and ice as may be required for the proper protection and prosecution of the Contract and to provide access to the Project Site.
- 4.5.6 In the event of temporary suspension of work, or during inclement weather, or whenever the Construction Manager shall direct, the Contractor shall protect, and shall cause its subcontractors to protect, carefully its and their work and materials against damage from the weather. If, in the opinion of the Construction Manager or NJ TRANSIT, work or materials have been damaged, such work or materials shall be removed and replaced at the expense of the Contractor.
- 4.5.7 Unless otherwise specified or shown in the Contract Documents, the Contractor shall provide protection of the entire construction area. It shall also install four foot high snow fence around trees that are to remain and that are located within the Contract Limit Line, at a distance equal to the branch spread of the tree.

#### 4.6 PROTECTION OF PRIVATE PROPERTY

The Contractor shall not enter on or make use of private property in the prosecution of the Project unless written permission therefor is secured, in duplicate, from the owner of the property, one copy of which shall be filed with NJ TRANSIT. The Contractor shall promptly restore or repair, without cost to NJ TRANSIT and in a manner satisfactory to the owner of the property, property damaged or destroyed by its operations. Special attention shall be given to the protection of existing landscape features and natural vegetation.

### 4.7 PROTECTION OF PUBLIC UTILITIES

4.7.1 The terms public utility or public utilities used in this Contract shall be construed to include those publicly and privately owned. Within the site of the Project there may be public utility facilities, and notwithstanding any other clause or clauses of this Contract, the Contractor shall not proceed with its work until it has made diligent inquiry at the offices of the Construction Manager, the utility companies and municipal authorities, NJ TRANSIT or other owners to determine their exact location. The Contractor shall notify, in writing, the utility companies and municipalities or other owners involved of the nature and scope of the Project and of its operations that may affect their facilities or property. Two copies of such notices shall be sent to the Construction Manager.

The Contractor's attention is called to the fact that the exact locations of the various overhead and underground lines, utilities, and structures located throughout the Project are unknown, and the Contractor is advised to use extreme caution during construction. The plans showing the approximate locations of the various overhead and underground lines, utilities, and structures are to used only as guidelines and are not guaranteed as to their accuracy or correctness.

- 4.7.2 The Contractor shall carry out its work carefully and skillfully and shall support and secure public utility facilities so as to avoid damage to them. Flow in drains and sewers shall be satisfactorily maintained. It shall not move any public utility facilities without the owner's written consent and, at the completion of the work, their condition shall be as safe and permanent as before. When public utility facilities are damaged by the Contractor, it shall notify their owner, who shall cause the damage to be repaired at the Contractor's expense. If the cost thereof is not paid by the Contractor within 30 days after repairs have been completed, the Contracting Officer shall deduct an amount sufficient to cover the cost from any monies due or that may become due the Contractor under this Contract. Service connections damaged by the Contractor shall be repaired by competent skilled mechanics.
- 4.7.3 During the normal course of construction the Contractor may find it necessary to temporarily relocate certain public utilities in order to proceed. The Contractor will be responsible for the coordination and scheduling of all such relocations with the utility owner. If the Bid Item "Protection of Public Utilities" appears in the Bidder's Proposal, NJ TRANSIT shall reimburse the Contractor for these relocation services upon receipt of an itemized invoice from the participating utility owner, and only for the amount of the invoices, to be submitted along with the Contractor's monthly invoice. If the Bid Item "Protection of Public Utilities" does not

appear in the Bidder's Proposal, the Contractor shall assume all costs associated with the temporary relocation of public utilities.

When facilities requiring relocation belong to NJ TRANSIT, the Contractor shall make requests for relocation by NJ TRANSIT personnel through the Construction Manager. The cost of such relocation shall be borne by NJ TRANSIT.

4.7.4 Under no circumstances shall the Contractor be entitled to damages of any kind arising from the need to relocate public utilities in order to complete the Work.

### 4.8 PROTECTION OF EXISTING MONUMENTS

Existing monuments and title stones which need not be removed shall be left in place and protected by the Contractor against damage and dislocation. When relocation or change in the grade of existing monuments is necessary, they shall be protected in their original position until their removal is approved by NJ TRANSIT, and shall be reset when directed and in conformance with the new lines and grades to be furnished by the Contractor. Monuments and title stones that are to be left in place or reset and are moved without approval of NJ TRANSIT shall be replaced at the Contractor's expense.

### 4.9 MAINTENANCE AND PROTECTION OF ROADWAY AND PEDESTRIAN TRAFFIC

- The Contractor shall conduct its work with the least possible obstruction of traffic. The convenience of the public and of the residents adjacent to the Project, and the protection of persons and property, are of primary importance and shall be provided for by the Contractor in an adequate and satisfactory manner. When a detour will be established, the Contractor shall make arrangements for establishing, maintaining, and signing for it and provide safety measures as are necessary to provide traffic guidance and protection. The signage shall include safety, directional and informational signals and devices necessary to provide effective pedestrian and vehicular circulation. The number and location of the signals and devices shall be subject to the Construction Manager's approval.
- 4.9.2 The Contractor shall erect or place, and maintain in good condition, appropriate and adequate barricades, signs, lights, beacons, flares, approved red flasher units, rubber cones, drums and other warning and danger signals and devices at working sites, closed roads, intersections, open excavations, locations of material storage, standing equipment and other obstructions; at points where the usable traffic width of the road is reduced; at points where traffic is deflected from its normal course of lanes; and at other places of danger to vehicular or pedestrian traffic or to completed work. Flagmen will be used as necessary. The various traffic control and warning devices shall be in accordance with Part VI of the Manual on Uniform Traffic Control Devices of the USDOT and approved by NJ TRANSIT's Construction Manager.
- 4.9.3 The Contractor shall provide, maintain and remove when no longer required, temporary driveways, parking areas and walkways that may be necessary to allow access to all parts of the Project, to adjacent property,

and for handling of materials and equipment. Should the Contractor elect to place materials that will be incorporated into the permanent driveways, parking areas or walks, it shall not do so without having prepared the subgrade as may be elsewhere required by the Specifications nor will it be relieved from responsibility for providing additional materials or for reworking the subgrade, if required to make the improvements conform fully with the Specifications.

- 4.9.4 The Contractor shall obtain permission in writing from the Construction Manager before using existing driveways or parking areas for construction purposes. It shall maintain such driveways and areas in good condition during the construction period, and at the completion of the Project, shall leave them in the essentially equal or better condition as at the start of the work to the satisfaction of the Construction Manager.
- 4.9.5 The Contractor shall employ construction methods and means that will keep flying dust to the minimum. Trucks hauling materials shall have tight tail gates and shall be loaded with adequate freeboard of not less than three inches, without precarious cones or piles of material. It shall provide for the containment of dust on the Project, and on roads, streets and other areas immediately adjacent to the Project limits, wherever traffic or buildings that are occupied or in use are affected by such dust. The materials and methods used for dust control shall be subject to the approval of NJ TRANSIT.
- 4.9.6 When vehicular or pedestrian traffic, or both, is to be maintained on new or existing roadways and pedestrian paths of travel, the Contractor shall plan and carry out its work to provide for the convenient and safe passage of such traffic. The Contractor shall provide for prompt removal from such roadways and pedestrian paths of all dirt and other materials that have been spilled, washed, tracked or otherwise deposited thereon by its hauling or other operations. Roadways and pedestrian paths within the limits of the Project which are reserved for traffic shall be maintained by the Contractor free from obstructions and in a smooth traveling condition at all times.
- 4.9.7 The Contractor shall not perform construction work above vehicular or pedestrian traffic until it obtains explicit written permission from the Construction Manager. Subject to such permission, the Contractor shall provide the necessary devices and means to protect such traffic from falling construction materials and other objects and from painting operations, during the time that construction work is carried on above traffic.
- 4.9.8 The Contractor shall comply with local codes and ordinances affecting complete or partial roadway closings, detours and roadway and pedestrian protective measures. All costs associated with maintaining and protecting roadway and pedestrian traffic is at the Contractor's sole expense and is considered included in the Contract Price.

#### 4.10 MAINTENANCE OF MARINE TRAFFIC

Work over, on or adjacent to navigable waters shall be so conducted that free navigation of the waterways will not be interfered with and the existing navigable depths will not be impaired except as allowed by permit issued by the U.S. Coast Guard and/or the U.S. Army Corps of Engineers, as applicable.

## 4.11 MAINTENANCE OF RAILROAD TRAFFIC

- 4.11.1 Where the Project includes work across, over, under or adjacent to railroad tracks or railroad right-of-way as specified in the Contract Documents, the Contractor shall safeguard the traffic, tracks and appurtenances, and other railroad property affected by its work. It shall comply with the regulations of NJ TRANSIT Rail Operations and those of any other operating railroad company relating to the work; shall keep the tracks clear of obstructions; shall provide barricades, warning signs, lights, flares, and other danger signals and means of protection; and shall arrange with the operating railroad company through NJ TRANSIT for the furnishing of watchmen and flagmen and other protective service that may be required by the railroad company. The Contractor's work activities shall be conducted in strict conformance with the governing rules of the specific railroad on whose track it is working as detailed in Appendix D to these General Provisions. The Contractor shall note that work around and adjacent to the railroad is severely restricted during the annual holiday moratorium on performance of work, which is defined as the period beginning 5 days prior to Thanksgiving and ending on January 2 of the following year. Track outages and fouling may not be permitted during this period.
- 4.11.2 Work done within NJ TRANSIT railroad right-of-way shall be subject to the approval of NJ TRANSIT in matters affecting railroad property and the safety and operation of its trains. The safety and continuity of railroad operation shall be of primary importance and shall be at all times protected and safeguarded. The Contractor, and any subcontractor, shall perform and arrange all pertaining construction work accordingly. Work shall be performed carefully and shall be regulated so as to avoid interruption of train movements and damage to the tracks and other facilities of the railroad. The Contractor agrees that delays in the performance of the Work attributable to the operations of the railroad shall not be the basis of claims for damages for delay or otherwise or for additional compensation. However, the Contractor may be entitled to an extension of time for completion of the Work, but only to the extent that the critical path of the work schedule is impacted by NJ TRANSIT railroad operations.
- 4.11.3 The maintenance and protection of railroad traffic will not be paid for under any specific scheduled item but the cost thereof, including the safeguarding of tracks, traffic and appurtenance of the railroad, watchmen, barricades, lights, signs, signals, warning, other protection and services is considered included in the Contract Price.
- 4.11.4 If work is done on or affecting the property of a railroad company other than NJ TRANSIT, the railroad company may assign inspectors or engineers during the time the Contractor is engaged in construction work on said railroad property for the general supervision of construction operations to insure adherence to Plans and

Specifications and to insure the use of approved construction methods pertaining to the safety and condition of the company's right-of-way. The salary and expense of said inspector and/or engineers and the cost of other engineering services furnished by the railroad company will be at no cost to the Contractor, unless otherwise specified in the Special Provisions. The same care taken to protect NJ TRANSIT railroad traffic as set forth above shall be exercised in the protection of railroad traffic on other affected railroads.

### 4.11.5 Detailed Protective Measures

- A. General
- Whenever in this Sub-article the term "Railroad" is used without further qualifications, it shall mean and be taken to mean NJ TRANSIT Rail Operations.
- The Contractor should note that the proposed work involves construction operations on and over property owned or controlled by the Railroad and will be performed adjacent to the high speed main line electrified tracks of the Railroad in the vicinity of high voltage lines of the Railroad. In working near these lines great care must be exercised and the Railroad's rules detailing requirements for clearance to be maintained between equipment and energized wires and other instructions in regard to working in the vicinity of their electric operations and requirements must be strictly observed whenever the tracks, structures, or properties of the Railroad are involved or affected.
- 3. Prior to commencement of work on Railroad property, the Contractor will name a qualified safety representative to interface with the Railroad's supervision. The Contractor safety representative will be responsible for ensuring full compliance with the Railroad's safety policies and procedures as they relate to the project. All Contractor personnel working within fouling distance shall attend the Railroad's safety orientation class which will be provided by the Railroad at no cost to Contractor. The Contractor's personnel may be required to travel to offices in Newark at One Penn Plaza East, or some other location convenient to the Railroad and remote from the site, for administration of this class. Each trained employee shall be issued a safety sticker to be placed on the employee's hard hat. The employee must display the sticker when working in the Railroad's limits. Contractor shall comply with the Railroad's safety requirements throughout the entire construction period. All costs encountered by Contractor due to complying with Railroad's safety requirements shall be at the sole expense of Contractor.
- 4. If, during the carrying out of the Work, the tracks or other facilities of the Railroad are endangered, the Contractor shall immediately do such work as directed by the Railroad to restore safety, and upon failure of the Contractor to carry out such orders immediately, the Railroad may take whatever steps as are necessary to restore safe conditions. The cost and expense to the Railroad of restoring safe conditions or of any damage to the Railroad's trains, tracks or other facilities caused by the Contractor or Subcontractor's operations, shall be considered a charge against the Contractor and shall be paid for by the Contractor, or may be deducted from any monies due or that may become due to Contractor under this Contract. Final payment to Contractor shall be

contingent upon a showing by the Contractor that the bills of the Railroad for such services have been paid by the Contractor.

- B. Rules and Regulations
- 1. Railroad traffic shall be maintained at all times with safety and continuity, and the Contractor shall conduct all of its operations on or over the Railroad's right-of-way fully within the rules, regulations, and requirements of the Railroad. The Contractor shall be responsible for acquainting itself with such requirements as the Railroad may demand. It is understood and agreed that the Contractor is cognizant of the limited ability of NJ TRANSIT to control the actions of the Railroad's operations and in its bid has made allowance for the fact that no additional compensation will be allowed for any delays, inconvenience or damages sustained by Contractor due to the actions, operations, inactions, or interference of the Railroad.
- The Contractor shall obtain verification of the time and schedule of track occupancy from the Railroad before proceeding with any construction or demolition work over, under, within, or adjacent to the Railroad's right-of-way. The Contractor shall submit for the approval of the Railroad a detailed description of the method of procedure which will be followed for Work within these areas. The Work in the field shall not proceed until the plans and method of procedure have been approved by the Railroad.
- 3. All work to be done under or over the Railroad's right-of-way shall be performed by the Contractor in a manner satisfactory to the Railroad and shall be performed at such times and in such manner as not to interfere with the movement of trains or traffic upon the tracks of the Railroad. The Contractor shall use all necessary care and precaution in order to avoid accidents, damage, delay or interference with the Railroad's trains or other property.
- 4. The Contractor shall give written notice to NJ TRANSIT's Construction Manager and the Railroad at least thirty (30) days prior to the commencement of any Work, or any portion of the Work, by the Contractor or its Subcontractors on, over or adjacent to the Railroad's right-of-way, in order to protect Railroad traffic.
- 5. If deemed necessary by the Railroad, it may furnish or assign an inspector who will be placed on the site of the Work during the time the Contractor or any Subcontractor is performing work under the Contract on Railroad property. The cost and expense will be paid directly by NJ TRANSIT.
- 6. Before proceeding with any construction or demolition work on, over or adjacent to the Railroad's property, a pre-construction meeting shall be held, at which time the Contractor shall submit for approval of the Railroad plans, computations, and a detailed description of Contractor's method of procedure for accomplishing the Work required under this Contract, including methods of protecting railroad traffic; however, such approval shall not serve in any way to relieve the Contractor of its complete responsibility for the adequacy and safety of its methods or procedures.
- 7. During the demolition procedures the Contractor must provide an approved shield to prohibit all debris from falling onto Railroad's right-of-way. The shield must be designed to provide a solid barrier between the work area and the tracks below. This shield must span over all tracks plus an

- additional 15 feet beyond the center line of each track. The Contractor is to submit details and calculations of the proposed shield for Railroad approval.
- 8. Whenever equipment or personnel are working closer than 15 feet from the nearest rail or 18 feet from the center line of track or over the top of track within this limitation, that track shall be considered fouled. Cranes, shovels, or any other equipment shall be considered to be fouling the track when located in such position that failure of same, with or without load, brings the equipment within the fouling limit. Operations within this fouling distance shall be conducted only with the permission of the Railroad and as directed by qualified railroad employees providing protection for track, signal, and catenary equipment. A power line is fouled and subject to hazard when any object is brought to a point less than 10 feet therefrom and a signal line or communication line shall be considered fouled and subject to hazard when any object is brought nearer than eight feet to any wire or cable.
- 9. The Contractor shall conduct its Work and handle its equipment and materials so that no part of any equipment shall foul an operated track or wire line without the written permission of the Railroad and NJ TRANSIT's Construction Manager. When the Contractor desires to foul an operated track, it must give the Railroad and NJ TRANSIT's Construction Manager written notice of its intentions thirty (30) days in advance, so that if approved, arrangements may be made for proper protection of the Railroad. Although the Railroad may shift or reroute traffic to accommodate Contractor, flagging protection shall still be provided when fouling a normal operating track as this track could be returned to operation on short notice as necessitated by demand. Contractor shall conform to working hours as determined by the Railroad with regard to fouled tracks and/or platform work. If railroad flagmen or protection is not available, construction work shall not be undertaken if this Work is to take place within the fouling limits. Should Contractor violate any of the conditions set forth herein, Railroad shall have the right to remedy the situation as appropriate, including suspending the Work, at the sole cost and expense of Contractor. The Contractor's employees and equipment will not be permitted to work near overhead wires or apparatus, except when protected by a Class A employee of the Railroad who will take necessary precautions for their safety before starting and during the progress of such Work. The Contractor must supply and install a grounding cable (4/0 copper or equivalent ACSR) for each piece of equipment working adjacent to any electrified lines. The ground must be an approved 'C' clamp type ground. When Contractor is working in existing electrified territory, it shall comply with the High Voltage Proximity Act, N.J.S.A. 34:6-47 et seq.
- 10. Equipment of the Contractor to be used adjacent to the tracks shall be in first-class condition so as to fully prevent failures of defective equipment that might cause delay in the operation of trains or damage to Railroad facilities. The Contractor's equipment shall not be placed or put into operation adjacent to tracks without first obtaining permission from the Railroad and NJ TRANSIT's Construction Manager. Under no circumstances shall any equipment or materials be placed or stored within eighteen (18) feet from the near rail of a track in operation.

- 11. Materials and equipment belonging to the Contractor shall not be stored on Railroad property without first having obtained permission from the Railroad NJ TRANSIT's Construction Manager and such permission will be on the condition that the Railroad and NJ TRANSIT will not be liable for damage to such materials and equipment from any cause. The Contractor shall keep tracks adjacent to the site clear of all refuse and debris that may accumulate from its operations, and shall leave the Railroad property in the condition existing before the start of its operations.
- 12. The Contractor shall consult the Railroad and NJ TRANSIT's Construction Manager in order to determine the type of protection required to insure safety and continuity of Railroad traffic incident to the particular methods of operation and equipment to be used on the Work. Any B & B Inspectors, track foremen or track watchmen, signalmen, electric traction linemen, or other employees deemed necessary for protective services by the Railroad, or its duly authorized representative, to insure the safety of trains, contingent upon the Contractor's operations, shall be obtained from the Railroad by the Contractor. The Contractor shall make all such requests through NJ TRANSIT's Construction Manager. The cost of same shall be paid by NJ TRANSIT.
- 13. The providing of such watchmen and other precautionary measures shall not, however, relieve the Contractor from liability for payment of damages caused by its operations.
- 14. The Railroad will require flagging and/or other protection of railroad traffic during all periods when the Contractor is working on or over the right-of-way of the Railroad, or as may be found necessary in the opinion of the Railroad Engineers. When protection is required the Contractor shall make the requests in writing to NJ TRANSIT's Construction Manager, who will forward same to the Railroad at least thirty (30) days before such protection is required. NJ TRANSIT shall be responsible for any compensation owing to the Railroad for such protection. Contractor shall not include the cost of such railroad protective services in its bid. However, the costs for safeguarding the tracks, barricades, lights, signs, signals warnings, other protections and services, including insurance shall be provided by the Contractor and shall be included in the bid price.
- 15. Prior to the beginning of Work, it must be determined whether the tracks near the work area must be taken out of service. The track must be taken out of service by a qualified Railroad employee when any of the following conditions exist:
  - Any construction machinery or equipment occupies the traffic envelope or is standing within 18 feet of the center line of an outside track.
  - b. Any unsecured construction materials are stored within 20 feet of the center line of any track.
  - c. Excavations will be performed under operating tracks or adjacent to where stability of tracks may be affected. Under no circumstances will excavations be permitted within the "track live load influence line". The live load influence occurs when an excavation nearest the adjacent track intersects a line from a point five feet horizontally from center line of adjacent track at the plane of the base or rail drawn on a slope of 1-1/2 foot horizontal to one foot vertical. If the excavation occurs within this perimeter, then temporary earth support plans, designed and sealed by a registered professional engineer, shall be

- submitted for approval. In any event, the excavation shall be no less than 25 feet from adjacent track, unless otherwise approved by the Railroad. Excavations shall be fenced, lighted, and otherwise protected as directed by the Railroad.
- d. Any other conditions, circumstances, or situation that may present a danger to the safe movement of trains.
- 16. It shall be expressly understood that this Contract includes no work for which the Railroad is to be billed by the Contractor, and it shall be further understood that the Contractor is not to bill the Railroad for any work which the Contractor may perform.
- 17. Upon completion of the work and as a condition of Final Acceptance, the Contractor shall remove from within the limits of the Railroad's right-of-way, all machinery, equipment, surplus materials, false work, rubbish and temporary buildings and other property of the Contractor, or Subcontractor, and shall leave the right-of-way in a condition satisfactory to the Railroad and NJ TRANSIT's Construction Manager.
- 18. Contractor notices for assignment of Railroad personnel and other written requests shall be directed to the Railroad through NJ TRANSIT's Construction Manager.
- 19. Crossing of tracks at-grade by equipment and personnel is prohibited except by prior arrangement with the Railroad.
- 20. All tunneling, jacking and boring operations within the railroad track influence lines shall be performed on a 24 hour/day basis to minimize the Railroad's exposure to construction hazards.
- 21. No Work across, over, under or adjacent to the Railroad shall commence until the Contractor's written notice is received and approved by the Railroad and all required personnel have attended the Railroad's safety class. Thereafter, rail protective personnel will be assigned, as required, for the Work.

## 4.12 WORK FURNISHED BY OTHERS

- 4.12.1 NJ TRANSIT may, and reserves the right to, enter upon the work site, or areas adjacent thereto, at any and all times during the progress of the work, or cause others to do so, for the purpose of performing work not included in these Contract Documents.
- 4.12.2 When such additional work is to be performed, the Contractor shall conduct its work so as not to interfere with or hinder the progress or completion of the work being performed by others. Moreover, the Contractor assumes the positive obligation of cooperating with such others and coordinating its activities with theirs. If there is a difference of opinion as to the respective rights of the Contractor and others doing work within the limits of or adjacent to the Project, NJ TRANSIT will decide as to the respective rights of the various parties involved in order to secure the completion of NJ TRANSIT's work in general harmony and in a satisfactory manner. NJ TRANSIT's decision shall be final and binding on, and shall not be cause for claims by the Contractor for additional compensation.

- 4.12.3 The Contractor shall assume all liability, financial or otherwise, in connection with this Contract and hereby waives any and all claims against NJ TRANSIT for additional compensation that may arise because of inconvenience, delay, or loss experienced by the Contractor because of the presence and operations of others working within the limits of or adjacent to the Project.
- 4.12.4 The Contractor will not be held responsible for damage or loss to work performed on the Contract or on other contracts within or adjacent to the site of the Project that may be caused by or on account of the work of others. The Contractor will be held responsible for any damage or loss done or caused by its work or forces to the work performed by other contractors within or adjacent to the site of the Project and it shall repair or make good any such damage or loss in a manner satisfactory and without cost to NJ TRANSIT.
- 4.12.5 The Contractor shall examine work or materials not included in this Contract, the installation of which will affect the work in this Contract, and should the same be imperfect, incorrect or insecure, it shall notify the Construction Manager immediately in order that the same may be rectified. The Contractor shall arrange its work and shall place and dispose of the materials being used so as not to interfere with the operation of others within the limits of the Project or adjacent thereto. The Contractor shall join its work with that of the others in an acceptable manner and shall perform it in proper sequence to that of the others.

### 4.13 ARTS IN TRANSIT

4.13.1 Art is included in NJ TRANSIT's capital program to enhance the appearance of NJ TRANSIT's buildings and infrastructure. As such, this Contract may incorporate art as designated by NJ TRANSIT.

NJ TRANSIT will be responsible for the selection of the Artist(s) and the Contractor shall enter into and abide by the Artist Agreement contained in Appendix E, Agreement between Contractor and Artist. The Contractor shall place no other contract requirements or conditions upon the Artist(s). The Contractor shall submit a copy of the executed agreement to the Construction Manager within fifteen calendar days after its execution. NJ TRANSIT retains the right to direct the Contractor to amend the Artist Agreement under the terms of Article 3.1, Change Orders.

4.13.2 The Contractor shall prosecute its Work so that installation of Artwork shall proceed in the manner and within the scheduled times directed by NJ TRANSIT and as incorporated in the Artist Agreement. The installation of the Artwork shall be included in the Contractor's Construction Progress Schedule.

Should the subject Artwork not be deemed substantially complete by NJ TRANSIT within the specified time allotted in the Artist Agreement, except for causes beyond the Contractor's and Artist's control, NJ TRANSIT may deduct from the amount due the Contractor \$100.00 per calendar day of delay as Artwork Liquidated Damages. The Artwork Liquidated Damages are to be separate from the Contract liquidated damages set forth in Article 2.1, TIME OF COMPLETION, DELAY, LIQUIDATED DAMAGES.

The Contractor shall not be charged with the Contract liquidated damages when the delay in substantial completion of the Work is solely due to late completion of the Artwork, unless the delay is attributable to acts or omissions of the Contractor. In that event, only the Artwork Liquidated Damages shall apply.

- 4.13.3 Upon completion and installation of the Artwork, NJ TRANSIT shall inspect the Artwork and installation and shall either accept or reject the Artwork. Upon NJ TRANSIT's acceptance of the Project and as part of the Contract close-out process, the Contractor shall assign the Agreement with the Artist(s) and all rights to the Artwork to NJ TRANSIT.
- 4.13.4 Payment for the above work and services shall be made from the Allowance identified in the Bidder's Proposal under the Item entitled: "Artwork Allowance". The Contractor shall pay the Artist in accordance with Attachment A, "Fees, Material and Payment Schedule" in the Artist Agreement. The Contractor is entitled to a markup of 5% overhead and 5% profit on the Artist's contract amount identified in the Artist Agreement as "Attachment D".

### 5. MOBILIZATION AND TEMPORARY FACILITIES

- 5.1 MOBILIZATION
- 5.1.1 When the item Mobilization and General Requirements (Mobilization) is included as a Contract Item it shall consist of initiating the Contract, and shall include such portions of the following as are required at the beginning of the Project: setting up the Contractor's general plant, offices, shops, storage areas, sanitary and other facilities as required by the Specifications, by Federal, State, or local law or by regulation; providing access to the Project site; obtaining necessary permits, grants and licenses, and payment of fees; protecting existing utilities; lighting work areas; providing shop drawings; sampling and testing of materials; providing required insurance and bonds other than the Performance Bond and Payment Bond, unless Insurance Bid Items are included in the Bidder's Proposal. Mobilization shall also be deemed to include the Contractor's cost of ongoing maintenance and protection of the work premises, demobilization and remobilization as necessary to accommodate sequencing the work, and all costs associated with the provision and maintenance of temporary facilities, unless a specific Bid Item has been provided in the Bidder's Proposal for a specific element of work (e.g. Field Offices).
- 5.1.2 Payment for Mobilization as hereinbefore specified will be made for the lump sum price bid therefore, regardless of the fact that the Contractor may have, for any reason, shut down its work on the Project or moved equipment away from the Project and back again.
- 5.1.3 Except where a specific Bid Item has been provided in the Bidder's Proposal for a specific element of work, the provisions for payment of the Contract Item Mobilization supersede any provisions elsewhere in the Contract for including the costs of these initial and ongoing services and facilities in the prices bid for the various Contract Items in the Proposal.

- 5.1.4 Payment to the Contractor for the item Mobilization will be made in accordance with the following schedule:
  - (a) When five (5) percent of the work is completed 25 percent of the amount bid for mobilization or two and one half (2 1/2) percent of the total Contract Price, whichever is less, will be paid.
  - (b) When ten (10) percent of the work is completed an additional 25 percent of the amount bid for mobilization or five (5) percent of the total Contract Price, whichever is less, will be paid.
  - (c) When 25 percent of the work is completed an additional 25 percent of the amount bid for mobilization or six (6) percent of the total Contract Price, whichever is less, will be paid.
  - (d) When 50 percent of the work is completed an additional 25 percent of the amount bid for mobilization or ten (10) percent of the total Contract Price, whichever is less, will be paid.
  - (e) The percentage of work completed shall be the total of payments earned, exclusive of the amount paid for this item, as shown on the monthly certificates of the approximate quantities of work done.
  - (f) Upon completion of all work on the Project, payment for any amount bid for mobilization in excess of ten (10) percent of the total Contract Price will be paid.
- 5.1.5 When the item Mobilization is not a Contract Item, no specific payment will be made for the work included in this Article. All costs thereof shall be included in the prices bid for the various scheduled Contract Items.

## 5.2 FIELD OFFICE AND SANITARY FACILITIES

- 5.2.1 The Contractor shall provide a field office on or as convenient to the job site as possible, subject to the approval of the Construction Manager and sufficient to accommodate NJ TRANSIT representatives assigned to the Project. Such space, together with necessary furnishings, equipment, supplies, etc., and all utilities shall be as required by this Article.
- 5.2.2 Within thirty calendar days of the Notice to Proceed, the Contractor shall provide and maintain the mobile trailer units described herein with parking facilities for five vehicles. The Field Office and the parking facilities will be for the use of the Construction Manager, Engineer/Architect, and their staff.
- The Field Office shall be a new or like new NJDCA approved weatherproof mobile trailer with a 7-foot minimum ceiling height, weatherproof windows (screened), doors each equipped with adequate locking devices, and a burglar and fire alarm system to be connected to a local 24-hour security service. The Field Office shall total at least 576 square feet and shall be divided into three rooms, one with a floor area of not less than 288 square feet and two with floor areas of not less than 144 square feet. All walls shall be paneled. The Field Office location shall be approved by the Construction Manager. The trailer shall be adequately tied down to resist high winds. The Contractor shall level the Field Office trailer and provide entrance steps, landing platforms, handrails, and under trailer enclosures as directed by the Construction Manager. The Contractor shall obtain required DCA permits and approvals for the Field Office as well as any subsequent permit renewals.

- All Field Office windows are to be protected by expanded metal grilles with angle frames which are to be through bolted top 2" x 2" x 1/4" plates. All external doors are to be heavy duty construction with cylinder locks and with two 2" x 2" x 1/4" angle bars which can be placed across the closed door and padlocked in place. Padlocks to be placed through eye bolts which are to be through-bolted to 2" x 2" x 1/4" plates. Contractor shall supply the padlocks and all keys (original and copies) to the Construction Manager.
- 5.2.5 The Contractor shall maintain and service the Field Office trailer as specified in this Article. Upon project completion, and only after receipt of written authorization from the Construction Manager, the Contractor shall remove the Field Office from the job site.
- 5.2.6 Any relocation of the Field Office trailer and utilities during the entire project duration shall be the Contractor's responsibility.
- 5.2.7 The Contractor shall provision the Field Office as follows:
  - Provide adequate lighting, electrical receptacles, and ground fault circuit interruptions as required by OSHA.
  - Provide lighting to furnish a minimum of 100 foot-candles at desk height uniformly in all areas.
  - 3. Provide heating and cooling equipment and any necessary fuel to maintain an ambient air temperature of 70 degrees F +/- 5 degrees F.
  - 4. Provide and maintain a source of hot and cold potable water for use in a flushing water closet, and for hand washing. The Contractor shall be responsible for plumbing hook-up to a sanitary line or for provision of a storage tank.
  - 5. Provide five separate phone lines; three lines to be equipped for voice; one for fax and one with modem capabilities. Provide and install the phone system with three new touch-tone phones with answering machine, speaker and hunting capabilities. All equipment shall be approved by the Construction Manager prior to installation.
  - 6. Provide OSHA required fire extinguisher.
  - 7. Furnish the CM Field Office with the following new equipment and furniture as approved by the Construction Manager:

Table 5.2.7.7: Field Office Equipment				
Description	Quantity			
Desk (60" x 30") with three lockable draws and rolling armchair	3			
Drafting table (60" x 36") with drawer and 54" straightedge and stool	1			
Reference table (54" x 30")	1			
Conference table (36" x 96")	1			
Metal folding chairs with saddle seat and steel back	10			
Storage cabinet (36" x 18" x 6') with lock and two keys	1			
Two file cabinets, four (4) drawers (legal size) with lock and key.	2			

Table 5.2.7.7: Field Office Equipment			
Two file cabinets, fireproof, four (4) drawers (legal size) with lock and key.	2		
Greensteel marker board (36" x 48"), mounted, and supply of markers including replacements as required	1		
Copier, using 8-1/2" x 11", 8-1/2" x 14", and 11" x 17" paper	1		
Two cubic feet refrigerator/freezer	1		
Wall clock (battery operated) 12" diameter face	1		
Plain paper laser and facsimile machine capable of 8½ x 11 and 8½ x 14" paper	1		
Automatic Drip Coffee machine (10 cup)	1		
Fully stocked first-aid cabinet in compliance with OSHA regulations	1		
Microwave	1		
Bookcase (36 x 42) with four shelves	1		

- 5.2.8 The Contractor shall maintain and service the Field Office in accordance with the following requirements.

  The Contractor shall:
  - Repair and clean the Field Office, including complete janitorial services, including cleaning and emptying of any temporary sanitary system, and trash removal, at a minimum frequency of once per week to the level approved by the Construction Manager.
  - 2. Repair, clean, and adjust equipment specified under Sub-article 5.2.7 and provide repair/maintenance service with 24 hour response/repair time for proper operation of all copiers, typewriters, computers, and any other office equipment whether supplied by the Contractor or supplied by others.
  - 3. During other than normal working hours, provide security measures and area protection adequate to insure the safety and integrity of the project site.
  - 4. Provide all necessary paper, including sanitary paper, and other office supplies as required by the Construction Manager.
  - 5. Provide adequate bottled water and paper cups inside the Field Office.
  - 6. Provide coffee, filters, plastic stirrers, sugar, cups, napkins, and non-dairy creamer.
  - 7. Maintain and restock the first-aid cabinet as required.
- 5.2.9 Payment for the Field Office materials and services identified in this Article shall be as follows:
  - A. Included and to be paid for under the Bid Item "Mobilization and General Requirements" shall be the costs for the following:
  - 1. Trailer site preparation and trailer delivery.
  - 2. Trailer utility and sanitary hookups.
  - 3. Trailer set-up, including: skirting, tying down, securing and making the trailer weatherproof; wooden stair and platform construction (including handrails); installation of burglar alarm system; and other miscellaneous efforts required to provide safe and orderly access to the trailer. Further,

any and all labor and materials required for repair and maintenance to the above for the duration of the project.

- 4. Obtaining and paying for any and all permits required for hauling, building and making utility connections for the trailer.
- 5. Any costs associated with the location and/or relocation, for any reason, of the Field Office and utilities.
- 6. All Field Office equipment and furnishings identified in this Article.
- 7. Trailer demobilization and removal at the completion of the project, including utility disconnections, temporary construction and disposal fees.
- 8. Any and all other costs associated with mobilizing, erecting, maintaining, repairing, demobilizing and removing the Field Office trailer and associated temporary improvements/structures.
- 9. All costs associated with Sub-article 5.2.8 Items 1, 2 and 3.
- B. Included and to be paid for under the Bid Item Allowance "Field Office" shall be the costs for the following:
- Monthly rental of the Field Office trailer.
- NJ TRANSIT telephone usage.
- 3. NJ TRANSIT electrical power usage.
- 4. NJ TRANSIT heating fuel expenses.
- 5. Office supply account covering all costs associated with Sub-article 5.2.8 Items 4, 5, 6 and 7.

The Contractor shall submit copies of invoices from the trailer rental and utility companies and receipts for office supply expenses along with the monthly applications for payment. The Contractor shall be reimbursed for the items listed in this Sub-article 5.2.9.B as a direct expense without any additional markups for overhead or profit.

All items purchased by the Contractor under Article 5.2 shall become the property of the Contractor for his use or disposition upon removal of the Field Office.

## 5.3 CONSTRUCTION SIGN

The Contractor shall construct and install construction sign(s) as indicated in the Contract Documents. Lettering shall be as shown in the Contract Documents and shall include the names of the Contractors engaged on the Project and such other persons or entities as directed. The sign(s) shall be securely installed to remain rigid and plumb, shall be maintained in good condition throughout the construction period, and shall be removed when directed by the Construction Manager. If the Contractor desires to install a sign other than those specified in the Contract Documents it shall first obtain the approval of the Construction Manager.

#### 5.4 TEMPORARY WATER

- 5.4.1 The Contractor shall provide, protect and maintain an adequate water supply for use on the Project during the period of construction, either by means of the permanent water supply line, or by the installation of a temporary water supply line. This water supply line shall be made available within 15 days after written authorization to proceed with the Project. If the source of water supply is a well, provisions covering the supply of water will include the installation of necessary power driven pumping facilities by the Contractor, as well as protection of well from contamination. The water supply shall be tested periodically by the Contractor and, if necessary, shall be chlorinated and filtered.
- 5.4.2 The Contractor will be required to install a valved temporary water supply connection at a point approximately 10 feet from the building or buildings and provide a meter, if required; the actual location of the point to which the water is brought shall be determined by the Contractor.
- 5.4.3 If there is a charge for water, said charges shall be paid by the Contractor. When temporary water lines are no longer required they shall be removed by the Contractor and any part, or parts, of the grounds or building disturbed or damaged shall be restored to the original condition by the Contractor. The Contractor shall install its permanent water lines to the boiler room and heating equipment in sufficient time to be available for supplying water for testing and operation of the heating system when needed to supply heat on the Project.

## 5.5 TEMPORARY LIGHT AND POWER

- 5.5.1 The Contractor shall extend electrical service to the building or buildings at locations approved by the Construction Manager; temporary electrical service shall be independent of the existing permanent service. Initial temporary service shall be three phase or single phase depending upon which phase is nearest to the Project site. This service shall be installed within 15 days after written Notice to Proceed with the Project. When the Contract calls for three phase permanent service, the Contractor shall install same within a reasonable time to permit use by all the trades.
- 5.5.2 The Contractor shall extend the service into the building and shall provide such receptacles and lighting as required for the proper conduct of the work.
- 5.5.3 The Contractor shall pay for cost of all electric energy used, and it shall also maintain and service any electrical equipment installed and necessary for maintaining heat after same is required in the building.
- 5.5.4 When the temporary electrical lines are no longer required they shall be removed by the Contractor and it shall restore to their original condition any part, or parts, of the grounds or building disturbed or damaged.
- 5.5.5 Any Contractor who fails to carry out its responsibility in the supplying of uninterrupted light and power to expedite the Project, as set forth in this Contract, shall be held responsible for such failure and the

Contracting Officer shall have the right to take such action as the Contracting Officer deems proper for the protection and conduct of the work and shall deduct the costs involved from the amount due the Contractor.

## 5.6 TEMPORARY HEAT

- 5.6.1 The Contractor shall provide, protect and maintain, at its own expense, sufficient heat to the Project during the entire period of construction either by using an NJ TRANSIT approved method of temporary heat or, when operational, the permanent heating system.
- Prior to any building being enclosed by walls and roof, if the outside temperature shall fall below 40 degrees F., at any time during the day or night, and the work in progress requires heat for execution and protection, the Contractor shall furnish acceptable means to provide sufficient heat to maintain a temperature of 40 degrees F., for that portion of the work which requires same.
- 5.6.3 Heating of field office, storage spaces, concrete and masonry materials and working area heating required prior to enclosure, as specified herein, shall be provided by the Contractor as specified in the Contract Documents.
- As soon as the building, or a major unit thereof, is generally enclosed by walls and roof, as determined by the Construction Manager, the responsibility for supplying working area heat shall rest with the Contractor. When the outside temperature falls below 40 degrees F., at any time during the day or night, the Contractor shall furnish sufficient heat, by the use and maintenance of LP gas heaters or other system approved by the Construction Manager, to maintain a temperature of 45 degrees F. within the enclosed area of the building at all times and shall remove same when no longer required. The Contractor shall provide or arrange at its own expense supervision of the LP gas heaters at all times prior to start of the permanent heating system. The Contractor shall furnish and pay for all fuel required for the above temporary installation during the term of this contract.
- 5.6.5 The Contractor will be held responsible for freeze ups following enclosure of the building. The Contractor shall remove soot, smudges, and other deposits from walls, ceilings, and exposed surfaces which are the result of the use of heating equipment including the permanent heating system during the period of its use for supplying heat. The Contractor shall not do any finish work until the areas are properly cleaned.
- A building, or major unit thereof, shall be considered "enclosed" when: (1) the exterior walls have been erected; (2) temporary roof or permanent roof is installed and in watertight condition; and (3) temporary or permanent doors are hung and window openings are closed with either permanent or temporary weather tight enclosures (cardboard, muslin and light canvas materials are not acceptable; any impervious transparent material is acceptable). A major unit of building as referred to herein shall be: (1) an entire separate structure; (2) a fully enclosed wing which shall have a floor area equal to at least 50 percent of the

total floor area of the Project; or (3) a section which shall have a floor area equal to at least 50 percent of the total floor area of the Project.

- 5.6.7 60 calendar days after the building, or major unit thereof, is enclosed and the Engineer has determined that heat is required for the proper execution of the construction work, the permanent heating system shall provide the heat. Regardless of whether the boiler room is within the confines of the major unit or not, it shall be enclosed and the floor installed at the time the permanent heating system shall supply heat. The boiler room floor area shall not be considered in determining the area comprising the major unit. The 60 day period shall apply only to the enclosed portion of this building.
- 5.6.8 The Contractor shall continue to provide acceptable means of temporary heat until the permanent heating system is operational. If the permanent heating system is not acceptable to the Construction Manager for providing sufficient heat, the Contractor shall continue to provide temporary heat as described above and as ordered by the Construction Manager or NJ TRANSIT.
- 5.6.9 When the heating system provided by the Contractor is designed for tie-in to existing steam lines for source of heat, NJ TRANSIT will provide steam for temporary heat through the Project's permanent heating system at no additional cost to the Contractor. The Contractor shall arrange, at its own cost, for connections.
- Valves, traps and other parts of the heating system which are permanently installed by the Contractor and used for supplying heat during the construction period need not be replaced, provided the system was in acceptable condition prior to its use, and further, that the system is properly cleaned and adjusted to operate after the permanent system is in use to the satisfaction of the Construction Manager. Seven (7) days prior to acceptance by NJ TRANSIT of the heating system as substantially complete, the Contractor shall replace disposable filters or turn over spare sets of filters to NJ TRANSIT.

## 5.7 TEMPORARY PARTITIONS, ENCLOSURES, GLAZING BREAKAGE AND CLEANING

- 5.7.1 Whenever necessary, in order to maintain proper temperatures for the prosecution of the work, or for the protection thereof, the Contractor shall furnish and maintain temporary enclosures and partitions. All openings in exterior walls not enclosed with finishing materials shall be closed temporarily. Window sashes may be installed and glazed. Temporary wood doors shall be provided at door openings. Temporary partitions shall be securely anchored, stable, well-constructed and maintained, and fit for the purpose intended, ie. Work area separation, protection of the public, delineation of pedestrian pathways, etc.
- 5.7.2 The Contractor shall be responsible for all breakage of glazing after same has been installed, no matter by whom or what caused, and shall replace all broken, scratched or otherwise damaged glazing before the completion and acceptance of the work. The Contractor shall wash all glazing on both sides at completion, or when directed, removing all paint spots, stains, plaster, etc.

5.7.3 The Contractor shall provide and maintain necessary temporary dustproof partitions around areas of work in any existing building.

## 5.8 TEMPORARY, INTERMEDIATE AND HIDDEN WORK

- The Contractor shall be responsible for temporary, intermediate and hidden work, including the furnishing and setting of sleeves, built in items, anchors, inserts, and chases for its work. The Contractor shall build these items into the construction. The Contractor shall build recesses, channels, chases, openings, and flues, and leave or create holes where shown on Drawings or where directed for steam, water or other piping, electrical conduits, switch boxes, panel boards, flues and ducts, or other features of the heating and ventilating work. Subcontractors requiring such recesses, channels, chases, openings, and flues shall furnish to the Contractor complete details and drawings of such as required in connection with the work. Such information shall be furnished in complete form and in ample time to allow the construction work to proceed without interruption or delay. These details and drawings shall be furnished in accordance with Article 6.5- SHOP AND WORKING DRAWING SUBMITTALS to the Construction Manager for review and approval prior to installation.
- 5.8.2 The Contractor shall close, build in, and finish around or over openings, chases, channels, pockets, and sleeves after installation has been completed.
- 5.8.3 Positive instructions in writing shall be obtained from the Engineer before cutting or boring floor beams, floor constructions, or supporting members.

#### 5.9 DEMOBILIZATION

At the completion of the Work and prior to final payment, the Contractor shall remove temporary facilities entirely from the site including, but not limited to the following: Field offices, trailers, shanties, sheds, temporary electric services, temporary water hydrants, temporary fences, project sign, job telephone, temporary roads, temporary toilets, temporary enclosures, dust barriers, and other temporary protection devices. The Contractor shall conduct final cleaning activities and restore all disturbed landscaping, street and sidewalk surfaces, subsurfaces and overhead structures, if any. Should the Contractor fail to remove such temporary facilities and restore disturbed conditions, NJ TRANSIT shall perform such activities as necessary and deduct the cost from the Contractor's final payment.

### 6. PROJECT ADMINISTRATION AND DOCUMENT CONTROL

#### 6.1 PROJECT MEETINGS

6.1.1 The Contractor, subcontractor, supplier or vendor whose presence is necessary, unless excused in writing by the Construction Manager, shall attend project meetings when called by the Construction Manager for the purpose of discussing the execution of the Work. The initial pre-construction meeting will generally be held prior to commencement of the work at a time, date and location to be set by the Contracting Officer.

## 6.1.2 General Requirements for Project Meetings:

- A. One of the persons designated by the Contractor to attend and participate in the project meetings shall have all required authority to commit the Contractor to solutions agreed upon in the project meetings.
- B. To the maximum extent practicable, advise the Construction Manager at least 24 hours in advance of project meetings regarding all items to be added to the agenda.
- C. The Construction Manager will compile the official minutes of each project meeting and will furnish three (3) copies to the Contractor.
- D. Except as noted below for the Pre-Construction Meeting, Project Meetings will be held once every two weeks. The Contractor and Construction Manager shall coordinate as necessary to establish a mutually acceptable schedule for meetings.

### 6.1.3 Pre-Construction Meeting:

- A. A pre-construction meeting will be scheduled by NJ TRANSIT. The Contractor shall provide attendance by an authorized representative and authorized representatives of all major subcontractors. The Construction Manager will advise other interested parties and request their attendance. The Construction Manager and the Contractor will arrange to review details of construction, and if appropriate, to walk the project with the Contract Drawings in hand and carefully observe all pertinent conditions relating to the construction of the Work, including the status of right-of-way, existing structures and obstructions to be removed, altered or changed.
- B. Minimum Pre-Construction Agenda: The Contractor shall be prepared to discuss:
  - (1) Organizational arrangement of Contractor's forces and personnel, and those of subcontractors, materials suppliers, Engineer and Construction Manager.
  - (2) Established channels and procedures for communications as approved by NJ TRANSIT.
  - (3) Construction schedule, including sequence of critical work as described in Article 6.2-CONSTRUCTION PROJECT SCHEDULE.
  - (4) Contract Documents, including distribution of required copies of original documents and revisions.
  - (5) Processing of shop drawings and other data submitted to the Construction Manager for review.
  - (6) Processing of field decisions and contract change orders.
  - (7) Rules and regulations governing performance of the Work.
  - (8) Procedures for safety and first aid, security, quality control, housekeeping, and other related matters.
  - (9) Existing conditions.
  - (10) Equal employment regulations.
  - (11) DBE requirements.

- (12) Quality assurance.
- (13) MSDS submittal requirements as set forth in Article 9.7- ENVIRONMENTAL COMPLIANCE AND LIABILITY.
- (14) Subcontractor submittals and approvals as set forth in Article 1.9- ASSIGNING AND SUBCONTRACTING CONTRACT. The Contractor shall provide the initial submittals noted therein.
- (15) Contract Completion and liquidated damages.

## 6.1.4 Project Meetings:

- A. Attendance: To the maximum extent practicable, assign the same person or persons to represent the Contractor and major Subcontractors, as requested by NJ TRANSIT, at project meetings throughout progress of the Work. If requested by NJ TRANSIT, subcontractors, material suppliers, and others shall attend those project meetings in which their aspects of the Work are involved.
- B. Minimum Agenda for Project Meetings:
  - (1) Review, revise as necessary, and approve minutes of previous meeting.
  - (2) Questions and issues unresolved at the previous Project Meeting.
  - (3) Engineer's, Construction Manager's and/or Contractor's unsatisfied request for information.
  - (4) Work accomplished since the previous Project Meeting, off-site fabrication problems, product delivery problems, proposed changes, and other circumstances which might delay progress of the Work.
  - (5) Corrective measures and procedures developed to regain planned and scheduled progress.
  - (6) Field observations, problems, Engineer's or Construction Manager's decisions, work quality, and employee work standards.
  - (7) Plan of the following month's Work.
  - (8) Status of DBE subcontractors.
  - (9) NPC and Change Order statuses.
  - (10) CPM status.
  - (11) Submittal schedule.
  - (12) Safety.
  - (13) Others, as required.

## 6.2 CONSTRUCTION PROJECT SCHEDULE

6.2.1 The Contractor shall be responsible for preparing and furnishing, at the pre-construction meeting, an initial draft of a coordinated combined project schedule that incorporates the project schedules of the Contractor and its subcontractors activities for the prosecution of the work. The schedule shall be a CPM (Critical Path Method) schedule in sufficient detail satisfactory to the Construction Manager.

- Float, or slack time, in the schedule is defined as the amount of time between the early start date and late start date or the early finish and late finish date of any activity. The definition of float or slack time also includes the amount of time between the late finish date of the Contractor's schedule and the time for completion specified in the Contract Documents, if the Contractor's scheduled late finish date is earlier than the Contract Time. Float or slack time is not for the exclusive use or benefit of either the Contractor or NJ TRANSIT, but for the overall benefit of the project as determined by NJ TRANSIT. Extensions of time for performance under any and all of the provisions of this Contract will be granted only to the extent that such equitable time adjustments for the activity or activities affected exceed the total float along the paths involved at the time the delay occurred or notification was issued for the change. The Contractor shall not sequester shared float through such strategies as excessively extending durations, artificially constraining resources, or introducing faulty logical relationships between schedule activities.
- 6.2.3 The Contractor shall prepare and maintain the Contract Schedule by the use of skilled and experienced scheduling personnel; each with at least five years experience or the equivalent thereof in detailed scheduling. Such personnel shall be directly involved in the planning, scheduling, evaluating, and progress reporting of the work. The Contractor shall submit the qualifications of the scheduler/scheduling consultant for approval at the Pre-Construction Meeting. Should the scheduler/scheduling consultant's qualifications prove unacceptable to NJ TRANSIT, the Contractor shall submit the qualifications of a substitute scheduler/scheduling consultant within seven (7) days of NJ TRANSIT's rejection of the originally proposed personnel.
- No later than 10 days after the Notice to Proceed, Contractor shall submit to the Construction Manager a Detailed Project Schedule (DPS). This DPS shall outline all activities and sequences of operations, as needed, for the orderly performance and timely completion of all work in accordance with the Contract, commencing with the Notice to Proceed and concluding with the Contract completion. The schedule should take into account mandatory sequencing, phasing, and restrictions of access to the Project Area, if any. The DPS is required to ensure adequate planning and scheduling of the work by Contractor and to enable the Construction Manager to evaluate work progress and to make progress payments. No progress payments (excluding payments for mobilization) will be made until a DPS is approved by the Construction Manager.
- 6.2.5 Within 14 days of receipt of Contractor's DPS, the Construction Manager will review the schedule for conformance with the Contract and provide the Contractor with the Construction Manager's comments. The Contractor shall incorporate the Construction Manager's comments into the DPS and shall resubmit the DPS to the Construction Manager within seven days of receipt of such comments. Contractor shall repeat this process (at its own expense) until the Construction Manager approves the DPS.

- Upon approval by the Construction Manager, the Contractor's DPS shall become the Baseline Schedule for the work. This schedule shall be used by Contractor for planning, scheduling and executing the work, for monitoring and reporting progress to the Construction Manager, and as a basis for progress payments. Progress shall be shown in terms of remaining duration, actual dates and percent complete for each activity. During the life of this Contract, Contractor shall make monthly progress updates to the DPS. The updated DPS reflecting progress through the end of the month, as determined by NJ TRANSIT during the schedule meetings, shall be submitted by the fifth work day of the following month. Under no circumstances at any time during the project shall the Contractor make any changes to the NJ TRANSIT-approved Baseline Schedule logic, durations and construction sequencing without first receiving the written approval of the Construction Manager.
- 6.2.7 The DPS shall be a CPM schedule prepared with the software "Primavera", latest version, using the precedence diagram method. The DPS shall show a clear and definable critical path for the work. All imposed or constrained dates shall be clearly identified. The DPS shall include all contractual milestones and activities for the complete scope of the work including interface activities with the Railroad and other parties such as utility companies and outside agencies. Contractor's activities shall delineate the individual components of the work such as design efforts, submittals, procurement activities, fabrication, deliveries, construction operations, application and receipt of permits, track usage requirements, and testing. For each activity in the DPS, Contractor shall include:
  - 1. Description, which shall clearly describe the operation and the location where it is occurring.
  - Durations, which shall be expressed in workdays. Durations shall not exceed 20 workdays except in the case of non-construction activities such as a procurement of materials, fabrication and delivery of equipment or other such activities. Durations shall include allowances for lost time and inefficiencies. Activities that have started shall show the remaining duration.
  - 3. Activity code, which will be utilized to allow for breakdown of the total schedule by work area, phase of work, activity type, etc. A responsibility code (as part of the activity code), shall individually and singularly denote Contractor, each subcontractor, the Railroad, outside agencies, utilities, and any other parties performing the activity.
  - 4. The number of person-hours required performing the activity. The number of person-hours shall be shown as a resource using integers.
  - The percent complete using integers, which represents the activity's progress as of the status date.
  - 6. The actual start and finish dates.
- 6.2.8 Every DPS Submittal shall include a 3-1/2 inch diskette containing the Contract schedule and all related files generated by the "Primavera" back up utility and shall include five hard copies of the following graphical and tabular reports:

### Graphical:

a. Activity Bar Chart (ABC), on 8-1/2 inch by 11 inch or 11 inch by 17 inch paper with activities grouped by work areas and sorted by early start.

### Tabular:

- a. Activity Listing Report (ALR), sorted by activity identifier and including predecessor activities, successor activities, resources, and allocated dollar amount. Show constraint dates on a separate line.
- b. Total Float Report (TFR), sorted by total float with a secondary sort by early start and including predecessor activities and successor activities.
- c. Early Start Report (ESR), sorted by early start with a secondary sort by total float.
- d. Cost Control Activity Report (CCAR), sorted by activity identifier and including the dollar amount earned to date for each activity (to be used as invoice back up).
- e. Predecessor-Successor Report (PSR), sorted by activity number.
- 6.2.9 Every DPS Submittal shall include a written Narrative Report explaining the CPM schedule and the Contractor's approach for meeting the interim and completion milestones. This report will include an analysis and summary of the contents of the computer reports and will address, as a minimum, the following:
  - Description of the project status.
  - Critical path analysis which takes into account construction sequencing, major procurement items
    that may influence the critical path, activities that influence interim contract milestones, and NJ
    TRANSIT approved constraint dates.
  - Total float
  - 4. Schedule slippage, including a comparison to the previous month's status.
  - 5. List of activities that may become critical within the next 30 day period.
  - 6. Logic revisions/other changes as approved by NJ TRANSIT.
  - 6. If the project falls behind, the measures the Contractor will take to get the project back on schedule.
- In the event that it is necessary for Contractor to revise the durations, construction sequencing or logic of the DPS, the revised DPS shall be submitted to the Construction Manager for approval, at no additional cost to NJ TRANSIT. Minor changes to the DPS, such as re-sequencing of activities, may be approved at a Project or Schedule Meeting; a minor change is not considered a revision in the context of this Paragraph. However, a revision shall incorporate all previously made changes, major or minor, to reflect current as-built and as-planned conditions. Reasons for revisions may include, but are not limited to, the incorporation of an approved change order or changes required to recover lost time if the Construction Manager determines that work is not progressing in accordance with the Baseline Schedule. In the case of minor changes or revisions that were made to improve Contractor's work progress and are not part of a change order, the

monetary value of the activities in the revised portion of the schedule shall be identical, in aggregate value, to the value of that same work as reflected in the initial Baseline Schedule.

- NJ TRANSIT will conduct Schedule Meetings as necessary with the Contractor to review and discuss the schedule. Schedule Meetings will generally be held as part of a progress meeting. If necessary, they shall be held as separate meetings. All Schedule Meetings shall be attended by the Contractor's Project Manager or a designee, who shall have the authority to make decisions on behalf of, and commit the resources of, the Contractor. The Contractor's superintendent and appropriate scheduling staff shall also attend the meetings. At these meetings, NJ TRANSIT will examine and comment on the Contractor's DPS. Schedule slippages will be analyzed and corrective actions will be discussed and agreed upon.
- The Construction Manager will plan the Schedule Meetings so that, regardless of frequency, there will always be a Schedule Meeting taking place on or about the 25th day of the month. During this "monthly" Schedule Meeting, in addition to the in-depth review of the DPS, the Project progress (i.e., completed activities and percent complete of partially completed activities) shall be presented by the Contractor and reviewed by NJ TRANSIT. NJ TRANSIT shall determine the percent of work complete and advise the Contractor accordingly. Subsequent to this meeting, the DPS shall be updated with the progress, as determined by NJ TRANSIT, and the Contractor shall submit the monthly Narrative Report and the revised DPS to the Construction Manager as part of its Payment Application. NJ TRANSIT's review of the DPS and Narrative Report shall not constitute NJ TRANSIT's approval of any Contractor changes to the logic, durations and construction sequencing of the previously approved Baseline Schedule unless NJ TRANSIT specifically confirms in writing its acceptance of such changes.
- In the event of a change order, the Contractor must clearly demonstrate how it proposes to incorporate the change order into the schedule. The Contractor shall provide, as part of its change order documentation and prior to change order negotiations, a schedule that clearly identifies the newly introduced change order work activities, the CPM path(s) affected and a narrative explaining the schedule impact of the change order to the DPS. If Contractor fails to notify the Construction Manager of the schedule changes associated with a change order, it will be deemed an acknowledgment by Contractor that the change order has no impact on the schedule.
- 6.2.14 All change order work activities shown in the schedule are considered to be tentative unless a Directive Letter or Change Order has been issued incorporating the changed Work into the Contract. Acceptance of a schedule containing change order work activities will not be construed to be approval of the value of the change, the duration of the work or constraints concerning the changed activities. The applicable Directive Letter or Change Order shall govern the monetary value and Contract Time impact of the changed work.
- 6.2.15 The Contractor shall furnish sufficient labor, plant and equipment to insure the prosecution of the work in accordance with the approved Project Schedule. If, in the opinion of the Construction Manager, the

Contractor falls behind in the prosecution of the Work as indicated in the Project Schedule, the Contractor shall take such steps as may be necessary to improve its progress. The Construction Manager may require the Contractor to increase the number of shifts, days of work, and/or the amount of plant and equipment, all without additional cost to NJ TRANSIT.

## 6.3 CONTRACT DRAWINGS AND SPECIFICATIONS

- Unless otherwise provided in the Contract Documents, NJ TRANSIT will furnish to the Contractor, free of charge, a maximum of six (6) full size copies of conformed Drawings and Specifications for the execution of the work. The Contractor shall at all times keep one copy of all Contract Documents up to date and in good order, available to the Engineer and to the Engineer's representatives. The Contractor shall keep its prints of the Contract Drawings up to date at all times by marking on them the final location of any changes in the Work. Prior to final payment the Contractor shall submit a copy of the marked-up drawings of all Contract Drawings whether altered or not to the Construction Manager. These marked up As-Built Drawings shall become the property of NJ TRANSIT.
- 6.3.2 The Engineer or NJ TRANSIT may furnish additional detail instructions to the Contractor through the Construction Manager, by means of supplemental drawings or otherwise, necessary for the proper execution of the work. Such drawings and instructions shall be consistent with the Contract Documents and reasonably inferable therefrom. The work shall be executed in conformity therewith and the Contractor shall do no work without proper drawings and instructions.
- 6.3.3 When the Contractor requests clarifications of Contract Drawings and Specifications it must give written notice to the Construction Manager with at least 14 calendar days lead time for the Construction Manager and Engineer to provide timely instruction or interpretation.
- 6.3.4 All Drawings referred to, together with such supplementary details as may be furnished or approved from time to time as the work progresses, are understood as being included in and a part of the Contract.
- 6.3.5 Dimensioned and full size drawings shall take precedence over scaled dimensions. Where the work is shown in complete detail on only half or a portion of a drawing or there is an indication of continuation, the remainder being shown in outline, the work drawn out in detail shall be understood to apply to other like portions of the Project.

#### 6.4 GENERAL REQUIREMENTS FOR SUBMITTALS

6.4.1 The Contractor shall make all submittals required by the Contract Documents, and revise and resubmit as necessary to establish compliance with the specified requirements. Individual requirements for submittals are described in the Technical Provisions of these Specifications. Prior to each submittal, the Contractor shall carefully review and coordinate all aspects of each item being submitted and verify that each item and

the submittal for it conforms in all respects with the requirements of the Contract Documents. By affixing the Contractor's signature to each submittal, the Contractor certifies that this coordination has been performed.

- No later than thirty-five (35) days after the Notice to Proceed, and before any items are submitted for review, the Contractor shall submit to the Construction Manager two (2) copies of the schedule described below.
- 6.4.3 The Contractor shall compile a complete and comprehensive schedule of all submittals anticipated to be made during progress of the Work which shall include a list of each type of item for which Contractor's drawings, shop drawings, Certificates of Compliance, material samples, guarantees, or other types of submittals are required. Upon review and approval of the Submittal Schedule by the Construction Manager, the Contractor will be required to adhere to the schedule except when specifically otherwise permitted in writing by the Construction Manager. The submittal schedule shall be incorporated into the Construction Project Schedule specified in Article 6.2.
- 6.4.4 The Contractor shall coordinate the submittal schedule with all necessary subcontractors and materials suppliers to ensure their understanding of the importance of adhering to the approved submittal schedule and their ability to so adhere. The Contractor shall coordinate as required to ensure the grouping of submittals as described in Sub-article 6.4.11herein.
- 6.4.5 The Contractor shall, on a monthly basis, revise, update and submit the submittal schedule to the Construction Manager reflecting the actual conditions and sequences highlighting any changes from the previously approved schedule.
- The Contractor shall submit documentation such as certificates, reports, test results, delivery tickets, manufacturers literature, etc., as specified in the Technical Provisions to the Construction Manager for NJ TRANSIT's use and approval. Where contents of submitted literature from manufacturers or other submittals includes data not pertinent to the submittal, the Contractor shall clearly indicate which portion of the contents is being submitted for review. The Contractor shall submit six (6) copies of each of the various items required to the Construction Manager, except that only one (1) copy of delivery ticket will be required. Three (3) copies will be returned to the Contractor.
- 6.4.7 The Contractor shall consecutively number all submittals and accompany each submittal with a letter of transmittal containing all pertinent information required for identification and checking of submittals to the satisfaction of the Construction Manager. The Contractor shall on at least the first page of each copy of each submittal, and elsewhere as required for positive identification, clearly indicate the submittal number in which the item was included.
- 6.4.8 When material is resubmitted for any reason, the Contractor shall transmit under a new letter of transmittal.

  All resubmittals shall carry the same submittal number as the original submittal except that an appendage

".01", ".02", ".03", etc. shall be added to indicate that the material is a first, second, third, etc. resubmission. For example, submission 177.01 would indicate the first resubmission; 177.02, would indicate a second resubmission; and 177.03, would indicate a third resubmission, etc.

- 6.4.9 The Contractor shall maintain an accurate submittal log for the duration of the Contract, showing current status of all submittals at all times and make the submittal log available for the Construction Manager's review upon request.
- 6.4.10 The Contractor shall, prior to submittal, use all means necessary to fully coordinate all material including, but not necessarily limited to:
  - (a) Determining and verifying all interface conditions, catalog numbers, and similar data.
  - (b) Coordinating with other trades as required.
  - (c) Clearly indicating all deviations from requirements of the Contract Documents.
- 6.4.11 Unless otherwise specified, the Contractor shall make all submittals in groups containing all associated items to ensure that information is available for checking each item when unit is received. Partial submittals may be rejected by the Construction Manager as not complying with the provisions of the Contract Documents and the Contractor shall be strictly liable for all delays by such non-compliance.
- 6.4.12 The Contractor shall make all submittals in advance of schedule dates for installation to provide sufficient time required for reviews, for securing necessary approvals, for possible revisions and resubmittals, and for placing orders and securing delivery. All submittals shall be made within the first six (6) months of the Project.
- 6.4.13 In scheduling, the Contractor shall allow at least thirty (30) calendar days from receipt of the submittal for review. The Construction Manager will stamp all submittals "Received", and the date so stamped shall be the official receipt date. Delays caused by tardiness in receipt by the Construction Manager of submittals will not be an acceptable basis for extension of the Contract Time.
- 6.4.14 The Engineer's review of submittals will be general, but should not be construed:
  - (a) As permitting any departure from the Contract requirements.
  - (b) As offering relief from the responsibility for any errors, omissions or negligence in the preparation by the Contractor of details, dimensions, materials, etc.
  - (c) As approving departures from details furnished by the Engineer, except as otherwise provided herein.

- 6.4.15 The Contractor shall take responsibility for and bear all cost of damages which may result from the ordering of any material or from proceeding with any part of the work prior to the final review by the Engineer of necessary submittals, including Shop and Working Drawings and all other required submittals.
- 6.4.16 Full compensation for furnishing all submittals shall be considered as included in the payments for the Contract Items to which such submittals relate and no additional compensation will be allowed therefor.
- 6.4.17 The provisions of Article 6.4 apply to all submittals.

### 6.5 SHOP AND WORKING DRAWING SUBMITTALS

- 6.5.1 The Contractor shall submit, with such promptness as to cause no delay in the work, a reproducible and five (5) legible copies and one sepia of all completed and detailed shop, setting or working drawings, details and schedules as are necessary to adequately perform the Work to the Construction Manager for review as to conformance to the design. By approving and submitting shop drawings, the Contractor thereby represents that it has determined and verified field measurements, field construction criteria, materials, catalog numbers, and similar data, or will do so, and that it has checked and coordinated each shop drawing with the requirements of the Work and of the Contract Documents. Drawings submitted by the Contractor on behalf of subcontractors shall have been checked by the Contractor before being submitted.
- The Engineer will review the shop and working drawings within 30 days. The sepia and three (3) copies will be returned to the Contractor reviewed and with comments. The Contractor shall make corrections if required by the Engineer and resubmit a reproducible and five (5) copies for approval. After final approval of the Drawings has been received, the Contractor shall immediately send the Engineer a minimum of three (3) prints of the finally approved drawings, plus the required number of approved prints each to every other affected Contractor. The Contractor shall prepare all work and shop drawings on sheets measuring 24 inches by 36 inches unless otherwise approved by the Construction Manager. The Contractor shall make all shop drawings and working drawings accurately to a scale sufficiently large to show all pertinent aspects of the item and its method of connection to the Work. The Contractor shall provide each drawing with a blank area 5 inches by 5 inches, located adjacent to the title block, and labeled as shown in the following Table. Failure to comply with these instructions will be sufficient reason to return such drawings to the Contractor without any action being taken. The title block shall display the following:

Table 6.5.2 Shop Drawing Labeling
Contract Number and Name
Number and Title of the Drawings
Date of Drawing and Revision Number
Name of Contractor and Subcontractor submitting Drawing
Clear identification of contents and location or work
Specification Article Number
Name; New Jersey State Registration Number and seal of professional Engineer certifying the drawings if engineering computations are involved or if original design work is depicted
Submittal Number

- 6.5.3 The Contractor's shop drawings shall show the general arrangement and such details as are necessary to provide a comprehensive description of the work to be performed. Shop Drawings shall consist of, but are not limited to, fabrication and erection drawings, schedule drawings, manufacturer's scale drawings, wiring and control diagrams, cuts of entire catalogs, pamphlets, descriptive literature, performance and test data.
- Shop drawings for steel structures shall consist of shop, erection, and other drawings, showing details, dimensions, sizes, and other information necessary for the complete fabrication and erection of the metal work. Shop drawings for concrete structures shall consist of such additional detailed drawings as may be required for the prosecution of the work and may include drawings of falsework, bracing, centering, formwork, and masonry layout diagrams. The Contractor shall check completely the rod lists and details of reinforcement steel shown on the plans and shall submit complete shop drawings for the reinforcement steel to the Construction Manager for the Engineer's review. Material specification designations for the various components of the structures shall be noted on the drawings. If structural steel is scheduled for payment on the basis of weight, shop drawings for steel structures shall include a shop bill of material on each individual drawing showing pertinent information including weights of items together with the total weight of steel for that shop drawing.
- 6.5.5 Working Drawings shall consist of, but are not limited to, plans for temporary structures such as decking, temporary bulkheads, support of excavation, support of utilities, groundwater control systems and forming and falsework; for underpinning; and for such other work as may be required for construction but which does not become an integral part of the completed project. They shall be accompanied by calculations or other sufficient information to completely explain the structure or system described and its intended manner of use. The Contractor shall coordinate drawings for work on utilities, streets and other facilities which are constructed for owners other than NJ TRANSIT so that the information required by these other owners is included on the Working Drawings.
- At the time of submission the Contractor shall inform the Engineer in writing of any deviation in the shop drawings from the requirements of the Contract Documents. If drawings show variations from the Contract requirements because of standard shop practice or for other reasons, the Contractor shall describe such variations in the letter of transmittal.

Failure to describe such variation to the Construction Manager, and the Engineer's review of shop drawings, shall not relieve the Contractor of responsibility for deviation from the requirements of the Contract Documents unless the Contractor has informed the Engineer in writing of such deviation at the time of submission and the Engineer has given written acceptance of the specific deviation. Neither shall the Engineer's review relieve the Contractor from responsibility for errors or omissions in the shop drawings nor relieve the Contractor from the responsibility for executing the work in accordance with the Contract.

### 6.5.7 Drawings Not In Conformance:

- (1) If corrections to the drawings are required, each print will be marked "REJECTED" or "FURNISH AS CORRECTED" or "REVISE AND RESUBMIT" and in each case the required corrections will be shown.
- (2) Each resubmittal will be handled in the same manner and review timeframe as the first submittal.
- (3) The Contractor shall direct specific attention, in writing or on the resubmitted drawings to revisions other than the corrections requested by the Engineer or Construction Manager on previous submittals.
- (4) If any corrections indicated on the drawings constitutes a change of the Contract requirements, the Contractor shall give direct and specific notice to the Construction Manager.
- (5) Work indicated on drawings marked "CONFORMS AS NOTED" or "FURNISH AS CORRECTED" may be carried out without resubmission if progressed "As Noted" or "As Corrected".

## 6.5.8 Drawings In Conformance:

- (1) Each copy of the drawings will be identified as conforming by being stamped, "REVIEWED", and dated by the Engineer.
- (2) When Shop and Working Drawings have been completed and stamped "REVIEWED," the Contractor shall carry out the construction in accordance therewith and make no further changes therein except upon written instructions from the Construction Manager.

#### 6.6 SAMPLES SUBMITTALS

The Contractor shall furnish samples as required by the Contract Documents and as directed by the Construction Manager for review and acceptance. The work shall be in accordance with accepted samples. Such samples shall be submitted promptly to the Construction Manager, at the beginning of the work, so as to give the Construction Manager ample time to obtain approval from the Engineer. A list of samples required by the Construction Manager is for NJ TRANSIT's convenience only, and shall not be construed as limiting the number or type of samples which the Contractor shall furnish.

# 6.6.2 Procedure for Submittal of Samples:

- A. The samples submitted by the Contractor shall be of the precise article, product or material proposed to be furnished.
- B. The Contractor shall submit all samples in the quantity identified.
- C. The Contractor shall prepay all shipping charges on samples.
- D. The Contractor shall label each sample indicating the following:
  - (1) Name of Project and Contract Number;
  - (2) Name of Contractor and Subcontractor;

- (3) Material or equipment represented;
- (4) Source;
- (5) Name of producer and brand (if any);
- (6) Specification Section, article or paragraph;
- (7) Location in Project; and
- (8) Submittal Number.

### 6.7 PRODUCT AND SUBMITTALS

- 6.7.1 The Contractor shall, within 30 days after the notice to proceed date, notify the Construction Manager in writing of the names of manufacturers, products, and equipment. The Construction Manager may reject products or installed equipment not in conformance with the specifications. The Contractor shall properly submit complete identifying information, note whether the item is included in the Specifications and state Specifications Section and Paragraph. Requests for approval of alternate products and equipment (approved equals or substitutions) shall comply with the provisions of Sub-articles 6.7.3 or 6.7.4, as applicable.
- 6.7.2 Where a particular brand or manufactured product is specified, it is to be regarded as a standard. Another brand or make which meets or exceeds the specifications, in the sole discretion of the Project Manager, may be accepted, in accordance with Article 6.7.3.
- 6.7.2.1 The designs in the Contract are based on the <u>named manufacturer's product(s)</u> in each Section of the Technical Provisions. Where the Contractor proposes to use a product other than the named product(s), the Contractor shall pay all costs for modifications of the design, including all re-engineering costs and any additional construction costs associated with the use of that product. Written approval shall be obtained from the Engineer through the Construction Manager prior to any use of a product other than the named.
- 6.7.2.2 Where a performance is specified and no manufacturer is listed, the Contractor shall submit in accordance with Sub-article 6.7.1 the name of the manufacturer, the product proposed, and detailed information showing its characteristics.
- 6.7.2.3 Where a choice of color, pattern, or texture is available for a specified product, the Engineer will make a selection from the manufacturer's highest and best standards.
- 6.7.2.4 Where the Contractor requests that a manufacturer's product be added to the named list it shall follow the procedure set forth in Sub-article 6.7.3 below. Any NJ TRANSIT approval of an additional approved equal is subject to the conditions of Sub-article 6.7.2.1 above.
- 6.7.2.5 Where the Contractor requests that an alternate product be substituted for that specified the terms of Subarticle 6.7.4. below shall apply.

- 6.7.3 Request for Approved Equal: Should the Contractor desire to use a product other than the named manufacturer(s) product, it shall first make application to the Engineer through the Construction Manager in writing, otherwise it will be held to what is specified. The application shall clearly identify that it is a "Request for Approved Equal".
- 6.7.3.1 The procedure for submitting a request for an approved equal will be as follows: The Contractor shall submit five (5) copies of the request and data. The Contractor shall amend and update data when changes concerning information on products become known. The Contractor shall include the following information:
  - Complete data substantiating compliance of proposed approved equal with requirements of the Specifications and Contract Drawings.
  - b. For products:
    - (1) Product identification, including manufacturer's name and address, model number and options.
    - (2) Installation characteristics, installation drawings, manufacturer's literature including product description, performance and test data, and reference standards if pertinent.
    - (3) Name and address of project(s) on which product was used under similar circumstances, and date of installation.
  - c. For construction methods:
    - (1) Detailed description of proposed method.
    - (2) Drawings illustrating methods.
  - d. Itemized comparison of proposed manufacturer's product with first-named product specified. Include differences in estimated life, estimated maintenance, availability of spare parts and repair services, energy consumption, performance capacity, salvageability, manufacturer's warranties, and other material differences. The Contractor may be required, at its own expense, to perform tests to demonstrate proof of equality.
  - e. Data relating to changes in Construction Schedule.
  - f. In making a request for an approved equal, Contractor is certifying:
    - (1) That it personally investigated the proposed product and method; that it believes, to the best of its knowledge and information, that product and method is either equivalent or superior to the product and method specified; and that it will update information as new or different data becomes known to the Contractor.
    - (2) That it will furnish the same guarantee as it would for the product and method specified.
    - (3) That it will coordinate installation of proposed product and method into the Work, and will make those changes required for the Work to be complete in all respects, all at no additional expense to NJ TRANSIT.

(4) That it waives all claims for additional costs and entitlement to any extension of Contract Time as a result of requesting approval of an approved equal, whether such approval is granted by

NJ TRANSIT or not.

6.7.3.2 NJ TRANSIT reserves the right, at its sole discretion, to deny requests for approved equals should it deem

the number of such requests to be excessive.

6.7.4 Requirements for Substitutions: The Contractor agrees that NJ TRANSIT is under no obligation to consider

substitutions of any kind and may direct the Contractor to proceed with the work as specified. NJ TRANSIT

shall not be liable for any costs or delays in action upon or for failure to act upon a proposed request for

substitution. Requests for substitutions will not be considered if, in the opinion of the Project Manager, the

substitutions are excessively broad in scope, require substantial revision of the Contract Drawings or

Specifications, require substantial administrative effort and expense to review or are otherwise not in NJ

TRANSIT's best interest.

Substitutions merely indicated or implied on shop drawings or product data submittals will not be considered

if no formal request for substitution has been submitted in accordance with this Article. NJ TRANSIT's

approval of such a shop drawing or data submittal shall not constitute approval of a substitution and the

Contractor shall be liable for all costs for corrective work to provide products in conformance with the

Contract Documents.

When making a request for substitution, the Contractor shall follow the same procedural and data

submission requirements as set forth in Sub-article 6.7.3 above, except that any such submission shall be

clearly identified as a "Request for Substitution". The Contractor shall include the following additional

information:

a) A detailed cost breakdown of the proposed product in comparison to the product specified, naming

the difference in cost in each case. The cost breakdown shall be submitted in the format specified

in Article 3.2.

b) A description of the benefit that will accrue to NJ TRANSIT should approval of the proposed

substitution be granted.

If a substitute item is approved and the substitute item changes the scope of work under this or other contracts

from the original specifications, then the Contractor offering the substitute item shall be responsible for all

added costs and additional Contract Time involved by reason of the change in its work and the work of other

contracts, including redesign. Any reduction in costs involved by reason of the change in its work shall be

deducted from the Contract Price by Change Order. No change involving cost shall be made without the

written consent of the Contracting Officer.

#### 6.8 VALUE ENGINEERING SUBMITTALS

- Value Engineering is defined as cost reduction proposals initiated and developed by the Contractor for changing the materials or other requirements of the Contract. This clause does not apply to such proposal unless it is identified by the Contractor at the time of submission to the Construction Manager or NJ TRANSIT as a proposal submitted pursuant to this clause. The cost reduction proposals contemplated are those that:
  - (a) Would result in less costly items or components of items than those specified herein without impairing any of the items' essential functions and characteristics such as service life, reliability, economy of operation, ease of maintenance, and necessary standardized features; and
  - (b) Would require, in order to be applied to this Contract, a Change Order to the Contract.
- 6.8.2 Cost reduction proposals as defined herein will be processed expeditiously and in the same manner as prescribed for any other proposal which would likewise necessitate issuance of a Change Order. As a minimum, the following information shall be submitted by the Contractor with each proposal:
  - (a) A description of the difference between the existing Contract requirements and the proposed change, and the comparative advantages and disadvantages of each;
  - (b) An itemization of the requirement of the Contract which must be changed if the proposal is adopted and suggested wording for revisions required;
  - (c) An estimate of the reduction in performance costs that will result from adoption of the proposal taking into account the costs of implementation by the Contractor and the basis for the estimate;
  - (d) A prediction of the effects the proposed change would have on other costs to NJ TRANSIT such as NJ TRANSIT furnished property costs, costs of related items, and costs of maintenance and operation;
  - (e) A statement of the time by which a Change Order adopting the Proposal must be issued so as to obtain the maximum cost reduction during the remainder of the Contract, noting any effect on maintaining the Contract delivery schedule.
- NJ TRANSIT shall not be liable for delays in action upon or for failure to act upon a proposal submitted pursuant to this clause. The decision of the Contracting Officer as to the acceptance or rejection of such proposal under this Contract shall be final and shall not be subject to the "Disputes" clause of this Contract. Unless and until a Change Order adding such proposal to the Contract is issued, the Contractor shall remain obligated to perform in accordance with the existing terms of the Contract. NJ TRANSIT may accept in whole or in part a cost reduction proposal submitted pursuant to this clause by issuing a Change Order which will identify the cost reduction on which it is based.
- 6.8.4 If a cost reduction proposal submitted pursuant to this clause is accepted under this Contract, an equitable adjustment in the Contract Price and in other affected provisions of this Contract shall be made in accordance with this clause. If the equitable adjustment involves a reduction in the Contract Price, it shall

be established by determining the amount of the total estimated decrease in the Contractor's cost of performance resulting from the adoption of the cost reduction proposal, taking into account the cost of implementing the change by the Contractor, and reducing the Contract Price by 50 percent of such decrease.

#### 6.9 PROGRESS PHOTOGRAPH SUBMITTALS

- 6.9.1 Prior to construction beginning and after construction operations have been started at the site, the Contractor shall have twenty (20) different color photographs taken each month, by a professional photographer, until completion of the Work. The Construction Manager will designate the location of views to be taken each month. The Contractor shall submit three (3) sets of prints of each photograph to the Construction Manager within ten (10) days after taking.
- 6.9.2 The prints shall be standard commercial quality, 8 x 10 inches, single weight glossy paper. Each print shall have an information box, stamped on the back, 1-1/2 x 3-1/2 inches, and arranged as follows:

Table 6.9.2 Photo Labeling
NJ TRANSIT
Contract Name
Contract Number
Contractor
Photograph Number
Date
Submittal Number
Information regarding view such as location, direction or site and significant points of interest

- 6.9.3 The Contractor shall enclose the three (3) sets of photographs back-to-back in a double-faced plastic sleeve punched and bound in separate standard three-ring binders.
- 6.9.4 Negatives: The Contractor shall submit the negatives with the photos to the Construction Manager.

# 6.10 REPORTS, RECORDS AND DATA SUBMITTALS

The Contractor shall submit to the Construction Manager such schedules of quantities and costs, progress schedules, certified payrolls, reports, estimates, records and other data as NJ TRANSIT may request concerning work performed or to be performed under this Contract. The cost of submitting all such data shall not be paid separately and is considered paid for under the various items contained in the Bidder's Proposal.

- 6.11 AS-BUILT DRAWINGS AND QUANTITIES SUBMITTAL
- 6.11.1 The Contractor shall keep its prints of the Contract Drawings up to date at all times by marking on them the final location of any changes in the Work. These drawings shall be identified as the "Marked-Up Drawings". The data shall be transferred regularly by the Contractor to transparencies furnished by the Construction Manager at the expense of the Contractor.
- 6.11.2 Prior to final payment the Contractor shall submit a copy of the Marked-Up Drawings of all Contract Drawings whether altered or not to the Construction Manager with the Contractor's certifications as to the accuracy of the information. As built drawings shall be entitled "AS-BUILT" above the Title Block and dated. This information shall be reviewed by the Construction Manager; such review by the Construction Manager is for content only and not for accuracy and does not relieve the Contractor of its certification. The Contractor shall pay for the cost of reproduction. Upon completion of the Work and prior to release of final payment the Contractor shall transfer all as-built data to 24" x 36" transparencies (sepia mylar or mylars) at the expense of the Contractor. These Marked-Up Drawings and As-Built Drawings shall be submitted to and become the property of NJ TRANSIT.
- 6.11.3 Following acceptance of the Project, the Construction Manager will proceed with the preparation of as-built quantities for all Contract Items and Extra Work which has been authorized and incorporated into the Project. When such as-built quantities are completed, they will be incorporated into a proposed Final Certificate of Payment. The Contractor assumes the positive obligation of assisting the Construction Manager wherever possible in the preparation of such as-built quantities.
- 6.11.4 The Construction Manager may from time to time, prior to acceptance, prepare as-built quantities and incorporate these quantities into monthly estimate certificates through an appropriate field order or change order. Such interim as-built quantities shall be subject to recalculation following acceptance of the Project. However, nothing contained in these specifications shall be construed to place on the Construction Manager the obligation of providing the Contractor with as-built quantities for the work performed prior to the issuance of a Final Certificate of Payment, nor to provide more than rough, approximate quantities of the work done for use in the preparation of monthly estimates.
- Should it appear to the Construction Manager at the time the Project is accepted that the calculation of asbuilt quantities might result in the Contractor being obliged to return money to NJ TRANSIT, NJ TRANSIT may, in its sole discretion, refuse to release retainages pending completion of the proposed Final Certificate of Payment. Where the estimate reveals that an overpayment has been made, the Contractor shall immediately return the amount of the overpayment. If the Contractor fails to remit the overpayment, NJ TRANSIT shall avail itself of other funds held on other projects with the same Contractor or against the retainages, and then if necessary proceed against the Contractor or its Surety. Where the proposed Final Certificate of Payment reveals that no overpayment has been made, the Contractor shall be entitled to payment thereunder and the release of retainages, but the Contractor shall have no claim of any kind for

additional compensation as a result of NJ TRANSIT's decision to withhold retainages or other monies pending issuance of the proposed Final Certificate.

### 7. QUALITY ASSURANCE AND QUALITY CONTROL

- 7.1 QUALITY ASSURANCE
- 7.1.1 General: The Contractor shall establish and maintain a quality assurance program in accordance with ANSI/ASQC Q9001-1994. The program shall ensure compliance with the requirements of the Contract Documents and shall include provisions ensuring compliance by subcontractors should any portion of the Work be subcontracted.
- 7.1.2 Six copies of the Contractor's quality assurance program shall be submitted to the Construction Manager at the pre-construction meeting for review and approval by the Construction Manager. Work undertaken by the Contractor before the Construction Manager's formal approval of the Contractor's program will be at the Contractor's sole risk and expense. A quality assurance audit of the Contractor's quality assurance program may be conducted by NJ TRANSIT at any time.
- 7.1.3 The Contractor's designated quality assurance program shall not be changed without the written concurrence of NJ TRANSIT. Work undertaken by the Contractor before receipt of written concurrence from NJ TRANSIT concerning such changes of the Contractor's quality assurance program will be at the Contractor's sole risk and expense.
- 7.1.4 The Contractor's quality assurance operations may be subject to NJ TRANSIT verification at any time. Verification shall include, but not be limited to: Audit of the quality assurance program; surveillance of the operations to determine that practices, methods, and procedures of the program are being properly implemented; inspection to measure the quality of items offered for acceptance; and inspection of items prior to release for shipment to ensure compliance with requirements of the Contract Documents.
- 7.1.5 Failure by the Contractor to promptly correct deficiencies discovered by the Contractor or of which the Contractor is notified by NJ TRANSIT may be cause for suspension of the Contract until corrective action has been taken or until conformance of the Work to prescribed criteria has been demonstrated to and approved by NJ TRANSIT. As a result of such suspension, no adjustment will be made with respect to increases in the cost or time.

#### 7.2 MATERIAL - WORKMANSHIP - LABOR

7.2.1 Only approved materials shall be used, and the work shall be carefully carried out in strict accordance with the general and detail drawings. The Construction Manager shall have full power at any time to reject such work or material which does not, in the Construction Manager's opinion, conform to the true intent and meaning of the Contract Documents.

7.2.2 Work when completed in a substantial and workmanlike manner, to the satisfaction of the Construction Manager, shall be accepted by NJ TRANSIT in writing. Unless otherwise specified all materials used shall

be new.

7.2.3 The Contractor shall furnish and pay for necessary transportation, scaffolding, centering, forms, water,

labor, tools, light, power, and mechanical appliances, permits for the installation and construction of work,

and all other means, materials, and supplies for properly prosecuting its work under the Contract, unless

expressly specified otherwise. The Contractors and all subcontractors shall rely on their own

measurements for the performance of their work.

7.2.4 The Contractor shall furnish necessary and approved materials in ample quantities and as frequently as

required to avoid delay in the progress of the work, and shall so store them as to prevent interference with

work not under this Contract.

7.2.5 The Contractor shall employ qualified and competent personnel in their respective lines of work. Should the

Construction Manager deem any employee incompetent or negligent or for any cause unfit for the

employee's duties, the Contractor shall dismiss that person, and that individual shall not again be employed

on the work. No Contractor will be required to employ for any work any person against whom it has a

reasonable objection.

7.2.6 The Contractor shall employ a full-time superintendent assigned solely to this Project who shall have full

authority to act for the Contractor. It is understood that such representative shall be acceptable to the

Construction Manager and shall be one who is to be continued in that capacity for the particular job involved

unless that individual ceases to be on the Contractor's payroll.

The various subcontractors shall have competent foremen in charge of their respective part of the work at

all times. They are not to employ on the work an unfit person or anyone not skilled in the work assigned to

that person.

The Contractor shall give the work its special supervision, lay out its own work, do the necessary leveling

and measuring or employ a competent New Jersey licensed engineer or land surveyor satisfactory to the

Construction Manager to do so. If, due to trade agreement, additional standby personnel are required to

supervise equipment or temporary services used by other trades, the Contractor shall provide such standby

services.

The superintendence and the number of workmen shall be sufficient to insure the completion of the Project

within the time stipulated therefore.

- 7.2.7 No materials or supplies for the work shall be purchased by the Contractor or by any subcontractor which are subject to a lien or other encumbrance or an agreement by which an interest is retained by the seller. The Contractor and all subcontractors warrant that they have good title to materials and supplies used by them in the work, free from liens, claims or encumbrances.
- 7.2.8 Manufacturer's Instructions: Where installations include manufactured products, the Contractor shall comply with the manufacturer's applicable instructions and recommendations for installation to the extent that these instructions and recommendations are more explicit or more stringent than the requirements indicated in the Contract Documents.
- 7.2.9 Where the specifications or the manufacturer's instructions or warranty require that the site be visited and inspected by a representative of the manufacturer prior to the commencement of a particular item of work, the Contractor shall ensure that said visit or inspection occurs and that the Construction Manager be given no less than 24 hours notice of arrival of the manufacturer's representative.

### 7.3 INSPECTION OF WORK

- 7.3.1 NJ TRANSIT shall at all times have access to the work whether it is in preparation or in progress and the Contractor shall provide proper facilities for such access and for inspection and testing. NJ TRANSIT reserves the right, at its option, to employ the services of professional consultants for any phase of the work as it may deem to be in the best interest of NJ TRANSIT. The Contractor shall cooperate with NJ TRANSIT and these consultants and shall provide access to the work and facilities for inspection and testing.
- 7.3.2 If the Specifications, the Engineer's or Construction Manager's instructions, laws, ordinances or public or private authority require work to be specifically tested or approved, the Contractor so affected shall give the Construction Manager three days notice in writing of its readiness for inspection, and if the inspection is by an authority other than the Construction Manager, of the date fixed for such inspection. Inspections by the Construction Manager will be promptly made. If such work should be covered up or otherwise concealed from view without approval or consent of the Construction Manager, it must, if required by the Construction Manager, be uncovered for examination and recovered after the examination at the Contractor's expense. There will be no extension of time to the Contract for uncovering or recovering work.
- 7.3.3 Except as otherwise provided herein, materials and installed equipment used in the construction of the Project shall be adequately tested according to standards of the trade, industry or as required by NJ TRANSIT, at the expense of the Contractor.
- 7.3.4 Whenever, in the Construction Manager's opinion, the Construction Manager considers it necessary or advisable to insure the proper implementation of the intent of the Contract Documents, the Construction Manager will have authority to require special inspection or testing of the work in addition to that required elsewhere in the Contract Documents, whether or not such work be then fabricated, installed or completed.

However, neither the Construction Manager's authority to act under this Subsection, nor any decision made by the Construction Manager either to exercise or not to exercise such authority, shall give rise to any duty or responsibility of the Construction Manager to the Contractor, subcontractor, their agents or employees, or any other person performing part of the work.

7.3.5 If after commencement of the Work the Construction Manager determines that some portion of the Work requires special inspection, testing or approval not provided for elsewhere in the Contract Documents, the Construction Manager will proceed with such inspection, testing or approval under contract with a third party for such services, or instruct the Contractor to order such special inspection, testing or approval. If such special inspection or testing reveals a failure of the Work to comply (1) with the requirements of the Contract Documents or (2) with respect to the performance of the Work, with laws, ordinances, rules, regulations or orders of public authority having jurisdiction, the Contractor shall bear all costs thereof, including the Construction Manager's additional services made necessary by such failure; otherwise NJ TRANSIT shall bear such costs, and an appropriate change order will be issued.

7.3.6 The Contractor shall cooperate fully with the Construction Manager and any testing company and supply materials for testing as required.

7.3.7 All construction subcode inspections shall be performed by the DCA Bureau of Code Services. The Contractor shall be responsible for requesting subcode inspections, as necessary, by contacting the DCA directly. The Contractor shall abide by all DCA instructions regarding subcode inspection procedures. The Contractor is responsible for notifying the Construction Manager of the time and date.

## 7.4 PLANT INSPECTION

NJ TRANSIT may undertake the inspection of materials at the source. Manufacturing plants may be inspected periodically for compliance with specified manufacturing methods. Material samples may be obtained for laboratory testing for compliance with materials quality requirements. This may be the basis for acceptance of manufactured lots as to quality.

In the event plant inspection is undertaken the following conditions shall be met:

- (a) NJ TRANSIT shall have the cooperation and assistance of the Contractor and the producer with whom it contracted for the provision of materials and equipment.
- (b) NJ TRANSIT shall have full entry at all times to such parts of the plant as may concern the manufacture or production of the materials being furnished.
- (c) If required by NJ TRANSIT, the Contractor shall arrange for approved office space for the use of the inspector; such space to be located conveniently in or near the plant.
- (d) Adequate safety measures shall be provided and maintained.

It is understood that NJ TRANSIT reserves the right to retest materials which have been tested and accepted at the source of supply after the same have been delivered and to reject materials which, when retested, do not meet the requirements of the Contract Documents. The cost of retesting in case of rejection shall be borne by the Contractor.

### 7.5 INSTALLED EQUIPMENT TESTING AND TRAINING

- 7.5.1 When mechanical, electrical or other equipment is installed it shall be the responsibility of the Contractor to operate it for a satisfactory period of time as required by the Contract Documents for proper testing of the equipment and instructing NJ TRANSIT operating personnel. Fuel, power, and other items or services required for proper testing of equipment and for the period of instruction shall be provided at the expense of the Contractor furnishing such equipment. The Contractor shall provide the Construction Manager with a minimum of three (3) days prior written notice of the performance of a test. Tests shall be conducted in the presence of NJ TRANSIT. Test results shall be submitted and approved by NJ TRANSIT prior to acceptance of the installation.
- 7.5.2 The Contractor shall furnish six (6) copies of each "Operating and Maintenance Booklet" which shall contain not less than the following (as applicable to each trade):
  - Manufacturer's service manuals and equipment parts list of all functional components of the system including control diagrams, wiring diagrams of controllers, and explanation and description of each system;
  - A complete typewritten list of all items of pertinent equipment including compressors, pumps, fans, motors, coils, etc. with nameplate date, capacities, model numbers, lubrication charts and preventive maintenance schedule;
  - 3. Trouble shooting guides and testing instructions;
  - 4. Manufacturer's parts list and ordering requirements;
  - 5. Names, addresses and telephone numbers of all manufacturers agents, Subcontractors, supply houses, etc. from which replacement parts, service and operating information can be obtained;
  - 6. The manuals shall be divided into indexed sections with tabs dividing sections, for A) Mechanical/HVAC, B) Electrical/Lighting, C) Finish Schedules.
- 7.5.3 The Contractor shall instruct NJ TRANSIT designated personnel as to the proper operation of all equipment and apparatus, and each of the various systems specified. No later than two weeks prior to conducting training sessions, the Contractor shall submit a Training Plan for the Construction Manager's approval. The Training Plan shall included the instructor's qualifications, the proposed training schedule and an outline of the instructor's lesson plan. Training session topics shall include, as a minimum, a detailed review of operating and maintenance procedures, spare parts, tool requirements, prescribed lubricants and fuels, hazards and warranties. The Contractor shall be responsible for providing all visual aids and training materials. Training sessions shall be conducted at a time and place convenient to NJ TRANSIT personnel.

The Contractor, after issuing complete instructions and direction to the NJ TRANSIT designated personnel, shall secure from such persons a signed acknowledgment in duplicate stating that complete and comprehensive instructions have been received and understood. The Contractor shall then forward the two copies of the signed acknowledgment to the Construction Manager for record purposes.

### 7.6 LABORATORY TESTING AND INSPECTION

#### 7.6.1 General Requirements:

NJ TRANSIT will reimburse the Contractor for the services of an independent Testing Laboratory to perform structural steel, reinforced concrete, soils and any other testing services required by NJ TRANSIT. The Contractor shall cooperate with the laboratory to facilitate the execution of its required services. Employment of the laboratory shall in no way relieve the Contractor of its obligations to either perform any other testing and inspection work as required by the Contract Documents or to perform any other item of Work.

## 7.6.2 Related Requirements:

- A. Laboratory Selection: The Contractor shall submit the name and qualifications of three independent testing laboratories for approval to the Construction Manager. The Contractor shall solicit pricing from each prospective testing and inspection laboratory for the services requested by NJ TRANSIT. The pricing information shall be submitted with the qualification submissions listed in Sub-article 7.6.3. The Construction Manager shall determine the best qualified laboratory.
- B. Payment: Payment for the services described herein shall be made only for work which is actually performed and approved by NJ TRANSIT. Payment shall be at the rates quoted for the services listed in the Special Provisions and shall be in effect from the Notice to Proceed date and for a two year period thereafter.

Rates and fees are to be based on work performed between 8:30 a.m. and 4:30 p.m.; overtime rate of 1-1/2 times the corresponding hourly rates may be applied for work performed after 4:30 p.m. and before 8:30 a.m. The overtime rate shall also be applicable for weekends and holidays.

Travel expenses for inspection and testing services rendered outside the 50 mile radius of Newark, New Jersey shall be completely documented to the satisfaction of NJ TRANSIT. Any out of town travel expenses shall be pre-approved via an estimate submitted by the laboratory.

NJ TRANSIT shall reimburse the Contractor for laboratory testing and inspection costs upon receipt of itemized invoices from the approved laboratory.

NJ TRANSIT shall pay only the amounts of the laboratory invoices, under the allowance amount included by NJ TRANSIT as a Bid Item in the Bidder's Proposal. The allowance may be adjusted upward or downward at NJ TRANSIT's sole discretion, to reflect actual costs. NJ TRANSIT shall

pay for the initial testing only. Should any material fail to satisfy the test requirements, the Contractor shall be responsible for any additional costs and delays to retest or test replacement material.

### 7.6.3 Qualification of Laboratory:

To be qualified the laboratory shall:

- A. Meet "Recommended Requirements for Independent Laboratory Qualification", published by American Council of Independent Laboratories.
- Meet basic requirements of ASTM E329, "Standards of Recommended Practice for Inspection and Testing Agencies for Concrete and Steel as Used in Construction."
- C. Be authorized to operate in the State of New Jersey.
- D. Submit copy of report inspection of facilities made by Materials Reference Laboratory of National Bureau of Standards during the most recent tour of inspection, with memorandum of remedies of any deficiencies reported by inspection.
- E. Have testing equipment:
  - 1. Calibrated at reasonable intervals by devices of accuracy traceable to either:
    - a. National Bureau of Standards; or
    - b. Accepted value of natural physical constants.

# 7.6.4 Laboratory Duties:

The laboratory shall promptly submit three (3) copies of a written report of each test and inspection to the Construction Manager for distribution. Each report shall include:

- 1. Date issued;
- Project title and number;
- 3. Testing laboratory name, address and telephone number;
- 4. Name and signature of laboratory inspector;
- Date and time of sampling or inspection;
- 6. Record of temperature and weather conditions;
- 7. Date of test;
- 8. Identification of product and specification section;
- 9. Location of inspection or test;
- 11. Results of tests and compliance with Contract Documents; and
- 12. Interpretation of test results, when requested by the Construction Manager.

# 7.6.5 Limitation of Authority of Testing Laboratory:

Laboratory is not authorized to:

- 1. Release, revoke, alter or enlarge on requirements of Contract Documents;
- 2. Approve or accept any portion of the Work; or
- 3. Perform any duties of the Contractor.

### 7.6.6 Contractor's Responsibilities:

- A. Cooperate with laboratory personnel, provide access to work, and to manufacturer's operations.
- B. Secure and deliver to the laboratory adequate quantities or representational samples of materials proposed to be used and which require testing.
- C. Provide to the laboratory the preliminary design mix proposed to be used for concrete, and other materials mixes which require control by the testing laboratory.
- D. Furnish copies of products test reports as required.
- E. Furnish incidental labor and facilities:
  - 1. To provide access to work to be tested;
  - 2. To obtain and handle samples at the Project Site or at the sources of the product to be tested;
  - 3. To facilitate inspections and tests; and
  - 4. For storage and curing of test samples.
- F. Notify laboratory sufficiently in advance of operations to allow for laboratory assignment of personnel and scheduling of tests. NJ TRANSIT reserves the right to have materials that were not properly tested, removed and replaced at no additional cost to NJ TRANSIT. When test or inspections cannot be performed after such notice, reimburse laboratory for laboratory personnel and travel expenses incurred due to Contractor's negligence.
- G. The Contractor shall have the laboratory perform additional tests as required by NJ TRANSIT

### 7.7 CERTIFICATION OF COMPLIANCE

Certain materials as specified elsewhere will be accepted on the basis of Certificates of Compliance stating that such materials or assemblies fully comply with the requirement of the Contract. The form of the Certificates of Compliance shall be approved by the Construction Manager.

Materials or assemblies used on the basis of Certificates of Compliance may be sampled and tested at any time and if found not to be in conformity with the Contract requirements, will be subject to rejection whether in place or not. The Contractor shall require the manufacturer or supplier to furnish three copies of Certificates of Compliance with each delivery of materials, components, and manufactured items that are acceptable by certification. Two copies shall be furnished to the Construction Manager and one copy shall be retained by the Contractor.

Certificates of Compliance shall contain the following information:

- (a) Project to which the material is consigned.
- (b) Name of the Contractor to which the material is supplied.
- (c) Kind of material supplied.
- (d) Quantity of material represented by the certificate.
- (e) Means of identifying the consignment, such as label marking, seal number, manufacturer or supplier, and such additional information as required to make positive identification.
- (f) Date and method of shipment.
- (g) A statement that the material has been tested and found in conformity with the pertinent Contract requirements stated in the certificate and that test results pertaining to the material are either on file with the producer and available upon request or attached to the certificate.
- (h) Signature of a person having legal authority to bind the supplier.
- (i) Signature attested to by a Notary Public.

No payments relative to materials specified to be accepted on the basis of Certificates of Compliance shall be made until the Construction Manager is in possession of an acceptable Certificate of Compliance.

A Certificate of Compliance shall not be construed as a waiver of NJ TRANSIT's right to test the material or assemblies supplied.

#### 7.8 NON-CONFORMING WORK AND MATERIALS

- 7.8.1 Materials or work found to be defective, or not in strict conformity with the requirements of the Contract Documents, or defaced or damaged through the acts or omissions of a Contractor or its subcontractors, or through action of fire, weather, vandalism or other causes, shall be removed immediately and new materials or work substituted therefor to the satisfaction of the Construction Manager without delays by the Contractor involved and at its sole cost and expense. Under no circumstances shall the Contractor be entitled to an extension of time for correcting defective work.
- 7.8.2 Should the Construction Manager determine that work, including work of an administrative nature, is not in conformance with the requirements of the Contract Documents, the Construction Manager shall issue a Non-Conformance Notice (NCN). The NCN shall state the work or material which is non-conforming and establish a reasonable time period for correcting the non-conforming work or material. Should the Contractor fail to correct, repair or replace the non-conforming work or material in a timely manner, NJ TRANSIT may take such actions as NJ TRANSIT deems necessary to protect NJ TRANSIT's and the public's interest, including but not limited to, withholding payments, suspending all or a portion of the Work, terminating the Contract for Default, denying future prequalification or subcontractor approvals, and/or suspending or debarring the Contractor from bidding on future NJ TRANSIT contracts.

- 7.8.3 No previous inspection or certification shall be held as an acceptance of defective work or materials or to relieve the Contractor from the obligation to furnish sound materials and to perform good satisfactory work.

  The Engineer shall be the final judge of the materials and work furnished.
- 7.8.4 The Contractor shall be given every opportunity to correct defective or damaged work; however, if the Contracting Officer deems it inexpedient to have the Contractor correct work damaged or done not in accordance with the Contract, the difference in value between such work and that specified, as determined by the Contracting Officer, together with the cost and expense of correcting the work, shall be deducted from the Contract Price.

#### 7.9 WARRANTY AGAINST DEFECTIVE WORK

- 7.9.1 In addition to other warranties set out elsewhere in this Contract, the Contractor warrants that work performed under this Contract conforms to the Contract requirements and is free of defect of equipment, material or design furnished, or workmanship performed by the Contractor or its subcontractors or suppliers at any tier. Such warranty shall continue for a period of one year from the date of Acceptance of the Work. Under this warranty, the Contractor shall remedy at its own expense any such failure to conform or any such defect. In addition, the Contractor shall remedy at its own expense damage to NJ TRANSIT owned or controlled real or personal property, when that damage is the result of the failure of the Contractor or its subcontractors or suppliers at any tier to conform to contract requirements or any such defect of equipment, material, workmanship, or design. The Contractor shall also restore work damaged in fulfilling the terms of this clause. The Contractor's warranty with respect to work repaired or replaced hereunder will run for one year from the date of the acceptance by NJ TRANSIT of such repair or replacement.
- 7.9.2 NJ TRANSIT shall notify the Contractor in writing within a reasonable time after the discovery of failure, defect, or damage. Should the Contractor fail to remedy failure, defect or damage described in the first paragraph of this Article within a reasonable time after receipt of notice thereof, NJ TRANSIT shall have the right to replace, repair, or otherwise remedy such failure, defect, or damage at the Contractor's expense.
- 7.9.3 In addition to the other rights and remedies provided by this clause, subcontractor's, manufacturers', and suppliers' warranties, expressed or implied, respecting work and materials shall, at the direction of the Contracting Officer, be enforced by the Contractor for the benefit of NJ TRANSIT. In such case if the Contractor's warranty under the first paragraph of this Article has expired, a suit directed by NJ TRANSIT to enforce a subcontractor's, manufacturer's or supplier's warranty shall be at the expense of NJ TRANSIT. The Contractor shall obtain warranties which the subcontractors, manufacturers, or suppliers would give in normal commercial practice.

If directed by the Contracting Officer, the Contractor shall require any such warranties to be executed in writing to NJ TRANSIT.

7.9.4 Notwithstanding other provisions of this Article, unless such a defect is caused by the negligence of the Contractor or its subcontractors or suppliers at any tier, the Contractor shall not be liable for the repair of defects of material or design furnished by NJ TRANSIT nor for the repair of damage which results from any such defect in NJ TRANSIT furnished material or design.

7.9.5 The warranty specified herein shall not limit NJ TRANSIT's rights under Article 13.3, COMPLETION AND FINAL ACCEPTANCE.

# 8. EXCAVATION AND DIFFERING SITE CONDITIONS

### 8.1 UNCLASSIFIED EXCAVATION

Unless otherwise provided elsewhere in this Contract, excavation work shall be considered unclassified excavation and shall consist of the removal of earth, rock, abandoned utilities, foundations and all other materials encountered of whatever nature.

### 8.2 MEASUREMENT OF PAY LIMITS FOR EXCAVATION

The method of measurement and establishment of pay limits for additions or deductions for excavation shall be as follows:

Basement Excavations: Pay limit for excavation shall be in accordance with cross sections limited by vertical parallel planes extending 24 inches outside of foundation walls shown on Contract Drawings, and horizontal plane along bottom of basement concrete slab or footings.

Pipelines and Encased Utilities: Pay limit for trench excavation shall be limited to width of 36 inches or the largest diameter of pipe barrel plus 24 inches, whichever is greatest and depth at bottom of pipe barrel. When rock is encountered, the Contractor shall excavate to six inches below bottom of pipe barrel. A compacted granular fill bid for the pipe shall be provided by the Contractor. No additional payment will be made for this addition of six inches and granular fill.

Encased Electrical Conduit, Steam Transmission Lines, Unformed Foundation Footings: Width and depth of trench shall be limited to same width and elevations of structure shown on Contract Drawings.

Where unsuitable foundation material is encountered, the Contractor shall excavate to elevations as directed by the Construction Manager. Unit prices for additional excavation and replacement with approved compacted granular fill, if stated in the proposal form, shall be used as a basis for additional payment by NJ TRANSIT.

#### 8.3 SOIL BORINGS

Where data pertaining to test pits, test borings, or any like information are given, by drawings or in writing, they are for general information only and shall not relieve the Contractor from the responsibility for making such investigations as may have been necessary to insure that its bid was based on actual conditions.

### 8.4 DIFFERING, LATENT OR UNUSUAL SITE CONDITIONS

- 8.4.1 The Contractor shall not proceed with the work at the site until it has satisfied itself that the topographic data in the Contract Documents are correct.
- 8.4.2 Should the Contractor encounter subsurface and latent conditions at the site materially differing from those shown on the plans or indicated in the Specifications (Type I), it shall immediately give notice to the Construction Manager of such conditions before they are disturbed. NJ TRANSIT will thereupon promptly investigate the conditions, and if it finds that they materially differ from those shown on the plans or indicated in the Specifications, it will promptly make such changes in the plans and/or Specifications as it may find necessary. Any increase or decrease of cost and time of completion resulting from such change shall be adjusted in the manner provided in Article 3.1, Contract Changes.
- 8.4.3 The Contractor shall promptly, and before such conditions are disturbed, notify the Construction Manager in writing of any other unknown physical conditions at the site, (Type II) of an unusual nature, differing materially from those ordinarily encountered and generally recognized as belonging in work of the character provided for in this Contract. NJ TRANSIT shall promptly investigate the conditions, and if it finds that such conditions do materially so differ that they could not have been discovered by the Contractor through employing the high standard of care required in the Contractor's pre-bid investigations and that they cause an increase or decrease in the Contractor's cost of, or the time required for, performance of any part of the Work under this Contract, an equitable adjustment shall be made in accordance with Article 3.2.
- 8.4.4 No claim of the Contractor under this Article shall be allowed unless the Contractor has given the written notice required above.
- 8.4.5 No claim by the Contractor for an equitable adjustment hereunder shall be allowed if asserted after substantial completion under this Contract.
- 8.4.6 The Contractor waives its right of claim if it disturbs the condition prior to submitting notice to the Construction Manager and before the Contracting Officer acts thereon.

#### 8.5 ARCHEOLOGICAL FINDINGS

When the Contractor's excavating operations encounter prehistoric remains or artifacts of historical or archeological significance, the operations shall be temporarily discontinued in that area. NJ TRANSIT will consult archeological authorities and determine the disposition of the remains or artifacts. The discontinuance of the work shall be governed by Article 2.3, SUSPENSION OF WORK.

### 9. INDEMNIFICATION AND LIABILITY

- 9.1 INDEMNIFICATION; RISKS ASSUMED BY THE CONTRACTOR
- 9.1.1 The Contractor shall defend, indemnify and save harmless the State of New Jersey, NJ TRANSIT, USDOT (if the Contract is in whole or part federally funded) and their officers, employees, servants and agents from all suits, actions, or claims of any character including, but not limited to, expenditures and costs of investigations, hiring of witnesses, court costs, counsel fees, settlements, judgments or otherwise, brought because of any injuries or damage received or sustained by any person, persons, or property on account of the operations of said Contractor or its subcontractors in the performance of the work specified in this Contract; or on account of or in consequence of any neglect in safeguarding the work as specified in this Contract; or because of any act or omission, neglect, or misconduct of said Contractor or its subcontractors in the performance of the work specified in this Contract; or from any claims or amounts arising or recovered under the Worker's Compensation Act, or any other law, ordinance, order, or decree. So much of the money due the said Contractor under and by virtue of this Contract as may be considered necessary by NJ TRANSIT for such purpose may be retained for the use of NJ TRANSIT; except that money due to the Contractor will not be withheld when the Contractor produces satisfactory evidence that it is adequately protected by public liability and property damage insurance. NJ TRANSIT shall, as soon as practicable after a claim has been made against it, give written notice thereof to the Contractor along with full and complete particulars of the claim. If the suit is brought against NJ TRANSIT, NJ TRANSIT shall immediately forward to the Contractor every claim, demand, complaint, notice, summons, pleading or other process received by NJ TRANSIT or its representatives. NJ TRANSIT shall have the right, but not the obligation, to participate, to the extent it deems appropriate, in the defense of the matter and must concur in the terms of any settlement or other voluntary disposition of the matter. In the defense of any such claims, demands, suits, actions and proceedings, the Contractor shall not raise or introduce, without the express written permission in advance of the Attorney General of the State of New Jersey, any defense involving in any way the immunity of NJ TRANSIT or the State of New Jersey, the jurisdiction of the tribunal over NJ TRANSIT or the State of New Jersey, or the provisions of any statutes respecting suits against NJ TRANSIT or the State of New Jersey.
- 9.1.2 The Contractor assumes the following distinct and several risks, whether they arise from acts or omissions whether negligent or not of the Contractor, its subcontractors, suppliers, employees, agents, and others working for the Contractor on the Project, of NJ TRANSIT or of third persons, or from any other cause, and whether such risks are within or beyond the control of the Contractor, excepting only risks which arise from

solely affirmative acts performed by NJ TRANSIT subsequent to the execution of the Contract with actual and willful intent to cause the loss, damage, and injuries described in Paragraphs (a) and (b) below:

(a) Risks of Loss or Damage to the Construction: Until completion of all work and the acceptance of the Project by NJ TRANSIT, the Contractor shall have the charge and care of the work and of the materials to be used therein, whether permanent or temporary, including materials for which it has received partial payment and shall take every precaution against injury or damage to any part thereof by the action of the elements or from any other cause, whether arising from the execution or from the nonexecution of the work. The Contractor shall rebuild, repair, restore, and make good all injuries or damages to any portion of the work occasioned by any of the above causes before acceptance and shall bear the expense thereof. However, the Contractor shall not assume the risk for damage to the work due to acts of war.

Where necessary to protect the work or materials from damage the Contractor shall in furtherance of the above Paragraph, but not by way of limitation, at its expense, provide suitable drainage for the Project and erect such temporary structures as are necessary to protect the work or materials from damage. The risks for failure to take such actions are assumed by the Contractor.

In case of suspension of work from any cause whatever, the Contractor shall continue to be responsible for the Project as provided above and shall take such precautions as may be necessary to prevent damage to the Project, shall provide for drainage and shall erect necessary temporary structures, signs or other facilities. During such period of suspension of work, the Contractor shall properly and continuously maintain in an acceptable growing condition living material in newly established plantings, seedings, and soddings furnished under this Contract, and shall take adequate precautions to protect new tree growth and other important vegetative growth against injury. If ordered by NJ TRANSIT, the Contractor shall properly store during such suspension of work materials which have been partially paid for by NJ TRANSIT or which have been furnished by NJ TRANSIT. Such storage by the Contractor shall be on behalf of NJ TRANSIT. NJ TRANSIT shall at all times be entitled to the possession of such materials, and the Contractor shall promptly return the same to the site of the work when requested. The Contractor shall not dispose of any of the materials so stored except on written authorization from NJ TRANSIT. The Contractor shall be solely responsible for the loss of or damage to such materials.

(b) Risk of Loss to Property in Performing the Work: The risk of loss or damage to any property of the Contractor, and of claims made against the Contractor or NJ TRANSIT for loss or damage to any property of subcontractors, supplier, workmen, and others performing the work, and to lessors, occurring at any time prior to completion of removal of such property from the construction site or NJ TRANSIT's premises, or the vicinity thereof shall be borne by the Contractor.

9.1.3 Neither the acceptance of the Project by NJ TRANSIT nor the making of final payment shall release the

Contractor from its obligations under this Article. Moreover, neither the enumeration in this Subparagraph

nor the enumeration elsewhere in this Contract of particular risks assumed by the Contractor or of particular

claims for which it is responsible shall be deemed: (a) to limit the effect of the provisions of this Article or of

any other provision of this Contract relating to such risks or claims; (b) to imply that it assumes or is

responsible for risks or claims only of the type enumerated in this Article or in any other provision of this

Contract; or (c) to limit the risks which it would assume or the claims for which it would be responsible in the

absence of such enumerations.

9.1.4 The Contractor expressly understands and agrees that insurance protection required by the Contract, or

otherwise provided by the Contractor, shall in no way limit the Contractor's responsibility to defend,

indemnify, and save harmless NJ TRANSIT and the State as herein provided.

9.2 INSURANCE

The Contractor shall and shall require its subcontractor(s) to procure and maintain until the issuance of

the Final Certificate of Payment, the types of insurance specified below:

9.2.1 Fire Insurance and Extended Coverage (Builder's Risk):

The Contractor shall effect and maintain "All Risk" Builder's Risk insurance coverage, including terrorism

coverage, for 100% of the Construction value upon the facility or facilities on which the work is to be

executed or which is to be constructed, and shall also cover materials, equipment, and supplies of all

kinds incident to the construction of said facility or facilities, in temporary structures, or on vehicles, or in

the open.

9.2.2 Workers' Compensation Insurance:

The Contractor shall procure and shall maintain during the life of this Contract, Workers' Compensation

Insurance, as required by applicable State law, for all of its employees to be engaged in work at the site of the

Project under this Contract and, in case any such work is sublet, the Contractor shall require the subcontractor

similarly to provide Workers' Compensation Insurance for all of the latter's employees to be engaged in such

work unless such employees are covered by the protection afforded by the Contractor's Workers Compensation

Insurance. In case any class of employees on the Project under this Contract is not protected under the

Workers' Compensation Statute, the Contractor shall provide and shall cause each subcontractor to provide

employer's liability insurance for the protection of such of its employees as are not otherwise protected. Limits

of Employer Liability are as follows:

Employer's Liability

Bodily Injury by Accident

\$1,000,000 each accident

Bodily Injury by Disease

\$1,000,000 each employee

Bodily Injury by Disease

\$1,000,000 policy limit

# 9.2.3 Commercial General Liability Insurance:

The Contractor shall and shall require its subcontractor(s) to procure and maintain during the life of this Contract, Commercial General Liability Insurance using ISO Occurrence Form CG0001 10/93 or equivalent. The policy shall provide a minimum amount of \$2,000,000 each occurrence, \$2,000,000 personal and advertising injury, \$4,000,000 general aggregate and \$4,000,000 products completed operations aggregate.

Coverage provided under this liability policy shall be on an occurrence basis and shall include, but not be limited to, bodily injury and property damage coverage including products liability/completed operations coverage, premises operations liability, blanket contractual liability, personal injury liability, independent contractors liability, mobile equipment, damage from explosion, collapse and underground hazards, and cross liability and severability of interests clause. Additional insured endorsement CG2026 11/85, CG 2010 11/85 or CG 2010 10/93 (but only if modified to include both ongoing and completed operations) naming NJ TRANSIT and the Indemnified Parties and coverage must apply on a primary and non-contributory basis. The policy shall allow the Contractor to waive its and its insurer's rights of subrogation. There shall be no coverage exceptions for property containing or adjacent to railroad facilities or other transportation facilities..

The Contractor shall furnish completed operations insurance written to the limits stipulated herein for Commercial General Liability Insurance. Coverage shall be required and maintained in force for a minimum of three years following acceptance of the overall Contract, regardless of any beneficial occupancy by NJ TRANSIT during the Contract term.

# 9.2.4 Umbrella Liability Insurance

The Contractor shall and shall require its subcontractors to procure and maintain umbrella liability insurance with a minimum limit of \$8,000,000 per occurrence and in the aggregate; coverage must follow form above underlying Commercial General Liability, Business Automobile Liability and Employer's Liability policies.

# 9.2.5 <u>Automobile Liability Insurance</u>

The Contractor shall and shall require its subcontractor(s) to procure and maintain during the life of the Contract, Automobile Liability Insurance applicable to all owned, non-owned, hired or leased vehicles with a minimum of \$1,000,000 combined single limit per accident for bodily injury and property damage liability. This policy shall name NJ TRANSIT as an additional insured.

9.2.6 <u>Asbestos Abatement Liability</u> –If applicable, the Contractor {or whomever is performing the removal of any Asbestos Containing Material {ACM} shall maintain throughout the entire period of their performance under this Contract <u>Asbestos Abatement Liability Insurance</u> in the amount of \$2,000,000 per loss and

\$2,000,000 in the aggregate. The Contractor {or whomever is responsible for transporting and disposing of the {ACM} shall maintain throughout the entire period of their performance under this Contract <a href="mailto:Transportation Pollution Coverage">Transportation Pollution Coverage {Form MCS90}</a> in the amount of \$2,000,000 or statutory minimum whichever is greater.

## 9.2.7 Contractor's Pollution Liability Insurance

The Contractor and any subcontractor performing construction and/or environmental remediation work must procure and maintain through the life of the Contract, Contractor's Pollution Liability Insurance, including lead abatement if required, covering the liability arising out of any sudden and non-sudden pollution or impairment of the environment, including bodily injury, property damage, clean-up costs and defense that arise from the work performed by the Contractor or its Sub-Contractor(s). Coverage under this policy shall have limits of liability with a minimum of \$5,000,000 per occurrence and shall be on an occurrence basis. The policy shall name NJ TRANSIT as an additional insured. Transport of any hazardous waste generated under this Contract shall require Hazardous Waste Haulers Insurance (MCS90) in an amount of \$5,000,000 per occurrence or statutory minimum, whichever is greater.

- 9.2.8 The Contractor and its subcontractor(s) shall, at its own expense, carry all insurance which may be required to provide the necessary protection against loss or damage to any property of the Contractor or to any property of subcontractors, suppliers, workmen, and others performing the work and to lessors, which insurance shall contain a waiver of any right of subrogation against NJ TRANSIT.
- 9.2.9 The insurance required herein shall provide adequate protection for the Contractor and its subcontractors, respectively, against damage claims which may arise from operations under this Contract, whether such operations be by the insured or by anyone directly or indirectly employed by the Contractor and, also against any of the special hazards which may be encountered in the performance of this Contract.
- 9.2.10 The insurance policies are to be written by good and solvent insurance companies authorized to do business in New Jersey with an A.M. BEST Insurance Rating of "A-" or better or by such other companies acceptable to NJ TRANSIT in its sole discretion.
- 9.2.11 The Contractor shall furnish NJ TRANSIT with two copies of all Certificates showing the types, amount, class of operations covered, effective dates, and dates of expiration of policies. The policies evidencing required insurance shall contain an endorsement to the effect that cancellation or any material change in the policies adversely affecting the interests of NJ TRANSIT in such insurance shall not be effective for less than thirty (30) days after written notice thereof to NJ TRANSIT. Evidence of such endorsement must be contained in the certificate of insurance. If requested by NJ TRANSIT, the Contractor shall also provide copies of the insurance policies covered by the certificate. The Contractor shall not commence work under this Contract until it has obtained the insurance required under this Paragraph and such insurance has been

approved by the Contracting Officer, nor shall the Contractor allow any subcontractor to commence work on its subcontract until the insurance required of the subcontractor has been so obtained and approved.

If the insurance provided by the Contractor or any of its subcontractor(s) fails to comply with the requirements listed herein, or if the Contractor or its subcontractor(s) fails to maintain such insurance, then NJ TRANSIT maintains the right to stop work until proper evidence is provided.

9.2.12 The cost of providing the required insurance shall be included under the Bid Item "Mobilization" whenever such a bid item is listed in the Bidder's Proposal. If no such item is listed then the cost shall be considered included under the total lump sum bid or allocated within the unit prices that sum to the total bid price.

# 9.2.13 Railroad Protective Comprehensive General Liability Insurance:

Should it be required, NJ TRANSIT will provide Railroad Protective Comprehensive General Liability Insurance coverage for this Contract.

### 9.3 LIMITATIONS OF LIABILITY

In no event, whether under the provisions of this Contract, as a result of breach of Contract, tort (including negligence) or otherwise, shall NJ TRANSIT, the State, or USDOT, be liable to the Contractor for special, consequential, incidental or penal damages including, but not limited to, loss of profit or revenues, loss of rental value for Contractor owned equipment, damages to associated equipment, additional risk, cost of capital or interest of any nature (whether characterized as damages for the retention of money, an increase in the cost of performance, a penalty, or otherwise).

# 9.4 NO THIRD PARTY BENEFICIARIES

It is specifically agreed between the parties executing this Contract that it is not intended by any of the provisions of any part of the Contract to create the public or any member thereof a third party beneficiary hereunder, or to authorize anyone not a party to the Contract to maintain a suit for personal injuries or property damage pursuant to the terms or provisions of the Contract.

It is the further intent of NJ TRANSIT and the Contractor in executing this Contract that no individual, firm, corporation or any combination thereof, which supplies materials, labor, services or equipment to the Contractor for the performance of the work becomes thereby a third party beneficiary of this Contract. NJ TRANSIT and the Contractor understand that such individual, firm, corporation or combination thereof, has no right to bring an action in the courts of this State against the State or NJ TRANSIT, by virtue of this lack of standing and also by virtue of the provisions of the New Jersey Contractual Liability Act, N.J.S.A. 59:13-1, et seq.

#### 9.5 PERSONAL LIABILITY OF PUBLIC OFFICIALS

In carrying out the provisions of the Contract, or in exercising power or authority granted to them by or within the scope of the Contract, there shall be no liability upon any NJ TRANSIT employee, either personally or as officials of the State, it being understood that in all such matters they act solely as agents and representatives of the State.

#### 9.6 INTELLECTUAL PROPERTY

If the Contractor uses any design, device or materials covered by letters, patent or copyright, it shall provide for such use by suitable agreement with the owner of such patented or copyrighted design, device or material. It is mutually agreed and understood that, without exception, the Contract Price shall include all royalties or costs arising from the use of such design, device or materials in any way involved in the Work. The Contractor shall defend, indemnify and save harmless the State, USDOT, NJ TRANSIT, and their officers, agents, servants, and employees from any and all claims for infringement by reason of the use of such patented or copyrighted design, device or materials or any trademark or copyright in connection with Work agreed to be performed under this Contract, and shall defend and indemnify the State, USDOT, NJ TRANSIT, and their officers, agents, servants, and employees, for any cost, expense or damage which it or they may be obliged to pay by reason of such infringement at any time during the prosecution of the Work or after completion of the Work.

### 9.7 ENVIRONMENTAL COMPLIANCE AND LIABILITY

- 9.7.1 The Contractor shall conduct all activities in compliance with all applicable federal, state and local laws, rules, regulations and permits designed to prevent or control the discharge of substances into the land, water and air and to protect individual health and safety. The Contractor will indemnify, hold harmless and defend NJ TRANSIT, the State of New Jersey, their directors, officers, employees, agents and assigns from and against any and all suits, actions, proceedings, costs, fines, penalties and claims arising from or alleged to have arisen from its violation of any such environmental, health or safety laws, rules, regulations or permits whenever such suits, actions, claims or proceedings shall be commenced, or whenever such costs are accrued, except for any violations, if any, at the NJ TRANSIT facility existing prior to the Contractor's activities. The Contractor shall take reasonable and necessary precautions to prevent the discharge of hazardous substances, including asbestos and petroleum products, onto NJ TRANSIT property or into the environment, including the air. Failure to comply will be considered grounds for default, and NJ TRANSIT may terminate the Contract in accordance with Article 2.5, TERMINATION FOR CAUSE. The indemnification obligations hereunder shall survive the completion or termination of this Contract.
- 9.7.2 No later than two weeks after the Notice to Proceed for this project, the Contractor shall supply to NJ TRANSIT a set of MSDS for any and all chemicals, materials or substances intended for use in the completion of the project that are covered by reference or definition by the OSHA Hazard Communication Standard (hereinafter HCS) and/or the New Jersey Worker and Community Right-to-Know Act, N.J.S.A. 34:5A-1 et seq. The chemical name and Chemical Abstract Service (hereinafter CAS) number must be

provided for all hazardous substances and for the five most predominant ingredients. If this information is not available on the MSDS, the information must be provided under separate cover when the MSDS is submitted. The Contractor shall also supply to NJ TRANSIT a copy of its written hazard communication program as defined by the OSHA-HCS and the New Jersey Worker and Community Right-to-Know Act, N.J.S.A. 34:5A-1 et seq.

- 9.7.3 In addition to supplying NJ TRANSIT with the MSDS, the Contractor shall obtain the expressed written approval of NJ TRANSIT to use any chemicals with a flammability or reactivity hazard classification of 2, 3, or 4 as defined by the National Fire Protection Association Standard NFPA704.
- 9.7.4 During the performance of this contract, the Contractor shall take any and all necessary precautions to ensure that personnel and property of NJ TRANSIT, the Contractor, third parties, and the general public are not exposed to physical or health hazards from any of the aforementioned chemicals, materials and substances. In addition, the aforementioned chemicals, materials and substances shall be labeled with the chemical name and CAS number of all hazardous substances including the five most predominant ingredients in accordance with the requirements of OSHA-HCS and the New Jersey Worker and Community Right-to-Know Act.
- 9.7.5 In the event the Contractor obtains any new information pertaining to the aforementioned chemicals, materials and substances during the performance of the work on this contract, the Contractor shall immediately make that information available to NJ TRANSIT.
- 9.7.6 The Contractor's format shall meet the requirements of OSHA-HCS. Alternative formats may be accepted provided they meet the requirements of the OSHA-HCS and New Jersey Worker and Community Right-to-Know Act.

#### 10. ETHICAL REQUIREMENTS

10.1 COVENANT AGAINST CONTINGENT FEES

The Contractor warrants that no person or selling agency has been employed or retained to solicit or secure this Contract upon an agreement or understanding for a commission, percentage, brokerage, or contingent fee, except bona fide employees or bona fide established commercial or selling agencies maintained by the Contractor for the purpose of securing business. For breach or violation of this warranty, NJ TRANSIT shall have the right to annul this Contract without liability and in its discretion to deduct from the Contract Price, or otherwise recover, the full amount of such commission, percentage, brokerage or contingent fee.

- 10.2 OFFICIALS NOT TO BENEFIT
- 10.2.1 <u>Interest of Members of Congress</u>: No member of or delegate to the Congress of the United States shall be admitted to any share or part of this Contract or to any benefit arising therefrom.

10.2.2 <u>Interest of Public Officials</u>: No member, officer or employee of NJ TRANSIT or the State shall have any interest, direct or indirect, in this Contract or the proceeds thereof. No former member, officer or employee of NJ TRANSIT who, during that person's tenure, had a direct, substantial involvement with matters that are related to this Contract, shall have any interest, direct or indirect, in this Contract or the proceeds thereof.

#### 10.3 GRATUITIES

- 10.3.1 The Contracting Officer may, by written notice to the Contractor, terminate the right of the Contractor to proceed under the Contract if it is found, after notice and hearing, that gratuities (in the form of entertainment, gifts or otherwise) were offered or given by the Contractor, or any agent or representative of the Contractor, to any officer, agent or employee of NJ TRANSIT with a view toward securing a contract or securing favorable treatment with respect to the performance of such Contract; provided that the existence of the facts upon which NJ TRANSIT makes its findings shall be in issue and may be reviewed in any competent court.
- 10.3.2 In the event this Contract is terminated as provided in the preceding Paragraph, NJ TRANSIT shall be entitled to pursue the same remedies against the Contractor as it could pursue in the event of a breach of the Contract by the Contractor.

### 10.4 NJ TRANSIT CODE OF ETHICS

- It is NJ TRANSIT policy that Contractors must avoid all situations where proprietary or financial interest, or the opportunity for financial gain, could lead an officer or employee of NJ TRANSIT to secure favored treatment for any organization or individual. Contractors must also avoid circumstances and conduct which may not constitute actual wrongdoing, or conflict of interest, but might nevertheless appear questionable to the general public, thus compromising the integrity of NJ TRANSIT. For the purposes of this Article only, NJ TRANSIT shall be deemed to include any person contracting with NJ TRANSIT to perform services on the Project. All Contractors must comply with NJ TRANSIT's Code of Ethics contained in this Article.
- 10.4.2 The Contractor shall not employ any NJ TRANSIT officer or employee in the business of the Contractor or in professional activity in which the Contractor is involved with the NJ TRANSIT officer or employee.
- The Contractor shall not offer or provide any interest, financial or otherwise, direct or indirect, to any NJ TRANSIT officer or employee, in the business of the Contractor or in professional activity in which the Contractor is involved with the NJ TRANSIT officer or employee. The Contractor shall not cause or influence, or attempt to cause or influence, any NJ TRANSIT officer or employee to act in his or her official capacity in any manner which might tend to impair the objectivity or independence of judgment of that NJ TRANSIT officer or employee.

- 10.4.4 The Contractor shall not cause or influence, or attempt to cause or influence, any NJ TRANSIT officer or employee to use or attempt to use his or her official position to secure any unwarranted privileges or advantages for that Contractor or any other person.
- 10.4.5 The Contractor shall not offer any NJ TRANSIT officer or employee any gift, favor, service or other thing of value under circumstances from which it might be reasonably inferred that such gift, service or other thing of value was given or offered for the purpose of influencing the recipient in the discharge of his or her official duties. In addition, employees or officers of NJ TRANSIT will not be permitted to accept breakfasts, lunches, dinners, alcoholic beverages, tickets to entertainment and/or sporting events, or any other item which could be construed as having more than nominal value.
- 10.4.6 In accordance with N.J.A.C. 16:72-4.1, the Contractor my be suspended and/or debarred if the Contractor:
  - A. Makes any offer or agreement to pay or to make payment of, either directly or indirectly, any fee, commission, compensation, gift, gratuity, or other thing of value of any kind to any NJ TRANSIT Board member, officer, or employee or to any member of the immediate family of such Board member, officer, or employee, or any partnership, firm, or corporation with which they are employed or associated, or in which such Board member, officer, or employee has an interest within the meaning of N.J.S.A. 52: 13D-13g:
  - B. Fails to report to the Attorney General and to the Executive Commission on Ethical Standards in writing forthwith the solicitation of any fee, commission, compensation, gift, gratuity or other thing of value by any NJ TRANSIT Board member, officer, or employee;
  - C. Undertakes, directly or indirectly, any private business, commercial, or entrepreneurial relationship with, whether or not pursuant to employment, contract or other agreement, express or implied, or sale, directly or indirectly of any interest in such Contractor to, any NJ TRANSIT Board member, officer, or employee having any duties or responsibilities in connection with the purchase, acquisition, or sale of any property or services by or to NJ TRANSIT, or with any person, firm, or entity with which he has an interest within the meaning of N.J.S.A. 52:13D-13g. Any relationship subject to this provision shall be reported in writing forthwith to the Executive Commission on Ethical Standards, which may grant a waiver of this restriction upon application of the NJ TRANSIT Board member, officer, or employee upon a finding that the present or proposed relationship does not present the potential, actuality, or appearance of a conflict of interest;
  - D. Influence or attempts to influence or causes to be influenced, any NJ TRANSIT Board member, officer, or employee in his official capacity in any manner which might tend to impair the objectivity or independence of judgment of such Board member, officer, or employee; or
  - E. Causes or influences or attempts to cause or influence, any NJ TRANSIT Board member, officer, or employee to use, or attempt to use, his official position to secure unwarranted privileges or advantages for the contractor or any other person.

### 11. SOCIAL AND ECONOMIC REQUIREMENTS

### 11.1 NEW JERSEY PREVAILING WAGE ACT

The Contractor and each subcontractor shall comply with the New Jersey Prevailing Wage Act, N.J.S.A. 34:11-56.25 et seq., and this Act is hereby made a part of this Contract. All workers shall be paid not less than the prevailing wage rate as designated by the Commissioner of Labor and Industry or the Commissioner's duly authorized deputy or representative.

In the event it is found that any worker has been paid a rate of wages less than the prevailing wage required to be paid by this Contract, the Contracting Officer may terminate the Contractor's or subcontractor's right to proceed with the work, or such part of the work as to which there has been a failure to pay required wages, and take such action it deems necessary or prosecute the work to completion.

NJ TRANSIT shall furnish as part of the Contract a copy of the prevailing minimum wage rates which shall be paid to the workers employed in the performance of the Contract.

- 11.1.2 Nothing contained in the Prevailing Wage Act shall prohibit the payment of more than the prevailing wage rate to any worker employed on the Project.
- 11.1.3 The Contractor and each subcontractor performing work for NJ TRANSIT who is subject to the provisions of the Prevailing Wage Act shall post the prevailing wage rates for each craft and classification involved, including the effective date of any changes thereof, in prominent and easily accessible places at the site of the work or at such place or places as are used by them to pay workers their wages.
- 11.1.4 The bidder's signature on the proposal is its guarantee that neither it nor any subcontractor it intends to contract with is currently listed by or on record with the Commissioner of Labor and Industry as one who failed to pay the prevailing wages according to the Prevailing Wage Act.
- 11.1.5 The Contractor and all of its subcontractors performing work at the site must prepare their bids as to labor costs in accordance with the prevailing wage (valid for the date the bids are to be submitted) for the geographical area of the Project Site.
- 11.1.6 After the completion of all construction work and before the proposed Final Certificate of Payment will be issued, the Contractor and subcontractors shall furnish the Construction Manager with written statements in form satisfactory to NJ TRANSIT certifying to the amounts then due and owing from the Contractor and subcontractors filing such statement to any and all workers for wages due on account of the Contract, setting forth therein the names of the persons whose wages are unpaid and the amount due to each respectively, which statement shall be verified by the oath of the Contractor or subcontractor, as the case may be, that it has read such statement subscribed by the Contractor or subcontractor, knows the contents thereof, and that the same is true of its own knowledge, provided, however, that nothing herein shall impair

the right of the Contractor to receive Final Payment because of failure of any subcontractor to comply with provisions of this Article.

#### 11.2 EQUAL OPPORTUNITY

### 11.2.1 Equal Employment Opportunity

The parties to this Contract do hereby agree that the provisions of N.J.S.A. 10:2-1 through 10:2-4 and N.J.S.A. 10:5-31 et seq. (P.L. 1975, c.127) as amended and supplemented and the rules and regulations promulgated pursuant thereto are hereby made a part of this Contract and are binding upon them.

The Contractor agrees that at the time the signed contract is returned to NJ TRANSIT, Contractor shall submit to NJ TRANSIT and the New Jersey Department of Treasury, Division of Public Contracts Equal Employment Opportunity Compliance an Initial Project Workforce Report (Form AA-201) consisting of forms provided by NJ TRANSIT and completed by Contractor in accordance with Subchapter 7 of the regulations promulgated by the Treasurer pursuant to P.L. 1975, c.127.

## 11.2.2 In accordance with N.J.S.A. 10:2-1 the Contractor agrees that:

- a. In the hiring of persons for the performance of work under this contract or any subcontract hereunder, or for the procurement, manufacture, assembling or furnishing of any such materials, equipment, supplies or services to be acquired under this contract, no contractor, nor any person acting on behalf of such contractor or subcontractor, shall, by reason of race, creed, color, national origin, ancestry, marital status, gender identity or expression, affectional or sexual orientation or sex, discriminate against any person who is qualified and available to perform the work to which the employment relates;
- b. No contractor, subcontractor, nor any person on his behalf shall, in any manner, discriminate against or intimidate any employee engaged in the performance of work under this contract or any subcontract hereunder, or engaged in the procurement, manufacture, assembling or furnishing of any such materials, equipment, supplies or services to be acquired under such contract, on account of race, creed, color, national origin, ancestry, marital status, gender identity or expression, affectional or sexual orientation or sex;
- c. There may be deducted from the amount payable to the contractor by the contracting public agency, under this contract, a penalty of \$ 50.00 for each person for each calendar day during which such person is discriminated against or intimidated in violation of the provisions of the contract; and
- d. This contract may be canceled or terminated by the contracting public agency, and all money due or to become due hereunder may be forfeited, for any violation of this section of the contract occurring after notice to the contractor from the contracting public agency of any prior violation of this section of the contract.

# 11.2.3 Executive Order 151 Mandatory EEO and Affirmative Action Requirements

It is the policy of the NJ TRANSIT that its contracts should create a workforce that reflects the diversity of the State of New Jersey. Therefore, contractors engaged by NJ TRANSIT to perform under a construction contract shall put forth a good faith effort to engage in recruitment and employment practices that further the goal of fostering equal opportunities to minorities and women.

The Contractor must demonstrate to the NJ TRANSIT's satisfaction that a good faith effort was made to ensure that minorities and women have been afforded equal opportunity to gain employment under the NJ TRANSIT's contract with the Contractor. Payment may be withheld from a contractor's contract for failure to comply with these provisions.

Evidence of a "good faith effort" includes, but is not limited to:

- A. The Contractor shall recruit prospective employees through the State Job bank website, managed by the Department of Labor and Workforce Development, available online at http://NJ.gov/JobCentralNJ.
- B. The Contractor shall keep specific records of its efforts, including records of all individuals interviewed and hired, including the specific numbers of minorities and women.
- C. The Contractor shall actively solicit and shall provide NJ TRANSIT with proof of solicitations for employment, including but not limited to advertisements in general circulation media, professional service publications and electronic media.
- D. The Contractor shall provide evidence of efforts described at 2 above to NJ TRANSIT no less frequently than once every 12 months.
- E. The Contractor shall comply with the requirements set forth at N.J.A.C. 17:27.

### 11.2.4 SUBCONTRACTS; EQUAL EMPLOYMENT GOALS

The contractor agrees to incorporate these State of New Jersey EEO Provisions for Construction Contracts in its construction subcontracts.

The New Jersey Department of Treasury, Division of Public Contracts Equal Employment Opportunity Compliance (hereinafter "Division") shall individually establish the targeted minority and women employment goals for determining good faith equal employment opportunity efforts by each construction contractor and subcontractor for each trade on each contract. The Division shall review the trades to be utilized during the completion of the work as reported on the Initial Project Workforce Report and determine the targeted employment goals based upon the number of qualified minorities and women available as reported by the New Jersey Department of Labor, Division of Planning and Research in its report, EEO Tabulations—Detailed Occupations by Race/Hispanic Groups.

In accordance with N.J.A.C. 17:27, Contractors and subcontractors are required to make a good faith effort to provide equal employment opportunity for minorities and women. Failure to make good faith efforts to provide equal employment opportunity for minorities and women may result in sanctions including fines/penalties, withholding of payment, termination of the contract, suspension/debarment or such other action as provided by law.

The contractor shall obtain the targeted employment goals from the Division's website at <a href="https://www.state.nj.us/treasury/contract">www.state.nj.us/treasury/contract</a> compliance or request the employment goals from the Division of Public Contracts Equal Employment Opportunity Compliance after submitting the initial project workforce report to the Division of Public Contracts Equal Employment Opportunity Compliance.

The contractor shall then be responsible for the completion and submittal of a Monthly Project Workforce Report (Form AA-202) for the duration of this contract.

(NOTE: FOR THE PURPOSE OF THIS CONTRACT THE "PUBLIC AGENCY COMPLIANCE OFFICER" REFERENCED BELOW IS NJ TRANSIT'S VP CIVIL RIGHTS & DIVERSITY PROGRAMS AND THE "PUBLIC AGENCY" IS NJ TRANSIT.)

11.2.5 MANDATORY EQUAL EMPLOYMENT OPPORTUNITY LANGUAGE FOR CONSTRUCTION CONTRACTS

During the performance of this contract, the Contractor agrees as follows:

The Contractor or subcontractor, where applicable, will not discriminate against any employee or applicant for employment because of age, race, creed, color, national origin, ancestry, marital status, affectional or sexual orientation, gender identity or expression, disability, nationality or sex. Except with respect to affectional or sexual orientation and gender identity or expression, the Contractor will take affirmative action to ensure that such applicants are recruited and employed, and that employees are treated during employment, without regard to their age, race, creed, color, national origin, ancestry, marital status, affectional or sexual orientation, gender identity or expression, disability, nationality or sex. Such action shall include, but not be limited to the following: employment, up-grading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The Contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided by the Public Agency Compliance Officer setting forth provisions of this nondiscrimination clause.

The Contractor or subcontractor, where applicable will, in all solicitations or advertisements for employees placed by or on behalf of the Contractor, state that all qualified applicants will receive consideration for employment without regard to age, race, creed, color, national origin, ancestry, marital status, affectional or sexual orientation, gender identity or expression, disability, nationality or sex.

The Contractor or subcontractor, where applicable, will send to each labor union or representative of workers with which it has a collective bargaining agreement or other contract or understanding, a notice, to be provided by the agency contracting officer advising the labor union or workers' representative of the Contractor's commitments under this act and shall post copies of the notice in conspicuous places available to employees and applicants for employment.

The Contractor or subcontractor, where applicable, agrees to comply with any regulations promulgated by the Treasurer, pursuant to N.J.S.A. 10:5-31 et seq., as amended and supplemented from time to time and the Americans with Disabilities Act.

When hiring or scheduling workers in each construction trade, the Contractor or subcontractor agrees to make good faith efforts to employ minority and women workers in each construction trade consistent with the applicable employment goal prescribed by N.J.A.C. 17:27-7.2; provided, however, that the Department of Labor and Workforce Development (LWD), Construction EEO Monitoring Program may, in its discretion, exempt a contractor or subcontractor from compliance with the good faith procedures prescribed by the following provisions, A, B and C, as long as the Dept.of LWD, Construction EEO Monitoring Program is satisfied that the Contractor or subcontractor is employing workers provided by a union which provides evidence, in accordance with standards prescribed by the Dept. of LWD, Construction EEO Monitoring Program, that its percentage of active "card carrying" members who are minority and women workers is equal to or greater than the targeted employment goal established in accordance with N.J.A.C. 17:27-7.2.

The Contractor or subcontractor agrees that a good faith effort shall include compliance with the following procedures:

(A) If the Contractor or subcontractor has a referral agreement or arrangement with a union for a construction trade, the Contractor or subcontractor shall, within three business days of the contract award, seek assurances from the union that it will cooperate with the Contractor or subcontractor as it fulfills its affirmative action obligations under this contract and in accordance with the rules promulgated by the Treasurer pursuant to N.J.S.A. 10:5-31 et seq., as supplemented and amended from time to time, and the Americans with Disabilities Act. If the Contractor or subcontractor is unable to obtain said assurances from the construction trade union at least five business days prior to the commencement of construction work, the Contractor or subcontractor agrees to afford equal employment opportunities to minority and women workers directly, consistent with this chapter. . If the Contractor's or subcontractor's prior experience with a construction trade union, regardless of whether the union has provided said assurances, indicates a significant possibility that the trade union will not refer sufficient minority and women workers consistent with affording equal employment opportunities as specified in this chapter, , the Contractor or subcontractor agrees to be prepared to provide such opportunities to minority and women workers directly, consistent with this chapter, by complying with the hiring or scheduling procedures prescribed under (B) below; and the Contractor or subcontractor further agrees to take said action immediately if it determines or is so notified by the Division that the union is not referring minority and women workers consistent with the equal employment opportunity goals set forth in this chapter.

- (B) If good faith efforts to meet targeted employment goals have not or cannot be met for each construction trade by adhering to the procedures of (A) above, or if the Contractor does not have a referral agreement or arrangement with a union for a construction trade, the Contractor or subcontractor agrees to take the following actions:
  - (I) To notify the public agency compliance officer, the Dept. of LWD, Construction EEO Monitoring Program, and minority and women referral organizations listed by the Division pursuant to N.J.A.C. 17:27-5.3, of its workforce needs, and request referral of minority and women workers:
  - (2) To notify any minority and women workers who have been listed with it as awaiting available vacancies;
  - (3) Prior to commencement of work, to request that the local construction trade union refer minority and women workers to fill job openings, provided the Contractor or subcontractor has a referral agreement or arrangement with a union for the construction trade;
  - (4) To leave standing requests for additional referral to minority and women workers with the local construction trade union, provided the Contractor or subcontractor has a referral agreement or arrangement with a union for the construction trade, the State Training and Employment Service and other approved referral sources in the area;
  - (5) If it is necessary to lay off some of the workers in a given trade on the construction site, layoffs shall be conducted in compliance with the equal employment opportunity and nondiscrimination standards set forth in this regulation, as well as with Federal and State court decisions:
  - (6) To adhere to the following procedure when minority and women workers apply or are referred to the Contractor or subcontractor:
    - (i) The Contractor or subcontractor shall interview the referred minority or woman worker.
    - If said individuals have never previously received any document or certification (ii) signifying a level of qualification lower than that required in order to perform the work of the construction trade, the Contractor or subcontractor shall in good faith determine the qualifications of such individuals. The Contractor or subcontractor shall hire or schedule those individuals who satisfy appropriate qualification standards in conformity with the equal employment opportunity and nondiscrimination principles set forth in this chapter. However, a contractor or subcontractor shall determine that the individual at least possesses the requisite skills, and experience recognized by a union, apprentice program or a referral agency, provided the referral agency is acceptable to the Dept. of LWD, EEO Monitoring Program. . If necessary, the Contractor or Construction subcontractor shall hire or schedule minority and women workers who qualify as trainees pursuant to these rules. All of the requirements, however, are limited by the provisions of (C) below.

- (iii) The name of any interested women or minority individual shall be maintained on a waiting list and shall be considered for employment as described in (i) above, whenever vacancies occur. At the request of the Dept. of LWD, Construction EEO Monitoring Program, the contractor or subcontractor shall provide evidence of its good faith efforts to employ women and minorities from the list to fill vacancies.
- (iv) If, for any reason, said Contractor or subcontractor determines that a minority individual or a woman is not qualified or if the individual qualifies as an advanced trainee or apprentice, the Contractor or subcontractor shall inform the individual in writing of the reasons for the determination, maintain a copy of the determination in its files, and send a copy to the public agency compliance officer and to the Dept. Of LWD, Construction EEO Monitoring Program.
- (7) To keep a complete and accurate record of all requests made for the referral of workers in any trade covered by the contract, on forms made available by the Dept. of LWD, EEO Monitoring Program and submitted promptly to the Dept. of LWD, Construction EEO Monitoring Program upon request.
- (C) The Contractor or subcontractor agrees that nothing contained in (B) above shall preclude the Contractor or subcontractor from complying with the union hiring hall or apprenticeship policies in any applicable collective bargaining agreement or union hiring hall arrangement, and, where required by custom or agreement, it shall send journeymen and trainees to the union for referral, or to the apprenticeship program for admission, pursuant to such agreement or arrangement. However, where the practices of a union or apprenticeship program will result in the exclusion of minorities and women or the failure to refer minorities and women consistent with the targeted county employment goal, the Contractor or subcontractor shall consider for employment persons referred pursuant to (B) above without regard to such agreement or arrangement; provided further, however, that the Contractor or subcontractor shall not be required to employ women and minority advanced trainees and trainees in numbers which result in the employment of advanced trainees and trainees as a percentage of the total workforce for the construction trade, which percentage significantly exceeds the apprentice to journey worker ratio specified in the applicable collective bargaining agreement, or in the absence of a collective bargaining agreement, exceeds the ratio established by practice in the area for said construction trade. Also, the Contractor or subcontractor agrees that, in implementing the procedures of (B) above, it shall, where applicable, employ minority and women workers residing within the geographical jurisdiction of the union.

After notification of award, but prior to signing a construction contract, the Contractor shall submit to the public agency compliance officer and the Dept. of LWD, Construction EEO Monitoring Program an initial project workforce report (Form AA 201) electronically provided to the public agency by the Dept. of LWD, Construction EEO Monitoring Program through its

website, for distribution to and completion by the Contractor, in accordance with N.J.A.C. 17:27-7. The Contractor also agrees to submit a copy of the Monthly Project Workforce Report once a month thereafter for the duration of this contract to the Dept. of LWD, Construction EEO Monitoring Program and to the public agency compliance officer.

The Contractor agrees to cooperate with the public agency in the payment of budgeted funds, as is necessary, for on-the-job and/or off-the-job programs for outreach and training of minorities and women.

(D) The Contractor and its subcontractors shall furnish such reports or other documents to the Dept. of LWD, Construction EEO Monitoring Program as may be requested by the Dept. of LWD, Construction EEO Monitoring Program from time to time in order to carry out the purposes of these regulations, and public agencies shall furnish such information as may be requested by the Dept. Of LWD, Construction EEO Monitoring Program for conducting a compliance investigation pursuant to Subchapter IO of the Administrative Code (NJAC 17:27).

# 11.2.6 Equal Opportunity for Individuals with Disabilities

The Contractor and NJ TRANSIT agree that the provisions of Title II of the Americans with Disabilities Act of 1990 (the "Act") (42 USC §12101 et seq.), which prohibits discrimination on the basis of disability by public entities in all services, programs, and activities provided or made available by public entities, and the rules and regulations promulgated thereto, are made a part of this Contract. In providing any aid, benefit, or service on behalf of NJ TRANSIT pursuant to this Contract, the Contractor agrees that the performance shall be in strict compliance with the Act. In the event that the Contractor, its agents, servants, employees, or subcontractors violate or are alleged to have violated the Act during the performance of this Contract, the Contractor shall defend NJ TRANSIT and the State of New Jersey in any action or administrative proceeding commenced pursuant to this Act. The Contractor shall indemnify, protect, and save harmless NJ TRANSIT and the State, their agents, servants, and employees from and against any and all suits, claims, losses, demands, or damages of whatever kind or nature arising out of or claimed to arise out of the alleged violation. The Contractor shall, at its own expense, appear, defend, and pay any and all charges for legal services and any and all costs and other expenses arising from such action or administrative proceeding or incurred in connection therewith. If any action or administrative proceeding results in an award of damages against NJ TRANSIT or the State or if NJ TRANSIT or the State incur any expense to cure a violation of the Act, the Contractor shall satisfy and discharge the same at its own expense.

NJ TRANSIT shall, as soon as practicable after a claim has been made against it, give written notice thereof to the Contractor along will full and complete particulars of the claim. If any action or administrative proceeding is brought against NJ TRANSIT or any of its agents, servants, and employees, NJ TRANSIT

shall expeditiously forward to the Contractor every demand, complaint, notice, summons, pleading, or other process received by it or its representatives.

It is expressly agreed and understood that any approval by NJ TRANSIT of the services provided by the Contractor pursuant to this Contract will not relieve the Contractor of the obligation to comply with the Act and to defend, indemnify, protect, and save harmless NJ TRANSIT pursuant to this paragraph.

The Contractor expressly understands and agrees that the provisions of this indemnification clause shall in no way limit the Contractor's obligations assumed in this Contract, nor shall they be construed to relieve the Contractor from any liability, nor preclude NJ TRANSIT from taking any other actions available to it under any other provisions of this Contract or otherwise at law.

## 11.3 UTILIZATION OF DISADVANTAGED BUSINESS ENTERPRISES

In connection with the performance of this Contract, the Contractor shall cooperate with NJ TRANSIT in meeting its commitments and goals with regard to the maximum utilization of disadvantaged business enterprises and will use its best efforts to insure that disadvantaged business enterprises shall have the maximum practicable opportunity to compete for subcontract work under this Contract.

The term "disadvantaged business enterprise" means a for-profit small business concern that is owned and controlled by one or more socially and economically disadvantaged persons, as defined by 49 CFR, Part 26. Such disadvantage may arise from cultural, racial, chronic economic circumstances or background or other similar cause. Such persons would include, but not be limited to, Black Americans (not of Hispanic origin); Hispanic Americans; Native Americans; Asian-Pacific Americans; Subcontinent Asian Americans; and Women, regardless of race or ethnicity.

The Contractor shall fully comply with the requirements and provisions set forth in the New Jersey Transit Corporation DBE Requirements for Procurement Activities.

## 12. MEASUREMENT AND PAYMENT

## 12.1 SCOPE OF PAYMENT

- 12.1.1 The Contractor shall receive and accept compensation provided for in the Contract as full payment for furnishing materials and for performing work under the Contract in a complete and acceptable manner and for risk, loss, damage or expense of whatever character arising out of the nature of the work or the prosecution thereof.
- 12.1.2 If the specifications relating to a unit price in the Proposal require that the said unit price cover and be considered compensation for certain work or material essential to the item, this same work or material will not also be measured or paid for under any other item which may appear elsewhere in the Contract.

- 12.1.3 If the specifications include Work for which no specific method of payment is provided, no separate payment will be made for that Work and the cost thereof shall be considered as included in the prices paid for the various scheduled Contract Items.
- 12.1.4 Except as specifically provided otherwise, no separate payment will be made for any of the requirements of the General and Special Provisions, and the cost thereof shall be considered as included in the various scheduled Contract Items.
- 12.1.5 Notwithstanding any other provision of this Contract, for a period of three years after acceptance, all estimates and payments (including the Final Certificate of Payment and payments made pursuant to the Final Certificate of Payment) shall be subject to correction and adjustment for clerical or other errors in the calculations involved in the determination of quantities and payments. The Contractor and NJ TRANSIT agree to pay to the other any sum due under the provisions of this Article.

### 12.2 QUANTITIES: MEASUREMENT AND PAYMENT

- 12.2.1 Work completed under the Contract will be measured by the Construction Manager according to United States standard measure. The method of measurement and computations to be used in determination of quantities of material furnished and of work performed under the Contract will be those methods generally recognized as conforming to good engineering practice.
- 12.2.2 Wherever the estimated quantities of work to be done and materials to be furnished under the Contract are shown in the documents including the proposal, they are given for use in comparing bids and the right is especially reserved, except as herein otherwise specifically limited, to increase or diminish them as may be deemed reasonably necessary or desirable by NJ TRANSIT to complete the work contemplated by this Contract, and such increase or diminution shall in no way abrogate this Contract, nor shall such increase or diminution give cause for claims or liability for damages.
- 12.2.3 When the Bidder's Proposal contains itemized quantities which are to be paid on a Unit Price basis, those quantities are designated as the Pay Quantities. When the estimated quantities for a specific portion of the work are designated as the Pay Quantities in the Contract, and if the work is actually performed as specified, they shall be the final quantities for which payment for such specific portion of the work will be made, unless the dimensions of said portions of the work shown on the plans are revised, or unless errors in the quantities are discovered. If revised dimensions result in an increase or decrease in the quantities of such work, the final quantities for payment will be revised in the amount represented by the authorized changes in the dimensions.
- 12.2.4 Wherever the actual quantity of work performed varies more than twenty-five percent (25%) above or below an estimated quantity shown in the Bidder's Proposal, an equitable adjustment to the contract shall be made upon demand of either NJ TRANSIT or the Contractor. The equitable adjustment shall be based upon an

increase or decrease in costs due solely to the twenty-five percent (25%) variation above or below the estimated quantity.

## 12.3 PARTIAL PAYMENTS

12.3.1 Monthly estimates will be made of the approximate quantities of work satisfactorily performed in accordance with the Contract Documents during the preceding month. Partial payments on account of such monthly estimate will be made based on the prices bid in the Proposal or as provided by change order. The Contractor will also be paid under the monthly estimates for materials delivered in accordance with Article 12.4, MATERIALS PAYMENTS and Article 8.2, MEASUREMENT OF PAY LIMITS FOR EXCAVATIONS.

For each lump sum bid item excluding Performance/Payment Bond, Mobilization and Allowances, the Contractor shall submit for the Construction Manager's review and approval a "Schedule of Values" (a detailed price breakdown of all individual items of work that are contained in said bid items) within ten (10) days of the Notice to Proceed. The approved Schedule of Values shall be incorporated into each Application for Payment and shall be used by NJ TRANSIT as the basis for partial payment and, if it so elects, as a basis for determining values of work it wishes to modify or delete.

- 12.3.2 No such estimate or payment shall be required to be made, in the judgment of NJ TRANSIT, when the work is not proceeding in accordance with the Contract Documents or following NJ TRANSIT giving the Contractor or surety notice of delay, neglect or default.
- 12.3.3 No such estimate or payment shall be construed to be an acceptance of any defective work or improper materials. NJ TRANSIT upon determining that any payment under a previous monthly estimate was improper or unwarranted for any reason may deduct the amount of such payment from the subsequent monthly estimate and partial payments made to the Contractor.
- 12.3.4 Material and work covered by partial payments made shall thereupon become the sole property of NJ TRANSIT but this provision shall not be construed as relieving the Contractor from the sole responsibility for care and protection of materials and work upon which payments have been made or restoration of any damaged work, or as a waiver of the right of NJ TRANSIT to require fulfillment of terms of the Contract.
- 12.3.5 NJ TRANSIT shall deduct from any monthly estimate and payment and the final payment such amounts as are required to be deducted pursuant to provisions of the Contract Documents.
- 12.3.6 In accordance with N.J.S.A. 52:32-40 and 52:32-41, prior to the issuance of a partial payment by NJ TRANSIT to the Contractor, the Contractor shall certify that a Subcontractor or Supplier has been paid any amount due from any previous partial payment and shall be paid any amount due from the current partial payment, or that a valid basis for withholding payment exists and the Contractor has complied with the applicable notice provisions.

## 12.4 MATERIALS PAYMENTS

- 12.4.1 The monthly estimates and payments made on account thereof will also include, when allowed by the Project Manager, an amount equal to the actual cost of materials furnished but not incorporated into the work, provided, however, that such amount shall not exceed 85 percent of the Contractor's bid price for the Contract Item into which the material will be incorporated, and the quantity allowed does not exceed the corresponding quantity estimated in the Contract.
- 12.4.2 Before including payments for such materials in an estimate, the Construction Manager must be satisfied that:
  - (a) The materials have been properly stored and protected by the Contractor or have been stored at locations owned or leased by NJ TRANSIT,
  - (b) The materials have been inspected and appear to be acceptable,
  - (c) The Contractor has provided NJ TRANSIT an invoice or bill of sale sufficient to show the price paid for the materials and proof that title, if applicable, has been transferred to NJ TRANSIT,
  - (d) The materials, if stored on property not belonging to NJ TRANSIT, are fenced in with access limited to NJ TRANSIT and the Contractor or their authorized agents and the fenced in materials are clearly identified in large letters as being without encumbrances and for use solely on this Project, and
  - (e) When such materials are stored in a leased area, the lease is made out to the Contractor and provides that it shall be cancelled only with the written permission of NJ TRANSIT.
- 12.4.3 The Contractor assumes full responsibility for the safe storage and protection of the materials and nothing in this Paragraph shall alter the provisions of Article 9.1, INDEMNIFICATION; RISKS ASSUMED BY THE CONTRACTOR. If materials paid for under this Article are damaged, stolen or prove to be unacceptable, the payment made therefor shall be deducted from subsequent estimates and payment.
- 12.4.4 Payment for materials as provided in this Article shall not be deemed to be an acceptance of such materials, and the Contractor shall be responsible for and must deliver to the site and properly incorporate in the work only those materials that comply with the Contract.
- 12.4.5 No payment for living or perishable plant materials will be made until they are accepted by NJ TRANSIT.
- 12.4.6 The Contractor shall pay all costs of handling and delivering materials to and from the place of storage to the site of the work, as well as storage rental. Taxes levied by any government against the materials shall be borne by the Contractor.

### 12.5 RETAINAGE

- 12.5.1 In making partial payments for work, there will be retained by NJ TRANSIT five percent of the estimated amount until completion and final acceptance of all work covered by the Contract and issuance of a Final Certificate of Payment.
- 12.5.2 The Contractor shall defend, indemnify and save NJ TRANSIT harmless from claims arising out of the demands of Subcontractors, laborers, workmen, mechanics, Suppliers, and furnishers of machinery and parts thereof, equipment, power tools, and supplies, including commissary, incurred in the furtherance of the performance of this Contract. The Contractor shall, at NJ TRANSIT's request, furnish satisfactory evidence that all obligations of the nature herein above designated have been paid, discharged or waived. The retainage specified herein shall not be paid to the Contractor until such obligations have been paid, discharged or waived. Recourse may also be made, if necessary, to the payment bond.

## 12.6 SUBCONTRACTOR PAYMENTS AND RETAINAGE

- 12.6.1 A Subcontractor shall look only to the Contractor for the payment of any claims of any nature whatsoever arising out of the said subcontract, and said Subcontractor agrees, as a condition of NJ TRANSIT's consent to the making of said subcontract, that it shall make no claim whatsoever against NJ TRANSIT, its officers, agents, servants or employees for any work performed or thing done by reason of said subcontract, or for any other cause whatsoever that may arise by reason of the relationship created between the Contractor and Subcontractor by the subcontract.
- 12.6.2 The Contractor agrees to pay each Subcontractor and Supplier under this Contract for satisfactory performance of its contract no later than ten (10) days from the receipt of each payment the Contractor receives from NJ TRANSIT. The Contractor shall ensure that all lower tier Subcontractors and suppliers are paid all invoiced amounts (less retainage) that meet all applicable requirements within 15 days from the receipt the Subcontractor receives from the Contractor.
- 12.6.3 In accordance with N.J.S.A. 52:32-40 and 52.32-41, the Contractor shall certify, prior to the issuance of a progress payment by NJ TRANSIT, that all Subcontractors and Suppliers have been paid any amounts due from previous progress payments and shall be paid any amounts due from the current progress payment. Alternatively, the Contractor shall certify that there exists a valid basis under the terms of the Subcontractor's or Supplier's contract to withhold payment from the Subcontractor or Supplier and therefore payment is withheld.
- 12.6.4 If the Contractor withholds payment from the Subcontractor or Supplier, the Contractor shall provide to the Subcontractor or Supplier written notice thereof. The notice shall detail the reason for withholding payment and state the amount of payment withheld. If a Performance/ Payment Bond has been provided under this Contract, the Contractor shall send a copy of the notice to the Surety providing the bond for the Contractor.

A copy of the notice shall also be submitted to NJ TRANSIT with the certification that payments are being withheld.

12.6.5 Should the Contractor provide notice and proceed to withhold payment from any Subcontractor or Supplier, NJ TRANSIT may elect, at its sole discretion, to help resolve the dispute. NJ TRANSIT's efforts shall be limited to meeting with the Contractor and the Subcontractor or Supplier and reviewing the relevant facts with both parties. NJ TRANSIT will not act as a decider of fact nor will NJ TRANSIT direct a settlement to the dispute. Any NJ TRANSIT effort is solely intended to assist the parties in understanding their respective positions and to encourage a reasonable resolution of the dispute.

12.6.6 The Contractor agrees to make retainage payments to each Subcontractor or Supplier within 15 days after the Subcontractor's or Supplier's work is completed. Only Subcontractors whose work has been 100% completed, including all Punchlist Work and any other Remaining Work, and who have supplied closeout documents shall be eligible for release of retainage. Any delay or postponement of payment from the above referenced time frame may occur only for good cause following written approval of NJ TRANSIT.

12.6.7 NJ TRANSIT agrees to release an equivalent amount of Contractor retainage provided that a) there are no offsetting claims from NJ TRANSIT (including, but not limited to, liquidated damages), other Subcontractors, Suppliers, materialmen or workers, and b) none of the other reasons to withhold payments specified under Article 12.7 exists. Prior to release of the Contractor's retainage, the Contractor shall provide to NJ TRANSIT executed copies of the following Subcontractor Closeout Documents, as appropriate: Consent of Surety to Final Payment to the Subcontractor, Subcontractor's Certificate of Amounts Due Workers For Wages, a Subcontractor Affidavit of Payment of Debts and Claims, a Subcontractor Affidavit of Release of Liens and a Certificate of Final Acceptance of Subcontractor Work, all in the form shown in Appendix B to the Contract.

12.6.8 Notwithstanding NJ TRANSIT's release or partial release of retainage, nothing in this Article shall be deemed to constitute NJ TRANSIT's partial or final acceptance of the Work, or any portion thereof, unless either a Certificate of Partial Acceptance or a Certificate of Final Acceptance has been executed by NJ TRANSIT, in the form(s) shown in Appendix B to the Contract.

## 12.7 PAYMENTS WITHHELD

- 12.7.1 NJ TRANSIT may withhold or, on account of subsequently discovered evidence, nullify the whole or a part of a certificate for payment to such extent as may be necessary to protect NJ TRANSIT from loss on account of:
  - Defective work not remedied;
  - (2) Claims filed, or reasonable evidence indicating probable filing of claims;

- (3) Failure of the Contractor to make payments promptly to Subcontractors or Suppliers for material or labor;
- (4) A reasonable doubt that the Contract can be completed for the balance then unpaid;
- (5) Damage to another contractor;
- (6) Lack of updated and approved CPM schedule;
- (7) Submission of incomplete payment invoice;
- (8) Liquidated damages;
- (9) Previous overpayments; and
- (10) Lack of compliance with Contract terms.
- 12.7.2 When the above grounds are removed, certificates of payment will be issued for amounts withheld because of them, less appropriate adjustments.
- 12.8 FINAL PAYMENT
- 12.8.1 Submissions required from the Contractor as a condition of final payment, include, but are not limited to, the following items:
  - (1) Completed Operations Insurance Certificate,
  - (2) Affidavit of Payment of Debts and Claims,
  - (3) Affidavit of Release of Liens,
  - (4) Consent of Surety to Final Payment,
  - (5) Certificate of amounts due workers for wages on the work pursuant to N.J.S.A. 34:11-56.25 et seq.,
  - (6) The one year and special written guarantees for periods of time in excess of the one year general guarantee,
  - (7) Operating instructions and maintenance manuals for equipment as required under Article 7.5.3. The maintenance and operating information shall be organized into suitable sets. Where applicable, these include: Operating and emergency instructions, replacement parts listing, maintenance contracts, warranties, guarantees, wiring diagrams, recommended "turn around" cycles, inspection procedures, shop drawings, product data, and similar applicable information for each type of equipment. Each set should be bound in a plastic covered binder. Identification should be printed clearly on both front and spine of each binder, and a complete typewritten index of contents should be provided. These shall be submitted to the Construction Manager for review by the Engineer. Corrections as required shall be made and then five (5) copies submitted in final form to NJ TRANSIT,
  - (8) Markup Drawings, as required under Article 6.11,
  - (9) Certificate of Final Acceptance,

- (10) Final payment request based on 100 percent completion of the work with all releases, certificates, consents, guarantees, warranties, and other documents, attached as required, including "Consent of Surety", and
- (11) Final Contractor Monthly DBE Payment Report.
- 12.8.2 The Proposed Final Certificate of Payment will show the total amount payable to the Contractor, including therein an itemization of said amount segregated as to Contract Item quantities, Extra Work and other basis for payment, and shall also show therein all deductions made or to be made for prior payments and as required pursuant to the provisions of the Contract Documents. All prior estimates and payments shall be subject to correction in the Proposed Final Certificate of Payment. Within 30 days after said Proposed Final Certificate of Payment has been issued to the Contractor, the Contractor shall submit to the Construction Manager its written approval of said Final Certificate of Payment or a written statement of all outstanding Contractor Initiated Change Order Requests (CICOR's) it has arising under or by virtue of the Contract or any action by any NJ TRANSIT employee, agent or officer in the prosecution of the Contract. CICOR's will not be considered unless the Contractor has strictly complied with the requirements of Article 3.4 CONTRACTOR INITIATED CHANGE ORDERS.
- 12.8.3 On the Contractor's approval, or if it files no statement of outstanding CICOR's within said period of 30 days, the Contracting Officer will issue a Final Certificate of Payment in writing in accordance with the proposed Final Certificate of Payment submitted to the Contractor and within 30 days thereafter, NJ TRANSIT will pay the entire sum due thereunder. Such Final Certificate of Payment and acceptance by the Contractor of the Final Payment based thereon shall operate as a release by the Contractor of the State and NJ TRANSIT, their agents, officers and employees, from all claims and liability of whatsoever nature for anything done or furnished or in any manner growing out of the performance of the Contract.
- 12.8.4 If the Contractor within said period of 30 days files a statement of outstanding CICOR's, the Contracting Officer will issue a Conditional Final Certificate of Payment in accordance with the proposed Final Certificate of Payment. Within 30 days thereafter, NJ TRANSIT will pay the sum due there under, provided the Contractor has in good faith provided the detailed CICOR cost information required by Article 3.4. The Contractor may request up to an additional 30 days within which to provide the required information.
- 12.8.5 Failure to submit such detailed cost information as to any CICOR within the 60 days provided from the date of the issuance of the proposed Final Certificate of Payment shall operate as a waiver of those CICOR's as to which such information is not provided and a release by the Contractor in favor of the State and NJ TRANSIT as to such CICOR. NJ TRANSIT will then issue a Conditional Final Payment based on the Conditional Final Certificate. Acceptance by the Contractor of this Conditional Final Payment shall constitute a release by the Contractor of the State and NJ TRANSIT, their agents, officers and employees, from all claims and liability of whatsoever nature for anything done or furnished or in any manner growing

out of the performance of the Contract except those CICOR's filed in response to the proposed Final Certificate and not waived as herein provided for failure to provide information and details.

- 12.8.6 The Contracting Officer's decision on outstanding CICOR's will be rendered in accordance with Article 1.15-DISPUTES.
- 12.8.7 Upon final resolution of the outstanding CICOR's, the Contracting Officer shall then make and issue a Final Certificate of Payment, and within 30 days thereafter, NJ TRANSIT will pay the entire sum, if any, found due thereon. Such Final Payment, if it resolves any of the CICOR's reserved under the Conditional Final Payment, will operate as a release in favor of the State, and NJ TRANSIT, their agents, officers and employees as to such claims.
- 12.8.8 No payment, however, final or otherwise, shall operate to release the Contractor or its Sureties from any obligations under this Contract or the Performance and Payment Bond.

### 12.9 SETTING OFF TAX ARREARS AGAINST SUMS OWED

Whenever any taxpayer under contract to provide goods or services to the State of New Jersey or its agencies or instrumentalities, and including the legislative and judicial branches of State government, or under contract for construction projects of the State of New Jersey or its agencies or instrumentalities, and including the legislative and judicial branches of State government, is entitled to payment for the goods or services or on that construction project and at the same time the taxpayer is indebted for any State tax, the Director of the Division of Taxation shall seek to set off so much of that payment as shall be necessary to satisfy the indebtedness. The Director, in consultation with the Director of the Division of Budget and Accounting in the Department of the Treasury, shall establish procedures and methods to effect a set-off. The Director shall give notice of the set-off to the taxpayer, the provider of goods or services or the contractor or subcontractor of construction projects and provide an opportunity for a hearing within 30 days of such notice under the procedures for protests established under N.J.S.A..54:49-18, but no request for conference, protest, or subsequent appeal to the Tax Court from any protest under this section shall stay the collection of the indebtedness. No payment shall be made to the taxpayer, the provider of goods or services or the contractor or subcontractor of construction projects pending resolution of the indebtedness. Interest that may be payable by the State pursuant to N.J.S.A.C.52:32-32 et seq., to the taxpayer, the provider of goods and services or the contractor or subcontractor of construction projects shall be stayed.

### 13. ACCEPTANCE AND COMPLETION

# 13.1 PARTIAL ACCEPTANCE

If at any time during the prosecution of the Project the Contractor completes a unit or portion of the Project, such as a structure or a section of right-of-way, it may request that NJ TRANSIT make a final inspection of that unit. NJ TRANSIT reserves the right to reject the request made by the Contractor if NJ TRANSIT, in its

sole discretion, determines that the unit or portion of the project should not be the subject of a Partial Acceptance.

If NJ TRANSIT determines that Partial Acceptance of the unit or portion of the Project is appropriate and finds upon inspection that the unit or portion is satisfactorily completed in compliance with the Contract, the Project Manager may accept that unit as being completed and the Contractor may be relieved of the responsibility of doing further work on or maintaining that unit or portion of the Project.

Such Partial Acceptance shall in no way void or alter the terms of the Contract, including Articles 9.1-INDEMNIFICATION; RISKS ASSUMED BY THE CONTRACTOR and 9.2- INSURANCE, nor shall it be construed as relieving the Contractor of full responsibility for making good defective work or materials found at any time before Final Acceptance pursuant to Article 13.3- COMPLETION AND FINAL ACCEPTANCE OF THE WORK.

## 13.2 SUBSTANTIAL COMPLETION

- 13.2.1 The Work shall be deemed substantially complete when, in the opinion of the Project Manager (whose judgment shall be conclusive), so much thereof has been completed in accordance with the terms of the Contract Documents that NJ TRANSIT may occupy the site of the work and use the work and the facilities resulting therefrom for the purposes for which they are intended. Unless the Project Manager determines that temporary pavement is sufficient, substantial completion will not be deemed to have occurred prior to the backfilling and restoration of street surfaces (if any) and the restoration of other surfaces, subsurfaces and overhead structures. Upon such substantial completion the Project Manager shall issue a Certificate of Substantial Completion. The issuance of this Certificate shall not relieve the Contractor from its obligation hereunder to finally complete all of the work of the Contract.
- 13.2.2 The work remaining to be completed after substantial completion in order for the Contractor to fulfill its obligations to fully complete the Work in accordance with the Contract shall be known as the "Remaining Work". The Remaining Work shall generally be limited to minor defects or omissions (also known as "Punch List Work"). However, NJ TRANSIT may include as part of Remaining Work, work which would ordinarily be required for substantial completion. Such other Remaining Work includes, but is not limited to, work not done because of seasonal factors or work which cannot be done until third persons perform other work which is not the Contractor's responsibility under the Contract. Nothing herein, however, shall diminish the right of NJ TRANSIT to determine what is necessary for substantial completion in accordance with Subarticle 13.2.1 above.
- 13.2.3 NJ TRANSIT shall advise the Contractor of the time required to complete Punch List Work and the time required to complete all other Remaining Work. Failure to complete in a timely manner all Remaining Work, other than Punch List Work, will result in the Contractor being liable for liquidated damages as set forth in Article 2.1, TIME OF COMPLETION DELAY LIQUIDATED DAMAGES. As an additional remedy for such

failure, and not in lieu of liquidated damages, NJ TRANSIT may complete the Remaining Work including Punch List Work, either by its own forces or by other contractors. The Contractor shall be entitled to payment according to the Contract Price upon such completion, subject however to NJ TRANSIT's right to reimbursement for its costs of completion. NJ TRANSIT may deduct such costs from any payment or payments due to the Contractor, and if such costs exceed the amount due the Contractor, the Contractor shall promptly pay such excess to NJ TRANSIT. NJ TRANSIT's entitlement to such reimbursement shall in no respect relieve the Contractor of its obligation to timely complete the Remaining Work.

13.2.4 Before final inspection, completion and acceptance of the Project, borrow and local material sources and areas occupied by the Contractor in connection with the work shall be cleaned of rubbish, excess materials, temporary structures and equipment, and the work shall be left in an acceptable condition. The final inspection and acceptance will not be made by NJ TRANSIT until the Project has been completed, including all work identified as "Remaining Work" (Punch List Work).

## 13.3 COMPLETION AND FINAL ACCEPTANCE OF THE WORK

13.3.1 When the Contractor believes that the Project has been fully completed, the Contractor shall provide written notification to the Construction Manager that the Project is ready for final inspection by NJ TRANSIT.

If the Project Manager finds the Work to be in compliance with the Contract, it will notify the Contracting Officer establishing completion as of the date of notification from the Contractor. If the Contracting Officer concurs, the Contractor will be issued a Certificate of Final Acceptance.

If the Project Manager's inspection discloses that the Work is not in conformance with the Contract, the Construction Manager will advise the Contractor as to the particular defects to be remedied. Upon correction of the defects, the Contractor shall provide written notification to the Construction Manager and another inspection shall be made. This procedure shall be repeated until the Project Manager finds the work to be in compliance with the Contract.

Payments made to the Contractor before the final acceptance do not commit NJ TRANSIT to acceptance of the Project.

13.3.2 NJ TRANSIT shall not be precluded or estopped, by any measurement, estimate or certificate made either before or after the completion and final acceptance of the Project and payment therefor if such measurement, estimate or certificate is found to be in error or untrue, from showing the true amount and character of the work performed and materials furnished by the Contractor, or from showing that any such measurement, estimate or certificate is incorrectly made or untrue, or that the work or materials do not conform in fact to the requirements of the Contract. NJ TRANSIT shall not be precluded or estopped, notwithstanding any such measurement, estimate or certificate, and payment made in accordance

therewith, from recovering from the Contractor and its Surety such damages as it may sustain by reason of the Contractor's failure to comply or to have complied with the terms of the Contract.

13.3.3 The Contractor, without prejudice to the terms of the Contract, shall be liable to NJ TRANSIT at any time both before and after acceptance for latent defects, fraud, or such gross mistakes as may amount to fraud, or as regards NJ TRANSIT's rights under any warranty or guarantee.

## 14. FEDERAL PROVISIONS

## 14.1 EMPLOYEE PROTECTIONS-CONSTRUCTION ACTIVITIES

The Contractor agrees to comply, and assures the compliance by each subcontractor at any tier, with the following employee protection requirements for construction employees:

- Davis-Bacon Act, as amended, 49 U.S.C. § 5333(a), the Davis-Bacon Act, 40 U.S.C. §§ 276a-276a(7), and U.S. DOL regulations, "Labor Standards Provisions Applicable to Contracts Governing Federally Financed and Assisted Construction", 29 CFR Part 5, and 29 CFR Parts 1 & 3.
- 2. Contract Work Hours and Safety Standards Act, as amended, in particular with the requirements of section 102 of the Act, 40 U.S.C. §§ 327-332; and U.S. DOL regulations, "Labor Standards Provisions Applicable to Contracts Governing Federally Financed and Assisted Construction", 29 C.F.R. Part 5; and with section 107 of the Act, 40 U.S.C. S 333, and U.S. DOL regulations, "Safety and Health Regulations for Construction", 29 C.F.R. Part 1926;
- Copeland "Anti-Kickback" Act, as amended, 18 U.S.C. § 874 and 40 U.S.C. S 276c, and U.S. DOL regulations, "Contractors and Subcontractors on Public Building or Public Work Financed in Whole or in part by Loans or Grants from the United States", 29 CFR Part 3.

## 14.1.1 MINIMUM WAGES

The Contractor shall comply with the following labor provisions. Should wage rates determined in accordance with the following conflict with those determined in accordance with Article 11.1, New Jersey Prevailing Wage Act, the greater of the two rates apply.

(a) All laborers and mechanics employed or working upon the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the Construction or Development of the Project), will be paid unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account (except such payroll deductions as are permitted by regulations issued by the Secretary of Labor under the Copeland Act, 29 CFR Part 3), the full amount of wages and bona fide fringe benefits (or cash equivalents thereof) due at the time of payment computed at rates not less than those contained in the wage determination of the

Secretary of Labor which is attached hereto and made a part hereof, regardless of any contractual relationship which may be alleged to exist between the Contractor and such laborers and mechanics.

Contributions made or costs reasonably anticipated for bona fide fringe benefits under Section 1(b)(2) of the Davis-Bacon Act on behalf of laborers or mechanics are considered wages paid to such laborers or mechanics, subject to the provisions of 29 CFR 5.5(a)(1)(iv); also, regular contributions made or costs incurred for more than a weekly period (but not less often than quarterly) under plans, funds, or programs that cover the particular weekly period, are deemed to be constructively made or incurred during such weekly period. Such laborers and mechanics shall be paid the appropriate wage rate and fringe benefits on the wage determination for the classification of work actually performed, without regard to skill, except as provided at 29 CFR 5.5(a)(4). Laborers or mechanics performing work in more than one classification may be compensated at the rate specified for each classification for the time actually worked therein: provided, that the employer's payroll records accurately set forth the time spent in each classification in which work is performed. The wage determination (including any additional classification and wage rates conformed 29 CPR 5.5(a)(1)(ii) of this section) and the Davis-Bacon poster (WH-1321) shall be posted at all times by the Contractor and its subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

- (b) 1. The Contracting Officer shall require that any class of laborers or mechanics, including helpers, which is not listed in the wage determination and which is to be employed under the Contract shall be classified in conformance with the wage determination. The Contracting Officer shall approve an additional classification and wage rate and fringe benefits therefore only when the following criteria have been met:
  - (a) The work to be performed by the classification requested is not performed by a classification in the wage determination; and
  - (b) The classification is utilized in the area by the construction industry; and
  - (c) The proposed wage rate, including any bona fide fringe benefits, bears a reasonable relationship to the wage rates contained in the wage determination.
  - 2. If the Contractor and the laborers and mechanics to be employed in the classification (if known), or their representatives, and the Contracting Officer agree on the classification and wage rate (including the amount designated for fringe benefits where appropriate), a report of the action taken shall be sent by the Contracting Officer to the Administrator of the Wage and Hour Division, Employment Standards Administration, U.S. Department of Labor, Washington, DC 20210. The Administrator, or an authorized representative, will

approve, modify, or disapprove every additional classification action within 30 days of receipt and so advise the Contracting Officer or will notify the Contracting Officer within the 30-day period that additional time is necessary.

- In the event the Contractor, the laborers or mechanics to be employed in the classification or their representatives, and the Contracting Officer do not agree on the proposed classification and wage rate (including the amount designated for fringe benefits, where appropriate), the Contracting Officer shall refer the questions, including the views of all interested parties and the recommendation of the Contracting Officer, to the Administrator for determination. The Administrator, or an authorized representative, will issue a determination within 30 days of receipt and so advise the Contracting Officer or will notify the Contracting Officer within the 30-day period that additional time is necessary.
- (4) The wage rate (including fringe benefits where appropriate) determined pursuant to 29CFR5.5 shall be paid to all workers performing work in the classification under this Contract from the first day on which work is performed in the classification.
- (c) Whenever the minimum wage rate prescribed in the Contract for a class of laborers or mechanics includes a fringe benefit which is not expressed as a hourly rate, the Contractor shall either pay the benefit as stated in the wage determination or shall pay another bona fide fringe benefit or an hourly cash equivalent thereof.
- (d) If the Contractor does not make payments to a trustee or other third person, the Contractor may consider as part of the wages of any laborer or mechanic the amount of any costs reasonably anticipated in providing bona fide fringe benefits under a plan or program, provided, that the Secretary of Labor has found, upon the written request of the Contractor, that the applicable standards of the Davis-Bacon Act have been met. The Secretary of Labor may require the Contractor to set aside in a separate account assets for the meeting of obligations under the plan or program.

# 14.1.2 WITHHOLDING

FTA shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld from the Contractor under this Contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to Davis-Bacon prevailing wage requirements, which is held by the same prime Contractor, so much of the accrued payments or advances as may be considered necessary to pay laborers and mechanics, including apprentices, trainees, and helpers, employed by the Contractor or any subcontractor the full amount of wages required by the Contract. In the event of failure to pay any laborer or mechanic, including any apprentice, trainee, or helper, employed or working on the site of the work (or under the United States Housing Act of 1937 or under the

Housing Act of 1949 in the Construction or Development of the Project), all or part of the wages required by the contract, FTA may, after written notice to the Contractor, sponsor, applicant or owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds until such violations have ceased.

## 14.1.3 PAYROLLS AND BASIC RECORDS

- Payrolls and basic records relating thereto shall be maintained by the Contractor during the course (a) of the work and preserved for a period of three years thereafter for all laborers and mechanics working at the site of the work (or under the United States Housing Act of 1937, or under the Housing Act of 1949, in the Construction or Development of the Project). Such records shall contain the name, address, and social security number of each such worker, his or her correct classification, hourly rates of wages paid (including rates of contributions or costs anticipated for bona fide fringe benefits or cash equivalents thereof of the types described in Section 1(b)(2)(B) of the Davis-Bacon Act), daily and weekly number of hours worked, deductions made and actual wages paid. Whenever the Secretary of Labor has found under 29 CFR 5.5(a)(1)(iv) that the wages of any laborer or mechanic include the amount of any costs reasonably anticipated in providing benefits under a plan or program described in Section 1(b)(2)(B) of the Davis-Bacon Act, the Contractor shall maintain records which show that the commitment to provide such benefits is enforceable, that the plan or program is financially responsible, and that the plan or program has been communicated in writing to the laborers or mechanics affected, and records which show the costs anticipated or the actual costs incurred in providing such benefits. Contractors employing apprentices or trainees under approved programs shall maintain written evidence of the registration of apprenticeship programs and certification of trainee programs, the registration of the apprentices and trainees, and the ratios and wage rates prescribed in the applicable programs.
- (b) 1. The contractor shall submit weekly for each week in which any contract work is performed a copy of all payrolls to the FTA if the agency is a party to the contract, but if the agency is not such a party, the contractor will submit the payrolls to the applicant, sponsor, or owner, as the case may be, for transmission to the FTA. The payrolls submitted shall set out accurately and completely all of the information required to be maintained under 29 CFR 5.5(a)(3)(i), except that full social security numbers and home addresses shall not be included on weekly transmittals. Instead the payrolls shall only need to include an individually identifying number for each employee (e.g., the last four digits of the employee's social security number). The required weekly payroll information may be submitted in any form desired. Optional Form WH-347 is available for this purpose from Web the Wage and Hour Division site at http://www.dol.gov/esa/whd/forms/wh347instr.htm or its successor site. The prime contractor is responsible for the submission of copies of payrolls by all subcontractors. Contractors and subcontractors shall maintain the full social security number and current

address of each covered worker, and shall provide them upon request to the FTA if the agency is a party to the contract, but if the agency is not such a party, the contractor will submit them to NJ TRANSIT for transmission to the FTA, the contractor, or the Wage and Hour Division of the Department of Labor for purposes of an investigation or audit of compliance with prevailing wage requirements. It is not a violation of 29 CFR 5.5(a)(3) for a prime contractor to require a subcontractor to provide addresses and social security numbers to the prime contractor for its own records, without weekly submission to the sponsoring government agency (or the applicant, sponsor, or owner).

- 2. Each payroll submitted shall be accompanied by "Statement of Compliance", signed by the Contractor or subcontractor or his or her agent who pays or supervises the payment of the persons employed under the Contract and shall certify the following:
  - (a) That the payroll for the payroll period contains the information required to be provided under 29 CFR 5.5 (a)(3)(ii), the appropriate information is being maintained under 29 CFR 5.5 (a)(3)(i) and that such information is correct and complete;
  - (b) That each laborer or mechanic (including each helper, apprentice, and trainee) employed on the Contract during the payroll period has been paid the full weekly wages earned, without rebate, either directly or indirectly, and that no deductions have been made either directly or indirectly from the full wages earned, other than permissible deductions as set forth at 29 CFR Part 3;
  - (c) That each laborer or mechanic has been paid not less than the applicable wage rates and fringe benefits or cash equivalents for the classification of work performed, as specified in the applicable wage determination incorporated into the contract.
- The weekly submission of a properly executed certification set forth on the reverse side of optional form WH-347 shall satisfy the requirement for submission of the "Statement of Compliance" required by 29 CFR 5.5(a)(3)(ii)(b).
- 4. The falsification of any of the above certifications may subject the Contractor or subcontractor to civil or criminal prosecution under 18 U.S.C., Section 1001 and 31 U.S.C., Section 231.
- (c) The Contractor or subcontractor shall make the records required under 29 CFR 5.5(a)(3)(i) available for inspection, copying, or transcription by authorized representatives of FTA or the Department of Labor, and shall permit such representatives to interview employees during working hours on the job. If the Contractor or subcontractor fails to submit the required records or make them available, FTA may, after written notice to the Contractor, sponsor, applicant, or owner, take

such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds. Furthermore, failure to submit the required records upon request or make such records available may be grounds for debarment action pursuant to 29 CFR 5.12.

### 14.1.4 APPRENTICES AND TRAINEES

- Apprentices Apprentices will be permitted to work at less than the predetermined rate for the (a) work they performed when they are employed pursuant to and individually registered in a bona fide apprenticeship program registered with the U.S. Department of Labor, Employment and Training Administration, Office of Apprenticeship Training, Employer and Labor Services, or with a State Apprenticeship Agency recognized by the Office, or if a person is employed in his or her first 90 days of probationary employment as an apprentice in such an apprenticeship program, who is not individually registered in the program, but who has been certified by the Office of Apprenticeship Training, Employer and Labor Services, or a State Apprenticeship Agency (where appropriate) to be eligible for probationary employment as an apprentice. The allowable ratio of apprentices to journeymen on the job site in any craft classification shall not be greater than the ratio permitted to the Contractor as to the entire work force under the registered program. Any worker listed on a payroll at an apprentice wage rate, who is not registered or otherwise employed as stated above, shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any apprentice performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. Where a Contractor is performing construction on a project in a locality other than that in which its program is registered, the ratios and wage rates (expressed in percentages of the journeyman's hourly rate) specified in the Contractor's or subcontractor's registered program shall be observed. Every apprentice must be paid at not less than the rate specified in the registered program for the apprentice's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Apprentices shall be paid fringe benefits in accordance with the provisions of the apprenticeship program. If the apprenticeship program does not specify fringe benefits, apprentices must be paid the full amount of fringe benefits listed on the wage determination for the applicable classification. If the Administrator determines that a different practice prevails for the applicable apprentice classification, fringes shall be paid in accordance with that determination. In the event the Office of Apprenticeship Training, Employer and Labor Services, or a State apprenticeship agency recognized by the Office, withdraws approval of an apprenticeship program, the Contractor will no longer be permitted to utilize apprentices at less than the applicable predetermined rate for the work performed until an acceptable program is approved.
- (b) <u>Trainees</u> Except as provided in 29 CFR 5.16, trainees will not be permitted to work at less than the predetermined rate for the work performed unless they are employed pursuant to and

individually registered in a program which has received prior approval, evidenced by formal certification by the U.S. Department of Labor, Employment and Training Administration. The ratio of trainees to journeymen on the job site shall not be greater than permitted under the plan approved by the Employment and Training Administration. Every trainee must be paid at not less than the rate specified in the approved program for the trainee's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Trainees shall be paid fringe benefits in accordance with the provisions of the trainee program. If the trainee program does not mention fringe benefits, trainees shall be paid the full amount of fringe benefits listed on the wage determination unless the Administrator of the Wage and Hour Division determines that there is an apprenticeship program associated with the corresponding journeyman wage rate on the wage determination which provides for less than full fringe benefits for apprentices. Any employee listed on the payroll at a trainee rate who is not registered and participating in a training plan approved by the Employment and Training Administration shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any trainee performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. In the event the Employment and Training Administration withdraws approval of a training program, the Contractor will no longer be permitted to utilize trainees at less than the applicable predetermined rate for the work performed until an acceptable program is approved.

(c) Equal Employment Opportunity - The utilization of apprentices, trainees, and journeymen under 29 CFR Part 5 shall be in conformity with the Equal Employment Opportunity requirements of Executive Order 11246, as amended, and 29 CFR Part 30.

## 14.1.5 COMPLIANCE WITH COPELAND ACT REQUIREMENTS

The Contractor shall comply with the requirements of 29 CFR Part 3, which are incorporated herein by reference.

## 14.1.6 CONTRACT TERMINATION: DEBARMENT

A breach of the Contract clauses in 29 CFR 5.5 may be grounds for termination of the contract, and for debarment as a Contractor and a subcontractor as provided in 29 CFR 5.12.

### 14.1.7 COMPLIANCE WITH DAVIS-BACON AND RELATED ACT REQUIREMENTS

All rulings and interpretations of the Davis-Bacon and related acts contained in 29 CFR Parts 1, 3, and 5 are incorporated herein by reference.

### 14.1.8 DISPUTES CONCERNING LABOR STANDARDS

Disputes arising out of the labor standards provisions of this Contract shall not be subject to the general disputes clause of this Contract. Such disputes shall be resolved in accordance with the procedures of the Department of Labor set forth in 29 CFR Parts 5, 6, and 7. Disputes within the meaning of this clause include disputes between the Contractor (or any of its subcontractors) and NJ TRANSIT, the U.S. Department of Labor, or the employees or their representatives.

### 14.1.9 CERTIFICATION OF ELIGIBILITY

- (a) By entering into this Contract, the Contractor certifies that neither it (nor he nor she) nor any person or firm that has an interest in the Contractor's firm is a person or firm ineligible to be awarded government contracts by virtue of Section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (b) No part of this Contract shall be subcontracted to any person or firm ineligible for award of a government contract by virtue of Section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (c) The penalty for making false statements is prescribed in the U.S. Criminal Code, 18 U.S.C. 1001.

## 14.1.10 OVERTIME REQUIREMENTS

No Contractor or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics shall require or permit any such laborer or mechanic in any work week in which he or she is employed on such work to work in excess of forty hours in such work week unless such laborer or mechanic receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of forty hours in such work week.

# 14.1.11 VIOLATION; LIABILITY FOR UNPAID WAGES; LIQUIDATED DAMAGES

In the event of any violation of the requirements of 29 CFR Part 5.5(b)(1), the Contractor and any subcontractor responsible therefor shall be liable for the unpaid wages. In addition, such Contractor and subcontractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a territory, to such District or to such territory) for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of 29 CFR 5.5(b)(1) in the sum of \$10 for each calendar day on which such individual was required or permitted to work in excess of the standard work week of forty hours without payment of the overtime wages required by 29 CFR 5.5(b)(1).

#### 14.1.12 WITHHOLDING FOR UNPAID WAGES AND LIQUIDATED DAMAGES

FTA or NJ TRANSIT shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld, from any moneys payable on account of work performed by the Contractor or subcontractor under any such contract or any other Federal contract with the same prime Contractor, or any other Federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same prime Contractor, such sums as may be determined to be necessary to satisfy any liabilities of such Contractor or subcontractor for unpaid wages and liquidated damages as provided in the clause set forth at 29 CFR 5.5(b)(2).

### 14.1.13 SUBCONTRACTS

The Contractor or subcontractor shall insert in any subcontracts the clauses set forth in this Article 14.1 and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime Contractor shall be responsible for compliance by any subcontractor or lower tier subcontractor with the clauses set forth in this Article 14.1.

## 14.2 CLEAN WATER AND CLEAN AIR ACTS

- 14.2.1 The Contractor shall comply with all applicable standards, orders, or regulations issued pursuant to the Federal Water Pollution Control Act, as amended, 33 U.S.C. § 1251 et seq.; among other things:
  - (a) With the provisions of the Safe Drinking Water Act of 1974, as amended, 42 U.S.C. § 300h et seq.
  - (b) With the notification of violating facilities provisions of Executive Order No. 11738; "Administration of the Clean Air Act and the Federal Water Pollution Control Act with Respect to Federal Contracts, Grants, or Loans", 42 U.S.C. S 7606 note. The Contractor agrees to report violations to NJ TRANSIT, FTA and to the USEPA Assistant Administrator or Enforcement.
- 14.2.2 The Contractor agrees to comply with all applicable regulations, standards, or orders implementing the Clean Air Act, as amended, 42 U.S.C. § 7401 et seq. The Contractor agrees to report violations to NJ TRANSIT, FTA and to the USEPA Assistant Administrator for Enforcement.
- 14.2.3 Both the Clean Water and Clean Air requirements flow down to all subcontracts which exceed \$100,000. The \$100,000 limit includes indefinite quantities where the amount is expected to exceed this limit in any year.

### 14.3 BUY AMERICA

Pursuant to 49 U.S.C. 5323(j) (P.L. 106-274), and the regulation found at 49 CFR Part 661, the Contractor agrees that all iron, steel and manufactured products purchased or used as a result of this Contract shall be of domestic manufacture or origin unless a waiver of these provisions is granted by the U.S. Secretary of Transportation.

There are four exceptions to this requirement:

- (a) That its application would be inconsistent with the public interest.
- (b) That such materials or products are not produced in the United States in sufficient and reasonably available quantities and of a satisfactory quality.
- (c) With respect to rolling stock only, but including train control, communications, traction power equipment, 40 percent may be non-domestic manufacture, but final assembly of such products must take place in the United States.
- (d) That inclusion of domestic material will increase the cost of the overall project contract by more than 25 percent.

### 14.4 FLY AMERICA REQUIREMENTS

The Contractor agrees to comply with 49 U.S.C. 40118 (the "Fly America" Act) in accordance with the General Services Administration's regulations at 41 CFR Part 301-10, which provide that recipients and subrecipients of Federal funds and their contractors are required to use U.S. Flag air carriers for U.S. Government-financed international air travel and transportation of their personal effects or property, to the extent such service is available, unless travel by foreign air carrier is a matter of necessity, as defined by the Fly America Act. The Contractor shall submit, if a foreign air carrier was used, an appropriate certification or memorandum adequately explaining why service by a U.S. flag air carrier was not available or why it was necessary to use a foreign air carrier and shall, in any event, provide a certificate of compliance with the Fly America requirements. The Contractor agrees to include the requirements of this section in all subcontracts that may involve international air transportation.

### 14.5 CARGO PREFERENCE - USE OF U.S. FLAG VESSELS

The Contractor agrees to utilize privately owned United States-flag commercial vessels to ship at least 50 percent of the gross tonnage (computed separately for dry bulk carriers, dry cargo liners, and tankers) involved, whenever shipping any equipment, materials, or commodities pursuant to 46 CFR Part 381, to the extent such vessels are available at fair and reasonable rates for United States-flag commercial vessels.

The Contractor further agrees to furnish within 20 working days following the date of loading for shipments originating within the United States, or within 30 working days following the date of loading for shipment originating outside the United States, a legible copy of a rated "on-board" commercial ocean bill-of-lading in English for each shipment of cargo described in the Paragraph above to NJ TRANSIT (through the prime Contractor in the case of subcontractor bills-of-lading) and to the Division of National Cargo, Office of Market Development, Maritime Administration, Washington, D.C. 20590, marked with appropriate identification of the Project.

The Contractor further agrees to insert the substance of the provisions of this clause in all subcontracts issued pursuant to this Contract.

## 14.6 ENERGY CONSERVATION

The Contractor shall comply with mandatory standards and policies relating to energy efficiency contained in applicable State of New Jersey energy conservation plans issued in compliance with the Energy Policy and Conservation Act (42 USC 6321 et seq.).

## 14.7 CONTRACT WORK HOURS AND SAFETY STANDARDS

Each Contractor shall be required to compute the wages of every mechanic and laborer on the basis of a standard work day of eight (8) hours and a standard work week of forty (40) hours. Work in excess of a work day or work week is permissible provided that the worker is compensated at a rate of not less than one and one half (1½) times the basic rate of pay for all hours worked in excess of eight (8) hours of any calendar day or forty (40) hours in the work week. No laborer or mechanic shall be required to work in surroundings or under working conditions which are unsanitary, hazardous or dangerous to his/her health and safety. The foregoing provisions are to be in compliance with Section 103 and 107 of the Contract Work Hours and Safety Standards Act (40 USC 327-330) as supplemented by Department of Labor Regulations (29 CFR, Part 5).

## 14.8 TITLE VI COMPLIANCE

During the performance of this Contract, the Contractor, for itself, its assignees and successors in interest and its subcontractors at every tier (hereinafter referred to as the "Contractor") agrees as follows:

### (a) Compliance with Regulations

The Contractor shall comply with the Regulations relative to nondiscrimination in federally-assisted programs of the United States Department of Transportation, Title 49, Code of Federal Regulations, Part 21, as they may be amended from time to time (hereinafter referred to as the Regulations), which are herein incorporated by reference and made a part of this Contract.

# (b) Nondiscrimination

In accordance with Title VI of the Civil Rights Act, as amended, 42 U.S.C. § 2000d, section 303 of the Age Discrimination Act of 1975, as amended, 42 U.S.C. § 6102, section 202 of the Americans with Disabilities Act of 1990, 42 U.S.C. § 12132, and Federal transit law at 49 U.S.C. § 5332, the Contractor agrees that it will not discriminate against any employee or applicant for employment because of race, color, creed, national origin, sex, age, or disability. In addition, the Contractor agrees to comply with applicable Federal implementing regulations and other implementing requirements FTA may issue.

# (c) Equal Employment Opportunity

The following equal employment opportunity requirements apply to the underlying contract:

## (1) Race, Color, Creed, National Origin, Sex

In accordance with Title VII of the Civil Rights Act, as amended, 42 U.S.C. § 2000e, and Federal transit laws at 49 U.S.C. § 5332, the Contractor agrees to comply with all applicable equal employment opportunity requirements of U.S. Department of Labor (U.S. DOL) regulations, "Office of Federal Contract Compliance Programs, Equal Employment Opportunity, Department of Labor," 41 C.F.R. Parts 60 et seq ., (which implement Executive Order No. 11246, "Equal Employment Opportunity," as amended by Executive Order No. 11375, "Amending Executive Order 11246 Relating to Equal Employment Opportunity," 42 U.S.C. § 2000e note), and with any applicable Federal statutes, executive orders, regulations, and Federal policies that may in the future affect construction activities undertaken in the course of the Project. The Contractor agrees to take affirmative action to ensure that applicants are employed, and that employees are treated during employment, without regard to their race, color, creed, national origin, sex, or age. Such action shall include, but not be limited to, the following: employment, upgrading, demotion or transfer, recruitment or recruitment advertising, layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. In addition, the Contractor agrees to comply with any implementing requirements FTA may issue.

## (2) Age

In accordance with section 4 of the Age Discrimination in Employment Act of 1967, as amended, 29 U.S.C. § § 623 and Federal transit law at 49 U.S.C. § 5332, the Contractor agrees to refrain from discrimination against present and prospective employees for reason of age. In addition, the Contractor agrees to comply with any implementing requirements FTA may issue.

## (3) Disabilities

In accordance with section 102 of the Americans with Disabilities Act, as amended, 42 U.S.C. § 12112, the Contractor agrees that it will comply with the requirements of U.S. Equal Employment Opportunity Commission, "Regulations to Implement the Equal Employment Provisions of the Americans with Disabilities Act," 29 C.F.R. Part 1630, pertaining to employment of persons with disabilities. In addition, the Contractor agrees to comply with any implementing requirements FTA may issue.

(d) The Contractor also agrees to include these requirements in each subcontract financed in whole or in part with Federal assistance provided by FTA, modified only if necessary to identify the affected parties.

## (e) <u>Information and Reports</u>

The Contractor shall provide all information and reports required by the Regulations or directives issued pursuant thereto, and shall permit access to its books, records, accounts, other sources of information, and its facilities as may be determined by the Recipient or NJ TRANSIT or the FTA to be pertinent to ascertain compliance with such Regulations, orders and instruction. Where any information is required or a Contractor is in the exclusive possession of another who fails or refuses to furnish this information, the Contractor shall so certify to NJ TRANSIT, or the FTA, as appropriate, and shall set forth what efforts it has made to obtain the information.

# (f) Sanctions for Noncompliance

In the event of the Contractor's noncompliance with the nondiscrimination provisions of this Contract, NJ TRANSIT shall impose such contract sanctions as it or the FTA may determine to be appropriate, including but not limited to:

- (1) Withholding of payments to the Contractor under the Contract until the Contractor complies; and/or
- (2) Cancellation, termination or suspension of the Contract, in whole or in part.

### 14.9 PATENT RIGHTS

If any invention, improvement, or discovery of the Contractor is conceived or first actually reduced to practice in the course of or under this Contract, which invention, improvement or discovery may be patentable under the laws of the United States of America or any foreign country, the Contractor shall immediately notify NJ TRANSIT and provide a detailed report for transmission to FTA.

The rights and responsibilities of NJ TRANSIT, the Contractor and FTA with respect to such invention, improvement, or discovery will be determined in accordance with applicable Federal laws, regulations, policies and any waiver thereof.

## 14.10 RCRA REQUIREMENTS: BUILDING INSULATION PRODUCTS AND FLY ASH IN CONCRETE

The contractor agrees to comply with all the requirements of Section 6002 of the Resource Conservation and Recovery Act (RCRA), as amended (42 USC 6962), including but not limited to the regulatory provisions of 40 CFR Part 247, and Executive Order 12873, as they apply to the procurement of the items designated in Subpart B of 40 CFR Part 247.

### 14.10.1 Building Insulation Products

NJ TRANSIT has adopted minimum recovered material content standards for the following types of insulating material:

Material Type	Percent By Weight
Celluose Loose Fill and Spray-On	75 percent post-consumer recovered paper

Perlite Composite Board	23 percent post-consumer recovered paper
Plastic Rigid Foam,	
Polyisocyanurate/Polyurethane	
Rigid Foam:	9 percent recovered material
Foam-in-place:	5 percent recovered material
Glass Fiber Reinforced:	6 percent recovered material
Phenolic Rigid Foam	5 percent recovered material
Rock Wool	50 percent recovered material

Note: The minimum content standards are based on the weight of material (not volume) in the insulating core.

The Contractor agrees to certify, prior to delivery, that building insulation products listed above which are to be supplied under the Contract shall meet or exceed the applicable minimum content standard.

Where NJ TRANSIT has specified building insulation products for which no minimum content standard has been established, the Contractor agrees to certify, prior to delivery, the minimum percentage of recovered materials that such products shall contain.

Also, the Contractor agrees to provide, at the time of installation, an estimate of the actual total percentage of recovered material contained in the building insulation product and an estimate of the product's cost.

The Contractor shall also obtain from the original manufacturer documentation that verifies the recovered material content of the building insulation products supplied. Such documentation should include manufacturer records identifying batch numbers and total percentage and type of recovered materials contained in the product. If specific batch data is unavailable, the average of recovered materials used by the manufacturer in a specific insulation product over a one-month period may be used.

The requirement to supply building insulation products containing recovered materials to the maximum extent practicable is subject to the following limitations:

- (1) Unsatisfactory level of competition;
- Unavailability within a reasonable period of time;
- (3) Inability to meet the performance standards in the applicable specifications;
- (4) Unavailability at a reasonable price.

Should the Contractor claim that one or more of the four limitations identified above applies to any building insulation products to be delivered or installed under this Contract, the Contractor shall so notify the Construction Manager in writing.

NJ TRANSIT reserves the right to reject any claims where the Contractor has failed to provide adequate

evidence that one or more of the four limitations applies.

Failure to deliver or install building insulation products in accordance with these requirements shall be

considered a material breach of the Contract.

14.10.2 Fly Ash In Concrete

Portland cement, Type II, which has been pre-blended with a maximum of 15 percent fly ash, by weight, and

conforming to ASTM C 595 may be used. When blended portland cement is used, no additional fly ash

shall be added.

Fly ash for portland cement concrete shall conform to ASTM C 618, Class C or Class F except that the loss

on ignition shall be not more than 3 percent. When Class C fly ash is used, the magnesium oxide shall not

exceed 2.5 percent. Before each source of fly ash is approved, certified results of test conducted by a

testing agency shall be submitted to and verified by the Department. Accompanying the certification shall

be a statement form the supplier listing the source and type of coal, the methods used to burn, collect, and

store the fly ash, and the quality control measures employed.

Conformance to the requirements for loss on ignition and fineness shall be determined by the supplier for

each truck load of fly ash delivered to the mixing site. The test values determined shall be included on the

delivery ticket. The Engineer may require that the fly ash not be used until the Department has performed

tests for loss on ignition and fineness.

Fly ash for other uses shall conform to ASTM C 593 except that the loss on ignition shall be not more than

10 percent and the combined content of silica and aluminum oxide shall be a minimum of 50 percent.

14.11 FEDERAL EEO PROVISIONS FOR CONSTRUCTION CONTRACTS

The Contractor, in addition to complying with Article 11.2, shall comply with the following Federal EEO

Provisions for Construction Contracts:

14.11.1 EQUAL EMPLOYMENT OPPORTUNITY

During the performance of this contract, the Contractor agrees as follows:

a.) In connection with the execution of this Contract, the Contractor shall not discriminate against any

employee or applicant for employment because of race, color, religion, sex or national origin. The

Contractor shall take affirmative action to ensure that applicants are employed, and that

employees are treated during their employment without regard to their race, color, religion, sex or

national origin. Such action shall include, but not be limited to the following: employment,

upgrading, demotion or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The Contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided setting forth the provisions of this nondiscrimination clause.

- b.) The Contractor will, in all solicitations or advertisements for employees placed by or on behalf of the Contractor, state that all qualified applicants will receive consideration for employment without regard to race, color, religion, sex, or national origin.
- c.) The Contractor will send to each labor union or representative of workers with which he has a collective bargaining agreement or other contract or understanding, a notice to be provided advising the said labor union or workers' representative of the Contractor's commitments under this Article, and shall post copies of the notice in conspicuous places available to employees and applicants for employment.
- d.) The Contractor will comply with all provisions of Executive Order 11246 of September 24, 1965, as amended by Executive Order 11375, and of the rules, regulations and relevant orders of the Secretary of Labor.
- e.) The Contractor will furnish all information and reports required by Executive Order 11246 of September 24, 1965, and by the rules, regulations, and orders of the Secretary of Labor or pursuant thereto, and will permit access to his books, records and accounts by the Federal Transit Administration (FTA) and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations and orders.
- f.) In the event of the Contractor's noncompliance with the nondiscrimination clauses of this contract or with any of the said rules, regulations or orders, this contract may be cancelled, terminated or suspended in whole or in part and the Contractor may be declared ineligible for further Government contracts or federally assisted construction contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions as may be imposed and remedies invoked as provided in the said Executive Order or by rule, regulation or order of the Secretary of Labor, or as otherwise provided by law.
- g.) The Contractor will include the provisions of paragraphs (a) through (g) in every subcontract or purchase order unless exempted by rules, regulations or orders of the Secretary of Labor issued pursuant to Section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each subcontractor or vendor. The Contractor will take such action with respect to any subcontract or purchase order as FTA may direct as a means of enforcing such provisions, including sanctions for noncompliance; provided, however, that in the event a

Contractor becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction by FTA, the Contractor may request the United States to enter into such litigation to protect the interests of the United States.

- 14.11.2 NOTICE OF REQUIREMENT FOR AFFIRMATIVE ACTION TO ENSURE EQUAL EMPLOYMENT OPPORTUNITY (EXECUTIVE ORDER 11246)
  - a.) The Contractor's attention is called to the "Equal Opportunity Clause" and the "Standard Federal Equal Employment Opportunity Construction Contract Specifications" set forth herein.
  - b.) The goals and timetables for minority and female participation expressed in percentage terms for the Contractor's aggregate work force in each trade on all construction work in the covered area, are as follows:

# **GOALS FOR MINORITY PARTICIPATION (Timetable Until Further Notice)**

County	Goal Percent
Atlantic	18.2
Bergen	22.6
Burlington	17.3
Camden	17.3
Cape May	14.5
Cumberland	16.0
Essex	17.3
Gloucester	17.3
Hudson	12.8
Hunterdon	17.0
Mercer	16.4
Middlesex	5.8
Monmouth	9.5
Morris	17.3
Ocean	17.0
Passaic	12.9
Salem	12.3
Somerset	17.3
Sussex	17.0
Union	17.3
Warren	1.6

# GOALS FOR FEMALE PARTICIPATION (Timetable Until Further Notice)

<u>County</u> <u>Goal Percent</u>

All 6.9

These goals are applicable to all the Contractor's construction work (whether or not it is Federal or federally assisted) performed in the covered area.

The Contractor's compliance with the Executive Order and the regulations in 41 CFR Part 60-4 shall be based on its implementation of the Equal Employment Opportunity Clause, specific affirmative action obligations required by the specifications set forth in 41 CFR 60-4.3(a), and its efforts to meet the goals established for the geographical area where the contract resulting from this solicitation is to be performed. The hours of minority and female employment and training must be substantially uniform throughout the length of the contract and in each trade, and the Contractor shall make a good faith effort to employ minorities and women on each of its projects. The transfer of minority or female employees or trainees from Contractor to Contractor or from project to project for the sole purpose of meeting the Contractor's goals shall be a violation of the contract, the Executive Order and the regulations in 41 CFR Part 60-4. Compliance with the goals will be measured against the total work hours performed.

- c.) The Contractor shall provide written notification to the Director of the Office of Federal Contract Compliance Programs within ten (10) working days of award of any construction subcontract in excess of \$10,000 at any tier for construction work under the contract resulting from this solicitation. The notification shall list the name, address and telephone number of the subcontractor; employer identification number; estimated dollar amount of the subcontract; estimated starting and completion dates of the subcontract; and the geographical area in which the contract is to be performed.
- d.) As used in this notice, and in the contract resulting from this solicitation, the "covered area" is the State of New Jersey.
- 14.11.3 STANDARD FEDERAL EQUAL EMPLOYMENT OPPORTUNITY CONTRACT SPECIFICATIONS (EXECUTIVE ORDER 11246)
  - As used in these specifications:
    - (a) "Covered area" means the geographical area described in the solicitation from which this contract resulted;
    - (b) "Director" means Director, Office of Federal Contract Compliance Programs, United States Department of Labor, or any person to whom the Director delegates authority;
    - (c) "Employer identification number" means the Federal Social Security number used on the Employer's Quarterly Federal Tax Return, U.S. Treasury Department Form 941.

- (d) "Minority" includes:
  - (i) Black (all persons having origins in any of the Black African racial groups not of Hispanic origin);
  - (ii) Hispanic (all persons of Mexican, Puerto Rican, Cuban, Central or South American or other Spanish culture or origin, regardless of race);
  - (iii) Asian and Pacific Islander (all persons having origins in any of the original peoples of the Far East, Southeast Asia, the Indian Subcontinent, or the Pacific Islands); and
  - (iv) American Indian or Alaskan Native (all persons having origins in any of the original peoples of North America and maintaining identifiable tribal affiliations through membership and participation or community identification).
- Whenever the Contractor, or any subcontractor at any tier, subcontracts a portion of the work involving any construction trade, it shall physically include in each subcontract in excess of \$10,000 the provisions of these specifications and the Notice which contains the applicable goals for minority and female participation and which is set forth in the solicitations from which this contract resulted.
- If the Contractor is participating (pursuant to 41 CFR 60-4.5) in a Hometown Plan approved by the U.S. Department of Labor in the covered area either individually or through an association, its affirmative action obligations on all work in the Plan area (including goals and timetables) shall be in accordance with that Plan for those trades which have unions participating in the Plan. Contractors must be able to demonstrate their participation in and compliance with the provisions of any such Hometown Plan. Each Contractor or subcontractor participating in an approved Plan is individually required to comply with its obligations under the EEO clause, and to make a good faith effort to achieve each goal under the Plan in each trade in which it has employees. The overall good faith performance by other Contractors or subcontractors toward a goal in an approved Plan does not excuse any covered Contractor's or subcontractor's failure to make good faith efforts to achieve the Plan goals and timetables.
- 4. The Contractor shall implement the specific affirmative action standards provided in paragraphs 7(a) through (p) of this Article. The goals set forth in the solicitation from which this contract resulted are expressed as percentages of the total hours of employment and training of minority and female utilization the Contractor should reasonably be able to achieve in each construction trade in which it has employees in the covered area. The Contractor is expected to make substantially uniform progress toward its goals in each craft during the period specified.

- 5. Neither the provisions of any collective bargaining agreement, nor the failure by a union with whom the Contractor has a collective bargaining agreement, to refer either minorities or women shall excuse the Contractor's obligations under these specifications, Executive Order 11246, or the regulations promulgated pursuant thereto.
- 6. In order for the nonworking training hours of apprentices and trainees to be counted in meeting the goals, such apprentices and trainees must be employed by the Contractor during the training period, and the Contractor must have made a commitment to employ the apprentices and trainees at the completion of their training, subject to the availability of employment opportunities. Trainees must be trained pursuant to training programs approved by the U.S. Department of Labor.
- 7. The Contractor shall take specific affirmative actions to ensure equal employment opportunity.

  The evaluation of the Contractor's compliance with these specifications shall be based upon its

  effort to achieve maximum results from its actions. The Contractor shall document these efforts
  fully, and shall implement affirmative action steps at least as extensive as the following:
  - (a) Ensure and maintain a working environment free of harassment, intimidation, and coercion at all sites, and in all facilities at which the Contractor's employees are assigned to work. The Contractor, where possible, will assign two or more women to each construction project. The Contractor shall specifically ensure that all foremen, superintendents, and other on-site supervisory personnel are aware of and carry out the Contractor's obligation to maintain such a working environment, with specific attention to minority or female individuals working at such sites or in such facilities.
  - (b) Establish and maintain a current list of minority and female recruitment sources, provide written notification to minority and female recruitment sources and to community organizations when the Contractor or its unions have employment opportunities available, and maintain a record of the organizations' responses.
  - (c) Maintain a current file of the names, addresses and telephone numbers of each minority and female off-the-street applicant and minority or female referral from a union, a recruitment source or community organization and of what action was taken with respect to each such individual. If such individual was sent to the union hiring hall for referral and was not referred back to the Contractor by the union, or, if referred, not employed by the Contractor, this shall be documented in the file with the reason therefor, along with whatever additional actions the Contractor may have taken.

- (d) Provide immediate written notification to the Director and the Contract Compliance Officer of NJ TRANSIT when the union or unions with which the Contractor has a collective bargaining agreement has not referred to the Contractor a minority person or a woman sent by the Contractor, or when the Contractor has other information that the union referral process has impeded the Contractor's efforts to meet its obligations.
- (e) Develop on-the-job training opportunities and/or participate in training programs for the area which expressly include minorities and women, including upgrading programs and apprenticeship and trainee programs relevant to the Contractor's employment needs, especially those programs funded or approved by the Department of Labor. The Contractor shall provide notice of these programs to the sources compiled under 7(b) above.
- (f) Disseminate the Contractor's EEO policy by providing notice of the policy to unions and training programs and requesting their cooperation in assisting the Contractor in meeting his EEO obligations; by including it in any policy manual and collective bargaining agreement; by publicizing it in the company newspaper, annual report, etc.; by specific review of the policy with all management personnel and with all minority and female employees at least once a year; and by posting the company EEO policy on bulletin boards accessible to all employees at each location where construction work is performed.
- (g) Review, at least annually, the company's EEO policy and affirmative action obligations under these specifications with all employees having any responsibility for hiring, assignment, layoff, termination or other employment decisions including specific review of these items with on-site supervisory personnel such as Superintendents, General Foremen, etc., prior to the initiation of construction work at any job site. A written record shall be made and maintained identifying the time and place of these meetings, persons attending, subject matter discussed, and disposition of the subject matter.
- (h) Disseminate the Contractor's EEO policy externally by including it in any advertising in the news media, specifically including minority and female news media, and providing written notification to and discussing the Contractor's EEO policy with other contractors and subcontractors with whom the Contractor does or anticipates doing business.

- (i) Direct its recruitment efforts, both oral and written, to minority, female and community organizations, to schools with minority and female students and to minority and female recruitment and training organizations serving the Contractor's recruitment area and employment needs. Not later than one month prior to the date for the acceptance of applications for apprenticeship or other training by any recruitment source, the Contractor shall send written notification to organizations such as the above, describing the openings, screening procedures, and tests to be used in the selection process.
- (j) Encourage present minority and female employees to recruit other minority persons and women and, where reasonable, provide after school, summer and vacation employment to minority and female youth both on the site and in other areas of a Contractor's work force.
- (k) Validate all tests and other selection requirements where there is an obligation to do so under 41 CFR Part 60-3.
- (I) Conduct, at least annually, an inventory and evaluation at least of all minority and female personnel for promotional opportunities and encourage these employees to seek or to prepare for, through appropriate training, etc., such opportunities.
- (m) Ensure that seniority practices, job classifications, work assignments and other personnel practices, do not have a discriminatory effect by continually monitoring all personnel and employment related activities to ensure that the EEO policy and the Contractor's obligations under these specifications are being carried out.
- (n) Document and maintain a record of all solicitations of offers for subcontracts from minority and female construction contractors and suppliers, including circulation of solicitations to minority and female contractors' associations and other business associations.
- (o) Conduct a review, at least annually, of all supervisors' adherence to and performance under the Contractor's EEO policies and affirmative action obligations.

- 8. Contractors are encouraged to participate in voluntary associations which assist in fulfilling one or more of their affirmative action obligations (7 (a) through (p)). The efforts of a contractor association, joint contractor-union, contractor-community, or other similar group of which the Contractor is a member and participant, may be asserted as fulfilling any one or more of its obligations under 7(a) through (p) of this Article, provided that the Contractor actively participates in the group, makes every effort to assure that the group has a positive impact on the employment of minorities and women in the industry, ensures that the concrete benefits of the program are reflected in the Contractor's minority and female work force participation, makes a good faith effort to meet its individual goals and timetables, and can provide access to documentation which demonstrates the effectiveness of actions taken on behalf of the Contractor. The obligation to comply, however, is the Contractor's and failure of such a group to fulfill an obligation shall not be a defense for the Contractor's noncompliance.
- 9. A single goal for minorities and a separate goal for women have been established. The Contractor, however is required to provide equal employment opportunity and to take affirmative action for all minority groups, both male and female, and all women, both minority and non-minority. Consequently, the Contractor may be in violation of the Executive Order if a particular group is employed in a substantially disparate manner (for example, even though the Contractor has achieved its goals for women generally, the Contractor may be in violation of the Executive Order if a specific minority group of women is underutilized).
- 10. The Contractor shall not use the goals and timetables or affirmative action standards to discriminate against any person because of race, color, religion, sex, or national origin.
- 11. The Contractor shall not enter into any subcontract with any person or firm debarred from Government contracts pursuant to Executive Order 11246, as amended.
- 12. The Contractor shall carry out such sanctions and penalties for violation of these specifications and of the Equal Employment Opportunity Clause, including suspension, termination and cancellation of existing subcontracts as may be imposed or ordered pursuant to Executive Order 11246, as amended, and its implementing regulations, by the Office of Federal Contract Compliance Programs. Any Contractor who fails to carry out such sanctions and penalties shall be in violation of these specifications and Executive Order 11246, as amended.

- 13. The Contractor, in fulfilling its obligations under this Article, shall implement specific affirmative action steps, at least as extensive as those standards prescribed in Paragraph 7 of these specifications, so as to achieve maximum results from its efforts to ensure equal employment opportunity. If the Contractor fails to comply with the requirements of the Executive Order, the implementing regulations, or this Article, the Director shall proceed in accordance with 41 CFR 60-4.8. In connection with the foregoing, NJ TRANSIT or its Contract Compliance Officer may utilize any remedies that may be available.
- 14. The Contractor shall designate a responsible official to monitor all employment related activity to ensure that the company EEO policy is being carried out, to submit reports relating to the provisions herein as may be required by the Government and to keep records. Records shall at least include for each employee the name, address, telephone numbers, construction trade, union affiliation if any, employee identification number when assigned, social security number, race, sex, status (e.g., mechanic, apprentice, trainee, helper, or laborer), dates of changes in status, hours worked per week in the indicated trade, rate of pay, and locations at which the work was performed. Records shall be maintained in an easily understandable and retrievable form; however, to the degree that existing records satisfy this requirement, Contractors shall not be required to maintain separate records.
- 15. Nothing herein provided shall be construed as a limitation upon the application of other laws which establish different standards of compliance or upon the application of requirements for the hiring of local or other area residents (e.g., those under the Public Works Employment Act of 1977 and the Community Development Block Grant Program).

#### 14.12 LIMITATIONS ON LOBBYING

The Contractor and its subcontractors shall comply with 31 USC 1352, entitled "Limitation on Use of Appropriated Funds to Influence Certain Federal Contracting and Financial Transactions".

- No appropriated funds may be expended by the recipient of a Federal contract, grant, loan or cooperative agreement to pay any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress or an employee of a Member of Congress in connection with any of the following covered Federal actions: the awarding of any Federal contract, the making of any Federal grant, the making of any Federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment or modification of any Federal contract, grant, loan or cooperative agreement.
- 2.) Any Contractor and any subcontractor at any tier who requests or receives a Federally-assisted contract or subcontract in excess of \$100,000 from NJ TRANSIT shall file with NJ TRANSIT the certification attached to this Contract and entitled "Certification for Contracts, Grants, Loans and

Cooperative Agreements", that the Contractor or subcontractor, as applicable, has not made, and will not make, any payment prohibited by Paragraph 1.) of this Article.

- 3.) Any Contractor and any subcontractor who has made or has agreed to make any payment using nonappropriated funds (to include profits from any covered Federal action) which would be prohibited under Paragraph 1.) of this Article if paid for with appropriated funds, shall file with NJ TRANSIT a disclosure form, which may be obtained from the Contracting Officer, entitled "Disclosure of Lobbying Activities".
- 4.) Any certification or disclosure form filed under Paragraphs 2.) and 3.) of this Article shall be forwarded from tier to tier until received by NJ TRANSIT. Any certification or disclosure form shall be treated as a material representation of fact upon which all receiving tiers shall rely. All liability arising from an erroneous representation shall be borne solely by the tier filing that representation and shall not be shared by any tier to which the erroneous representation is forwarded.
- 5.) The prohibition on the use of appropriated funds does not apply in the case of a payment of reasonable compensation to an officer or employee of a Contractor or subcontractor if the payment is for agency and legislative liaison activities not directly related to a covered Federal action.
- 6.) The prohibition on the use of appropriated funds does not apply in the case of any reasonable payment to an officer or employee of a Contractor or subcontractor or to a person, other than an officer or employee of a Contractor or subcontractor, if the payment is for professional or technical services rendered directly in the preparation, submission or negotiation of any bid, proposal or application for a Federal contract, grant, loan or cooperative agreement.

#### 14.13 SEISMIC SAFETY

The Contractor agrees that any new building or addition to an existing building will be designed and constructed in accordance with the standards for Seismic Safety required in USDOT Seismic Safety Regulations, 49 CFR Part 41, and will certify to compliance to the extent required by the regulation. The Contractor also agrees to ensure that all work performed under this Contract including work performed by a subcontractor is in compliance with the standards required by the Seismic Safety Regulations and the certification of compliance issues on the project.

#### 14.14 EXCLUSIONARY OR DISCRIMINATORY SPECIFICATIONS

Except where otherwise required by Federal statute or regulations, the Contractor agrees that it will comply with the requirements of 49 USC § 5323(h)(2) by refraining from using exclusionary or discriminatory specifications in the performance of the Work.

#### 14.15 NO FEDERAL GOVERNMENT OBLIGATIONS TO THIRD PARTIES

The Contractor agrees that, absent the Federal Government's express written consent, the Federal Government shall not be subject to any obligations or liabilities to the Contractor in connection with the performance of the Project. Notwithstanding any concurrence provided by the Federal Government in or approval of the Contract, the Federal Government continues to have no obligations or liabilities to the Contractor.

#### 14.16 FALSE OR FRAUDULENT STATEMENTS AND CLAIMS

- 14.16.1 The Contractor recognizes that the requirements of the Program Fraud Civil Remedies Act of 1986, as amended, 49 USC §§ 3801 et seq. and USDOT regulations, "Program Fraud Civil Remedies," 49 CFR Part 31, apply to its actions pertaining to the Project. Accordingly, by signing the Contract, the Contractor certifies or affirms the truthfulness and accuracy of any statement it has made, it makes, or it may make pertaining to the Contract. In addition to other penalties that may be applicable, the Contractor also acknowledges that if it makes a false, fictitious,, or fraudulent claim, statement, submission, or certification, the Federal Government reserves the right to impose the penalties of the Program Fraud Civil Remedies Act of 1986, as amended, on the Contractor to the extent the Federal Government deems appropriate.
- 14.16.2 The Contractor also acknowledges that if it makes a false, fictitious, or fraudulent claim, statement, submission, or certification to the Federal Government in connection with an urbanized area formula project financed with Federal assistance authorized by 49 USC § 5307, the Government reserves the right to impose on the Contractor the penalties of 18 USC § 1001 and 49 USC § 5307(n)(1), to the extent the Federal Government deems appropriate.

#### 14.17 FEDERAL AID, LAWS AND REGULATIONS

This Contract is funded in whole or in part by the United States Government and is subject to all Federal laws and regulations governing Federally-financed projects. The Contractor acknowledges that Federal requirements may change and the changed requirements will apply to the project as required, unless the Federal Government determines otherwise.

All Contractual provisions required by DOT, as set forth in FTA circular 4220.1E, dated April 15, 1996, are hereby incorporated by reference. Anything to the contrary herein notwithstanding, all FTA mandated terms shall be deemed to control in the event of a conflict with other provisions contained in this Agreement. The Contractor shall not perform any act, fail to perform any act, or refuse to comply with any NJ Agreement. The Contractor shall not perform any act, fail to perform any act, or refuse to comply with any NJ TRANSIT requests which would cause NJ TRANSIT to be in violation of the FTA Terms and conditions.

### **APPENDIX A**

# SUBCONTRACTOR EVALUATION DATA FORM

(PROJECT)	
(SUBCONTRACTOR)	
(DATE)	

### SUBCONTRACTOR EVALUATION DATA FORM

1.	Proposed Project:	
2.	Proposed work or trade:	Division Nos.:
3.	Approximate Contract Amount:	
4.	Company Name:	
		Partnership Individual Joint Venture
5.	Principal Office Location:	
	Phone:	Person in Charge:
6.	Name and Title of Principals:	
7.	Branch Office Locations:	
		Phone:
		Phone:
8.	•	esponsible for work at this project.
9.	Subcontractor Information Data	Form as required by Federal Regulation, submitted herewith.
10.	Contractor has confirmed that the listed Subcontractor is not listed on the State of New Jersey, Department of Labor and Workforce Development, Division of Wage and Hour Compliance, Prevailing Wage Debarment List or on the Department of Treasury, Consolidated Debarment Report and Contractor has also confirmed that the listed Subcontractor is currently registered and active with no exclusion on the consolidated U.S. Government, System for Award Management (SAM) database.	
11.	Number of years organization h	as been under present name:
12.	List all other names under which	h the company has done business and for how many years.
13.	State Licensed or Certified to de	o business in:
14.		rsuant to P.L.1999, C.238, "THE PUBLIC WORKS CONTRACTOR attached) Yes: No:
15.	Business Registration Certificat	re (P.L. 2004, c.57): (Copy attached) Yes: No:

ivallic.	Dhone:		Δttn:
Address.	Phone: Work:	//alue:	Aun
Name:	Phone:	value	Attn.
Address:	Work:	Value:	7401.
Name:	Phone:		Attn:
Address:	Phone: Work: Phone: Work		Value:
Do you qualify as a DBl	E/SBE/ESBE on this project:		
Surety company that no	ormally handles bonding:		
Company Name	:	No. of `	/ears:
Agent:		Phone:	
Address:			
	t four projects where you perforn		
NIANI	AUUDEGG		
<u>NAME</u> 	ADDRESS	<u>PHONE</u>	<u>CONTACT</u>
<u>NAME</u> 	<u>ADDRESS</u>	<u>PHONE</u>	<u>CONTACT</u>
	ADDRESS  Bradstreet:		
Are you listed in Dunn &		What City? _	
Are you listed in Dunn & List all projects that you	Bradstreet:are participating in litigation agai	What City? _	
Are you listed in Dunn & List all projects that you Annual Billings:	Bradstreet:are participating in litigation agai	What City? _ inst the owner:	
Are you listed in Dunn & List all projects that you Annual Billings:  The undersigned certifies the second control of the second con	Bradstreet:are participating in litigation agai	What City? _ inst the owner:	
Are you listed in Dunn & List all projects that you Annual Billings:	Bradstreet:are participating in litigation agai	What City? _ inst the owner:	
Are you listed in Dunn & List all projects that you  Annual Billings:  The undersigned certifies to requirements of the Prime	Bradstreet:are participating in litigation agai	What City? inst the owner: ditions, specification	ns and other Contract

### **NJ TRANSIT**

### SUBCONTRACTOR INFORMATION

### **DATA FORM**

#### **PURPOSE:**

Federal Regulations require that NJ TRANSIT obtain information on an ongoing basis from all companies seeking to do business with NJ TRANSIT as prime contractors or subcontractors. As part of the subcontractor evaluation process, prime contractors shall provide the required information for all subcontractors performing work under the contract.

Information may be completed on-line by visiting the following website: http://www.tcgionline.com/njtransitbidresponse.asp

Prime contractors will need to create a company profile and password by clicking on the "Add or Update Bidder Profile" link.

Once the company profile and password are created, prime contractors may add bid and subcontractor information by clicking on appropriate links.

Include the printed form with the Subcontractor Evaluation Data Form.\*

Completion and submission of this form will:

- 1. Place the firm in NJ TRANSIT'S database for use in developing bidder's lists;
- 2. Assist NJ TRANSIT in establishing appropriate small and disadvantaged business enterprise goals; and,
- 3. Alert NJ TRANSIT of any changes in the goods and services provided.

#### **Important Notice about DBE/SBE/ESBE Certification:**

Submission of this form does not certify a firm as a Disadvantaged Business with either NJ TRANSIT or the State of New Jersey. Contact NJ TRANSIT'S Office of Business of Diversity (973) 491-8060 for clarification of this form and for information regarding certification. For questions regarding the bid package, contact the contract specialist.

# NEW JERSEY TRANSIT CORPORATION SUBCONTRACTOR INFORMATION DATA FORM

(MANDATORY FORM TO BE FILLED OUT ENTIRELY)

T Contract No:	Project Name:	
me Contractor Name:		
dress:		
Complete th	e information below for AL	L SUBCONTRACTORS
		SUBCONTRACTOR
Company's Full Name		
Address		
City		
Zip		
County		
Phone		
Fax		
E-mail		
Owner		
Date Established		
Date Certified		
Ethnicity		
Gender		
Certification Status: Non S SBE I, SBE II, SBE III, SBE indicate all that apply)		
Federal Tax ID # / SSN #		
Annual Gross Receipts:  *indicate the letter that app	A – Less than \$500K B - \$500K to \$1M C - \$1M to \$2M D - \$2M to \$5M E - \$5M and over	

Primary Industry Operation Code:

### APPENDIX B- CLOSEOUT FORMS

Affidavit of Payment of Debts and Claims

Affidavit of Release of Liens

Consent of Surety to Final Payment

Certificate of amounts due workers for wages on the work pursuant

to N.J.S.A. 34:11-56.25 et seq.

Certificate of Substantial Completion

Certificate of Partial Acceptance

Certificate of Final Acceptance

Proposed Final Certificate of Payment

Final Certificate of Payment

Subcontractor Affidavit of Payment of Debts and Claims

Subcontractor Affidavit of Release of Liens

Consent of Surety to Final Payment to the Subcontractor

Subcontractor's Certificate of amounts due workers for wages on the work pursuant

to N.J.S.A. 34:11-56.25 et seq.

Certificate of Final Acceptance of Subcontractor's Work

### AFFIDAVIT OF PAYMENT OF DEBTS AND CLAIMS

то:	New Jersey Transit Corporation One Penn Plaza East New Jersey 07105-2246	CONTRACT NO.:	
Newark, New Jersey 07105-2246 PROJECT:		CONTRACTOR:	
<i></i>	JJECT:	CONTRACT DATE:	
STA	ATE OF:		
CO	UNTY OF:		
here for a inde with	undersigned, pursuant to Sub-article 12.8.1 aby certifies that, except as listed below he hall materials and equipment furnished, for all ebtedness and claims against the CONTRAC the performance of the Contract referen	las paid in full or has otherwise sati work, labor, and services performed TOR for damages arising in any ma iced above for which the NEW .	isfied all obligations d, and for all known anner in connection
<b>EXCEPTIONS:</b> (If none, write "NONE". The CONTRACTOR shall furnish bond satisfactory to the NEW JERSEY TRANSIT CORPORATION for each exception.)			satisfactory to the
SIG	NED,	Subscribed and sworn to be	fore me this
	NED, NTRACTOR		
COI	VINACION	day of	, 20
		Notary Public of	
Bv.		My commission expires	, 20
Title	<u> </u>	(SEAL)	

### AFFIDAVIT OF RELEASE OF LIENS

TO: New Jersey Transit Corporation One Penn Plaza East	CONTRACT NO.:	
Newark, New Jersey 07105-2246	CONTRACTOR:	
PROJECT:	CONTRACT DATE:	
STATE OF:		
COUNTY OF:		
hereby certifies that to the best of his kn are no liens filed against any property	e 12.8.1 of the Contract's General Provisions towledge, information and belief, except as low of the NEW JERSEY TRANSIT CORPORTION of materials and equipment, or personance of the Contract referenced above.	isted below, there ORATION by the
<b>EXCEPTIONS:</b> (If none, write "NONE" NEW JERSEY TRANSIT CORPORATION	'. The CONTRACTOR shall furnish bond s	satisfactory to the
SIGNED,	Subscribed and sworn to before	re me this
CONTRACTOR	day of	, 20
	Notary Public of	
Ву:	My commission expires	, 20
Title:		
	(SEAL)	

CONSENT OF SURETY TO FINAL PAYMENT

Bond No.	707110,217
TO: New Jersey Transit Corporation One Penn Plaza East	CONTRACT NO.:
Newark, New Jersey 07105-2246	CONTRACTOR:
PROJECT:	CONTRACT DATE:
In accordance with the General Provisions for C the NEW JERSEY TRANSIT CORPORATION a	Construction, Sub-article 12.8.1 of the Contract between and the CONTRACTOR as indicated above, the
	, SURETY COMPANY
on bond of	
	, CONTRACTOR
hereby approves to the final payment to the CONTRACTOR shall not relieve the SURETY C	CONTRACTOR, and agrees that final payment to the COMPANY of any of its obligations to
NEW JERSEY TRANSIT CORPORATION One Penn Plaza East Newark, New Jersey 07105-2246	, OWNER
as set forth in said Surety Company's bond.	
IN WITNESS WHEREOF, the SURETY COM, 20	MPANY has hereunto set its hand this day of
	Surety Company
Attest:	Signature of Authorized Representative
(Seal):	Title

### CERTIFICATE OF AMOUNTS DUE WORKERS FOR WAGES

TO: New Jersey Transit Corporation One Penn Plaza East	CONTRACT NO.:		
Newark, New Jersey 07105-2246 PROJECT:	CONTRACTOR:		
PROJECT:	CONTRACT DATE:		
The undersigned CONTRACTOR, pursuant to Provisions for Construction, hereby certifies earned on the work to all laborers, mechanics general employed by him or by any subcontratis further certified that all laborers, mechanics general have been paid wages at rates not pursuant to N.J.S.A. 34:11-56.25 et seq., and apprentice, trainee, watchman, guard, and win the Contract or training program provisions	that, except as listed below he has paid is, apprentices, trainees, watchmen, guard actor performing work under the Contract is, apprentices, trainees, watchmen, guard less than those required by the Contract that the work performed by each such later in general conformed to the classical applicable to the wage rate paid.	I in full all wages is, and workers in on the Project. It is, and workers in ot provisions and aborer, mechanic, fications set forth	
<b>EXCEPTIONS:</b> (If none, write "NONE"; for each and detailed explanation.)	ach exception, the CONTRACTOR shall	provide complete	
SIGNED,	Subscribed and sworn to befor	e me this	
CONTRACTOR	day of	, 20	
	Notary Public of		
	My commission expires	, 20	
	(SEAL)		
Ву:	<u> </u>		
Title:	<u> </u>		
Date:	<u> </u>		

### CERTIFICATE OF SUBSTANTIAL COMPLETION

TO: New Jersey Transit Corporation One Penn Plaza East	CONTRACT NO.:
Newark, New Jersey 07105-2246 PROJECT:	CONTRACTOR:
	CONTRACT DATE:
, 20 at the above cap	certifies that a field inspection performed on tioned Project revealed that the CONTRACTOR has all authorized additional work for the following items of
All Contract Bid Items	
Bid Item(s) No.	
Other	
	ove enumerated have been satisfactorily completed in that as of the aforementioned date the NEW JERSEY occupancy of the subject work.
The Remaining Work to be completed in order hereby listed in the Punch List Work attached hereby	to achieve one hundred percent (100%) completion is ereto.
SIGNED,	ACCEPTED,
NJ TRANSIT	CONTRACTOR
Ву:	By:
Title:	Title:
Date:	Date:

### CERTIFICATE OF PARTIAL ACCEPTANCE

то:	New Jersey Transit Corporation One Penn Plaza East	CONTRACT NO.:
	Newark, New Jersey 07105-2246	CONTRACTOR:
PRO	DJECT:	CONTRACT DATE:
The	undersigned Project Manager hereby	certifies that a field inspection performed on
has con		ioned Project revealed that the CONTRACTOR of the work described below within its base the following items of work:
	Bid Item(s) No.	
	Other	
con the	npleted in accordance with the Contract Do	k above enumerated have been satisfactorily cuments, and that as of the aforementioned date N may take beneficial occupancy of the subject Sub-Article 13.1.
SIG	NED,	ACCEPTED,
	NJ TRANSIT	CONTRACTOR
Ву:		Ву:
	9:	Title:
Dəf	a·	Date:

### CERTIFICATE OF FINAL ACCEPTANCE

TO: New Jersey Transit Corporation One Penn Plaza East Newark, New Jersey 07105-2246	CONTRACT NO.: CONTRACTOR:	
PROJECT:	CONTRACT DATE:	
Construction, the undersigned CONTRACTOR I 20 the base contract and all authorized addit	d 13.3.1 of the Contract's General Provisions for hereby notifies that as of, ional work has been 100% completed, including but not ork) listed on the attached Certificate(s) of Substantial	
	SIGNED,	
	CONTRACTOR	
	Ву:	
	Title:	
found to be 100% complete in accordance with t	the Project and all base and authorized additional work	
SIGNED,	CONCURRENCE,	
Project Manager	Contracting Officer	
Title	Title	

### PROPOSED FINAL CERTIFICATE OF PAYMENT

TO: New Jersey Transit Corpora One Penn Plaza East		CONTRACT NO.:
Newark, New Jersey 07105-2246		CONTRACTOR:
PROJECT:		CONTRACT DATE:
The undersigned Project Manager,	considering t	that
	,	, CONTRACTOR
all authorized additional work, and Sub-article 12.8.1 of the Contract's Payment in the amount of	fulfilled all his General Pro	d one hundred percent (100%) of the base contract and s contractual obligations including those enumerated in ovisions for Construction, hereby proposes that a Final be issued to said balance of his work under this Contract.
Original Contract Value:	\$	
Total Contract Changes Value Changed Contract Value:	\$	
Less: Payments to Date: Balance Due Contractor: Less: Deductions:	\$	(1)
Total Final Payment Amount:	<u>\$</u>	
(1) Deductions Explanations:		
	•	includes all direct and indirect costs attributable to this eek further compensation for any other costs related to
SIGNED,		ACCEPTED CONTRACTOR,
Title:		Title:
Date:		Date:

### FINAL CERTIFICATE OF PAYMENT (Attach to Final Payment Request)

TO: New Jersey Transit Corporation One Penn Plaza East Newark, New Jersey 07105-2246		CONTRACT NO.:	
		CONTRACTOR:	
PROJECT:		CONTRACT DATE:	
The undersigned Contracting Office	er, considering	g that	
	,	CONTRACTOR	
all authorized additional work, and Sub-article 12.8.1 of the Contract's Payment in the amount of	fulfilled all his General Pro	d one hundred percent (100%) of the base contract and contractual obligations including those enumerated in visions for Construction, hereby proposes that a Final be issued to said balance of his work under this Contract.	
Original Contract Value: Total Contract Changes Value Changed Contract Value: Less: Payments to Date: Balance Due Contractor:	\$ \$ \$		
Less: Deductions:  Total Final Payment Amount: \$		(1)	
(1) Deductions Explanations:			
		includes all direct and indirect costs attributable to this eek further compensation for any other costs related to	
SIGNED,		ACCEPTED CONTRACTOR,	
Title:		Title:	
Date:		Date:	

### SUBCONTRACTOR AFFIDAVIT OF PAYMENT OF DEBTS AND CLAIMS

TO: New Jersey Transit Corporation	CONTRACT NO.:					
One Penn Plaza East Newark, New Jersey 07105-2246	CONTRACTOR:	CONTRACTOR:				
PROJECT:	SUBCONTRACTOR:	SUBCONTRACTOR:				
	CONTRACT DATE:					
STATE OF:						
COUNTY OF:						
The undersigned, pursuant to Sub-article 12. hereby certifies that, except as listed below, I for all materials and equipment furnished for indebtedness and claims against the SUB connection with the performance of the Corresponding TRANSIT CORPORATION or its property mig	ne has paid in full or has otherwise sa all work, labor, and services performe CONTRACTOR for damages arising ontract referenced above for which tht in any way be held responsible.	atisfied all obligations ed, and for all known g in any manner in the NEW JERSEY				
NEW JERSEY TRANSIT CORPORATION for		u sausiaciory to the				
SIGNED,	Subscribed and sworn to be	efore me this				
SUBCONTRACTOR	day of	, 20				
	Notary Public of					
Ву:	My commission expires	, 20				
Title:		(SEAL)				
SIGNED,	Subscribed and sworn to be	efore me this				
CONTRACTOR	day of	, 20				
	Notary Public of					
Ву:	My commission expires	, 20				
Title:		)				

### SUBCONTRACTOR AFFIDAVIT OF RELEASE OF LIENS

TO: New Jersey Transit Corporation One Penn Plaza East	CONTRACT NO.:						
Newark, New Jersey 07105-2246	CONTRACTOR:	CONTRACTOR:					
PROJECT:	SUBCONTRACTOR:						
	CONTRACT DATE:						
STATE OF:							
COUNTY OF:							
The undersigned, pursuant to Sub-article 12.6 hereby certifies that to the best of his knowled are no liens filed against any property CORPORATION by the SUBCONTRACTOR, of work, labor or services arising from the perfe	dge, information and belief, except as or Contract funds of the NEW or or its suppliers of materials and equip	s listed below, there IERSEY TRANSIT ment, or performers					
<b>EXCEPTIONS:</b> (If none, write "NONE". Th NEW JERSEY TRANSIT CORPORATION for		satisfactory to the					
SIGNED,	Subscribed and sworn to be	fore me this					
SUBCONTRACTOR	day of	, 20					
	Notary Public of						
	My commission expires	, 20					
Ву:							
Title:	_						
SIGNED,	Subscribed and sworn to be	Subscribed and sworn to before me this					
CONTRACTOR	day of	, 20					
	Notary Public of						
	My commission expires	, 20					
Ву:	(SEAL)						
Title:	<u></u>						

### CONSENT OF SURETY PAYMENT TO THE SUBCONTRACTOR

Bor	nd No.					
TO:	New Jersey Transit Corporation	CONTRACT NO.:				
One Penn Plaza East Newark, New Jersey 07105-2246		CONTRACTOR:				
PRO	OJECT:	SUBCONTRACTOR:				
		CONTRACT DATE:				
		Construction, Sub-article 12.6.7, of the Contract between and the CONTRACTOR as indicated above, the				
		, SURETY COMPANY				
on b	oond of					
		, CONTRACTOR				
	eby consents to the final payment to the SU BCONTRACTOR shall not relieve the SURE	JBCONTRACTOR, and agrees that final payment to the TY COMPANY of any of its obligations to				
One	N JERSEY TRANSIT CORPORATION Penn Plaza East vark, New Jersey 07105-2246	, OWNER				
	set forth in said Surety Company's bond.					
IN	WITNESS WHEREOF, the SURETY COI	MPANY has hereunto set its hand this day of				
Λ ++ c	. of:	Surety Company				
Alle	est:	Signature of Authorized Representative				
(Se	al):	Signature of Authorized Representative				
		Title				

Page 13 of 15 EBS MFC 9/20/2013

### SUBCONTRACTOR'S CERTIFICATE OF AMOUNTS DUE WORKERS FOR WAGES

TO: New Jersey Transit Corporation One Penn Plaza East Newark, New Jersey 07105-2246	CONTRACT NO.:  CONTRACTOR:			
PROJECT:	SUBCONTRACTOR:  CONTRACT DATE:			
The undersigned SUBCONTRACTOR, pursual General Provisions for Construction, hereby cerwages earned on the work to all laborers, medworkers in general employed by him. It is furth trainees, watchmen, guards, and workers in generalized by the Contract provisions and pursual performed by each such laborer, mechanic, ageneral conformed to the classifications set applicable to the wage rate paid.	tifies that, except as listed below, he chanics, apprentices, trainees, watcher certified that all laborers, mechaneral have been pai wages at rates ant to N.J.S.A. 34:11-56.25 et sequapprentice, trainee, watchman, gua	e has paid in full all chmen, guards, and nanics, apprentices, not less than those , and that the work ard, and worker in		
<b>EXCEPTIONS:</b> (If none, write "NONE"; for ecomplete and detailed explanation.)	ach exception, the SUBCONTRAC	CTOR shall provide		
SIGNED,	Subscribed and sworn to before me this			
SUBCONTRACTOR	day of	, 20		
	Notary Public of			
Ву:	My commission expires	, 20		
Title:	(SEAL)			
Date:				
SIGNED,	Subscribed and sworn to be	fore me this		
CONTRACTOR	day of	, 20		
	Notary Public of			
Ву:	My commission expires	, 20		
Title:	(SEAL)			
Date:				

### CERTIFICATE OF FINAL ACCEPTANCE OF SUBCONTRACTOR'S WORK

TO: New Jersey Transit Corporation One Penn Plaza East Newark, New Jersey 07105-2246	CONTRACT NO.:  CONTRACTOR:				
PROJECT:	SUBCONTRACTOR:				
FROJECT.	CONTRACT DATE:				
	CONTRACT DATE.				
undersigned SUBCONTRACTOR hereby notifie 20 its Subcontract work and all authorized a	Contract's General Provisions for Construction, the sthe NJ TRANSIT Corporation that as of, dditional work has been 100% completed, including but Work) listed on the attached Certificate(s) of Substantial				
SIGNED,	SIGNED,				
SUBCONTRACTOR	CONTRACTOR				
	Ву:				
Ву:	Title:				
Title:	_				
The undersigned Project Manager hereby certification 20 a final field inspection was performed on authorized additional work found to be 100% co	ies that on, the Subcontractor's work and all Subcontract work and mplete in accordance with the Contract Documents.				
This Certificate of Final Acceptance of Subcontr, 20	ractor's work is signed on this day of				
SIGNED,	CONCURRENCE,				
Project Manager	Contracting Officer				
Title	 Title				

# APPENDIX C CONTRACT EXECUTION FORM

### CONTRACT NO.

	Th	is Agreer	ment mad	le this	day of			20	00,	between
the	New	Jersey	Transit	Corporation	hereinafter	referred	to as	NJ	TRANS	SIT, and
			_ hereina	fter referred t	o as the Cont	ractor.				
				V	VITNESSETH	l:				
	Wi	nereas, t	he said	Contractor, fo	or and in col	nsideratioi	n of the	payn	nents h	ereinafter
spe	cified a	and agree	ed to be i	made by NJ 1	RANSIT, her	eby coven	ants an	d agre	es to fu	ırnish and
deli	ver all	materia	ls and to	o do and pe	erform all wo	rk and la	bor req	uired	to com	plete the
				within	cons	secutive c	alendar	days f	rom the	issuance
of t	he offi	cial notic	e to prod	ceed, in strict	and entire o	conformity	with the	e spec	cification	ns for the
proj	iect, wi	hich said	specifica	tions are as t	ollows and ar	e hereby i	made a	part o	f this Co	ontract as
fully	and v	with the s	same effe	ect as if the s	same had bee	en set for	h at len	gth in	the bo	dy of this
Cor	ntract:							_		
		Bida	lers Prop	osal						
		Perf	ormance	/ Payment Bo	ond					
		_		n Affidavit Prevailing Wa	ne Determina	tion				
				ailing Wage D		uon				
		Prev	ailing Wa	age Affidavit						
			-	ntractors Certi	ficate					
				ompliance Provisions fo	r Construction	Contract	s			
				rovisions for C			•			
				or Contracts,	Grants, Loans	s and				
				Agreements Certificate						
		-	Provisio							
			eral Prov							
		Spe	cial Provi	sions						

The Contractor agrees to make payment of all proper charges for labor and materials required in the aforementioned work.

Technical Provisions
Addendum:

payment by NJ TRANSIT shall be considered as a release in full of all claims against the Executive Director and NJ TRANSIT out of, or by reason of, the work done and materials furnished under this Contract. In consideration of the premises, NJ TRANSIT hereby agrees to pay to the Contractor for the said work when completed in accordance with the said specifications, the sum of \$\_\_\_\_\_\_. It is understood that payments shall be the total of the unit prices written in this Contract for the work actually done. In Witness Whereof, the Contracting Officer of NJ TRANSIT has signed this instrument and caused it to be attested, and the Contractor has caused this instrument to be signed by its and attested by its the day and year first written. **NEW JERSEY TRANSIT CORPORATION** ATTEST: Contracting Officer ATTEST: CONTRACTOR Name: Name: Title: Title: The above Contract has been reviewed and approved as to form only. JOHN JAY HOFFMAN ACTING ATTORNEY GENERAL OF NEW

It is also agreed and understood that the acceptance by the Contractor of the final

**JERSEY** 

Deputy Attorney General

#### APPENDIX D

## GOVERNING RULES FOR WORKING WITHIN THE RAILROAD RIGHT OF WAY

# A. GENERAL REQUIREMENTS FOR WORKING WITHIN NJ TRANSIT'S RIGHT OF WAY

Contractors shall cooperate at all times with officials of NJ TRANSIT and use all reasonable care and diligence in their work to avoid accidents, damage or unnecessary delay to, or interference with, passenger trains and other property of NJ TRANSIT. Contractors are to be advised that a pre-construction meeting will be required prior to any work commencing on NJ TRANSIT property (within the Right-of-Way). In advance of any work on the property, a two-hour "NJTRO Contractor Safety Program" is to be administered by the NJTRO Safety Department for the sponsor agency resident Engineer and Contractor's Safety Coordinator and Crane Groundmen. The Contractor's Construction Superintendent should contact Mr. Joseph Codella at (973) 491-7887 to arrange for the scheduling of this program.

Contractors are to be advised that all construction operations within and over the limits of NJ TRANSIT's Right-of-Way shall be accomplished by methods which will in no way cause damage to the tracks, facilities, aerial or underground lines, embankments or drainage systems. It shall be the Contractor's responsibility to provide for protection of the tracks and embankments as shown on approved plans or as field approved in a safe and satisfactory manner, to install and maintain such shoring, sheeting and bracing as may be required, and to remove and dispose of such protective facilities upon completion of the work. Blasting will not be permitted on or along the right-of-way without prior written approval of NJ TRANSIT. All damage to NJ TRANSIT property caused by the Contractor's operations shall be repaired by the Contractor, or at his cost by NJ TRANSIT at the discretion of NJTRO. Work shall not continue until such damage is repaired and the railroad is back in service.

Whenever, in the judgement of NJTRO, work within or above the railroad's right-of-way may affect or involve the safe movement of its trains, the time and method of doing such work shall first be submitted in writing and approved by NJTRO. This approval shall not be considered as releasing the Contractor from responsibility or liability for any damage which NJTRO may suffer, or for which it may be held liable, by the action or omissions of the Contractor or those of his Sub-Contractors, or his or their employees.

Contractors shall provide written notice not less than ten (10) business days in advance of any work to be performed within or above the right-of-way, or other work which may affect railroad safety to: Mr. Eli Charchar, Engineer of Projects, R.O.W., Engineering, NJ TRANSIT Rail Operations, One Penn Plaza East, Newark, NJ 07105.

NJTRO will require protective personnel to be on duty to protect its operations when the Contractor is working within the property right-of-way. Flag protection will be required when the contractor is on, above or below, or immediately adjacent to NJTRO property or having the capability of obstructing an adjacent track. The specific responsibilities of the NJTRO Flagman are to provide enforcement of NJTRO Safety and Operating rules and other items as provided in these General Requirements as discussed in the "Contractor Safety Program", and is <u>not</u> provided for Engineering related matters.

Where such work is in proximity (15 feet or less) or has the potential to come in contact with overhead electrical wires or facilities, before any work proceeds, an on-site safety meeting <u>must</u> be conducted to determine the identity of such wires or facilities and appropriate steps to be taken. If owned by NJTRO, a qualified Class A employee(s) will be assigned who will take the necessary precautions in accordance with the NJTRO-3 Electrical Operating Instructions. All cranes and hoisting equipment used in this application must be properly grounded in accordance with NJTRO Specification MW-252.

When Crane Operators' visibility is impaired during any hoisting operation; Spotters or qualified Groundmen shall be utilized to guide the Operator. Universal hand signals shall be utilized and their meaning clearly understood between Operator and Spotter. When visual contact between the Operator and Spotter is impaired, two-way radio contact must be utilized.

If a temporary crossing of railroad tracks is necessary, Contractors shall apply to NJ TRANSIT in writing for such a crossing, and, if required, NJTRO shall execute its regular form of private grade crossing agreement covering the crossing desired and providing for the Contractor to pay all construction, maintenance, removal, protection, and other costs and affording contractual liability insurance in the amounts required.

The minimum hours per day for employees engaged in flagging or protection purposes will be eight (8) hours, plus appropriate travel time. For all time over eight (8) hours, the overtime rate will be charged. Personnel used in flagging service will be paid deadhead (traveling) time to and from headquarters each day, plus transportation from headquarters to the site of the work if required [in accordance with the current collective bargaining agreement]. It will be the responsibility of the Contractor to provide transportation for the Flagmen from and to the nearest NJTRO train station, as necessary.

NJTRO will assign Inspectors an/or Engineers during the time the Contractor is engaged in construction work on railroad property to provide general coordination of construction operations, to insure adherence to plans and specifications, and to insure the use of approved construction methods. It is to be understood that the providing of Inspectors, Engineers, Operators, Conductors, Flagmen or other forces, and the taking of any other precautions deemed necessary by NJTRO shall not relieve the Contractor

or his Sub-Contractor from liability for payment of damages caused by their respective operations.

All of NJTRO's costs shall be at the prevailing rates of pay in accordance with railroad accounts, and shall include overtime burden, [if overtime pay is warranted], and Workmen's Compensation Insurance, Public Liability Insurance, Property Damage Insurance, Railroad Unemployment Insurance, Railroad Retirement, Excise Tax, Vacation allowance, and other standard and legal costs, including overhead for supervision and accounting. In general, a recommendation is made that final payment to Contractors, not be made until NJTRO has been reimbursed in full for all of the costs.

Typically, use of NJ TRANSIT property will be restricted as follows, unless specifically authorized by the on-site NJTRO qualified employee (Flagman):

- (a) All workers must maintain a distance of no less than eighteen (18) feet from the track.
- (b) Any tools and equipment being utilized must not extend closer than eighteen (18) feet from the track.
- (c) When a train is approaching, all workers must cease work, stand clear of the track, and face the approaching train.
- (d) No worker is permitted to cross the railroad tracks at any area other than designated grade crossings.
- (e) No tools or working materials are permitted to be left along the NJTRO right-of-way.
- (f) In no event shall equipment or material be transported across a track or tracks without special permission and appropriate flagging protection.
- (g) All personnel, equipment and materials to be used during the construction shall be kept at all times at least fifteen (15) feet from all electrical, signal and communication systems unless protected by an Electrical, Signal or Communication's Department representative. The Contractor is responsible for damage to NJTRO property and any utilities located thereon, whether above or below ground.
- (h) All personnel, equipment and materials to be used during the construction in electrified territory shall also be kept at all times at least fifteen (15) feet from overhead trolley, messenger, static and transmission lines unless clearance and protection is provided by a qualified Electric Traction Department Class A High Tension Lineman.

- (i) All lifting operations shall be reviewed in meeting a standard requirement for a positive block to be installed on the hoisting equipment. This positive block is required to avoid contact with facilities or interfere with safe train operations.
- (j) When construction activity involves any type of hoisting procedure adjacent to aerial lines, the Contractor shall furnish NJTRO with sufficient florescent orange rubber goods, as determined by NJTRO, to be installed as an aid for equipment Operators and Groundmen in visually locating the aerial lines, and as additional protection against damage.

### ENGINEERING DATA REQUIRED FOR ERECTION, DEMOLITION OR OTHER HOISTING OPERATIONS PROPOSED

- A detailed plan must be submitted and approved by NJTRO, showing locations of cranes or hoisting devices (both horizontally and vertically), operating radii, positive stops and delivery or disposal locations. The location of all tracks and other railroad facilities should also be clearly shown on the submission (with distance and dimension information).
- 2. Crane rating sheets are to be provided showing cranes to be adequate for 150% of the lift. Crane and boom nomenclature is to be indicated.
- 3. Plans and computations are to be included with the submission showing weight of pick, safety factors and what work is being performed over NJTRO. Plans must also show a positive stop attachment and general plans of the existing and proposed facilities, showing complete and sufficient details with supporting data for the demolition or erection of the structure.
- 4. A location plan showing all physical limitations, restrictions or obstructions such as wires, poles, adjacent structures, etc., showing that the proposed swings are possible. In the event of any hoisting operation proposed to be adjacent to aerial lines, the Contractor shall furnish NJTRO with sufficient florescent orange rubber goods, as determined by NJTRO, to be installed as an aid in visually locating the aerial lines and as additional protection.
- 5. A data sheet shall be prepared listing the type, size and arrangements of slings, shackles, or other connecting equipment. Include copies of a catalog or information sheets of specialized equipment being used.
- 6. A complete lifting procedure is to be included, indicating the order of lifts and any repositioning of crane or cranes including reattachment of positive stop.
- 7. Temporary support of any components or intermediate stages is to be shown.

- 8. A time schedule of various stages must be provided as well as a schedule for the entire lifting procedure.
- 9. All bridge erection or demolition procedures submitted will be signed and sealed by a registered professional Engineer licensed in the State of New Jersey.
- 10. At least six (6) copies of the plan should be sent to Mr. Eli Charchar, NJ TRANSIT, R.O.W. Engineering, One Penn Plaza East, Newark, NJ 07105.
- 11. The Contractor is to be advised that they can expect a minimum thirty (30) day review period for this and any other shop drawing submissions.

# MINIMUM STANDARDS FOR GEO-TECHNICAL INVESTIGATIONS ALONG THE RIGHT-OF-WAY

Subsurface investigations made on or adjacent to the right-of-way should meet the minimum recommended practices as provided in Chapter 1, Volume 1, of the current American Railway Engineering association Manual for Railway Engineering. Additionally, the following requirements must be met:

- Borings shall be advanced using casing or mud rotary techniques. Use of hollow stem augers below the water table is prohibited.
- All borings shall be tremie grouted with non-shrink cement grout from the bottom to the top of the bore hole at completion. Subsequent minor surface settlement shall be back-filled with tamped earth, asphalt or finished concrete, as appropriate.
- No observation or monitoring wells shall be installed on railroad property without the prior express authorization of the Environmental Services Unit.
- No sampling of any kind shall be done on railroad property without the prior express authorization of the Environmental Services Unit.
- No work shall be done that interferes with operation and/or maintenance of the railroad unless specifically approved in the Temporary Access Permit.
- The crossing of tracks or use of tracks for or by personnel, equipment of material shall only be done under the protection of a qualified NJTRO representative.
- The presence of buried railroad or foreign utilities may or may not be known and any damage resulting from the investigation will be repaired as required,

- and all charges resulting from such damage shall be paid promptly by the Permittee in accordance with the terms of the Temporary Access Permit.
- In advance of the authorized investigation, the locations of proposed pits, boring location, or monitoring well location shall be marked out in the field to review the site for possible location of buried utilities or conflict with operating systems.

A reproducible location plan of proposed boring layout, test pits' locations, or monitoring well locations will be submitted with the technical specification for the work for review and approvals prior to start of work. Detail of the plan shall be sufficient to permit review and comment by the Engineering and Environmental Services Departments. These plans and specifications shall be accompanied by a brief narrative of how the work will be carried out.

The location plan should provide from a licensed Land Surveyor the proposed State plane coordinates and approximate ground surface elevations of the work, and reference centerline alignment and profile of near tracks, support poles and guy anchors, existing foundations and overhead or undergrade wire, conduit, pipelines or structures. NJ TRANSIT uses State coordinate systems for horizontal control as appropriate and vertical datum based on Mean Sea Level equal to 0.0 feet in plan and nearest 0.1 feet in elevation.

Property information should be coordinated with the applicable Right-of-Way and Track Maps or Valuation Sheets. Copies of these maps can be obtained from the NJTRO Real Estate Department. These maps should be used to locate the work with respect to railroad stationing, structure number and mile post.

A draft summary Engineering Report shall be prepared signed and sealed by the licensed professional Engineer in charge of the work. Upon review and comment by NJTRO, three copies of the final report shall be submitted for record.

### ADDITIONAL REQUIREMENTS FOR PIPELINE OCCUPANCIES

The Contractor shall be responsible for compliance with all provisions of NJTRO Specification EP-2 and shall comply with all reasonable requests from NJTRO.

The Contractor shall be responsible to furnish all labor, materials and equipment necessary to install the casing and carrier pipes as referred to in the executed Occupancy Agreement and as shown on the approved contract documents. The Contractor shall be responsible for notification to the appropriate utility companies for surface markout and NJTRO shall be responsible for markout of its own facilities potentially affected by the installation.

If the jacking pit/boring equipment is constructed such that verbal communications are limited, universal hand signals shall be utilized and their meaning clearly understood between all employees. When visual contact between key operators and support Groundmen cannot be adequately maintained, two-way radio contact must be utilized.

The Contractor must provide material certifications for all material to be installed and must prepare and submit for review (allowing 30 days) detailed drawings and supporting calculations (all signed and sealed by a licensed professional Engineer) showing the proposed methods of crossing; including jacking pit details, shoring, bracing, dewatering methods, pushing backstops, receiving pits, grade and alignment controls, catalog cuts on jacking equipment, and narrative methods for installing casing and carrier pipe. The Contractors must be prepared to work continuously and complete the jacking operation below the tracks once the live load influence line has been entered by the auger/casing.

The use of water or other liquids to facilitate conventional casing emplacement and soil removal is prohibited. If during installation, an obstruction is encountered which prevents installation of the pipe in accordance with the approved plans,, the pipe shall be abandoned in place and immediately filled with grout. A revised installation plan must be submitted for approval.

When water is known or expected to be encountered, a designed dewatering system with pumps of sufficient capacity shall be utilized to handle the flow in such a fashion which does not allow groundwater to affect the installation. When dewatering, close observation shall be maintained to detect any settlement or displacement of the embankments, tracks and facilities.

As part of the jacking operation, the Contractor shall be responsible for the completion of Survey Control Monitoring to verify track movement prior to, during, and at a point after completion. The survey monitoring procedure and location layout can be site-specific modified, but, must generally follow those as shown on the attached. The survey monitoring procedure must be completed and signed by a Licensed Land Surveyor.

### NJ TRANSIT SURVEY MONITORING PROCEDURES FOR PIPELINE OCCUPANCIES

<u>ACTIVITY</u>	LOCATION OF SETTLEMENT POINTS	FREQUENCY
Prior to installation and disturbance of the property.	The top of each rail at the centerline of crossing, 5', 10' and 25' on each side of the crossing or as directed by the Manager of Right-of-Way Engineering or his designee. When temporary track supports have been installed, only the running rails shall be monitored.	Take three (3) sets of reading with at least one (1) train passing the area between readings for a base level measurement. All readings to be measured to the nearest 0.001 ft.
Installation of the casing, grouting operation, and, or during dewatering operation.	All points noted above and as required by NJTRO on the Contractors approved dewatering plans.	Immediately prior to start and trains continuously during jacking, and week all dewatering activities for the first activities for the first seven (7) days and then twice weekly.
At completion of jacking, as, necessary,	All points noted above.	After one (1) train, after five (5) after one (1) day, after one (1) after one (1) month or as directed by the Manager, Right-of-Way Engineering or his designee

Should the total changes in rail elevations for any pair of adjacent points exceed the established base elevations by 0.02 ft., the Surveyor shall immediately notify the Manager, Right-of-Way, Engineering or his designee. All readings shall be transmitted directly to the NJ TRANSIT Resident Engineer within twenty-four (24) hours of taking the measurement. Elevations shall be referenced to a U.S.G.S. benchmark and survey runs shall be reported with the proper closure errors. Readings and elevations shall be certified by a licensed Land Surveyor. Copies of the field data shall be maintained on- site by the permittee's Engineer. Review of the survey data by NJ TRANSIT will be made at the end of the first month following the jacking operation to determine the need for additional elevation measurements.

# TRACK SURVEY MONITORING POINTS (TYPICAL FOR ALL TRACKS)

	25'	10'	5'	CENTERLINE OF CROSSING		5'	10'	25'
	Α	С	E		G	I	K	M
RAIL	<del> </del>							
RAIL	·							
	В	D	F		Н	J	L	N

NOTE: All field markouts shall be accomplished using a paint or keel marker, chisel cuts are prohibited on the rail.

# B. <u>SPECIFICATIONS REGARDING SAFETY AND PROTECTION OF RAILROAD TRAFFIC AND PROPERTY WHEN WORKING WITHIN THE NATIONAL RAILROAD PASSENGER CORPORATION (RAILROAD) RIGHT OF WAY</u>

In the following Specifications "Chief Engineer" shall, mean Railroad's Vice President, Chief Engineer, "Railroad" shall mean the National Railroad Passenger Corporation, 'and/or his/her duly authorized representative.

- (1) <u>Pre-Entry Meeting</u>: Before entry of Permittee and/or Contractors onto Railroad's property, a pre-entry meeting shall be held at which time Permittee and/or Contractors shall submit for written approval of the Chief Engineer, plans, computations and a detailed description of proposed methods for accomplishing the work, including methods for protecting Railroad's traffic. Any such written approval shall not relieve Permittee and/or Contractor of their complete responsibility for the adequacy and safety of their operations.
- (2) Rules, Regulations and Requirements: Railroad traffic shall be maintained at all times with safety and continuity, and Permittee and/or Contractors shall conduct their operations in compliance with all rules, regulations, and requirements of Railroad (including these Specifications) with respect to any work performed on, over, under, within or adjacent to Railroad's property. Permittee and/or Contractors shall be responsible for acquainting themselves with such rules, regulations and requirements. Any violation of Railroad's safety rules, regulations, or requirements shall be grounds for the immediate suspension of the Permittee and/or Contractor work, and the retraining of all personnel, at the Permittee's expense.
- (3) <u>Maintenance of Safe Conditions</u>: If tracks or other property of Railroad are endangered during the work, Permittee and/or Contractor shall immediately take such steps as may be directed by Railroad to restore safe conditions, and upon failure of Permittee and/or Contractor to immediately carry out such direction, Railroad may take whatever steps are reasonably necessary to restore safe conditions. All costs and expenses of restoring safe conditions, and of repairing any damage to Railroad's trains, tracks, right-of-way or other property caused by the operations of Permittee and/or Contractors, shall be paid by Permittee.
- (4) <u>Protection in General</u>: Permittee and/or Contractors shall consult with the Chief Engineer to determine the type and extent of protection required to insure safety and continuity of railroad traffic. Any Inspectors, Track Foremen, Track Watchmen, Flagmen, Signalmen, Electric Traction Linemen, or other employees deemed necessary by Railroad, at its sole discretion, for protective services shall be obtained from Railroad by Permittee and/or Contractors. The cost of same shall be paid directly to Railroad by Permittee. The provision of such employees by Railroad, and any other precautionary measures taken by Railroad, shall not relieve Permittee and/or Contractors from their complete responsibility for the adequacy and safety of their operations.

- (5) Protection for Work Near Electrified Track or Wire: Whenever work is performed in the vicinity of electrified tracks and/or high voltage wires, particular care must be exercised, and Railroad's requirements regarding clearance to be maintained between equipment and tracks and/or energized wires, and otherwise regarding work in the vicinity of electrified tracks, must be strictly observed. No employees or equipment will be permitted to work near overhead wires, except when protected by a Class A employee of Railroad. Permittee and/or Contractors must supply an adequate length of grounding cable (4/0 copper with approved clamps) for each piece of equipment working near or adjacent to any overhead wire.
- (6) Fouling of Track or Wire: No work will be permitted within twenty-five (25) feet of the centerline of track or the energized wire or have potential of getting within twenty-five (25) feet of track wire without the approval of the Chief Engineer's representative. Permittee and/or Contractors shall conduct their work so that no part of any equipment or material shall foul an active track or overhead wire without the written permission of the Chief Engineer's representative. When Permittee and/or Contractors desire to foul an active track, they must provide the Chief Engineer's representative with their site-specific work plan a minimum of twenty-one (21) working days in advance, so that, if approved, arrangements may be made for proper protection of Railroad. Any equipment shall be considered to be fouling a track or overhead wire when located (a) within fifteen (15) feet from the centerline of the track or within fifteen (15) feet from the wire, or (b) in such a position that failure of same, with or without a load, would bring it within fifteen (15) feet from the centerline of the track or within fifteen (15) feet from the wire and requires the presence of the proper Railroad protection personnel.

If acceptable to the Chief Engineer's representative, a safety barrier (approved temporary fence or barricade) may be installed at fifteen (15) feet from centerline of track or overhead wire to afford the Permittee and/or Contractor with a work area that is not considered fouling. Nevertheless, protection personnel may be required at the discretion of the Chief Engineer's representative.

- (7) <u>Track Outages</u>: Permittee and/or Contractors shall verify the time and schedule of track outages from Railroad before scheduling any of their work on, over, under, within, or adjacent to Railroad's right-of-way. Railroad does not guarantee the availability of any track outage at any particular time. Permittee and/or Contractors shall schedule all work to be performed in such a manner as not to interfere with Railroad operations. Permittee and/or Contractors shall use all necessary care and precaution to avoid accidents, delay or interference with Railroad's trains or other property.
- (8) <u>Demolition:</u> During any demolition, the Contractor must provide horizontal and vertical shields, designed by a Professional Engineer registered in the state in which the work takes place. These shields shall be designed in accordance with the Railroad's specifications and approved by the Railroad, so as to prevent any debris from falling onto the Railroad's right-of-way or other property. A grounded temporary vertical protective barrier must be provided if an existing vertical protective barrier is 'removed during demolition. In addition, if any openings are left in an existing bride deck, a protective fence must be erected at both ends of the bridge to prohibit unauthorized persons from entering onto the bridge.

- (9) <u>Equipment Condition</u>: All equipment to be used in the vicinity of operating tracks shall be in "certified" first-class condition so as to prevent failures that might cause delay to trains or damage to Railroad's property. No equipment shall be placed or put into operation near or adjacent to operating tracks without first obtaining permission from the Chief Engineer's representative. Under no circumstances shall any equipment or materials be placed or stored within twenty-five (25) feet from the centerline of an outside track, except as approved by the Site Specific Safety Work Plan. To insure compliance with this requirement, Permittee and/or Contractors must establish a twenty-five (25) foot foul line prior to the start of work by either driving stakes, taping off or erecting a temporary fence, or providing an alternate method as approved by the Chief Engineer's representative. Permittee and/or Contractors will be issued warning stickers which must be placed in the operating cabs of all equipment as a constant reminder of the twenty-five (25) foot clearance envelope.
- (10) **Storage of Materials and Equipment**: No material or equipment shall be stored on Railroad's property without first having obtained permission from the Chief Engineer. Any such storage will be on the condition that Railroad will not be liable for loss of or damage to such materials or equipment from any cause.
- (11) <u>Condition of Railroad's Property:</u> Permittee and/or Contractors shall keep Railroad's property clear of all refuse and debris from its operations. Upon completion of the work, Permittee and/or Contractors shall remove from Railroad's property all machinery, equipment, surplus materials, falsework, rubbish, temporary structures, and other property of the Permittee and/or Contractors and shall leave Railroad's property in a condition satisfactory to the Chief Engineer.
- (12) <u>Safety Training</u>: All individuals, including representatives and employees of the Permittee and/or Contractors, before entering onto Railroad's property or coming within twenty-five (25) feet of the centerline of the track or energized wire shall first attend Railroad's Safety Contractor/Leasee Employee Training Class. The Safety Orientation Class will be provided by Railroad's Safety Representative at Permittee's expense. A photo I.D. will be issued and must be worn/displayed while on Railroad property. All costs of complying with Railroad's safety training shall be at the sole expense of Permittee. Permittee and/or Contractors shall appoint a qualified person as their Safety Representative. He/she shall continuously assure that all individuals comply with Railroad's safety requirements. All safety training records shall be maintained with site specific work plan.
- (13) <u>No Charges to Railroad</u>: It is expressly understood that neither these Specifications, nor any document to which they are attached, include any work for which Railroad is to be billed by Permittee and/or Contractors, unless Railroad gives a written request that such work be performed at Railroad's expense.

# C. <u>SPECIFIC REQUIREMENTS OF CONSOLIDATED RAIL CORPORATION</u> FOR WORK ON ITS RIGHT OF WAY

#### **SCOPE**

It must be clearly understood that Conrail owns and uses its right of way for the primary purpose of operating a railroad. All work shall therefore be done in a manner such that the rail operations and facilities are not interfered with, interrupted or endangered. In addition, any facilities that are a result of the proposed work shall be located to minimize encumbrance to the right of way so that Conrail will have unrestricted use of its property for current and future operations.

The sponsor of the project shall be ultimately responsible for assuring that its agents, consultants, contractors and sub-contractors fully comply with the specifications contained herein. The term 'sponsor' used throughout these specifications shall mean the sponsor, its employees, its agents, consultants, contractors, sub-contractors, etc. The following terms and conditions shall apply to any project which requires performance of work on the right of way or other property of Conrail.

#### RIGHT OF ENTRY ON CONRAIL PROPERTY

No entry upon Conrail property shall be permitted without the proper authorization by Conrail to the sponsor in the form of an agreement or a proper permit-to-enter prepared by Conrail. The applicant shall pay the associated fees and execute the permit-to-enter prior to entering Conrail property. The location and design of that portion of the access route to the project site that is on Conrail property shall be shown clearly on any plans for the project and approved by Conrail.

It is to be clearly understood that the issuance of a permit-to-enter does not constitute authority to proceed with any construction work. Construction cannot begin until a formal agreement between Conrail and the sponsor is executed, and the sponsor receives permission from Conrail's representative to proceed with the work.

#### **INSURANCE**

In addition to any other forms of insurance or bonds required under the terms of any contract or specifications and except to the extent that any of the requirements of this section are expressly waived or revised in writing by Conrail, prior to the commencement of any work, contractor, at his own cost and expense, shall maintain insurance of the following kinds and amounts and deliver to Conrail satisfactory evidence of such insurance as indicated herein:

# 1. Public Liability Insurance

Public Liability Insurance, including contractual liability insurance of not less than \$5,000,000 combined single limit for bodily injury and/or property damage for damages arising out of bodily injuries to or death of all persons in any one occurrence and for damage to or destruction of

property, including the loss of use thereof, in any one occurrence. Conrail shall be named as an additional insured under this insurance.

# 2. **Automobile Liability Insurance**

Automobile Liability Insurance with a limit of not less than \$5,000,000 combined single limit for bodily injury and/or property damage per occurrence. Conrail shall be named as an additional insured under this insurance.

#### 3. Workers' Compensation / Employers' Liability Insurance

Employers' Liability and Occupational Disease Insurance with limits of \$1,000,000 each accident, \$1,000,000 policy limit and \$1,000,000 each employee. Such policy shall include a waiver of subrogation in favor of Conrail.

# 4. **General Contractor's Pollution Legal Liability Insurance**

General Contractor's Pollution Liability Insurance with limits of not less than \$5,000,000 per occurrence/ \$5,000,000 aggregate bodily injury, property damage and cleanup expenses resulting from pollution conditions. Conrail shall be named as an additional insured under this insurance.

#### 5. Railroad Protective Liability Insurance

Should it be required, NJ TRANSIT will provide Railroad Protective Liability Insurance Coverage for this Contract.

# **CHANGES IN RAILROAD FACILITIES**

Temporary and permanent changes of signal, communication, power transmission lines, trailers, drainage and other railroad facilities required in connection with the project to clear temporary and/or permanent work of the sponsor as shown on the approved construction plans, shall be made or caused to be made by Conrail at the sole cost and expense of the sponsor in accordance with Conrail's force account estimate. Any other changes made or services furnished by Conrail at the request of the sponsor shall be the sole cost and expense of the sponsor.

# PROTECTION OF RAILROAD OPERATIONS

The sponsor shall conduct the work in such a manner as to safeguard the operations, facilities, right-of-way and property of Conrail. All work affecting the above items shall be subject to the approval of Conrail. The sponsor's operations adjacent to, over or under Conrail's tracks, facilities, right-of-way, and property shall be governed by Conrail's standards and by such other requirements as specified by Conrail's representative so as to insure the safe operation of

trains, prevent delay to trains and insure the safety of all concerned, including the sponsor's forces.

An operating track shall be considered obstructed or fouled when any object is brought closer than fifteen (15) feet (4.6 m) horizontally from the centerline of track and projects above the top of tie or as determined by Conrail's representative. A power line shall be considered fouled when any object is brought to a point less than eight (8) feet (2.5 m) therefrom. A signal line shall be considered fouled when any object is brought nearer than six (6) feet (1.8 m) to any wire or cable. Cranes, trucks and other equipment shall be considered as fouling the track, power line or signal line when failure of equipment, whether working or idle, with or without load, will obstruct the track or other Conrail facilities.

Equipment used by the sponsor shall be in first-class condition to preclude any failure that would cause interference with the operation of Conrail trains or damage to its facilities. The sponsor's equipment shall not be placed or put in operation adjacent to the tracks or facilities of Conrail without obtaining clearance from Conrail's representative. All such equipment shall be operated by the sponsor in a manner satisfactory to Conrail. No equipment or material shall be stored on Conrail property.

In general, a hazard occurs and a flagman is necessary in the following circumstances: (1) the driving of sheeting or piles within twenty five (25) feet (7.6 m) of the tracks, (2) the removal or demolition of all or part of an overhead or adjacent structure, (3) the erection of any structural material, or (4) the performance of any other operation that could obstruct or foul (as described above) the tracks or other facilities of Conrail as determined by Conrail's representative.

Minimum overhead and lateral clearances as specified by Conrail, shall be maintained during the performance of all work. Existing overhead and lateral clearances shall be maintained during construction unless a temporary reduction in clearance for construction purposes is approved, in writing, by Conrail. The sponsor shall erect a highly visible construction fence no closer than fifteen (15) feet (4.6 m) from the centerline of the track through the work area to insure that the lateral clearance requirement is being met.

All wire and attachments shall be treated as live unless notified by Conrail's representative that same have been grounded and de-energized. Particular attention shall be given to the use of hand lines containing metal strands which cannot be used when working near or above exposed live wires. When working over wires, tools and materials not in use shall be stored in a manner to prevent them from falling. Tools or materials shall not be thrown to or by men working over the wires. The sponsor shall be responsible for locating and protecting all underground facilities.

Painting and paint removal procedures shall be approved by the Conrail and inspected by Conrail's representative prior to beginning the work over railroad right of way. The sponsor shall protect the track structure and railroad property from any material used in conjunction with performing the work. A flagman shall be required whenever the above described work fouls or is likely to foul the track, as previously defined.

The sponsor shall give notice to Conrail's representative at least fourteen (14) days in advance of the time work is to be commenced. Conrail shall assign, at the sole cost and expense of the

sponsor, conductors and/or flagmen, or other similar qualified employees to protect Conrail's trains and facilities when in the opinion of its representative, the construction work will cause or may cause a hazard to Conrail facilities and the safe operation of trains. No operations of the sponsor shall be carried out without all the necessary protection to properly safeguard the work. The minimum hours per day for railroad employees engaged in flagging service shall be eight (8) hours. The overtime rate will be charged for all time in excess of eight (8) hours. Flagmen are paid from the time they leave headquarters until they arrive back at headquarters. The travel time to and from project site is known as "deadheading" and is paid at full rate of pay, plus travel expenses. No conductor or flagman may remain on duty longer than twelve (12) hours in any twenty-four (24) hour period.

The providing of flagmen or inspectors or the taking of other precautionary measures, shall not, however, relieve the sponsor from liability for payment of damages caused by their operations. The sponsor must obtain permission from the flagman before fouling or obstructing any track.

The sponsor shall be responsible for damage to Conrail facilities or property arising out of the execution of its work. Conrail shall undertake any necessary repair work at the sole cost and expense of the sponsor. Billing for the work shall be in accordance with Conrail's standard billing procedures.

Conrail labor shall be charged to sponsor at actual rate plus amount paid for insurance, railroad retirement, excise tax, vacation allowance, holidays, health and welfare benefits, small tools, 401k payment and overhead in accordance with Conrail's standard billing procedures. Materials shall be charged to the sponsor at actual cost to Conrail plus transportation costs, handling expense and applicable taxes.

# RAILROAD ENGINEERING AND INSPECTION

Conrail, at its sole discretion, may assign an engineer or inspector for the general protection of railroad property and operations during the construction of the project. This inspection service shall be supplied at the sole cost and expense of the sponsor.

# **PAYMENT OF RAILROAD SERVICES**

It is a requirement that the sponsor shall reimburse Conrail in full for work undertaken by Conrail in accordance with any provision of these special requirements. Final contract payment shall not be made by the sponsor to its contractor, sub-contractor, consultant or agent, until Conrail certifies that all railroad bills against them, if any, have been paid in full.

# TEMPORARY GRADE CROSSING

Under most circumstances, a grade crossing of our track will not be permitted. Should the sponsor demonstrate a necessity for a temporary grade crossing of Conrail's tracks, the sponsor shall be required to apply for and execute the standard private grade crossing agreement for

each crossing required. Application for the crossing shall be made to Conrail at least twelve (12) weeks before the crossing is required and addressed to:

Consolidated Rail Corporation 3501 Island Ave. Philadelphia, PA 19153 Attention: Robert J. Dempsey Manager, Real Estate Services Telephone: (215) 937-7562

A letter size plan showing the location, size, construction details, and access to the requested crossing should accompany the letter of application. The plan shall be fully detailed and dimensioned with all Conrail facilities shown and referenced. The sponsor shall state the purpose for which the crossing is needed and the expected life of the crossing. All application fees, construction, maintenance, protection and removal costs shall be at the sole cost and expense of the sponsor. The roadbed and all other Conrail facilities will be restored to the original condition subject to the approval of Conrail's designated representative.

### SHEETING AND SHORING REQUIREMENTS

The following items are to be included in the design and construction procedures for all permanent and temporary facilities adjacent to Conrail tracks:

- 1) Footings for all piers, columns, walls or other facilities shall be located and designed so that any temporary sheeting and shoring for support of adjacent track or tracks during construction shall not be closer than ten (10) feet (3.0 m) from the centerline of the nearest track.
- When excavation for construction of the above mentioned facilities is within the theoretical railroad embankment line (see Conrail Drawing SK-1, attached), interlocking steel sheet piling, driven prior to excavation, must be used to protect track stability. The use of trench boxes or similar devices is not acceptable. Soldier piling and lagging will be considered for supporting adjacent track(s) only when its use is approved by Conrail. Consideration for the use of soldier piling and lagging shall be made if the required penetration of steel sheet piling cannot be obtained and when dry, non-running, stable material will be encountered.
- The sheeting shall be designed to support all lateral forces caused by the earth, railroad and other surcharge loads. The railroad loading to be applied is an E-80 loading. This loading consists of 80 Kip (356 KN) axles spaced five (5) feet (1.5 m) on centers. The lateral forces acting on the sheeting shall be computed as follows:
  - (a) The Rankine Theory shall be used to compute the active earth pressure due to the weight of the soil.

- (b) The Boussinesq analysis shall be used to determine the lateral pressure caused by the railroad loading. The load on the track shall be taken as a strip load with a width equal to the length of the ties (8'-6" or 2.6 m). The vertical surcharge, q (psf), caused by each axle, shall be uniform and equal to the axle weight divided by the tie length and the axle spacing (5'-0" or 1.5 m). For an E-80 loading, this results in:  $q = 80,000 /(8.5 \times 5) = 1882$  psf (90.1 KPa).
- (c) The horizontal pressure due to the live load surcharge at any point on the sheet piling wall is Ph and can be calculated by the following: Ph =  $(2q/\pi)$ (B-sin B cos 2 oc) (see Conrail Drawing SK-2, attached).
- 4) Deflection design criteria is as follows:
  - (a) 1/2" (1.27 cm) maximum deflection for sheet piling ten (10) feet (3.0 m) from centerline of the nearest track.
  - (b) 1" (2.54 cm) maximum deflection for sheet piling greater than ten (10) (3.0m) feet from centerline of the nearest track.
  - (c) Use K (at-rest earth pressure) for design of all braced and tie-back excavations.
- 5) The allowable stresses for the sheet piling and other steel members (wales, struts, etc.) shall be in accordance with AREA Chapter 15, Part 1. These allowable stresses may be increased ten percent (10%) due to the temporary nature of the installations. A factor of safety of at least 1.5 must be used on temporary sheeting for the embedment length (i.e. multiply calculated embedment depth by 1.5).
- Where soil or rock anchors are used, all anchors must be tested. Testing shall be in accordance with industry standards with ten percent (10%) of the anchors "Performance Tested" and all others "Proof Tested". All tie-back anchor stresses are to be in accordance with AREA Chapter 8, Part 20.5.7.
- 7) Exploratory trenches, three (3) feet (0.9 m) deep and fifteen (15) inches (0.4 m) wide in the form of an "H" with outside dimensions matching the outside of sheeting dimensions are to be hand dug, prior to placing and driving steel sheeting, in areas where railroad underground installations are known to exist. These trenches are for exploratory purposes only and are to be backfilled with the backfill compacted immediately. This work must be done in the presence of Conrail's representative.
- 8) Absolute use of track is required while driving sheeting within fifteen (15) feet (4.6 m) from centerline of a live track. The procedure for arranging the use of track shall be as outlined on pages Three and Four.
- 9) Cavities adjacent to the sheet piling, created by the driving of sheet piling, shall be filled with sand and any disturbed ballast must be restored and tamped immediately.

- 10) Sheet piling shall be cut off at the top of tie during construction. After construction and backfilling has been completed, piling within ten (10) feet (3.0 m) from centerline of track, or when bottom of excavation is below a line extending a 1:1 slope from end of tie to point of intersection with sheeting, shall be cut off eighteen (18) inches (0.5 m) below existing ground line and left in place.
- 11) Any excavation adjacent to track shall be covered and ramped and provided with barricades as required by Conrail. A lighted walkway with a handrail must be provided adjacent to the track for any excavation within ten (10) feet (3.0 m) of the centerline.
- 12) Final backfilling of excavation shall be as required by project specifications.
- 13) The sponsor is to advise Conrail of the time schedule of each operation and obtain approval of Conrail for all work to be performed adjacent to Conrail tracks so that it may be properly supervised by railroad personnel.
- 14) All drawings for temporary sheeting and shoring shall be prepared and stamped by a Registered Professional Engineer and shall be accompanied by complete design computations when submitted for approval.
- 15) Where physical conditions of design impose insurmountable restrictions requiring the placing of sheeting closer than specified above, the matter must be submitted to Conrail for approval of any modifications.
- 16) Five (5) copies of the submission are to be sent to Conrail's Area Engineer. The sponsor is advised to expect a minimum thirty (30) day review period from the day the submission is received by the Area Engineer.
- 17) Conrail's representative must be present at the site during the entire sheeting and shoring procedure period. The sponsor must notify the railroad representative at least seventy-two (72) hours in advance of the work. No changes will be accepted after that time.

# **ERECTION, HOISTING AND DEMOLITION REQUIREMENTS**

- 1) A plan showing the locations of cranes, horizontally and vertically, operating radii, with delivery or disposal locations shown. The location of all tracks and other railroad facilities should also be shown.
- 2) Crane rating sheets showing cranes to be adequate for 150% of the actual weight of the pick. A complete set of crane charts, including crane, counterweight, and boom nomenclature is to be submitted.
- Plans and computations showing weight of picks must be submitted. Where beams are being removed over Conrail facilities, the weight shall include the weight of concrete or other material that will be included in each pick. Calculations shall be made from plans of

the existing and/or proposed structure showing complete and sufficient details with supporting data for the demolition or erection of the structure.

- 4) If the sponsor can prove to Conrail that plans do not exist and weights must be calculated from field measurements, the field measurements are to be made under the supervision of the Registered Professional Engineer submitting the procedure and he shall include sketches and estimated weight calculations with his procedure. If possible, field measurements shall be taken with a Conrail representative present. Weights shall include the weight of concrete, or other material, that will be included in the lifts.
- 5) If the procedure involves either the cutting of steel or the bolting of joints which would affect Conrail operations, a detailed staging plan with estimated durations will be required.
- 6) A location plan showing all obstructions such as wires, poles, adjacent structures, etc., must be provided to show that the proposed lifts are clear of these obstructions.
- 7) A data sheet shall be prepared listing the type, size and arrangements of slings, shackles, or other connecting equipment. Include copies of a catalog or information sheets for specialized equipment.
- 8) A complete procedure is to be included, indicating the order of lifts and any repositioning or rehitching of the crane or cranes.
- 9) Demolition shield submittals must include a plan showing the details of the shield, a written installation and removal procedure and design calculations verifying the capacity of the shield. The shield should be designed for a minimum load of fifty (50) pounds/sq.ft (245 kgs/sq.m)plus the weight of the equipment, debris and any other load to be carried.
- 10) Temporary support of any components (overhead or undergrade) or intermediate stages is to be shown and detailed. A guardrail (railroad) will be required to be installed in a track where a temporary bent is located within twelve (12) feet (3.7 m) from the centerline of that track.
- 11) A time schedule of the various stages must be shown as well as a schedule for the entire lifting procedure.
- 12) All bridge erection or demolition procedures submitted will be prepared, signed and sealed by a Registered Professional Engineer.
- 13) Five (5) copies of the lifting procedures are to be sent to Conrail's Area Engineer. The sponsor is to expect a minimum thirty (30) day review period from the day the submission is received by The Area Engineer.
- 14) Conrail's representative must be present at the site during the entire demolition and erection procedure period. The sponsor must notify the railroad representative at least

- seventy-two (72) hours in advance of the work. No changes will be accepted after that time.
- 15) The name and experience of the employee supervising the operation must be supplied to Conrail.

# **OVERGRADE BRIDGE REQUIREMENTS**

#### **CLEARANCES**

- 1) The minimum vertical clearance above the top of the higher rail shall be twenty three (23) feet (7 m) at all times. In areas where the railroad has been electrified with a catenary wire, and areas which are likely to be electrified, the minimum vertical clearance must be twenty four (24) feet, six (6) inches (7.5 m) above the top of the higher rail.
- 2) The minimum horizontal clearance measured from the centerline of track to the near face of the obstruction must be twenty (20) feet (6.1 m) for tangent track and twenty one (21) feet (6.4 m) for curves. See Conrail Standard Plan 48754-B, attached.
- Whenever practicable, bridge structures must have the piers and abutments located outside of the railroad right of way. All piers located less than twenty five (25) feet (7.6 m) from the centerline of track require a crash wall designed in accordance with specifications outlined in the current AREA manual.
- 4) All piers should be located so that they do not interfere with ditches. Where special conditions make this impossible, an explanation of these conditions must be submitted with the drainage plans for review by Conrail.
- 5) The permanent clearances should be correlated with the methods of construction so that temporary construction clearances will not be less than the minimum allowed.
- 6) Bridge structures shall provide sufficient lateral and vertical clearance for anticipated future tracks, changes in track centers and raising of tracks for maintenance purposes. The locations of these tracks shall be determined by inquiry to Conrail.
- 7) The profile of the top of rail should be examined to determine if the track is in a sag at the location of the bridge. If the track is in a sag, the vertical clearance from the track to the bridge should be increased sufficiently to allow raising the track to remove the sag.
- 8) Plans for bridges must show dimensioned locations of all utilities which might be located on the railroad right of way.
- 9) Vertical and horizontal clearances must be adjusted so that the sight distance to railroad signals is not reduced from what is existing.

- 10) All proposed temporary clearances which are less than those listed above must be submitted to Conrail for review and must be approved by Conrail prior to construction.
- 11) Clearances are subject to the requirements of the state in which the construction takes place and must be approved by the State and Conrail if less than those prescribed by law.

#### **DRAINAGE**

- 1) Maintaining the existing drainage and providing for future drainage improvements is of the utmost importance. Conrail will give special attention to reviewing drainage details.
- 2) Drainage plans must be included with the general plans submitted to Conrail for approval. These plans must include hydrologic and hydraulic studies and computations showing the frequency and duration of the design storm used, as well as the method of analysis such as Soil Conservation Service or the Rational method. Conrail uses storms with a 100-year recurrence interval as the minimum design storm.
- 3) Lateral clearances must provide sufficient space for construction of the required track ditch parallel to the standard roadbed section. If the ditch cannot be provided, or the pier will interfere with the ditch, then a culvert of sufficient size must be provided. See Conrail Standard Plans 48754-B and 48747, attached.
- Ditches and culverts must be sized to accommodate all increased run-off due to the construction and the increased size must continue to the natural outlet of the ditch. Ditches must be designed in accordance with good drainage engineering practices and must meet all local codes and ordinances.
- 5) No scuppers or other deck drains, roadway drainage, catch basins, inlets or outlets are permitted to drain onto Conrail property. Any variation of this policy must have the prior approval of Conrail. If an exception is ultimately granted, maintenance of such should not be Conrail's. Drainage from scuppers and deck drains must be conveyed through pipes, preferably to a point which is off Conrail property. If the drainage must be conveyed into a railroad ditch, calculations must be provided to Conrail which indicate the ability of the ditch to carry the additional runoff.
- 6) Additional drainage may require the installation of a pipe or pipes, new ditch or reprofiling of the existing ditch.

#### **EROSION CONTROL**

1) Embankment slopes on Conrail property adjacent to the track must have a slope of 2:1 or less and be paved for a minimum of two (2) feet (0.6 m) beyond the outside edge of the bridge foundation structure. The purpose of the pavement is to minimize erosion of the embankment material and to reduce deterioration of the sub-grade material by drainage

- water. The pavement shall consist of a prepared sub-base and/or filter fabric with grouted rip-rap on the surface.
- The general plans for the bridge should indicate the proposed methods of erosion control during construction and must specifically address means to prevent silt accumulation in ditches and culverts and to prevent fouling the track ballast and sub-ballast. If the plans do not show erosion control, the contractor must submit a proposed method of erosion control and must have this method approved by Conrail prior to beginning any grading on the site.
- 3) Existing track ditches must be maintained at all times throughout the construction period. After the construction has been completed, all erosion and siltation must be removed and the ditches must be restored.
- 4) Conrail's approval of drainage and erosion control plans will not relieve the sponsor submitting these plans from ultimate responsibility for a satisfactory plan.

#### **REFERENCES**

- 1) In areas where underground utilities may be affected, Conrail's C.E. 8, "Specifications for Pipeline Occupancy" will govern.
- 2) In areas where power or communication lines will be affected, Conrail's C.E. 4, "Specifications for Wire, Conduit and Cable Occupations" will govern.

#### **APPENDIX E**

# AGREEMENT BETWEEN CONTRACTOR

#### AND

#### **ARTIST**

This Agreement made the day of	2005
between	(hereinafter referred to as
"Contractor") having its principal office at	and
	(hereinafter referred to as
the "Artist").	

#### WITNESSETH:

WHEREAS, pursuant to N.J.S.A. 52:16A-29 et seq., NJ TRANSIT is desirous of incorporating fine arts in the following project:

#### Name of Project:

#### **Location of Project:**

WHEREAS, the Artist has submitted a proposal to design such Artwork, under the terms and conditions set forth herein and in accordance with the following Attachments which are hereby made a part of this Agreement:

**Appendix "1"** - Federal Contract Requirements

Attachment "A" - Fees, Materials and Payment Schedule

Attachment "B" - Progress Schedule

Attachment "C" - Scope of Work

Attachment "D" - Artist Proposal for Artwork

The proposal to design and deliver the Artwork shall be incorporated in this Agreement as Attachment "D". To the extent that this contract contains requirements different from the proposal, the contract shall control.

NOW, THEREFORE, in consideration of the mutual promises and undertakings contained herein, the parties covenant and agree as follows:

- 1. The Artist shall provide all design, engineering, project coordination and project management services which are necessary in the performance of work required by this Agreement. The Artist is responsible for all of the Artist's travel and transportation expenses associated with the performance of this Agreement.
- 2. The Artist is responsible for the provision of the Artwork design, documents and submittals which include engineering drawings and lighting design, physical layout, complete fabrication, transportation, installation and documentation of the Artwork. The Artist shall ensure that all structural elements of the Artwork and mounting devices are designed and certified by an engineer licensed in the State of New Jersey. The Artist agrees to provide all necessary lighting fixtures and fixture-ready attachments including cords and plug connections, bulbs and mounting bracket/tracks. The Artist shall consult with the project architect prior to procurement of lighting equipment for review and approval of manufacturer, type and number of fixtures, and to ensure proper interface with electrical source(s) in the building.
- 3. It shall be Contractor's responsibility to prepare and make available the site for preparation, modification, or other necessary work by the Artist and Contractors performing services relevant to the Artwork.
- 4. Contractor and NJ TRANSIT, through their authorized representative or representatives, may at reasonable times and on reasonable notice, set up mutually convenient times to review the Artwork while in the process of being designed, completed and installed. Such visits shall be preceded by a telephone communication to the Artist with verbal confirmation by the Artist.
- 5. The Artist will complete the Artwork in accordance with the Progress Schedule contained herein in Attachment B.

In the event the installation is not properly completed by the agreed-upon date, except for causes beyond the Artist's control (such as, but not limited to fire, theft, strikes,

lock-outs, materials shortages, illness and acts of God) and as liquidated damages (not as a penalty), Contractor may deduct from the amount due Artist \$\_\_\_\_\_\_ for each day's delay encountered until installation is properly completed, or the Agreement terminated. The Artist may request Contractor, not later than 10 days before the installation date set forth in this contract, for an extension of that installation date. Such extension, if granted, must be in writing. If the site is not available to receive the Artwork when the Artist has completed on schedule, Contractor will pay for all excess costs of storage and transportation. Contractor may at its own expense take possession of the Artwork and store same at its own site. In that event, Contractor shall bear risk of loss. Subsequent delivery and installation shall be in accordance with the terms stated herein.

- 6. The Artist shall regularly confer with Contractor's Project Management staff and attend required meetings in order to verify and insure that the Artwork will be incorporated into the project at the proper time. Meetings shall be held at a time which is mutually convenient for Contractor and the Artist.
- 7. The Artist must take all reasonable precautions to protect the property of NJ TRANSIT and all others at and adjacent to the installation site of the work. The Contractor will be responsible for adequate protection of the public during installation.
- 8. To the extent possible, NJ TRANSIT will not intentionally remove, alter, modify or change the Artwork unless it is necessary for the preservation, safety, repair, destruction or renovation of the premises in which the Artwork is installed as determined by NJ TRANSIT at its sole discretion. NJ TRANSIT shall not otherwise remove, alter, modify, destroy or change the Artwork without the agreement of the Artist. Such agreement shall be governed by the applicable provisions of 17 U.S.C.A. □113. The Artist further waives any rights he may have against and agrees not to institute suit against NJ TRANSIT for any removal, alterations, modifications, damage, destruction or changes that may occur to the Artwork in the event that the premises are no longer under the ownership, supervision, or control of NJ TRANSIT. The Artist hereby agrees to provide a suggested maintenance

schedule for the Artwork to NJ TRANSIT at the time of completion of the project. These obligations and waivers shall survive the termination of this Agreement.

- 9. The Artist agrees that for one year following the installation and acceptance of the Artwork, he shall correct, without charge, any defects in the Artwork which require repair or correction.
- 10. A plaque containing the title of the Artwork, name of the Artist, and recognition of NJ TRANSIT shall be publicly displayed and identified with the Artwork. Such plaque shall be provided by the Artist upon completion of the work. The design and content of the plaque is subject to approval in writing by NJ TRANSIT.
- 11. The Artist agrees to register the completed Artwork for copyright immediately upon acceptance by NJ TRANSIT. Copyright notice shall appear on the work and project plaque. The Artist grants to NJ TRANSIT and the NJSCA the non-exclusive right to photograph the Artwork and otherwise reproduce it in film, tape, digital format, or any other reproductive media, and publish the same with proper identification of copyright, including identification of Artist. With the exception of the above listed parties, Artist reserves to himself all copyrights in the work, the preliminary design, and any incidental works, made in the creation of the work.

The Artist further agrees that the Artwork completed under this Agreement is a unique work and that copies or editions of the Artwork will not be created and installed at any other location worldwide.

NJ TRANSIT will consult with the Artist prior to the production of any commercially marketable items which depict the Artwork and are intended to be offered for sale to the general public.

12. The Artist agrees to supply NJ TRANSIT and the NJSCA with full archival documentation of the completed Artwork including, but not limited to 35mm slides, black and white and color photographs, original presentation maquette, original drawings and a written physical description. Such documentation is to be placed in the archives of the NJSCA. The submitted material shall not be offered for sale at any time.

This documentation is to be supplied by the Artist not later than 30 days after installation of the Artwork and must comply with the NJSCA documentation format for visual artwork.

- 13. NJ TRANSIT may elect to publicly dedicate/unveil the Artwork and the Artist will attend said dedication/unveiling, upon adequate written notice from NJ TRANSIT, unless excused by NJ TRANSIT.
- 14: Contractor/NJ TRANSIT may terminate this Agreement for convenience by giving written notice to the ARTIST of such termination sent to the ARTIST at the address set forth in paragraph 34. In that event the ARTIST shall be paid for all reasonable costs of work properly completed on the date of termination including all reasonable costs of materials and equipment purchased and/or rented to date, and any penalties due on leased space and subcontracts for fabrication of the ARTWORK.
  - 15: Contractor/NJ TRANSIT may terminate this Agreement for cause at any time

if the ARTIST has materially failed to comply with terms of the Agreement. In the event of such failure, NJ TRANSIT shall promptly give written notification to the ARTIST of its intent to terminate and the reasons thereof. The ARTIST shall have ten (10) days, or such additional time as NJ TRANSIT may grant, after receipt of notice to cure its failure. If the failure is not cured to the satisfaction of NJ TRANSIT, the ARTIST shall be held in breach of contract and the contract terminated (in whole or in part) effective immediately.

After receipt of notice of termination, the ARTIST shall not incur any new obligations without the approval of NJ TRANSIT and shall cancel as many outstanding obligations as possible. NJ TRANSIT will evaluate each obligation deemed non-cancellable by the ARTIST in order to determine its eligibility for inclusion in compensable costs. Compensation shall be made for work properly completed and approved by NJ TRANSIT prior to the date of termination.

If this Agreement or any part thereof is terminated for cause, NJ TRANSIT may procure services similar to those so terminated. The ARTIST shall be liable to NJ TRANSIT for any reasonable excess costs incurred for such similar services.

No damages of any nature shall be claimed against NJ TRANSIT in the event it exercises this right of termination. The rights and remedies available to in this section shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Agreement.

If, after notice of termination of this Agreement under the provisions of this section, it is determined for any reason that the ARTIST was not in default under the provisions of this section, or that the default was excusable under the provisions of this section, the rights and obligations of the parties shall be the same as if the notice of termination had been issued pursuant to Section 14.

- 16. Ownership of all materials, designs and drawings related to the Artwork, originated and prepared for NJ TRANSIT after the date of signing of this Agreement, shall rest with NJ TRANSIT, and shall be submitted by the Artist to the project archives of the NJSCA. The Artist shall provide NJ TRANSIT and the NJSCA with copies of all pertinent maintenance information and structural design information in accordance with the requirements of this Agreement.
- Artist will receive a full review and discussion of plans to date and complete background information relevant to all considerations affecting the Artist's responsibility to fulfill its obligations. In particular, Contractor shall make available to the Artist all necessary background and orientation materials, adequate notice of scheduled presentations, meetings dealing with matters affecting the Artist's participation in construction schedules, deadlines and plans affecting the Artist's installation of his/her work, and shall invite the Artist and afford the Artist full participation at all such presentations and meetings.
- b. The quality of the finishes of the completed Artwork shall be as described by the Artist in his Design Development Plans unless Contractor and NJ TRANSIT

have authorized necessary changes. NJ TRANSIT reserves the right to reject the completed Artwork if it does not meet the stated specifications of the Design Development Plans.

- 18. The Contracting Officer for this Agreement shall be: \_\_\_\_\_\_\_\_, (NAME OF CONTRACTING OFFICER). Wherever this Agreement requires any notice to be given to or by Contractor, or any determination to be made by Contractor, the Contracting Officer or his Designee shall represent and act for the Contractor.
- 19. The Artist shall schedule any on-site activities with Contractor, and will provide notice of on-site activities to Contractor and NJ TRANSIT.
- 20. The Artist's status shall be that of an independent contractor and not as agent or employee of Contractor or NJ TRANSIT.
- 21. The Artist agrees not to assign this Agreement in whole or in part, or any moneys hereunder, without the prior written approval of NJ TRANSIT. Following installation of the artwork and its acceptance by NJ TRANSIT, and acceptance of the overall project, the Contractor will assign this Agreement and all its rights hereunder to NJ TRANSIT.
- 22. The Artist agrees that in the performance of this Agreement he will obey, abide and comply with applicable Federal, State, County and Municipal laws and regulations.
- 23. If incorporated in a state other than the State of New Jersey, the Artist shall promptly file with the Secretary of State an application for a Certificate of Authority to do business in the State of New Jersey and shall comply with the provisions of the laws of the State of New Jersey in that regard.
- 24. This Agreement shall be governed by and construed and the rights and obligations of the parties hereto shall be determined in accordance with the laws of the State of New Jersey. The courts of New Jersey Shall have sole jurisdiction over any suits brought concerning this Agreement.
- 25. The Artist hereby covenants and agrees to take, use, provide and make all proper and necessary and sufficient precautions, safeguards and protections against the occurrence or happening of any accident, injuries, damages or hurt to any person or property during the progress of the work herein covered and to indemnify, defend, and save harmless

the Contractor and NJ TRANSIT, its officers, agents, servants and employees for the payment of all sums of money by reason of all, or any, such accidents, injuries, damages or hurt that may occur upon or about such work and all fines, penalties and loss incurred for or by reason of the violation of any ordinance, regulation, or statute, while the said work is in progress.

- 26. The risk of loss to the Artwork shall remain on the Artist at all times until the Artwork is delivered to Contractor. The Artist shall ensure that All Risk insurance on the Artwork, which shall protect the Artist's, Contractor and NJ TRANSIT's interests, is carried by the Artist or the Artist's fabricators and other subcontractors during the period of fabrication.
- 27. The Artist will carry Comprehensive General Liability insurance in the amount of \$5,000,000 Combined Single Limit for each occurrence. Such insurance shall name NJ TRANSIT, Contractor, (LIST OTHER INSURED) as additional insureds. This insurance to provide coverage during the period of fabrication and installation and until the installation is accepted in writing by Contractor.

The certificate of such insurance shall be submitted to the Contractor and NJ TRANSIT upon signing of this Agreement and shall be kept current through acceptance of the artwork.

28. The parties to this contract do hereby agree that the provisions of N.J.S.A. 10:2-1 through 10:2-4, dealing with discrimination in employment on public contracts, and the rules and regulations promulgated pursuant thereunto, are hereby made a part of this contract and are binding upon them. The Artist further agrees to comply with the requirements of N.J.S.A. 10:5-31, et seq., and all rules and regulations promulgated thereunder.

For the purpose of this Agreement the "Public Agency Compliance Officer" referenced below is NJ TRANSIT's Director of Records Management & Procurement Compliance, the "Public Agency" is NJ TRANSIT and the contractor is the Artist.

During the performance of this contract, the contractor agrees as follows:

The contractor or subcontractor, where applicable, will not discriminate against any employee or applicant for employment because of age, race, creed, color, national origin, ancestry, marital status, sex, affectional or sexual orientation. The contractor will take affirmative action

to ensure that such applicants are recruited and employed, and that employees are treated during employment, without regard to their age, race, creed, color, national origin, ancestry, marital status, sex, affectional or sexual orientation. Such action shall include but not be limited to the following: employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided by the Public Agency Compliance Officer setting forth provisions of this non-discrimination clause;

The contractor or subcontractor, where applicable, will in all solicitations or advertisements for employees placed by or on behalf of the contractor, state that all qualified applicants will receive consideration for employment without regard to age, race, creed, color, national origin, ancestry, marital status, sex, affectional or sexual orientation;

The contractor or subcontractor, where applicable, will send to each labor union or representative of workers with which it has a collective bargaining agreement or other contract or understanding, a notice, to be provided by the agency contracting officer advising the labor union or workers' representative of the contractor's commitments under this act and shall post copies of the notice in conspicuous places available to employees and applicants for employment.

The contractor or subcontractor, where applicable, agrees to comply with any regulations promulgated by the Treasurer pursuant to P.L. 1975, c.127, as amended and supplemented from time to time and the Americans with Disabilities Act.

The contractor or subcontractor agrees to attempt in good faith to employ minority and female workers consistent with the applicable county employment goals prescribed by N.J.A.C. 17:27-5.2 promulgated by the Treasurer pursuant to P.L. 1975, c.127, as amended and supplemented from time to time or in accordance with a binding determination of the applicable county employment goals determined by the Affirmative Action Office pursuant to N.J.A.C. 17:27-5.2 promulgated by the Treasurer pursuant to P.L. 1975, c.127, as amended and supplemented from time to time.

The contractor or subcontractor agrees to inform in writing appropriate recruitment agencies in the area, including employment agencies, placement bureaus, colleges, universities, labor unions, that it does not discriminate on the basis of age, creed, color, national origin, ancestry, marital status, sex, affectional or sexual orientation and that it will discontinue the use of any recruitment agency which engages in direct or indirect discriminatory practices.

The contractor or subcontractor agrees to revise any of its testing procedures, if necessary, to assure that all personnel testing conforms with the principles of job-related testing, as established by the statutes and court decisions of the State of New Jersey and as established by applicable Federal law and applicable Federal court decisions.

The contractor or subcontractor agrees to review all procedures relating to transfer, upgrading, downgrading and layoff to ensure that all such actions are taken without regard to age, creed, color, national origin, ancestry, marital status, sex, affectional or sexual orientation and conform with the applicable employment goals, consistent with the statutes and court decisions of the State of New Jersey, and applicable Federal law and applicable Federal court decisions.

The contractor and its subcontractors shall furnish such reports or other documents to the Affirmative Action Office as may be requested by the office from time to time in order to carry out the purposes of these regulations, and public agencies shall furnish such information as may be requested by the Affirmative Action Office for conducting a compliance investigation pursuant to <u>Subchapter 10 of the Administrative Code (N.J.A.C. 17:27)</u>.

- 29. The New Jersey prevailing wage laws are hereby made a part of this contract. The Artist's signature on the proposal and on this Agreement is his guarantee that neither he nor any subcontractors he might employ to perform work are listed by or are recorded in the New Jersey Department of Labor as one who has failed to pay prevailing wages in accordance with the provisions of the Prevailing Wage Act.
- 30. The Artist does hereby warrant and represent that this Agreement has not been solicited or secured, directly or indirectly, in a manner contrary to the laws of the State of

New Jersey or of the United States and that said laws have not been violated as they relate to the procurement or the performance of this Agreement by any conduct, including the paying or giving of any fee, commission, compensation, gift, gratuity or consideration of any kind, directly or indirectly, to any NJ TRANSIT Contractor, or State of New Jersey employee, officer or official.

- 31. The Artist warrants and represents that no person has been employed to solicit or secure this contract in violation of any other laws and further represents that all applicable laws and regulations shall be complied with in the performance of this contract.
- 32. Prohibited Interest: No member, officer, or employee of NJ TRANSIT or its subsidiaries shall have any interest, direct or indirect, in this Agreement or the proceeds thereof. No former member, officer or employee of NJ TRANSIT or its subsidiaries who, during his tenure, had a direct, substantial involvement with matters that are closely related to this Agreement, shall have any interest, direct or indirect, in this Agreement or the proceeds thereof.
- 33. Release of Claims: It is agreed the Artist acceptance of final payment from Contractor/NJ TRANSIT shall release in full all claims and liability of whatsoever nature against Contractor and NJ TRANSIT for anything done or furnished or in any manner growing out of the performance of this Contract except such claims as may specifically excepted in writing by the Artist from the operation of such release.
- 34. All notices should be sent certified mail, return receipt requested, to the parties at the addresses listed below:

#### (Name and Address of Contractor)

and;

#### (Name and Address of Artist)

35. All changes to this Agreement shall be in writing and signed by the Contractor and the Artist.

	(2) (3) (4)	Certificate of Final Acceptance, Final Release of Claims, Affidavit of Payment of Debts and Claims, Affidavit of Release of Liens, Final Certificate of Payment							
IN	WITNES	WHEREOF,	Contracto	r and	the	Artist	have	executed	this
Agreement the d	ay and yea	ar above written							
			ARTIS	ŝΤ					
Witness			Ву	Artist					
			CONT	RACT	OR				
Witness			Ву:	Title					

but are not limited to, the following items:

Project Close-Out: Submissions required prior to final payment, include,

36.

#### **APPENDIX 1**

#### FEDERAL CONTRACT REQUIREMENTS

- 1. NO FEDERAL GOVERNMENT OBLIGATIONS TO THIRD PARTIES: The Artist agrees that, absent the Federal Government's express written consent, the Federal Government shall not be subject to any obligations or liabilities to any subrecipient, any third party contractor, or any other person not a party to the contract in connection with the performance of the project. Notwithstanding any concurrence provided by the Federal Government in or approval of any solicitation, subagreement, or third party contract, the Federal Government continues to have no obligations or liabilities to any party, including the subrecipient and third party contractor.
  - 2. FALSE OR FRAUDULENT STATEMENTS AND CLAIMS:
- A.) The Artist recognizes that the requirements of the Program Fraud Civil Remedies Act of 1986, as amended, 49 USC § 3801 et seq. and USDOT regulations, "Program Fraud Civil Remedies," 49 CFR Part 31, apply to its actions pertaining to the project. Accordingly, by signing the Agreement, the Artist certifies or affirms the truthfulness and accuracy of any statement it has made, it makes, or it may make pertaining to the Agreement. In addition to other penalties that may be applicable, the Artist also acknowledges that if it makes a false, fictitious, or fraudulent claim, statement, submission, or certification, the Federal Government reserves the right to impose the penalties of the Program Fraud Civil Remedies Act of 1986, as amended, on the Artist to the extent the Federal Government deems appropriate.
- B.) The Artist also acknowledges that if it makes a false, fictitious, or fraudulent claim, statement, submission, or certification to the Federal Government in connection with an urbanized area formula project financed with Federal Assistance authorized by 49 USC §5307, the Government reserves the right to impose on the Artist the penalties of 18 USC § 1001 and 49 USC §5307(n)(1), to the extent the Federal Government deems appropriate.
- 3. EXCLUSIONARY OR DISCRIMINATORY SPECIFICATIONS: Apart from inconsistent requirements imposed by Federal statute or regulations, the Artist agrees that it will comply with the requirements of 49 USC § 5323(h)(2) by refraining from using any Federal Assistance awarded by FTA to support procurements using exclusionary or discriminatory specifications.
- 4. INTERESTS OF MEMBERS OF OR DELEGATES TO CONGRESS: No member of or delegate to the Congress of the United States shall be admitted to any share or part of this Agreement or to any benefit arising therefrom.
- 5. CERTIFICATION OF DEBARMENT AND SUSPENSION: During the performance of this Agreement, the Artist must be in compliance with the requirements of the United States Department of Transportation, 49 CFR Part 29, and FTA Circular 2015.1. The attached Exhibit sets forth detailed requirements and is hereby made a part of this Agreement.
- 6. TITLE VI COMPLIANCE: During the performance of this Agreement, the Artist, for itself, its assignees and successors in interest (hereinafter referred to as the "Artist") agrees as follows:

- A.) Compliance with Regulations: The Artist shall comply with the Regulations relative to nondiscrimination in federally-assisted programs of the Department of Transportation (hereinafter, "DOT") Title 49, Code of Federal Regulations, Part 21, as they may be amended from time to time (hereinafter referred to as the Regulations), which are herein incorporated by reference and made a part of this Agreement.
- B.) Nondiscrimination: The Artist, with regard to the work performed by it during the Agreement, shall not discriminate on the grounds of race, religion, color, sex, age, or national origin in the selection and retention of subcontractors, including procurements of materials and leases of equipment. The Artist shall not participate either directly or indirectly in the discrimination prohibited by Section 21.5 of the Regulations, including employment practices when contract covers a program set forth in Appendix B of the Regulations.
- C.) Solicitations for Subcontracts Including Procurements of Materials and Equipment: In all solicitations either by competitive bidding or negotiation made by the Artist for work to be performed under a subcontract, including procurements of materials or leases of equipment, each potential subcontractor or supplier shall be notified by the Artist of the Artist's obligations under this Agreement and the Regulations relative to nondiscrimination on the grounds of race, religion, color, sex, age, or national origin.
- D.) Information and Reports: The Artist shall provide all information and reports required by the Regulations or directives issued pursuant thereto, and shall permit access to its books, records, accounts, other sources of information, and its facilities as may be determined by NJ TRANSIT or the FTA to be pertinent to ascertain compliance with such Regulations, orders and instruction. Where any information is required or an Artist is in the exclusive possession of another who fails or refuses to furnish this information, the Artist shall so certify to NJ TRANSIT, or the FTA, as appropriate, and shall set forth what efforts it has made to obtain the information.
- E.) Sanctions for Noncompliance: In the event of the Artist's noncompliance with the nondiscrimination provisions of this Agreement, NJ TRANSIT shall impose such contract sanctions as it or the FTA may determine to be appropriate, including but not limited to:
- 1.) Withholding of payments to the Artist under the Agreement until the Artist complies; and/or
  - 2.) Cancellation, termination or suspension of the Agreement, in whole or in part.
- F.) Incorporation of Provisions: The Artist shall include the provisions of Paragraphs A through F of this Section in every subcontract, including procurements of materials and leases of equipment, unless exempt by the Regulations, or directives issued pursuant thereto. The Artist shall take such action with respect to any subcontract or procurement as NJ TRANSIT or the FTA may direct as a means of enforcing such provisions including sanctions for noncompliance: provided, however, that, in the event an Artist becomes involved in, or is threatened with, litigation with a subcontractor or supplier as a result of such direction, the Artist may request NJ TRANSIT to enter into such litigation to protect the interests of NJ TRANSIT and, in addition, the Artist may request NJ TRANSIT to enter into such litigation to protect the interest of the United States.

- 7. CONTRACT WORK HOURS AND SAFETY STANDARDS: During the performance of this Agreement, the Artist, for itself, its assignees and successors in interest (hereinafter referred to as the "Artist") agrees as follows:
- A.) Overtime Requirements: No Artist or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics shall require or permit any such laborer or mechanic in any work week in which he or she is employed on such work to work in excess of eight hours in any calendar day or in excess of forty hours in such work week unless such laborer or mechanic receives compensation at rate not less than one and one-half times the basic rate of pay for all hours worked in excess of eight hours in any calendar day or in excess of forty hours in such work week, whichever is greater.
- B.) Violation; Liability for Unpaid Wages; Liquidated Damages: In the event of any violation of the clause set forth in subparagraph (b)(1) of 29 CFR Section 5.5, the Artist and any subcontractor responsible therefore shall be liable for the unpaid wages. In addition, such Artist and subcontractor shall be liable to the United States (in case the work done under contract for the District of Columbia or a territory, to such district or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of the clause set forth in subparagraph (b)(1) of 29 CFR Section 5.5 in the sum of \$10 for each calendar day on which such individual was required or permitted to work in excess of eight hours or in excess of the standard work week of forty hours without payment of the overtime wages required by the clause set forth in subparagraph (b)(1) of 29 CFR Section 5.5.
- C.) Withholding for Unpaid Wages and Liquidated Damages: The Contractor, on behalf of DOT or NJ TRANSIT, shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld, from any monies payable on account of work performed by the Artist or subcontractor under any such contract or any other Federal contract with the same Artist, or any other Federally-assigned contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same Artist, such sums as may be determined to be necessary to satisfy any liabilities of such Artist or subcontractor for unpaid wages and liquidated damages as provided in the clause set forth in subparagraph (B)(2) of 29 CFR Section 5.5.
- D.) Nonconstruction Grants: The Artist or subcontractor shall maintain payrolls and basic payroll records during the course of the work and shall preserve them for a period of three (3) years from the completion of the Agreement for all laborers and mechanics, including guards and watchmen, working on the Agreement. Such records shall contain the name and address of each such employee, social security number, correct classifications, hourly rates of wages paid, daily and weekly number of hours worked, deductions made, and actual wages paid. These records shall be made available by the Artist or subcontractor for inspection, copying, or transcription by authorized representatives of NJ TRANSIT, the FTA and the Department of Labor, and the Artist or subcontractor will permit such representatives to interview employees during working hours on the job.
- E.) Subcontracts: The Artist or subcontractor shall insert in any subcontracts the clauses set forth in Paragraphs A through E of this Section and also a clause requiring the subcontractors to

include these clauses in any lower tier subcontracts. The Artist shall be responsible for compliance by any subcontractor or lower tier subcontractor with the clauses set forth in Paragraphs A through E of this Section.

- 8. LIMITATIONS ON LOBBYING: The Artist shall comply with 31 USC 1352, entitled "Limitation on use of appropriated funds to influence certain Federal contracting and financial transactions".
- A.) No appropriated funds may be expended by the recipient of a Federal contract, grant, loan or cooperative agreement to pay any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress or an employee of a Member of Congress in connection with any of the following covered Federal actions: the awarding of any Federal contract, the making of any Federal grant, the making of any Federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment or modification of any Federal contract, grant, loan or cooperative agreement.
- B.) Any Artist who requests or receives a Federally-assisted contract or subcontract in excess of \$100,000 from NJ TRANSIT shall file with NJ TRANSIT the certification attached to this Agreement and entitled "Certification for Contracts, Grants, Loans and Cooperative Agreements" which certifies that the Artist or subcontractor, as applicable, has not made, and will not make, any payment prohibited by paragraph A.) of this Article.
- C.) Any Artist who has made or has agreed to make any payment using non appropriated funds (to include profits from any covered Federal action) which would be prohibited under paragraph A.) of this Article if paid for with appropriated funds, shall file with NJ TRANSIT a disclosure form entitled "Disclosure of Lobbying Activities", which is available from NJ TRANSIT.
- D.) Any certification or disclosure form filed under paragraphs B.) and C.) of this Article shall be forwarded from tier to tier until received by NJ TRANSIT. Any certification or disclosure form shall be treated as a material representation of fact upon which all receiving tiers shall rely. All liability arising from an erroneous representation shall be borne solely by the tier filing that representation and shall not be shared by any tier to which the erroneous representation is forwarded.
- E.) The prohibition on the use of appropriated funds does not apply in the case of a payment of reasonable compensation to an officer or employee of the Artist or subcontractor if the payment is for agency and legislative liaison activities not directly related to a covered Federal action.
- F.) The prohibition on the use of appropriated funds does not apply in the case of any reasonable payment to an officer or employee of the Artist or to a person, other than an officer or employee of the Artist, if the payment is for professional or technical services rendered directly in the preparation, submission or negotiation of any bid, proposal or application for a Federal contract, grant, loan or cooperative agreement.
- 9. AUDIT AND INSPECTION OF RECORDS: The Artist shall retain all Agreement records and permit the authorized representatives of the State of New Jersey, NJ TRANSIT, U.S. Department of Transportation (USDOT) and the Comptroller General of the United States, upon request, to inspect and audit all data and records of the Artist relating to its performance and that of its subcontractors and

assignees, if any, under this Agreement from the effective date hereof through and until the expiration of three (3) years after completion of and final payment for the Project Services. The Artist further agrees to include in all its subcontracts hereunder a provision to the effect that the subcontractor agrees that it will keep all Agreement records and that the State of New Jersey, NJ TRANSIT, the USDOT, and the Comptroller General of the United States and any of their duly authorized representatives shall, until the expiration of three years after final payment under the subcontract, have access to and the right to examine any books, documents, and other records related to the subcontractor's performance under the subcontract.

Documents of every nature prepared pursuant to this Agreement shall be available to and become the property of NJ TRANSIT, and basic notes and other pertinent data shall be made available to NJ TRANSIT upon request without restriction as to their future use.

At least until the expiration of three (3) years after the completion of, and final payment by, NJ TRANSIT for the Project Services, the Artist shall keep and maintain complete financial records showing actual time devoted and cost incurred in connection with services rendered under this Agreement, and it shall make same subject to inspection and audit by NJ TRANSIT, should NJ TRANSIT desire. Following passage of three (3) years from the date of completion of and final payment by NJ TRANSIT for the Project Services, the Artist may request from NJ TRANSIT permission to dispose of the various records. NJ TRANSIT may either order the records be destroyed or surrendered to NJ TRANSIT.

#### 10. BUY AMERICA

Pursuant to Section 165 of the Surface Transportation Assistance Act of 1982 (P.L. 97-424), and the regulation found at 49 CFR Part 661, the Artist agrees that all iron, steel and manufactured products purchased or used as a result of this Agreement shall be of domestic manufacture or origin unless a waiver of these provisions is granted by the U.S. Secretary of Transportation.

There are four exceptions to this requirement:

- (a) That its application would be inconsistent with the public interest.
- (b) That such materials or products are not produced in the United States in sufficient and reasonably available quantities and of a satisfactory quality.
- (c) With respect to rolling stock only, but including train control, communications, traction power equipment, 40 percent may be non-domestic manufacture, but final assembly of such products must take place in the United States.
- (d) That inclusion of domestic material will increase the cost of the overall project contract by more than 25 percent.

#### 11. CLEAN WATER AND CLEAN AIR ACTS

If this Agreement shall be in an amount greater than \$100,000, the Artist shall comply with Section 306 of the Clean Air Act (42 USC 1857(h)), Section 508 of the Clean Water Act (33 USC 1368), Executive Order 11738, Environmental Protection Agency Regulations (40 CFR Part 15), and any other applicable standard, order or requirement issued pursuant to Federal statute or regulation. The Artist shall report violations to NJ TRANSIT, FTA and to the USEPA Assistant Administrator for Enforcement.

#### 12. FEDERAL AID, LAWS AND REGULATIONS

This Agreement is funded in whole or in part by the United States Government and is subject to all Federal laws and regulations governing Federally-financed projects. The Artist acknowledges that Federal requirements may change and the changed requirements will apply to the project as required, unless the Federal Government determines otherwise.

#### **ATTACHMENT A**

#### FEES, MATERIALS AND PAYMENT SCHEDULE

1.	The Contract Amount totals \$	This	amount repi	resents all	costs, wh	ether
	identified as a specific item or not,					
	described in the Artist's proposal, and	l as such, it incl	udes any an	d all items	and in tern	ns as
	described in the NJSCA publication	"Procedures f	<u>or Artists in</u>	the Perfo	rmance of	Arts
	Inclusion Contracts" which are necess	sary to complete	e this project	and shall	be made p	art of
	this contract.					

- 2. As payment for the Artwork, including all fees, supplies, design and transportation of same, Contractor will pay the Artist the total sum of (\$000,000.00) as follows:\*\*\*
  - a. \$00,000.00 upon consultation, acceptance of proposal, establishment of project timeline, acceptance of a date for submission of Design Development Plans and signing of contract.
  - b. \$00,000.00 upon acceptance of Design Development Plans (including installation plan) for Artwork.
  - c. \$00,000.00 upon receipt and approval of all signed and sealed □shop□ drawings for the Artwork.
  - d. \$00,000.00 upon delivery of materials to the Artist, presentation of receipts or certifications of ownership for said materials, and start of construction.
  - e. \$00,000.00 upon 50% completion of the Artwork.
  - f. \$00,000.00 upon 100 installation of the Artwork.
  - g. \$00,000.00 upon 100% project completion including receipt of archival documentation, copyright application to NJSCA; acceptance of the Artwork by Contractor and NJ TRANSIT; submission of final invoice and close-out documents.

#### **Payment Notes**

Artist shall submit an invoice for each milestone to Contractor at the address identified under Article 34 of the contract.

Payment to the Artist shall be made not later than ten (10) calendar days after Terminal receives payment from NJ TRANSIT.

NJ TRANSIT is exempt from sales tax; please provide sales tax exemption certificates to supplier.)

# ATTACHMENT B

# **PROGRESS SCHEDULE**

15 Days After Signing of the Agreement (Design and Development Plans) to NJ TRANSIT for Approval	The Artist submits complete Plans and Drawings					
10 Work Days After Receipt of the Design and Development Plans	Design Development and Installation Plan approved by Contractor and NJ TRANSIT.					
10 Work Days After Receipt of materials.	The Artist begins creation of Artwork.					
To Be Determined	Meetings at the Artist's studio and/or fabricator to review and approve progress of Artwork. Pictural Documentation may be substituted					
To Be Determined	Artwork ready for delivery and installation.					
To Be Determined	Installation completed.					
30 Days After Installation of the Artwork	Final approval by Contractor and completion of Agreement. The Artist submits all required project close-out documents.					

#### ATTACHMENT C

#### SCOPE OF WORK AND BUDGET

#### 1. Purpose

The purpose of this Agreement is to provide the Artwork design, engineering design, construction documents, construction management, complete fabrication, complete transportation and complete installation for Artwork to be installed in the (NAME OF PROJECT).

#### 2. Project Budget

The total contract for the Artwork is \$000,000 including all costs.

#### 3. Schedule

The Artwork must be completed and ready for installation by MONTH, DAY, YEAR. Installation of the Artwork must be completed by MONTH, DAY, YEAR. All close-out submissions must be completed and submitted to NJ TRANSIT within 30 days after completion of installation.

#### 4. Site Location

The Artwork will be installed in the (LOCATION) of the (NAME OF PROJECT). The exact location(s) for the Artwork(s) shall be proposed by the Artist in the Design Development Plans and approved by the project architect in writing.

#### 5. Artwork Description

The Artist will design, fabricate, and deliver the following work of art:

a. An Artwork which shall consist of (DETAILED DESCRIPTION OF ARTWORK (S). The surface textures and color of the Artwork shall remain in accordance with the Artist's Design Development Plans as reviewed and accepted by Contractor and NJ TRANSIT.

The Artist shall execute and complete fabrication, transportation and installation of the Artwork upon consultation and coordination with Contractor. All dimensions for the Artwork shall be formally documented as a part of the Design Development Plans. Any structural plans shall be reviewed and approved by engineers who hold license in the State of New Jersey. The Artist will coordinate and produce pertinent plans, engineering and technical data and shall promptly submit such information in accordance with the project schedule as determined by Contractor.

The artist shall be responsible for the structural integrity and proper installation of the Artwork. The installation plans shall be submitted and approved by NJ TRANSIT prior to installation of the artwork. Close contact shall be maintained with NJ TRANSIT project management staff to properly schedule installation activities.

- b. Project Control Specifications: The following are control specifications which address specific issues of the ARTIST's proposal are hereby incorporated into this Agreement.
- 1. There shall be no reduction in the scale of the elements which comprise the Artwork.

- 2. The Artist must provide clean connection devices for the mounting of the Artwork to ensure access for potential service needs/ease of maintenance/or removal of same.
- 3. The elements comprising the Artwork, must be absolutely secure in their mounting methods as called for in engineering calculations sealed by a New Jersey licensed engineer.
- 4. The Artist shall control the overall weight and structural integrity of each of the elements of the Artwork ensuring that they will not exceed live load design criteria as called for in engineering calculations sealed by a New Jersey licensed engineer and approved by Contractor's project architect/engineer (NAME OF ENGINEER).
- 5. The Artist shall use paints which ensure maximum color fading protection for the Artwork from exposure to UV rays.
- 6. The Artist shall provide all lighting equipment and fixtures (brackets and plugs) necessary for installation.
- c. The Artist shall provide complete construction management services for the Artwork, which shall include, but not be limited to the following:
- 1. Schedule all work to be performed on the Artwork by both on and off site suppliers, fabricators or workers.
- 2. Provide all necessary engineering information and supervision for layout of the proper installation devices to insure that the eventual installation of the Artwork will properly interface with the structure.
- d. The Artist shall provide the following upon completion of the Artwork.
- 1. Install, a plaque containing the title of the Artwork, Artist's name, credits and other information as per NJSCA guidelines. The plaque shall be publicly displayed at a size to be mutually agreed upon by NJ TRANSIT, and the Artist.
- 2. Delivery of complete archival documentation of the finished Artwork to the **NJSCA**:
  - a. All original designs, original drawings, and original presentation maquette.
  - b. 35mm slides, color and black and white photographs
  - c. A written physical description and statement of intent.

Such documentation shall be placed in the archives of the NJSCA and is to be supplied by the Artist not later than 30 days after the installation of the Artwork. Materials submitted must comply with the NJSCA documentation format for visual artwork. All such materials will remain joint property of NJ TRANSIT and the NJSCA.

# ATTACHMENT D

# **CONTRACT DELIVERABLES**

The following is a listing of contract deliverables that are required at the completion of each phase of this project. The detailed requirements of each deliverable item are specified in the NJSCA publication entitled "Procedures for Artists," a copy of which is attached hereto as part of this attachment.

Ref. 1.1.0 1.3.0 1.4.0 1.7.0	Deliverable Art Design and Installation Plan Time Schedule Comprehensive Site Plan/Elevations, Details Site Inspection
Construction/Project Phase Ref. 2.1.0 2.2.0 2.3.0 2.4.0 2.5.0 2.6.0 2.7.0 2.8.0 2.9.0	Deliverable Schedule and Progress Reports Meetings Correspondence Pre-Construction/Project Meeting Review and Approve Project Schedule Coordinate with Project Management Coordinate and Maintain Documentation Certification of Insurance Site Inspection/Field Observation Reports
Project Close-Out Phase Ref. 3.1.0 3.2.0	Deliverable Responsibilities Commencement
3.3.0	Project Close-Out Requirements 3.3.1 Complete Photo 3.3.2 Project Plaque 3.3.3 Submission of Copyright Form VA 3.3.4 Maintenance Schedule 3.3.5 Publicity Agreement with NJSCA 3.3.6 Final Cost Analysis 3.3.7 Final Payment

**Design Development Phase** 

# ATTACHMENT D ARTIST PROPOSAL FOR ARTWORK

### STATE OF NEW JERSEY

# EEO PROVISIONS FOR CONSTRUCTION CONTRACTS

#### I. BID REQUIREMENTS

This contract is subject to the provisions of N.J.S.A. 10:2-1 through 10:2-4 and N.J.S.A. 10:5-31 <u>et seq.</u> (P.L. 1975, c.127), and in accordance with the rules and regulations promulgated pursuant thereto, the bidder agrees to comply with the following:

At the time the signed contract is returned to NJ TRANSIT, the said bidder (contractor) shall submit to NJ TRANSIT and the New Jersey Department of Treasury, Division of Public Contracts Equal Employment Opportunity Compliance an Initial Project Workforce Report (Form AA-201) consisting of forms provided by NJ TRANSIT and completed by the bidder (contractor) in accordance with Subchapter 7 of the regulations promulgated by the Treasurer pursuant to P.L. 1975, c.127.

# II. EXECUTIVE ORDER 151 MANDATORY EEO AND AFFIRMATIVE ACTION REQUIREMENTS

It is the policy of the New Jersey Transit Corporation that its contracts should create a workforce that reflects the diversity of the State of New Jersey. Therefore, contractors engaged by the New Jersey Transit Corporation to perform under a construction contract shall put forth a good faith effort to engage in recruitment and employment practices that further the goal of fostering equal opportunities to minorities and women.

The contractor must demonstrate to the New Jersey Transit Corporation's satisfaction that a good faith effort was made to ensure that minorities and women have been afforded equal opportunity to gain employment under the New Jersey Transit Corporation's contract with the contractor. Payment may be withheld from a contractor's contract for failure to comply with these provisions.

Evidence of a "good faith effort" includes, but is not limited to:

- 1. The Contractor shall recruit prospective employees through the State Job bank website, managed by the Department of Labor and Workforce Development, available online at http://NJ.gov/JobCentralNJ.
- 2. The Contractor shall keep specific records of its efforts, including records of all individuals interviewed and hired, including the specific numbers of minorities and women.
- 3. The Contractor shall actively solicit and shall provide the New Jersey Transit Corporation with proof of solicitations for employment, including but not limited to advertisements in general circulation media, professional service publications and electronic media.
- 4. The Contractor shall provide evidence of efforts described at 2 above to the New Jersey Transit Corporation no less frequently than once every 12 months.
- 5. The Contractor shall comply with the requirements set forth at N.J.A.C. 17:27.

# III. SUBCONTRACTS; EQUAL EMPLOYMENT GOALS

The contractor agrees to incorporate these State of New Jersey EEO Provisions for Construction Contracts in its construction subcontracts.

The New Jersey Department of Treasury, Division of Public Contracts Equal Employment Opportunity Compliance (hereinafter "Division") shall individually establish the targeted minority and women employment goals for determining good faith equal employment opportunity efforts by each construction

- 1 - **10/09** 

contractor and subcontractor for each trade on each contract. The Division shall review the trades to be utilized during the completion of the work as reported on the Initial Project Workforce Report and determine the targeted employment goals based upon the number of qualified minorities and women available as reported by the New Jersey Department of Labor, Division of Planning and Research in its report, EEO Tabulations—Detailed Occupations by Race/Hispanic Groups.

In accordance with N.J.A.C. 17:27, Contractors and subcontractors are required to make a good faith effort to provide equal employment opportunity for minorities and women. Failure to make good faith efforts to provide equal employment opportunity for minorities and women may result in sanctions including fines/penalties, withholding of payment, termination of the contract, suspension/debarment or such other action as provided by law.

The contractor shall obtain the targeted employment goals from the Division's website at <a href="https://www.state.nj.us/treasury/contract\_compliance">www.state.nj.us/treasury/contract\_compliance</a> or request the employment goals from the Division of Public Contracts Equal Employment Opportunity Compliance after submitting the initial project workforce report to the Division of Public Contracts Equal Employment Opportunity Compliance.

The contractor shall then be responsible for the completion and submittal of a Monthly Project Workforce Report (Form AA-202) for the duration of this contract.

(NOTE: FOR THE PURPOSE OF THIS CONTRACT THE "PUBLIC AGENCY COMPLIANCE OFFICER" REFERENCED BELOW IS NJ TRANSIT'S VICE PRESIDENT OF DIVERSITY PROGRAMS AND THE "PUBLIC AGENCY" IS NJ TRANSIT.)

#### IV. MANDATORY CONTRACT LANGUAGE

N.J.S.A. 10:5-31 et seq. and N.J.A.C. 17:27 MANDATORY AFFIRMATIVE ACTION LANGUAGE FOR CONSTRUCTION CONTRACTS

During the performance of this contract, the contractor agrees as follows:

The contractor or subcontractor, where applicable, will not discriminate against any employee or applicant for employment because of age, race, creed, color, national origin, ancestry, marital status, affectional or sexual orientation, gender identity or expression, disability, nationality or sex. Except with respect to affectional or sexual orientation and gender identity or expression, the contractor will ensure that equal employment opportunity is afforded to such applicants in recruitment and employment, and that employees are treated during employment, without regard to their age, race, creed, color, national origin, ancestry, marital status, affectional or sexual orientation, gender identity or expression, disability, nationality or sex. Such equal employment opportunity shall include, but not be limited to the following: employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided by the Public Agency Compliance Officer setting forth provisions of this nondiscrimination clause.

The contractor or subcontractor, where applicable will, in all solicitations or advertisements for employees placed by or on behalf of the contractor, state that all qualified applicants will receive consideration for employment without regard to age, race, creed, color, national origin, ancestry, marital status, affectional or sexual orientation, gender identity or expression, disability, nationality or sex.

The contractor or subcontractor, where applicable, will send to each labor union or representative of workers with which it has a collective bargaining agreement or other contract or understanding, a notice, to be provided by the agency contracting officer advising the labor union or workers' representative of the contractor's commitments under this act and shall post copies of the notice in conspicuous places available to employees and applicants for employment.

- 2 - **10/09** 

The contractor or subcontractor where applicable, agrees to comply with any regulations promulgated by the Treasurer pursuant to N.J.S.A. 10:5-31 et seq., as amended and supplemented from time to time and the Americans with Disabilities Act.

When hiring or scheduling workers in each construction trade, the contractor or subcontractor agrees to make good faith efforts to employ minority and women workers in each construction trade consistent with the targeted employment goal prescribed by N.J.A.C. 17:27-7.2; provided, however, that the Division may, in its discretion, exempt a contractor or subcontractor from compliance with the good faith procedures prescribed by the following provisions, A, B and C, as long as the Division is satisfied that the contractor or subcontractor is employing workers provided by a union which provides evidence, in accordance with standards prescribed by the Division, that its percentage of active "card carrying" members who are minority and women workers is equal to or greater than the targeted employment goal established in accordance with N.J.A.C. 17:27-7.2.

The contractor or subcontractor agrees that a good faith effort shall include compliance with the following procedures:

- (A). If the contractor or subcontractor has a referral agreement or arrangement with a union for a construction trade, the contractor or subcontractor shall, within three business days of the contract award, seek assurances from the union that it will cooperate with the contractor or subcontractor as it fulfills its affirmative action obligations under this contract and in accordance with the rules promulgated by the Treasurer pursuant to N.J.S.A. 10:5-31 et seq., as supplemented and amended from time to time, and the Americans with Disabilities Act. If the contractor or subcontractor is unable to obtain said assurances from the construction trade union at least five business days prior to the commencement of construction work. the contractor or subcontractor agrees to afford equal employment opportunities to minority and women workers directly, consistent with this chapter. If the contractor's or subcontractor's prior experience with a construction trade union, regardless of whether the union has provided said assurances, indicates a significant possibility that the trade union will not refer sufficient minority and women workers consistent with affording equal employment opportunities as specified in this chapter, the contractor or subcontractor agrees to be prepared to provide such opportunities to minority and women workers directly, consistent with this chapter, by complying with the procedures prescribed under (B) below; and the contractor or subcontractor further agrees to take said action immediately if it determines or is so notified by the Division that the union is not referring minority and women workers consistent with the equal employment opportunity goals set forth in this chapter.
- (B). If good faith efforts to meet targeted employment goals have not or cannot be met for each construction trade by adhering to the procedures of (A) above, or if the contractor does not have a referral agreement or arrangement with a union for a construction trade, the contractor or subcontractor agrees to take the following actions:
- (1). To notify the public agency compliance officer, the Division, and minority and women referral organizations listed by the Division pursuant to N.J.A.C. 17:27-5.3, of its workforce needs, and request referral of minority and women workers;
- (2). To notify any minority and women workers who have been listed with it as awaiting available vacancies;
- (3). Prior to commencement of work, to request that the local construction trade union refer minority and women workers to fill job openings, provided the contractor or subcontractor has a referral agreement or arrangement with a union for the construction trade;
- (4). To leave standing requests for additional referral to minority and women workers with the local construction trade union, provided the contractor or subcontractor has a referral agreement or arrangement with a union for the construction trade, the State Training and Employment Service and other approved referral sources in the area;

- 3 - **10/09** 

- (5). If it is necessary to lay off some of the workers in a given trade on the construction site, layoffs shall be conducted in compliance with the equal employment opportunity and non-discrimination standards set forth in this regulation, as well as with applicable Federal and State court decisions;
- (6). To adhere to the following procedure when minority and women workers apply or are referred to the contractor or subcontractor:
- (i) If said individuals have never previously received any document or certification signifying a level of qualification lower than that required in order to perform the work of the construction trade, the contractor or subcontractor shall in good faith determine the qualifications of such individuals. The contractor or subcontractor shall hire or schedule those individuals who satisfy appropriate qualification standards in conformity with the equal employment opportunity and non-discrimination principles set forth in this chapter. However, a contractor or subcontractor shall determine that the individual at least possesses the requisite skills, and experience recognized by a union, apprentice program or a referral agency, provided the referral agency is acceptable to the Division. If necessary, the contractor or subcontractor shall hire or schedule minority and women workers who qualify as trainees pursuant to these rules. All of the requirements, however, are limited by the provisions of (C) below.
- (ii). The name of any interested women or minority individual shall be maintained on a waiting list, and shall be considered for employment as described in paragraph (i) above, whenever vacancies occur. At the request of the Division, the contractor or subcontractor shall provide evidence of its good faith efforts to employ women and minorities from the list to fill vacancies.
- (iii). If, for any reason, said contractor or subcontractor determines that a minority individual or a woman is not qualified or if the individual qualifies as an advanced trainee or apprentice, the contractor or subcontractor shall inform the individual in writing of the reasons for the determination, maintain a copy of the determination in its files, and send a copy to the public agency compliance officer and to the Division.
- (7). To keep a complete and accurate record of all requests made for the referral of workers in any trade covered by the contract, on forms made available by the Division and submitted promptly to the Division upon request.
- (C). The contractor or subcontractor agrees that nothing contained in (B) above shall preclude the contractor or subcontractor from complying with the union hiring hall or apprenticeship policies in any applicable collective bargaining agreement or union hiring hall arrangement, and, where required by custom or agreement, it shall send journeymen and trainees to the union for referral, or to the apprenticeship program for admission, pursuant to such agreement or arrangement. However, where the practices of a union or apprenticeship program will result in the exclusion of minorities and women or the failure to refer minorities and women consistent with the targeted county employment goal, the contractor or subcontractor shall consider for employment persons referred pursuant to (B) above without regard to such agreement or arrangement; provided further, however, that the contractor or subcontractor shall not be required to employ women and minority advanced trainees and trainees in numbers which result in the employment of advanced trainees and trainees as a percentage of the total workforce for the construction trade, which percentage significantly exceeds the apprentice to journey worker ratio specified in the applicable collective bargaining agreement, or in the absence of a collective bargaining agreement, exceeds the ratio established by practice in the area for said construction trade. Also, the contractor or subcontractor agrees that, in implementing the procedures of (B) above it shall, where applicable, employ minority and women workers residing within the geographical jurisdiction of the union.

After notification of award, but prior to signing a construction contract, the contractor shall submit to the public agency compliance officer and the Division an initial project workforce report (Form AA 201) provided to the public agency by the Division for distribution to and completion by the contractor, in accordance with N.J.A.C. 17:27-7. The contractor also agrees to submit a copy of the Monthly Project Workforce Report once a month thereafter for the duration of this contract to the Division and to the public agency compliance officer.

- 4 - **10/09** 

The contractor agrees to cooperate with the public agency in the payment of budgeted funds, as is necessary, for on-the-job and/or off-the-job programs for outreach and training of minorities and women.

(D). The contractor and its subcontractors shall furnish such reports or other documents to the Division of Public Contracts Equal Employment Opportunity Compliance as may be requested by the Division from time to time in order to carry out the purposes of these regulations, and public agencies shall furnish such information as may be requested by the Division of Public Contracts Equal Employment Opportunity Compliance for conducting a compliance investigation pursuant to Subchapter 10 of the Administrative Code at N.J.A.C. 17:27.

- 5 - **10/09** 

# NEW JERSEY TRANSIT CORPORATION DBE REQUIREMENTS FOR RACE-CONSCIOUS

# FEDERAL PROCUREMENT ACTIVITIES

# **TABLE OF CONTENTS**

TITLE	OF ARTICLE		PAGE
1.1	POLICY 2		
1.2	ASSURANCE	2	
1.3	DBE GOAL FOR THIS PROJECT	3	
1.4	GUIDANCE TO BIDDER/PROPOSERS/PRIMES	. 3	
1.5	TRANSIT VEHICLE MANUFACTURERS (TVM)		3-4
1.6	RESPONSIBLE BID/PROPOSAL CRITERIA	4	
2.0	GUIDANCE ON A GOOD FAITH	4	
2.2	A GOOD FAITH EFFORT	5	
2.3	REQUIRED FORMS	5-6	
2.4	SUPPLEMENTAL REQUIRED FORMS (IF APPLICABLE)		7
2.5	INSTRUCTIONS FOR COMPLETING REQUIRED FORMS		7-9
2.6	INSTRUCTIONS FOR COMPLETING SUPPLEMENTAL REQUIRED FORMS		9-10
3.0	GUIDANCE ON COUNTING DBE PARTICIPATION		11
3.2	COUNTING DBE PARTICIPATION	11-12	
3.3	DBE PRIME CONTRACTOR GUIDANCE	.12	
3.4	DBE TRUCKING FIRM GUIDANCE	. 12	
3.5	LEASING TRUCKS	12-13	
3.6	COUNTING MATERIALS AND SUPPLIES	•••	13
4.0	TERMINATION OF DBE(s)	13	
4.2	ADDITION OF DBE(s)	14	
4.3	REPLACEMENT OR REMOVAL OF DBE(s)		14-15
4.4	WITHDRAWN DBE(s)	15	
4.5	DECERTIFIED DBE(s)	15-16	
5.0	AWARD OBLIGATIONS	16	
5.2	POST AWARD DELIVERABLES	16-17	
5.3	PROMPT PAYMENT OF INVOICES TO SUBCONTRACTORS		17-18
5.4	SUBCONTRACT PAYMENT DISPUTE RESOLUTION		18
5.5	PROMPT PAYMENT OF SUBCONTRACTOR RETAINAGE		18-19
5.6	REMEDIES AND PENALITES	19	
5.7	LIQUIDATED DAMAGES	19-20	
APPENI	DIX 1 (GLOSSARY)	21-22	

NEW JERSEY TRANSIT CORPORATION

DBE REQUIREMENTS FOR

**RACE-CONSCIOUS** 

FEDERAL PROCUREMENT ACTIVITIES

The following pages provide Bidders/Proposers/Primes on federal contracts with New Jersey Transit (NJT), information about NJT's Disadvantaged Business Enterprise (DBE) Program, administered by the Office of Business Development (OBD). Prospective Bidders/Proposers/Primes will have an opportunity to ask questions regarding the directives contained in the DBE specifications at the pre-bid/pre-proposal conference(s). Further clarification of the DBE specifications, along with assistance in completing the forms, can be obtained by calling

A list of certified DBE firms may be found in the NJ Unified Certification Program (NJUCP) Directory at <a href="https://www.njucp.net">www.njucp.net</a>. Note: Use of this list does not relieve the Bidder/Proposer/Prime contractor/consultant of responsibility to seek DBE participation from other sources. The list is updated daily and must be checked periodically, as firms are certified and decertified daily.

These DBE specifications are a part of the Contract and shall be binding upon the successful Bidder/Proposer and Prime in the pre and post-award stages of NJT professional services, construction, and goods and services contracts. These specifications shall be binding upon sub-recipients and imposed on their contractors.

1.1 POLICY

(973) 491-7593.

As defined in the U.S. Department of Transportation (USDOT) Regulation 49 CFR Part 26, it is the policy of NJT that Disadvantaged Business Enterprises shall have the opportunity to compete for and participate in the performance of contracts financed in whole or in part with federal funds. Each subcontract a Prime signs with a subcontractor/subconsultant must include the following assurance referenced in **article 1.2.** 

1.2 ASSURANCE

1.2.1 The Prime contractor/consultant, or subcontractor/subconsultant shall not discriminate on the basis of race, color, national origin, or sex in the performance of this contract. The Prime contractor/consultant shall carry out applicable requirements of 49 CFR Part 26 in the award and administration of DOT-assisted contracts. Failure by the contractor/consultant to carry out these requirements is a material breach of this contract, which may result in the termination of this contract or such other remedy, as the

recipient deems appropriate (49 CFR 26.13).

1.2.2 This language is included to comply with relevant Federal law and to ensure that all persons who enter into any direct or indirect form of contractual agreement with NJT are aware of their responsibilities and the commitment of NJT to see that NJT's DBE Policy is carried out in all instances.

# 1.3 DBE GOAL FOR THIS PROJECT

As an aid in meeting the commitment of its DBE Program, NJT is setting a *race conscious* goal of awarding \_\_\_\_\_\_ percent of the gross sum bid/proposal to certified NJUCP DBE firms. Should the actual contract amount increase or decrease, through approved change order(s), the assigned goal may remain. The OBD will determine if the change orders and/or contract phases will result in an adjustment to the DBE participation goal.

# 1.4 GUIDANCE TO BIDDER/PROPOSERS/PRIMES

- 1.4.1 Failure by a Bidder/Proposer/Prime to comply with any of the requirements contained herein shall result in breach of contract and it shall be subject to the appropriate penalties, remedies, or liquidated damage(s).

  Refer to articles 5.6-5.7
- 1.4.2 All required forms, including the supplemental section (see articles 2.3-2.4) must be submitted in accordance with the requirements. Firms must be certified under the NJUCP at the time of contract award in order to obtain DBE credit toward the goal.
- 1.4.3 Price alone is not an acceptable basis for rejecting a DBE subcontractor/subconsultant' s bid.
- 1.4.4 The Bidder/Proposer/Prime shall, at a minimum, seek DBEs in the same geographic area in which it generally seeks subcontractors/subconsultants. However, the Bidder/Proposer/Prime may be required to expand its search under specific circumstances as determined by OBD. **Refer to article 2.0**
- 1.4.5 Agreements between a Bidder/Proposer/Prime and a DBE in which the DBE promises not to provide subcontracting quotations to other Bidder/Proposers are prohibited.
- 1.4.6 The desire of a Bidder/Proposer/Prime to self-perform the work of a contract with its own organization is not an acceptable basis to not meet the goal or demonstrate a good faith effort to do so.
- 1.4.7 The Bidder/Proposer/Prime is responsible for verifying that the DBE is certified under the appropriate NAICS code for the scope of work identified. DBE credit shall be given only for work performed in the NAICS code(s) under which the DBE is certified.
- 1.4.8 A DBE firm listed on the First-Tier DBE Utilization Form (Form A) shall constitute a binding representation to NJT, by the Bidder/Proposer/Prime, that the DBE firm is qualified, available, and certified under the appropriate and required NAICS code to perform the scope of work identified. **Refer to article 2.5a**

# 1.5 TRANSIT VEHICLE MANUFACTURERS (TVM)

- 1.5.1 As a transit vehicle manufacturer, you must establish and submit for FTA's approval an annual overall DBE percentage goal. A TVM must certify that it submitted the annual DBE goal required by 49 CFR and FTA has approved it or not disapproved it.
- (a) As a condition of being authorized to bid or propose on FTA assisted transit vehicle procurements, the Bidder/Proposer must complete and submit the TVM Certification form with the bid/proposal certifying that it has complied with the requirements of 49 CFR 26.49.
- 1.5.2 NJT may, with FTA approval, establish project-specific goals for DBE participation in the procurement of transit vehicles in lieu of complying with the procedures of this section.

### 1.6 RESPONSIBLE BID/PROPOSAL CRITERIA

- 1.6.1 As a matter of responsibility, the two lowest Bidders or two highest ranked Proposers must submit the required forms, including the supplemental section (if applicable), with the bid/proposal or within seven (7) days after the bid opening or proposal due date. *NJT may grant a formal written request to extend this 7-day requirement at its sole discretion on a case-by-case basis.*
- 1.6.2 Failure to satisfactorily complete or submit all required forms when due may result in determination by NJT that the Bidder/Proposer is non-responsible and may cause rejection of the bid or proposal.
- 1.6.3 If the two lowest Bidders/highest ranked Proposers submit the DBE forms, but fail to meet the DBE goal, the OBD will consider the efforts made to determine if a Bidder/Proposer/Prime has in fact, demonstrated a good faith effort. See article 2.0
- 1.6.4 If it is determined that efforts were made to include DBE participation on the contract, however these efforts did not result in meeting the goal, NJT may request that additional efforts be made within 10 business days of the request. If at this time the Bidder/Proposer fails to demonstrate a good faith effort to achieve the goal, NJT shall consider awarding the contract to the next lowest bidder or highest ranked proposer who offers a reasonable price and meets the DBE goal or demonstrates a good faith effort and other bid requirements or requirements of 49 CFR Part 26.

### 2. GUIDANCE ON A GOOD FAITH EFFORT

- 2.1 To demonstrate a good faith effort to meet the DBE goal, a Bidder/Proposer/Prime shall provide written documentation in addition to Form D (article 2.3e), of the steps it has taken, prior to the bid opening/proposal due date, or during the life of the contract to obtain DBE participation. <a href="https://document.nih.google.com/">The Bidder/Proposer/Prime can meet this requirement in either of two ways:</a>
  - (1) The Bidder/ Proposer/Prime can meet the goal.
  - (2) The Bidder/Proposer/Prime shall exhaust the available options referenced in **article 2.2** in making a continuous good faith effort to meet the assigned contract goal for the life of the contract.
- (a) The efforts employed by the bidder should be those that one could reasonably expect a bidder to take if the bidder were actively and aggressively trying to obtain DBE participation sufficient to meet the DBE Requirements for Federal Procurement Activities

contract goal.

- (b) In determining a good faith effort, the OBD will consider the <u>quality</u>, <u>quantity</u>, and <u>intensity</u> of the different kinds of efforts that the Bidder/Proposer/Prime has made. Mere *pro forma* efforts will not be considered as demonstration of good faith effort to meet the DBE contract requirements.
- (c) The Bidder/Proposer/Prime shall use good business judgment and consider a number of factors in negotiating with subcontractors/subconsultants, including DBE subcontractors/ subconsultants, and should take a firm's price and capabilities as well as contract goals into consideration. The fact that there may be some additional costs involved in finding and using DBEs is not in itself sufficient reason for a Bidder' s/Proposer's failure to meet the contract DBE goal, as long as such costs are reasonable as determined by NJT. Primes are not, however, required to accept higher quotes from DBEs if the price difference is excessive or unreasonable.
- (d) The Bidder/Proposer/Prime's ability or desire to perform the work of a contract with its own organization (self-performance) does not relieve the Bidder/Proposer/Prime of the responsibility to meet the goal or demonstrate a good faith effort.
- (e) The Bidder/Proposer/Prime shall <u>not reject</u> DBEs as being unqualified without sound reasons based on a thorough investigation of their capabilities. The Bidder/Proposer/Prime's standing within its industry, membership in specific groups, organizations, or associations and political or social affiliations (for example, union vs. non-union employee status) are not legitimate causes for the rejection or non-solicitation of bids in the Bidder/Proposer/Prime's efforts to meet the assigned project goal.
- (f) The OBD will support the Bidder/Proposer/Prime in indentifying ways to meet the assigned contract goal.

#### 2.2 A GOOD FAITH EFFORT

The following is a list of actions that NJT will consider as evidence of a Bidder/Proposer/Prime's good faith effort to obtain DBE participation. While exhausting the available options in this list may count as a good faith effort, this list is not intended to be a mandatory checklist, nor is this list intended to be exclusive or exhaustive of all the efforts a Bidder/Proposer/Prime might make to achieve the assigned DBE goal. NJT may require a Bidder/Proposer/Prime to take action above and beyond those listed below to meet the assigned DBE goal.

- (a) The Bidder/Proposer/Prime shall solicit through all reasonable and available means (e.g. attendance at pre-bid meetings, advertising and/or written notices) the interest of all certified DBEs who have the capacity to perform the work of the contract.
  - (1) The Bidder/Proposer/Prime must solicit this interest within sufficient time to allow the DBE to respond to the solicitation.
  - (2) The Bidder/Proposer/Prime must take appropriate steps to follow up on initial solicitations in order to determine with certainty if the DBE firms are interested.
- (b) The Bidder/Proposer/Prime shall select portions of the work to be performed by DBEs in order to increase the likelihood that the DBE goals will be achieved. This includes, where appropriate, breaking out

- contract work items into economically feasible units to facilitate DBE participation, even when the Prime might otherwise prefer to perform these work items with its own forces.
- (c) The Bidder/Proposer/Prime shall provide interested DBEs with adequate information about the plans, specifications, and requirements of the contract in a timely manner to assist them in responding to a solicitation.
- (d) The Bidder/Proposer/Prime shall negotiate with a DBE(s) with the intent to enter into a contract. It is the Bidder/Proposer's responsibility to make a portion of the work available to DBE subcontractors/subconsultants and suppliers and to select those portions of the work or material needs consistent with the available DBE Primes and suppliers, so as to facilitate DBE participation.
  - (1) Evidence of such negotiation includes: the names, addresses, and telephone numbers of DBEs that were considered; a description of the information provided regarding the plans and specifications for the work selected for subcontracting; and evidence as to why additional agreements could not be reached for DBEs to perform the work.
- **(e)** The Bidder/Proposer/Prime shall make efforts to assist interested DBEs in obtaining bonding, lines of credit, or insurance as required by NJT or the Prime contractor.
- (f) The Bidder/Proposer/Prime shall make efforts to assist interested DBEs in obtaining necessary equipment, supplies, materials, or related assistance or services.
- (g) The Bidder/Proposer/Prime shall effectively use the services of available minority/women community organizations; minority/women Prime contractors groups; local, State and Federal minority/women business assistance offices; and other organizations as allowed on a case-by-case basis to provide assistance in the recruitment and placement of DBEs.

### 2.3 REQUIRED FORMS

- (a) Form A First Tier DBE Utilization: Lists all First Tier DBE firms scheduled to participate on this contract.
- **Form A1 Bidder/Proposer Solicitation and Contractor Information:** Lists all DBE and Non-DBE sub contractor/subconsultants, including suppliers, solicited for, **and** participating on this contract.
- (c) <u>Form A2- Non-DBE Sub Utilization</u>: Lists all **DBE and Non-DBE** subcontractors/ subconsultants, including suppliers, participating on this contract.
- (d) <u>Form B Intent to Perform as a DBE Sub</u>: Identifies the work the 1<sup>st</sup> Tier DBE intends to perform including scope of work, subcontract dollar value, etc.
- (e) <u>DBE Good Faith Effort Form</u> (<u>if applicable</u>): Identifies any DBE subcontractor invited to quote, but declined to do so for any reason.
- (f) <u>Trucking Commitment Agreement (if applicable):</u> Identifies all trucking firms (DBE and Non-DBE) participating on this contract, at any tier.
- (g) NJ UCP DBE Certification & NAICS Code Verification: Confirms the DBE status and NAICS code(s) of

each First Tier DBE subcontractor/subconsultant.

- (h) \* Form E Contractor's Monthly DBE Payment Report & Payment Certification Voucher (Post-Award): Records monthly payments <u>issued</u> to each DBE subcontractor/subconsultant/supplier and monthly payments issued by NJ TRANSIT to the Prime. Certifies that DBE subs have been paid for previous month's invoices.
- (i) <u>Form E2 DBE's Monthly Payment Report (Post-Award)</u>: Records monthly invoices submitted by the DBE, payments <u>owed</u> to the DBE on past due invoices and payments <u>received</u> from the prime by each DBE subcontractor/subconsultant.
- (j) <u>Form E1- DBE Prime's Monthly Payment Report (For DBE Prime Only):</u> Records monthly payments <u>issued</u> to each DBE Prime by NJ TRANSIT to.

# 2.4 SUPPLEMENTAL REQUIRED FORMS (IF APPLICABLE)

- (a) <u>Form AA Second Tier DBE Utilization:</u> Lists all Second Tier DBE firms scheduled to participate on the DBE sub-Prime's contract.
- (b) <u>Form AA1 Second Tier Bidder/Proposer Solicitation and Contractor Information</u>: Lists all Second Tier DBE firms participating on this contract as indicated on Form AA and Form AA2.
- (c) <u>Form AA2- Second Tier Non-DBE Sub Utilization:</u> Lists all **DBE and Non-DBE firms** including suppliers participating on the DBE sub-Prime's contract.
- (d) <u>Form BB Intent to Perform as a Second Tier DBE Sub</u>: Identifies the work the 2<sup>nd</sup> Tier DBE intends to perform including scope of work, subcontract dollar value, etc.
- (e) <u>NJ UCP DBE Certification & NAICS Code Verification:</u> Confirms the DBE status and NAICS code(s) of each Second Tier DBE subcontractor/subconsultant.

### 2.5 INSTRUCTIONS FOR COMPLETING REQUIRED FORMS (see glossary for definition of terms)

# (a) Form A - First Tier DBE Utilization:

Form A is a formal agreement between the Bidder/Proposer and the DBE(s). Replacement/removal of DBE subcontractors/subconsultants/supplier identified on Form A is prohibited after the bid or proposal is submitted to NJT. **Refer to article 4.3**. A DBE Bidder/Proposer, which lists itself on Form A, is committed to performing the work indicated with its own personnel.

DBEs performing as second tier sub(s) to a **non-DBE sub Prime** should be listed with the name of the non-DBE sub Prime's firm name in parenthesis next to the DBE sub's name. { *Ex: DBE Electric Co. (Prime Contractor, Inc.)*}

<sup>\*</sup> This form is due from the Prime in each month following the notice to proceed issued by NJ TRANSIT. **Refer to**article 5.2.4

- (1) A first Tier DBE is required to perform at least 51% of its subcontract value with its own forces. Bidders/Proposers/Primes will not receive any credit for DBEs performing less than 51% and therefore must not be listed on this form.
- (2) For DBE suppliers, identify all manufacturers, regular dealers, and brokers. If a DBE supplier is a *manufacturer*, indicate the full value of its subcontract. If a DBE supplier is a *regular dealer*, show its total contract value multiplied by 60% (Ex. \$100K x 60%= \$60K). If a DBE supplier is *neither a manufacturer nor a dealer*, indicate the <u>fee/commission only</u>, *not the cost of materials* or supplies. See article 3.0 for direction on determining credit toward the goal.
- (3) A detailed scope of work must be provided; <u>one-word descriptions are not acceptable</u>. (Ex. Haul and dispose of approximately 192,000 tons of contaminated soil to a clean earth facility at \$34.00 a ton).

# (b) Form A1 - Bidder/Proposer/Prime Solicitation and Contractor Information:

The Bidder/Proposer must complete and submit **page one** (1). The DBE and non-DBE subcontractors/subconsultants, including suppliers, solicited for, participating on, or expressed interest in this contract must complete **page two** (2).

# (c) Form A2- Non-DBE Sub Utilization:

Bidders/Proposers are required to report and submit all dollars committed to non-DBE subcontractors/subconsultants/suppliers. The non-DBE portion of work is <u>not</u> counted toward the assigned DBE goal. **See article 3.0** 

A detailed scope of work must be provided; one-word descriptions are not acceptable.

(Ex. Haul and dispose of approximately 192,000 tons of contaminated soil to a clean earth facility at \$34.00 a ton).

# (d) Form B - Intent to Perform as a First Tier DBE Sub:

Each DBE subcontractor/subconsultant/supplier listed on Form A must complete and sign Form B. **Note:**The Bidder/Proposer/Prime is prohibited from completing any portion of the form and from directing DBE(s) to sign a blank form.

- (1) The Bidder/Proposer/Prime must provide interested DBEs with a copy of appropriate plans, specifications, and requirements of the contract in a timely manner to allow the DBE to prepare an appropriate price quote and submit on time.
- (2) First Tier DBEs must perform at least 51% of the total dollar value of its subcontract, with its own forces. The firm must indicate the percentage of the total portion of work to be subcontracted to DBE and non-DBE firms. The <u>non-DBE</u> percentage of work is <u>not counted</u> toward the assigned goal.

- (3) The OBD encourages DBE-to-DBE subcontracting in order to preserve DBE participation credit. See article 3.0
- (4) The DBE must provide a detailed scope of work; one-word descriptions are not acceptable. Descriptions should include: type of services provided, total number of units, price per unit, total cost, etc.

# (e) <u>DBE Good Faith Effort</u>: (If Applicable)

Form D applies to any Bidder/Proposer/Prime who failed to meet the assigned DBE goal. This form will assist the Bidder/Proposer/Prime in demonstrating a good faith effort.

If the DBE(s) solicited declines to sign this form, the completed form should be submitted with the Bidder/Proposer's signature only and the OBD will verify the information provided with the firm. **Refer to articles 2.0-2.2** for guidance.

# (f) Trucking Commitment Agreement: (If Applicable)

DBEs must provide information for all DBE and non-DBE trucking firms it will lease from or subcontract to. Subcontracting to a non-DBE trucker means that the non-DBE will perform a portion of the DBE firm's subcontract. **Refer to article 3.4** 

The following documents must be attached for all trucks owned: copy of title(s)/finance agreement(s), registration card(s), insurance card(s), apportioned cab card(s) and/or hazardous material license(s) if applicable. A copy of the title or finance agreement is the only acceptable proof of ownership.

The following documents must be attached for all trucks leased: copy of lease agreement(s) established between both firms, title(s), registration card(s), insurance card(s), lease agreement(s), apportioned cab card(s) and/or hazardous material license(s) if applicable.

# (g) NJ UCP DBE Certification and NAICS Code Verification:

All DBEs listed on Form A must be certified at the time of contract award. It is the Bidder/
Proposer's responsibility to ensure that DBEs are certified and that their NAICS code(s) match the scope of work to be performed on this contract. Credit will not be given for any work to be performed without the appropriate NAICS code. Status can be verified through <a href="https://www.njucp.net">www.njucp.net</a> and <a href="https://wwww.njucp.net">www.njucp.net</a> and <a href="h

# (h) Form E - Contractor's Monthly DBE Payment Report & Payment Certification Voucher:

Beginning the month following the contract's notice to proceed, the Prime must report monthly payment activity for each DBE subcontractor/subconsultant/supplier; certifies each DBE sub has been paid any amounts due from previous or current progress payments paid to the Prime. (article 5.2.4)

All invoices 30 days past due from NJT must be listed in the appropriate field.

This report is due even if there is no payment activity. This form must be completed and submitted to the OBD by the 7<sup>th</sup> of each month to the attention of the OBD's Manager of Contract Compliance.

# (i) Form E1- DBE Prime's Monthly Payment Report (For DBE Prime Only)

Beginning the subsequent month following the contract's execution date, the DBE Prime must report its monthly payments received by NJT. **Refer to article 5.2.5**.

All invoices 30 days past due from NJT must be listed in the appropriate field.

This report is due even if there is no payment activity. This form must be completed and submitted to the OBD by the 7<sup>th</sup> of each month to the attention of the OBD's Manager of Contract Compliance.

# (j) Form E2 – DBE's Monthly Payment Report:

The Prime must provide a copy of the Form E2 to each DBE subcontractor/subconsultant/ supplier(s). Beginning the subsequent month following the DBE's execution date, the DBE firm must report its monthly payment activity.

This report is due even if there is no payment activity. <u>This form must be completed and submitted by</u> the DBE only to the OBD by the 7<sup>th</sup> of each month to the attention of the OBD's Manager of Contract Compliance.

All invoices 30 days past due must be listed in the appropriate field. Identify concerns or issues in the comments section to be addressed by the OBD. (Refer to article 5.2.6)

#### 2.6 INSTRUCTIONS FOR COMPLETING SUPPLEMENTAL REQUIRED FORMS:

(a) <u>Form AA – Second Tier DBE Utilization:</u> The Second Tier DBE must perform 100% of its subcontract with its own forces. A formal request to waive this requirement may be granted, solely at the discretion of the OBD; however, approval is required.

A detailed scope of work must be provided; one-word descriptions are not acceptable.

(Ex. Haul and dispose of approximately 192,000 tons of contaminated soil to a clean earth facility at \$34.00 a ton).

### (b) Form AA1 – Second Tier Bidder/Proposer Solicitation and Contractor Information:

The DBE sub-Prime must submit and complete **page one** (1). Second Tier DBE(s) solicited for and participating on this contract must complete **page two** (2).

# (c) Form AA2- Second Tier Non-DBE Subcontractor Utilization:

DBE sub-Primes are required to report and submit all dollars committed to non-DBEs. The non-DBE portion of work is not counted toward DBE participation credit. **Refer to article 3.0** 

A detailed scope of work must be provided; one-word descriptions are not acceptable.

(Ex. Haul and dispose of approximately 192,000 tons of contaminated soil to a clean earth facility at \$34.00 a ton).

# (d) Form BB - Intent to Perform as a Second Tier DBE Subcontractor:

Each DBE firm listed on Form AA, must complete, and sign. Only Second Tier DBE(s) must complete and sign this form.

The Second Tier DBE must provide a detailed scope of work; one-word descriptions are not acceptable. Descriptions should include: type of services provided, total number of units, price per unit, total cost, etc

# (e) NJ UCP DBE Certification & NAICS Code Verification:

All DBEs listed on Form AA must be certified at the time of contract award. It is the Bidder/Proposer/Prime's responsibility to ensure that DBEs are certified and that their NAICS code(s) match the scope of work to be performed on this contract. Credit will <u>not</u> be given for any work to be performed without the appropriate NAICS code. Status can be verified through <u>www.njcup.net</u> and www.census.gov/eos/www/naics/.

### 3.0 GUIDANCE ON COUNTING DBE PARTICIPATION

- 3.1 If a firm is not currently certified as a DBE in accordance with 49 CFR part 26 at the time of the execution of the contract, the firm's participation will not count toward the DBE goal.
- 3.1.1 A DBE performing less than 51% of its subcontract will not count toward the assigned goal and should not be listed on any forms.
- 3.1.2 When a DBE subcontracts part of the work of its contract to another firm, the value of the subcontracted work may be counted toward DBE goals only if the DBEs subcontractor/subconsultant is a DBE.
- (a) Work that a DBE subcontracts to a <u>non-DBE firm does not count</u> toward the DBE contract goal.
- (b) When a DBE performs as a participant in a **joint venture with a Non-DBE**, count the portion of the total dollar value of the contract equal to the distinct, clearly defined portion of the work of the contract that the DBE performs with its own forces toward DBE goals.
- (c) A DBE performs a *commercially useful function* when it is responsible for execution of the work of the contract and is carrying out its responsibilities by actually performing, managing and supervising the work involved.

A DBE does not perform a *commercially useful function* if its role is limited to that of an extra participant in a transaction, contract, or project through which funds are passed in order to obtain the appearance of DBE participation.

# 3.2 COUNTING DBE PARTICIPATION

3.2.1 When a DBE participates in a contract, only the value of the work actually performed by the DBE is counted toward DBE goals.

- (a) The entire amount of that portion of a contract that is performed by the DBE's own forces is counted. This includes the cost of supplies and materials obtained by the DBE for the work of the contract, as well as supplies purchased or equipment leased by the DBE (except supplies and equipment the DBE subcontractor/subconsultant purchases or leases from the Prime contractor or its affiliate).
- 3.2.2 The entire amount of fees or commissions charged by a DBE firm for providing a bona fide service, such as professional, technical, consultant, or managerial services, or for providing bonds or insurance specifically required for the performance of a contract, is counted toward DBE goals, provided the fee is reasonable and not excessive as compared with fees customarily allowed for similar services by a DBE.

### 3.3 DBE PRIME CONTRACTOR GUIDANCE

- 3.3.1 If a <u>DBE Prime</u>, expenditures are counted toward DBE goals only if the DBE is performing a commercially useful function on that contract.
- 3.3.2 A DBE Prime must perform or be responsible at least 30% of the total cost of its contract with its own workforce.
- 3.3.3 If a DBE Prime does not perform or exercise responsibility for at least 30% of the total cost of its contract with its own workforce or subcontracts a greater portion of the work of a contract than would be expected on the basis of normal industry practice for the type of work involved, NJT will consider that it is not performing a commercially useful function and the DBE Prime shall be in breach of the contract and subject to the appropriate remedies and penalties. Refer to Articles 5.6-5.7

#### 3.4 DBE TRUCKING FIRMS GUIDANCE

- 3.4.1 A DBE trucking firm is performing a commercially useful function if:
- (a) The DBE is responsible for the management and supervision of the entire trucking operation for which it is responsible on a particular contract, and there is not a contrived arrangement for the purpose of meeting DBE goals.
- (b) The DBE itself <u>owns and operates</u> <u>at least one</u> fully licensed, insured, and operational truck <u>to be used on the contract</u>.
- 3.4.2 The DBE receives credit for the total value of the transportation services it provides on the contract using trucks it owns, insures, and operates using drivers it employs.

# 3.5 LEASING TRUCKS

- 3.5.1 Leased trucks must display the name and identification number of the DBE.
- 3.5.2 The DBE may lease trucks from another DBE firm, including an owner-operator that is certified as a DBE.
  The DBE who leases trucks from another DBE receives credit for the total value of the transportation services the lessee DBE provides on the contract.
- 3.5.3 The DBE may also lease trucks from a non-DBE firm, including an owner-operator.

- (a) The DBE who leases trucks from a non-DBE <u>is entitled to credit only for the fee or commission it receives</u>
  <u>as a result of the lease arrangement.</u>
- 3.5.4 For the purposes of this section (Leasing), a lease must indicate that the DBE has exclusive use of and control over the truck.
- (a) This does not preclude the leased truck from working for others during the term of the lease with the consent of the DBE, so long as the lease gives the DBE absolute priority for use of the leased truck.

### 3.6 COUNTING MATERIALS AND SUPPLIES

- 3.6.1 Expenditures with DBEs for materials or supplies are counted toward DBE goals as provided in the following:
- (a) If the materials or supplies are obtained from a **DBE manufacturer**, 100% of the cost of the materials or supplies are counted toward DBE goals.
  - (1) For purposes of this paragraph 3.6.1(a), a manufacturer is a firm that operates or maintains a factory or establishment that produces, on the premises, the material, supplies, articles, or equipment required under the contract and of the general character described by the specifications.
- (b) If the materials or supplies are purchased from a **DBE regular dealer**, sixty percent (60%) of the cost of the materials or supplies is counted toward DBE goals.
  - (1) For purposes of this paragraph 3.6.1(b), a <u>regular dealer</u> is a firm that owns, operates, maintains a store, warehouse, or other establishment in which the materials, supplies, articles or equipment of the general character described by the specifications and required under the contract are bought, kept in stock, and regularly sold or leased to the public in the usual course of business.
  - (2) The firm must be an established, regular business that engages, as its principal business and under its own name, in the purchase and sale or lease of the products in question.
  - (3) A person may be a <u>regular dealer</u> in such bulk items as petroleum products, steel, cement, gravel, stone, or asphalt without owning, operating, or maintaining a place of business as provided above if the person both owns and operates distribution equipment for the products. Any supplementing of regular dealers' own distribution equipment shall be by a long-term lease agreement and not on an ad hoc or contract-by-contract basis.
  - (4) Packagers, brokers, manufacturers' representatives, or other persons who arrange or expedite transactions are not regular dealers within the meaning of this paragraph 3.6.1(b).
- (c) With respect to materials or supplies purchased from a **DBE**, which is neither a manufacturer nor a regular dealer, only the entire amount of fees or commissions charged for assistance in the procurement of the materials and supplies, or fees or transportation charges for the delivery of materials or supplies required on a job site, is counted toward DBE goals, provided the fees are determined to be reasonable and not excessive as compared with fees customarily allowed for similar service. **However, any portion**

of the cost of the materials and supplies themselves do not count toward DBE goals.

# 4.0 TERMINATION OF DBE(s)

4.1 The Bidder/Proposer/Prime may not terminate for convenience, or any other reason, and then perform the work of the terminated subcontractor/subconsultant with its own forces or those of an affiliate without NJT's prior written consent. Refer to Article 4.3.

Submission of all REQUIRED FORMS is mandatory for the following Articles 4.2 through 4.5

# 4.2 ADDITION OF DBE(s)

- 4.2.1 Should the Bidder/Proposer/Prime wish to add a DBE not listed on Form A, a written request for the addition of a DBE(s) must be submitted by the Bidder/Proposer/Prime.
- 4.2.2 The Bidder/Proposer/Prime must receive written approval of the OBD **prior to** the addition of the DBE subcontractor/subconsultant in order for the addition to be credited toward the goal.

# 4.3 REPLACEMENT OR REMOVAL OF DBE(s)

- 4.3.1 When a Prime is considering replacing or removing a DBE due to performance issues, the OBD must be contacted as soon as possible.
- 4.3.2 Request for DBE replacement or removal may be made under the following conditions:
  - 1) The DBE materially fails to successfully perform the contract tasks.
  - 2) Under unusual situations referenced in article 4.3.8.
- 4.3.3 A written request for replacement or removal of a DBE(s) listed on Form A, must be submitted by the Bidder/Proposer/Prime to the OBD with complete justification for the request. The process to follow such requests is as follows:
- (a) Written communications (over a period) from the Prime and/or NJT's PM/CM team to the DBE, notifying the DBE of its poor performance must be provided to the OBD.
- (b) The OBD will arrange a meeting with the DBE, the Prime, and a representative from Procurement and project management to discuss the specifics of the performance issue.
- (c) The DBE must provide a written plan identifying the efforts it will make to correct the deficiencies.
- (d) The Prime must provide the DBE with a minimum of 30 calendar days from acceptance of its plan to improve its performance. Throughout the 30-day window, the Prime and/or NJT PM/CM team must provide written communication to the DBE of any additional/continued performance issues, with a copy to the OBD.
- 4.3.4 The Bidder/Proposer/Prime must receive written approval of the OBD <u>prior to</u> replacement or removal of the DBE subcontractor/subconsultant can be made, <u>regardless of the reason for the replacement or removal</u>.
- 4.3.5 If the OBD issues written approval for the removal of a DBE(s), NJT will require a Bidder/Proposer/Prime to continue to demonstrate a good faith effort to replace the removed DBE to the extent needed to meet the contract goal established by NJT for the procurement.

These good faith efforts shall be directed at finding another DBE to perform at least the same amount of work under the contract as the removed DBE.

- 4.3.6 Failure to obtain approval from the OBD **prior to** replacing or removing a DBE shall result in the Bidder/Proposer/Prime being found in breach of the contract and subject to the appropriate remedies, audits and penalties. **Articles 5.6-5.7**
- 4.3.7 If NJT finds that the Bidder/Proposer/Prime upon submission of its bid/proposal committed itself to the goal in good faith, the Bidder/Proposer/Prime may, in "unusual situations", be permitted to substitute a DBE subcontractor(s)/subconsultant(s).
- 4.3.8 The term "unusual situations", includes, but is not limited to, the following circumstances:
- (a) Failure to qualify as a DBE, or maintain DBE certification status.
- (b) Death or physical disability of a key individual.
- (c) Dissolution, if a corporation or partnership.
- (d) Bankruptcy of the subcontractor/subconsultant, subject to applicable bankruptcy law, and only in instances where the bankruptcy affects the subcontractor/subconsultant's ability to perform.
- (e) Inability to obtain, or loss of, a license necessary for the performance of the particular category of work.
- (f) Failure or inability to comply with a requirement of law applicable to Primes or, subcontractors/subconsultants.

# 4.4 WITHDRAWN DBE(s)

- 4.4.1 When a DBE is unable to complete a subcontract (withdraws), for any reason, NJT will require a Bidder/Proposer/Prime to make a good faith effort to replace a withdrawn DBE at least to the extent needed to ensure that the Prime contractor is able to meet the contract goal established by NJT for the procurement. These good faith efforts shall be directed at finding another DBE to perform at least the same amount of work under the contract as the removed DBE.
- 4.4.2 The Bidder/Proposer/Prime is required to make a good faith effort to seek other DBE subcontractors/subconsultants in substitution of the original DBE. The good faith efforts described in article 2 are required in finding another DBE to perform at least the same amount of work under the contract as the DBE that was terminated, to the extent needed to meet the contract goal established for the procurement.

### 4.5 DECERTIFIED DBE(S)

- 4.5.1 If the Prime has reason to doubt that a proposed DBE is still eligible due to change in ownership, management, or size, the Prime shall, within 10 days of learning this information, notify NJT of that fact in writing.
- 4.5.2 If the subcontract has not been executed before the DBE's ineligibility occurs, the Prime will not receive credit toward the contract goal for the ineligible DBE. The Prime may continue to use the ineligible DBE, however, DBE participation credit will not be granted.

- (a) To the extent necessary to meet the assigned contract goal, the Prime will make a good faith effort to replace the ineligible DBE within 10 days after notification by the OBD. The OBD will support the Prime in its efforts to replace with an eligible DBE firm in order to meet the contract goal.
- 4.5.3 If the contract has been executed before the firm has been notified of its ineligibility, the Prime may continue to use the firm on the contract and receive credit toward its DBE goal for the duration of that particular phase of or option in the contract. This may not apply to future phases or options, which will be handled on a case-by-case basis at the sole discretion of the OBD.
- 4.5.4 *Exception*: If the DBEs ineligibility is caused solely by its having exceeded the size standard during the performance of the contract the Prime may continue to count its participation on that contract only toward the contract goals. This may not apply to future phases or options and will be subject to determination by the OBD.

### 5.0 AWARD OBLIGATIONS

- 5.1 The Prime must designate a **DBE Liaison Officer**. The liaison officer will be responsible to NJT regarding DBE subcontract matters.
- 5.1.1 If at any point during the contract's life, the Prime's DBE participation falls below the assigned goal, the Prime must identify additional work or new work items for which it will subcontract to DBEs to the extent necessary to meet the assigned goal. Any new scope of work issued to the Prime shall still be subject to the assigned goal.
- 5.1.2 The OBD will support the Prime in identifying current/future opportunities in the contract to meet the assigned contract goal.
- 5.1.3 Should the Prime seek a change that addresses the DBE's performance, or affects the work scope and/or compensation, the OBD must be notified, prior to implementation, for its review and approval of the changes as soon as possible. No change will be allowed without <u>prior review and approval</u> by the OBD. Failure to notify the office and obtain approval prior to a change shall result in breach of the contract and may be subject to the appropriate remedies, audits, and penalties.
- 5.1.4 Whenever NJT issues project change orders the goal may still apply; the OBD will determine if increased DBE participation will be required.
- 5.1.5 To ensure that all obligations under subcontracts awarded to DBEs are met NJT shall review the Prime's DBE involvement efforts during the performance of the contract.

### **5.2 POST AWARD DELIVERABLES**

- 5.2.1 After the execution of a contract with NJT, **signed copies** of subcontractor/subconsultant agreements between the Prime and DBE subcontractors must be submitted to the OBD <u>no later than 10 business days after the Prime's contract execution date</u>. The agreement between the Prime and DBE subcontractor shall remain firm for the duration of the contract.
- 5.2.2 The Prime shall provide a list of the anticipated job start date for all DBE subcontractors/subconsultants

- no later than two days after the initial pre-construction meeting.
- 5.2.3 **Certification of DBE(s) Payments** submit monthly with the Form E to the Manager of the OBD and with its monthly invoice submittal to NJT project manager of this project. **Refer to article 5.3.2**
- (a) The Prime will certify, <u>prior to the issuance of each progress payment by NJT</u>, that all DBE subs have been paid any amounts due on past due invoices from previous or current progress payments.
- 5.2.4 Form E (Contractor's Monthly DBE Payment Report & DBE Payment Certification Voucher) submit monthly to the Manager of the OBD. Refer to articles 2.5h and 5.3.2.
  - Failure to submit this report on a monthly basis may result in breach of the contract and be subject to the appropriate remedies, penalties or liquidated damages as indicated in articles 5.6-5.7.
- 5.2.5 Form E1 (DBE Prime's Monthly Payment Report) (For DBE Prime Only) submit monthly to the Manager of the OBD.
- 5.2.6 Form E2 (DBE's Monthly Payment Report) Refer to article 2.5j
- (a) Forms E/E1 and E2 will be reviewed monthly to determine compliance with the assigned DBE goal, the subcontractor prompt payment regulation, and the DBE Program.
- (b) Attainment of goals will be monitored and based upon actual payments <u>received</u> by the DBE. Failure to submit Form E/E1 may result in suspension of payments or such other remedies as provided in article 5.6. If at any time, NJT has reason to believe that any person or firm has willfully and knowingly provided incorrect information or made false statements, it shall refer the matter to the USDOT for inquiry.

### 5.3 PROMPT PAYMENT OF INVOICES TO SUBCONTRACTORS

- 5.3.1 The Prime must pay each subcontractor under this contract for satisfactory performance of its work **no** later than ten (10) days from the receipt of each payment the Prime receives from NJT for the subcontractor's work. Failure to comply with this requirement shall result in breach of the contract and shall be subject to the appropriate remedies as determined by OBD.
- 5.3.2 In accordance with 49 CFR 26.29, the Prime shall certify, prior to the issuance of a progress payment by <a href="NJT">NJT</a> that all subcontractors have been paid any amounts due on past due invoices (greater than 30 days) from previous or current progress payments. The Prime must submit the Certification of DBE(s) Payments with its monthly invoice submittals to NJT project manager, and with its Form E to the OBD.
- 5.3.3 The Prime will not be reimbursed for work performed by subcontractors/subconsultants unless and until the Prime ensures that the subs are promptly paid for the work performed. Alternatively, the Prime shall certify that a valid basis exists under the terms of the subcontractor's/subconsultant's or supplier's contract to withhold payment from the subcontractor/subconsultant and therefore payment is withheld.
- 5.3.4 If the Prime withholds payment from the subcontractor/subconsultant, the Prime shall provide to the subcontractor/subconsultant or supplier written notice thereof. The notice shall detail the reason for withholding payment and state the amount of the payment withheld. If a performance/payment bond has been provided under this contract, the Prime shall send a copy of the notice to the surety providing the

- bond for the Prime. A copy of the notice shall also be submitted to NJT with the certification that payments are being withheld.
- 5.3.5 If withholding payment is due to the Prime's failure to promptly pay the DBE in accordance with the prompt payment of invoices and/or retainage clauses, the OBD may request proof of payment to DBE(s) for delinquent invoices and/or retainage in order to issue release of payment to Prime.
- 5.3.6 Failure to comply with the above shall result in breach of the contract and may be subject to the appropriate penalties. **See article 5.6**

### 5.4 SUBCONTRACTOR PAYMENT DISPUTE RESOLUTION

- 5.4.1 The Prime is required to notify the OBD of its intention to withhold payment from a DBE as soon as possible and in advance of taking action. Should the Prime provide notice and proceed to withhold payment from any subcontractor/subconsultant or supplier due to a performance issue or unapproved work performed, an OBD representative shall make an effort to resolve the dispute.
- (a) OBD's efforts shall be limited to meeting with the Prime and the subcontractor/ subconsultant, and reviewing the relevant facts with both parties.
- (b) OBD will not act as a decider of fact nor will OBD direct a settlement to the dispute.
- (c) Any OBD effort is solely intended to assist the parties in understanding their respective positions and to encourage a reasonable resolution of the dispute. The Prime is required to send written notification of the above to the OBD immediately.
- 5.4.2 Should payments be withheld that are not related to the previous items mentioned, and/or a determination can be made that the withholding of payments violates the prompt payment clause, NJT may execute the appropriate remedies in accordance with article 5.6.

# 5.5 PROMPT PAYMENT OF SUBCONTRACTOR RETAINAGE (FOR CONSTRUCTION CONTRACTS ONLY)

- 5.5.1 The Prime must include a contract clause in the subcontractor agreement obligating the Prime to pay <u>all</u> retainage owed to the subcontractor/subconsultant for satisfactory completion of the accepted scope of work no later than 15 days after the DBE subcontractor' s/subconsultant' s work is satisfactorily completed.
- 5.5.2 Only subcontractors/subconsultants whose work has been 100% completed, including all punch list work or remaining work, and who have supplied closeout documents shall be eligible for release of retainage. Any delay or postponement of payment from the above referenced time frame may occur only for good cause <u>following written approval of NJT</u>.
- 5.5.3 NJT may agree to release an equivalent amount of Prime retainage provided that:
- (a) There is no offsetting claims from NJT (including, but not limited to, liquidated damages), other subcontractors/subconsultants, material men, or workers;
- (b) None of the other reasons to withhold payments specified under the Prime contract exists.
- 5.5.4 Prior to release of the Prime's retainage, the Prime shall provide to NJT executed copies of the following

  DBE Requirements for Federal Procurement Activities

subcontractor closeout documents, (shown in Appendix A of the contract) as appropriate:

- (a) Consent of Surety to Final Payment to the Subcontractor
- (b) Certificate of Amounts Due and Owing to Subcontractor Employees
- (c) Subcontractor Release of Claims
- (d) Subcontractor Release of Liens and a Certificate of Final Acceptance of Subcontractor Work
- 5.5.5 Notwithstanding NJT's release or partial release of retainage, nothing in this clause shall be deemed to constitute NJT's partial or final acceptance of the work, or any portion thereof, unless either a Certificate of Partial Acceptance or a Certificate of Final Acceptance has been executed by NJT, in the form(s).

# 5.6 Audit and Penalties

During the performance of any contract and for a period of up to three (3) years following completion of the contract work, NJ TRANSIT may conduct reviews for compliance with the requirements of the DBE Program. Such reviews may include, but not be limited to, the evaluation of monthly reports, desk audits and site visitations.

- 5.6.1 Where a Prime is found to be in breach of the requirements of the DBE Program during the performance of the contract, and does not promptly take corrective action, the following sanctions may be instituted (singularly, in any combination, and in addition to any other contractual remedies or otherwise provided by law):
- (a) The Prime may be ordered to stop work without penalty to NJT.
- **(b)** The contract may be terminated for breach.
- (c) Suspension or debarment proceedings may be commenced in accordance with New Jersey law.
- (d) The relevant performance bond may be enforced.
- (e) NJT may withhold payment of specific invoices.

#### 5.7 LIQUIDATED DAMAGES

- 5.7.1 Liquidated damages (LD) may be assessed when the Prime fails to meet the established DBE goal on the contract.
- 5.7.2 If the DBE goal is not met, and the Prime has not demonstrated a good faith effort to do so, NJT may elect to subtract from the Prime' s payment, as liquidated damages and not a penalty, the following: The amount equal to the difference (in dollars) between the total contract value multiplied by the assigned DBE goal percentage, (originally established or as subsequently modified) and the actual DBE participation percentage (total dollars paid to DBEs divided by total dollars paid to the Prime).
- 5.7.3 This may be withheld from a series of payments or from the Prime's final payment, depending on the size of the liquidated damage.
- 5.7.4 If the Prime's final payment is not sufficient to satisfy the LD in full, the balance shall be due and owing from the Prime and subject to repayment terms as determined by NJT. NJT shall waive liquidated damages DBE Requirements for Federal Procurement Activities

where good	causo is sh	nown for the	a deficiency	in DRE par	ticination u	non determi	nation by th	
where good	cause is si	iown for the	e deficiency	iii DDL pai	ricipation u	pon determ	nation by th	e ODD.

# APPENDIX I

#### **GLOSSARY**

<u>A Good Faith Effort</u>-the efforts employed by the bidder, which should be those that one could reasonably expect a bidder to take if the bidder were actively and aggressively trying to obtain DBE participation sufficient to meet the contract goal.

<u>Certification</u> - means the process by which a business is determined to be a bona fide DBE. Any business applying for DBE certification must complete the appropriate NJ Unified Certification Program Application. Certification Applications are available at the OBD.

#### Disadvantaged Business Enterprise or DBE - means a small business concern:

Which is at least 51 percent owned by one or more socially and economically disadvantaged individuals, or in the case of any publicly owned business, at least 51 percent of the stock of which is owned by one or more socially and economically disadvantaged individuals; and where one or more of the socially and economically disadvantaged owners controls management and daily business operations. A DBE shall not include a small business concern where that concern or a group of concerns controlled by the same socially and economically disadvantaged individual or individuals has annual average gross receipts in excess of \$22,410,000 over the previous three fiscal years or is not otherwise eligible as a small business as defined by the Small Business Administration in 13 CFR Part 121.

<u>DBE Goal</u> - means numerically expressed objectives for DBE participation on federal contracts Prime contractors are required to make a good faith effort to achieve to the extent necessary to meet the assigned DBE goal.

<u>DBE Sub-Prime</u> - means any 1<sup>st</sup> Tier DBE subcontractor/subconsultant listed on the Form A that will subcontract any portion of its subcontract/scope of work to a DBE(s) and/or non-DBE(s) firm(s).

<u>DBE Ineligibility</u> – means a firm's DBE status changes or ceases due to change in ownership, management, or size, etc.

**DBE Prime** - means the successful Bidder is a DBE firm and has a direct contract with NJT.

<u>DBE Trucking Firm</u> – owns and operates at least one fully licensed, insured, and operational truck used on the contract. Is responsible for the management and supervision of the entire trucking operation for which it is responsible on a particular contract.

<u>First Tier DBE</u> - refers to any DBE listed on the Bidder/Proposer/Prime's Form A and having a direct contract with the Prime.

<u>Joint Venture</u>—means an association of a DBE firm and one or more other firms to carry out a single, for-profit business enterprise, for which parties combine their property, capital, efforts, skills and knowledge, and in which the DBE is responsible for a distinct, clearly defined portion of the work of the contract and whose share in the capital contribution, control, management, risks, and profits of the joint venture are commensurate with its ownership interest.

<u>Prime</u> - means any contractor or consultant, including a DBE contractor/consultant, who enters into a direct contractual relationship with NJT.

Race-conscious Measure or Program - is one that is focused specifically on assisting only DBEs, including women-owned

DBEs.

<u>Race-neutral Measure or Program-</u> is one that is focused specifically on assisting all small businesses equally, including DBEs. Such activities include bonding, insurance, and technical assistance. For the purposes of this part, race-neutral includes gender-neutrality.

**Reasonable Bid Price** - means a price that shall be considered reasonable if the Bidder/Proposer/Prime would have been awarded the contract had the firm submitted the only bid.

<u>Regular Dealer</u> - means a firm that owns, operates, or maintains an establishment in which the materials or supplies required for the performance of a contract are bought, kept in stock and regularly sold to the public in the usual course of business.

The firm must engage in, as its principal business, and in its own name, the purchase and sale of products in question. Bulk items such as steel, cement and petroleum products need not be stocked, if the dealer owns or operates distribution equipment.

Note: Brokers and packagers are not regarded as regular dealers.

**Second Tier DBE** - refers to any DBE listed on the DBE Sub-Prime's Form AA.

<u>Subcontractor/ Subconsultant</u> - means any contractor/consultant, including suppliers, who enters into a contract issued by a Prime contractor.

<u>Transit Vehicle Manufacturer (TVM)</u> - is a manufacturer of vehicles used by NJT for the primary program purpose of public mass transportation (e.g., buses, railcars, vans). The term does not apply to firms, which rehabilitate old vehicles, or to manufacturers of locomotives or ferryboats. The term refers to distributors of or dealers in transit vehicles with respect to the requirements of 49 CFR Section 26.49.

<u>U.S. DOT</u> – means the U.S. Department of Transportation, including the Office of the Secretary, the Federal Highway Administration (FHWA), the Federal Transit Administration (FTA), and the Federal Railroad Administration (FRA).

<u>Withdrawn DBE</u> – a DBE withdraws, or drops out of its contract, or fails to complete its work on the contract for any reason.

### STATE OF NEW JERSEY

# EEO PROVISIONS FOR CONSTRUCTION CONTRACTS

#### I. BID REQUIREMENTS

This contract is subject to the provisions of N.J.S.A. 10:2-1 through 10:2-4 and N.J.S.A. 10:5-31 <u>et seq.</u> (P.L. 1975, c.127), and in accordance with the rules and regulations promulgated pursuant thereto, the bidder agrees to comply with the following:

At the time the signed contract is returned to NJ TRANSIT, the said bidder (contractor) shall submit to NJ TRANSIT and the New Jersey Department of Treasury, Division of Public Contracts Equal Employment Opportunity Compliance an Initial Project Workforce Report (Form AA-201) consisting of forms provided by NJ TRANSIT and completed by the bidder (contractor) in accordance with Subchapter 7 of the regulations promulgated by the Treasurer pursuant to P.L. 1975, c.127.

# II. EXECUTIVE ORDER 151 MANDATORY EEO AND AFFIRMATIVE ACTION REQUIREMENTS

It is the policy of the New Jersey Transit Corporation that its contracts should create a workforce that reflects the diversity of the State of New Jersey. Therefore, contractors engaged by the New Jersey Transit Corporation to perform under a construction contract shall put forth a good faith effort to engage in recruitment and employment practices that further the goal of fostering equal opportunities to minorities and women.

The contractor must demonstrate to the New Jersey Transit Corporation's satisfaction that a good faith effort was made to ensure that minorities and women have been afforded equal opportunity to gain employment under the New Jersey Transit Corporation's contract with the contractor. Payment may be withheld from a contractor's contract for failure to comply with these provisions.

Evidence of a "good faith effort" includes, but is not limited to:

- 1. The Contractor shall recruit prospective employees through the State Job bank website, managed by the Department of Labor and Workforce Development, available online at http://NJ.gov/JobCentralNJ.
- 2. The Contractor shall keep specific records of its efforts, including records of all individuals interviewed and hired, including the specific numbers of minorities and women.
- 3. The Contractor shall actively solicit and shall provide the New Jersey Transit Corporation with proof of solicitations for employment, including but not limited to advertisements in general circulation media, professional service publications and electronic media.
- 4. The Contractor shall provide evidence of efforts described at 2 above to the New Jersey Transit Corporation no less frequently than once every 12 months.
- 5. The Contractor shall comply with the requirements set forth at N.J.A.C. 17:27.

# III. SUBCONTRACTS; EQUAL EMPLOYMENT GOALS

The contractor agrees to incorporate these State of New Jersey EEO Provisions for Construction Contracts in its construction subcontracts.

The New Jersey Department of Treasury, Division of Public Contracts Equal Employment Opportunity Compliance (hereinafter "Division") shall individually establish the targeted minority and women employment goals for determining good faith equal employment opportunity efforts by each construction

- 1 - **10/09** 

contractor and subcontractor for each trade on each contract. The Division shall review the trades to be utilized during the completion of the work as reported on the Initial Project Workforce Report and determine the targeted employment goals based upon the number of qualified minorities and women available as reported by the New Jersey Department of Labor, Division of Planning and Research in its report, EEO Tabulations—Detailed Occupations by Race/Hispanic Groups.

In accordance with N.J.A.C. 17:27, Contractors and subcontractors are required to make a good faith effort to provide equal employment opportunity for minorities and women. Failure to make good faith efforts to provide equal employment opportunity for minorities and women may result in sanctions including fines/penalties, withholding of payment, termination of the contract, suspension/debarment or such other action as provided by law.

The contractor shall obtain the targeted employment goals from the Division's website at <a href="https://www.state.nj.us/treasury/contract\_compliance">www.state.nj.us/treasury/contract\_compliance</a> or request the employment goals from the Division of Public Contracts Equal Employment Opportunity Compliance after submitting the initial project workforce report to the Division of Public Contracts Equal Employment Opportunity Compliance.

The contractor shall then be responsible for the completion and submittal of a Monthly Project Workforce Report (Form AA-202) for the duration of this contract.

(NOTE: FOR THE PURPOSE OF THIS CONTRACT THE "PUBLIC AGENCY COMPLIANCE OFFICER" REFERENCED BELOW IS NJ TRANSIT'S VICE PRESIDENT OF DIVERSITY PROGRAMS AND THE "PUBLIC AGENCY" IS NJ TRANSIT.)

# IV. MANDATORY CONTRACT LANGUAGE

N.J.S.A. 10:5-31 et seq. and N.J.A.C. 17:27 MANDATORY AFFIRMATIVE ACTION LANGUAGE FOR CONSTRUCTION CONTRACTS

During the performance of this contract, the contractor agrees as follows:

The contractor or subcontractor, where applicable, will not discriminate against any employee or applicant for employment because of age, race, creed, color, national origin, ancestry, marital status, affectional or sexual orientation, gender identity or expression, disability, nationality or sex. Except with respect to affectional or sexual orientation and gender identity or expression, the contractor will ensure that equal employment opportunity is afforded to such applicants in recruitment and employment, and that employees are treated during employment, without regard to their age, race, creed, color, national origin, ancestry, marital status, affectional or sexual orientation, gender identity or expression, disability, nationality or sex. Such equal employment opportunity shall include, but not be limited to the following: employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided by the Public Agency Compliance Officer setting forth provisions of this nondiscrimination clause.

The contractor or subcontractor, where applicable will, in all solicitations or advertisements for employees placed by or on behalf of the contractor, state that all qualified applicants will receive consideration for employment without regard to age, race, creed, color, national origin, ancestry, marital status, affectional or sexual orientation, gender identity or expression, disability, nationality or sex.

The contractor or subcontractor, where applicable, will send to each labor union or representative of workers with which it has a collective bargaining agreement or other contract or understanding, a notice, to be provided by the agency contracting officer advising the labor union or workers' representative of the contractor's commitments under this act and shall post copies of the notice in conspicuous places available to employees and applicants for employment.

- 2 - **10/09** 

The contractor or subcontractor where applicable, agrees to comply with any regulations promulgated by the Treasurer pursuant to N.J.S.A. 10:5-31 et seq., as amended and supplemented from time to time and the Americans with Disabilities Act.

When hiring or scheduling workers in each construction trade, the contractor or subcontractor agrees to make good faith efforts to employ minority and women workers in each construction trade consistent with the targeted employment goal prescribed by N.J.A.C. 17:27-7.2; provided, however, that the Division may, in its discretion, exempt a contractor or subcontractor from compliance with the good faith procedures prescribed by the following provisions, A, B and C, as long as the Division is satisfied that the contractor or subcontractor is employing workers provided by a union which provides evidence, in accordance with standards prescribed by the Division, that its percentage of active "card carrying" members who are minority and women workers is equal to or greater than the targeted employment goal established in accordance with N.J.A.C. 17:27-7.2.

The contractor or subcontractor agrees that a good faith effort shall include compliance with the following procedures:

- (A). If the contractor or subcontractor has a referral agreement or arrangement with a union for a construction trade, the contractor or subcontractor shall, within three business days of the contract award, seek assurances from the union that it will cooperate with the contractor or subcontractor as it fulfills its affirmative action obligations under this contract and in accordance with the rules promulgated by the Treasurer pursuant to N.J.S.A. 10:5-31 et seq., as supplemented and amended from time to time, and the Americans with Disabilities Act. If the contractor or subcontractor is unable to obtain said assurances from the construction trade union at least five business days prior to the commencement of construction work. the contractor or subcontractor agrees to afford equal employment opportunities to minority and women workers directly, consistent with this chapter. If the contractor's or subcontractor's prior experience with a construction trade union, regardless of whether the union has provided said assurances, indicates a significant possibility that the trade union will not refer sufficient minority and women workers consistent with affording equal employment opportunities as specified in this chapter, the contractor or subcontractor agrees to be prepared to provide such opportunities to minority and women workers directly, consistent with this chapter, by complying with the procedures prescribed under (B) below; and the contractor or subcontractor further agrees to take said action immediately if it determines or is so notified by the Division that the union is not referring minority and women workers consistent with the equal employment opportunity goals set forth in this chapter.
- (B). If good faith efforts to meet targeted employment goals have not or cannot be met for each construction trade by adhering to the procedures of (A) above, or if the contractor does not have a referral agreement or arrangement with a union for a construction trade, the contractor or subcontractor agrees to take the following actions:
- (1). To notify the public agency compliance officer, the Division, and minority and women referral organizations listed by the Division pursuant to N.J.A.C. 17:27-5.3, of its workforce needs, and request referral of minority and women workers;
- (2). To notify any minority and women workers who have been listed with it as awaiting available vacancies;
- (3). Prior to commencement of work, to request that the local construction trade union refer minority and women workers to fill job openings, provided the contractor or subcontractor has a referral agreement or arrangement with a union for the construction trade;
- (4). To leave standing requests for additional referral to minority and women workers with the local construction trade union, provided the contractor or subcontractor has a referral agreement or arrangement with a union for the construction trade, the State Training and Employment Service and other approved referral sources in the area;

- 3 - **10/09** 

- (5). If it is necessary to lay off some of the workers in a given trade on the construction site, layoffs shall be conducted in compliance with the equal employment opportunity and non-discrimination standards set forth in this regulation, as well as with applicable Federal and State court decisions;
- (6). To adhere to the following procedure when minority and women workers apply or are referred to the contractor or subcontractor:
- (i) If said individuals have never previously received any document or certification signifying a level of qualification lower than that required in order to perform the work of the construction trade, the contractor or subcontractor shall in good faith determine the qualifications of such individuals. The contractor or subcontractor shall hire or schedule those individuals who satisfy appropriate qualification standards in conformity with the equal employment opportunity and non-discrimination principles set forth in this chapter. However, a contractor or subcontractor shall determine that the individual at least possesses the requisite skills, and experience recognized by a union, apprentice program or a referral agency, provided the referral agency is acceptable to the Division. If necessary, the contractor or subcontractor shall hire or schedule minority and women workers who qualify as trainees pursuant to these rules. All of the requirements, however, are limited by the provisions of (C) below.
- (ii). The name of any interested women or minority individual shall be maintained on a waiting list, and shall be considered for employment as described in paragraph (i) above, whenever vacancies occur. At the request of the Division, the contractor or subcontractor shall provide evidence of its good faith efforts to employ women and minorities from the list to fill vacancies.
- (iii). If, for any reason, said contractor or subcontractor determines that a minority individual or a woman is not qualified or if the individual qualifies as an advanced trainee or apprentice, the contractor or subcontractor shall inform the individual in writing of the reasons for the determination, maintain a copy of the determination in its files, and send a copy to the public agency compliance officer and to the Division.
- (7). To keep a complete and accurate record of all requests made for the referral of workers in any trade covered by the contract, on forms made available by the Division and submitted promptly to the Division upon request.
- (C). The contractor or subcontractor agrees that nothing contained in (B) above shall preclude the contractor or subcontractor from complying with the union hiring hall or apprenticeship policies in any applicable collective bargaining agreement or union hiring hall arrangement, and, where required by custom or agreement, it shall send journeymen and trainees to the union for referral, or to the apprenticeship program for admission, pursuant to such agreement or arrangement. However, where the practices of a union or apprenticeship program will result in the exclusion of minorities and women or the failure to refer minorities and women consistent with the targeted county employment goal, the contractor or subcontractor shall consider for employment persons referred pursuant to (B) above without regard to such agreement or arrangement; provided further, however, that the contractor or subcontractor shall not be required to employ women and minority advanced trainees and trainees in numbers which result in the employment of advanced trainees and trainees as a percentage of the total workforce for the construction trade, which percentage significantly exceeds the apprentice to journey worker ratio specified in the applicable collective bargaining agreement, or in the absence of a collective bargaining agreement, exceeds the ratio established by practice in the area for said construction trade. Also, the contractor or subcontractor agrees that, in implementing the procedures of (B) above it shall, where applicable, employ minority and women workers residing within the geographical jurisdiction of the union.

After notification of award, but prior to signing a construction contract, the contractor shall submit to the public agency compliance officer and the Division an initial project workforce report (Form AA 201) provided to the public agency by the Division for distribution to and completion by the contractor, in accordance with N.J.A.C. 17:27-7. The contractor also agrees to submit a copy of the Monthly Project Workforce Report once a month thereafter for the duration of this contract to the Division and to the public agency compliance officer.

- 4 - **10/09** 

The contractor agrees to cooperate with the public agency in the payment of budgeted funds, as is necessary, for on-the-job and/or off-the-job programs for outreach and training of minorities and women.

(D). The contractor and its subcontractors shall furnish such reports or other documents to the Division of Public Contracts Equal Employment Opportunity Compliance as may be requested by the Division from time to time in order to carry out the purposes of these regulations, and public agencies shall furnish such information as may be requested by the Division of Public Contracts Equal Employment Opportunity Compliance for conducting a compliance investigation pursuant to Subchapter 10 of the Administrative Code at N.J.A.C. 17:27.

- 5 - **10/09** 

TECHNICAL SPECIFICATIONS AND DRAWINGS FOR THIS PROJECT ARE NOT REPRINTED HERE DUE TO SIZE